

COMBINED PURCHASE & SALE AND LEASE AGREEMENTS

CONTRACTS INCLUDED

I. PURCHASE & SALE AGREEMENTS (PSA)

- 1. Offer, Agreement and Escrow Instructions for Purchase of Real Estate
- 2. Purchase and Sale Agreement Comparison Sheet
- 3. Offer, Agreement and Escrow Instructions for
- Purchase of Real Estate Vacant Land
- 4. Offer, Agreement and Escrow Instructions for Purchase of Residential Income Properties
 - a. 4.1 Chart for Authority
 - b. 4.2 Chart for Contingency

II. PSA ADDENDUM

- 5. Buyer's Notice Disapproving of Contingencies
- 6. Seller's Response to Buyer's Requests
- 7. Buyer's Reply to Seller's Response
- 8. Confidentiality Agreement
- 9. Counter Offer Seller
- 10. Counter Offer Buyer
- 11. Addendum to Purchase Agreement
- 12. Amendment to Purchase and Sale Agreement
- 13. Seller's Mandatory Disclosure Statement
- 14. Property Information Sheet
- 15. Lead-Based Paint Hazards Addendum
- 16. Estoppel Certificate By Lessee
- 17. Estoppel Certificate By Lessor
- 18. Release of Funds Addendum
- 19. Assignment and Assumption of Lessor's Interest in Lease

III. LEASES

- 20. Single-Tenant Net
- 21. Lease Comparison Sheet
- 22. Single-Tenant Gross
- 23. Land Lease Gross
- 24. Multi-Tenant Lease Net
- 25. Multi-Tenant Lease Gross
- 26. Multi-Tenant Office Lease Net
- 27. Multi-Tenant Office Lease Gross
- 28. Multi-Tenant Month to Month Lease Gross Short Form
- 29. Shopping Center Lease Net

IV. LEASE ADDENDUM

- 30. Chart of Authority
- 31. Property Information Sheet
- 32. Agency Disclosure
- 33. Rent Adjustment
- 34. Option to Extend
- 35. Notice of Election to Extend Term of Lease
- 36. Option to Purchase
- 37. Notice of Election to Purchase
- 38. Right of First Refusal to Purchase
- 39. Right of First Offer to Purchase
- 40. Right of First Refusal to Lease Additional Space
- 41. Guaranty of Lease
- 42. Lessor's Right to Terminate on Assignment or Sublease
- 43. Profit Sharing on Assignment or Sublease
- 44. Work Letter
- 45. Work Letter Office



Dated: ___

STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS FOR PURCHASE OF REAL ESTATE

(Non-Residential)

1.	Buyer.	/"P	ar") haraby offers to	nurchase the real	nronorty horoinaf	for described from the	owner thereof (" Seller ") (colle	ctivaly the "Barties" or
indiv				•		· ·	ction of the Buyer's Continger	• •
Clos and	i ng Date ") conditions	to be held by set forth in this a	("Escrow greement ("Agreem	Holder") whose ace	ldress is have the right to a	, Phone No	, Facsimile No under, but any such assignme	upon the terms
subs	1.2 The equent co	term " Date of Ag unteroffer thereto		erein shall be the d	ate when by exec	• •	fined in paragraph 20.2) of the	
2.	Property	d by both Parties.						
	2.1 The	real property ("P					tion) is located	l in the County of
	2.2 If th	ne legal descriptio	n of the Property is	not complete or is i	naccurate, this Ag	reement shall not be inv	alid and the legal description). shall be completed or
part duct cond	2.3 The of the pro ing, condulitioning ed	Property include: perty, as well as tl its, disconnects, li	s, at no additional control of the following items, it is fighting fixtures); telewith a fire spring are spring the spring sprin	ost to Buyer, the pe f any, owned by Sel ephone distribution	rmanent improver ler and at present systems (lines, ja	located on the Property cks and connections only	tter described. those items which pursuant electrical distribution system s	ns (power panel, bus utilating, air
	2.4 The	fire sprinkler mo	nitor: is owned	by Seller and includ	ded in the Purchas	e Price, is leased by	Seller, and Buyer will need to	negotiate a new lease
	2.5 Exc		Paragraph 2.3, the		_	there is no fire sprin 's personal property, fu	kler monitor. Initure and furnishings, and	all of
3.	Purchase	Price.						
		purchase price (" ike any not applic		be paid by Buyer to	Seller for the Pro	perty shall be	, payable as follows:	
	(a)	Cash down payr	ment, including the	Deposit as defined	in paragraph 4.3 (or if an all cash transaction	on, the Purchase Price):	
	(b)	Amount of "Nev	w Loan" as defined i	n paragraph 5.1, if a	any:			
	(c)	Trust") securing	the existing promis	sory note(s) (" Exist i	ing Note(s)"):	ing existing deed(s) of the Closing of approxi	rust ("Existing Deed(s) of mately:	
		until paid	(and/or the entire u	npaid balance is du	e on	erest at the rate of		
		Said Secor	nd Note is payable a	t per	month, including	interest at the rate of	% per annum	
	(d)	Buyer shall give		npaid balance is du st (" Purchase Mon o	e on ey Deed of Trust")	_). on the property, to secu		
		of Buyer to Selle	er described in para	graph 6 (" Purchase	Money Note") in	the amount of:		
Tota	Purchase	Price:						
	es includir	ng, but not limited		ng fees, and apprai	sal fees as a condi		of trust permits the beneficia e Property, Buyer agrees to pa	
4.	Deposits							
	4.1						e delivered by Broker to Escr	
		1 \		=		-	en delivered to Escrow Holde en delivered to Escrow Holde	
term Agre	inate this ement. Sl	transaction by giv	ring written notice o	f such election to E	scrow Holder whe	eupon neither Party sha	e period then Seller may elec all have any further liability to nall, upon request by Buyer, b	the other under this
Buye	4.2 Add	litional deposits: Within 5 busine	ess days after the Da	te of Agreement, B	uyer shall deposit	with Escrow Holder the	additional sum of	to be applied to the
Purc	hase Price	at the Closing.	•	_			proved or waived, Buyer shall	
Hold	er the add	litional sum of	to be ap	plied to the Purcha	se Price at the Clo	sing.	eller may notify Buyer, Escrov	·
	ng that, ui						ce, the Escrow shall be deeme	
	tered bank	in an interest be	aring account whose	term is appropriat	e and consistent v	vith the timing requirem	collectively the " Deposit "), in a ents of this transaction. The applicable instrument is rede	interest therefrom shall
spec	ified matu		eral Tax Identification				unt cannot be opened until Bu	
_								
	IALS				INIT	IALS		
© 2	U19 AIR (RE. All Rights F	keserved.				Last Edited: 10/	/13/2022 1:44 PM

4.4 Notwithstanding the foregoing, within 5 days after Escrow Holder receives the monies described in paragraph 4.1 above, Escrow Holder shall release \$100 of said monies to Seller as and for independent consideration for Seller's' execution of this Agreement and the granting of the contingency period to Buyer as herein provided. Such independent consideration is non-refundable to Buyer but shall be credited to the Purchase Price in the event that the purchase of the Property is completed. 4.5 Upon waiver of all of Buyer's contingencies the Deposit shall become non-refundable but applicable to the Purchase Price except in the event of a Seller breach, or in the event that the Escrow is terminated pursuant to the provisions of Paragraph 9.1(n) (Destruction, Damage or Loss) or 9.1(o) (Material Change). Financing Contingency. (Strike if not applicable) 5.1 This offer is contingent upon Buyer obtaining from an insurance company, financial institution or other lender, a commitment to lend to Buyer a sum equal % of the Purchase Price, on terms acceptable to Buyer. Such loan ("New Loan") shall be secured by a first deed of trust or mortgage on the to at least Property. If this Agreement provides for Seller to carry back junior financing, then Seller shall have the right to approve the terms of the New Loan. Seller shall have 7 days following receipt of the commitment setting forth the proposed terms of the New Loan to approve or disapprove of such proposed terms. If Seller fails to notify Escrow Holder, in writing, of the disapproval within said 7 days it shall be conclusively presumed that Seller has approved the terms of the New Loan. 5.2 If Buyer shall fail to notify its Broker, Escrow Holder and Seller, in writing within days following the Date of Agreement, that the New Loan has not been obtained, it shall be conclusively presumed that Buyer has either obtained said New Loan or has waived this New Loan contingency. 5.3 If Buyer shall notify its Broker, Escrow Holder and Seller, in writing, within the time specified in paragraph 5.2 hereof, that Buyer has not obtained said New Loan, this Agreement shall be terminated, and Buyer shall be entitled to the prompt return of the Deposit, plus any interest earned thereon, less only Escrow Holder and Title Company cancellation fees and costs, which Buyer shall pay. Seller Financing. (Purchase Money Note). (Strike if not applicable) 6.1 If Seller approves Buyer's financials (see paragraph 6.5) the Purchase Money Note shall provide for interest on unpaid principal at the rate of % per annum, with principal and interest paid as follows: _. The Purchase Money Note and Purchase Money Deed of Trust shall be on the current forms nmonly used by Escrow Holder, and be junior and subordinate only to the Existing Note(s) and/or the New Loan expressly called for by this Agreement 6.2 The Purchase Money Note and/or the Purchase Money Deed of Trust shall contain provisions regarding the following (see also paragraph 10.3 (b)) Prepayment. Principal may be prepaid in whole or in part at any time without penalty, at the option of the Buyer. Late Charge. A late charge of 6% shall be payable with respect to any payment of principal, interest, or other charges, not made within 10 days after (b) it is due. Due On Sale. In the event the Buyer sells or transfers title to the Property or any portion thereof, then the Seller may, at Seller's option, require the (c) entire unpaid balance of said Note to be paid in full. 6.3 If the Purchase Money Deed of Trust is to be subordinate to other financing, Escrow Holder shall, at Buyer's expense prepare and record on Seller's behalf a request for notice of default and/or sale with regard to each mortgage or deed of trust to which it will be subordinate. 6.4 WARNING: CALIFORNIA LAW DOES NOT ALLOW DEFICIENCY JUDGEMENTS ON SELLER FINANCING. IF BUYER ULTIMATELY DEFAULTS ON THE LOAN, SELLER'S SOLE REMEDY IS TO FORECLOSE ON THE PROPERTY. 6.5 Seller's obligation to provide financing is contingent upon Seller's reasonable approval of Buyer's financial condition. Buyer to provide a current financial statement and copies of its Federal tax returns for the last 3 years to Seller within 10 days following the Date of Agreement. Seller has 10 days following receipt of such documentation to satisfy itself with regard to Buyer's financial condition and to notify Escrow Holder as to whether or not Buyer's financial condition is acceptable. If Seller fails to notify Escrow Holder, in writing, of the disapproval of this contingency within said time period, it shall be conclusively presumed that Seller has approved Buyer's financial condition. If Seller is not satisfied with Buyer's financial condition or if Buyer fails to deliver the required documentation then Seller may notify Escrow Holder in writing that Seller Financing will not be available, and Buyer shall have the option, within 10 days of the receipt of such notice, to either terminate this transaction or to purchase the Property without Seller financing. If Buyer fails to notify Escrow Holder within said time period of its election to terminate this transaction then Buyer shall be conclusively presumed to have elected to purchase the Property without Seller financing. If Buyer elects to terminate, Buyer's Deposit shall be refunded less Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation. 7.1 Each Party acknowledges receiving a Disclosure Regarding Real Estate Agency Relationship, confirms and consents to the following agency relationships in this transaction with the following real estate broker(s) ("Brokers") and/or their agents ("Agent(s)"): is the broker of (check one): Late Seller; or both the Buyer and Seller (dual agent). License No. Seller's Brokerage Firm License No. the Seller's Agent (salesperson or broker associate); or both the Seller's Agent is (check one): Seller's Agent and the Buyer's Agent (dual agent). is the broker of (check one): $\ \ \ \ \$ the Buyer; or $\ \ \ \ \$ both the Buyer and Seller (dual agent). Buyer's Brokerage Firm License No. the Buyer's Agent (salesperson or broker associate); or 🔲 both the Buyer's Agent is (check one): Buyer's Agent License No. and the Seller's Agent (dual agent). The Parties acknowledge that other than the Brokers and Agents listed above, there are no other brokers or agents representing the Parties or due any fees and/or commissions under this Agreement. Buyer shall use the services of Buyer's Broker exclusively in connection with any and all negotiations and offers with respect to the Property for a period of 1 year from the date inserted for reference purposes at the top of page 1. 7.2 Buyer and Seller each represent and warrant to the other that he/she/it has had no dealings with any person, firm, broker, agent or finder in connection with the negotiation of this Agreement and/or the consummation of the purchase and sale contemplated herein, other than the Brokers and Agents named in paragraph 7.1, and no broker, agent or other person, firm or entity, other than said Brokers and Agents is/are entitled to any commission or finder's fee in connection with this transaction as the result of any dealings or acts of such Party. Buyer and Seller do each hereby agree to indemnify, defend, protect and hold the other harmless from and against any costs, expenses or liability for compensation, commission or charges which may be claimed by any broker, agent, finder or other similar party, other than said named Brokers and Agents by reason of any dealings or act of the indemnifying Party. 8.1 Upon acceptance hereof by Seller, this Agreement, including any counteroffers incorporated herein by the Parties, shall constitute not only the agreement of purchase and sale between Buyer and Seller, but also instructions to Escrow Holder for the consummation of the Agreement through the Escrow. Escrow Holder shall not prepare any further escrow instructions restating or amending the Agreement unless specifically so instructed by the Parties or a Broker herein. Subject to the reasonable approval of the Parties, Escrow Holder may, however, include its standard general escrow provisions. In the event that there is any conflict between the provisions of the Agreement and the provisions of any additional escrow instructions the provisions of the Agreement shall prevail as to the Parties and the Escrow Holden 8.2 As soon as practical after the receipt of this Agreement and any relevant counteroffers, Escrow Holder shall ascertain the Date of Agreement as defined in paragraphs 1.2 and 20.2 and advise the Parties and Brokers, in writing, of the date ascertained. Escrow Holder is hereby authorized and instructed to conduct the Escrow in accordance with this Agreement, applicable law and custom and practice of the community in which Escrow Holder is located, including any reporting requirements of the Internal Revenue Code. In the event of a conflict between the law of the state where the Property is located and the law of the state where the Escrow Holder is located, the law of the state where the Property is located shall prevail. 8.4 Subject to satisfaction of the contingencies herein described, Escrow Holder shall close this escrow (the "Closing") by recording a general warranty deed (a grant deed in California) and the other documents required to be recorded, and by disbursing the funds and documents in accordance with this Agreement. 8.5 Buyer and Seller shall each pay one-half of the Escrow Holder's charges and Seller shall pay the usual recording fees and any required documentary transfer taxes. Seller shall pay the premium for a standard coverage owner's or joint protection policy of title insurance. (See also paragraph 11.)

taxes. Seller shall pay the premium for a standard coverage owner's or joint protection policy of title insurance. (See also paragraph 11.)

8.6 Escrow Holder shall verify that all of Buyer's contingencies have been satisfied or waived prior to Closing. The matters contained in paragraphs 9.1 subparagraphs (b), (c), (d), (e), (g), (i), (n), and (o), 9.4, 12, 13, 14, 16, 18, 20, 21, 22, and 24 are, however, matters of agreement between the Parties only and are not

subparagraphs (b), (c), (d), (e), (g), (i), (n), and (o), 9.4, 12, 13, 14, 16, 18, 20, 21, 22, and 24 are, however, matters of agreement between the Parties only and are not instructions to Escrow Holder.

8.7 If this transaction is terminated for non-satisfaction and non-waiver of a Buyer's Contingency, as defined in paragraph 9.2 or disapproval of any other matter subject to Buyer's approval, then neither of the Parties shall thereafter have any liability to the other under this Agreement, except to the extent of a breach of any affirmative covenant or warranty in this Agreement. In the event of such termination, Buyer shall, subject to the provisions of paragraph 8.10, be promptly refunded all funds deposited by Buyer with Escrow Holder, less only the \$100 provided for in paragraph 4.4 and the Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation. If this transaction is terminated as a result of Seller's breach of this Agreement then Seller shall pay the Title Company and Escrow Holder cancellation fees and costs.

8.8 The Closing shall occur on the Expected Closing Date, or as soon thereafter as the Escrow is in condition for Closing; provided, however, that if the Closing

INITIALS

© 2019 AIR CRE. All Rights Reserved.

does not occur by the Expected Closing Date and said Date is not extended by mutual instructions of the Parties, a Party not then in default under this Agreement may notify the other Party, Escrow Holder, and Brokers, in writing that, unless the Closing occurs within 5 business days following said notice, the Escrow shall be deemed terminated without further notice or instructions.

8.9 Except as otherwise provided herein, the termination of Escrow shall not relieve or release either Party from any obligation to pay Escrow Holder's fees and costs or constitute a waiver, release or discharge of any breach or default that has occurred in the performance of the obligations, agreements, covenants or warranties contained therein.

8.10 If this Escrow is terminated for any reason other than Seller's breach or default, then as a condition to the return of Buyer's deposit, Buyer shall within 5 days after written request deliver to Seller, at no charge, copies of all surveys, engineering studies, soil reports, maps, master plans, feasibility studies and other similar items prepared by or for Buyer that pertain to the Property.

9. Contingencies to Closing.

9.1 IF, BEFORE EXPIRATION OF THE APPLICABLE TIME, BUYER FAILS TO PROVIDE ESCROW HOLDER WRITTEN NOTICE OF BUYER'S DISAPPROVAL OF ANY OF
BUYER'S CONTINGENCIES OR ANY OTHER MATTER THAT IS SUBJECT TO BUYER'S APPROVAL IN THIS AGREEMENT, THEN BUYER SHALL BE CONCLUSIVELY DEEMED TO HAVE SATISFIED SUCH BUYER'S CONTINGENCIES AND/OR APPROVED OF SUCH OTHER MATTERS. If a number of days is completed in any of the optional spaces
in subparagraphs 9.1 (a) through (m), then such number shall apply and override the pre-printed number, even if the pre-printed number is not stricken. The Closing
of this transaction is contingent upon the satisfaction or waiver of the following contingencies: (a) Disclosure. Seller shall make to Buyer, through Escrow, all of the applicable disclosures required by law (See AIR CRE ("AIR") standard form entitled "Seller's
Mandatory Disclosure Statement") and provide Buyer with a completed Property Information Sheet ("Property Information Sheet") concerning the Property, duly
executed by or on behalf of Seller in the current form or equivalent to that published by the AIR within 10 or days following the Date of Agreement.
Buyer has 10 days from the receipt of said disclosures to approve or disapprove the matters disclosed.
(b) Physical Inspection. Buyer has 10 or days following the receipt of the Property Information Sheet or the Date of Agreement, whichever
is later, to satisfy itself with regard to the physical aspects and size of the Property.
(c) Hazardous Substance Conditions Report. Buyer has 30 or days following the receipt of the Property Information Sheet or the Date of Agreement, whichever is later, to satisfy itself with regard to the environmental aspects of the Property. Seller recommends that Buyer obtain a Hazardous Substance
Conditions Report concerning the Property and relevant adjoining properties. Any such report shall be paid for by Buyer. A "Hazardous Substance" for purposes of
this Agreement is defined as any substance whose nature and/or quantity of existence, use, manufacture, disposal or effect, render it subject to Federal, state or local
regulation, investigation, remediation or removal as potentially injurious to public health or welfare. A "Hazardous Substance Condition" for purposes of this
Agreement is defined as the existence on, under or relevantly adjacent to the Property of a Hazardous Substance that would require remediation and/or removal
under applicable Federal, state or local law. (d) Soil Inspection. Buyer has 30 or days following the receipt of the Property Information Sheet or the Date of Agreement, whichever is
later, to satisfy itself with regard to the condition of the soils on the Property. Seller recommends that Buyer obtain a soil test report. Any such report shall be paid for
by Buyer. Seller shall provide Buyer copies of any soils report that Seller may have within 10 days following the Date of Agreement.
(e) Governmental Approvals. Buyer has 30 or days following the Date of Agreement to satisfy itself with regard to approvals and permits
from governmental agencies or departments which have or may have jurisdiction over the Property and which Buyer deems necessary or desirable in connection with
its intended use of the Property, including, but not limited to, permits and approvals required with respect to zoning, planning, building and safety, fire, police,
handicapped and Americans with Disabilities Act requirements, transportation and environmental matters. (f) Conditions of Title. Escrow Holder shall cause a current commitment for title insurance ("Title Commitment") concerning the Property issued by the
Title Company, as well as legible copies of all documents referred to in the Title Commitment (" Underlying Documents "), and a scaled and dimensioned plot showing
the location of any easements to be delivered to Buyer within 10 or days following the Date of Agreement. Buyer has 10 days from the receipt of the
Title Commitment, the Underlying Documents and the plot plan to satisfy itself with regard to the condition of title. The disapproval by Buyer of any monetary
encumbrance, which by the terms of this Agreement is not to remain against the Property after the Closing, shall not be considered a failure of this contingency, as
Seller shall have the obligation, at Seller's expense, to satisfy and remove such disapproved monetary encumbrance at or before the Closing. (g) Survey. Buyer has 30 or days following the receipt of the Title Commitment and Underlying Documents to satisfy itself with regard to
any ALTA title supplement based upon a survey prepared to American Land Title Association ("ALTA") standards for an owner's policy by a licensed surveyor, showing
the legal description and boundary lines of the Property, any easements of record, and any improvements, poles, structures and things located within 10 feet of either
side of the Property boundary lines. Any such survey shall be prepared at Buyer's direction and expense. If Buyer has obtained a survey and approved the ALTA title
supplement, Buyer may elect within the period allowed for Buyer's approval of a survey to have an ALTA extended coverage owner's form of title policy, in which even
Buyer shall pay any additional premium attributable thereto.
(h) Existing Leases and Tenancy Statements. Seller shall within 10 or days following the Date of Agreement provide both Buyer and Escrow
Holder with legible copies of all leases, subleases or rental arrangements (collectively, "Existing Leases") affecting the Property, and with a tenancy statement ("Estoppel Certificate") in the latest form or equivalent to that published by the AIR, executed by Seller and/or each tenant and subtenant of the Property. Seller shall
use its best efforts to have each tenant complete and execute an Estoppel Certificate. If any tenant fails or refuses to provide an Estoppel Certificate then Seller shall
complete and execute an Estoppel Certificate for that tenancy. Buyer has 10 days from the receipt of said Existing Leases and Estoppel Certificates to satisfy itself with
regard to the Existing Leases and any other tenancy issues.
(i) Owner's Association. Seller shall within 10 or days following the Date of Agreement provide Buyer with a statement and transfer package from any owner's association servicing the Property. Such transfer package shall at a minimum include: copies of the association's bylaws, articles of
incorporation, current budget and financial statement. Buyer has 10 days from the receipt of such documents to satisfy itself with regard to the association.
(j) Other Agreements. Seller shall within 10 or days following the Date of Agreement provide Buyer with legible copies of all other
agreements ("Other Agreements") known to Seller that will affect the Property after Closing. Buyer has 10 days from the receipt of said Other Agreements to satisfy
itself with regard to such Agreements.
(k) Financing. If paragraph 5 hereof dealing with a financing contingency has not been stricken, the satisfaction or waiver of such New Loan contingency.
(I) Existing Notes. If paragraph 3.1(c) has not been stricken, Seller shall within 10 or days following the Date of Agreement provide Buyer with legible copies of the Existing Notes, Existing Deeds of Trust and related agreements (collectively, "Loan Documents") to which the Property will remain subject
after the Closing. Escrow Holder shall promptly request from the holders of the Existing Notes a beneficiary statement ("Beneficiary Statement") confirming: (1) the
amount of the unpaid principal balance, the current interest rate, and the date to which interest is paid, and (2) the nature and amount of any impounds held by the
beneficiary in connection with such loan. Buyer has 10 or days following the receipt of the Loan Documents and Beneficiary Statements to satisfy itself
with regard to such financing. Buyer's obligation to close is conditioned upon Buyer being able to purchase the Property without acceleration or change in the terms
of any Existing Notes or charges to Buyer except as otherwise provided in this Agreement or approved by Buyer, provided, however, Buyer shall pay the transfer fee
referred to in paragraph 3.2 hereof. Likewise if Seller is to carry back a Purchase Money Note then Seller shall within 10 or days following the Date of
Agreement provide Buyer with a copy of the proposed Purchase Money Note and Purchase Money Deed of Trust. Buyer has 10 or days following the receipt of such documents to satisfy itself with regard to the form and content thereof.
(m) Personal Property. In the event that any personal property is included in the Purchase Price, Buyer has 10 or days following the Date of
Agreement to satisfy itself with regard to the title condition of such personal property. Seller recommends that Buyer obtain a UCC-1 report. Any such report shall be
paid for by Buyer. Seller shall provide Buyer copies of any liens or encumbrances affecting such personal property that it is aware of within 10 or days
following the Date of Agreement.
(n) Destruction, Damage or Loss. Subsequent to the Date of Agreement and prior to Closing there shall not have occurred a destruction of, or damage or
loss to, the Property or any portion thereof, from any cause whatsoever, which would cost more than \$10,000.00 to repair or cure. If the cost of repair or cure is \$10,000.00 or less, Seller shall repair or cure the loss prior to the Closing. Buyer shall have the option, within 10 days after receipt of written notice of a loss costing
more than \$10,000.00 to repair or cure, to either terminate this Agreement or to purchase the Property notwithstanding such loss, but without deduction or offset
against the Purchase Price. If the cost to repair or cure is more than \$10,000.00, and Buyer does not elect to terminate this Agreement, Buyer shall be entitled to any
insurance proceeds applicable to such loss. Unless otherwise notified in writing, Escrow Holder shall assume no such destruction, damage or loss has occurred prior
to Closing. (a) Material Change, Buyer shall have 10 days following receipt of written notice of a Material Change within which to satisfy itself with regard to such
(o) Material Change. Buyer shall have 10 days following receipt of written notice of a Material Change within which to satisfy itself with regard to such change. "Material Change" shall mean a substantial adverse change in the use, occupancy, tenants, title, or condition of the Property that occurs after the date of this
offer and prior to the Closing. Unless otherwise notified in writing, Escrow Holder shall assume that no Material Change has occurred prior to the Closing.
(p) Seller Performance. The delivery of all documents and the due performance by Seller of each and every undertaking and agreement to be performed
by Seller under this Agreement.
INITIALS

INITIALS
© 2019 AIR CRE. All Rights Reserved.
OFA-20.30, Revised 10-13-2022

- (q) Brokerage Fee. Payment at the Closing of such brokerage fee as is specified in this Agreement or later written instructions to Escrow Holder executed by Seller and Brokers ("Brokerage Fee"). It is agreed by the Parties and Escrow Holder that Brokers are a third party beneficiary of this Agreement insofar as the Brokerage Fee is concerned, and that no change shall be made with respect to the payment of the Brokerage Fee specified in this Agreement, without the written consent of Brokers.
- 9.2 The contingencies specified in subparagraphs 9.1(a) through (m) are for the benefit of, and may be waived by, Buyer, and are referred to collectively as "Buyer's Contingencies" and individually as a "Buyer's Contingency."
- 9.3 Buyer's timely and written disapproval or conditional approval of a Buyer's Contingency or any other matter that is subject to Buyer's approval in this Agreement shall constitute disapproval thereof ("Disapproved Item(s)"). Concurrent with notice of a Disapproved Item, Buyer may make a request to Seller regarding such Disapproved Item ("Buyer's Request"). If Buyer fails to make a timely and written Buyer's Request, then this Agreement shall terminate due to the non-satisfaction and non-waiver of a contingency. Seller may respond to a Buyer's Request within 10 days following Seller's receipt thereof ("Seller's Response"). Seller's acceptance of a Buyer's Request shall amend this Agreement accordingly. If Seller fails to provide a timely and written Seller's Response, then Seller's Response shall be deemed to be a rejection of Buyer's Request. Buyer may, within 10 days following the earlier of Buyer's receipt of a Seller's Response (which is not an acceptance of Buyer's Request) or the date of Seller's deemed rejection of a Buyer's Request ("Buyer's Reply Period"), reply to a Seller's Response ("Buyer's Reply") and elect to (i) terminate this Agreement due to the non-satisfaction and non-waiver of the applicable contingency, (ii) accept the Seller's Response in which event this Agreement shall be amended accordingly, or (iii) withdraw Buyer's Request and waive the Disapproved Item in which event Buyer shall accept the Property subject to the Disapproved Item. If Buyer fails to provide a timely and written Buyer's Reply, then Buyer shall be deemed to have elected to terminate this Agreement as of the end of the Buyer's Reply Period. The date Buyer accepts a Seller's Response or withdraws a Buyer's Request and waives a Disapproved Item shall be the date of Buyer's approval of the Disapproved Item. A Party shall provide to Escrow Holder copy of all notices of a Disapproved Item, Buyer's Request, Seller's Response and Buyer's Reply and Escrow Holder shall promptly provide copies thereof to the other Party. Unless the Parties in writing agree otherwise, if the Expected Closing Date is a specific calendar date and a Buyer's Reply Period expires after such specific calendar date, then notwithstanding paragraph 1.1, the Expected Closing Date shall be extended to be 3 business days after the earlier of the date Buyer withdraws a Buyer's Request and waives the applicable Disapproved Item or Buyer accepts the applicable Seller's Response.
- 9.4 The Parties acknowledge that extensive local, state and Federal legislation establish broad liability upon owners and/or users of real property for the investigation and remediation of Hazardous Substances. The determination of the existence of a Hazardous Substance Condition and the evaluation of the impact of such a condition are highly technical and beyond the expertise of Brokers. The Parties acknowledge that they have been advised by Brokers to consult their own technical and legal experts with respect to the possible presence of Hazardous Substances on the Property or adjoining properties, and Buyer and Seller are not relying upon any investigation by or statement of Brokers with respect thereto. The Parties hereby assume all responsibility for the impact of such Hazardous Substances upon their respective interests herein.

10. Documents and Other Items Required at or Before Closing.

- 10.1 Five days prior to the Closing date Escrow Holder shall obtain an updated Title Commitment concerning the Property from the Title Company and provide copies thereof to each of the Parties.
 - 10.2 Seller shall deliver to Escrow Holder in time for delivery to Buyer at the Closing:
 - (a) Grant or general warranty deed, duly executed and in recordable form, conveying fee title to the Property to Buyer.
 - (b) If applicable, the Beneficiary Statements concerning Existing Note(s).
- (c) If applicable, the Existing Leases and Other Agreements together with duly executed assignments thereof by Seller and Buyer. The assignment of Existing Leases shall be on the most recent Assignment and Assumption of Lessor's Interest in Lease form published by the AIR or its equivalent.
- (d) An affidavit executed by Seller to the effect that Seller is not a "foreign person" within the meaning of Internal Revenue Code Section 1445 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Internal Revenue Service such sum as is required by applicable Federal law with respect to purchases from foreign sellers.
- (e) If the Property is located in California, an affidavit executed by Seller to the effect that Seller is not a "nonresident" within the meaning of California Revenue and Tax Code Section 18662 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Franchise Tax Board such sum as is required by such statute.
 - (f) If applicable, a bill of sale, duly executed, conveying title to any included personal property to Buyer.
 - (g) If the Seller is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the sale of the Property.

 10.3 Buyer shall deliver to Seller through Escrow:
- (a) The cash portion of the Purchase Price and such additional sums as are required of Buyer under this Agreement shall be deposited by Buyer with Escrow Holder, by federal funds wire transfer, or any other method acceptable to Escrow Holder in immediately collectable funds, no later than 2:00 P.M. on the business day prior to the Expected Closing Date provided, however, that Buyer shall not be required to deposit such monies into Escrow if at the time set for the deposit of such monies Seller is in default or has indicated that it will not perform any of its obligations hereunder. Instead, in such circumstances in order to reserve its rights to proceed Buyer need only provide Escrow with evidence establishing that the required monies were available.
- (b) If a Purchase Money Note and Purchase Money Deed of Trust are called for by this Agreement, the duly executed originals of those documents, the Purchase Money Deed of Trust being in recordable form, together with evidence of fire insurance on the improvements in the amount of the full replacement cost naming Seller as a mortgage loss payee, and a real estate tax service contract (at Buyer's expense), assuring Seller of notice of the status of payment of real property taxes during the life of the Purchase Money Note.
 - (c) The Assignment and Assumption of Lessor's Interest in Lease form specified in paragraph 10.2(c) above, duly executed by Buyer.
 - (d) Assumptions duly executed by Buyer of the obligations of Seller that accrue after Closing under any Other Agreements.
 - (e) If applicable, a written assumption duly executed by Buyer of the loan documents with respect to Existing Notes.
 - (f) If the Buyer is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the purchase of the Property.
- 10.4 At Closing, Escrow Holder shall cause to be issued to Buyer a standard coverage (or ALTA extended, if elected pursuant to 9.1(g)) owner's form policy of title insurance effective as of the Closing, issued by the Title Company in the full amount of the Purchase Price, insuring title to the Property vested in Buyer, subject only to the exceptions approved by Buyer. In the event there is a Purchase Money Deed of Trust in this transaction, the policy of title insurance shall be a joint protection policy insuring both Buyer and Seller.

IMPORTANT: IN A PURCHASE OR EXCHANGE OF REAL PROPERTY, IT MAY BE ADVISABLE TO OBTAIN TITLE INSURANCE IN CONNECTION WITH THE CLOSE OF ESCROW SINCE THERE MAY BE PRIOR RECORDED LIENS AND ENCUMBRANCES WHICH AFFECT YOUR INTEREST IN THE PROPERTY BEING ACQUIRED. A NEW POLICY OF TITLE INSURANCE SHOULD BE OBTAINED IN ORDER TO ENSURE YOUR INTEREST IN THE PROPERTY THAT YOU ARE ACQUIRING.

11. Prorations and Adjustments.

- 11.1 Taxes. Applicable real property taxes and special assessment bonds shall be prorated through Escrow as of the date of the Closing, based upon the latest tax bill available. The Parties agree to prorate as of the Closing any taxes assessed against the Property by supplemental bill levied by reason of events occurring prior to the Closing. Payment of the prorated amount shall be made promptly in cash upon receipt of a copy of any supplemental bill.
- 11.2 Insurance. WARNING: Any insurance which Seller may have maintained will terminate on the Closing. Buyer is advised to obtain appropriate insurance to cover the Property.
- 11.3 Rentals, Interest and Expenses. Scheduled rentals, interest on Existing Notes, utilities, and operating expenses shall be prorated as of the date of Closing. The Parties agree to promptly adjust between themselves outside of Escrow any rents received after the Closing.
 - 11.4 Security Deposit. Security Deposits held by Seller shall be given to Buyer as a credit to the cash required of Buyer at the Closing.
- 11.5 Post Closing Matters. Any item to be prorated that is not determined or determinable at the Closing shall be promptly adjusted by the Parties by appropriate cash payment outside of the Escrow when the amount due is determined.
- 11.6 Variations in Existing Note Balances. In the event that Buyer is purchasing the Property subject to an Existing Deed of Trust(s), and in the event that a Beneficiary Statement as to the applicable Existing Note(s) discloses that the unpaid principal balance of such Existing Note(s) at the closing will be more or less than the amount set forth in paragraph 3.1(c) hereof ("Existing Note Variation"), then the Purchase Money Note(s) shall be reduced or increased by an amount equal to such Existing Note Variation. If there is to be no Purchase Money Note, the cash required at the Closing per paragraph 3.1(a) shall be reduced or increased by the amount of such Existing Note Variation.
- 11.7 Variations in New Loan Balance. In the event Buyer is obtaining a New Loan and the amount ultimately obtained exceeds the amount set forth in paragraph 5.1, then the amount of the Purchase Money Note, if any, shall be reduced by the amount of such excess.
- 11.8 Owner's Association Fees. Escrow Holder shall: (i) bring Seller's account with the association current and pay any delinquencies or transfer fees from Seller's proceeds, and (ii) pay any up front fees required by the association from Buyer's funds.

12. Representations and Warranties of Seller and Disclaimers.

12.1 Seller's warranties and representations shall survive the Closing and delivery of the deed for a period of 3 years, and any lawsuit or action based upon them

INITIALS

© 2019 AIR CRE. All Rights Reserved.

must be commenced within such time period. Seller's warranties and representations are true, material and relied upon by Buyer and Brokers in all respects. Seller hereby makes the following warranties and representations to Buyer and Brokers:

- (a) Authority of Seller. Seller is the owner of the Property and/or has the full right, power and authority to sell, convey and transfer the Property to Buyer as provided herein, and to perform Seller's obligations hereunder.
- (b) Maintenance During Escrow and Equipment Condition At Closing. Except as otherwise provided in paragraph 9.1(n) hereof, Seller shall maintain the Property until the Closing in its present condition, ordinary wear and tear excepted.
- (c) Hazardous Substances/Storage Tanks. Seller has no knowledge, except as otherwise disclosed to Buyer in writing, of the existence or prior existence on the Property of any Hazardous Substance, nor of the existence or prior existence of any above or below ground storage tank.
- (d) Compliance. Except as otherwise disclosed in writing, Seller has no knowledge of any aspect or condition of the Property which violates applicable laws, rules, regulations, codes or covenants, conditions or restrictions, or of improvements or alterations made to the Property without a permit where one was required, or of any unfulfilled order or directive of any applicable governmental agency or casualty insurance company requiring any investigation, remediation, repair, maintenance or improvement be performed on the Property.
- (e) Changes in Agreements. Prior to the Closing, Seller will not violate or modify any Existing Lease or Other Agreement, or create any new leases or other agreements affecting the Property, without Buyer's written approval, which approval will not be unreasonably withheld.
- (f) Possessory Rights. Seller has no knowledge that anyone will, at the Closing, have any right to possession of the Property, except as disclosed by this Agreement or otherwise in writing to Buyer.
 - (g) Mechanics' Liens. There are no unsatisfied mechanics' or materialmens' lien rights concerning the Property.
- (h) Actions, Suits or Proceedings. Seller has no knowledge of any actions, suits or proceedings pending or threatened before any commission, board, bureau, agency, arbitrator, court or tribunal that would affect the Property or the right to occupy or utilize same.
- (i) Notice of Changes. Seller will promptly notify Buyer and Brokers in writing of any Material Change (see paragraph 9.1(o)) affecting the Property that becomes known to Seller prior to the Closing.
- (j) No Tenant Bankruptcy Proceedings. Seller has no notice or knowledge that any tenant of the Property is the subject of a bankruptcy or insolvency proceeding.
 - (k) No Seller Bankruptcy Proceedings. Seller is not the subject of a bankruptcy, insolvency or probate proceeding.
- (I) Personal Property. Seller has no knowledge that anyone will, at the Closing, have any right to possession of any personal property included in the Purchase Price nor knowledge of any liens or encumbrances affecting such personal property, except as disclosed by this Agreement or otherwise in writing to Buyer.
- 12.2 Buyer hereby acknowledges that, except as otherwise stated in this Agreement, Buyer is purchasing the Property in its existing condition and will, by the time called for herein, make or have waived all inspections of the Property Buyer believes are necessary to protect its own interest in, and its contemplated use of, the Property. The Parties acknowledge that, except as otherwise stated in this Agreement, no representations, inducements, promises, agreements, assurances, oral or written, concerning the Property, or any aspect of the occupational safety and health laws, Hazardous Substance laws, or any other act, ordinance or law, have been made by either Party or Brokers, or relied upon by either Party hereto.
- 12.3 In the event that Buyer learns that a Seller representation or warranty might be untrue prior to the Closing, and Buyer elects to purchase the Property anyway then, and in that event, Buyer waives any right that it may have to bring an action or proceeding against Seller or Brokers, regarding said representation or warranty.
- 12.4 Any environmental reports, soils reports, surveys, and other similar documents which were prepared by third party consultants and provided to Buyer by Seller or Seller's representatives, have been delivered as an accommodation to Buyer and without any representation or warranty as to the sufficiency, accuracy, completeness, and/or validity of said documents, all of which Buyer relies on at its own risk. Seller believes said documents to be accurate, but Buyer is advised to retain appropriate consultants to review said documents and investigate the Property.

13. Possession.

Possession of the Property shall be given to Buyer at the Closing subject to the rights of tenants under Existing Leases.

14. Buyer's Entry

At any time during the Escrow period, Buyer, and its agents and representatives, shall have the right at reasonable times and subject to rights of tenants, to enter upon the Property for the purpose of making inspections and tests specified in this Agreement. No destructive testing shall be conducted, however, without Seller's prior approval which shall not be unreasonably withheld. Following any such entry or work, unless otherwise directed in writing by Seller, Buyer shall return the Property to the condition it was in prior to such entry or work, including the re-compaction or removal of any disrupted soil or material as Seller may reasonably direct. All such inspections and tests and any other work conducted or materials furnished with respect to the Property by or for Buyer shall be paid for by Buyer as and when due and Buyer shall indemnify, defend, protect and hold harmless Seller and the Property of and from any and all claims, liabilities, losses, expenses (including reasonable attorneys' fees), damages, including those for injury to person or property, arising out of or relating to any such work or materials or the acts or omissions of Buyer, its agents or employees in connection therewith.

15. Further Documents and Assurances.

The Parties shall each, diligently and in good faith, undertake all actions and procedures reasonably required to place the Escrow in condition for Closing as and when required by this Agreement. The Parties agree to provide all further information, and to execute and deliver all further documents, reasonably required by Escrow Holder or the Title Company.

16. Attorneys' Fees.

If any Party or Broker brings an action or proceeding (including arbitration) involving the Property whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees and costs. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred.

17. Prior Agreements/Amendments.

- 17.1 This Agreement supersedes any and all prior agreements between Seller and Buyer regarding the Property.
- 17.2 Amendments to this Agreement are effective only if made in writing and executed by Buyer and Seller.

18. Broker's Rights.

18.1 If this sale is not consummated due to the default of either the Buyer or Seller, the defaulting Party shall be liable to and shall pay to Brokers the Brokerage Fee that Brokers would have received had the sale been consummated. If Buyer is the defaulting party, payment of said Brokerage Fee is in addition to any obligation with respect to liquidated or other damages.

18.2 Upon the Closing, Brokers are authorized to publicize the facts of this transaction.

19. Notices.

19.1 Whenever any Party, Escrow Holder or Brokers herein shall desire to give or serve any notice, demand, request, approval, disapproval or other communication, each such communication shall be in writing and shall be delivered personally, by messenger, or by mail, postage prepaid, to the address set forth in this agreement or by facsimile transmission, electronic signature, digital signature, or email.

19.2 Service of any such communication shall be deemed made on the date of actual receipt if personally delivered, or transmitted by facsimile transmission, electronic signature, digital signature, or email. Any such communication sent by regular mail shall be deemed given 48 hours after the same is mailed. Communications sent by United States Express Mail or overnight courier that guarantee next day delivery shall be deemed delivered 24 hours after delivery of the same to the Postal Service or courier. If such communication is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.

19.3 Any Party or Broker hereto may from time to time, by notice in writing, designate a different address to which, or a different person or additional persons to whom, all communications are thereafter to be made.

tion of Offer.
tion of Offer

20.1 If this offer is not accepted by Seller on or before 5:00 P.M. according to the time standard applicable to the city of	on the date of
, it shall be deemed automatically revoked.	

20.2 The acceptance of this offer, or of any subsequent counteroffer hereto, that creates an agreement between the Parties as described in paragraph 1.2, shall be deemed made upon delivery to the other Party or either Broker herein of a duly executed writing unconditionally accepting the last outstanding offer or counteroffer.

INITIALS
© 2019 AIR CRE. All Rights Reserved.

21. LIQUIDATED DAMAGES. (This Liquidated Damages paragraph is applicable only if initialed by both Parties). THE PARTIES AGREE THAT IT WOULD BE IMPRACTICABLE OR EXTREMELY DIFFICULT TO FIX, PRIOR TO SIGNING THIS AGREEMENT, THE ACTUAL DAMAGES WHICH WOULD BE SUFFERED BY SELLER IF BUYER FAILS TO PERFORM ITS OBLIGATIONS UNDER THIS AGREEMENT. THEREFORE, IF, AFTER THE SATISFACTION OR WAIVER OF ALL CONTINGENCIES PROVIDED FOR THE BUYER'S BENEFIT, BUYER BREACHES THIS AGREEMENT, SELLER SHALL BE ENTITLED TO LIQUIDATED DAMAGES IN THE AMOUNT OF UPON PAYMENT OF SAID SUM TO SELLER, BUYER SHALL BE RELEASED FROM ANY FURTHER LIABILITY TO SELLER, AND ANY ESCROW CANCELLATION FEES AND TITLE COMPANY CHARGES SHALL BE PAID BY SELLER. BUYER'S InitialS Seller'S InitialS						
22.1 ANY CONTROVERSY AS SHALL BE DETERMINED BY BINDIN ITS COMMERCIAL ARBITRATION RUSUCH CONTROVERSY SHALL BE AR EXPERIENCE IN THE TYPE OF REAL ACCORDANCE WITH APPLICABLE LAGREEMENT AND ANY AMENDME PERMITTED IN ACCORDANCE WITH AWARD WITHIN 30 DAYS AFTER THE PARAGRAPH 16 HEREOF AND SHAIL OF THE DEPOSITS FOR ARBITRATO CROSS-EXAMINE WITNESSES, BUT AND LEGAL ARGUMENT AS THE AR COMPETENT JURISDICTION NOTW 22.2 BUYER'S RESORT TO OR BUYER FOR DAMAGES AND/OR SPEWHICH EVENT SUCH AWARD SHALL 22.3 NOTICE: BY INITIALING IN OF DISPUTES" PROVISION DECIDED HAVE THE DISPUTE LITIGATED IN A APPEAL, UNLESS SUCH RIGHTS ARE AGREEING TO THIS PROVISION, 10 TO THIS ARBITRATION PROVISION IN	22. ARBITRATION OF DISPUTES. (This Arbitration of Disputes paragraph is applicable only if initialed by both Parties.) 22.1 ANY CONTROVERSY AS TO WHETHER SELLER IS ENTITLED TO LIQUIDATED DAMAGES AND/OR BUYER IS ENTITLED TO THE RETURN OF THE DEPOSIT SHALL BE DETERMINED BY BINDING ARBITRATION ADMINISTERED BY THE JUDICIAL ARBITRATION & MEDIATION SERVICES, INC. ("JAMS") IN ACCORDANCE WITH ITS COMMERCIAL ARBITRATION RULES ("COMMERCIAL RULES"). ARBITRATION HEARINGS SHALL BE HELD IN THE COUNTY WHERE THE PROPERTY IS LOCATED. SUCH CONTROVERSY SHALL BE ARBITRATED BY A SINGLE ARBITRATOR, APPOINTED UNDER THE COMMERCIAL RULES WHO HAS HAD AT LEAST 5 YEARS OF EXPERIENCE IN THE TYPE OF REAL ESTATE THAT IS THE SUBJECT OF THIS AGREEMENT. THE ARBITRATOR SHALL HEAR AND DETERMINE SAID CONTROVERSY IN ACCORDANCE WITH APPLICABLE LAW OF THE JURISDICTION WHERE THE PROPERTY IS LOCATED, THE INTENTION OF THE PARTIES AS EXPRESSED IN THIS AGREEMENT AND ANY AMENDMENTS THERETO, AND UPON THE EVIDENCE PRODUCED AT AN ARBITRATION HEARING. PRE-ARBITRATION DISCOVERY SHALL BE PERMITTED IN ACCORDANCE WITH THE COMMERCIAL RULES OR STATE LAW APPLICABLE TO ARBITRATION PROCEEDINGS. THE ARBITRATION SHALL RENDER AN AWARD WITHIN 30 DAYS AFTER THE CONCLUSION OF THE HEARING, WHICH MAY INCLUDE ATTORNEYS' FEES AND COSTS TO THE PREVAILING PARTY PER PARAGRAPH 16 HEREOF AND SHALL BE ACCOMPANIED BY A REASONED OPINION. THE FAILURE OR REFUSAL OF A PARTY TO PAY SUCH PARTY'S REQUIRED SHARE OF THE DEPOSITS FOR ARBITRATOR COMPENSATION OR ADMINISTRATIVE CHARGES SHALL CONSTITUTE A WAIVER BY SUCH PARTY TO PRESENT EVIDENCE OR CROSS-EXAMINE WITNESSES, BUT SUCH WAIVER SHALL NOT ALLOW FOR A DEFAULT JUDGMENT MAY BE ENTERED ON THE AWARD IN ANY COURT OF COMPETENT JURISDICTION NOTWITHSTANDING THE FAILURE OF A PARTY DULY NOTHIFIED OF THE ARBITRATION HEARING TO APPEAR THEREAT. 22.2 BUYER'S RESORT TO OR PARTICIPATION IN SUCH ARBITRATION PROCEEDINGS SHALL NOT BAR SUIT IN A COURT OF COMPETENT JURISDICTION BY THE BUYER FOR DAMAGES AND/OR SPECIFIC PERFORMANCE. 22.3 NOTICE: BY INITIALING					
WE HAVE READ AND UNDERSTAND PROVISION TO NEUTRAL ARBITRATI	THE FOREGOING AND AGREE TO SUBMIT D ON.	ISPUTES ARISING OUT OF THE M.	ATTERS INCLUDED IN THE "ARBIT	RATION OF DISPUTES"		
		A	~			
	Buyer's Initials	26	ller's Initials			
Paragraphs 21 and 22 are each inco Agreement accomplished by means 23.2 Applicable Law. This Agreement accomplished by means 23.2 Applicable Law. This Agreement are of Essence. Time in 23.4 Counterparts. This Agreement is combined the signed signature paragraph 23.5 Waiver of Jury Trial. THE PROPERTY OR ARISING OUT OF THE 23.6 Conflict. Any conflict be typewritten or handwritten provision 23.7 1031 Exchange. Both Separty initiating an exchange shall be exchanging Party in the event that the state of the sta	eement shall be binding on the Parties with reporated into this Agreement only if initiale to of electronic signature or similar technology reement shall be governed by, and paragrag in the Parties hereto concerning this Agreement of the essence of this Agreement. The ement may be executed by Buyer and Sellentstrument. Escrow Holder, after verifying the ges on one of the counterparts, which shall the parties Hereby WAIVE THEIR RESPECTIVES AGREEMENT. The tween the printed provisions of this Agreements. Seller and Buyer must initial any and a seller and costs of such exchange. The cooperate she sale is delayed and/or that the sale other specifically indicated to the contrary, the workship in the sale of the sale is delayed and/or that the sale of the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and in the contrary, the workship is the sale is delayed and in the contrary, the workship is the sale is delayed and in the contrary in the sale is delayed and in the contrary.	d by both Parties at the time that sy shall be legal and binding. The shall be legal and binding. The shall be initiated in the countries of t	the Agreement is executed. Signal laws of the state in which the Property is locate shall be deemed an original, and except for the signatures, is authory ACTION OR PROCEEDING IN written provisions shall be controlled by the control of the signal with the provisions shall be controlled by the control of t	Property is located. d. all of which together corized and instructed VOLVING THE blled by the B1 exchange. Any chages to the		
	sture of a Real Estate Agency Relationship.	unad butha principles set forth i	n the applicable coefficies of the C	alifornia Civil Codo ac		
summarized in paragraph 24.2. 24.2 When entering into a disof agency relationship or represent transaction, as follows: (a) Seller's Agent. A S following affirmative obligations: (1 Seller: a. Diligent exercise of reason all facts known to the agent material Parties. An agent is not obligated to	agree that their relationship(s) shall be gove scussion with a real estate agent regarding a ation it has with the agent or agents in the teller's agent under a listing agreement with) To the Seller: A fiduciary duty of utmost canable skills and care in performance of the a ally affecting the value or desirability of the preveal to either Party any confidential info	real estate transaction, a Buyer ransaction. Buyer and Seller ack the Seller acts as the agent for the re, integrity, honesty, and loyalty gent's duties. b. A duty of hones property that are not known to, or	or Seller should from the outset of nowledge being advised by the Bone Seller only. A Seller's agent or in dealings with the Seller. (2) Total and fair dealing and good faith. Or within the diligent attention ar	understand what type rokers in this subagent has the othe Buyer and the c. A duty to disclose and observation of, the		
agent, even if by agreement the age the following affirmative obligation the Seller: a. Diligent exercise of rea disclose all facts known to the agen observation of, the Parties. An age affirmative duties set forth above. (c) Agent Representing	elling agent can, with a Buyer's consent, agrent may receive compensation for services r. (1) To the Buyer: A fiduciary duty of utmosonable skills and care in performance of the materially affecting the value or desirability is not obligated to reveal to either Party and Both Seller and Buyer. A real estate agent yer in a transaction, but only with the know	endered, either in full or in part is st care, integrity, honesty, and lo se agent's duties. b. A duty of ho sy of the property that are not kn my confidential information obtain, either acting directly or through	from the Seller. An agent acting of yalty in dealings with the Buyer. nest and fair dealing and good fa own to, or within the diligent atte ined from the other Party which one or more associate licensees.	only for a Buyer has (2) To the Buyer and ith. c. A duty to ention and does not involve the , can legally be the		
agent has the following affirmative either Seller or the Buyer. b. Other both Seller and Buyer, the agent manot limited to, facts relating to eithe including Seller's willingness to according to the self-buyer.	obligations to both the Seller and the Buyer duties to the Seller and the Buyer aduties to the Seller and the Buyer as stated by not, without the express permission of the Buyer's or Seller's financial position, motiept a price less than the listing price or Buyer on trelieve a Seller or Buyer from the response	: a. A fiduciary duty of utmost ca above in their respective sections e respective Party, disclose to the vations, bargaining position, or o r's willingness to pay a price grea	re, integrity, honesty and loyalty s (a) or (b) of this paragraph 24.2. e other Party confidential informa ther personal information that m ater than the price offered. (3) Th	in the dealings with (2) In representing ation, including, but ay impact price, ne above duties of the		

INITIALS

agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Buyer has the duty to exercise reasonable care to protect Buyer, including as to those facts about the Property which are known to Buyer or within Buyer's diligent attention and observation. Both Seller and Buyer should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.

- (d) Further Disclosures. Throughout this transaction Buyer and Seller may receive more than one disclosure, depending upon the number of agents assisting in the transaction. Buyer and Seller should each read its contents each time it is presented, considering the relationship between them and the real estate agent in this transaction and that disclosure. Buyer and Seller each acknowledge receipt of a disclosure of the possibility of multiple representation by the Broker representing that principal. This disclosure may be part of a listing agreement, buyer representation agreement or separate document. Buyer understands that Broker representing Buyer may also represent other potential buyers, who may consider, make offers on or ultimately acquire the Property. Seller understands that Broker representing Seller may also represent other sellers with competing properties that may be of interest to this Buyer. Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this transaction may be brought against Broker more than one year after the Date of Agreement and that the liability (including court costs and attorneys' fees), of any Broker with respect to any breach of duty, error or omission relating to this Agreement shall not exceed the fee received by such Broker pursuant to this Agreement; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such
- 24.3 Confidential Information. Buyer and Seller agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- Construction of Agreement. In construing this Agreement, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Agreement. Whenever required by the context, the singular shall include the plural and vice versa. This Agreement shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.

26. Additional Provisions.	
Additional provisions of this offer, if any, are as follows or are attached hereto by an addendum or addenda consisting of paragraphs	through
(If there are no additional provisions write "NONE".)	

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS AGREEMENT OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

- SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS AGREEMENT.
- RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PROPERTY. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING OF THE PROPERTY, THE INTEGRITY AND CONDITION OF ANY STRUCTURES AND OPERATING SYSTEMS, AND THE SUITABILITY OF THE PROPERTY FOR BUYER'S INTENDED USE.

WARNING: IF THE PROPERTY IS LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THIS AGREEMENT MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PROPERTY IS LOCATED.

NOTE:

- THIS FORM IS NOT FOR USE IN CONNECTION WITH THE SALE OF RESIDENTIAL PROPERTY.

2. IF EITHER PARTY IS A CORPORATION, IT IS RECOMMENDED THAT THIS A	AGREEMENT BE SIGNED BY TWO CORPORATE OFFICERS.
The undersigned Buyer offers and agrees to buy the Property on the terms and con	ditions stated and acknowledges receipt of a copy hereof.
•	Date
BROKER	BUYER
Attn:	Dec
Title:	By: Name Printed:
Address:	Title:
Phone:	Phone: Fax:
Fax:	Email:
Email: Federal ID No.:	
Broker DRE License #:	Ву:
Agent DRE License #:	Name Printed:
	Title:
	Phone:
	Fax:
	Email:
	Address:
	Federal ID No.:
27. Acceptance.	
	grees to sell the Property to Buyer on the terms and conditions therein specified.
27.2 In consideration of real estate brokerage service rendered by Brokers, Se	Seller's Broker % and Buyer's Broker %. This
Agreement shall serve as an irrevocable instruction to Escrow Holder to pay such Br	·
Closing.	σ · · · · · · · · · · · · · · · · · · ·
27.3 Seller acknowledges receipt of a copy hereof and authorizes Brokers to o	
NOTE: A PROPERTY INFORMATION SHEET IS REQUIRED TO BE DELIVERED TO BUYE	ER BY SELLER UNDER THIS AGREEMENT.
	Date:
BROKER	SELLER
Attn:	By:
Title:	Name Printed:
Address:	

INITIALS

© 2019 AIR CRE. All Rights Reserved. OFA-20.30, Revised 10-13-2022

INITIALS

Dhana	Title.
Phone: Fax:	Title: Phone:
Email:	Fax:
Federal ID No.:	Email:
Broker DRE License #:	
Agent's DRE License #:	By: Name Printed:
	Title:
	Phone:
	Fax: Email:
	Email:
	Address:
	Federal ID No.:
	AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.



STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS

FOR PURCHASE OF REAL ESTATE

The following is a list of some of the major differences between the various purchase agreements published by AIR CRE.

The 'Standard' Purchase Agreement v. Purchase Agreement for Vacant Land

Header changed from (Non-Residential) to (Vacant Land)

- **2.3** Eliminates all references to mechanical systems and fixtures leaving only a generic reference to "improvements" in place.
- **2.4** Eliminates the reference to fire sprinklers.
- **3.1** Gives two choices for determining price: 1. Fixed price or 2. Price per unit, such as price per square foot.
- **9.1e** Adds a warning that past uses may no longer be allowed. Buyer's responsibility to rezone and Seller must cooperate.
- **12.4** Feasibility studies has been added to the list of studies/reports that 3rd parties prepare and are passed from Seller to Buyer.

The 'Standard' Purchase Agreement v. Purchase Agreement for Residential Income Properties

Header changed from (Non-Residential) to (Do Not Use for Properties Containing Less Than Five Units)

- **2.3** List of equipment has been modified to reflect differences between commercial and residential properties.
- **9.1a** Lead based paint has been added as part of the mandatory disclosures in certain circumstances.

- **9.1b** A termite inspection/report to be paid for by Buyer has been added.
- **9.1i** Discusses Seller providing Buyer with statement and transfer package in the event that an Owner's Association exists.
- **10.2i** Adds a requirement that Seller provide the Buyer with all keys to the Property.
- **10.4** Again discusses keys.
- **23.8** Adds a provision regarding the fact that the Property is being sold in compliance with Equal Housing Opportunity laws.

Warning #2 Modifies the standard warning so as to also discuss Rent Control and Termites.

Note #1 This form is NOT to be used with properties containing less than 5 units. Residential transactions involving 1-4 units is governed by a different set of laws. As a consequence, we decided to avoid those issues and let CAR deal with them.

NOTE: The above is not intended to be a complete list of all of the differences between the various purchase agreements.



STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS FOR PURCHASE OF REAL ESTATE

(Vacant Land)

Dated:	
1. Buyer.	reinafter described from the owner thereof ("Caller") (collectively the "Dartier"
1.1, ("Buyer") hereby offers to purchase the real property, he individually, a "Party"), through an escrow ("Escrow") to close 30 or ("Escrow Holder") whose address is	
and conditions set forth in this agreement ("Agreement"). Buyer shall have the righ	
Buyer of Buyer's obligations herein unless Seller expressly releases Buyer. 1.2 The term "Date of Agreement" as used herein shall be the date when by	
subsequent counteroffer thereto, Buyer and Seller have reached agreement in writiterms accepted by both Parties.	, , , , ,
2. Property.	
2.1 The real property ("Property") that is the subject of this offer consists of	
, is commonly known as (street address, city, state, zip)	and is legally described as: (APN:
corrected to meet the requirements of ("Title Company"), which sha	Il issue the title policy hereinafter described.
2.3 The Property includes, at no additional cost to Buyer, the permanent impart of the property, as well as the following items, if any, owned by Seller and at property.	provements thereon, including those items which pursuant to applicable law are a resent located on the Property: (collectively, the "Improvements")
2.4 Except as provided in Paragraph 2.3, the Purchase Price does not include	
which shall be removed by Seller prior to Closing.	
3. Purchase Price.	
3.1 The purchase price (" Purchase Price ") to be paid by Buyer to Seller for the will be determined based on a per unit cost instead of a fixed price)	
	re number of units shall be based on a calculation of total area of the Property as
certified to the Parties by a licensed surveyor in accordance with paragraph 9.1(g). It calculation: . The Purchase Price shall be payable as follows:	
(Strike any not applicable)	
(a) Cash down payment, including the Deposit as defined in paragraph	4.3 (or if an all cash transaction, the Purchase Price):
(b) Amount of "New Loan" as defined in paragraph 5.1, if any:	
(c) Buyer shall take title to the Property subject to and/or assume the	following existing deed(s) of trust ("Existing Deed(s) of
Trust") securing the existing promissory note(s) ("Existing Note(s)" (i) An Existing Note ("First Note") with an unpaid principal balan):
Said First Note is payable at per month, includi	ng interest at the rate of % per annum
until paid (and/or the entire unpaid balance is due on	
(ii) An Existing Note ("Second Note") with an unpaid principal ba	lance as of the Closing of approximately:
Said Second Note is payable at per month, incl	uding interest at the rate of% per
annum until paid (and/or the entire unpaid balance is due on).
(d) Buyer shall give Seller a deed of trust ("Purchase Money Deed of T of Buyer to Seller described in paragraph 6 ("Purchase Money Note	
3.2 If Buyer is taking title to the Property subject to, or assuming, an Existing of fees including, but not limited to, points, processing fees, and appraisal fees as a maximum of 1.5% of the unpaid principal balance of the applicable Existing Note.	Deed of Trust and such deed of trust permits the beneficiary to demand paymen condition to the transfer of the Property, Buyer agrees to pay such fees up to a
4. Deposits.	
	payable to Escrow Holder, to be delivered by Broker to Escrow Holder within 2 or
	ne executed Agreement has been delivered to Escrow Holder, or within 2 or
business days after both Parties have executed this Agreement and the	ne executed Agreement has been delivered to Escrow Holder Buyer shall deliver to
Escrow Holder a check in the sum of If said check is not received by terminate this transaction by giving written notice of such election to Escrow Holde	
Agreement. Should Buyer and Seller not enter into an agreement for purchase and	
Buyer. 4.2 Additional deposits:	
	posit with Escrow Holder the additional sum of to be applied to the
Purchase Price at the Closing. (b) Within 5 husiness days after the contingencies discussed in paragram	aph 9.1 (a) through (m) are approved or waived, Buyer shall deposit with Escrow
Holder the additional sum of to be applied to the Purchase Price at the	he Closing.
(c) If an Additional Deposit is not received by Escrow Holder within the writing that, unless the Additional Deposit is received by Escrow Holder within 2 bu.	e time period provided then Seller may notify Buyer, Escrow Holder, and Brokers, siness days following said notice, the Escrow shall be deemed terminated without
further notice or instructions.	•
chartered bank in an interest bearing account whose term is appropriate and consist	_ ·
accrue to the benefit of Buyer, who hereby acknowledges that there may be penalti specified maturity. Buyer's Federal Tax Identification Number is NO	
Identification Number is provided.	
4.4 Notwithstanding the foregoing, within 5 days after Escrow Holder received	es the monies described in paragraph 4.1 above, Escrow Holder shall release \$100
	
INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	Last Edited: 10/13/2022 1:44 PM

of said monies to Seller as and for independent consideration for Seller's' execution of this Agreement and the granting of the contingency period to Buyer as herein provided. Such independent consideration is non-refundable to Buyer but shall be credited to the Purchase Price in the event that the purchase of the Property is completed.

completed.
4.5 Upon waiver of all of Buyer's contingencies the Deposit shall become non-refundable but applicable to the Purchase Price except in the event of a Seller breach, or in the event that the Escrow is terminated pursuant to the provisions of Paragraph 9.1(n) (Destruction, Damage or Loss) or 9.1(o) (Material Change).

5. Financing Contingency. (Strike if not applicable)

- 5.2 If Buyer shall fail to notify its Broker, Escrow Holder and Seller, in writing within ______ days following the Date of Agreement, that the New Loan has not been obtained, it shall be conclusively presumed that Buyer has either obtained said New Loan or has waived this New Loan contingency.
- 5.3 If Buyer shall notify its Broker, Escrow Holder and Seller, in writing, within the time specified in paragraph 5.2 hereof, that Buyer has not obtained said New Loan, this Agreement shall be terminated, and Buyer shall be entitled to the prompt return of the Deposit, plus any interest earned thereon, less only Escrow Holder and Title Company cancellation fees and costs, which Buyer shall pay.

6. Seller Financing. (Purchase Money Note). (Strike if not applicable)

- 6.1 If Seller approves Buyer's financials (see paragraph 6.5) the Purchase Money Note shall provide for interest on unpaid principal at the rate of _________. The Purchase Money Note and Purchase Money Deed of Trust shall be on the current forms commonly used by Escrow Holder, and be junior and subordinate only to the Existing Note(s) and/or the New Loan expressly called for by this Agreement.
 - 6.2 The Purchase Money Note and/or the Purchase Money Deed of Trust shall contain provisions regarding the following (see also paragraph 10.3 (b)):
 - (a) Prepayment. Principal may be prepaid in whole or in part at any time without penalty, at the option of the Buyer.
- (b) Late Charge. A late charge of 6% shall be payable with respect to any payment of principal, interest, or other charges, not made within 10 days after it is due.
- (c) Due On Sale. In the event the Buyer sells or transfers title to the Property or any portion thereof, then the Seller may, at Seller's option, require the entire unpaid balance of said Note to be paid in full.
- 6.3 If the Purchase Money Deed of Trust is to be subordinate to other financing, Escrow Holder shall, at Buyer's expense prepare and record on Seller's behalf a request for notice of default and/or sale with regard to each mortgage or deed of trust to which it will be subordinate.
- 6.4 WARNING: CALIFORNIA LAW DOES NOT ALLOW DEFICIENCY JUDGEMENTS ON SELLER FINANCING. IF BUYER ULTIMATELY DEFAULTS ON THE LOAN, SELLER'S SOLE REMEDY IS TO FORECLOSE ON THE PROPERTY.
- 6.5 Seller's obligation to provide financing is contingent upon Seller's reasonable approval of Buyer's financial condition. Buyer to provide a current financial statement and copies of its Federal tax returns for the last 3 years to Seller within 10 days following the Date of Agreement. Seller has 10 days following receipt of such documentation to satisfy itself with regard to Buyer's financial condition and to notify Escrow Holder as to whether or not Buyer's financial condition is acceptable. If Seller fails to notify Escrow Holder, in writing, of the disapproval of this contingency within said time period, it shall be conclusively presumed that Seller has approved Buyer's financial condition. If Seller is not satisfied with Buyer's financial condition or if Buyer fails to deliver the required documentation then Seller may notify Escrow Holder in writing that Seller Financing will not be available, and Buyer shall have the option, within 10 days of the receipt of such notice, to either terminate this transaction or to purchase the Property without Seller financing. If Buyer fails to notify Escrow Holder within said time period of its election to terminate this transaction then Buyer shall be conclusively presumed to have elected to purchase the Property without Seller financing. If Buyer elects to terminate, Buyer's Deposit shall be refunded less Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation.

7. Real Estate Brokers.

	_	Regarding Real Estate Agency Relationship, confirms and consents to the following agency relationships in
this transaction with the following	g real estate broker(s) (" Br	okers") and/or their agents ("Agent(s)"):
Seller's Brokerage Firm	License No.	is the broker of (check one): the Seller; or both the Buyer and Seller (dual agent).
Seller's Agent	License No.	is (check one): the Seller's Agent (salesperson or broker associate); or both the Seller's Agent
and the Buyer's Agent (dual agent).	
Buyer's Brokerage Firm	License No.	is the broker of (check one): the Buyer; or both the Buyer and Seller (dual agent).
Buyer's Agent	License No.	_ is (check one): — the Buyer's Agent (salesperson or broker associate); or — both the Buyer's Agent
and the Seller's Agent (dual agent).	
_		gents listed above, there are no other brokers or agents representing the Parties or due any fees and/or
commissions under this Agreemen	nt. Buyer shall use the se	vices of Buyer's Broker exclusively in connection with any and all negotiations and offers with respect to
the Property for a period of 1 year	r from the date inserted for	or reference purposes at the top of page 1.

7.2 Buyer and Seller each represent and warrant to the other that he/she/it has had no dealings with any person, firm, broker, agent or finder in connection with the negotiation of this Agreement and/or the consummation of the purchase and sale contemplated herein, other than the Brokers and Agents named in paragraph 7.1, and no broker, agent or other person, firm or entity, other than said Brokers and Agents is/are entitled to any commission or finder's fee in connection with this transaction as the result of any dealings or acts of such Party. Buyer and Seller do each hereby agree to indemnify, defend, protect and hold the other harmless from and against any costs, expenses or liability for compensation, commission or charges which may be claimed by any broker, agent, finder or other similar party, other than said named Brokers and Agents by reason of any dealings or act of the indemnifying Party.

8. Escrow and Closing.

- 8.1 Upon acceptance hereof by Seller, this Agreement, including any counteroffers incorporated herein by the Parties, shall constitute not only the agreement of purchase and sale between Buyer and Seller, but also instructions to Escrow Holder for the consummation of the Agreement through the Escrow. Escrow Holder shall not prepare any further escrow instructions restating or amending the Agreement unless specifically so instructed by the Parties or a Broker herein. Subject to the reasonable approval of the Parties, Escrow Holder may, however, include its standard general escrow provisions. In the event that there is any conflict between the provisions of the Agreement and the provisions of any additional escrow instructions the provisions of the Agreement shall prevail as to the Parties and the Escrow Holder.
- 8.2 As soon as practical after the receipt of this Agreement and any relevant counteroffers, Escrow Holder shall ascertain the Date of Agreement as defined in paragraphs 1.2 and 20.2 and advise the Parties and Brokers, in writing, of the date ascertained.
- 8.3 Escrow Holder is hereby authorized and instructed to conduct the Escrow in accordance with this Agreement, applicable law and custom and practice of the community in which Escrow Holder is located, including any reporting requirements of the Internal Revenue Code. In the event of a conflict between the law of the state where the Property is located and the law of the state where the Property is located shall prevail.
- 8.4 Subject to satisfaction of the contingencies herein described, Escrow Holder shall close this escrow (the "Closing") by recording a general warranty deed (a grant deed in California) and the other documents required to be recorded, and by disbursing the funds and documents in accordance with this Agreement.
- 8.5 Buyer and Seller shall each pay one-half of the Escrow Holder's charges and Seller shall pay the usual recording fees and any required documentary transfer taxes. Seller shall pay the premium for a standard coverage owner's or joint protection policy of title insurance. (See also paragraph 11.)
- 8.6 Escrow Holder shall verify that all of Buyer's contingencies have been satisfied or waived prior to Closing. The matters contained in paragraphs 9.1 subparagraphs (b), (c), (d), (e), (g), (i), (n), and (o), 9.4, 12, 13, 14, 16, 18, 20, 21, 22, and 24 are, however, matters of agreement between the Parties only and are not instructions to Escrow Holder.
- 8.7 If this transaction is terminated for non-satisfaction and non-waiver of a Buyer's Contingency, as defined in Paragraph 9.2 or disapproval of any other matter subject to Buyer's approval, then neither of the Parties shall thereafter have any liability to the other under this Agreement, except to the extent of a breach of any affirmative covenant or warranty in this Agreement. In the event of such termination, Buyer shall, subject to the provisions of paragraph 8.10, be promptly refunded all funds deposited by Buyer with Escrow Holder, less only the \$100 provided for in paragraph 4.4 and the Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation. If this transaction is terminated as a result of Seller's breach of this Agreement then Seller shall pay the Title Company and Escrow Holder cancellation fees and costs.
- 8.8 The Closing shall occur on the Expected Closing Date, or as soon thereafter as the Escrow is in condition for Closing; provided, however, that if the Closing does not occur by the Expected Closing Date and said Date is not extended by mutual instructions of the Parties, a Party not then in default under this Agreement may

INITIALS

notify the other Party, Escrow Holder, and Brokers, in writing that, unless the Closing occurs within 5 business days following said notice, the Escrow shall be deemed terminated without further notice or instructions.

8.9 Except as otherwise provided herein, the termination of Escrow shall not relieve or release either Party from any obligation to pay Escrow Holder's fees and costs or constitute a waiver, release or discharge of any breach or default that has occurred in the performance of the obligations, agreements, covenants or warranties contained therein.

8.10 If this Escrow is terminated for any reason other than Seller's breach or default, then as a condition to the return of Buyer's deposit, Buyer shall within 5 days after written request deliver to Seller, at no charge, copies of all surveys, engineering studies, soil reports, maps, master plans, feasibility studies and other similar items prepared by or for Buyer that pertain to the Property.

9.1 IF, BEFORE EXPIRATION OF THE APPLICABLE TIME, BUYER FAILS TO PROVIDE ESCROW HOLDER WRITTEN NOTICE OF BUYER'S DISAPPROVAL OF ANY OF

9. Contingencies to Closing.

BUYER'S CONTINGENCIES OR ANY OTHER MATTER THAT IS SUBJECT TO BUYER'S APPROVAL II TO HAVE SATISFIED SUCH BUYER'S CONTINGENCIES AND/OR APPROVED OF SUCH OTHER MA	•
in subparagraphs 9.1 (a) through (m), then such number shall apply and override the pre-printe	
of this transaction is contingent upon the satisfaction or waiver of the following contingencies:	at number, even if the pre-printed number is necessioned.
(a) Disclosure. Seller shall make to Buyer, through Escrow, all of the applicable dis	sclosures required by law (See AIR CRE ("AIR") standard form entitled
"Seller's Mandatory Disclosure Statement") and provide Buyer with a completed Property Info	
duly executed by or on behalf of Seller in the current form or equivalent to that published by the	
Agreement. Buyer has 10 days from the receipt of said disclosures to approve or disapprove the	
(b) Physical Inspection. Buyer has 10 or days following the receipt	of the Property Information Sheet or the Date of Agreement, whichever
is later, to satisfy itself with regard to the physical aspects and size of the Property. (c) Hazardous Substance Conditions Report. Buyer has 30 or days f	following the receipt of the Droporty Information Shoot or the Date of
Agreement, whichever is later, to satisfy itself with regard to the environmental aspects of the Conditions Report concerning the Property and relevant adjoining properties. Any such report this Agreement is defined as any substance whose nature and/or quantity of existence, use, may regulation, investigation, remediation or removal as potentially injurious to public health or we Agreement is defined as the existence on, under or relevantly adjacent to the Property of a Hazunder applicable Federal, state or local law. (d) Soil Inspection. Buyer has 30 or days following the receipt of the later, to satisfy itself with regard to the condition of the soils on the Property. Seller recommendations are consistent with the soils on the property.	Property. Seller recommends that Buyer obtain a Hazardous Substance shall be paid for by Buyer. A "Hazardous Substance" for purposes of anufacture, disposal or effect, render it subject to Federal, state or local effect. A "Hazardous Substance Condition" for purposes of this zardous Substance that would require remediation and/or removal e Property Information Sheet or the Date of Agreement, whichever is
by Buyer. Seller shall provide Buyer copies of any soils report that Seller may have within 10 da (e) Governmental Approvals. Buyer has 30 or days following the Days	lys following the Date of Agreement. ate of Agreement to satisfy itself with regard to approvals and permits
from governmental agencies or departments which have or may have jurisdiction over the Propits intended use of the Property, including, but not limited to, permits and approvals required valuacion handicapped and Americans with Disabilities Act requirements, transportation and environment	with respect to zoning, planning, building and safety, fire, police, ntal matters.
NOTE: Past uses of the Property may no longer be allowed. In the event that the Property mus appropriate government agencies. Seller shall sign all documents Buyer is required to file in co development approvals.	nnection with rezoning, conditional use permits and/or other
(f) Conditions of Title. Escrow Holder shall cause a current commitment for title i Title Company, as well as legible copies of all documents referred to in the Title Commitment ('the location of any easements to be delivered to Buyer within 10 or days following Title Commitment, the Underlying Documents and the plot plan to satisfy itself with regard to the encumbrance, which by the terms of this Agreement is not to remain against the Property after Seller shall have the obligation, at Seller's expense, to satisfy and remove such disapproved mo	"Underlying Documents"), and a scaled and dimensioned plot showing ng the Date of Agreement. Buyer has 10 days from the receipt of the the condition of title. The disapproval by Buyer of any monetary r the Closing, shall not be considered a failure of this contingency, as
(g) Survey. Buyer has 30 or days following the receipt of the Title C	
any ALTA title supplement based upon a survey prepared to American Land Title Association ("A	
the legal description and boundary lines of the Property, any easements of record, and any imposide of the Property boundary lines. Any such survey shall be prepared at Buyer's direction and	d expense. If Buyer has obtained a survey and approved the ALTA title
supplement, Buyer may elect within the period allowed for Buyer's approval of a survey to have Buyer shall pay any additional premium attributable thereto.	e an ALIA extended coverage owner's form of title policy, in which event
(h) Existing Leases and Tenancy Statements. Seller shall within 10 or	days following the Date of Agreement provide both Ruyer and Escrow
Holder with legible copies of all leases, subleases or rental arrangements (collectively, "Existing ("Estoppel Certificate") in the latest form or equivalent to that published by the AIR, executed use its best efforts to have each tenant complete and execute an Estoppel Certificate. If any te complete and execute an Estoppel Certificate for that tenancy. Buyer has 10 days from the recregard to the Existing Leases and any other tenancy issues.	s Leases") affecting the Property, and with a tenancy statement by Seller and/or each tenant and subtenant of the Property. Seller shall nant fails or refuses to provide an Estoppel Certificate then Seller shall eipt of said Existing Leases and Estoppel Certificates to satisfy itself with
(i) Owner's Association. Seller shall within 10 or days following the	
package from any owner's association servicing the Property. Such transfer package shall at a rincorporation, current budget and financial statement. Buyer has 10 days from the receipt of s (j) Other Agreements. Seller shall within 10 or days following the lagreements ("Other Agreements") known to Seller that will affect the Property after Closing. Extractly with regard to such Agreements.	uch documents to satisfy itself with regard to the association. Date of Agreement provide Buyer with legible copies of all other Buyer has 10 days from the receipt of said Other Agreements to satisfy
 (k) Financing. If paragraph 5 hereof dealing with a financing contingency has not (l) Existing Notes. If paragraph 3.1(c) has not been stricken, Seller shall within 10 	- · · · · · · · · · · · · · · · · · · ·
with legible copies of the Existing Notes, Existing Deeds of Trust and related agreements (collection after the Closing. Escrow Holder shall promptly request from the holders of the Existing Notes amount of the unpaid principal balance, the current interest rate, and the date to which interest beneficiary in connection with such loan. Buyer has 10 or days following the rection with regard to such financing. Buyer's obligation to close is conditioned upon Buyer being able of any Existing Notes or charges to Buyer except as otherwise provided in this Agreement or appreferred to in paragraph 3.2 hereof. Likewise if Seller is to carry back a Purchase Money Note to Agreement provide Buyer with a copy of the proposed Purchase Money Note and Purchase Note Purchase Note Purchase Note Purchase Note Purchase Note	ctively, "Loan Documents") to which the Property will remain subject a beneficiary statement ("Beneficiary Statement") confirming: (1) the st is paid, and (2) the nature and amount of any impounds held by the seipt of the Loan Documents and Beneficiary Statements to satisfy itself to purchase the Property without acceleration or change in the terms approved by Buyer, provided, however, Buyer shall pay the transfer fee then Seller shall within 10 or days following the Date of
receipt of such documents to satisfy itself with regard to the form and content thereof.	auys nom tile
(m) Personal Property. In the event that any personal property is included in the P	Purchase Price, Buyer has 10 or days following the Date of
Agreement to satisfy itself with regard to the title condition of such personal property. Seller repaid for by Buyer. Seller shall provide Buyer copies of any liens or encumbrances affecting such	ecommends that Buyer obtain a UCC-1 report. Any such report shall be
following the Date of Agreement.	At Clasicathon shall not be a second of the second
(n) Destruction, Damage or Loss. Subsequent to the Date of Agreement and prior loss to, the Property or any portion thereof, from any cause whatsoever, which would cost mor \$10,000.00 or less, Seller shall repair or cure the loss prior to the Closing. Buyer shall have the more than \$10,000.00 to repair or cure, to either terminate this Agreement or to purchase the against the Purchase Price. If the cost to repair or cure is more than \$10,000.00, and Buyer doc insurance proceeds applicable to such loss. Unless otherwise notified in writing, Escrow Holde to Closing.	e than \$10,000.00 to repair or cure. If the cost of repair or cure is option, within 10 days after receipt of written notice of a loss costing Property notwithstanding such loss, but without deduction or offset es not elect to terminate this Agreement, Buyer shall be entitled to any
(o) Material Change. Buyer shall have 10 days following receipt of written notice change. "Material Change" shall mean a substantial adverse change in the use, occupancy, ten offer and prior to the Closing. Unless otherwise notified in writing, Escrow Holder shall assume	nants, title, or condition of the Property that occurs after the date of this
	-
	-

INITIALS
© 2019 AIR CRE. All Rights Reserved.
OFAL-15.30, Revised 10-13-2022

- (p) Seller Performance. The delivery of all documents and the due performance by Seller of each and every undertaking and agreement to be performed by Seller under this Agreement.
- (q) Brokerage Fee. Payment at the Closing of such brokerage fee as is specified in this Agreement or later written instructions to Escrow Holder executed by Seller and Brokers ("Brokerage Fee"). It is agreed by the Parties and Escrow Holder that Brokers are a third party beneficiary of this Agreement insofar as the Brokerage Fee is concerned, and that no change shall be made with respect to the payment of the Brokerage Fee specified in this Agreement, without the written consent of Brokers.
- 9.2 The contingencies specified in subparagraphs 9.1(a) through (m) are for the benefit of, and may be waived by, Buyer, and are referred to collectively as "Buyer's Contingencies" and individually as a "Buyer's Contingency."
- 9.3 Buyer's timely and written disapproval or conditional approval of a Buyer's Contingency or any other matter that is subject to Buyer's approval in this Agreement shall constitute disapproval thereof ("Disapproved Item(s)"). Concurrent with notice of a Disapproved Item, Buyer may make a request to Seller regarding such Disapproved Item ("Buyer's Request"). If Buyer fails to make a timely and written Buyer's Request, then this Agreement shall terminate due to the non-satisfaction and non-waiver of a contingency. Seller may respond to a Buyer's Request within 10 days following Seller's receipt thereof ("Seller's Response"). Seller's acceptance of a Buyer's Request shall amend this Agreement accordingly. If Seller fails to provide a timely and written Seller's Response, then Seller's Response shall be deemed to be a rejection of Buyer's Request. Buyer may, within 10 days following the earlier of Buyer's receipt of a Seller's Response (which is not an acceptance of Buyer's Request) or the date of Seller's deemed rejection of a Buyer's Request ("Buyer's Reply Period"), reply to a Seller's Response ("Buyer's Reply") and elect to (i) terminate this Agreement due to the non-satisfaction and non-waiver of the applicable contingency, (ii) accept the Seller's Response in which event this Agreement shall be amended accordingly, or (iii) withdraw Buyer's Request and waive the Disapproved Item in which event Buyer shall accept the Property subject to the Disapproved Item. If Buyer fails to provide a timely and written Buyer's Reply, then Buyer shall be deemed to have elected to terminate this Agreement as of the end of the Buyer's Reply Period. The date Buyer accepts a Seller's Response or withdraws a Buyer's Request and waives a Disapproved Item shall be the date of Buyer's approval of the Disapproved Item. A Party shall provide to Escrow Holder copy of all notices of a Disapproved Item, Buyer's Request, Seller's Response and Buyer's Reply and Escrow Holder shall promptly provide copies thereof to the other Party. Unless the Parties in writing agree otherwise, if the Expected Closing Date is a specific calendar date and a Buyer's Reply Period expires after such specific calendar date, then notwithstanding paragraph 1.1, the Expected Closing Date shall be extended to be 3 business days after the earlier of the date Buyer withdraws a Buyer's Request and waives the applicable Disapproved Item or Buyer accepts the applicable Seller's Response.
- 9.4 The Parties acknowledge that extensive local, state and Federal legislation establish broad liability upon owners and/or users of real property for the investigation and remediation of Hazardous Substances. The determination of the existence of a Hazardous Substance Condition and the evaluation of the impact of such a condition are highly technical and beyond the expertise of Brokers. The Parties acknowledge that they have been advised by Brokers to consult their own technical and legal experts with respect to the possible presence of Hazardous Substances on the Property or adjoining properties, and Buyer and Seller are not relying upon any investigation by or statement of Brokers with respect thereto. The Parties hereby assume all responsibility for the impact of such Hazardous Substances upon their respective interests herein.

10. Documents Required at or Before Closing.

- 10.1 Five days prior to the Closing date Escrow Holder shall obtain an updated Title Commitment concerning the Property from the Title Company and provide copies thereof to each of the Parties.
 - 10.2 Seller shall deliver to Escrow Holder in time for delivery to Buyer at the Closing:
 - (a) Grant or general warranty deed, duly executed and in recordable form, conveying fee title to the Property to Buyer.
 - (b) If applicable, the Beneficiary Statements concerning Existing Note(s).
- (c) If applicable, the Existing Leases and Other Agreements together with duly executed assignments thereof by Seller and Buyer. The assignment of Existing Leases shall be on the most recent Assignment and Assumption of Lessor's Interest in Lease form published by the AIR or its equivalent.
- (d) An affidavit executed by Seller to the effect that Seller is not a "foreign person" within the meaning of Internal Revenue Code Section 1445 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Internal Revenue Service such sum as is required by applicable Federal law with respect to purchases from foreign sellers.
- (e) If the Property is located in California, an affidavit executed by Seller to the effect that Seller is not a "nonresident" within the meaning of California Revenue and Tax Code Section 18662 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Franchise Tax Board such sum as is required by such statute.
 - f) If applicable, a bill of sale, duly executed, conveying title to any included personal property to Buyer.
 - (g) If the Seller is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the sale of the Property.
 - 10.3 Buyer shall deliver to Seller through Escrow:
- (a) The cash portion of the Purchase Price and such additional sums as are required of Buyer under this Agreement shall be deposited by Buyer with Escrow Holder, by federal funds wire transfer, or any other method acceptable to Escrow Holder in immediately collectable funds, no later than 2:00 P.M. on the business day prior to the Expected Closing Date provided, however, that Buyer shall not be required to deposit such monies into Escrow if at the time set for the deposit of such monies Seller is in default or has indicated that it will not perform any of its obligations hereunder. Instead, in such circumstances in order to reserve its rights to proceed Buyer need only provide Escrow with evidence establishing that the required monies were available.
- (b) If a Purchase Money Note and Purchase Money Deed of Trust are called for by this Agreement, the duly executed originals of those documents, the Purchase Money Deed of Trust being in recordable form, together with evidence of fire insurance on the improvements in the amount of the full replacement cost naming Seller as a mortgage loss payee, and a real estate tax service contract (at Buyer's expense), assuring Seller of notice of the status of payment of real property taxes during the life of the Purchase Money Note.
 - (c) The Assignment and Assumption of Lessor's Interest in Lease form specified in paragraph 10.2(c) above, duly executed by Buyer.
 - (d) Assumptions duly executed by Buyer of the obligations of Seller that accrue after Closing under any Other Agreements.
 - (e) If applicable, a written assumption duly executed by Buyer of the loan documents with respect to Existing Notes.
 - (f) If the Buyer is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the purchase of the Property.

10.4 At Closing, Escrow Holder shall cause to be issued to Buyer a standard coverage (or ALTA extended, if elected pursuant to 9.1(g)) owner's form policy of title insurance effective as of the Closing, issued by the Title Company in the full amount of the Purchase Price, insuring title to the Property vested in Buyer, subject only to the exceptions approved by Buyer. In the event there is a Purchase Money Deed of Trust in this transaction, the policy of title insurance shall be a joint protection policy insuring both Buyer and Seller.

IMPORTANT: IN A PURCHASE OR EXCHANGE OF REAL PROPERTY, IT MAY BE ADVISABLE TO OBTAIN TITLE INSURANCE IN CONNECTION WITH THE CLOSE OF ESCROW SINCE THERE MAY BE PRIOR RECORDED LIENS AND ENCUMBRANCES WHICH AFFECT YOUR INTEREST IN THE PROPERTY BEING ACQUIRED. A NEW POLICY OF TITLE INSURANCE SHOULD BE OBTAINED IN ORDER TO ENSURE YOUR INTEREST IN THE PROPERTY THAT YOU ARE ACQUIRING.

11. Prorations and Adjustments.

- 11.1 Taxes. Applicable real property taxes and special assessment bonds shall be prorated through Escrow as of the date of the Closing, based upon the latest tax bill available. The Parties agree to prorate as of the Closing any taxes assessed against the Property by supplemental bill levied by reason of events occurring prior to the Closing. Payment of the prorated amount shall be made promptly in cash upon receipt of a copy of any supplemental bill.
- 11.2 *Insurance*. **WARNING**: Any insurance which Seller may have maintained will terminate on the Closing. Buyer is advised to obtain appropriate insurance to cover the Property.
- 11.3 Rentals, Interest and Expenses. Scheduled rentals, interest on Existing Notes, utilities, and operating expenses shall be prorated as of the date of Closing. The Parties agree to promptly adjust between themselves outside of Escrow any rents received after the Closing.
 - 11.4 Security Deposit. Security Deposits held by Seller shall be given to Buyer as a credit to the cash required of Buyer at the Closing.
- 11.5 Post Closing Matters. Any item to be prorated that is not determined or determinable at the Closing shall be promptly adjusted by the Parties by appropriate cash payment outside of the Escrow when the amount due is determined.
- 11.6 Variations in Existing Note Balances. In the event that Buyer is purchasing the Property subject to an Existing Deed of Trust(s), and in the event that a Beneficiary Statement as to the applicable Existing Note(s) discloses that the unpaid principal balance of such Existing Note(s) at the closing will be more or less than the amount set forth in paragraph 3.1(c) hereof ("Existing Note Variation"), then the Purchase Money Note(s) shall be reduced or increased by an amount equal to such Existing Note Variation. If there is to be no Purchase Money Note, the cash required at the Closing per paragraph 3.1(a) shall be reduced or increased by the amount of such Existing Note Variation.
- 11.7 Variations in New Loan Balance. In the event Buyer is obtaining a New Loan and the amount ultimately obtained exceeds the amount set forth in paragraph 5.1, then the amount of the Purchase Money Note, if any, shall be reduced by the amount of such excess.
- 11.8 Owner's Association Fees. Escrow Holder shall: (i) bring Seller's account with the association current and pay any delinquencies or transfer fees from Seller's proceeds, and (ii) pay any up front fees required by the association from Buyer's funds.

INITIALS		INITIALS
@ 2040 AIR CRE	All Di Li D	

12. Representations and Warranties of Seller and Disclaimers.

- 12.1 Seller's warranties and representations shall survive the Closing and delivery of the deed for a period of 3 years, and any lawsuit or action based upon them must be commenced within such time period. Seller's warranties and representations are true, material and relied upon by Buyer and Brokers in all respects. Seller hereby makes the following warranties and representations to Buyer and Brokers:
- (a) Authority of Seller. Seller is the owner of the Property and/or has the full right, power and authority to sell, convey and transfer the Property to Buyer as provided herein, and to perform Seller's obligations hereunder.
- (b) Maintenance During Escrow and Equipment Condition At Closing. Except as otherwise provided in paragraph 9.1(n) hereof, Seller shall maintain the Property until the Closing in its present condition, ordinary wear and tear excepted.
- (c) Hazardous Substances/Storage Tanks. Seller has no knowledge, except as otherwise disclosed to Buyer in writing, of the existence or prior existence on the Property of any Hazardous Substance, nor of the existence or prior existence of any above or below ground storage tank.
- (d) Compliance. Except as otherwise disclosed in writing, Seller has no knowledge of any aspect or condition of the Property which violates applicable laws, rules, regulations, codes or covenants, conditions or restrictions, or of improvements or alterations made to the Property without a permit where one was required, or of any unfulfilled order or directive of any applicable governmental agency or casualty insurance company requiring any investigation, remediation, repair, maintenance or improvement be performed on the Property.
- (e) Changes in Agreements. Prior to the Closing, Seller will not violate or modify any Existing Lease or Other Agreement, or create any new leases or other agreements affecting the Property, without Buyer's written approval, which approval will not be unreasonably withheld.
- Possessory Rights. Seller has no knowledge that anyone will, at the Closing, have any right to possession of the Property, except as disclosed by this Agreement or otherwise in writing to Buyer.
 - Mechanics' Liens. There are no unsatisfied mechanics' or materialmens' lien rights concerning the Property.
- Actions, Suits or Proceedings. Seller has no knowledge of any actions, suits or proceedings pending or threatened before any commission, board, (h) bureau, agency, arbitrator, court or tribunal that would affect the Property or the right to occupy or utilize same.
- (i) Notice of Changes. Seller will promptly notify Buyer and Brokers in writing of any Material Change (see paragraph 9.1(o)) affecting the Property that becomes known to Seller prior to the Closing.
- No Tenant Bankruptcy Proceedings. Seller has no notice or knowledge that any tenant of the Property is the subject of a bankruptcy or insolvency (j) proceeding.
 - (k) No Seller Bankruptcy Proceedings. Seller is not the subject of a bankruptcy, insolvency or probate proceeding.
- Personal Property. Seller has no knowledge that anyone will, at the Closing, have any right to possession of any personal property included in the Purchase Price nor knowledge of any liens or encumbrances affecting such personal property, except as disclosed by this Agreement or otherwise in writing to Buyer.
- 12.2 Buyer hereby acknowledges that, except as otherwise stated in this Agreement, Buyer is purchasing the Property in its existing condition and will, by the time called for herein, make or have waived all inspections of the Property Buyer believes are necessary to protect its own interest in, and its contemplated use of, the Property. The Parties acknowledge that, except as otherwise stated in this Agreement, no representations, inducements, promises, agreements, assurances, oral or written, concerning the Property, or any aspect of the occupational safety and health laws, Hazardous Substance laws, or any other act, ordinance or law, have been made by either Party or Brokers, or relied upon by either Party hereto.
- 12.3 In the event that Buyer learns that a Seller representation or warranty might be untrue prior to the Closing, and Buyer elects to purchase the Property anyway then, and in that event, Buyer waives any right that it may have to bring an action or proceeding against Seller or Brokers regarding said representation or warranty.
- 12.4 Any environmental reports, soils reports, surveys, feasibility studies, and other similar documents which were prepared by third party consultants and provided to Buyer by Seller or Seller's representatives, have been delivered as an accommodation to Buyer and without any representation or warranty as to the sufficiency, accuracy, completeness, and/or validity of said documents, all of which Buyer relies on at its own risk. Seller believes said documents to be accurate, but Buyer is advised to retain appropriate consultants to review said documents and investigate the Property.

Possession.

Possession of the Property shall be given to Buyer at the Closing subject to the rights of tenants under Existing Leases.

14. Buyer's Entry.

At any time during the Escrow period, Buyer, and its agents and representatives, shall have the right at reasonable times and subject to rights of tenants, to enter upon the Property for the purpose of making inspections and tests specified in this Agreement. No destructive testing shall be conducted, however, without Seller's prior approval which shall not be unreasonably withheld. Following any such entry or work, unless otherwise directed in writing by Seller, Buyer shall return the Property to the condition it was in prior to such entry or work, including the recompaction or removal of any disrupted soil or material as Seller may reasonably direct. All such inspections and tests and any other work conducted or materials furnished with respect to the Property by or for Buyer shall be paid for by Buyer as and when due and Buyer shall indemnify, defend, protect and hold harmless Seller and the Property of and from any and all claims, liabilities, losses, expenses (including reasonable attorneys' fees), damages, including those for injury to person or property, arising out of or relating to any such work or materials or the acts or omissions of Buyer, its agents or employees in connection therewith.

Further Documents and Assurances.

The Parties shall each, diligently and in good faith, undertake all actions and procedures reasonably required to place the Escrow in condition for Closing as and when required by this Agreement. The Parties agree to provide all further information, and to execute and deliver all further documents, reasonably required by Escrow Holder or the Title Company.

If any Party or Broker brings an action or proceeding (including arbitration) involving the Property whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees and costs. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred.

Prior Agreements/Amendments.

- 17.1 This Agreement supersedes any and all prior agreements between Seller and Buyer regarding the Property.
- 17.2 Amendments to this Agreement are effective only if made in writing and executed by Buyer and Seller.

Broker's Rights.

18.1 If this sale is not consummated due to the default of either the Buyer or Seller, the defaulting Party shall be liable to and shall pay to Brokers the Brokerage Fee that Brokers would have received had the sale been consummated. If Buyer is the defaulting party, payment of said Brokerage Fee is in addition to any obligation with respect to liquidated or other damages.

18.2 Upon the Closing, Brokers are authorized to publicize the facts of this transaction.

19.1 Whenever any Party, Escrow Holder or Brokers herein shall desire to give or serve any notice, demand, request, approval, disapproval or other communication, each such communication shall be in writing and shall be delivered personally, by messenger, or by mail, postage prepaid, to the address set forth in this agreement or by facsimile transmission, electronic signature, digital signature, or email.

19.2 Service of any such communication shall be deemed made on the date of actual receipt if personally delivered, or transmitted by facsimile transmission, electronic signature, digital signature, or email. Any such communication sent by regular mail shall be deemed given 48 hours after the same is mailed. Communications sent by United States Express Mail or overnight courier that guarantee next day delivery shall be deemed delivered 24 hours after delivery of the same to the Postal Service or courier. If such communication is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.

19.3 Any Party or Broker hereto may from time to time, by notice in writing, designate a different address to which, or a different person or additional persons to whom, all communications are thereafter to be made.

20.	Duration	of	Offer.

.0.	diation of Office.
	0.1 If this offer is not accepted by Seller on or before 5:00 P.M. according to the time standard applicable to the city of on the date of
	, it shall be deemed automatically revoked.
	0.2 The acceptance of this offer, or of any subsequent counteroffer hereto, that creates an agreement between the Parties as described in paragraph 1.2, shall
	
NIT	LS INITIALS

be deemed made upon delivery to the other Party or either Broker herein of a duly executed w counteroffer.	riting unconditionally accepting the last outstanding offer or
21. LIQUIDATED DAMAGES. (This Liquidated Damages paragraph is applicable only if initials the parties agree that it would be impracticable or extremely difficult to fix, preserved by seller if buyer fails to perform its obligations under this of all contingencies provided for the buyer's benefit, buyer breaches this agree amount of Upon payment of said sum to seller, buyer shall be relecancellation fees and title company charges shall be paid by seller.	IOR TO SIGNING THIS AGREEMENT, THE ACTUAL DAMAGES WHICH AGREEMENT. THEREFORE, IF, AFTER THE SATISFACTION OR WAIVER
Buyer's Initials	Seller's Initials

22. ARBITRATION OF DISPUTES. (This Arbitration of Disputes paragraph is applicable only if initialed by both Parties.)

22.1 ANY CONTROVERSY AS TO WHETHER SELLER IS ENTITLED TO LIQUIDATED DAMAGES AND/OR BUYER IS ENTITLED TO THE RETURN OF THE DEPOSIT SHALL BE DETERMINED BY BINDING ARBITRATION ADMINISTERED BY THE JUDICIAL ARBITRATION & MEDIATION SERVICES, INC. ("JAMS") IN ACCORDANCE WITH ITS COMMERCIAL ARBITRATION RULES ("COMMERCIAL RULES"). ARBITRATION HEARINGS SHALL BE HELD IN THE COUNTY WHERE THE PROPERTY IS LOCATED. SUCH CONTROVERSY SHALL BE ARBITRATED BY A SINGLE ARBITRATOR. APPOINTED UNDER THE COMMERCIAL RULES WHO HAS HAD AT LEAST 5 YEARS OF EXPERIENCE IN THE TYPE OF REAL ESTATE THAT IS THE SUBJECT OF THIS AGREEMENT. THE ARBITRATOR SHALL HEAR AND DETERMINE SAID CONTROVERSY IN ACCORDANCE WITH APPLICABLE LAW OF THE JURISDICTION WHERE THE PROPERTY IS LOCATED. THE INTENTION OF THE PARTIES AS EXPRESSED IN THIS AGREEMENT AND ANY AMENDMENTS THERETO, AND UPON THE EVIDENCE PRODUCED AT AN ARBITRATION HEARING. PRE-ARBITRATION DISCOVERY SHALL BE PERMITTED IN ACCORDANCE WITH THE COMMERCIAL RULES OR STATE LAW APPLICABLE TO ARBITRATION PROCEEDINGS. THE ARBITRATOR SHALL RENDER AN AWARD WITHIN 30 DAYS AFTER THE CONCLUSION OF THE HEARING, WHICH MAY INCLUDE ATTORNEYS' FEES AND COSTS TO THE PREVAILING PARTY PER PARAGRAPH 16 HEREOF AND SHALL BE ACCOMPANIED BY A REASONED OPINION. THE FAILURE OR REFUSAL OF A PARTY TO PAY SUCH PARTY'S REQUIRED SHARE OF THE DEPOSITS FOR ARBITRATOR COMPENSATION OR ADMINISTRATIVE CHARGES SHALL CONSTITUTE A WAIVER BY SUCH PARTY TO PRESENT EVIDENCE OR CROSS-EXAMINE WITNESSES, BUT SUCH WAIVER SHALL NOT ALLOW FOR A DEFAULT JUDGMENT AGAINST THE NON-PAYING PARTY IN THE ABSENCE OF EVIDENCE AND LEGAL ARGUMENT AS THE ARBITRATOR MAY REQUIRE FOR MAKING AN AWARD. JUDGMENT MAY BE ENTERED ON THE AWARD IN ANY COURT OF COMPETENT JURISDICTION NOTWITHSTANDING THE FAILURE OF A PARTY DULY NOTIFIED OF THE ARBITRATION HEARING TO APPEAR THEREAT.

22.2 BUYER'S RESORT TO OR PARTICIPATION IN SUCH ARBITRATION PROCEEDINGS SHALL NOT BAR SUIT IN A COURT OF COMPETENT JURISDICTION BY THE BUYER FOR DAMAGES AND/OR SPECIFIC PERFORMANCE UNLESS AND UNTIL THE ARBITRATION RESULTS IN AN AWARD TO THE SELLER OF LIQUIDATED DAMAGES, IN WHICH EVENT SUCH AWARD SHALL ACT AS A BAR AGAINST ANY ACTION BY BUYER FOR DAMAGES AND/OR SPECIFIC PERFORMANCE.

22.3 NOTICE: BY INITIALING IN THE SPACE BELOW YOU ARE AGREEING TO HAVE ANY DISPUTE ARISING OUT OF THE MATTERS INCLUDED IN THE "ARBITRATION" OF DISPUTES" PROVISION DECIDED BY NEUTRAL ARBITRATION AS PROVIDED BY CALIFORNIA LAW AND YOU ARE GIVING UP ANY RIGHTS YOU MIGHT POSSESS TO HAVE THE DISPUTE LITIGATED IN A COURT OR JURY TRIAL. BY INITIALING IN THE SPACE BELOW YOU ARE GIVING UP YOUR JUDICIAL RIGHTS TO DISCOVERY AND APPEAL, UNLESS SUCH RIGHTS ARE SPECIFICALLY INCLUDED IN THE "ARBITRATION OF DISPUTES" PROVISION. IF YOU REFUSE TO SUBMIT TO ARBITRATION AFTER AGREEING TO THIS PROVISION, YOU MAY BE COMPELLED TO ARBITRATE UNDER THE AUTHORITY OF THE CALIFORNIA CODE OF CIVIL PROCEDURE. YOUR AGREEMENT TO THIS ARBITRATION PROVISION IS VOLUNTARY.

WE HAVE READ AND UNDERSTAND THE FOREGOING AND AGREE TO SUBMIT DISPUTES ARISING OUT OF THE MATTERS INCLUDED IN THE "ARBITRATION OF DISPUTES" PROVISION TO NEUTRAL ARBITRATION.

Buyer's Initials	_	Seller's Initials	

23. Miscellaneous.

- 23.1 Binding Effect. Buyer and Seller both acknowledge that they have carefully read and reviewed this Agreement and each term and provision contained herein. In addition, this Agreement shall be binding on the Parties without regard to whether or not paragraphs 21 and 22 are initialed by both of the Parties. Paragraphs 21 and 22 are each incorporated into this Agreement only if initialed by both Parties at the time that the Agreement is executed. Signatures to this
- Agreement accomplished by means of electronic signature or similar technology shall be legal and binding.

 23.2 **Applicable Law**. This Agreement shall be governed by, and paragraph 22.3 is amended to refer to, the laws of the state in which the Property is located. Any litigation or arbitration between the Parties hereto concerning this Agreement shall be initiated in the county in which the Property is located.
 - 23.3 Time of Essence. Time is of the essence of this Agreement.
- 23.4 **Counterparts**. This Agreement may be executed by Buyer and Seller in counterparts, each of which shall be deemed an original, and all of which together shall constitute one and the same instrument. Escrow Holder, after verifying that the counterparts are identical except for the signatures, is authorized and instructed to combine the signature pages on one of the counterparts, which shall then constitute the Agreement.
- 23.5 Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY OR ARISING OUT OF THIS AGREEMENT.
- 23.6 Conflict. Any conflict between the printed provisions of this Agreement and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions. Seller and Buyer must initial any and all handwritten provisions.

 23.7 1031 Exchange. Both Seller and Buyer agree to cooperate with each other in the event that either or both wish to participate in a 1031 exchange. Any
- party initiating an exchange shall bear all costs of such exchange. The cooperating Party shall not have any liability (special or otherwise) for damages to the exchanging Party in the event that the sale is delayed and/or that the sale otherwise fails to qualify as a 1031 exchange.
 - 23.8 Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Agreement shall mean and refer to calendar days.

Disclosures Regarding The Nature of a Real Estate Agency Relationship.

- 24.1 The Parties and Brokers agree that their relationship(s) shall be governed by the principles set forth in the applicable sections of the California Civil Code, as summarized in paragraph 24.2.
- 24.2 When entering into a discussion with a real estate agent regarding a real estate transaction, a Buyer or Seller should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Buyer and Seller acknowledge being advised by the Brokers in this transaction, as follows:
- (a) Seller's Agent. A Seller's agent under a listing agreement with the Seller acts as the agent for the Seller only. A Seller's agent or subagent has the following affirmative obligations: (1) To the Seller: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Seller. (2) To the Buyer and the Seller: a. Diligent exercise of reasonable skills and care in performance of the agent's duties. b. A duty of honest and fair dealing and good faith. c. A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set
- Buyer's Agent. A selling agent can, with a Buyer's consent, agree to act as agent for the Buyer only. In these situations, the agent is not the Seller's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Seller. An agent acting only for a Buyer has the following affirmative obligations. (1) To the Buyer: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Buyer. (2) To the Buyer and the Seller: a. Diligent exercise of reasonable skills and care in performance of the agent's duties. b. A duty of honest and fair dealing and good faith. c. A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (c) Agent Representing Both Seller and Buyer. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Seller and the Buyer in a transaction, but only with the knowledge and consent of both the Seller and the Buyer. (1) In a dual agency situation, the agent has the following affirmative obligations to both the Seller and the Buyer: a. A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Seller or the Buyer. b. Other duties to the Seller and the Buyer as stated above in their respective sections (a) or (b) of this paragraph 24.2. (2) In representing both Seller and Buyer, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

not limited to, facts relating to either Buyer's or Seller's financial position, motivations, bargaining position, or other personal information that may impact price, including Seller's willingness to accept a price less than the listing price or Buyer's willingness to pay a price greater than the price offered. (3) The above duties of the agent in a real estate transaction do not relieve a Seller or Buyer from the responsibility to protect their own interests. Buyer and Seller should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Buyer has the duty to exercise reasonable care to protect Buyer, including as to those facts about the Property which are known to Buyer or within Buyer's diligent attention and observation. Both Seller and Buyer should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.

(d) Further Disclosures. Throughout this transaction Buyer and Seller may receive more than one disclosure, depending upon the number of agents assisting in the transaction. Buyer and Seller should each read its contents each time it is presented, considering the relationship between them and the real estate agent in this transaction and that disclosure. Buyer and Seller each acknowledge receipt of a disclosure of the possibility of multiple representation by the Broker representing that principal. This disclosure may be part of a listing agreement, buyer representation agreement or separate document. Buyer understands that Broker representing Buyer may also represent other potential buyers, who may consider, make offers on or ultimately acquire the Property. Seller understands that Broker representing Seller may also represent other sellers with competing properties that may be of interest to this Buyer. Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this transaction may be brought against Broker more than one year after the Date of Agreement and that the liability (including court costs and attorneys' fees), of any Broker with respect to any breach of duty, error or omission relating to this Agreement; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.

24.3 Confidential Information. Buyer and Seller agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.

25. Construction of Agreement. In construing this Agreement, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Agreement. Whenever required by the context, the singular shall include the plural and vice versa. This Agreement shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.

26.	Additional Provision	anc

Additional provisions of this offer, if any, are as follows or are attached hereto by an addendum or addenda consisting of paragraphs ______ through _______. (If there are no additional provisions write "NONE".)

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS AGREEMENT OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

- 1. SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS AGREEMENT.
- 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PROPERTY. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING OF THE PROPERTY, THE INTEGRITY AND CONDITION OF ANY STRUCTURES AND OPERATING SYSTEMS, AND THE SUITABILITY OF THE PROPERTY FOR BUYER'S INTENDED USE.

WARNING: IF THE PROPERTY IS LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THIS AGREEMENT MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PROPERTY IS LOCATED.

NOTE:

- 1. THIS FORM IS NOT FOR USE IN CONNECTION WITH THE SALE OF RESIDENTIAL PROPERTY.
- 2. IF EITHER PARTY IS A CORPORATION, IT IS RECOMMENDED THAT THIS AGREEMENT BE SIGNED BY TWO CORPORATE OFFICERS.

The undersigned Buyer offers and agrees to buy the Property on the terms and conditions stated and acknowledges receipt of a copy hereof.

\	Butt.
BROKER	BUYER
	\
Atta	
Attn:	Ву:
Title:	Name Printed:
Address	Title:
Address:	Phone:
Phone:	Fax:
Fax:	Email:
Email:	
Federal ID No.:	D
Broker DRE License #:	By:
Agent DRE License #:	Name Printed:
	Title:
	Phone:
	Fax:
	Email:
	Address:
	Federal ID No.:
_ Y	

27. Acceptance.

27.1 Seller accepts the foregoing offer to purchase the Property and hereby agrees to sell the Property to Buyer on the terms and conditions therein specified.

27.2 In consideration of real estate brokerage service rendered by Brokers, Seller agrees to pay Brokers a real estate Brokerage Fee in a sum equal to

% of the Purchase Price to be divided between the Brokers as follows: Seller's Broker ______ % and Buyer's Broker ______ %. This

Agreement shall serve as an irrevocable instruction to Escrow Holder to pay such Brokerage Fee to Brokers out of the proceeds accruing to the account of Seller at the Closing.

27.3 Seller acknowledges receipt of a copy hereof and authorizes Brokers to deliver a signed copy to Buyer.

NOTE: A PROPERTY INFORMATION SHEET IS REQUIRED TO BE DELIVERED TO BUYER BY SELLER UNDER THIS AGREEMENT.

		Date:
BROKER		SELLER
Attn:		
INITIALS	All Bights Recogned	INITIALS
(C) JOJO AID CDE	All Dights Dosonyod	

Title:	
	Ву:
Address:	Name Printed:
Phone: Fax:	Title:
Email:	Phone: Fax:
Federal ID No.:	Email:
Broker DRE License #:	
Agent's DRE License #:	Ву:
	Name Printed:
	Title: Phone:
	Fax:
	Email:
	Address:
	Federal ID No.:
AIR CRE * https://www.aircre.com * NOTICE: No part of these works may be repro	* 213-687-8777 * contracts@aircre.com oduced in any form without permission in writing.
No nezi No part of these works may be repre	National Manager Permission in Manager
	< \
	< / ·
\ / ·	
\sim \times	
1	



Dated:

STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS FOR PURCHASE OF RESIDENTIAL INCOME PROPERTY

(Do Not Use for a Property Which Contains Less than Five Units. Do Not Use for a Condominium or for a Common Interest Development)

1.	Buyer.		/II= II) I						5 (II = 11 II) (
indiv	1.1 idually a				chase the real proper close 30 or						
					der") whose address						
and of Buye	conditions or of Buyer 1.2 The equent co	set fo r's oblig term untero	rth in this agreer gations herein ur " Date of Agree m	nent (" Agreement " less Seller expressl ent " as used hereir). Buyer shall have th	ne right to assig	n Buyer's rights he	ereunder, bu defined in p	t any such assig aragraph 20.2)	nment shall not re	elieve or a
2.	Property										
					ect of this offer consi y, state, zip)					//	/ of
part telep equi	2.2 If the ected to m 2.3 The of the prophone distorment; fillite dishes	ne lega neet the Prope perty, ributio ire spri	I description of t e requirements o erty includes, at r as well as the fol n systems (lines, nkler systems; so and spa equipme	the Property is not of f ("T o additional cost to cowing items, if any jacks and connection courity and fire detern; garage door op	omplete or is inaccur itle Company"), whice Buyer, the permane cons only); built-in appection systems; carpet eners; and	rate, this Agreei ch shall issue th ent improvemen d at present loca pliances; plumb ts; window cov (collective	ment shall not be e title policy herei ts thereon, includ ated on the Propeing fixtures; ceilingerings and screens ly, the "Improvem"	invalid and t inafter descr ling those ite erty: electrica g fans; firepl s; attached w nents").	he legal descrip ibed. ms which pursu I distribution sy aces; heating, v vall and floor co	tion shall be comp ant to applicable I stems and lighting entilating, air cond verings; television	law are a g fixtures; litioning antennas
with				•	e determined during E nase Price does not in			//	/ v	nd a	all of
whic	h shall be	remov	ed by Seller prio	to Closing.							
3.	Purchase	Price.									
		-	ase price (" Purcl / not applicable)	nase Price") to be p	aid by Buyer to Seller	r for the Proper	ty shall be	, paya	ible as follows:		
	,	,	, ,	teal alternative Decre	-11 d-6' d-1						
	(a)	Casn	i down payment,	including the Depo	sit as defined in para	agraph 4.3 (or if	an all cash transa	ction, the Pu	rchase Price):		
	(b)	Amo	ount of "New Loa	n" as defined in par	ragraph 5.1, if any:						
	(c)	Trus	$oldsymbol{t}$ ") securing the $oldsymbol{\epsilon}$	existing promissory	ject to and/or assum note(s) (" Existing No n an unpaid principal	te(s)"):			ting Deed(s) of		
	40	(ii)	until paid (and/ An Existing Not Said Second No until paid (and/	or the entire unpaid ("Second Note") te is payable at or the entire unpaid	per month, iu d balance is due on _ with an unpaid princip per month d balance is due on _). pal balance as o h, including inte).	of the Closing of aperest at the rate of	pproximately	r: % per annum		_
	(d)	•	-	•	Purchase Money Dee 1 6 ("Purchase Mone	•		ecure the pro	omissory note		
Total	Purchase	Price:									
	es includi	ng, but .5% of	not limited to, p	oints, processing fe	to, or assuming, an Exes, and appraisal fees applicable Existing No	s as a condition					
	4.1	Buye	r has delivered to	Broker a check in	the sum of	, payable to	o Escrow Holder, t	o be delivere	ed by Broker to	Escrow Holder wit	hin 2 or
term	ow Holder inate this ement. Si	busine a ched transa hould E	ss days after botl ck in the sum of ction by giving w	n Parties have execu . If sa ritten notice of suc	Ited this Agreement a Ited this Agreement a id check is not receive the election to Escrow I Igreement for purchase	and the execute ed by Escrow H Holder whereup	ed Agreement has older within said to oon neither Party	been delive time period t shall have an	red to Escrow H hen Seller may ly further liabili	older Buyer shall of elect to unilaterally ty to the other und	deliver to ly der this
			•	ys after the Date of	Agreement, Buyer sh	hall deposit with	n Escrow Holder th	he additional	sum of	to be appli	ed to the
Purc	hase Price		•	s after the contine	oncine discussed in	aragraph 0 1 /=	through (m) are	annroyed s-	waived Books	hall donosit with	Eccross
Hold				_	encies discussed in pa I to the Purchase Pric			approved or	waived, Buyer s	snaii deposit with i	:SCrOW
char accru spec	ng that, under notice 4.3 Escape tered bankue to the bified matu	nless the or instance or insta	he Additional De ructions. older shall depos interest bearing of Buyer, who he	oosit is received by t the funds deposit account whose terr ereby acknowledge	oy Escrow Holder with Escrow Holder within ed with it by Buyer p in is appropriate and o is that there may be p inber is	n 2 business day oursuant to para consistent with penalties or inte	ys following said n graphs 4.1 and 4.2 the timing requir rest forfeitures if t	otice, the Es 2 (collectively ements of th the applicabl	crow shall be do y the "Deposit" is transaction. le instrument is	eemed terminated), in a State or Fed The interest theref redeemed prior to	without erally from shall o its
							_				
INIT	IALS					INITIAL	<u></u>				
		CRE. A	All Rights Reser	ved.			-		Last Edited	: 10/12/2022 3:0)4 PM

ROFA-11.30, Revised 10-12-2022

4.4 Notwithstanding the foregoing, within 5 days after Escrow Holder receives the monies described in paragraph 4.1 above, Escrow Holder shall release \$100 of said monies to Seller as and for independent consideration for Seller's' execution of this Agreement and the granting of the contingency period to Buyer as herein provided. Such independent consideration is non-refundable to Buyer but shall be credited to the Purchase Price in the event that the purchase of the Property is completed. 4.5 Upon waiver of all of Buyer's contingencies the Deposit shall become non-refundable but applicable to the Purchase Price except in the event of a Seller breach, or in the event that the Escrow is terminated pursuant to the provisions of Paragraph 9.1(n) (Destruction, Damage or Loss) or 9.1(o) (Material Change). Financing Contingency. (Strike if not applicable) 5.1 This offer is contingent upon Buyer obtaining from an insurance company, financial institution or other lender, a commitment to lend to Buyer a sum equal % of the Purchase Price, on terms acceptable to Buyer. Such loan ("New Loan") shall be secured by a first deed of trust or mortgage on the to at least Property. If this Agreement provides for Seller to carry back junior financing, then Seller shall have the right to approve the terms of the New Loan. Seller shall have 7 days from receipt of the commitment setting forth the proposed terms of the New Loan to approve or disapprove of such proposed terms. If Seller fails to notify Escrow Holder, in writing, of the disapproval within said 7 days it shall be conclusively presumed that Seller has approved the terms of the New Loan. 5.2 If Buyer shall fail to notify its Broker, Escrow Holder and Seller, in writing within days following the Date of Agreement, that the New Loan has not been obtained, it shall be conclusively presumed that Buyer has either obtained said New Loan or has waived this New Loan contingency. 5.3 If Buyer shall notify its Broker, Escrow Holder and Seller, in writing, within the time specified in paragraph 5.2 hereof, that Buyer has not obtained said New Loan, this Agreement shall be terminated, and Buyer shall be entitled to the prompt return of the Deposit, plus any interest earned thereon, less only Escrow Holder and Title Company cancellation fees and costs, which Buyer shall pay. Seller Financing. (Purchase Money Note). (Strike if not applicable) 6.1 If Seller approves Buyer's financials (see paragraph 6.5) the Purchase Money Note shall provide for interest on unpaid principal at the rate of % per annum, with principal and interest paid as follows: _. The Purchase Money Note and Purchase Money Deed of Trust shall be on the current forms nmonly used by Escrow Holder, and be junior and subordinate only to the Existing Note(s) and/or the New Loan expressly called for by this Agreement 6.2 The Purchase Money Note and/or the Purchase Money Deed of Trust shall contain provisions regarding the following (see also paragraph 10.3 (b)) Prepayment. Principal may be prepaid in whole or in part at any time without penalty, at the option of the Buyer. Late Charge. A late charge of 6% shall be payable with respect to any payment of principal, interest, or other charges, not made within 10 days after (b) it is due. Due On Sale. In the event the Buyer sells or transfers title to the Property or any portion thereof, then the Seller may, at Seller's option, require the (c) entire unpaid balance of said Note to be paid in full. 6.3 If the Purchase Money Deed of Trust is to be subordinate to other financing, Escrow Holder shall, at Buyer's expense prepare and record on Seller's behalf a request for notice of default and/or sale with regard to each mortgage or deed of trust to which it will be subordinate. 6.4 WARNING: CALIFORNIA LAW DOES NOT ALLOW DEFICIENCY JUDGEMENTS ON SELLER FINANCING. IF BUYER ULTIMATELY DEFAULTS ON THE LOAN, SELLER'S SOLE REMEDY IS TO FORECLOSE ON THE PROPERTY. 6.5 Seller's obligation to provide financing is contingent upon Seller's reasonable approval of Buyer's financial condition. Buyer to provide a current financial statement and copies of its Federal tax returns for the last 3 years to Seller within 10 days following the Date of Agreement. Seller has 10 days following receipt of such documentation to satisfy itself with regard to Buyer's financial condition and to notify Escrow Holder as to whether or not Buyer's financial condition is acceptable. If Seller fails to notify Escrow Holder, in writing, of the disapproval of this contingency within said time period, it shall be conclusively presumed that Seller has approved Buyer's financial condition. If Seller is not satisfied with Buyer's financial condition or if Buyer fails to deliver the required documentation then Seller may notify Escrow Holder in writing that Seller Financing will not be available, and Buyer shall have the option, within 10 days of the receipt of such notice, to either terminate this transaction or to purchase the Property without Seller financing. If Buyer fails to notify Escrow Holder within said time period of its election to terminate this transaction then Buyer shall be conclusively presumed to have elected to purchase the Property without Seller financing. If Buyer elects to terminate, Buyer's Deposit shall be refunded less Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation. 7.1 Each Party acknowledges receiving a Disclosure Regarding Real Estate Agency Relationship, confirms and consents to the following agency relationships in $this \, transaction \, \, with \, the \, \, following \, real \, estate \, \, broker(s) \, ("\textbf{Brokers"}) \, \, and/or \, their \, agents \, ("Agent(s)"): \, \, ("\textbf{Agent}(s)"): \, \, ("\textbf{Agent}(s)"): \, \, ((\textbf{Agent}(s))"): \, \, ((\textbf{Agent}($ is the broker of (check one): Late Seller; or both the Buyer and Seller (dual agent). License No. Seller's Brokerage Firm the Seller's Agent (salesperson or broker associate); or both the Seller's Agent is (check one): License No. Seller's Agent and the Buyer's Agent (dual agent). is the broker of (check one): $\ \ \ \ \$ the Buyer; or $\ \ \ \ \$ both the Buyer and Seller (dual agent). Buyer's Brokerage Firm License No. the Buyer's Agent (salesperson or broker associate); or 🔲 both the Buyer's Agent is (check one): Buyer's Agent License No. and the Seller's Agent (dual agent). The Parties acknowledge that other than the Brokers and Agents listed above, there are no other brokers or agents representing the Parties or due any fees and/or commissions under this Agreement. Buyer shall use the services of Buyer's Broker exclusively in connection with any and all negotiations and offers with respect to the Property for a period of 1 year from the date inserted for reference purposes at the top of page 1. 7.2 Buyer and Seller each represent and warrant to the other that he/she/it has had no dealings with any person, firm, broker, agent or finder in connection with the negotiation of this Agreement and/or the consummation of the purchase and sale contemplated herein, other than the Brokers and Agents named in paragraph 7.1, and no broker, agent or other person, firm or entity, other than said Brokers and Agents is/are entitled to any commission or finder's fee in connection with this transaction as the result of any dealings or acts of such Party. Buyer and Seller do each hereby agree to indemnify, defend, protect and hold the other harmless from and against any costs, expenses or liability for compensation, commission or charges which may be claimed by any broker, agent, finder or other similar party, other than said named Brokers and Agents by reason of any dealings or act of the indemnifying Party. 8.1 Upon acceptance hereof by Seller, this Agreement, including any counteroffers incorporated herein by the Parties, shall constitute not only the agreement of purchase and sale between Buyer and Seller, but also instructions to Escrow Holder for the consummation of the Agreement through the Escrow. Escrow Holder shall not prepare any further escrow instructions restating or amending the Agreement unless specifically so instructed by the Parties or a Broker herein. Subject to the reasonable approval of the Parties, Escrow Holder may, however, include its standard general escrow provisions. In the event that there is any conflict between the provisions of the Agreement and the provisions of any additional escrow instructions the provisions of the Agreement shall prevail as to the Parties and the Escrow Holden 8.2 As soon as practical after the receipt of this Agreement and any relevant counteroffers, Escrow Holder shall ascertain the Date of Agreement as defined in paragraphs 1.2 and 20.2 and advise the Parties and Brokers, in writing, of the date ascertained. Escrow Holder is hereby authorized and instructed to conduct the Escrow in accordance with this Agreement, applicable law and custom and practice of the community in which Escrow Holder is located, including any reporting requirements of the Internal Revenue Code. In the event of a conflict between the law of the state where the Property is located and the law of the state where the Escrow Holder is located, the law of the state where the Property is located shall prevail. 8.4 Subject to satisfaction of the contingencies herein described, Escrow Holder shall close this escrow (the "Closing") by recording a general warranty deed (a grant deed in California) and the other documents required to be recorded, and by disbursing the funds and documents in accordance with this Agreement. 8.5 Buyer and Seller shall each pay one-half of the Escrow Holder's charges and Seller shall pay the usual recording fees and any required documentary transfer taxes. Seller shall pay the premium for a standard coverage owner's or joint protection policy of title insurance. (See also paragraph 11.) 8.6 Escrow Holder shall verify that all of Buyer's contingencies have been satisfied or waived prior to Closing. The matters contained in paragraphs 9.1 subparagraphs (b), (c), (d), (e), (g), (i), (n), and (o), 9.4, 12, 13, 14, 16, 18, 20, 21, 22, and 24 are, however, matters of agreement between the Parties only and are not instructions to Escrow Holder. 8.7 If this transaction is terminated for non-satisfaction and non-waiver of a Buyer's Contingency, as defined in Paragraph 9.2 or disapproval of any other matter subject to Buyer's approval, then neither of the Parties shall thereafter have any liability to the other under this Agreement, except to the extent of a breach of any affirmative covenant or warranty in this Agreement. In the event of such termination, Buyer shall, subject to the provisions of paragraph 8.10, be promptly

8.8 The Closing shall occur on the Expected Closing Date, or as soon thereafter as the Escrow is in condition for Closing; provided, however, that if the Closing

INITIALS

refunded all funds deposited by Buyer with Escrow Holder, less only the \$100 provided for in paragraph 4.4 and the Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation. If this transaction is terminated as a result of Seller's breach of this Agreement then Seller shall pay the Title

Company and Escrow Holder cancellation fees and costs.

does not occur by the Expected Closing Date and said Date is not extended by mutual instructions of the Parties, a Party not then in default under this Agreement may notify the other Party, Escrow Holder, and Brokers, in writing that, unless the Closing occurs within 5 business days following said notice, the Escrow shall be deemed terminated without further notice or instructions.

8.9 Except as otherwise provided herein, the termination of Escrow shall not relieve or release either Party from any obligation to pay Escrow Holder's fees and costs or constitute a waiver, release or discharge of any breach or default that has occurred in the performance of the obligations, agreements, covenants or warranties contained therein.

8.10 If this Escrow is terminated for any reason other than Seller's breach or default, then as a condition to the return of Buyer's deposit, Buyer shall within 5 days after written request deliver to Seller, at no charge, copies of all surveys, engineering studies, soil reports, maps, master plans, feasibility studies and other similar items prepared by or for Buyer that pertain to the Property.

9.1 IF, BEFORE EXPIRATION OF THE APPLICABLE TIME, BUYER FAILS TO PROVIDE ESCROW HOLDER WRITTEN NOTICE OF BUYER'S DISAPPROVAL OF ANY OF

9. Contingencies to Closing.

BUYER'S CONTINGENCIES OR ANY OTHER MATTER THAT IS SUBJECT TO BUYER'S APPROVAL IN THIS AGREEMENT, THEN BUYER SHALL BE CONCLUSIVELY DEEMED
TO HAVE SATISFIED SUCH BUYER'S CONTINGENCIES AND/OR APPROVED OF SUCH OTHER MATTERS. If a number of days is completed in any of the optional spaces in subparagraphs 9.1 (a) through (m), then such number shall apply and override the pre-printed number, even if the pre-printed number is not stricken. The Closing
of this transaction is contingent upon the satisfaction or waiver of the following contingencies:
(a) Disclosure. Seller shall make to Buyer, through Escrow, all of the applicable disclosures required by law (See AIR CRE ("AIR") standard form entitled "Seller's Mandatory Disclosure Statement"; if the Property was built prior to 1978, then attach Lead-Based Paint Hazards Addendum to this Agreement) and provide
Buyer with a completed Property Information Sheet ("Property Information Sheet") concerning the Property, duly executed by or on behalf of Seller in the current
form or equivalent to that published by the AIR within 10 or days following the Date of Agreement. Buyer has 10 days from the receipt of said disclosures to approve or disapprove the matters disclosed.
(b) Physical Inspection. Buyer has 10 or days following the receipt of the Property Information Sheet or the Date of Agreement, whichever
is later, to satisfy itself with regard to the physical aspects and size of the Property. Seller recommends that the Buyer have the Property inspected with regard to the
possible presence of wood destroying pests and organisms by a licensed structural pest control company. Any such report shall be paid for by Buyer. (c) Hazardous Substance Conditions Report. Buyer has 30 or days following the receipt of the Property Information Sheet or the Date of
Agreement, whichever is later, to satisfy itself with regard to the environmental aspects of the Property. Seller recommends that Buyer obtain a Hazardous Substance
Conditions Report concerning the Property and relevant adjoining properties. Any such report shall be paid for by Buyer. A "Hazardous Substance" for purposes of this Agreement is defined as any substance whose nature and/or quantity of existence, use, manufacture, disposal or effect, render it subject to Federal, state or local regulation, investigation, remediation or removal as potentially injurious to public health or welfare. A "Hazardous Substance Condition" for purposes of this
Agreement is defined as the existence on, under or relevantly adjacent to the Property of a Hazardous Substance that would require remediation and/or removal under applicable Federal, state or local law.
(d) Soil Inspection. Buyer has 30 or days following the receipt of the Property Information Sheet or the Date of Agreement, whichever is
later, to satisfy itself with regard to the condition of the soils on the Property. Seller recommends that Buyer obtain a soil test report. Any such report shall be paid for
by Buyer. Seller shall provide Buyer copies of any soils report that Seller may have within 10 days following the Date of Agreement.
(e) Governmental Approvals. Buyer has 30 or days following the Date of Agreement to satisfy itself with regard to approvals and permits from governmental agencies or departments which have or may have jurisdiction over the Property and which Buyer deems necessary or desirable in connection with
its intended use of the Property, including, but not limited to, permits and approvals required with respect to zoning, planning, building and safety, fire, police,
handicapped and Americans with Disabilities Act requirements, transportation and environmental matters.
(f) Conditions of Title. Escrow Holder shall cause a current commitment for title insurance ("Title Commitment") concerning the Property issued by the Title Company, as well as legible copies of all documents referred to in the Title Commitment ("Underlying Documents"), and a scaled and dimensioned plot showing
the location of any easements to be delivered to Buyer within 10 or days following the Date of Agreement. Buyer has 10 days from the receipt of the
Title Commitment, the Underlying Documents and the plot plan to satisfy itself with regard to the condition of title. The disapproval by Buyer of any monetary
encumbrance, which by the terms of this Agreement is not to remain against the Property after the Closing, shall not be considered a failure of this contingency, as
Seller shall have the obligation, at Seller's expense, to satisfy and remove such disapproved monetary encumbrance at or before the Closing.
(g) Survey. Buyer has 30 or days following the receipt of the Title Commitment and Underlying Documents to satisfy itself with regard to any ALTA title supplement based upon a survey prepared to American Land Title Association ("ALTA") standards for an owner's policy by a licensed surveyor, showing
the legal description and boundary lines of the Property, any easements of record, and any improvements, poles, structures and things located within 10 feet of either
side of the Property boundary lines. Any such survey shall be prepared at Buyer's direction and expense. If Buyer has obtained a survey and approved the ALTA title
supplement, Buyer may elect within the period allowed for Buyer's approval of a survey to have an ALTA extended coverage owner's form of title policy, in which event
Buyer shall pay any additional premium attributable thereto.
(h) Existing Leases and Tenancy Statements. Seller shall within 10 or days following the Date of Agreement provide both Buyer and Escrow Holder with legible copies of all leases, subleases or rental arrangements (collectively, "Existing Leases") affecting the Property, and with a tenancy statement
(" Estoppel Certificate ") in the latest form or equivalent to that published by the AIR, executed by Seller and/or each tenant and subtenant of the Property. Seller shall use its best efforts to have each tenant complete and execute an Estoppel Certificate. If any tenant fails or refuses to provide an Estoppel Certificate then Seller shall
complete and execute an Estoppel Certificate for that tenancy. Buyer has 10 days from the receipt of said Existing Leases and Estoppel Certificates to satisfy itself with
regard to the Existing Leases and any other tenancy issues.
(i) Owner's Association. Seller shall within 10 or days following the Date of Agreement provide Buyer with a statement and transfer
package from any owner's association servicing the Property. Such transfer package shall at a minimum include: copies of the association's bylaws, articles of incorporation, current budget and financial statement. Buyer has 10 days from the receipt of such documents to satisfy itself with regard to the association.
(j) Other Agreements. Seller shall within 10 or days following the Date of Agreement provide Buyer with legible copies of all other
agreements ("Other Agreements") known to Seller that will affect the Property after Closing. Buyer has 10 days from the receipt of said Other Agreements to satisfy itself with regard to such Agreements.
(k) Financing. If paragraph 5 hereof dealing with a financing contingency has not been stricken, the satisfaction or waiver of such New Loan contingency.
(I) Existing and new Purchase Money Notes. If paragraph 3.1(c) has not been stricken, Seller shall within 10 or days following the Date of
Agreement provide Buyer with legible copies of the Existing Notes, Existing Deeds of Trust and related agreements (collectively, "Loan Documents") to which the Property will remain subject after the Closing. Escrow Holder shall promptly request from the holders of the Existing Notes a beneficiary statement ("Beneficiary
Statement") confirming: (1) the amount of the unpaid principal balance, the current interest rate, and the date to which interest is paid, and (2) the nature and
amount of any impounds held by the beneficiary in connection with such loan. Buyer has 10 or days following the receipt of the Loan Documents and
Beneficiary Statements to satisfy itself with regard to such financing. Buyer's obligation to close is conditioned upon Buyer being able to purchase the Property without acceleration or change in the terms of any Existing Notes or charges to Buyer except as otherwise provided in this Agreement or approved by Buyer, provided,
however, Buyer shall pay the transfer fee referred to in paragraph 3.2 hereof. Likewise if Seller is to carry back a Purchase Money Note then Seller shall within 10 or
days following the Date of Agreement provide Buyer with a copy of the proposed Purchase Money Note and Purchase Money Deed of Trust. Buyer has
days from the receipt of such documents to satisfy itself with regard to the form and content thereof.
(m) Personal Property. In the event that any personal property is included in the Purchase Price, Buyer has 10 or days following the Date of
Agreement to satisfy itself with regard to the title condition of such personal property. Seller recommends that Buyer obtain a UCC-1 report. Any such report shall be paid for by Buyer. Seller shall provide Buyer copies of any liens or encumbrances affecting such personal property that it is aware of within 10 or days
following the Date of Agreement.
(n) Destruction, Damage or Loss. Subsequent to the Date of Agreement and prior to Closing there shall not have occurred a destruction of, or damage or
loss to, the Property or any portion thereof, from any cause whatsoever, which would cost more than \$10,000.00 to repair or cure. If the cost of repair or cure is
\$10,000.00 or less, Seller shall repair or cure the loss prior to the Closing. Buyer shall have the option, within 10 days after receipt of written notice of a loss costing more than \$10,000.00 to repair or cure, to either terminate this Agreement or to purchase the Property notwithstanding such loss, but without deduction or offset
against the Purchase Price. If the cost to repair or cure is more than \$10,000.00, and Buyer does not elect to terminate this Agreement, Buyer shall be entitled to any
insurance proceeds applicable to such loss. Unless otherwise notified in writing, Escrow Holder shall assume no such destruction, damage or loss has occurred prior
to Closing. (a) Material Change Buyer shall have 10 days following receipt of written notice of a Material Change within which to satisfy itself with regard to such
(o) Material Change. Buyer shall have 10 days following receipt of written notice of a Material Change within which to satisfy itself with regard to such change. "Material Change" shall mean a substantial adverse change in the use, occupancy, tenants, title, or condition of the Property that occurs after the date of this offer and prior to the Closing. Unless otherwise notified in writing, Escrow Holder shall assume that no Material Change has occurred prior to the Closing.

INITIALS

INITIALS
© 2019 AIR CRE. All Rights Reserved.
ROFA-11.30, Revised 10-12-2022

- (p) Seller Performance. The delivery of all documents and the due performance by Seller of each and every undertaking and agreement to be performed by Seller under this Agreement.
- (q) Brokerage Fee. Payment at the Closing of such brokerage fee as is specified in this Agreement or later written instructions to Escrow Holder executed by Seller and Brokers ("Brokerage Fee"). It is agreed by the Parties and Escrow Holder that Brokers are a third party beneficiary of this Agreement insofar as the Brokerage Fee is concerned, and that no change shall be made with respect to the payment of the Brokerage Fee specified in this Agreement, without the written consent of Brokers.
- 9.2 The contingencies specified in subparagraphs 9.1(a) through (m) are for the benefit of, and may be waived by, Buyer, and are referred to collectively as "Buyer's Contingencies" and individually as a "Buyer's Contingency."
- 9.3 Buyer's timely and written disapproval or conditional approval of a Buyer's Contingency or any other matter that is subject to Buyer's approval in this Agreement shall constitute disapproval thereof ("Disapproved Item(s)"). Concurrent with notice of a Disapproved Item, Buyer may make a request to Seller regarding such Disapproved Item ("Buyer's Request"). If Buyer fails to make a timely and written Buyer's Request, then this Agreement shall terminate due to the non-satisfaction and non-waiver of a contingency. Seller may respond to a Buyer's Request within 10 days following Seller's receipt thereof ("Seller's Response"). Seller's acceptance of a Buyer's Request shall amend this Agreement accordingly. If Seller fails to provide a timely and written Seller's Response, then Seller's Response shall be deemed to be a rejection of Buyer's Request. Buyer may, within 10 days following the earlier of Buyer's receipt of a Seller's Response (which is not an acceptance of Buyer's Request) or the date of Seller's deemed rejection of a Buyer's Request ("Buyer's Reply Period"), reply to a Seller's Response ("Buyer's Reply") and elect to (i) terminate this Agreement due to the non-satisfaction and non-waiver of the applicable contingency, (ii) accept the Seller's Response in which event this Agreement shall be amended accordingly, or (iii) withdraw Buyer's Request and waive the Disapproved Item in which event Buyer shall accept the Property subject to the Disapproved Item. If Buyer fails to provide a timely and written Buyer's Reply, then Buyer shall be deemed to have elected to terminate this Agreement as of the end of the Buyer's Reply Period. The date Buyer accepts a Seller's Response or withdraws a Buyer's Request and waives a Disapproved Item shall be the date of Buyer's approval of the Disapproved Item. A Party shall provide to Escrow Holder copy of all notices of a Disapproved Item, Buyer's Request, Seller's Response and Buyer's Reply and Escrow Holder shall promptly provide copies thereof to the other Party. Unless the Parties in writing agree otherwise, if the Expected Closing Date is a specific calendar date and a Buyer's Reply Period expires after such specific calendar date, then notwithstanding paragraph 1.1, the Expected Closing Date shall be extended to be 3 business days after the earlier of the date Buyer withdraws a Buyer's Request and waives the applicable Disapproved Item or Buyer accepts the applicable Seller's Response.
- 9.4 The Parties acknowledge that extensive local, state and Federal legislation establish broad liability upon owners and/or users of real property for the investigation and remediation of Hazardous Substances. The determination of the existence of a Hazardous Substance Condition and the evaluation of the impact of such a condition are highly technical and beyond the expertise of Brokers. The Parties acknowledge that they have been advised by Brokers to consult their own technical and legal experts with respect to the possible presence of Hazardous Substances on the Property or adjoining properties, and Buyer and Seller are not relying upon any investigation by or statement of Brokers with respect thereto. The Parties hereby assume all responsibility for the impact of such Hazardous Substances upon their respective interests herein.

10. Documents and Other Items Required at or Before Closing.

- 10.1 Five days prior to the Closing date Escrow Holder shall obtain an updated Title Commitment concerning the Property from the Title Company and provide copies thereof to each of the Parties.
 - 10.2 Seller shall deliver to Escrow Holder in time for delivery to Buyer at the Closing:
 - (a) Grant or general warranty deed, duly executed and in recordable form, conveying fee title to the Property to Buyer.
 - (b) If applicable, the Beneficiary Statements concerning Existing Note(s).
- (c) If applicable, the Existing Leases and Other Agreements together with duly executed assignments thereof by Seller and Buyer. The assignment of Existing Leases shall be on the most recent Assignment and Assumption of Lessor's Interest in Lease form published by the AIR or its equivalent.
- (d) An affidavit executed by Seller to the effect that Seller is not a "foreign person" within the meaning of Internal Revenue Code Section 1445 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Internal Revenue Service such sum as is required by applicable Federal law with respect to purchases from foreign sellers.
- (e) If the Property is located in California, an affidavit executed by Seller to the effect that Seller is not a "nonresident" within the meaning of California Revenue and Tax Code Section 18662 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Franchise Tax Board such sum as is required by such statute.
 - f) If applicable, a bill of sale, duly executed, conveying title to any included personal property to Buyer.
 - (g) If the Seller is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the sale of the Property.
 - (h) Keys and/or other means to operate all locks, security systems, alarms, mailboxes and garage doors (collectively "Keys")
 - 10.3 Buyer shall deliver to Seller through Escrow:
- (a) The cash portion of the Purchase Price and such additional sums as are required of Buyer under this Agreement shall be deposited by Buyer with Escrow Holder, by federal funds wire transfer, or any other method acceptable to Escrow Holder in immediately collectable funds, no later than 2:00 P.M. on the business day prior to the Expected Closing Date provided, however, that Buyer shall not be required to deposit such monies into Escrow if at the time set for the deposit of such monies Seller is in default or has indicated that it will not perform any of its obligations hereunder. Instead, in such circumstances in order to reserve its rights to proceed Buyer need only provide Escrow with evidence establishing that the required monies were available.
- (b) If a Purchase Money Note and Purchase Money Deed of Trust are called for by this Agreement, the duly executed originals of those documents, the Purchase Money Deed of Trust being in recordable form, together with evidence of fire insurance on the improvements in the amount of the full replacement cost naming Seller as a mortgage loss payee, and a real estate tax service contract (at Buyer's expense), assuring Seller of notice of the status of payment of real property taxes during the life of the Purchase Money Note.
 - (c) The Assignment and Assumption of Lessor's Interest in Lease form specified in paragraph 10.2(c) above, duly executed by Buyer.
 - (d) Assumptions duly executed by Buyer of the obligations of Seller that accrue after Closing under any Other Agreements.
 - (e) If applicable, a written assumption duly executed by Buyer of the loan documents with respect to Existing Notes.
 - (f) If the Buyer is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the purchase of the Property.
- 10.4 At Closing, Escrow Holder shall: (i) deliver all Keys to Buyer, and (ii) cause to be issued to Buyer a standard coverage (or ALTA extended, if elected pursuant to 9.1(g)) owner's form policy of title insurance effective as of the Closing, issued by the Title Company in the full amount of the Purchase Price, insuring title to the Property vested in Buyer, subject only to the exceptions approved by Buyer. In the event there is a Purchase Money Deed of Trust in this transaction, the policy of title insurance shall be a joint protection policy insuring both Buyer and Seller.

IMPORTANT: IN A PURCHASE OR EXCHANGE OF REAL PROPERTY, IT MAY BE ADVISABLE TO OBTAIN TITLE INSURANCE IN CONNECTION WITH THE CLOSE OF ESCROW SINCE THERE MAY BE PRIOR RECORDED LIENS AND ENCUMBRANCES WHICH AFFECT YOUR INTEREST IN THE PROPERTY BEING ACQUIRED. A NEW POLICY OF TITLE INSURANCE SHOULD BE OBTAINED IN ORDER TO ENSURE YOUR INTEREST IN THE PROPERTY THAT YOU ARE ACQUIRING.

11. Prorations and Adjustments.

- 11.1 Toxes. Applicable real property taxes and special assessment bonds shall be prorated through Escrow as of the date of the Closing, based upon the latest tax bill available. The Parties agree to prorate as of the Closing any taxes assessed against the Property by supplemental bill levied by reason of events occurring prior to the Closing. Payment of the prorated amount shall be made promptly in cash upon receipt of a copy of any supplemental bill.
- 11.2 Insurance. WARNING: Any insurance which Seller may have maintained will terminate on the Closing. Buyer is advised to obtain appropriate insurance to cover the Property.
- 11.3 Rentals, Interest and Expenses. Scheduled rentals, interest on Existing Notes, utilities, and operating expenses shall be prorated as of the date of Closing. The Parties agree to promptly adjust between themselves outside of Escrow any rents received after the Closing.
 - 11.4 Security Deposit. Security Deposits held by Seller shall be given to Buyer as a credit to the cash required of Buyer at the Closing.
- 11.5 Post Closing Matters. Any item to be prorated that is not determined or determinable at the Closing shall be promptly adjusted by the Parties by appropriate cash payment outside of the Escrow when the amount due is determined.
- 11.6 Variations in Existing Note Balances. In the event that Buyer is purchasing the Property subject to an Existing Deed of Trust(s), and in the event that a Beneficiary Statement as to the applicable Existing Note(s) discloses that the unpaid principal balance of such Existing Note(s) at the closing will be more or less than the amount set forth in paragraph 3.1(c) hereof ("Existing Note Variation"), then the Purchase Money Note(s) shall be reduced or increased by an amount equal to such Existing Note Variation. If there is to be no Purchase Money Note, the cash required at the Closing per paragraph 3.1(a) shall be reduced or increased by the amount of such Existing Note Variation.
- 11.7 Variations in New Loan Balance. In the event Buyer is obtaining a New Loan and the amount ultimately obtained exceeds the amount set forth in paragraph 5.1, then the amount of the Purchase Money Note, if any, shall be reduced by the amount of such excess.
 - 11.8 Owner's Association Fees. Escrow Holder shall: (i) bring Seller's account with the association current and pay any delinquencies or transfer fees from

INITIALS		INITIALS	
----------	--	----------	--

Seller's proceeds, and (ii) pay any up front fees required by the association from Buyer's funds.

12. Representations and Warranties of Seller and Disclaimers.

- 12.1 Seller's warranties and representations shall survive the Closing and delivery of the deed for a period of 3 years, and any lawsuit or action based upon them must be commenced within such time period. Seller's warranties and representations are true, material and relied upon by Buyer and Brokers in all respects. Seller hereby makes the following warranties and representations to Buyer and Brokers:
- (a) Authority of Seller. Seller is the owner of the Property and/or has the full right, power and authority to sell, convey and transfer the Property to Buyer as provided herein, and to perform Seller's obligations hereunder.
- (b) Maintenance During Escrow and Equipment Condition At Closing. Except as otherwise provided in paragraph 9.1(n) hereof, Seller shall maintain the Property until the Closing in its present condition, ordinary wear and tear excepted.
- (c) Hazardous Substances/Storage Tanks. Seller has no knowledge, except as otherwise disclosed to Buyer in writing, of the existence or prior existence on the Property of any Hazardous Substance, nor of the existence or prior existence of any above or below ground storage tank.
- (d) Compliance. Except as otherwise disclosed in writing, Seller has no knowledge of any aspect or condition of the Property which violates applicable laws, rules, regulations, codes or covenants, conditions or restrictions, or of improvements or alterations made to the Property without a permit where one was required, or of any unfulfilled order or directive of any applicable governmental agency or casualty insurance company requiring any investigation, remediation, repair, maintenance or improvement be performed on the Property.
- (e) Changes in Agreements. Prior to the Closing, Seller will not violate or modify any Existing Lease or Other Agreement, or create any new leases or other agreements affecting the Property, without Buyer's written approval, which approval will not be unreasonably withheld.
- (f) Possessory Rights. Seller has no knowledge that anyone will, at the Closing, have any right to possession of the Property, except as disclosed by this Agreement or otherwise in writing to Buyer.
 - (g) Mechanics' Liens. There are no unsatisfied mechanics' or materialmens' lien rights concerning the Property.
- (h) Actions, Suits or Proceedings. Seller has no knowledge of any actions, suits or proceedings pending or threatened before any commission, board, bureau, agency, arbitrator, court or tribunal that would affect the Property or the right to occupy or utilize same.
- (i) Notice of Changes. Seller will promptly notify Buyer and Brokers in writing of any Material Change (see paragraph 9.1(o)) affecting the Property that becomes known to Seller prior to the Closing.
- (j) No Tenant Bankruptcy Proceedings. Seller has no notice or knowledge that any tenant of the Property is the subject of a bankruptcy or insolvency proceeding.
 - (k) No Seller Bankruptcy Proceedings. Seller is not the subject of a bankruptcy, insolvency or probate proceeding.
- (I) Personal Property. Seller has no knowledge that anyone will, at the Closing, have any right to possession of any personal property included in the Purchase Price nor knowledge of any liens or encumbrances affecting such personal property, except as disclosed by this Agreement or otherwise in writing to Buyer.
- 12.2 Buyer hereby acknowledges that, except as otherwise stated in this Agreement, Buyer is purchasing the Property in its existing condition and will, by the time called for herein, make or have waived all inspections of the Property Buyer believes are necessary to protect its own interest in, and its contemplated use of, the Property. The Parties acknowledge that, except as otherwise stated in this Agreement, no representations, inducements, promises, agreements, assurances, oral or written, concerning the Property, or any aspect of the occupational safety and health laws, Hazardous Substance laws, or any other act, ordinance or law, have been made by either Party or Brokers, or relied upon by either Party hereto.
- 12.3 In the event that Buyer learns that a Seller representation or warranty might be untrue prior to the Closing, and Buyer elects to purchase the Property anyway then, and in that event, Buyer waives any right that it may have to bring an action or proceeding against Seller or Brokers regarding said representation or warranty.
- 12.4 Any environmental reports, soils reports, surveys, and other similar documents which were prepared by third party consultants and provided to Buyer by Seller or Seller's representatives, have been delivered as an accommodation to Buyer and without any representation or warranty as to the sufficiency, accuracy, completeness, and/or validity of said documents, all of which Buyer relies on at its own risk. Seller believes said documents to be accurate, but Buyer is advised to retain appropriate consultants to review said documents and investigate the Property.

13. Possession.

Possession of the Property shall be given to Buyer at the Closing subject to the rights of tenants under Existing Leases.

14. Buyer's Entry.

At any time during the Escrow period, Buyer, and its agents and representatives, shall have the right at reasonable times and subject to rights of tenants, to enter upon the Property for the purpose of making inspections and tests specified in this Agreement. No destructive testing shall be conducted, however, without Seller's prior approval which shall not be unreasonably withheld. Following any such entry or work, unless otherwise directed in writing by Seller, Buyer shall return the Property to the condition it was in prior to such entry or work, including the re-compaction or removal of any disrupted soil or material as Seller may reasonably direct. All such inspections and tests and any other work conducted or materials furnished with respect to the Property by or for Buyer shall be paid for by Buyer as and when due and Buyer shall indemnify, defend, protect and hold harmless Seller and the Property of and from any and all claims, liabilities, losses, expenses (including reasonable attorneys' fees), damages, including those for injury to person or property, arising out of or relating to any such work or materials or the acts or omissions of Buyer, its agents or employees in connection therewith.

15. Further Documents and Assurances.

The Parties shall each, diligently and in good faith, undertake all actions and procedures reasonably required to place the Escrow in condition for Closing as and when required by this Agreement. The Parties agree to provide all further information, and to execute and deliver all further documents, reasonably required by Escrow Holder or the Title Company.

16. Attorneys' Fees.

If any Party or Broker brings an action or proceeding (including arbitration) involving the Property whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees and costs. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred.

17. Prior Agreements/Amendments.

- 17.1 This Agreement supersedes any and all prior agreements between Seller and Buyer regarding the Property.
- 17.2 Amendments to this Agreement are effective only if made in writing and executed by Buyer and Seller.

18. Broker's Rights.

18.1 If this sale is not consummated due to the default of either the Buyer or Seller, the defaulting Party shall be liable to and shall pay to Brokers the Brokerage Fee that Brokers would have received had the sale been consummated. If Buyer is the defaulting party, payment of said Brokerage Fee is in addition to any obligation with respect to liquidated or other damages.

18.2 Upon the Closing, Brokers are authorized to publicize the facts of this transaction.

19. Notices.

19.1 Whenever any Party, Escrow Holder or Brokers herein shall desire to give or serve any notice, demand, request, approval, disapproval or other communication, each such communication shall be in writing and shall be delivered personally, by messenger, or by mail, postage prepaid, to the address set forth in this agreement or by facsimile transmission, electronic signature, digital signature, or email.

19.2 Service of any such communication shall be deemed made on the date of actual receipt if personally delivered, or transmitted by facsimile transmission, electronic signature, digital signature, or email. Any such communication sent by regular mail shall be deemed given 48 hours after the same is mailed. Communications sent by United States Express Mail or overnight courier that guarantee next day delivery shall be deemed delivered 24 hours after delivery of the same to the Postal Service or courier. If such communication is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.

19.3 Any Party or Broker hereto may from time to time, by notice in writing, designate a different address to which, or a different person or additional persons to whom, all communications are thereafter to be made.

20. Duration of Offer.

20.1 If this offer is not accepted by Seller on or before 5:00 P.M. according to the time standard applicable to the city of ______ on the date of

INITIALS
© 2019 AIR CRE. All Rights Reserved.

•	•		ment between the Parties as described in pa conditionally accepting the last outstanding	•
THE PARTIES AGREE THAT IT WOU WOULD BE SUFFERED BY SELLER I OF ALL CONTINGENCIES PROVIDE AMOUNT OF UPO	F BUYER FAILS TO PERFORM ITS OBLIGA D FOR THE BUYER'S BENEFIT, BUYER BR	DIFFICULT TO FIX, PRIOR TO S ATIONS UNDER THIS AGREEM EACHES THIS AGREEMENT, SE UYER SHALL BE RELEASED FRI	th Parties). IIGNING THIS AGREEMENT, THE ACTUAL DA ENT. THEREFORE, IF, AFTER THE SATISFACTI ELLER SHALL BE ENTITLED TO LIQUIDATED D OM ANY FURTHER LIABILITY TO SELLER, AN	ION OR WAIVER AMAGES IN THE
	Buyer's Initials		Seller's Initials	
22.1 ANY CONTROVERSY AS SHALL BE DETERMINED BY BINDIF ITS COMMERCIAL ARBITRATION RESUCH CONTROVERSY SHALL BE AS EXPERIENCE IN THE TYPE OF REAL ACCORDANCE WITH APPLICABLE AGREEMENT AND ANY AMENDM PERMITTED IN ACCORDANCE WITH AWARD WITHIN 30 DAYS AFTER TO PARAGRAPH 16 HEREOF AND SHALL AND LEGAL ARGUMENT AS THE ACCOMPETENT JURISDICTION NOTW 22.2 BUYER'S RESORT TO OF BUYER FOR DAMAGES AND/OR SPONICH EVENT SUCH AWARD SHALL 22.3 NOTICE: BY INITIALING OF DISPUTES" PROVISION DECIDED HAVE THE DISPUTE LITIGATED IN A APPEAL, UNLESS SUCH RIGHTS AR AGREEING TO THIS PROVISION, YOU TO THIS ARBITRATION PROVISION	ING ARBITRATION ADMINISTERED BY THE SULES ("COMMERCIAL RULES"). ARBITRATED BY A SINGLE ARBITRATOR, AND LESTATE THAT IS THE SUBJECT OF THIS A LAW OF THE JURISDICTION WHERE THE ENTS THERETO, AND UPON THE EVIDEN HIT THE COMMERCIAL RULES OR STATE LAME CONCLUSION OF THE HEARING, WHI LAW OF THE COMMERCIAL RULES OR STATE LAW OF COMPENSATION OR ADMINISTRATIVE OF COMPENSATION OR ADMINISTRATIVE OF COMPENSATION OR ADMINISTRATIVE OF A PARTY OF THE PROPERTY OF THE PROPER	IQUIDATED DAMAGES AND/O E JUDICIAL ARBITRATION & M ATION HEARINGS SHALL BE H PPOINTED UNDER THE COMM AGREEMENT. THE ARBITRATOI PROPERTY IS LOCATED, THE I ICE PRODUCED AT AN ARBITRAT ICH MAY INCLUDE ATTORNEY DPINION. THE FAILURE OR RE E CHARGES SHALL CONSTITU A DEFAULT JUDGMENT AGAI S AN AWARD. JUDGMENT MAY POCCEDINGS SHALL NOT BAR I'LL THE ARBITRATION RESULTS Y BUTCH FOR ANY DISPUTE ARIS ED BY CALIFORNIA LAW AND Y N THE SPACE BELOW YOU ARE RATION OF DISPUTES" PROVIS NDER THE AUTHORITY OF THE	OR BUYER IS ENTITLED TO THE RETURN OF THE BUYER IS ENTITLED TO THE RETURN OF THE BUYER IS ENTITLED TO THE RETURN OF THE BUYER IN ACCURED IN THE COUNTY WHERE THE PROPERTY MERCIAL RULES WHO HAS HAD AT LEAST 5 YER SHALL HEAR AND DETERMINE SAID CONTINITED OF THE PARTIES AS EXPRESSED MATION HEARING. PRE-ARBITRATION DISCOMINED PROCEEDINGS. THE ARBITRATOR SHALES OF A PARTY TO PAY SUCH PARTY'S RETE A WAIVER BY SUCH PARTY TO PRESENT IN THE ABSENTY OF THE NON-PAYING PARTY IN THE ABSENTY OF THE NON-PAYING PARTY IN THE ABSENTY OF THE MATION HEARING TO APPEAR THEREAT. RESULT IN A COURT OF COMPETENT JURISDICATION OF THE MATTERS INCLUDED IN THE COURT OF THE MATTERS INCLUDED IN THE COUNTY OF THE MATTERS INCLUDED IN THE COU	ORDANCE WITH Y IS LOCATED. YEARS OF ROVERSY IN IN THIS VERY SHALL BE LL RENDER AN RTY PER EQUIRED SHARE EVIDENCE OR NCE OF EVIDENCE JIST OF CTION BY THE ED DAMAGES, IN HE "ARBITRATION T POSSESS TO COVERY AND RATION AFTER YOUR AGREEMENT
WE HAVE READ AND UNDERSTAND PROVISION TO NEUTRAL ARBITRAT		IIT DISPUTES ARISING OUT OF	THE MATTERS INCLUDED IN THE "ARBITRAT	TON OF DISPUTES
			*	
	Buyer's Initials	<u> </u>	Seller's Initials	
Paragraphs 21 and 22 are each inc Agreement accomplished by mear 23.2 Applicable Law; Venue	orporated into this Agreement only if inings of electronic signature or similar technors. This Agreement shall be governed by, a	tialed by both Parties at the ti nology shall be legal and bindi and paragraph 22.3 is amende	not paragraphs 21 and 22 are initialed by bo me that the Agreement is executed. Signati ng. ed to refer to, the laws of the state in which is iated in the county in which the Property is	ures to this the Property is

- - 23.3 **Time of Essence**. Time is of the essence of this Agreement.

. it shall be deemed automatically revoked.

- 23.4 Counterparts. This Agreement may be executed by Buyer and Seller in counterparts, each of which shall be deemed an original, and all of which together shall constitute one and the same instrument. Escrow Holder, after verifying that the counterparts are identical except for the signatures, is authorized and instructed to combine the signed signature pages on one of the counterparts, which shall then constitute the Agreement.
- 23.5 Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY OR ARISING OUT OF THIS AGREEMENT.
- 23.6 Conflict. Any conflict between the printed provisions of this Agreement and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions. Seller and Buyer must initial any and all handwritten provisions.
- 23.7 1031 Exchange. Both Seller and Buyer agree to cooperate with each other in the event that either or both wish to participate in a 1031 exchange. Any party initiating an exchange shall bear all costs of such exchange. The cooperating Party shall not have any liability (special or otherwise) for damages to the exchanging Party in the event that the sale is delayed and/or that the sale otherwise fails to qualify as a 1031 exchange.
- 23.8 Equal Housing Opportunity. The Property is being sold in compliance with local, state and federal anti-discrimination laws.
 23.9 NOTICE REGARDING GAS AND HAZARDOUS LIQUID TRANSMISSION PIPELINES. This notice is being provided simply to inform you that information about the general location of gas and hazardous liquid transmission pipelines is available to the public via the National Pipeline Mapping System (NPMS) Internet Web site maintained by the United States Department of Transportation at http://www.npms.phmsa.dot.gov/. To seek further information about possible transmission pipelines near the property, you may contact your local gas utility or other pipeline operators in the area. Contact information for pipeline operators is searchable by ZIP Code and county on the NPMS Internet Web site.
- Disclosures Regarding The Nature of a Real Estate Agency Relationship.
- 24.1 The Parties and Brokers agree that their relationship(s) shall be governed by the principles set forth in the applicable sections of the California Civil Code, as summarized in paragraph 24.2.
- 24.2 When entering into a discussion with a real estate agent regarding a real estate transaction, a Buyer or Seller should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Buyer and Seller acknowledge being advised by the Brokers in this transaction, as follows:
- (a) Seller's Agent. A Seller's agent under a listing agreement with the Seller acts as the agent for the Seller only. A Seller's agent or subagent has the following affirmative obligations: (1) To the Seller: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Seller. (2) To the Buyer and the Seller: a. Diligent exercise of reasonable skills and care in performance of the agent's duties. b. A duty of honest and fair dealing and good faith. c. A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set
- Buyer's Agent. A selling agent can, with a Buyer's consent, agree to act as agent for the Buyer only. In these situations, the agent is not the Seller's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Seller. An agent acting only for a Buyer has the following affirmative obligations. (1) To the Buyer: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Buyer. (2) To the Buyer and the Seller: a. Diligent exercise of reasonable skills and care in performance of the agent's duties. b. A duty of honest and fair dealing and good faith. c. A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

affirmative duties set forth above.

- (c) Agent Representing Both Seller and Buyer. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Seller and the Buyer in a transaction, but only with the knowledge and consent of both the Seller and the Buyer. (1) In a dual agency situation, the agent has the following affirmative obligations to both the Seller and the Buyer: a. A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Seller or the Buyer. b. Other duties to the Seller and the Buyer as stated above in their respective sections (a) or (b) of this paragraph 24.2. (2) In representing both Seller and Buyer, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating to either Buyer's or Seller's financial position, motivations, bargaining position, or other personal information that may impact price, including Seller's willingness to accept a price less than the listing price or Buyer's willingness to pay a price greater than the price offered. (3) The above duties of the agent in a real estate transaction do not relieve a Seller or Buyer from the responsibility to protect their own interests. Buyer and Seller should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Buyer has the duty to exercise reasonable care to protect Buyer, including as to those facts about the Property which are known to Buyer or within Buyer's diligent attention and observation. Both Seller and Buyer should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.
- (d) Further Disclosures. Throughout this transaction Buyer and Seller may receive more than one disclosure, depending upon the number of agents assisting in the transaction. Buyer and Seller should each read its contents each time it is presented, considering the relationship between them and the real estate agent in this transaction and that disclosure. Buyer and Seller each acknowledge receipt of a disclosure of the possibility of multiple representation by the Broker representing that principal. This disclosure may be part of a listing agreement, buyer representation agreement or separate document. Buyer understands that Broker representing Buyer may also represent other potential buyers, who may consider, make offers on or ultimately acquire the Property. Seller understands that Broker representing Seller may also represent other sellers with competing properties that may be of interest to this Buyer. Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this transaction may be brought against Broker more than one year after the Date of Agreement and that the liability (including court costs and attorneys' fees), of any Broker with respect to any breach of duty, error or omission relating to this Agreement shall not exceed the fee received by such Broker pursuant to this Agreement; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker
- 24.3 Confidential Information. Buyer and Seller agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- 25. Construction of Agreement. In construing this Agreement, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Agreement. Whenever required by the context, the singular shall include the plural and vice versa. This Agreement shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.

26.	^ 44:4:4	Provisions.

Additional provisions of this offer, if any, are as follows or are attached hereto by an addendum or addenda consisting of paragraphs ______ through ______ (If there are no additional provisions write "NONE".)

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS AGREEMENT OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

- SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS AGREEMENT.
- 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PROPERTY. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF WOOD DESTROYING PESTS OR ORGANISMS AND/OR HAZARDOUS SUBSTANCES, THE ZONING OF THE PROPERTY, THE INTEGRITY AND CONDITION OF ANY STRUCTURES AND OPERATING SYSTEMS, WHETHER OR NOT THE PROPERTY MEETS CURRENT CODES AND WHETHER OR NOT THE PROPERTY IS SUBJECT TO RENT CONTROL OR OTHER LOCAL RENTAL PROPERTY REGULATIONS.

Date:

WARNING: IF THE PROPERTY IS LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THIS AGREEMENT MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PROPERTY IS LOCATED.

NOTE:

- 1. THIS FORM IS NOT TO BE USED WITH PROPERTIESS CONTAINING LESS THAN 5 RESIDENTIAL UNITS.
- 2. IF EITHER PARTY IS A CORPORATION, IT IS RECOMMENDED THAT THIS AGREEMENT BE SIGNED BY TWO CORPORATE OFFICERS.

The undersigned Buyer offers and agrees to buy the Property on the terms and conditions stated and acknowledges receipt of a copy hereof.

BROKER	BUYER
Attn:	Ву:
Title:	Name Printed:
	Title:
Address:	Phone:
Phone:	Fax:
Fax:	Email:
Email:	
Federal ID No.:	By:
Broker DRE License #:	By: Name Printed:
Agent DRE License #:	Title:
	Phone:
	Fax:
\ \	Email:
	Address:
V	Federal ID No.:
27. Acceptance.	
•	and hereby agrees to sell the Property to Buyer on the terms and conditions therein specified.
27.2 In consideration of real estate brokerage service rendered b	y Brokers, Seller agrees to pay Brokers a real estate Brokerage Fee in a sum equal to
% of the Purchase Price to be divided between the Broke	rs as follows: Seller's Broker % and Buyer's Broker %. This
<u> </u>	p pay such Brokerage Fee to Brokers out of the proceeds accruing to the account of Seller at the
Closing.	
27.3 Seller acknowledges receipt of a copy hereof and authorizes	Brokers to deliver a signed copy to Buyer.
NOTE: A DRODERTY INCORMATION SHEET IS REQUIRED TO BE DELIVER	RED TO RUYER BY SELLER LINDER THIS AGREEMENT

INITIALS

INITIALS
© 2019 AIR CRE. All Rights Reserved.
ROFA-11.30, Revised 10-12-2022

	Date:
BROKER	SELLER
Attn: Title:	By:
nue	Name Printed: Title:
Address:	Phone:
Phone:	Fax:
Fax: Email:	Email:
Federal ID No.:	
Broker DRE License #:	By:
Agent's DRE License #:	Name Printed: Title:
	Phone:
	Fax:
	Email:
	Address:
	Federal ID No.:
	* 213-687-8777 * contracts@aircre.com
NOTICE: No part of these works may be repr	oduced in any form without permission in writing.
	<) \
	() '
	X
() \	
	,
() '	
X	
() \	
'	
1	

Limited liability company: Manager(s)

Husband and wife: Community Property or Separate property

Joint tenants: All joint tenants

Tenants in common: All tenants in common

Corporation:

Board chairman, president or any vice president

and

Secretary, asst. secretary, CFO or asst. treasurer

General partnership: General partner with authority

Limited Partnership: General partner

Para. 9.1	Contingency topic	Time Seller has to deliver	Time Buyer has to approve
(a)	Seller's Mandatory Disclosure Statement and Property Information Sheet	to Escrow within 10 days after Date of Agreement ("DOA")	10 days after Buyer's receipt
(b)	Physical Inspection		10 days after the later of Buyer's receipt of Property Information Sheet or DOA
(c)	Hazardous Substance Condition Report		30 days after later of receipt of Property Information Sheet or DOA
(d)	Soil Inspection	within 10 days after DOA	30 days after later of receipt of Property Information Sheet or DOA
(e)	Governmental approvals		30 days after DOA
(f)	Title	within 10 days of DOA, Escrow to deliver Title Commitment, Underlying Documents and scaled and dimensioned plot of easements	10 days after Buyer's receipt of Title Commitment, Underlying Documents and plot plan
(g)	ALTA Survey		30 days after Buyer's receipt of Title Commitment and Underlying Documents
(h)	Leases and Estoppels	within 10 days after DOA	10 days after Buyer's receipt
(i)	Owner's Association	within 10 days after DOA	10 days after Buyer's receipt
(j)	Other Agreements	within 10 days after DOA	10 days after Buyer's receipt
(k)	New Loan Financing		[] after DOA
(1)	Existing Notes	within 10 days after DOA provide copies of loan documents	10 days after Buyer's receipt of loan documents and beneficiary statement
(m)	Personal Property	within 10 days after DOA provide copies of liens	10 days after DOA



BUYER'S NOTICE DISAPPROVING OF CONTINGENCIES

Date of Notice:
Buyer's Disapproval Notice Number:
Escrow Holder:
In accordance with the terms, conditions and provisions of the Standard Offer, Agreement And Escrow Instructions For Purchase Of Real Estate dated for reference purposes only as of (as amended) ("Agreement") by and between ("Buyer") and ("Seller") regarding the property commonly known as ("Property"), the defined terms of which Agreement shall have the same meaning in this Notice, Buyer disapproves of the following contingencies (check all that apply):
Disclosures (Para. 9.1(a))
Physical Inspection (Para. 9.1(b))
Hazardous Substance Conditions Report (Para. 9.1(c))
Soil Inspection (Para. 9.1(d))
Governmental approvals (Para. 9.1(e))
Conditions of Title (Para. 9.1(f))
ALTA Survey (Para. 9.1(g))
Existing Leases and Tenancy Statements (Para. 9.1(h))
Owner's Association (Para. 9.1(i))
Other Agreements (Para. 9.1(j))
Financing (Para. 5 and 9.1(k))
Existing and New Purchase Money Notes (Para. 9.1(I))
Personal Property (Para. 9.1(m))
All of the above contingencies
All of the above contingencies except (add description):
Other matter that is subject to Buyer's approval (add description):
Buyer requests ("Buyer's Request(s)") Seller to (check and complete as appropriate):
At Seller's cost, make the following repairs, replacements and/or take the following actions by the following dates (describe):
Reduce the Purchase Price by the amount of (insert amount):
At the Closing, credit Buyer the amount of (insert amount):
Extend time for the following contingencies by the following periods of time (describe contingencies and extended periods of time):
Extend the Expected Closing Date as follows (insert):
Other (add description):
BUYER
Dug.
By: Name Printed:
Title:
AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com
NOTICE: No part of these works may be reproduced in any form without permission in writing.

NITIALS
© 2022 AIR CRE. All Rights Reserved.
BNDC-1.00, Revised 10-12-2022

INITIALS



SELLER'S RESPONSE TO BUYER'S REQUESTS

BUYER'S REQUESTS
Responding to Buyer's Request Number: Date of Seller's Response: Escrow Holder:
In accordance with the terms, conditions and provisions of the Standard Offer, Agreement And Escrow Instructions For Purchase Of Real Estate dated for reference purposes only as of (as amended) ("Agreement") by and between ("Buyer") and ("Seller") regarding the property commonly known as ("Property"), the defined terms of which Agreement shall have the same meaning in this Notice, Seller responds to the above-referenced Buyer's Request as follows (check and complete as appropriate): Seller Denies Buyer's Request. Seller agrees to the entirety of Buyer's Request; the Agreement is amended accordingly. Seller does not agree to Buyer's Request except the following parts (check all that apply): At Seller's cost, make the following repairs, replacements and/or take the following actions by the following dates (describe): Reduce the Purchase Price by the amount of (insert amount) At the Closing, credit Buyer the amount of (insert amount) Extend time for Buyer's approval of the following contingency by the following period of time (describe contingency and period of time): Extend Expected Closing Date as follows (insert): Other (add description):
SELLER
By:Name Printed:
Title:
AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2022 AIR CRE. All Rights Reserved.
SRBR-1.00, Revised 10-12-2022



BUYER'S REPLY TO SELLER'S RESPONSE

IO SELLER 5 RESPONSE
Buyer's Request Number: Date of Buyer's Reply: Escrow Holder:
In accordance with the terms, conditions and provisions of the Standard Offer, Agreement And Escrow Instructions For Purchase Of Real Estate dated for reference purposes only as of (as amended) ("Agreement") by and between ("Buyer") and ("Seller") regarding the property commonly known as ("Property"), the defined terms of which Agreement shall have the same meaning in this Notice, Buyer replies ("Buyer's Reply Seller's Response to the above-referenced Buyer's Request as follows (select one only):
Buyer accepts Seller's Response; the Agreement is amended accordingly.
Buyer withdraws Buyer's Request and waives the applicable Disapproved Item; Buyer accepts the Property subject to such waived Disapproved Item. Buyer elects to terminate the Agreement due to the non-satisfaction and non-waiver of the applicable contingency.
BUYER
By: Name Printed:
Title:
AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS



CONFIDENTIALITY AGREEMENT

This Agreement ("Agreement") is made and entered into by and between	("Owner") and ("Potential Buyer") and shall be effective as of
WHEREAS, Owner owns certain real property commonly known as (street address, city,	state, zip) ("Property"), and
WHEREAS, Potential Buyer is considering acquiring the Property from Owner, and	
WHEREAS, in order to evaluate whether or not to acquire the Property Potential Buyer documents, $ \\$	wants Owner to provide Potential Buyer with certain information and/or
NOW THEREFORE, in consideration of the promises contained herein and other good at acknowledged Owner and Potential Buyer agree as follows:	nd valuable consideration, the receipt and sufficiency of which is hereby
1. Agreement to Provide Information. Owner agrees to make available to Potential concerning the Property. As a condition to furnishing such information and documents Buyer agrees to treat such information in accordance with the provisions of this Agreemany person or entity, collectively all of such person's or entity's officers, directors, empladvisors, including, without limitation, attorneys, accountants, consultants, bankers and	to Potential Buyer and its "Representatives" (as defined below), Potential ment. As used herein the term "Representatives" shall mean with respect to oyees, affiliates, partners, members, representatives, agents, brokers or
2. Definition of Evaluation Material. The term "Evaluation Material" means all infor representatives provides to Potential Buyer or to its Representatives and all notes, anal Potential Buyer or its Representatives that contain, reflect or are based upon, in whole pursuant hereto; provided, however, that the foregoing shall not require Potential Buyer compilations, studies, interpretations or other documents prepared by Potential Buyer information which is generally available to the public.	yses, compilations, studies, interpretations or other documents prepared by or in part, the information furnished to Potential Buyer or its Representatives er to reveal or disclose to Owner in any manner any such notes, analyses,
3. Use of Evaluation Material and Confidentiality.	< _ \
(a) Potential Buyer hereby agrees that it and its Representatives shall use the Event Property, that the Evaluation Material shall be kept confidential and that Potential Buyer any manner whatsoever and, that such information shall be disclosed only to such of Posole purpose of evaluating the purchase of the Property and who are informed that such for any disclosure by any of its Representatives of any Evaluation Materials which if managrees, at its sole expense, to take all reasonable measures to prevent or restrain its Re Evaluation Material. In no event shall Potential Buyer or its Representatives use the Evaluation Material is strictly prohibited.	otential Buyer's Representatives who need to know such information for the chinformation is confidential. In any event, Potential Buyer shall be responsible de by Potential Buyer would be a breach of this Agreement, and Potential Buyer presentatives from prohibited or unauthorized disclosure or use of the
(b) Potential Buyer agrees that, without the prior written consent of Owner, it a Evaluation Material has been made available to Potential Buyer, the fact that discussion any of the terms, conditions or other matters then being discussed with respect thereto Property's tenants, partners, members, employees, contractors, managers, insurers, managers, mana	o (including the status thereof), and (ii) shall not contact any of Owner's or the
(c) If Potential Buyer decides that it does not wish to proceed with the purchase upon the request of Owner for any reason, Potential Buyer and shall promptly deliver to Representatives by or on behalf of Owner pursuant hereto. At such time Potential Buyer Potential Buyer or its Representatives and no copy thereof shall be retained. Notwithst and its Representatives shall continue to be bound by Potential Buyer's obligations of continue to be bound by Potential Buyer's obligations.	er shall also destroy or deliver to Owner all Evaluation Material prepared by tanding the return or destruction of the Evaluation Material, Potential Buyer
4. Accuracy of Evaluation Material . Potential Buyer understands and acknowledges Seller and those representing Potential Buyer) are not making any representation or wa Material. Potential Buyer agrees that Owner, its Representatives and/or the Brokers sh relating to or resulting from the use of the Evaluation Material or any errors therein or made in a final definitive agreement regarding the purchase, as, when and if executed a	arranty, express or implied, as to the accuracy or completeness of the Evaluation all not have any liability to Potential Buyer or to any of its Representatives omissions therefrom. Only those representations or warranties which are
5. Remedies . It is understood and agreed that money damages would not be a suffi Representatives and that Owner shall be entitled to equitable relief, including injunctio shall not be deemed to be the exclusive remedies for a breach by Potential Buyer or an remedies available at law or equity to Owner.	ns and specific performance, as a remedy for any such breach. Such remedies
6. Miscellaneous.	
(a) Waivers and Amendments. No failure or delay by Owner in exercising any ri any single or partial exercise thereof preclude any other or future exercise thereof or the may be amended only with the written consent of Owner and Potential Buyer.	
(b) Effect of Agreement. Neither Potential Buyer nor Owner nor any of their respect to conducting negotiations relating to or consummating the purchase of the Prodiscretion, to reject any and all proposals made by Potential Buyer with regard to purch into a definitive agreement, with regard to any other transaction, with any other person discussions and negotiations with Potential Buyer at any time.	operty by virtue of this Agreement. Owner reserves the right, in its sole hasing the Property and to engage in discussions and negotiations, and to enter
(c) Governing Law. The parties hereby agree that this Agreement shall be gove Property is located. The parties hereby irrevocably consent to the exclusive jurisdiction actions, suits or proceedings arising out of or relating to this Agreement.	rned by and construed in accordance with the laws of the State in which the of the state courts located in the State in which the Property is located for any
(d) Severability. If any provision of this Agreement is found to violate any statut agency or exchange, such invalidity shall not be deemed to affect any other provision h provision shall be deemed deleted herefrom to the minimum extent necessary to cure	ereof or the validity of the remainder of this Agreement, and such invalid
(e) Counterparts. This Agreement may be executed in multiple counterparts, eagreement. Telecopied and PDF signatures may be used in place of original signatures such telecopied document or PDF copies. Signatures to this Agreement accomplished binding.	on this Agreement, and the parties intend to be bound by the signatures on
(f) Entire and Sole Agreement. This Agreement constitutes the entire understa hereof and supersedes any and all prior or contemporaneous oral or written communic	nding and agreement between the parties with respect to the subject matter cations with respect hereto, all of which are merged herein.
WHEREFORE, by their signature below, the parties hereto acknowledge that they have legally binding Agreement.	reviewed carefully the terms of this Agreement and they understand it is a

INITIALS

© 2017 AIR CRE. All Rights Reserved. CA-1.02, Revised 10-22-2020

POTENTIAL BUYER:	OWNER:
Ву:	Ву:
Name Printed:	Name Printed:
Title:	Title:

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS



COUNTER OFFER - SELLER

Dated: By and Between Seller:	
Buyer:	
Property Address:	(street address, city, state, zip)
APN:	
	d Agreement for Purchase of Real Estate" dated, regarding the above-referenced Property and Parties by of the Offer is attached hereto and incorporated herein by reference.
 Seller hereby accepts the Offer provided th 	e Buyer agrees to the following changes in the Offer:
(Please check the appropriate box or boxes	
Initialed pen and ink changes have bee	
Seller shall be deemed to have initiale	d paragraph 21 regarding liquidated damages and paragraph 22 regarding arbitration of disputes
The changes are detailed in the adden	dum attached hereto consisting of paragraphs through
The changes are set forth below.	
	JIRE INITIALS BY ALL PARTIES, BUT ARE NOT INITIALED BY ALL PARTIES, ARE EXCLUDED FROM THE FINAL AGREEMENT
UNLESS SPECIFICALLY REFERENCED FOR INCLUSION	,
 REMAINING TERMS: All of the terms and onereby approved by Seller. 	onditions contained in the Offer which have not been specifically modified or deleted by this Counter Offer are
,	(If checked and initialed) MULTIPLE COUNTER OFFER: Concurrently herewith, Seller is also making a Counter Offer to or may not be the same as in this Counter Offer. Acceptance of this Counter Offer by Buyer shall not, therefore, be need by Seller in paragraph 11 below.
receipt by Seller's authorized agent,	eserves the right to continue to offer the Property for sale, and to accept any other offer at any time prior to actual, of a copy of this Counter Offer duly accepted and signed by Buyer and, if applicable, the re-signing of this Counter 's acceptance of such a prior offer shall revoke this Counter Offer.
	e and be revoked, without further notice, at 5:00 pm on, unless it is accepted and signed by Buyer and r's agent (as specified in paragraph 4) prior to said time. Upon expiration, Buyer's Deposit, if any, shall be returned.
 TIME: Time is of the essence. 	
7. BINDING EFFECT : This Counter Offer, wher Signatures to this Counter Offer accomplished by	fully executed by both Buyer and Seller, and delivered and received as specified above, shall be a binding contract. The means of electronic signature or similar technology shall be legal and binding. NO REPRESENTATION OR THIS AGREEMENT OR THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS AGREEMENT OR THE
NOTE: 1. THIS FORM IS NOT FOR USE IN CONN 2. IF EITHER PARTY IS A CORPORATION,	ES ARE ADVISED TO SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS AGREEMENT. ECTION WITH THE SALE OF RESIDENTIAL PROPERTY. T IS RECOMMENDED THAT THIS AGREEMENT BE SIGNED BY TWO OF ITS CORPORATE OFFICERS. operty on the terms and conditions stated herein.
SELLER'S BROKER	SELLER
Attn:	Ву:
Title:	Name Printed:
Address: Phone:	Title: Phone:
Fax:	Fax:
Email:	Email:
Federal ID No.: Broker DRE License #:	D. u.
Agent DRE License #:	By: Name Printed:
	Title:
)	Phone:
	Fax: Email:
*	Address:
	Federal ID No.:
 ACCEPTANCE: Buyer accepts the foregoing deliver a signed copy to Seller. 	Counter Offer and agrees to purchase the Property on the terms and conditions specified, and authorizes Brokers to
BUYER'S BROKER	BUYER
Attn:	By:
Title:	Name Printed:
INITIALS	INITIALS

© 2017 AIR CRE. All Rights Reserved. COS-3.02, Revised 10-22-2020

Address:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Federal ID No.:	
Broker DRE License #:	By: Name Printed:
Agent Bitz Election II.	Title:
	Phone:
	Fax:
	Email:
	Address:
	Federal ID No.:
10. ACKNOWLEDGMENT OF RECEIPT am/pm.	: Seller's agent hereby acknowledges that a copy of the Counter Offer executed by Buyer was received on at
	By: Name Printed:
	Name Finiteu.
	ATURE LINE: (Applicable only if paragraph 3 is checked and initialed.) By signing below, Seller accepts this Multiple Counter Note to Seller: Do NOT sign in this paragraph until after Buyer has accepted the Counter Offer by signing in paragraph 9).
SELLER	
- -	
	Ву:
	Name Printed:
	Title:
	Phone: Fax:
	Email:
	Ву:
	Name Printed:
	Title:
	Phone:
	Fax: Email:
NOT	AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com ICE: No part of these works may be reproduced in any form without permission in writing.



COUNTER OFFER - BUYER

Dated:			
By and Between Seller:			
Buyer:			
Property Address:			
APN:	(street address, city, state, zip)		
APN.			
	Thereafter, Seller executed and deliv	chase of Real Estate" dated, regarding the ered to Buyer the "Counter Offer - Seller" dated cated herein by reference.	
Buyer hereby accepts Seller's Counter Of	fer provided the Seller agrees to the f	ollowing changes in said Counter Offer:	
(Please check the appropriate box or box	es)		./ \
Initialed pen and ink changes have b	een made to the Offer.		
Seller shall be deemed to have initia	led paragraph 21 regarding liquidated	d damages and paragraph 22 regarding arbitration of di	sputes
The changes are detailed in the adde	endum attached hereto consisting of	paragraphs through	X
The changes are set forth below.			
-			
	OURSE INITIALS BY ALL BARTIES BUT I		
UNLESS SPECIFICALLY REFERENCED FOR INCLU		ARE NOT INITIALED BY ALL PARTIES, ARE EXCLUDED FRO	JM THE FINAL AGREEMENT
2. REMAINING TERMS : All of the terms an Counter Offer are hereby approved by Buyer.	d conditions contained in the Seller's	Counter Offer which have not been specifically modified	ed or deleted by this
3. EXPIRATION : This Counter Offer shall exp			red and signed by Seller and
a fully executed copy is actually received by Bu Counter Offer may be executed in counterparts		d time. Upon expiration, Buyer's Deposit, if any, shall b	e promptly returned. This
4. TIME : Time is of the essence.			
5. BINDING EFFECT : This Counter Offer, wh	en fully executed by both Buyer and S	Seller, and delivered and received as specified above, sl	hall be a binding contract.
RECOMMENDATION IS MADE BY AIR CRE OR B	Y ANY BROKER AS TO THE LEGAL SUFF	imilar technology shall be legal and binding. NO REPRE FICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF TH	HIS AGREEMENT OR THE
TRANSACTION TO WHICH IT RELATES. THE PAR	RTIES ARE ADVISED TO SEEK ADVICE O	F COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCE	S OF THIS AGREEMENT.
NOTE:	INECTION WITH THE SALE OF RESIDEN	ITIAL DRODERTY	
2. IF EITHER PARTY IS A CORPORATION	N, IT IS RECOMMENDED THAT THIS AG	REEMENT BE SIGNED BY TWO OF ITS CORPORATE OFF	ICERS.
6. The undersigned Buyer agrees to purchase	se the Property on the terms and con-	ditions stated herein.	
BUYER'S BROKER		BUYER	
Attn:		By:	
Title:		Name Printed:	
Address:	•	Title:	
Phone:		Phone:	
Fax:		Fax: Email:	
Email:			
Federal ID No.: Broker DRE License #:		Ву:	
Agent DRE License #:		Name Printed:	
		Title:	
		Phone: Fax:	
		Email:	
//'			
\nearrow		Address:	
		Federal ID No.:	
7. ACCEPTANCE : Seller accepts the foregoin a signed copy to Buyer.	ng Counter Offer and agrees to sell the	e Property on the terms and conditions specified, and a	authorizes Brokers to deliver
SELLER'S BROKER		SELLER	
Atta			
Attn: Title:		By:	
		Name Printed: Title:	
Address:		Phone:	
Phone:		Fax:	
Fax: Email:		Email:	
INITIALS		INITIALS	
HINITIALS		HINTHALA	

© 2017 AIR CRE. All Rights Reserved. COB-2.02, Revised 10-22-2020

By:
Name Printed:
Title:
Phone:
Fax:
Email:
Address:
Federal ID No.:
a copy of the Counter Offer executed by Seller was received on at
By: Name Printed:
Numer integral
* 212 CO7 C777 * contract Onlines
* 213-687-8777 * contracts@aircre.com oduced in any form without permission in writing.
•

INITIALS



ADDENDUM TO THE STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS FOR PURCHASE OF REAL ESTATE

By and Between	
Buyer:	
Seller:	_
	_
Property Address:	(street address situ state 7in)
	(street address, city, state, zip)
"Agreement"). In the event of any conflict	of the above-referenced Agreement (said Agreement and the Addendum are hereinafter collectively referred to as the between the provisions of this Addendum shall control. AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2017 AIR CRE. All Rights Reserved.
APA-1.03, Revised 10-22-2020

INITIALS



_____ AMENDMENT TO PURCHASE AND SALE AGREEMENT

THIS AMENDMENT TO PURCHASE AND SALE AGREEMENT AND ESCROW INSTRUCTI between ("Buyer") and ("Seller") (collectively, the "Prate of the collection	•
WHEREAS, Buyer and Seller are parties to that certain Standard Offer, Agreement as ("Agreement") with respect to the property commonly known as (street address, ci	
WHEREAS, the Agreement has not been previously amended except (if there are no	exceptions write "NONE"):
WHEREAS, the Parties have opened an escrow with("Escrow Holder");
WHEREAS, the Parties now desire to amend the Agreement and their instructions to	e Escrow Holder ("Escrow Instructions");
NOW, THEREFORE, for Buyer's payment of ten dollars (\$10) to Seller and other good acknowledged, the Parties hereby amend the Agreement and the Escrow Instructio	
The defined terms of the Agreement shall have the same meaning in this Amendment or similar technology shall be legal and binding. This Amendment may be executed together shall constitute one and the same instrument. This Amendment shall not as a whole, as if both Parties had prepared it. Except as specifically amended by this and continue in full force and effect.	in counterparts, each of which shall be deemed an original, and all of which be construed as if prepared by one of the Parties, but according to its fair meaning
ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE O CONSEQUENCES OF THIS AMENDMENT. THE PARTIES ARE URGED TO SEEK ADVICE AMENDMENT.	
EXECUTED as of the date first written above. By Seller:	By Buyer:
Ву:	Ву:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax: Email:	Fax: Email:
Ву:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Address:	Address:
Federal ID No.:	Federal ID No.:
AIR CRE * https://www.aircra.com *	212-687-8777 * contracts@aircra.com

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS

© 2020 AIR CRE All Rights Reserved



SELLER'S MANDATORY DISCLOSURE STATEMENT

(DO NOT USE IN SALE OF RESIDENTIAL PROPERTIES WITH 1-4 UNITS)

This Seller's Mandatory Disclosure Statement ("Disclosure") is made pursuant to the	e requirements of and shall be a part of the STANDARD OFFER, AG	REEMENT
AND ESCROW INSTRUCTIONS FOR PURCHASE OF REAL ESTATE (See paragraph 9.1(a)	·	
purposes as of, regarding that certain real property commonly known is the Seller and is the Buyer. Note: This Disclosure does	in as (street address, city, state, zip): (" Property ") wherein ses not replace the Property Information Sheet published by AIR CRE ("A	
unless not required by the Purchase Agreement, Seller should also provide Buyer th		
1. NATURAL HAZARDS DISCLOSURES. To make the disclosures required by Califor Government Code Section 8589.4 (dam failure inundation area), California Government cone), California Public Resources Code Sections 4125, 4142 and 4291 (wildlands fire California Public Resources Code Section 2696 (seismic hazard zone) ("Natural Hazar Report") prepared by:	ment Code Sections 51178 or 51179, and 51182 (very high fire hazard se re area), California Public Resources Code Section 2622 (earthquake fau	everity ılt zone), and
myNHD Inc., 800-814-2922, https://www.myNHD.com., or	,, and the Natural Hazards Report is atta	ached.
2. EARTHQUAKE SAFETY. If the improvements on the Property were constructed or reinforced masonry walls together with wood frame floors or roofs, or (ii) unreinf Government Code §§8875.6, 8875.9, 8893.2, and 8893.3; and California Civil Code § Owner's Guide to Earthquake Safety (the "Booklet") published by the California Seis	forced masonry walls, then California Business & Professions Code §10: §2079.9 require Seller to provide Buyer a copy of The Commercial Prop	147;
https://ssc.ca.gov/forms_pubs/cog. Seller is not required to provide the Bookle Seller is hereby providing to Buyer.	let to Buyer. Seller is required to provide the Booklet to Buyer, whi	ich Booklet
3. FLOOD DISASTER INSURANCE. If Seller or Seller's predecessor-in-interest has perconditioned upon obtaining and maintaining flood insurance on the Property, Federathat they will be required to obtain and maintain such insurance on the Property and by a flood disaster, the purchaser may be required to reimburse the Federal Government.	ral law, i.e., 42 U.S.C. §5154a, mandates that prospective purchasers be not that if such insurance is not maintained and the Property is thereafte	e advised er damaged
Seller's knowledge, Federal flood disaster assistance has or has not been put has been received, the law specifies that the required notice be "contained in document of the contained in docu	previously received with regard to the Property. Note: if such disaster as ments evidencing the transfer of ownership".	ssistance
4. WATER HEATER BRACING. The Property does or does not contain or or more water heaters of 120 gallons or less, then as required by California Health a have or have not been braced, strapped and/or anchored in accordance wi		
5. TITLE INSURANCE. If the Purchase Agreement does not provide that Buyer wil title insurance. In accordance with California Civil Code §1057.6, Buyer is advised as	ill be provided with title insurance, Buyer is strongly urged to consider p	ourchasing
IMPORTANT: IN A PURCHASE OR EXCHANGE OF REAL PROPERTY, IT MAY BE ADVISAL SINCE THERE MAY BE PRIOR RECORDED LIENS AND ENCUMBRANCES WHICH AFFECT INSURANCE SHOULD BE OBTAINED IN ORDER TO ENSURE YOUR INTEREST IN THE PR	T YOUR INTEREST IN THE PROPERTY BEING ACQUIRED. A NEW POLICY	
6. METHAMPHETAMINE LABORATORY ACTIVITY. As required by California Health	n and Safety Code §25400.28, Seller notifies Buyer that a government o	rder
identifying the Property as contaminated by methamphetamine laboratory activity been received by Seller.	has been received by Seller (copy of such order is attached), or	has not
7. HAZARDOUS SUBSTANCES. If the Property is non-residential property, then as	s required by California Health and Safety Code §25359.7 Seller notifies	Buyer that:
Seller neither knows nor has reasonable cause to believe that any release of ha	azardous substance (as such terms are defined by California Health and	Safety Code
§§25320 and 25316) has come to be located on or beneath the Property, or Se substance has come to be located on or beneath the Property.	eller knows or has reasonable cause to believe that a release of hazardo	ous
8. WATER CONSERVING PLUMBING FIXTURES. For multi-family residential (conta California Civil Code §1101.5 requires all noncompliant plumbing fixtures to be repla	• • • • • • • • • • • • • • • • • • • •	
by California Civil Code §1101.5(e), Seller discloses that the Property was built a	after January 1, 1994 and is exempt from this statute; U does include	е
noncompliant plumbing fixtures; or — does not include any noncompliant plumbi	ing fixtures.	
9. DEATH ON THE PROPERTY. As required by California Civil Code §1710.2, Seller the Property in the three years before the date of Buyer's offer to purchase the Property		cupant of
10. OTHER.		
Seller believes the information provided in this Disclosure by Seller is true and corre that Seller has exercised good faith in the selection of a Natural Hazards Expert, Sell-information contained in the Natural Hazards Report, and makes no representation Hazards Report. Buyer agrees the Natural Hazards Report satisfies Seller's legal duty Report shall be deemed to be an expert dealing with matters within the scope of its Natural Hazards Disclosures, and California Civil Code §1103.4 which deals with a se knowledge shall be deemed to apply to Seller.	ler has not independently verified, and Seller is not obligated to verify, to warranty as to the truth or accuracy of any information contained in ty to make the Natural Hazards Disclosures, the provider of the Natural sexpertise with respect to the examination and written report regarding	the n the Natura Hazards g the
The hazards referenced in this Disclosure may limit the ability of the owner of the Pradisaster. The maps on which the Natural Hazards Report are based estimate when property will be affected by a natural disaster. Buyer is advised to obtain profession	re natural hazards exist and are not definitive indicators of whether or r	not a
Applicable laws and the areas covered by the various natural hazard zones, etc. may the Natural Hazards Report becomes inaccurate as a result of any governmental acti has actual knowledge that the information has become inaccurate. Prior to the clos is advised to verify with appropriate local governmental agencies, such as the city ar established natural hazard zones in addition to those listed above. Signatures to this shall be legal and binding. This Disclosure may be executed in counterparts.	tion, map revision, changed information, or other act or occurrence, unl se of escrow, Buyer may wish to again check the status of the Property. and/or county in which the Property is located, to determine if they have	less Seller Also, Buyer e
The statements herein are not intended to be full or complete disclosure of all post the Property, Buyer is advised to: 1. Review the entirety of all laws applicable to such disclosed matter.	ssible impacts on Buyer and/or the Property. If a matter is disclosed r	egarding
INITIALS	INITIALS	
© 2017 AIR CRE. All Rights Reserved.	Last Edited: 9/30/2022 2	2:03 PM

SMD-7.01, Revised 01-18-2022

- 2. Seek advice of counsel as to the legal consequence of any disclosed matters.
- 3. Retain appropriate consultants to review and investigate the impact of all such disclosed matters.

Neither AIR CRE nor any of the brokers make any representation, warranty or recommendation regarding the legal sufficiency, legal effect or any other consequences of this Disclosure.

Date:			
SELLER			
Ву:			
Name Printed:			
Title:			

Buyer acknowledges receipt of this Disclosure together with the Natural Hazards Report, the Booklet and any other documents referenced herein (as applicable) as being provided to Buyer, and agrees to the above terms and conditions.

Date: _____
BUYER

By: _____
Name Printed: _____
Title: ____

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2017 AIR CRE. All Rights Reserved.
SMD-7.01, Revised 01-18-2022



PROPERTY INFORMATION SHEET

(For the sale or leasing of non-residential properties)

PREFACE:

Purpose: This Statement is NOT a warranty as to the actual condition of the Property/Premises. The purpose is, instead, to provide the brokers and the potential buyer/lessee with important information about the Property/Premises which is currently in the actual knowledge of the Owner and which the Owner is required by law to disclose.

Actual Knowledge: For purposes of this Statement the phrase 'actual knowledge' means: the awareness of a fact, or the awareness of sufficient information and circumstances so as to cause one to believe that a certain situation or condition probably exists. TO WHOM IT MAY CONCERN: ("Owner"), owns the Property/Premises commonly known as (street address, city, state, zip) located in the County of generally described as (describe briefly the nature of the Premises or Property) (hereinafter "Property"), and certifies that: Material Physical Defects. Owner has no actual knowledge of any material physical defects in the Property or any improvements and structures thereon, including, but not limited to the roof, except (if there are no exceptions write "NONE"): Equipment. Owner has no actual knowledge that the heating, ventilating, air conditioning, plumbing, loading doors, electrical and lighting systems, life safety systems, security systems and mechanical equipment existing on the Property as of the date hereof, if any, are not in good operating order and condition, except (if there are no exceptions write "NONE"): Owner has no actual knowledge of any leases, financing agreements, liens or other agreements affecting any equipment which is being included with the Property, except (if there are no exceptions write "NONE"): Soil Conditions. Owner has no actual knowledge that the Property has any slipping, sliding, settling, flooding, ponding or any other grading, drainage or soil problems, except (if there are no exceptions write "NONE"): **Utilities.** Owner represents and warrants that the Property is served by the following utilities (check the appropriate boxes) public sewer system and the cost of installation thereof has been fully paid, \square private septic system, \square electricity, \square natural gas, \square domestic water, telephone, and U other: Insurance. Owner has no actual knowledge of any insurance claims filed regarding the Property during the preceding 3 years, except (if there are no exceptions write "NONE"): Compliance With Laws. Owner has no actual knowledge of any aspect or condition of the Property which violates applicable laws, rules, regulations, codes, or covenants, conditions or restrictions, or of improvements or alterations made to the Property without a permit where one was required, or of any unfulfilled order or directive of any applicable government agency or of any casualty insurance company that any work of investigation, remediation, repair, maintenance or improvement is to be performed on the Property, except (if there are no exceptions write "NONE"): Hazardous Substances and Mold. A. Owner has no actual knowledge of the Property ever having been used as a waste dump, of the past or present existence of any above or below ground storage tanks on the Property, or of the current existence on the Property of asbestos, transformers containing PCB's or any hazardous, toxic or infectious substance whose nature and/or quantity of existence, use, manufacture or effect, render it subject to Federal, state or local regulation, investigation, remediation or removal as potentially injurious to public health or welfare, except (if there are no exceptions write "NONE"): Owner represents and warrants that it is not currently, and never has been engaged in the business of hauling waste, and never stored hazardous substances on the Property, except (if there are no exceptions write "NONE").

C. Owner has no actual knowledge of the existence on the Property of hazardous levels of any mold or fungi defined as toxic under applicable state or Federal law, except (if there are no exceptions write "NONE"); Fire Damage. Owner has no actual knowledge of any structure on the Property having suffered material fire damage, except (if there are no exceptions write "NONE"): Actions, Suits or Proceedings. Owner has no actual knowledge that any actions, suits or proceedings are pending or threatened before any court, arbitration tribunal, governmental department, commission, board, bureau, agency or instrumentality that would affect the Property or the right or ability of an owner or tenant to convey, occupy or utilize the Property, except (if there are no exceptions write "NONE"): Owner has not served any Notices of Default on any of the tenants of the Property which have not been resolved except (if there are no exceptions write "NONE"): 10. Governmental Proceedings. Owner has no actual knowledge of any existing or contemplated condemnation, environmental, zoning, redevelopment agency plan or other land use regulation proceedings which could detrimentally affect the value, use and operation of the Property, except (if there are no exceptions write "NONE"): 11. Unrecorded Title Matters. Owner has no actual knowledge of any encumbrances, covenants, conditions, restrictions, easements, licenses, liens, charges or other matters which affect the title of the Property that are not recorded in the official records of the county recorder where the Property is located, except (if there are no exceptions write "NONE"): Leases. Owner has no actual knowledge of any leases, subleases or other tenancy agreements affecting the Property, except (if there are no exceptions write "NONF"): 13. Options. Owner has no actual knowledge of any options to purchase, rights of first refusal, rights of first offer or other similar agreements affecting the Property, except (if there are no exceptions write "NONE"): 14. Short Sale/Foreclosure. The ability of the Owner to complete a sale of the Property is contingent is not contingent upon obtaining the consent of one or more lenders to conduct a 'short sale', ie. a sale for less that the amount owing on the Property. (This paragraph only needs to be completed if this Property Information Sheet is being completed in connection with the proposed sale of the Property) One or more of any loans secured by the Property is is is not in 15. **Energy Efficiency**. The Property has has not been granted an energy efficiency rating or certification such as one from the U.S. Green Building Council's Leadership in Energy and Environmental Design (LEED) or Seller/Lessor does not know if the Property has been granted such a rating or certificate. If such a rating or certification has been obtained please describe the rating or certification and provide the name of the organization that granted it: 16. Other. (It will be presumed that there are no additional items which warrant disclosure unless they are set forth herein):

INITIALS

INITIALS
© 2017 AIR CRE. All Rights Reserved.
PI-8.03, Revised 10-22-2020

The statements herein will be relied upon by brokers, buyers, lessees, lenders and others. Therefore, Owner and/or the Owner's Property Manager has reviewed and modified this printed statement as necessary to accurately and completely state all the known material facts concerning the Property. To the extent such modifications are not made, this statement may be relied upon as printed. This statement, however, shall not relieve a buyer or lessee of responsibility for independent investigation of the Property. Owner agrees to promptly notify, in writing, all appropriate parties of any material changes which may occur in the statements contained herein from the date this statement is signed until title to the Property is transferred, or the lease is executed. Signatures to this Statement accomplished by means of electronic signature or similar technology shall be legal and binding.

Date: (fill in date of execution)	OWNER
	By: Name Printed: Title:
Buyer/lessee hereby acknowledges receipt of a copy of this Property Information	Sheet on (Fill in date received) BUYER/LESSEE
	By: Name Printed: Title:
	213-687-8777 * contracts@aircre.com duced in any form without permission in writing.
QY.	

INITIALS



LEAD-BASED PAINT HAZARDS ADDENDUM TO STANDARD OFFER, AGREEMENT, AND ESCROW INSTRUCTIONS FOR PURCHASE OF RESIDENTIAL INCOME PROPERTY

(For residential property built prior to 1978)

Date:	
By and Between	
Seller:	
Buyer:	
Property Address:	
	(street address, city, state, zip)
aragraph:	
his Addendum is attached to and mad greement, this Addendum shall contr	e part of the above-referenced Agreement. In the event of any conflict between the provisions of this Addendum and the ol.
the Property is residential property b	uilt prior to 1978, then as required by federal law (42 USC §4852d):
. LEAD WARNING STATEMENT. Th	e following is included in the Agreement:
nay present exposure to lead from hildren may produce permanent i mpaired memory. Lead poisoning o provide the buyer with any infor	residential real property on which a residential dwelling was built prior to 1978 is notified that such property in lead-based paint that may place young children at risk of developing lead poisoning. Lead poisoning in young neurological damage, including learning disabilities, reduced intelligence quotient, behavioral problems, and also poses a particular risk to pregnant women. The seller of any interest in residential real property is required rmation on lead-based paint hazards from risk assessments or inspections in the seller's possession and notify the int hazards. A risk assessment or inspection for possible lead-based paint hazards is recommended prior to
. SELLER'S DISCLOSURES.	
(a) Seller must provide Buyer a	lead hazard information pamphlet, as prescribed by the Administrator of the Environmental Protection Agency. (This pamphlet
an be obtained via the following link:	https://www.epa.gov/lead/protect-your-family-lead-your-home.) Seller has provided to Buyer this pamphlet (or an in the state where the Property is located).
(b) Seller discloses that Seller hoo exceptions, write "NONE"):	as no knowledge of the presence in the Property of any lead-based paint or any lead-based paint hazards, except (if there are
(c) Seller discloses that Seller h xceptions, write "NONE"):	as no lead hazard evaluation report available to Seller, except the following which Seller has provided Buyer (if there are no
	t Buyer a 10-day period (unless the parties mutually agree upon a different period of time) to conduct a risk assessment or ed paint and/or lead-based paint hazards.
. BUYER'S ACKNOWLEDGMENTS.	Buyer certifies that Buyer has:
(a) Read the Lead Warning Stat	ement above and understands its contents;
(b) Received from Seller a lead	hazard information pamphlet described above (or an equivalent pamphlet approved for use in the state where the Property is n reports, if any, described above; and
(c) Either (<i>check the applicable</i>	box):
i. A 10-day opportu	nity (unless the parties mutually agreed upon a different period of time) before becoming obligated under the Agreement to assessment or inspection for the presence of lead-based paint and/or lead-based paint hazards; or
ii. Waived the oppo	rtunity to conduct a risk assessment or inspection for the presence of lead-based paint and/or lead-based paint hazards.
NOT	AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com ICE: No part of these works may be reproduced in any form without permission in writing.
	ice. No part of these works may be reproduced in any form without permission in writing.

INITIALS

INITIALS
© 2022 AIR CRE. All Rights Reserved.
PSADD-1.00, Revised 10-13-2022



STANDARD ESTOPPEL CERTIFICATE - BY LESSEE

TO W	HOM IT MAY CONCERN:					
(stree		("Premises"	as Lessor, and "), which Lease was amendedare specified above).			
Lesse	e hereby certifies as follows:					
amer	ndments and guarantees.) Other	than the documents included	he guarantees, if any, are attached herel I in Exhibit 1 there are no oral or writter ons, write "NONE")	•	• •	
2.	The Lease term commenced or	and expires or	·			
3.	The current monthly rent and e	expense pass-through, if any, a	re as follows:			
		Amount	Day of Month Due	Paid Up Through	<u>Year</u>	
No re	Through ints or pass-throughs have been above.)	prepaid except as reflected in	n the Lease. (It will be presumed that no	o expense pass-throughs are curr	ently required unless set	
4.	The current amount of security	deposit held by Lessor is		7		
			s execution, except as hereinabove iden no exceptions, write "NONE"):		nd effect and contains the	
6. of an	The improvements and space r inducement nature by Lessor h	equired to be provided by Less ave been fulfilled except (if th	sor have been furnished and completed ere are no exceptions, write "NONE"): _	in all respects to the satisfaction	of Lessee, and all promises	
7.	Lessee has no knowledge of an	y uncured defaults by Lessor o	r Lessee under the Lease, except (if the	re are no exceptions, write "NON	IE"):	
	There are no disputes between otions, write "NONE"):		the Lease, the Premises or the improve	ements therein or thereon, excep	ot (if there are no	
	Lessee is in full and complete p	ossession of the Premises and	has not assigned or sublet any portion	of the Premises, except (if there	are no exceptions, write	
	Lessee has no knowledge of an IE"):	y prior sale, transfer, assignme	ent or encumbrance of the Lessor's inter	rest in the Lease, except (if there	are no exceptions, write	
been in cor	made by Lessee, Lessee represe mpliance with all applicable law	ents that to the best of its kno rs, rules and regulations, exce	except (if there are no exceptions, write wledge, all such alterations and addition pt (if there are no exceptions, write "NC Il force and effect, except (if there are no	ns were done in accordance with DNE"):	the terms of the Lease and	
	13. Lessee is not currently the subject of a bankruptcy proceeding and to the best of its knowledge neither Lessor nor any Guarantor is involved in such a proceeding, except (if there are no exceptions, write "NONE"):					
as ne state	cessary to make it an accurate s	statement of the current facts ignatures to this Certificate ac	the statements made in this Estoppel Concerning the Lease. If no such adjust complished by means of electronic sign IE"):	ments have been made, said par	ties may rely upon the	
DATE	:(Fill in date of execution)	<u> </u>	 By:			
			Name Printed: Title: Phone: Fax: Email:			
			By: Name Printed:			
			Title: Phone: Fax:			
			Email:			
	Address:					

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

ESTLE-3.02, Revised 10-22-2020



STANDARD ESTOPPEL CERTIFICATE - BY LESSOR

TO W	HOM IT MAY CONCERN:					
conc	ease ("Lease") dated erning the real property known ("Guarantor(s)") (i	as (street address, city, state, z	rip):	("Premises"), wh	ich Lease was amended	and guaranteed
In lig	ht of the fact that the Lessee ha	as failed to provide an Estoppel	Certificate, Lessor herek	y certifies as follows	:	
	True copies of the above referendments and guarantees.) Othere with respect to the Premises		d in Exhibit 1 there are n	o oral or written agre	, , , ,	•
2.	The Lease term commenced o	n and expires on	·			
3.	The current monthly rent and	expense pass-through, if any, a	re as follows:			
		<u>Amount</u>	Day of Month Due		Paid Up Through	<u>Year</u>
Rent						
No re	Through ents or pass-throughs have been above.)	n prepaid except as reflected in	the Lease. (It will be pro	esumed that no expe	ense pass-throughs are current	ly required unless set
4.	The current amount of security	y deposit held by Lessor is				
5. entir	The Lease has not been modifi e agreement between Lessor a	ied, orally or in writing, since its nd Lessee, except (if there are r				effect and contains the
6. Lesso	The improvements and space or have been fulfilled except (if	required to be provided by Less there are no exceptions, write	•	•	respects, and all promises of a	n inducement nature by
7.	Lessor has no knowledge of an	ny uncured defaults by Lessor o	r Lessee under the Lease	, except (if there are	no exceptions, write "NONE")	:
8. exce	There are no disputes between ptions, write "NONE"):	n Lessor and Lessee concerning	the Lease, the Premises	or the improvement	s therein or thereon, except (i	f there are no
	To the best of Lessor's knowled e are no exceptions, write "NON	-	ete possession of the Pre	mises and has not a	ssigned or sublet any portion o	of the Premises, except (if
	Lessor has no knowledge of ar NE"):	ny prior sale, transfer, assignme	nt or encumbrance of the	e Lessee's interest in	the Lease, except (if there are	e no exceptions, write
If alte	To the best of Lessor's knowled erations or additions have been the terms of the Lease and in c	made by Lessee, Lessor repres	ents that to the best of i	ts knowledge, all suc	ch alterations and additions we	ere done in accordance
12.	The guarantees of the Guarant	tors named above are still in ful	ll force and effect, except	(if there are no exce	eptions, write "NONE"):	
	Lessor is not currently the subjections, except (if there are no e		g and to the best of its kr	nowledge neither Les	ssee nor any Guarantor is invol	ved in such a
as ne	Lessor is aware that buyers, le cessary to make it an accurate ments in this form as printed.	statement of the current facts	concerning the Lease. If	no such adjustments	s have been made, said parties	may rely upon the
15.	The Lessee may be contacted a Mailing address: Telephone: Facsimile: Email:	at:				
16.	Additional items (if there are n	no additional items, write "NON	IE"):			
DATE	(fill in date of execution)			rinted:	-	
•						
	//					
) `		Email: _			
			Ву:			
	7		Name P	rinted:		
						
		AIR CRE * https://www	w.aircre.com * 213-687	'-8777 * contracts@	Daircre.com	

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2017 AIR CRE. All Rights Reserved.



RELEASE OF FUNDS STANDARD SALE ADDENDUM

Date: By and Betwe			
Seller:			
Buyer:			
Property Add	ess:	(street address, city, state, zip)	
		(Street address), city, state, 2197	
Paragraph:			
without any further written or v	erbal instruction		
portion of the purchase price at release of these funds is author	the close of esc zed prior to the e completion or	Buyer for any reason other than a default by Seller under this Agreement. Said funds being released will apply a row and the Seller's demand for cash at close of escrow is reduced accordingly. The Buyer acknowledges that the recording of any document as required under the Buyer's previous instructions, and regardless of the current outcome of this escrow. Payment of the funds prior to the close of escrow as herein authorized is made without	; a
		CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com part of these works may be reproduced in any form without permission in writing.	

INITIALS



ASSIGNMENT AND ASSUMPT	TION OF LESSOR'S INTEREST IN LEASE
	("Assignee") the interest of Assignor under that certain Lease (the "Lease") dated
zip), by and between, as Lessor, and, the "Premises"). Signatures to this Assignment accomplises.	as Lessee, concerning the real property commonly known as (street address, city, state, shed by means of electronic signature or similar technology shall be legal and binding.
2. Assignor also assigns to Assignee the security deposit under said Lease Lease.	in the sum of and the interest of Lessor in any applicable guaranty of said
3. This Assignment shall be effective from and after the recordation, in the Assignee.	e county where the Premises are located, of the deed passing fee title to the Premises to
	Dated:
	ASSIGNOR
	ASSIGNOR
	By:
	Name Printed:
	Title:
	Phone:
	By: Name Printed:
	Title:
	Phone:
ACCEPTAN	ICE AND ASSUMPTION
	and obligations of the Lessor accruing from and after the recordation, in the county to Assignee. Assignee shall defend, indemnify, and hold Assignor harmless with respect te.
	Dated:
	ASSIGNEE
	Ву:
\	Name Printed:
	Title:Phone:
__	By:
	Name Printed:
	Title: Phone:
	om * 213-687-8777 * contracts@aircre.com
NOTICE: No part of these works may be	reproduced in any form without permission in writing.

INITIALS

INITIALS © 2017 AIR CRE. All Rights Reserved. AA-2.03, Revised 10-22-2020



STANDARD INDUSTRIAL/COMMERCIAL SINGLE-TENANT LEASE - NET (DO NOT USE THIS FORM FOR MULTI-TENANT BUILDINGS)

 Basic Provisions ("Basic Provisions"). 1.1 Parties. This Lease ("Lease"), dated for reference purposes only 	ic made by and between	("Lossor") and
("Lessee"), (collectively the "Parties," or individually a "Party").		
1.2 Premises: That certain real property, including all improvements therein (street address, city, state, zip): ("Premises"). The Premises are located that it is a property including all improvements therein (street address, city, state, zip): ("Premises").		
the nature of the property and , if applicable, the " Project ," if the property is located		
1.3 Term: years and months ("Original Term") o	ommencing ("Commencement	nt Date") and ending
("Expiration Date"). (See also Paragraph 3) 1.4 Early Possession: If the Premises are available Lessee may have non-excli	usive possession of the Premises commencin	ng ("Early Possession
Date"). (See also Paragraphs 3.2 and 3.3)	de effection and become	(Consolina Developments A)
1.5 Base Rent: per month ("Base Rent"), payable on the		
Lif this box is checked, there are provisions in this Lease for the Base Rent t 1.6 Base Rent and Other Monies Paid Upon Execution:	be adjusted. See Paragraph	
(a) Base Rent: for the period	1.5	
(b) Security Deposit: ("Security Deposit"). (See also Particular for the period .	agraph 5)	
(d) Other : for		
(e) Total Due Upon Execution of this Lease: 1.7 Agreed Use: (See also Paragraph 6)		_ / > ,
1.8 Insuring Party. Lessor is the "Insuring Party" unless otherwise stated her	ein. (See also Paragraph 8)) /
1.9 Real Estate Brokers. (See also Paragraph 15 and 25)(a) Representation: Each Party acknowledges receiving a Disclosure Re	garding Poal Estato Agonov Polationship, cor	afirms and consents to the following
agency relationships in this Lease with the following real estate brokers ("Broker(s)")		initins and consents to the following
Lessor's Brokerage Firm License No Is the broke	r of (check one): the Lessor; or bo	th the Lessee and Lessor (dual agent).
Lessor's Agent License No is (check one): License No is (check one): License No.	ne Lessor's Agent (salesperson or broker asso	ociate); or both the Lessee's
Agent and the Lessor's Agent (dual agent).		
Lessee's Brokerage Firm License No Is the broke agent).	r of (check one): the Lessee; or bo	oth the Lessee and Lessor (dual
Lessee's Agent License No is (check one): t	he Lessee's Agent (salesperson or broker ass	ociate); or both the Lessee's
Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by I	both Darties Lesson shall pourte the Drelians t	the brokeress for especial to in a
separate written agreement (or if there is no such agreement, the sum of		
by the Brokers.	(Curamentanii) (Car	a alaa Dawaanan 27\
1.10 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed.1.11 Attachments. Attached hereto are the following, all of which constitute and the second of the constitute of the constitute		e also Paragraph 37)
an Addendum consisting of Paragraphs through	;	
a plot plan depicting the Premises;		
a current set of the Rules and Regulations;		
a Work Letter;		
other (specify):		
 Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Le 	account has Drawings for the term of the rent	al and upon all of the terms
covenants and conditions set forth in this Lease. While the approximate square foot	age of the Premises may have been used in t	he marketing of the Premises for
purposes of comparison, the Base Rent stated herein is NOT tied to square footage a different. NOTE: Lessee is advised to verify the actual size prior to executing this L		ctual size be determined to be
2.2 Condition. Lessor shall deliver the Premises to Lessee broom clean and fi	ee of debris on the Commencement Date or	
first occurs (" Start Date "), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkle		
doors, sump pumps, if any, and all other such elements in the Premises, other than t		• , , ,
that the structural elements of the roof, bearing walls and foundation of any building Premises do not contain hazardous levels of any mold or fungi defined as toxic under		
of the Start Date, or if one of such systems or elements should malfunction or fail wi		
respect to such matter, except as otherwise provided in this Lease, promptly after re-		
extent of such non-compliance, malfunction or failure, rectify same at Lessor's exper and (ii) 30 days as to the remaining systems and other elements of the Building. If Le		
period, correction of any such non-compliance, malfunction or failure shall be the ob	oligation of Lessee at Lessee's sole cost and e	xpense. Lessor also warrants, that
unless otherwise specified in writing, Lessor is unaware of (i) any recorded Notices o secured by the Premises; and (iii) any bankruptcy proceeding affecting the Premises.		quent amounts due under any loan
2.3 Compliance . Lessor warrants that to the best of its knowledge the impro		= ::
covenants or restrictions of record, regulations, and ordinances ("Applicable Require thereof, was constructed. Said warranty does not apply to the use to which Lessee v		-
Disabilities Act or any similar laws as a result of Lessee's use (see Paragraph 50), or to be made by Lessee. NOTE: Lessee is responsible for determining whether or not th		
Lessee's intended use, and acknowledges that past uses of the Premises may no lo		
except as otherwise provided, promptly after receipt of written notice from Lessee's the same at Lessor's expense. If Lessee does not give Lessor written notice of a non-		· · · · · · · · · · · · · · · · · · ·
correction of that non-compliance shall be the obligation of Lessee at Lessee's sole c		_
require during the term of this Lease the construction of an addition to or an alteration of the reinforcement or other physical modification of the Unit, Premises and/or Bui		
work as follows:		
(a) Subject to Paragraph 2.3(c) below, if such Capital Expenditures are r compared with uses by tenants in general, Lessee shall be fully responsible for the compared with uses by tenants in general, Lessee shall be fully responsible for the compared with uses by tenants in general, Lessee shall be fully responsible for the compared with uses by tenants in general, Lessee shall be fully responsible for the compared with uses by tenants in general, Lessee shall be fully responsible for the compared with uses by tenants in general, Lessee shall be fully responsible for the compared with uses by tenants in general, Lessee shall be fully responsible for the compared with uses by tenants in general, Lessee shall be fully responsible for the compared with uses by tenants in general, Lessee shall be fully responsible for the compared with uses by tenants in general, Lessee shall be fully responsible for the compared with uses by tenants in general with uses the compared with uses by tenants in general with uses by tenants in general with uses the compared with uses the c		
		Experience to required during
INITIALS	INITIALS	

© 2019 AIR CRE. All Rights Reserved. STN-27.30, Revised 10-22-2020 the last 2 years of this Lease and the cost thereof exceeds 6 months' Base Rent, Lessee may instead terminate this Lease unless Lessor notifies Lessee, in writing, within 10 days after receipt of Lessee's termination notice that Lessor has elected to pay the difference between the actual cost thereof and an amount equal to 6 months' Base Rent. If Lessee elects termination, Lessee shall immediately cease the use of the Premises which requires such Capital Expenditure and deliver to Lessor written notice specifying a termination date at least 90 days thereafter. Such termination date shall, however, in no event be earlier than the last day that Lessee could legally utilize the Premises without commencing such Capital Expenditure.

- (b) If such Capital Expenditure is not the result of the specific and unique use of the Premises by Lessee (such as, governmentally mandated seismic modifications), then Lessor shall pay for such Capital Expenditure and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date that on which the Base Rent is due, an amount equal to 1/144th of the portion of such costs reasonably attributable to the Premises. Lessee shall pay Interest on the balance but may prepay its obligation at any time. If, however, such Capital Expenditure is required during the last 2 years of this Lease or if Lessor reasonably determines that it is not economically feasible to pay its share thereof, Lessor shall have the option to terminate this Lease upon 90 days prior written notice to Lessee unless Lessee notifies Lessor, in writing, within 10 days after receipt of Lessor's termination notice that Lessee will pay for such Capital Expenditure. If Lessor does not elect to terminate, and fails to tender its share of any such Capital Expenditure, Lessee may advance such funds and deduct same, with Interest, from Rent until Lessor's share of such costs have been fully paid. If Lessee is unable to finance Lessor's share, or if the balance of the Rent due and payable for the remainder of this Lease is not sufficient to fully reimburse Lessee on an offset basis, Lessee shall have the right to terminate this Lease upon 30 days written notice to Lessor.
- (c) Notwithstanding the above, the provisions concerning Capital Expenditures are intended to apply only to non-voluntary, unexpected, and new Applicable Requirements. If the Capital Expenditures are instead triggered by Lessee as a result of an actual or proposed change in use, change in intensity of use, or modification to the Premises then, and in that event, Lessee shall either: (i) immediately cease such changed use or intensity of use and/or take such other steps as may be necessary to eliminate the requirement for such Capital Expenditure, or (ii) complete such Capital Expenditure at its own expense. Lessee shall not, however, have any right to terminate this Lease.
- 2.4 **Acknowledgements**. Lessee acknowledges that: (a) it has been given an opportunity to inspect and measure the Premises, (b) it has been advised by Lessor and/or Brokers to satisfy itself with respect to the size and condition of the Premises (including but not limited to the electrical, HVAC and fire sprinkler systems, security, environmental aspects, and compliance with Applicable Requirements and the Americans with Disabilities Act), and their suitability for Lessee's intended use, (c) Lessee has made such investigation as it deems necessary with reference to such matters and assumes all responsibility therefor as the same relate to its occupancy of the Premises, (d) it is not relying on any representation as to the size of the Premises made by Brokers or Lessor, (e) the square footage of the Premises was not material to Lessee's decision to lease the Premises and pay the Rent stated herein, and (f) neither Lessor, Lessor's agents, nor Brokers have made any oral or written representations or warranties with respect to said matters other than as set forth in this Lease. In addition, Lessor acknowledges that: (i) Brokers have made no representations, promises or warranties concerning Lessee's ability to honor the Lease or suitability to occupy the Premises, and (ii) it is Lessor's sole responsibility to investigate the financial capability and/or suitability of all proposed tenants.
- 2.5 Lessee as Prior Owner/Occupant. The warranties made by Lessor in Paragraph 2 shall be of no force or effect if immediately prior to the Start Date Lessee was the owner or occupant of the Premises. In such event, Lessee shall be responsible for any necessary corrective work.

3. Term.

- 3.1 Term. The Commencement Date, Expiration Date and Original Term of this Lease are as specified in Paragraph 1.3
- 3.2 **Early Possession**. Any provision herein granting Lessee Early Possession of the Premises is subject to and conditioned upon the Premises being available for such possession prior to the Commencement Date. Any grant of Early Possession only conveys a non-exclusive right to occupy the Premises. If Lessee totally or partially occupies the Premises prior to the Commencement Date, the obligation to pay Base Rent shall be abated for the period of such Early Possession. All other terms of this Lease (including but not limited to the obligations to pay Real Property Taxes and insurance premiums and to maintain the Premises) shall be in effect during such period. Any such Early Possession shall not affect the Expiration Date.
- 3.3 **Delay In Possession**. Lessor agrees to use commercially reasonable efforts to deliver exclusive possession of the Premises to Lessee by the Commencement Date. If, despite said efforts, Lessor is unable to deliver possession by such date, Lessor shall not be subject to any liability therefor, nor shall such failure affect the validity of this Lease or change the Expiration Date. Lessee shall not, however, be obligated to pay Rent or perform its other obligations until Lessor delivers possession of the Premises and any period of rent abatement that Lessee would otherwise have enjoyed shall run from the date of delivery of possession and continue for a period equal to what Lessee would otherwise have enjoyed under the terms hereof, but minus any days of delay caused by the acts or omissions of Lessee. If possession is not delivered within 60 days after the Commencement Date, as the same may be extended under the terms of any Work Letter executed by Parties, Lessee may, at its option, by notice in writing within 10 days after the end of such 60 day period, cancel this Lease, in which event the Parties shall be discharged from all obligations hereunder. If such written notice is not received by Lessor within said 10 day period, Lessee's right to cancel shall terminate. If possession of the Premises is not delivered within 120 days after the Commencement Date, this Lease shall terminate unless other agreements are reached between Lessor and Lessee, in writing.
- 3.4 Lessee Compliance. Lessor shall not be required to tender possession of the Premises to Lessee until Lessee complies with its obligation to provide evidence of insurance (Paragraph 8.5). Pending delivery of such evidence, Lessee shall be required to perform all of its obligations under this Lease from and after the Start Date, including the payment of Rent, notwithstanding Lessor's election to withhold possession pending receipt of such evidence of insurance. Further, if Lessee is required to perform any other conditions prior to or concurrent with the Start Date, the Start Date shall occur but Lessor may elect to withhold possession until such conditions are satisfied.

Rent.

- 4.1 Rent Defined. All monetary obligations of Lessee to Lessor under the terms of this Lease (except for the Security Deposit) are deemed to be rent ("Rent").
- 4.2 **Payment**. Lessee shall cause payment of Rent to be received by Lessor in lawful money of the United States, without offset or deduction (except as specifically permitted in this Lease), on or before the day on which it is due. All monetary amounts shall be rounded to the nearest whole dollar. In the event that any invoice prepared by Lessor is inaccurate such inaccuracy shall not constitute a waiver and Lessee shall be obligated to pay the amount set forth in this Lease. Rent for any period during the term hereof which is for less than one full calendar month shall be prorated based upon the actual number of days of said month. Payment of Rent shall be made to Lessor at its address stated herein or to such other persons or place as Lessor may from time to time designate in writing. Acceptance of a payment which is less than the amount then due shall not be a waiver of Lessor's rights to the balance of such Rent, regardless of Lessor's endorsement of any check so stating. In the event that any check, draft, or other instrument of payment given by Lessee to Lessor is dishonored for any reason, Lessee agrees to pay to Lessor the sum of \$25 in addition to any Late Charge and Lessor, at its option, may require all future Rent be paid by cashier's check. Payments will be applied first to accrued late charges and attorney's fees, second to accrued interest, then to Base Rent, Insurance and Real Property Taxes, and any remaining amount to any other outstanding charges or costs.
- 4.3 **Association Fees.** In addition to the Base Rent, Lessee shall pay to Lessor each month an amount equal to any owner's association or condominium fees levied or assessed against the Premises. Said monies shall be paid at the same time and in the same manner as the Base Rent.
- Security Deposit. Lessee shall deposit with Lessor upon execution hereof the Security Deposit as security for Lessee's faithful performance of its obligations under this Lease. If Lessee fails to pay Rent, or otherwise Defaults under this Lease, Lessor may use, apply or retain all or any portion of said Security Deposit for the payment of any amount already due Lessor, for Rents which will be due in the future, and/or to reimburse or compensate Lessor for any liability, expense, loss or mage which Lessor may suffer or incur by reason thereof. If Lessor uses or applies all or any portion of the Security Deposit, Lessee shall within 10 days after written request therefor deposit monies with Lessor sufficient to restore said Security Deposit to the full amount required by this Lease. If the Base Rent increases during the term of this Lease, Lessee shall, upon written request from Lessor, deposit additional monies with Lessor so that the total amount of the Security Deposit shall at all times bear the same proportion to the increased Base Rent as the initial Security Deposit bore to the initial Base Rent. Should the Agreed Use be amended to accommodate a material change in the business of Lessee or to accommodate a sublessee or assignee, Lessor shall have the right to increase the Security Deposit to the extent necessary, in Lessor's reasonable judgment, to account for any increased wear and tear that the Premises may suffer as a result thereof. If a change in control of Lessee occurs during this Lease and following such change the financial condition of Lessee is, in Lessor's reasonable judgment, significantly reduced, Lessee shall deposit such additional monies with Lessor as shall be sufficient to cause the Security Deposit to be at a commercially reasonable level based on such change in financial condition. Lessor shall not be required to keep the Security Deposit separate from its general accounts. Within 90 days after the expiration or termination of this Lease, Lessor shall return that portion of the Security Deposit not used or applied by Lessor. Lessor shall upon written request provide Lessee with an accounting showing how that portion of the Security Deposit that was not returned was applied. No part of the Security Deposit shall be considered to be held in trust, to bear interest or to be prepayment for any monies to be paid by Lessee under this Lease. THE SECURITY DEPOSIT SHALL NOT BE USED BY LESSEE IN LIEU OF PAYMENT OF THE LAST MONTH'S RENT.

6. Use.

6.1 **Use.** Lessee shall use and occupy the Premises only for the Agreed Use, or any other legal use which is reasonably comparable thereto, and for no other purpose. Lessee shall not use or permit the use of the Premises in a manner that is unlawful, creates damage, waste or a nuisance, or that disturbs occupants of or causes damage to neighboring premises or properties. Other than guide, signal and seeing eye dogs, Lessee shall not keep or allow in the Premises any pets, animals,

birds, fish, or reptiles. Lessor shall not unreasonably withhold or delay its consent to any written request for a modification of the Agreed Use, so long as the same will not impair the structural integrity of the improvements on the Premises or the mechanical or electrical systems therein, and/or is not significantly more burdensome to the Premises. If Lessor elects to withhold consent, Lessor shall within 7 days after such request give written notification of same, which notice shall include an explanation of Lessor's objections to the change in the Agreed Use.

6.2 Hazardous Substances

- (a) Reportable Uses Require Consent. The term "Hazardous Substance" as used in this Lease shall mean any product, substance, or waste whose presence, use, manufacture, disposal, transportation, or release, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentially injurious to the public health, safety or welfare, the environment or the Premises, (ii) regulated or monitored by any governmental authority, or (iii) a basis for potential liability of Lessor to any governmental agency or third party under any applicable statute or common law theory. Hazardous Substances shall include, but not be limited to, hydrocarbons, petroleum, gasoline, and/or crude oil or any products, by-products or fractions thereof. Lessee shall not engage in any activity in or on the Premises which constitutes a Reportable Use of Hazardous Substances without the express prior written consent of Lessor and timely compliance (at Lessee's expense) with all Applicable Requirements. "Reportable Use" shall mean (i) the installation or use of any above or below ground storage tank, (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and/or (iii) the presence at the Premises of a Hazardous Substance with respect to which any Applicable Requirements requires that a notice be given to persons entering or occupying the Premises or neighboring properties. Notwithstanding the foregoing, Lessee may use any ordinary and customary materials reasonably required to be used in the normal course of the Agreed Use, ordinary office supplies (copier toner, liquid paper, glue, etc.) and common household cleaning materials, so long as such use is in compliance with all Applicable Requirements, is not a Reportable Use, and does not expose the Premises or neighboring property to any meaningful risk of contamination or damage or expose Lessor to any liability therefor. In addition, Lessor may condition its consent to any Reportable Use upon receiving such additional assurances as Lessor reasonably deems necessary to protect itself, the public, the Premises and/or the environment against damage, contamination, injury and/or liability, including, but not limited to, the installation (and removal on or before Lease expiration or termination) of protective modifications (such as concrete encasements) and/or increasing the Security Deposit.
- (b) **Duty to Inform Lessor**. If Lessee knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises, other than as previously consented to by Lessor, Lessee shall immediately give written notice of such fact to Lessor, and provide Lessor with a copy of any report, notice, claim or other documentation which it has concerning the presence of such Hazardous Substance.
- (c) Lessee Remediation. Lessee shall not cause or permit any Hazardous Substance to be spilled or released in, on, under, or about the Premises (including through the plumbing or sanitary sewer system) and shall promptly, at Lessee's expense, comply with all Applicable Requirements and take all investigatory and/or remedial action reasonably recommended, whether or not formally ordered or required, for the cleanup of any contamination of, and for the maintenance, security and/or monitoring of the Premises or neighboring properties, that was caused or materially contributed to by Lessee, or pertaining to or involving any Hazardous Substance brought onto the Premises during the term of this Lease, by or for Lessee, or any third party.
- (d) Lessee Indemnification. Lessee shall indemnify, defend and hold Lessor, its agents, employees, lenders and ground lessor, if any, harmless from and against any and all loss of rents and/or damages, liabilities, judgments, claims, expenses, penalties, and attorneys' and consultants' fees arising out of or involving any Hazardous Substance brought onto the Premises by or for Lessee, or any third party (provided, however, that Lessee shall have no liability under this Lease with respect to underground migration of any Hazardous Substance under the Premises from adjacent properties not caused or contributed to by Lessee). Lessee's obligations shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Lessee, and the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease. No termination, cancellation or release agreement entered into by Lessor and Lessee shall release Lessee from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Lessor in writing at the time of such agreement.
- (e) Lessor Indemnification. Except as otherwise provided in paragraph 8.7, Lessor and its successors and assigns shall indemnify, defend, reimburse and hold Lessee, its employees and lenders, harmless from and against any and all environmental damages, including the cost of remediation, which result from Hazardous Substances which existed on the Premises prior to Lessee's occupancy or which are caused by the gross negligence or willful misconduct of Lessor, its agents or employees. Lessor's obligations, as and when required by the Applicable Requirements, shall include, but not be limited to, the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease.
- (f) Investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by governmental entities having jurisdiction with respect to the existence of Hazardous Substances on the Premises prior to Lessee's occupancy, unless such remediation measure is required as a result of Lessee's use (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises, in which event Lessee shall be responsible for such payment. Lessee shall cooperate fully in any such activities at the request of Lessor, including allowing Lessor and Lessor's agents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remedial responsibilities.
- (g) Lessor Termination Option. If a Hazardous Substance Condition (see Paragraph 9.1(e)) occurs during the term of this Lease, unless Lessee is legally responsible therefor (in which case Lessee shall make the investigation and remediation thereof required by the Applicable Requirements and this Lease shall continue in full force and effect, but subject to Lessor's rights under Paragraph 6.2(d) and Paragraph 13), Lessor may, at Lessor's option, either (i) investigate and remediate such Hazardous Substance Condition, if required, as soon as reasonably possible at Lessor's expense, in which event this Lease shall continue in full force and effect, or (ii) if the estimated cost to remediate such condition exceeds 12 times the then monthly Base Rent or \$100,000, whichever is greater, give written notice to Lessee, within 30 days after receipt by Lessor of knowledge of the occurrence of such Hazardous Substance Condition, of Lessor's desire to terminate this Lease as of the date 60 days following the date of such notice. In the event Lessor elects to give a termination notice, Lessee may, within 10 days thereafter, give written notice to Lessor of Lessee's commitment to pay the amount by which the cost of the remediation of such Hazardous Substance Condition exceeds an amount equal to 12 times the then monthly Base Rent or \$100,000, whichever is greater. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days following such commitment. In such event, this Lease shall continue in full force and effect, and Lessor shall proceed to make such remediation as soon as reasonably possible after the required funds are available. If Lessee does not give such notice and provide the required funds or assurance thereof within the time provided, this Lease shall terminate as of the date specified in Lessor's notice of termination.
- 6.3 Lessee's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lessee's sole expense, fully, diligently and in a timely manner, materially comply with all Applicable Requirements, the requirements of any applicable fire insurance underwriter or rating bureau, and the recommendations of Lessor's engineers and/or consultants which relate in any manner to the Premises, without regard to whether said Applicable Requirements are now in effect or become effective after the Start Date. Lessee shall, within 10 days after receipt of Lessor's written request, provide Lessor with copies of all permits and other documents, and other information evidencing Lessee's compliance with any Applicable Requirements specified by Lessor, and shall immediately upon receipt, notify Lessor in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving the failure of Lessee or the Premises to comply with any Applicable Requirements. Likewise, Lessee shall immediately give written notice to Lessor of: (i) any water damage to the Premises and any suspected seepage, pooling, dampness or other condition conducive to the production of mold; or (ii) any mustiness or other odors that might indicate the presence of mold in the Premises. In addition, Lessee shall provide copies of all relevant material safety data sheets (MSDS) to Lessor within 10 days of the receipt of a written request therefor. In addition, Lessee shall provide Lessor with copies of its business license, certificate of occupancy and/or any similar document within 10 days of the receipt of a written request therefor.
- 6.4 Inspection; Compliance. Lessor and Lessor's "Lender" (as defined in Paragraph 30) and consultants authorized by Lessor shall have the right to enter into Premises at any time in the case of an emergency, and otherwise at reasonable times after reasonable notice, for the purpose of inspecting and/or testing the condition of the Premises and/or for verifying compliance by Lessee with this Lease. The cost of any such inspections shall be paid by Lessor, unless a violation of Applicable Requirements, or a Hazardous Substance Condition (see paragraph 9.1(e)) is found to exist or be imminent, or the inspection is requested or ordered by a governmental authority. In such case, Lessee shall upon request reimburse Lessor for the cost of such inspection, so long as such inspection is reasonably related to the violation or contamination. In addition, Lessee shall provide copies of all relevant material safety data sheets (MSDS) to Lessor within 10 days of the receipt of a written request therefor. Lessee acknowledges that any failure on its part to allow such inspections or testing will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to allow such inspections and/or testing in a timely fashion the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for the remainder to the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to allow such inspection and/or testing. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to such failure nor prevent the exercise of any of the other rights and remedies granted hereunder.

7. Maintenance; Repairs; Utility Installations; Trade Fixtures and Alterations.

7.1 Lessee's Obligations.

(a) In General. Subject to the provisions of Paragraph 2.2 (Condition), 2.3 (Compliance), 6.3 (Lessee's Compliance with Applicable Requirements), 7.2 (Lessor's Obligations), 9 (Damage or Destruction), and 14 (Condemnation), Lessee shall, at Lessee's sole expense, keep the Premises, Utility Installations (intended for Lessee's exclusive use, no matter where located), and Alterations in good order, condition and repair (whether or not the portion of the Premises requiring repairs, or the means of repairing the same, are reasonably or readily accessible to Lessee, and whether or not the need for such repairs occurs as a result of Lessee's use, any

INITIALS

prior use, the elements or the age of such portion of the Premises), including, but not limited to, all equipment or facilities, such as plumbing, HVAC equipment, electrical, lighting facilities, boilers, pressure vessels, fire protection system, fixtures, walls (interior and exterior), foundations, ceilings, roofs, roof drainage systems, floors, windows, doors, plate glass, skylights, landscaping, driveways, parking lots, fences, retaining walls, signs, sidewalks and parkways located in, on, or adjacent to the Premises. Lessee, in keeping the Premises in good order, condition and repair, shall exercise and perform good maintenance practices, specifically including the procurement and maintenance of the service contracts required by Paragraph 7.1(b) below. Lessee's obligations shall include restorations, replacements or renewals when necessary to keep the Premises and all improvements thereon or a part thereof in good order, condition and state of repair. Lessee shall, during the term of this Lease, keep the exterior appearance of the Building in a first-class condition (including, e.g. graffiti removal) consistent with the exterior appearance of other similar facilities of comparable age and size in the vicinity, including, when necessary, the exterior repainting of the Building.

- (b) Service Contracts. Lessee shall, at Lessee's sole expense, procure and maintain contracts, with copies to Lessor, in customary form and substance for, and with contractors specializing and experienced in the maintenance of the following equipment and improvements, if any, if and when installed on the Premises: (i) HVAC equipment, (ii) boiler, and pressure vessels, (iii) fire extinguishing systems, including fire alarm and/or smoke detection, (iv) landscaping and irrigation systems, (v) roof covering and drains, and (vi) clarifiers. However, Lessor reserves the right, upon notice to Lessee, to procure and maintain any or all of such service contracts, and Lessee shall reimburse Lessor, upon demand, for the cost thereof.
- (c) Failure to Perform. If Lessee fails to perform Lessee's obligations under this Paragraph 7.1, Lessor may enter upon the Premises after 10 days' prior written notice to Lessee (except in the case of an emergency, in which case no notice shall be required), perform such obligations on Lessee's behalf, and put the Premises in good order, condition and repair, and Lessee shall promptly pay to Lessor a sum equal to 115% of the cost thereof.
- (d) **Replacement**. Subject to Lessee's indemnification of Lessor as set forth in Paragraph 8.7 below, and without relieving Lessee of liability resulting from Lessee's failure to exercise and perform good maintenance practices, if an item described in Paragraph 7.1(b) cannot be repaired other than at a cost which is in excess of 50% of the cost of replacing such item, then such item shall be replaced by Lessor, and the cost thereof shall be prorated between the Parties and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date on which Base Rent is due, an amount equal to the product of multiplying the cost of such replacement by a fraction, the numerator of which is one, and the denominator of which is 144 (i.e. 1/144th of the cost per month). Lessee shall pay Interest on the unamortized balance but may prepay its obligation at any time.
- 7.2 **Lessor's Obligations**. Subject to the provisions of Paragraphs 2.2 (Condition), 2.3 (Compliance), 9 (Damage or Destruction) and 14 (Condemnation), it is intended by the Parties hereto that Lessor have no obligation, in any manner whatsoever, to repair and maintain the Premises, or the equipment therein, all of which obligations are intended to be that of the Lessee. It is the intention of the Parties that the terms of this Lease govern the respective obligations of the Parties as to maintenance and repair of the Premises.

7.3 Utility Installations; Trade Fixtures; Alterations.

- (a) **Definitions**. The term "**Utility Installations**" refers to all floor and window coverings, air and/or vacuum lines, power panels, electrical distribution, security and fire protection systems, communication cabling, lighting fixtures, HVAC equipment, plumbing, and fencing in or on the Premises. The term "**Trade Fixtures**" shall mean Lessee's machinery and equipment that can be removed without doing material damage to the Premises. The term "Alterations" shall mean any modification of the improvements, other than Utility Installations or Trade Fixtures, whether by addition or deletion. "**Lessee Owned Alterations and/or Utility Installations**" are defined as Alterations and/or Utility Installations made by Lessee that are not yet owned by Lessor pursuant to Paragraph 7.4(a).
- (b) Consent. Lessee shall not make any Alterations or Utility Installations to the Premises without Lessor's prior written consent. Lessee may, however, make non-structural Alterations or Utility Installations to the interior of the Premises (excluding the roof) without such consent but upon notice to Lessor, as long as they are not visible from the outside, do not involve puncturing, relocating or removing the roof or any existing walls, will not affect the electrical, plumbing, HVAC, and/or life safety systems, do not trigger the requirement for additional modifications and/or improvements to the Premises resulting from Applicable Requirements, such as compliance with Title 24, and the cumulative cost thereof during this Lease as extended does not exceed a sum equal to 3 month's Base Rent in the aggregate or a sum equal to one month's Base Rent in any one year. Notwithstanding the foregoing, Lessee shall not make or permit any roof penetrations and/or install anything on the roof without the prior written approval of Lessor. Lessor may, as a precondition to granting such approval, require Lessee to utilize a contractor chosen and/or approved by Lessor. Any Alterations or Utility Installations that Lessee shall desire to make and which require the consent of the Lessor shall be presented to Lessor in written form with detailed plans. Consent shall be deemed conditioned upon Lessee's: (i) acquiring all applicable governmental permits, (ii) furnishing Lessor with copies of both the permits and the plans and specifications prior to commencement of the work, and (iii) compliance with all conditions of said permits and other Applicable Requirements in a prompt and expeditious manner. Any Alterations or Utility Installations shall be performed in a workmanlike manner with good and sufficient materials. Lessee shall promptly upon completion furnish Lessor with as-built plans and specifications. For work which costs an amount in excess of one month's Base Rent, Lessor may condition its consent upon Lessee's posting an
- (c) Liens; Bonds. Lessee shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lessee at or for use on the Premises, which claims are or may be secured by any mechanic's or materialmen's lien against the Premises or any interest therein. Lessee shall give Lessor not less than 10 days notice prior to the commencement of any work in, on or about the Premises, and Lessor shall have the right to post notices of non-responsibility. If Lessee shall contest the validity of any such lien, claim or demand, then Lessee shall, at its sole expense defend and protect itself, Lessor and the Premises against the same and shall pay and satisfy any such adverse judgment that may be rendered thereon before the enforcement thereof. If Lessor shall require, Lessee shall furnish a surety bond in an amount equal to 150% of the amount of such contested lien, claim or demand, indemnifying Lessor against liability for the same. If Lessor elects to participate in any such action, Lessee shall pay Lessor's attorneys' fees and costs.

7.4 Ownership; Removal; Surrender; and Restoration.

- (a) **Ownership.** Subject to Lessor's right to require removal or elect ownership as hereinafter provided, all Alterations and Utility Installations made by Lessee shall be the property of Lessee, but considered a part of the Premises. Lessor may, at any time, elect in writing to be the owner of all or any specified part of the Lessee Owned Alterations and Utility Installations. Unless otherwise instructed per paragraph 7.4(b) hereof, all Lessee Owned Alterations and Utility Installations shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.
- (b) Removal. By delivery to Lessee of written notice from Lessor not earlier than 90 and not later than 30 days prior to the end of the term of this Lease, Lessor may require that any or all Lessee Owned Alterations or Utility Installations be removed by the expiration or termination of this Lease. Lessor may require the removal at any time of all or any part of any Lessee Owned Alterations or Utility Installations made without the required consent.
- (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements, parts and surfaces thereof broom clean and free of debris, and in good operating order, condition and state of repair, ordinary wear and tear excepted. "Ordinary wear and tear" shall not include any damage or deterioration that would have been prevented by good maintenance practice. Notwithstanding the foregoing and the provisions of Paragraph 7.1(a), if the Lessee occupies the Premises for 12 months or less, then Lessee shall surrender the Premises in the same condition as delivered to Lessee on the Start Date with NO allowance for ordinary wear and tear. Lessee shall repair any damage occasioned by the installation, maintenance or removal of Trade Fixtures, Lessee owned Alterations and/or Utility Installations, furnishings, and equipment as well as the removal of any storage tank installed by or for Lessee. Lessee shall also completely remove from the Premises any and all Hazardous Substances brought onto the Premises by or for Lessee, or any third party (except Hazardous Substances which were deposited via underground migration from areas outside of the Premises) to the level specified in Applicable Requirements. Trade Fixtures shall remain the property of Lessee and shall be removed by Lessee. Any personal property of Lessee not removed on or before the Expiration Date or any earlier termination date shall be deemed to have been abandoned by Lessee and may be disposed of or retained by Lessor as Lessor may desire. The failure by Lessee to timely vacate the Premises pursuant to this Paragraph 7.4(c) without the express written consent of Lessor shall constitute a holdover under the provisions of Paragraph 26 below.

8. Insurance; Indemnity.

8.1 Payment For Insurance. Lessee shall pay for all insurance required under Paragraph 8 except to the extent of the cost attributable to liability insurance carried by Lessor under Paragraph 8.2(b) in excess of \$2,000,000 per occurrence. Premiums for policy periods commencing prior to or extending beyond the Lease term shall be prorated to correspond to the Lease term. Payment shall be made by Lessee to Lessor within 10 days following receipt of an invoice.

8.2 Liability Insurance

- (a) Carried by Lessee. Lessee shall obtain and keep in force a Commercial General Liability policy of insurance protecting Lessee and Lessor as an additional insured against claims for bodily injury, personal injury and property damage based upon or arising out of the ownership, use, occupancy or maintenance of the Premises and all areas appurtenant thereto. Such insurance shall be on an occurrence basis providing single limit coverage in an amount not less than \$1,000,000 per occurrence with an annual aggregate of not less than \$2,000,000. Lessee shall add Lessor as an additional insured by means of an endorsement at least as broad as the Insurance Service Organization's "Additional Insured-Managers or Lessors of Premises" Endorsement. The policy shall not contain any intra-insured exclusions as between insured persons or organizations, but shall include coverage for liability assumed under this Lease as an "insured contract" for the performance of Lessee's indemnity obligations under this Lease. The limits of said insurance shall not, however, limit the liability of Lessee nor relieve Lessee of any obligation hereunder. Lessee shall provide an endorsement on its liability policy(ies) which provides that its insurance shall be primary to and not contributory with any similar insurance carried by Lessor, whose insurance shall be considered excess insurance only.
- (b) Carried by Lessor. Lessor shall maintain liability insurance as described in Paragraph 8.2(a), in addition to, and not in lieu of, the insurance required to be maintained by Lessee. Lessee shall not be named as an additional insured therein.

INITIALS

INITIALS

8.3 Property Insurance - Building, Improvements and Rental Value.

- (a) **Building and Improvements**. The Insuring Party shall obtain and keep in force a policy or policies in the name of Lessor, with loss payable to Lessor, any ground-lessor, and to any Lender insuring loss or damage to the Premises. The amount of such insurance shall be equal to the full insurable replacement cost of the Premises, as the same shall exist from time to time, or the amount required by any Lender, but in no event more than the commercially reasonable and available insurable value thereof. Lessee Owned Alterations and Utility Installations, Trade Fixtures, and Lessee's personal property shall be insured by Lessee not by Lessor. If the coverage is available and commercially appropriate, such policy or policies shall insure against all risks of direct physical loss or damage (except the perils of flood and/or earthquake unless required by a Lender), including coverage for debris removal and the enforcement of any Applicable Requirements requiring the upgrading, demolition, reconstruction or replacement of any portion of the Premises as the result of a covered loss. Said policy or policies shall also contain an agreed valuation provision in lieu of any coinsurance clause, waiver of subrogation, and inflation guard protection causing an increase in the annual property insurance coverage amount by a factor of not less than the adjusted U.S. Department of Labor Consumer Price Index for All Urban Consumers for the city nearest to where the Premises are located. If such insurance coverage has a deductible clause, the deductible amount shall not exceed \$5,000 per occurrence, and Lessee shall be liable for such deductible amount in the event of an Insured Loss.
- (b) Rental Value. The Insuring Party shall obtain and keep in force a policy or policies in the name of Lessor with loss payable to Lessor and any Lender, insuring the loss of the full Rent for one year with an extended period of indemnity for an additional 180 days ("Rental Value insurance"). Said insurance shall contain an agreed valuation provision in lieu of any coinsurance clause, and the amount of coverage shall be adjusted annually to reflect the projected Rent otherwise payable by Lessee, for the next 12 month period. Lessee shall be liable for any deductible amount in the event of such loss.
- (c) Adjacent Premises. If the Premises are part of a larger building, or of a group of buildings owned by Lessor which are adjacent to the Premises, the Lessee shall pay for any increase in the premiums for the property insurance of such building or buildings if said increase is caused by Lessee's acts, omissions, use or occupancy of the Premises.

8.4 Lessee's Property; Business Interruption Insurance; Worker's Compensation Insurance.

- (a) **Property Damage**. Lessee shall obtain and maintain insurance coverage on all of Lessee's personal property, Trade Fixtures, and Lessee Owned Alterations and Utility Installations. Such insurance shall be full replacement cost coverage with a deductible of not to exceed \$1,000 per occurrence. The proceeds from any such insurance shall be used by Lessee for the replacement of personal property, Trade Fixtures and Lessee Owned Alterations and Utility Installations.
- (b) **Business Interruption**. Lessee shall obtain and maintain loss of income and extra expense insurance in amounts as will reimburse Lessee for direct or indirect loss of earnings attributable to all perils commonly insured against by prudent lessees in the business of Lessee or attributable to prevention of access to the Premises as a result of such perils.
- (c) Worker's Compensation Insurance. Lessee shall obtain and maintain Worker's Compensation Insurance in such amount as may be required by Applicable Requirements. Such policy shall include a 'Waiver of Subrogation' endorsement. Lessee shall provide Lessor with a copy of such endorsement along with the certificate of insurance or copy of the policy required by paragraph 8.5.
- (d) **No Representation of Adequate Coverage**. Lessor makes no representation that the limits or forms of coverage of insurance specified herein are adequate to cover Lessee's property, business operations or obligations under this Lease.
- 8.5 Insurance Policies. Insurance required herein shall be by companies maintaining during the policy term a "General Policyholders Rating" of at least A-, VII, as set forth in the most current issue of "Best's Insurance Guide", or such other rating as may be required by a Lender. Lessee shall not do or permit to be done anything which invalidates the required insurance policies. Lessee shall, prior to the Start Date, deliver to Lessor certified copies of policies of such insurance or certificates with copies of the required endorsements evidencing the existence and amounts of the required insurance. No such policy shall be cancelable or subject to modification except after 30 days prior written notice to Lessor. Lessee shall, at least 10 days prior to the expiration of such policies, furnish Lessor with evidence of renewals or "insurance binders" evidencing renewal thereof, or Lessor may increase his liability insurance coverage and charge the cost thereof to Lessee, which amount shall be payable by Lessee to Lessor upon demand. Such policies shall be for a term of at least one year, or the length of the remaining term of this Lease, whichever is less. If either Party shall fail to procure and maintain the insurance required to be carried by it, the other Party may, but shall not be required to, procure and maintain the same.
- 8.6 **Waiver of Subrogation**. Without affecting any other rights or remedies, Lessee and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is not limited by the amount of insurance carried or required, or by any deductibles applicable hereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Lessee, as the case may be, so long as the insurance is not invalidated thereby.
- 8.7 Indemnity. Except for Lessor's gross negligence or willful misconduct, Lessee shall indemnify, protect, defend and hold harmless the Premises, Lessor and its agents, Lessor's master or ground lessor, partners and Lenders, from and against any and all claims, loss of rents and/or damages, liens, judgments, penalties, attorneys' and consultants' fees, expenses and/or liabilities arising out of, involving, or in connection with, a Breach of the Lease by Lessee and/or the use and/or occupancy of the Premises and/or Project by Lessee and/or by Lessee's employees, contractors or invitees. If any action or proceeding is brought against Lessor by reason of any of the foregoing matters, Lessee shall upon notice defend the same at Lessee's expense by counsel reasonably satisfactory to Lessor and Lessor shall cooperate with Lessee in such defense. Lessor need not have first paid any such claim in order to be defended or indemnified.
- 8.8 **Exemption of Lessor and its Agents from Liability**. Notwithstanding the negligence or breach of this Lease by Lessor or its agents, neither Lessor nor its agents shall be liable under any circumstances for: (i) injury or damage to the person or goods, wares, merchandise or other property of Lessee, Lessee's employees, contractors, invitees, customers, or any other person in or about the Premises, whether such damage or injury is caused by or results from fire, steam, electricity, gas, water or rain, indoor air quality, the presence of mold or from the breakage, leakage, obstruction or other defects of pipes, fire sprinklers, wires, appliances, plumbing, HVAC or lighting fixtures, or from any other cause, whether the said injury or damage results from conditions arising upon the Premises or upon other portions of the building of which the Premises are a part, or from other sources or places, (ii) any damages arising from any act or neglect of any other tenant of Lessor or from the failure of Lessor or its agents to enforce the provisions of any other lease in the Project, or (iii) injury to Lessee's business or for any loss of income or profit therefrom. Instead, it is intended that Lessee's sole recourse in the event of such damages or injury be to file a claim on the insurance policy(ies) that Lessee is required to maintain pursuant to the provisions of paragraph 8.
- 8.9 **Failure to Provide Insurance**. Lessee acknowledges that any failure on its part to obtain or maintain the insurance required herein will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, for any month or portion thereof that Lessee does not maintain the required insurance and/or does not provide Lessor with the required binders or certificates evidencing the existence of the required insurance, the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater. The parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to maintain the required insurance. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to maintain such insurance, prevent the exercise of any of the other rights and remedies granted hereunder, nor relieve Lessee of its obligation to maintain the insurance specified in this Lease.

9. Damage or Destruction.

9.1 Definitions.

- (a) "Premises Partial Damage" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations, which can reasonably be repaired in 6 months or less from the date of the damage or destruction. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (b) "Premises Total Destruction" shall mean damage or destruction to the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which cannot reasonably be repaired in 6 months or less from the date of the damage or destruction. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (c) "Insured Loss" shall mean damage or destruction to improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which was caused by an event required to be covered by the insurance described in Paragraph 8.3(a), irrespective of any deductible amounts or
- coverage limits involved.

 (d) "Replacement Cost" shall mean the cost to repair or rebuild the improvements owned by Lessor at the time of the occurrence to their condition existing immediately prior thereto, including demolition, debris removal and upgrading required by the operation of Applicable Requirements, and without deduction for depreciation.
- (e) "Hazardous Substance Condition" shall mean the occurrence or discovery of a condition involving the presence of, or a contamination by, a Hazardous Substance, in, on, or under the Premises which requires restoration.
- 9.2 Partial Damage Insured Loss. If a Premises Partial Damage that is an Insured Loss occurs, then Lessor shall, at Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonably possible and this Lease shall continue in full force and effect; provided, however, that Lessee shall, at Lessor's election, make the repair of any damage or destruction the total cost to repair of which is \$10,000 or less, and, in such event, Lessor shall make any applicable insurance proceeds available to Lessee on a reasonable basis for that purpose. Notwithstanding the foregoing, if the required insurance was not in force or the insurance proceeds are not sufficient to effect such repair, the Insuring Party shall promptly contribute the shortage in proceeds

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

(except as to the deductible which is Lessee's responsibility) as and when required to complete said repairs. In the event, however, such shortage was due to the fact that, by reason of the unique nature of the improvements, full replacement cost insurance coverage was not commercially reasonable and available, Lessor shall have no obligation to pay for the shortage in insurance proceeds or to fully restore the unique aspects of the Premises unless Lessee provides Lessor with the funds to cover same, or adequate assurance thereof, within 10 days following receipt of written notice of such shortage and request therefor. If Lessor receives said funds or adequate assurance thereof within said 10 day period, the party responsible for making the repairs shall complete them as soon as reasonably possible and this Lease shall remain in full force and effect. If such funds or assurance are not received, Lessor may nevertheless elect by written notice to Lessee within 10 days thereafter to: (i) make such restoration and repair as is commercially reasonable with Lessor paying any shortage in proceeds, in which case this Lease shall remain in full force and effect, or (ii) have this Lease terminate 30 days thereafter. Lessee shall not be entitled to reimbursement of any funds contributed by Lessee to repair any such damage or destruction. Premises Partial Damage due to flood or earthquake shall be subject to Paragraph 9.3, notwithstanding that there may be some insurance coverage, but the net proceeds of any such insurance shall be made available for the repairs if made by either Party.

- 9.3 Partial Damage Uninsured Loss. If a Premises Partial Damage that is not an Insured Loss occurs, unless caused by a negligent or willful act of Lessee (in which event Lessee shall make the repairs at Lessee's expense), Lessor may either: (i) repair such damage as soon as reasonably possible at Lessor's expense, in which event this Lease shall continue in full force and effect, or (ii) terminate this Lease by giving written notice to Lessee within 30 days after receipt by Lessor of knowledge of the occurrence of such damage. Such termination shall be effective 60 days following the date of such notice. In the event Lessor elects to terminate this Lease, Lessee shall have the right within 10 days after receipt of the termination notice to give written notice to Lessor of Lessee's commitment to pay for the repair of such damage without reimbursement from Lessor. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days after making such commitment. In such event this Lease shall continue in full force and effect, and Lessor shall proceed to make such repairs as soon as reasonably possible after the required funds are available. If Lessee does not make the required commitment, this Lease shall terminate as of the date specified in the termination notice.
- 9.4 **Total Destruction**. Notwithstanding any other provision hereof, if a Premises Total Destruction occurs, this Lease shall terminate 60 days following such Destruction. If the damage or destruction was caused by the gross negligence or willful misconduct of Lessee, Lessor shall have the right to recover Lessor's damages from Lessee, except as provided in Paragraph 8.6.
- 9.5 **Damage Near End of Term.** If at any time during the last 6 months of this Lease there is damage for which the cost to repair exceeds one month's Base Rent, whether or not an Insured Loss, Lessor may terminate this Lease effective 60 days following the date of occurrence of such damage by giving a written termination notice to Lessee within 30 days after the date of occurrence of such damage. Notwithstanding the foregoing, if Lessee at that time has an exercisable option to extend this Lease or to purchase the Premises, then Lessee may preserve this Lease by, (a) exercising such option and (b) providing Lessor with any shortage in insurance proceeds (or adequate assurance thereof) needed to make the repairs on or before the earlier of (i) the date which is 10 days after Lessee's receipt of Lessor's written notice purporting to terminate this Lease, or (ii) the day prior to the date upon which such option expires. If Lessee duly exercises such option during such period and provides Lessor with funds (or adequate assurance thereof) to cover any shortage in insurance proceeds, Lessor shall, at Lessor's commercially reasonable expense, repair such damage as soon as reasonably possible and this Lease shall continue in full force and effect. If Lessee fails to exercise such option and provide such funds or assurance during such period, then this Lease shall terminate on the date specified in the termination notice and Lessee's option shall be extinguished.

9.6 Abatement of Rent: Lessee's Remedies.

- (a) Abatement. In the event of Premises Partial Damage or Premises Total Destruction or a Hazardous Substance Condition for which Lessee is not responsible under this Lease, the Rent payable by Lessee for the period required for the repair, remediation or restoration of such damage shall be abated in proportion to the degree to which Lessee's use of the Premises is impaired, but not to exceed the proceeds received from the Rental Value insurance. All other obligations of Lessee hereunder shall be performed by Lessee, and Lessor shall have no liability for any such damage, destruction, remediation, repair or restoration except as provided herein.
- (b) **Remedies**. If Lessor is obligated to repair or restore the Premises and does not commence, in a substantial and meaningful way, such repair or restoration within 90 days after such obligation shall accrue, Lessee may, at any time prior to the commencement of such repair or restoration, give written notice to Lessor and to any Lenders of which Lessee has actual notice, of Lessee's election to terminate this Lease on a date not less than 60 days following the giving of such notice. If Lessee gives such notice and such repair or restoration is not commenced within 30 days thereafter, this Lease shall terminate as of the date specified in said notice. If the repair or restoration is commenced within such 30 days, this Lease shall continue in full force and effect. "Commence" shall mean either the unconditional authorization of the preparation of the required plans, or the beginning of the actual work on the Premises, whichever first occurs.
- 9.7 **Termination; Advance Payments**. Upon termination of this Lease pursuant to Paragraph 6.2(g) or Paragraph 9, an equitable adjustment shall be made concerning advance Base Rent and any other advance payments made by Lessee to Lessor. Lessor shall, in addition, return to Lessee so much of Lessee's Security Deposit as has not been, or is not then required to be, used by Lessor.

10. Real Property Taxes.

- 10.1 **Definition**. As used herein, the term "**Real Property Taxes**" shall include any form of assessment; real estate, general, special, ordinary or extraordinary, or rental levy or tax (other than inheritance, personal income or estate taxes); improvement bond; and/or license fee imposed upon or levied against any legal or equitable interest of Lessor in the Premises or the Project, Lessor's right to other income therefrom, and/or Lessor's business of leasing, by any authority having the direct or indirect power to tax and where the funds are generated with reference to the Building address. Real Property Taxes shall also include any tax, fee, levy, assessment or charge, or any increase therein: (i) imposed by reason of events occurring during the term of this Lease, including but not limited to, a change in the ownership of the Premises, and (ii) levied or assessed on machinery or equipment provided by Lessor to Lessee pursuant to this Lease.
- 10.2 **Payment of Taxes**. In addition to Base Rent, Lessee shall pay to Lessor an amount equal to the Real Property Tax installment due at least 20 days prior to the applicable delinquency date. If any such installment shall cover any period of time prior to or after the expiration or termination of this Lease, Lessee's share of such installment shall be prorated. In the event Lessee incurs a late charge on any Rent payment, Lessor may estimate the current Real Property Taxes, and require that such taxes be paid in advance to Lessor by Lessee monthly in advance with the payment of the Base Rent. Such monthly payments shall be an amount equal to the amount of the estimated installment of taxes divided by the number of months remaining before the month in which said installment becomes delinquent. When the actual amount of the applicable tax bill is known, the amount of such equal monthly advance payments shall be adjusted as required to provide the funds needed to pay the applicable taxes. If the amount collected by Lessor is insufficient to pay such Real Property Taxes when due, Lessee shall pay Lessor, upon demand, such additional sum as is necessary. Advance payments may be intermingled with other moneys of Lessor and shall not bear interest. In the event of a Breach by Lessee in the performance of its obligations under this Lease, then any such advance payments may be treated by Lessor as an additional Security Deposit.
- 10.3 **Joint Assessment**. If the Premises are not separately assessed, Lessee's liability shall be an equitable proportion of the Real Property Taxes for all of the land and improvements included within the tax parcel assessed, such proportion to be conclusively determined by Lessor from the respective valuations assigned in the assessor's work sheets or such other information as may be reasonably available.
- 10.4 **Personal Property Taxes**. Lessee shall pay, prior to delinquency, all taxes assessed against and levied upon Lessee Owned Alterations, Utility Installations, Trade Fixtures, furnishings, equipment and all personal property of Lessee. When possible, Lessee shall cause its Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all other personal property to be assessed and billed separately from the real property of Lessor. If any of Lessee's said property shall be assessed with Lessor's real property, Lessee shall pay Lessor the taxes attributable to Lessee's property within 10 days after receipt of a written statement setting forth the taxes applicable to Lessee's property.

11. Utilities and Services.

- 11.1 Lessee shall pay for all water, gas, heat, light, power, telephone, trash disposal and other utilities and services supplied to the Premises, together with any taxes thereon. If any such services are not separately metered or billed to Lessee, Lessee shall pay a reasonable proportion, to be determined by Lessor, of all charges jointly metered or billed. There shall be no abatement of rent and Lessor shall not be liable in any respect whatsoever for the inadequacy, stoppage, interruption or discontinuance of any utility or service due to riot, strike, labor dispute, breakdown, accident, repair or other cause beyond Lessor's reasonable control or in cooperation with governmental request or directions.
- 11.2 Within fifteen days of Lessor's written request, Lessee agrees to deliver to Lessor such information, documents and/or authorization as Lessor needs in order for Lessor to comply with new or existing Applicable Requirements relating to commercial building energy usage, ratings, and/or the reporting thereof.

12. Assignment and Subletting.

12.1 Lessor's Consent Required.

- (a) Lessee shall not voluntarily or by operation of law assign, transfer, mortgage or encumber (collectively, "assign or assignment") or sublet all or any part of Lessee's interest in this Lease or in the Premises without Lessor's prior written consent.
- (b) Unless Lessee is a corporation and its stock is publicly traded on a national stock exchange, a change in the control of Lessee shall constitute an assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of the voting control of Lessee shall constitute a change in control for this purpose.
- (c) The involvement of Lessee or its assets in any transaction, or series of transactions (by way of merger, sale, acquisition, financing, transfer, leveraged buy-out or otherwise), whether or not a formal assignment or hypothecation of this Lease or Lessee's assets occurs, which results or will result in a reduction of the Net Worth of Lessee by an amount greater than 25% of such Net Worth as it was represented at the time of the execution of this Lease or at the time of the most recent assignment to which Lessor has consented, or as it exists immediately prior to said transaction or transactions constituting such reduction, whichever was or is

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

greater, shall be considered an assignment of this Lease to which Lessor may withhold its consent. "Net Worth of Lessee" shall mean the net worth of Lessee (excluding any guarantors) established under generally accepted accounting principles.

- (d) An assignment or subletting without consent shall, at Lessor's option, be a Default curable after notice per Paragraph 13.1(d), or a noncurable Breach without the necessity of any notice and grace period. If Lessor elects to treat such unapproved assignment or subletting as a noncurable Breach, Lessor may either: (i) terminate this Lease, or (ii) upon 30 days written notice, increase the monthly Base Rent to 110% of the Base Rent then in effect. Further, in the event of such Breach and rental adjustment, (i) the purchase price of any option to purchase the Premises held by Lessee shall be subject to similar adjustment to 110% of the price previously in effect, and (ii) all fixed and non-fixed rental adjustments scheduled during the remainder of the Lease term shall be increased to 110% of the scheduled adjusted rent.
 - (e) Lessee's remedy for any breach of Paragraph 12.1 by Lessor shall be limited to compensatory damages and/or injunctive relief.
 - (f) Lessor may reasonably withhold consent to a proposed assignment or subletting if Lessee is in Default at the time consent is requested.
- (g) Notwithstanding the foregoing, allowing a de minimis portion of the Premises, ie. 20 square feet or less, to be used by a third party vendor in connection with the installation of a vending machine or payphone shall not constitute a subletting.

12.2 Terms and Conditions Applicable to Assignment and Subletting.

- (a) Regardless of Lessor's consent, no assignment or subletting shall: (i) be effective without the express written assumption by such assignee or sublessee of the obligations of Lessee under this Lease, (ii) release Lessee of any obligations hereunder, or (iii) alter the primary liability of Lessee for the payment of Rent or for the performance of any other obligations to be performed by Lessee.
- (b) Lessor may accept Rent or performance of Lessee's obligations from any person other than Lessee pending approval or disapproval of an assignment. Neither a delay in the approval or disapproval of such assignment nor the acceptance of Rent or performance shall constitute a waiver or estoppel of Lessor's right to exercise its remedies for Lessee's Default or Breach.
 - (c) Lessor's consent to any assignment or subletting shall not constitute a consent to any subsequent assignment or subletting.
- (d) In the event of any Default or Breach by Lessee, Lessor may proceed directly against Lessee, any Guarantors or anyone else responsible for the performance of Lessee's obligations under this Lease, including any assignee or sublessee, without first exhausting Lessor's remedies against any other person or entity responsible therefor to Lessor, or any security held by Lessor.
- (e) Each request for consent to an assignment or subletting shall be in writing, accompanied by information relevant to Lessor's determination as to the financial and operational responsibility and appropriateness of the proposed assignee or sublessee, including but not limited to the intended use and/or required modification of the Premises, if any, together with a fee of \$500 as consideration for Lessor's considering and processing said request. Lessee agrees to provide Lessor with such other or additional information and/or documentation as may be reasonably requested. (See also Paragraph 36)
- (f) Any assignee of, or sublessee under, this Lease shall, by reason of accepting such assignment, entering into such sublease, or entering into possession of the Premises or any portion thereof, be deemed to have assumed and agreed to conform and comply with each and every term, covenant, condition and obligation herein to be observed or performed by Lessee during the term of said assignment or sublease, other than such obligations as are contrary to or inconsistent with provisions of an assignment or sublease to which Lessor has specifically consented to in writing.
- (g) Lessor's consent to any assignment or subletting shall not transfer to the assignee or sublessee any Option granted to the original Lessee by this Lease unless such transfer is specifically consented to by Lessor in writing. (See Paragraph 39.2)
- 12.3 Additional Terms and Conditions Applicable to Subletting. The following terms and conditions shall apply to any subletting by Lessee of all or any part of the Premises and shall be deemed included in all subleases under this Lease whether or not expressly incorporated therein:
- (a) Lessee hereby assigns and transfers to Lessor all of Lessee's interest in all Rent payable on any sublease, and Lessor may collect such Rent and apply same toward Lessee's obligations under this Lease; provided, however, that until a Breach shall occur in the performance of Lessee's obligations, Lessee may collect said Rent. In the event that the amount collected by Lessor exceeds Lessee's then outstanding obligations any such excess shall be refunded to Lessee. Lessor shall not, by reason of the foregoing or any assignment of such sublease, nor by reason of the collection of Rent, be deemed liable to the sublessee for any failure of Lessee to perform and comply with any of Lessee's obligations to such sublessee. Lessee hereby irrevocably authorizes and directs any such sublessee, upon receipt of a written notice from Lessor stating that a Breach exists in the performance of Lessee's obligations under this Lease, to pay to Lessor all Rent due and to become due under the sublease. Sublessee shall rely upon any such notice from Lessor and shall pay all Rents to Lessor without any obligation or right to inquire as to whether such Breach exists, notwithstanding any claim from Lessee to the contrary.
- (b) In the event of a Breach by Lessee, Lessor may, at its option, require sublessee to attorn to Lessor, in which event Lessor shall undertake the obligations of the sublessor under such sublease from the time of the exercise of said option to the expiration of such sublease; provided, however, Lessor shall not be liable for any prepaid rents or security deposit paid by such sublessee to such sublessor or for any prior Defaults or Breaches of such sublessor.
 - (c) Any matter requiring the consent of the sublessor under a sublease shall also require the consent of Lessor.
 - (d) No sublessee shall further assign or sublet all or any part of the Premises without Lessor's prior written consent.
- (e) Lessor shall deliver a copy of any notice of Default or Breach by Lessee to the sublessee, who shall have the right to cure the Default of Lessee within the grace period, if any, specified in such notice. The sublessee shall have a right of reimbursement and offset from and against Lessee for any such Defaults cured by the sublessee.

13. Default; Breach; Remedies.

- 13.1 **Default; Breach**. A "**Default**" is defined as a failure by the Lessee to comply with or perform any of the terms, covenants, conditions or Rules and Regulations under this Lease. A "**Breach**" is defined as the occurrence of one or more of the following Defaults, and the failure of Lessee to cure such Default within any applicable grace period:
- (a) The abandonment of the Premises; the vacating of the Premises prior to the expiration or termination of this Lease without providing a commercially reasonable level of security, or where the coverage of the property insurance described in Paragraph 8.3 is jeopardized as a result thereof, or without providing reasonable assurances to minimize potential vandalism; or failure to deliver to Lessor exclusive possession of the entire Premises in accordance herewith prior to the expiration or termination of this Lease.
- (b) The failure of Lessee to make any payment of Rent or any Security Deposit required to be made by Lessee hereunder, whether to Lessor or to a third party, when due, to provide reasonable evidence of insurance or surety bond, or to fulfill any obligation under this Lease which endangers or threatens life or property, where such failure continues for a period of 3 business days following written notice to Lessee. THE ACCEPTANCE BY LESSOR OF A PARTIAL PAYMENT OF RENT OR SECURITY DEPOSIT SHALL NOT CONSTITUTE A WAIVER OF ANY OF LESSOR'S RIGHTS, INCLUDING LESSOR'S RIGHT TO RECOVER POSSESSION OF THE PREMISES.
- (c) The failure of Lessee to allow Lessor and/or its agents access to the Premises or the commission of waste, act or acts constituting public or private nuisance, and/or an illegal activity on the Premises by Lessee, where such actions continue for a period of 3 business days following written notice to Lessee. In the event that Lessee commits waste, a nuisance or an illegal activity a second time then, the Lessor may elect to treat such conduct as a non-curable Breach rather than a Default.
- (d) The failure by Lessee to provide (i) reasonable written evidence of compliance with Applicable Requirements, (ii) the service contracts, (iii) the rescission of an unauthorized assignment or subletting, (iv) an Estoppel Certificate or financial statements, (v) a requested subordination, (vi) evidence concerning any guaranty and/or Guarantor, (vii) any document requested under Paragraph 42, (viii) material safety data sheets (MSDS), or (ix) any other documentation or information which Lessor may reasonably require of Lessee under the terms of this Lease, where any such failure continues for a period of 10 days following written notice to Lessee.
- (e) A Default by Lessee as to the terms, covenants, conditions or provisions of this Lease, or of the rules adopted under Paragraph 40 hereof, other than those described in subparagraphs 13.1(a), (b), (c) or (d), above, where such Default continues for a period of 30 days after written notice; provided, however, that if the nature of Lessee's Default is such that more than 30 days are reasonably required for its cure, then it shall not be deemed to be a Breach if Lessee commences such cure within said 30 day period and thereafter diligently prosecutes such cure to completion.
- (f) The occurrence of any of the following events: (i) the making of any general arrangement or assignment for the benefit of creditors; (ii) becoming a "debtor" as defined in 11 U.S.C. § 101 or any successor statute thereto (unless, in the case of a petition filed against Lessee, the same is dismissed within 60 days); (iii) the appointment of a trustee or receiver to take possession of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where possession is not restored to Lessee within 30 days; or (iv) the attachment, execution or other judicial seizure of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where such seizure is not discharged within 30 days; provided, however, in the event that any provision of this subparagraph is contrary to any applicable law, such provision shall be of no force or effect, and not affect the validity of the remaining provisions.
 - (g) The discovery that any financial statement of Lessee or of any Guarantor given to Lessor was materially false.
- (h) If the performance of Lessee's obligations under this Lease is guaranteed: (i) the death of a Guarantor, (ii) the termination of a Guarantor's liability with respect to this Lease other than in accordance with the terms of such guaranty, (iii) a Guarantor's becoming insolvent or the subject of a bankruptcy filing, (iv) a Guarantor's refusal to honor the guaranty, or (v) a Guarantor's breach of its guaranty obligation on an anticipatory basis, and Lessee's failure, within 60 days following written notice of any such event, to provide written alternative assurance or security, which, when coupled with the then existing resources of Lessee, equals or exceeds the combined financial resources of Lessee and the Guarantors that existed at the time of execution of this Lease.

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

- 13.2 **Remedies**. If Lessee fails to perform any of its affirmative duties or obligations, within 10 days after written notice (or in case of an emergency, without notice), Lessor may, at its option, perform such duty or obligation on Lessee's behalf, including but not limited to the obtaining of reasonably required bonds, insurance policies, or governmental licenses, permits or approvals. Lessee shall pay to Lessor an amount equal to 115% of the costs and expenses incurred by Lessor in such performance upon receipt of an invoice therefor. In the event of a Breach, Lessor may, with or without further notice or demand, and without limiting Lessor in the exercise of any right or remedy which Lessor may have by reason of such Breach:
- (a) Terminate Lessee's right to possession of the Premises by any lawful means, in which case this Lease shall terminate and Lessee shall immediately surrender possession to Lessor. In such event Lessor shall be entitled to recover from Lessee: (i) the unpaid Rent which had been earned at the time of termination; (ii) the worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that the Lessee proves could have been reasonably avoided; (iii) the worth at the time of award of the amount by which the unpaid rent for the balance of the term after the time of award exceeds the amount of such rental loss that the Lessee proves could be reasonably avoided; and (iv) any other amount necessary to compensate Lessor for all the detriment proximately caused by the Lessee's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom, including but not limited to the cost of recovering possession of the Premises, expenses of reletting, including necessary renovation and alteration of the Premises, reasonable attorneys' fees, and that portion of any leasing commission paid by Lessor in connection with this Lease applicable to the unexpired term of this Lease. The worth at the time of award of the amount referred to in provision (iii) of the immediately preceding sentence shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of the District within which the Premises are located at the time of award plus one percent. Efforts by Lessor to mitigate damages caused by Lessee's Breach of this Lease shall not waive Lessor's right to recover any damages to which Lessor is otherwise entitled. If termination of this Lease is obtained through the provisional remedy of unlawful detainer, Lessor shall have the right to recover in such proceeding any unpaid Rent and damages as are recoverable therein, or Lessor may reserve the right to recover all or any part thereof in a separate suit. If a notice and grace period required under Paragraph 13.1 was not previously given, a notice to pay rent or quit, or to perform or quit given to Lessee under the unlawful detainer statute shall also constitute the notice required by Paragraph 13.1. In such case, the applicable grace period required by Paragraph 13.1 and the unlawful detainer statute shall run concurrently, and the failure of Lessee to cure the Default within the greater of the two such grace periods shall constitute both an unlawful detainer and a Breach of this Lease entitling Lessor to the remedies provided for in this Lease and/or by said statute.
- (b) Continue the Lease and Lessee's right to possession and recover the Rent as it becomes due, in which event Lessee may sublet or assign, subject only to reasonable limitations. Acts of maintenance, efforts to relet, and/or the appointment of a receiver to protect the Lessor's interests, shall not constitute a termination of the Lessee's right to possession.
- (c) Pursue any other remedy now or hereafter available under the laws or judicial decisions of the state wherein the Premises are located. The expiration or termination of this Lease and/or the termination of Lessee's right to possession shall not relieve Lessee from liability under any indemnity provisions of this Lease as to matters occurring or accruing during the term hereof or by reason of Lessee's occupancy of the Premises.
- 13.3 Inducement Recapture. Any agreement for free or abated rent or other charges, the cost of tenant improvements for Lessee paid for or performed by Lessor, or for the giving or paying by Lessor to or for Lessee of any cash or other bonus, inducement or consideration for Lessee's entering into this Lease, all of which concessions are hereinafter referred to as "Inducement Provisions," shall be deemed conditioned upon Lessee's full and faithful performance of all of the terms, covenants and conditions of this Lease. Upon Breach of this Lease by Lessee, any such Inducement Provision shall automatically be deemed deleted from this Lease and of no further force or effect, and any rent, other charge, bonus, inducement or consideration theretofore abated, given or paid by Lessor under such an Inducement Provision shall be immediately due and payable by Lessee to Lessor, notwithstanding any subsequent cure of said Breach by Lessee. The acceptance by Lessor of rent or the cure of the Breach which initiated the operation of this paragraph shall not be deemed a waiver by Lessor of the provisions of this paragraph unless specifically so stated in writing by Lessor at the time of such acceptance.
- 13.4 Late Charges. Lessee hereby acknowledges that late payment by Lessee of Rent will cause Lessor to incur costs not contemplated by this Lease, the exact amount of which will be extremely difficult to ascertain. Such costs include, but are not limited to, processing and accounting charges, and late charges which may be imposed upon Lessor by any Lender. Accordingly, if any Rent shall not be received by Lessor within 5 days after such amount shall be due, then, without any requirement for notice to Lessee, Lessee shall immediately pay to Lessor a one-time late charge equal to 10% of each such overdue amount or \$100, whichever is greater. The Parties hereby agree that such late charge represents a fair and reasonable estimate of the costs Lessor will incur by reason of such late payment. Acceptance of such late charge by Lessor shall in no event constitute a waiver of Lessee's Default or Breach with respect to such overdue amount, nor prevent the exercise of any of the other rights and remedies granted hereunder. In the event that a late charge is payable hereunder, whether or not collected, for 3 consecutive installments of Base Rent, then notwithstanding any provision of this Lease to the contrary, Base Rent shall, at Lessor's option, become due and payable quarterly in advance.
- 13.5 Interest. Any monetary payment due Lessor hereunder, other than late charges, not received by Lessor, when due shall bear interest from the 31st day after it was due. The interest ("Interest") charged shall be computed at the rate of 10% per annum but shall not exceed the maximum rate allowed by law. Interest is payable in addition to the potential late charge provided for in Paragraph 13.4.

13.6 Breach by Lessor.

- (a) Notice of Breach. Lessor shall not be deemed in breach of this Lease unless Lessor fails within a reasonable time to perform an obligation required to be performed by Lessor. For purposes of this Paragraph, a reasonable time shall in no event be less than 30 days after receipt by Lessor, and any Lender whose name and address shall have been furnished to Lessee in writing for such purpose, of written notice specifying wherein such obligation of Lessor has not been performed; provided, however, that if the nature of Lessor's obligation is such that more than 30 days are reasonably required for its performance, then Lessor shall not be in breach if performance is commenced within such 30 day period and thereafter diligently pursued to completion.
- (b) **Performance by Lessee on Behalf of Lessor**. In the event that neither Lessor nor Lender cures said breach within 30 days after receipt of said notice, or if having commenced said cure they do not diligently pursue it to completion, then Lessee may elect to cure said breach at Lessee's expense and offset from Rent the actual and reasonable cost to perform such cure, provided however, that such offset shall not exceed an amount equal to the greater of one month's Base Rent or the Security Deposit, reserving Lessee's right to seek reimbursement from Lessor for any such expense in excess of such offset. Lessee shall document the cost of said cure and supply said documentation to Lessor.
- 14. Condemnation. If the Premises or any portion thereof are taken under the power of eminent domain or sold under the threat of the exercise of said power (collectively "Condemnation"), this Lease shall terminate as to the part taken as of the date the condemning authority takes title or possession, whichever first occurs. If more than 10% of the Building, or more than 25% of that portion of the Premises not occupied by any building, is taken by Condemnation, Lessee may, at Lessee's option, to be exercised in writing within 10 days after Lessor shall have given Lessee written notice of such taking (or in the absence of such notice, within 10 days after the condemning authority shall have taken possession) terminate this Lease as of the date the condemning authority takes such possession. If Lessee does not terminate this Lease in accordance with the foregoing, this Lease shall remain in full force and effect as to the portion of the Premises remaining, except that the Base Rent shall be reduced in proportion to the reduction in utility of the Premises caused by such Condemnation. Condemnation awards and/or payments shall be the property of Lessor, whether such award shall be made as compensation for diminution in value of the leasehold, the value of the part taken, or for severance damages; provided, however, that Lessee shall be entitled to any compensation paid by the condemnor for Lessee's relocation expenses, loss of business goodwill and/or Trade Fixtures, without regard to whether or not this Lease is terminated pursuant to the provisions of this Paragraph. All Alterations and Utility Installations made to the Premises by Lessee, for purposes of Condemnation only, shall be considered the property of the Lessee and Lessee shall be entitled to any and all compensation which is payable therefor. In the event that this Lease is not terminated by reason of the Condemnation, Lessor shall repair any damage to the Premises caused by such Condemnation.

15. Brokerage Fees.

- 15.1 Additional Commission. In addition to the payments owed pursuant to Paragraph 1.9 above, Lessor agrees that: (a) if Lessee exercises any Option, (b) if Lessee or anyone affiliated with Lessee acquires any rights to the Premises or other premises owned by Lessor and located within the same Project, if any, within which the Premises is located, (c) if Lessee remains in possession of the Premises, with the consent of Lessor, after the expiration of this Lease, or (d) if Base Rent is increased, whether by agreement or operation of an escalation clause herein, then, Lessor shall pay Brokers a fee in accordance with the fee schedule of the Brokers in effect at the time the Lease was executed. The provisions of this paragraph are intended to supersede the provisions of any earlier agreement to the contrary.
- 15.2 **Assumption of Obligations**. Any buyer or transferee of Lessor's interest in this Lease shall be deemed to have assumed Lessor's obligation hereunder. Brokers shall be third party beneficiaries of the provisions of Paragraphs 1.9, 15, 22 and 31. If Lessor fails to pay to Brokers any amounts due as and for brokerage fees pertaining to this Lease when due, then such amounts shall accrue Interest. In addition, if Lessor fails to pay any amounts to Lessee's Broker when due, Lessee's Broker may send written notice to Lessor and Lessee of such failure and if Lessor fails to pay such amounts within 10 days after said notice, Lessee shall pay said monies to its Broker and offset such amounts against Rent. In addition, Lessee's Broker shall be deemed to be a third party beneficiary of any commission agreement entered into by and/or between Lessor and Lessor's Broker for the limited purpose of collecting any brokerage fee owed.
- 15.3 Representations and Indemnities of Broker Relationships. Lessee and Lessor each represent and warrant to the other that it has had no dealings with any person, firm, broker, agent or finder (other than the Brokers and Agents, if any) in connection with this Lease, and that no one other than said named Brokers and Agents is entitled to any commission or finder's fee in connection herewith. Lessee and Lessor do each hereby agree to indemnify, protect, defend and hold the other harmless from and against liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any

	INITIALS

dealings or actions of the indemnifying Party, including any costs, expenses, attorneys' fees reasonably incurred with respect thereto.

16. Estoppel Certificates.

- (a) Each Party (as "Responding Party") shall within 10 days after written notice from the other Party (the "Requesting Party") execute, acknowledge and deliver to the Requesting Party a statement in writing in form similar to the then most current "Estoppel Certificate" form published by AIR CRE, plus such additional information, confirmation and/or statements as may be reasonably requested by the Requesting Party.
- (b) If the Responding Party shall fail to execute or deliver the Estoppel Certificate within such 10 day period, the Requesting Party may execute an Estoppel Certificate stating that: (i) the Lease is in full force and effect without modification except as may be represented by the Requesting Party, (ii) there are no uncured defaults in the Requesting Party's performance, and (iii) if Lessor is the Requesting Party, not more than one month's rent has been paid in advance. Prospective purchasers and encumbrancers may rely upon the Requesting Party's Estoppel Certificate, and the Responding Party shall be estopped from denying the truth of the facts contained in said Certificate. In addition, Lessee acknowledges that any failure on its part to provide such an Estoppel Certificate will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to execute and/or deliver a requested Estoppel Certificate in a timely fashion the monthly Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for remainder of the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to provide the Estoppel Certificate. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to provide the Estoppel Certificate nor prevent the exercise of any of the other rights and remedies granted hereunder.
- (c) If Lessor desires to finance, refinance, or sell the Premises, or any part thereof, Lessee and all Guarantors shall within 10 days after written notice from Lessor deliver to any potential lender or purchaser designated by Lessor such financial statements as may be reasonably required by such lender or purchaser, including but not limited to Lessee's financial statements for the past 3 years. All such financial statements shall be received by Lessor and such lender or purchaser in confidence and shall be used only for the purposes herein set forth.
- 17. Definition of Lessor. The term "Lessor" as used herein shall mean the owner or owners at the time in question of the fee title to the Premises, or, if this is a sublease, of the Lessee's interest in the prior lease. In the event of a transfer of Lessor's title or interest in the Premises or this Lease, Lessor shall deliver to the transferee or assignee (in cash or by credit) any unused Security Deposit held by Lessor. Upon such transfer or assignment and delivery of the Security Deposit, as aforesaid, the prior Lessor shall be relieved of all liability with respect to the obligations and/or covenants under this Lease thereafter to be performed by the Lessor. Subject to the foregoing, the obligations and/or covenants in this Lease to be performed by the Lessor shall be binding only upon the Lessor as hereinabove defined.
- 18. Severability. The invalidity of any provision of this Lease, as determined by a court of competent jurisdiction, shall in no way affect the validity of any other provision hereof.
- 19. Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Lease shall mean and refer to calendar days.
- 20. Limitation on Liability. The obligations of Lessor under this Lease shall not constitute personal obligations of Lessor, or its partners, members, directors, officers or shareholders, and Lessee shall look to the Premises, and to no other assets of Lessor, for the satisfaction of any liability of Lessor with respect to this Lease, and shall not seek recourse against Lessor's partners, members, directors, officers or shareholders, or any of their personal assets for such satisfaction.
- 21. Time of Essence. Time is of the essence with respect to the performance of all obligations to be performed or observed by the Parties under this Lease.
- 22. No Prior or Other Agreements; Broker Disclaimer. This Lease contains all agreements between the Parties with respect to any matter mentioned herein, and no other prior or contemporaneous agreement or understanding shall be effective. Lessor and Lessee each represents and warrants to the Brokers that it has made, and is relying solely upon, its own investigation as to the nature, quality, character and financial responsibility of the other Party to this Lease and as to the use, nature, quality and character of the Premises. Brokers have no responsibility with respect thereto or with respect to any default or breach hereof by either Party.

23. Notices.

- 23.1 Notice Requirements. All notices required or permitted by this Lease or applicable law shall be in writing and may be delivered in person (by hand or by courier) or may be sent by regular, certified or registered mail or U.S. Postal Service Express Mail, with postage prepaid, or by facsimile transmission, or by email, and shall be deemed sufficiently given if served in a manner specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specify a different address for notice, except that upon Lessee's taking possession of the Premises, the Premises shall constitute Lessee's address for notice. A copy of all notices to Lessor shall be concurrently transmitted to such party or parties at such addresses as Lessor may from time to time hereafter designate in writing.
- 23.2 **Date of Notice**. Any notice sent by registered or certified mail, return receipt requested, shall be deemed given on the date of delivery shown on the receipt card, or if no delivery date is shown, the postmark thereon. If sent by regular mail the notice shall be deemed given 72 hours after the same is addressed as required herein and mailed with postage prepaid. Notices delivered by United States Express Mail or overnight courier that guarantees next day delivery shall be deemed given 24 hours after delivery of the same to the Postal Service or courier. Notices delivered by hand, or transmitted by facsimile transmission or by email shall be deemed delivered upon actual receipt. If notice is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.
- 23.3 **Options.** Notwithstanding the foregoing, in order to exercise any Options (see paragraph 39), the Notice must be sent by Certified Mail (return receipt requested), Express Mail (signature required), courier (signature required) or some other methodology that provides a receipt establishing the date the notice was received by the Lessor.

24. Waivers.

- (a) No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any other term, covenant or condition hereof. Lessor's consent to, or approval of, any act shall not be deemed to render unnecessary the obtaining of Lessor's consent to, or approval of, any subsequent or similar act by Lessee, or be construed as the basis of an estoppel to enforce the provision or provisions of this Lease requiring such consent.
- (b) The acceptance of Rent by Lessor shall not be a waiver of any Default or Breach by Lessee. Any payment by Lessee may be accepted by Lessor on account of monies or damages due Lessor, notwithstanding any qualifying statements or conditions made by Lessee in connection therewith, which such statements and/or conditions shall be of no force or effect whatsoever unless specifically agreed to in writing by Lessor at or before the time of deposit of such payment.
- (c) THE PARTIES AGREE THAT THE TERMS OF THIS LEASE SHALL GOVERN WITH REGARD TO ALL MATTERS RELATED THERETO AND HEREBY WAIVE THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE TO THE EXTENT THAT SUCH STATUTE IS INCONSISTENT WITH THIS LEASE.

25. Disclosures Regarding The Nature of a Real Estate Agency Relationship.

- (a) When entering into a discussion with a real estate agent regarding a real estate transaction, a Lessor or Lessee should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Lessor and Lessee acknowledge being advised by the Brokers in this transaction, as follows:
- (i) <u>Lessor's Agent</u>. A Lessor's agent under a listing agreement with the Lessor acts as the agent for the Lessor only. A Lessor's agent or subagent has the following affirmative obligations: <u>To the Lessor</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessor. <u>To the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (ii) <u>Lessee's Agent</u>. An agent can agree to act as agent for the Lessee only. In these situations, the agent is not the Lessor's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Lessor. An agent acting only for a Lessee has the following affirmative obligations. <u>To the Lessee</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessee. <u>To the Lessee and the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (iii) <u>Agent Representing Both Lessor and Lessee</u>. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Lessor and the Lessee in a transaction, but only with the knowledge and consent of both the Lessor and the Lessee. In a dual agency situation, the agent has the following affirmative obligations to both the Lessor and the Lessee: (a) A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Lessor or the Lessee. (b) Other duties to the Lessor and the Lessee as stated above in subparagraphs (i) or (ii). In representing both Lessor and Lessee, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating

INITIALS
INTIALS

to either Lessee's or Lessor's financial position, motivations, bargaining position, or other personal information that may impact rent, including Lessor's willingness to accept a rent less than the listing rent or Lessee's willingness to pay rent greater than the rent offered. The above duties of the agent in a real estate transaction do not relieve a Lessor or Lessee from the responsibility to protect their own interests. Lessor and Lessee should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Both Lessor and Lessee should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.

- (b) Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this Lease may be brought against Broker more than one year after the Start Date and that the liability (including court costs and attorneys' fees), of any Broker with respect to any such lawsuit and/or legal proceeding shall not exceed the fee received by such Broker pursuant to this Lease; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.
- (c) Lessor and Lessee agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- 26. No Right To Holdover. Lessee has no right to retain possession of the Premises or any part thereof beyond the expiration or termination of this Lease. At or prior to the expiration or termination of this Lease Lessee shall deliver exclusive possession of the Premises to Lessor. For purposes of this provision and Paragraph 13.1(a), exclusive possession shall mean that Lessee shall have vacated the Premises, removed all of its personal property therefrom and that the Premises have been returned in the condition specified in this Lease. In the event that Lessee does not deliver exclusive possession to Lessor as specified above, then Lessor's damages during any holdover period shall be computed at the amount of the Rent (as defined in Paragraph 4.1) due during the last full month before the expiration or termination of this Lease (disregarding any temporary abatement of Rent that may have been in effect), but with Base Rent being 150% of the Base Rent payable during such last full month. Nothing contained herein shall be construed as consent by Lessor to any holding over by Lessee.
- 27. Cumulative Remedies. No remedy or election hereunder shall be deemed exclusive but shall, wherever possible, be cumulative with all other remedies at law or in equity.
- 28. Covenants and Conditions; Construction of Agreement. All provisions of this Lease to be observed or performed by Lessee are both covenants and conditions. In construing this Lease, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Lease. Whenever required by the context, the singular shall include the plural and vice versa. This Lease shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.
- 29. Binding Effect; Choice of Law. This Lease shall be binding upon the Parties, their personal representatives, successors and assigns and be governed by the laws of the State in which the Premises are located. Any litigation between the Parties hereto concerning this Lease shall be initiated in the county in which the Premises are located. Signatures to this Lease accomplished by means of electronic signature or similar technology shall be legal and binding.

30. Subordination; Attornment; Non-Disturbance.

- 30.1 **Subordination**. This Lease and any Option granted hereby shall be subject and subordinate to any ground lease, mortgage, deed of trust, or other hypothecation or security device (collectively, "Security Device"), now or hereafter placed upon the Premises, to any and all advances made on the security thereof, and to all renewals, modifications, and extensions thereof. Lessee agrees that the holders of any such Security Devices (in this Lease together referred to as "Lender" shall have no liability or obligation to perform any of the obligations of Lessor under this Lease. Any Lender may elect to have this Lease and/or any Option granted hereby superior to the lien of its Security Device by giving written notice thereof to Lessee, whereupon this Lease and such Options shall be deemed prior to such Security Device, notwithstanding the relative dates of the documentation or recordation thereof.
- 30.2 **Attornment**. In the event that Lessor transfers title to the Premises, or the Premises are acquired by another upon the foreclosure or termination of a Security Device to which this Lease is subordinated (i) Lessee shall, subject to the non-disturbance provisions of Paragraph 30.3, attorn to such new owner, and upon request, enter into a new lease, containing all of the terms and provisions of this Lease, with such new owner for the remainder of the term hereof, or, at the election of the new owner, this Lease will automatically become a new lease between Lessee and such new owner, and (ii) Lessor shall thereafter be relieved of any further obligations hereunder and such new owner shall assume all of Lessor's obligations, except that such new owner shall not: (a) be liable for any act or omission of any prior lessor or with respect to events occurring prior to acquisition of ownership; (b) be subject to any offsets or defenses which Lessee might have against any prior lessor, (c) be bound by prepayment of more than one month's rent, or (d) be liable for the return of any security deposit paid to any prior lessor which was not paid or credited to such new owner.
- 30.3 **Non-Disturbance**. With respect to Security Devices entered into by Lessor after the execution of this Lease, Lessee's subordination of this Lease shall be subject to receiving a commercially reasonable non-disturbance agreement (a "Non-Disturbance Agreement") from the Lender which Non-Disturbance Agreement provides that Lessee's possession of the Premises, and this Lease, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises. Further, within 60 days after the execution of this Lease, Lessor shall, if requested by Lessee, use its commercially reasonable efforts to obtain a Non-Disturbance Agreement from the holder of any pre-existing Security Device which is secured by the Premises. In the event that Lessor is unable to provide the Non-Disturbance Agreement within said 60 days, then Lessee may, at Lessee's option, directly contact Lender and attempt to negotiate for the execution and delivery of a Non-Disturbance Agreement.
- 30.4 **Self-Executing**. The agreements contained in this Paragraph 30 shall be effective without the execution of any further documents; provided, however, that, upon written request from Lessor or a Lender in connection with a sale, financing or refinancing of the Premises, Lessee and Lessor shall execute such further writings as may be reasonably required to separately document any subordination, attornment and/or Non-Disturbance Agreement provided for herein.
- 31. Attorneys' Fees. If any Party or Broker brings an action or proceeding involving the Premises whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred. In addition, Lessor shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is subsequently commenced in connection with such Default or resulting Breach (\$200 is a reasonable minimum per occurrence for such services and consultation).
- **32.** Lessor's Access; Showing Premises; Repairs. Lessor and Lessor's agents shall have the right to enter the Premises at any time, in the case of an emergency, and otherwise at reasonable times after reasonable prior notice for the purpose of showing the same to prospective purchasers, lenders, or tenants, and making such alterations, repairs, improvements or additions to the Premises as Lessor may deem necessary or desirable and the erecting, using and maintaining of utilities, services, pipes and conduits through the Premises and/or other premises as long as there is no material adverse effect on Lessee's use of the Premises. All such activities shall be without abatement of rent or liability to Lessee.
- **33. Auctions.** Lessee shall not conduct, nor permit to be conducted, any auction upon the Premises without Lessor's prior written consent. Lessor shall not be obligated to exercise any standard of reasonableness in determining whether to permit an auction.
- **34. Signs.** Lessor may place on the Premises ordinary "For Sale" signs at any time and ordinary "For Lease" signs during the last 6 months of the term hereof. Except for ordinary "for sublease" signs, Lessee shall not place any sign upon the Premises without Lessor's prior written consent. All signs must comply with all Applicable Requirements.
- **35. Termination; Merger.** Unless specifically stated otherwise in writing by Lessor, the voluntary or other surrender of this Lease by Lessee, the mutual termination or cancellation hereof, or a termination hereof by Lessor for Breach by Lessee, shall automatically terminate any sublease or lesser estate in the Premises; provided, however, that Lessor may elect to continue any one or all existing subtenancies. Lessor's failure within 10 days following any such event to elect to the contrary by written notice to the holder of any such lesser interest, shall constitute Lessor's election to have such event constitute the termination of such interest.
- **36. Consents.** All requests for consent shall be in writing. Except as otherwise provided herein, wherever in this Lease the consent of a Party is required to an act by or for the other Party, such consent shall not be unreasonably withheld or delayed. Lessor's actual reasonable costs and expenses (including but not limited to architects', attorneys', engineers' and other consultants' fees) incurred in the consideration of, or response to, a request by Lessee for any Lessor consent, including but not limited to consents to an assignment, a subletting or the presence or use of a Hazardous Substance, shall be paid by Lessee upon receipt of an invoice and supporting documentation therefor. Lessor's consent to any act, assignment or subletting shall not constitute an acknowledgment that no Default or Breach by Lessee of this Lease exists, nor shall such consent be deemed a waiver of any then existing Default or Breach, except as may be otherwise specifically stated in writing by Lessor at the time of such consent. The failure to specify herein any particular condition to Lessor's consent shall not preclude the imposition by Lessor at the time of

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

consent of such further or other conditions as are then reasonable with reference to the particular matter for which consent is being given. In the event that either Party disagrees with any determination made by the other hereunder and reasonably requests the reasons for such determination, the determining party shall furnish its reasons in writing and in reasonable detail within 10 business days following such request.

37. Guarantor.

- 37.1 **Execution**. The Guarantors, if any, shall each execute a guaranty in the form most recently published by AIR CRE, and each such Guarantor shall have the same obligations as Lessee under this Lease.
- 37.2 **Default**. It shall constitute a Default of the Lessee if any Guarantor fails or refuses, upon request to provide: (a) evidence of the execution of the guaranty, including the authority of the party signing on Guarantor's behalf to obligate Guarantor, and in the case of a corporate Guarantor, a certified copy of a resolution of its board of directors authorizing the making of such guaranty, (b) current financial statements, (c) an Estoppel Certificate, or (d) written confirmation that the guaranty is still in effect.
- **38. Quiet Possession.** Subject to payment by Lessee of the Rent and performance of all of the covenants, conditions and provisions on Lessee's part to be observed and performed under this Lease, Lessee shall have quiet possession and quiet enjoyment of the Premises during the term hereof.
- 39. Options. If Lessee is granted any Option, as defined below, then the following provisions shall apply.
- 39.1 **Definition**. "**Option**" shall mean: (a) the right to extend or reduce the term of or renew this Lease or to extend or reduce the term of or renew any lease that Lessee has on other property of Lessor; (b) the right of first refusal or first offer to lease either the Premises or other property of Lessor; (c) the right to purchase, the right of first offer to purchase or the right of first refusal to purchase the Premises or other property of Lessor.
- 39.2 **Options Personal To Original Lessee**. Any Option granted to Lessee in this Lease is personal to the original Lessee, and cannot be assigned or exercised by anyone other than said original Lessee and only while the original Lessee is in full possession of the Premises and, if requested by Lessor, with Lessee certifying that Lessee has no intention of thereafter assigning or subletting.
- 39.3 **Multiple Options**. In the event that Lessee has any multiple Options to extend or renew this Lease, a later Option cannot be exercised unless the prior Options have been validly exercised.

39.4 Effect of Default on Options.

- (a) Lessee shall have no right to exercise an Option: (i) during the period commencing with the giving of any notice of Default and continuing until said Default is cured, (ii) during the period of time any Rent is unpaid (without regard to whether notice thereof is given Lessee), (iii) during the time Lessee is in Breach of this Lease, or (iv) in the event that Lessee has been given 3 or more notices of separate Default, whether or not the Defaults are cured, during the 12 month period immediately preceding the exercise of the Option.
- (b) The period of time within which an Option may be exercised shall not be extended or enlarged by reason of Lessee's inability to exercise an Option because of the provisions of Paragraph 39.4(a).
- (c) An Option shall terminate and be of no further force or effect, notwithstanding Lessee's due and timely exercise of the Option, if, after such exercise and prior to the commencement of the extended term or completion of the purchase, (i) Lessee fails to pay Rent for a period of 30 days after such Rent becomes due (without any necessity of Lessor to give notice thereof), or (ii) if Lessee commits a Breach of this Lease.
- **40. Multiple Buildings.** If the Premises are a part of a group of buildings controlled by Lessor, Lessee agrees that it will abide by and conform to all reasonable rules and regulations which Lessor may make from time to time for the management, safety, and care of said properties, including the care and cleanliness of the grounds and including the parking, loading and unloading of vehicles, and to cause its employees, suppliers, shippers, customers, contractors and invitees to so abide and conform. Lessee also agrees to pay its fair share of common expenses incurred in connection with such rules and regulations.
- **41. Security Measures.** Lessee hereby acknowledges that the Rent payable to Lessor hereunder does not include the cost of guard service or other security measures, and that Lessor shall have no obligation whatsoever to provide same. Lessee assumes all responsibility for the protection of the Premises, Lessee, its agents and invitees and their property from the acts of third parties.
- **42. Reservations.** Lessor reserves to itself the right, from time to time, to grant, without the consent or joinder of Lessee, such easements, rights and dedications that Lessor deems necessary, and to cause the recordation of parcel maps and restrictions, so long as such easements, rights, dedications, maps and restrictions do not unreasonably interfere with the use of the Premises by Lessee. Lessee agrees to sign any documents reasonably requested by Lessor to effectuate any such easement rights, dedication, map or restrictions.
- 43. Performance Under Protest. If at any time a dispute shall arise as to any amount or sum of money to be paid by one Party to the other under the provisions hereof, the Party against whom the obligation to pay the money is asserted shall have the right to make payment "under protest" and such payment shall not be regarded as a voluntary payment and there shall survive the right on the part of said Party to institute suit for recovery of such sum. If it shall be adjudged that there was no legal obligation on the part of said Party to pay such sum or any part thereof, said Party shall be entitled to recover such sum or so much thereof as it was not legally required to pay. A Party who does not initiate suit for the recovery of sums paid "under protest" within 6 months shall be deemed to have waived its right to protest such payment.

44. Authority; Multiple Parties; Execution.

- (a) If either Party hereto is a corporation, trust, limited liability company, partnership, or similar entity, each individual executing this Lease on behalf of such entity represents and warrants that he or she is duly authorized to execute and deliver this Lease on its behalf. Each Party shall, within 30 days after request, deliver to the other Party satisfactory evidence of such authority.
- (b) If this Lease is executed by more than one person or entity as "Lessee", each such person or entity shall be jointly and severally liable hereunder. It is agreed that any one of the named Lessees shall be empowered to execute any amendment to this Lease, or other document ancillary thereto and bind all of the named Lessees, and Lessor may rely on the same as if all of the named Lessees had executed such document.
- (c) This Lease may be executed by the Parties in counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument.
- **45. Conflict.** Any conflict between the printed provisions of this Lease and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions.
- **46. Offer**. Preparation of this Lease by either Party or their agent and submission of same to the other Party shall not be deemed an offer to lease to the other Party. This Lease is not intended to be binding until executed and delivered by all Parties hereto.
- **47. Amendments.** This Lease may be modified only in writing, signed by the Parties in interest at the time of the modification. As long as they do not materially change Lessee's obligations hereunder, Lessee agrees to make such reasonable non-monetary modifications to this Lease as may be reasonably required by a Lender in connection with the obtaining of normal financing or refinancing of the Premises.
- 48. Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY OR ARISING OUT OF THIS LEASE.

OR ARISING OUT OF THIS LEASE.
49. Arbitration of Disputes. An Addendum requiring the Arbitration of all disputes between the Parties and/or Brokers arising out of this Lease is is not
attached to this Lease.
50. Accessibility; Americans with Disabilities Act.
(a) The Premises:
have not undergone an inspection by a Certified Access Specialist (CASp). Note: A Certified Access Specialist (CASp) can inspect the subject premises and determine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not require a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection the subject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the arrangements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to correct violations of construction-related accessibility standards within the premises.
have undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises met all applicable construction-related accessibility standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this Lease and agrees to keep such report confidential.

INITIALS

INITIALS
© 2019 AIR CRE. All Rights Reserved.
STN-27.30, Revised 10-22-2020

have undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises did not meet all applicable construction-related accessibility standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this Lease and agrees to keep such report confidential except as necessary to complete repairs and corrections of violations of construction related accessibility standards.

In the event that the Premises have been issued an inspection report by a CASp the Lessor shall provide a copy of the disability access inspection certificate to Lessee within 7 days of the execution of this Lease.

(b) Since compliance with the Americans with Disabilities Act (ADA) and other state and local accessibility statutes are dependent upon Lessee's specific use of the Premises, Lessor makes no warranty or representation as to whether or not the Premises comply with ADA or any similar legislation. In the event that Lessee's use of the Premises requires modifications or additions to the Premises in order to be in compliance with ADA or other accessibility statutes, Lessee agrees to make any such necessary modifications and/or additions at Lessee's expense.

LESSOR AND LESSEE HAVE CAREFULLY READ AND REVIEWED THIS LEASE AND EACH TERM AND PROVISION CONTAINED HEREIN, AND BY THE EXECUTION OF THIS LEASE SHOW THEIR INFORMED AND VOLUNTARY CONSENT THERETO. THE PARTIES HEREBY AGREE THAT, AT THE TIME THIS LEASE IS EXECUTED, THE TERMS OF THIS LEASE ARE COMMERCIALLY REASONABLE AND EFFECTUATE THE INTENT AND PURPOSE OF LESSOR AND LESSEE WITH RESPECT TO THE PREMISES.

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS LEASE OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

- 1. SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS LEASE.
- 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PREMISES. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING OF THE PREMISES, THE STRUCTURAL INTEGRITY, THE CONDITION OF THE ROOF AND OPERATING SYSTEMS, AND THE SUITABILITY OF THE PREMISES FOR LESSEE'S INTENDED USE.

WARNING: IF THE PREMISES ARE LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THE LEASE MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PREMISES ARE LOCATED.

The parties hereto have executed this Lease at the place and on the dates specified $\boldsymbol{\alpha}$	above their respective signatures.
Executed at:	Executed at:
On:	On:
By LESSOR:	By LESSEE:
By:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
By:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Address:	Address:
Federal ID No.:	Federal ID No.:
BROKER	BROKER
Attn:	Attn:
Title:	Title:
Address:	Address:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Federal ID No.:	Federal ID No.:
Broker DRE License #:	Broker DRE License #:
Agent DRE License #:	Agent DRE License #:
\sim \vee	

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2019 AIR CRE. All Rights Reserved.

STN-27.30, Revised 10-22-2020



LEASES

The following is a list of some of the major differences between the various leases published by AIR CRE.

SINGLE TENANT NET v. SINGLE TENANT GROSS

Paragraph 7

Net - the tenant is responsible for all maintenance

Gross – the landlord is responsible for the maintenance of the roof and foundations, but the tenant retains responsibility for keeping the roof and roof drainage systems free of debris.

Paragraph 8

Net – tenant pay and/or reimburses landlord for all insurance premiums

Gross – with regard to the insurance obtained by the landlord, tenant only reimburses landlord for the premiums to the extent that they exceed the base year premiums

Paragraph 10

Net – tenant pays all of the real property taxes

Gross- tenant only pays the increase over the base year

SINGLE TENANT GROSS v. LAND LEASE

The Land Lease is designed for use where the property is vacant or minimally improved. It is not intended for use re properties that are improved with buildings.

Paragraphs 1.6 and 4.3

In the Single Tenant Gross lease, the tenant pays any owner's association fees but there are no corresponding provisions in the Land lease.

Paragraph 2

Since the Land lease is intended for use with vacant properties the Lessor's warranties are much more limited (2.2) and the provisions re governmentally mandated capital expenditures have been eliminated (2.3).

Paragraph 7.1

Since the property is vacant or minimally improved, the repair and maintenance obligations of the tenant have been reduced, service contracts are limited to the landscaping (b) and the 'replacement' provisions (d) have been eliminated.

Paragraph 7.2

The landlord's maintenance and repair obligations have been eliminated.

Paragraph 9

Again, since it is contemplated that the property is vacant or only minimally improved, the provisions regarding damage and destruction in the Land lease are substantially more limited.

Paragraph 30.3

Unlike the Single Tenant Gross lease, in the Land lease the landlord is not required to attempt to obtain a non-disturbance agreement from existing lenders.

SINGLE TENANT INDUSTRIAL/COMMERCIAL v. MULTI-TENANT INDUSTRIAL/COMMERCIAL

Paragraph 1.6

Multi – sets out tenant's share of the CAMs

Single – no corresponding provision, but see paragraph 40 re situations where there are multiple buildings.

Paragraphs 2.6 - 2.10

Multi – common area definitions, rules, parking, etc.

Single – no corresponding provisions except paragraph 40.

Paragraphs 4.1 and 4.2

Multi – CAM charges defined, etc.

Single – no corresponding provision except paragraph 40

Paragraph 7.2

Multi – landlord responsible for maintenance of common areas including the roof subject to reimbursement via the CAM charges

Single – no corresponding provision since no common areas

MULTI-TENANT COMMERCIAL/INDUSTRIAL LEASES v. MULTI-TENANT OFFICE LEASES

Paragraph 1.2

Office – embraces the concept of 'rentable square footage'

Industrial – no corresponding provision

Paragraph 1.12

Office – defines building hours, hours and days of operation

Industrial – no corresponding provision

Paragraph 1.13

Office – discusses services that the landlord will NOT be providing

Industrial – no corresponding provision

Paragraph 2.6

Office – provides for charging for parking

Industrial – no corresponding provision

Paragraph 4.2

Office – discusses CAMs in terms of 95% occupancy

Industrial – no corresponding provision

Paragraph 7.1 and 7.2

Office – landlord responsible for most maintenance within the Premises

Industrial – tenant responsible for interior maintenance

Paragraph 11

Office – landlord provides utilities (except those separately metered)

Industrial – tenant pays for all utilities

Paragraph 41

Office – landlord may move tenant to other space within the project

Industrial – no corresponding provision

Rules

Office – lease has attachment listing rules re operation and parking

Industrial – no corresponding provision

MULTI-TENANT COMMERCIAL/INDUSTRIAL LEASES v. MULTI-TENANT RETAIL LEASE

Paragraph 1.2

Retail – provides for name of shopping center

Industrial – no corresponding provision

Paragraph 1.6

Retail – percentage rent

Industrial – no corresponding provision

Paragraph 1.8

Retail – merchant's association

Industrial – no corresponding provision

Paragraph 1.11

Retail – restriction re tenant's trade name

Industrial – no corresponding provision

Paragraph 2.6

Retail – employee parking restrictions

Industrial – no corresponding provision

Paragraph 2.12

Retail – allows landlord to remodel the shopping center

Industrial – no corresponding provision

Paragraph 4.2e

Retail – allows landlord to treat major tenants differently re CAMs

Industrial – no corresponding provision

Paragraph 6.1

Retail – deals with trade names, required hours of operation, exclusive use, and similar items

Industrial – no corresponding provision

Paragraph 34

Retail – strict rules re signs

Industrial – rules re signs significantly more lenient

Paragraph 42

Retail – allows landlord to relocate tenant to other space in the center

Industrial – no corresponding provision

NOTE: The above list is not intended to be a complete list of all of the differences between the various leases.



1. Basic Provisions ("Basic Provisions").

STANDARD INDUSTRIAL/COMMERCIAL SINGLE-TENANT LEASE - GROSS (DO NOT USE THIS FORM FOR MULTI-TENANT BUILDINGS)

1.1 Parties. This Lease ("Lease"), dated for reference		nd between ("Lessor"	") and
("Lessee"), (collectively the "Parties," or individually a "Parties." 1.2 Premises: That certain real property, including al	•	w Lossor under the terms of this Los	so, commonly known as
(street address, city, state, zip): ("Premises").			
the nature of the property and , if applicable, the " Project ,"			
1.3 Term: years and mo			
("Expiration Date"). (See also Paragraph 3)			
1.4 Early Possession: If the Premises are available Le	ssee may have non-exclusive possession of	the Premises commencing	("Early Possession
Date"). (See also Paragraphs 3.2 and 3.3)			
1.5 Base Rent: per month ("Base Ren	t"), payable on the day of eac	ch month commencing	(See also Paragraph 4)
If this box is checked, there are provisions in this L	ease for the Base Rent to be adjusted. See	Paragraph	
1.6 Base Rent and Other Monies Paid Upon Execution			/ \
(a) Base Rent: for the period			
(b) Security Deposit: ("Security			
(c) Association Fees: for the pe	riod		X
(d) Other: for			
(e) Total Due Upon Execution of this Lease:		\	
1.7 Agreed Use: (See also Paragraph			
1.8 Insuring Party. Lessor is the "Insuring Party". Th		(See also Paragraph 8)	\
1.9 Real Estate Brokers. (See also Paragraph 15 and(a) Representation: Each Party acknowledges	•	Agency Relationship, confirms and	consents to the following
agency relationships in this Lease with the following real est			consents to the following
Lessor's Brokerage Firm License No	, , , , , , , , , , , , , , , , , , , ,		on and Laccor (dual agent)
Lessor's Agent License No	_ is (check one): the Lessor's Agent (s	alesperson or broker associate); or	both the Lessee's
Agent and the Lessor's Agent (dual agent).	_		
Lessee's Brokerage Firm License No	Is the broker of (check one):	the Lessee; or both the Less	see and Lessor (dual
agent).			
Lessee's Agent License No	_ is (check one): the Lessee's Agent (salesperson or broker associate); or	both the Lessee's
Agent and the Lessor's Agent (dual agent).			
(b) Payment to Brokers. Upon execution and o			
separate written agreement (or if there is no such agreemen	t, the sum of or	_ % of the total Base Rent) for the b	rokerage services rendered
by the Brokers.		(11.0	1 0=)
1.10 Guarantor. The obligations of the Lessee under t	/ -	, , ,	graph 37)
1.11 Attachments. Attached hereto are the following,			
an Addendum consisting of Paragraphs	through;		
a plot plan depicting the Premises;	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \		
a current set of the Rules and Regulations;	\		
a Work Letter;			
)		
other (specify):			
2. Premises.	V		
2.1 Letting . Lessor hereby leases to Lessee, and Less			
covenants and conditions set forth in this Lease. While the			
purposes of comparison, the Base Rent stated herein is NOT different. NOTE: Lessee is advised to verify the actual size		adjustment should the actual size b	e determined to be
2.2 Condition . Lessor shall deliver the Premises to Le		Commencement Date or the Farly	Possession Date, whichever
first occurs (" Start Date "), and, so long as the required service			
following the Start Date, warrants that the existing electrical			
doors, sump pumps, if any, and all other such elements in the	e Premises, other than those constructed b	y Lessee, shall be in good operating	condition on said date,
that the surface and structural elements of the roof, bearing	, •	, ,	· ·
and that the Unit does not contain hazardous levels of any n			•
exists as of the Start Date, or if one of such systems or eleme obligation with respect to such matter, except as otherwise	-		
the nature and extent of such non-compliance, malfunction			
the HVAC systems, and (ii) 30 days as to the remaining system			
appropriate warranty period, correction of any such non-cor	npliance, malfunction or failure shall be the	obligation of Lessee at Lessee's sole	e cost and expense, except
for the roof, foundations, and bearing walls which are handle	·		_
unaware of (i) any recorded Notices of Default affecting the	Premise; (ii) any delinquent amounts due u	nder any loan secured by the Premis	ses; and (iii) any bankruptcy
proceeding affecting the Premises. 2.3 Compliance. Lessor warrants that to the best of	ts knowledge the improvements on the Pre	emises comply with the huilding cod	es annlicable laws
covenants or restrictions of record, regulations, and ordinan			
thereof, was constructed. Said warranty does not apply to the		-	•
Disabilities Act or any similar laws as a result of Lessee's use	(see Paragraph 50), or to any Alterations or	Utility Installations (as defined in Pa	aragraph 7.3(a)) made or to
be made by Lessee. NOTE: Lessee is responsible for determ	_		
Lessee's intended use, and acknowledges that past uses of			•
except as otherwise provided, promptly after receipt of writt the same at Lessor's expense. If Lessee does not give Lessor			
correction of that non-compliance shall be the obligation of		-	_
require during the term of this Lease the construction of an	•		_
or the reinforcement or other physical modification of the U		=	Territoria de la companya della companya della companya de la companya della comp
work as follows:		6.1	
(a) Subject to Paragraph 2.3(c) below, if such Ca	pital Expenditures are required as a result	of the specific and unique use of the	e Premises by Lessee as
INITIALS	INITIALS		
© 2019 AIR CRE. All Rights Reserved.		Last Edited	: 9/30/2022 2:33 PM

© 2019 AIR CRE. All Rights Reserved. STG-27.40, Revised 10-22-2020

compared with uses by tenants in general, Lessee shall be fully responsible for the cost thereof, provided, however, that if such Capital Expenditure is required during the last 2 years of this Lease and the cost thereof exceeds 6 months' Base Rent, Lessee may instead terminate this Lease unless Lessor notifies Lessee, in writing, within 10 days after receipt of Lessee's termination notice that Lessor has elected to pay the difference between the actual cost thereof and an amount equal to 6 months' Base Rent. If Lessee elects termination, Lessee shall immediately cease the use of the Premises which requires such Capital Expenditure and deliver to Lessor written notice specifying a termination date at least 90 days thereafter. Such termination date shall, however, in no event be earlier than the last day that Lessee could legally utilize the Premises without commencing such Capital Expenditure.

- (b) If such Capital Expenditure is not the result of the specific and unique use of the Premises by Lessee (such as, governmentally mandated seismic modifications), then Lessor shall pay for such Capital Expenditure and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date that on which the Base Rent is due, an amount equal to 1/144th of the portion of such costs reasonably attributable to the Premises. Lessee shall pay Interest on the balance but may prepay its obligation at any time. If, however, such Capital Expenditure is required during the last 2 years of this Lease or if Lessor reasonably determines that it is not economically feasible to pay its share thereof, Lessor shall have the option to terminate this Lease upon 90 days prior written notice to Lessee unless Lessee notifies Lessor, in writing, within 10 days after receipt of Lessor's termination notice that Lessee will pay for such Capital Expenditure. If Lessor does not elect to terminate, and fails to tender its share of any such Capital Expenditure, Lessee may advance such funds and deduct same, with Interest, from Rent until Lessor's share of such costs have been fully paid. If Lessee is unable to finance Lessor's share, or if the balance of the Rent due and payable for the remainder of this Lease is not sufficient to fully reimburse Lessee on an offset basis, Lessee shall have the right to terminate this Lease upon 30 days written notice to Lessor.
- (c) Notwithstanding the above, the provisions concerning Capital Expenditures are intended to apply only to non-voluntary, unexpected, and new Applicable Requirements. If the Capital Expenditures are instead triggered by Lessee as a result of an actual or proposed change in use, change in intensity of use, or modification to the Premises then, and in that event, Lessee shall either: (i) immediately cease such changed use or intensity of use and/or take such other steps as may be necessary to eliminate the requirement for such Capital Expenditure, or (ii) complete such Capital Expenditure at its own expense. Lessee shall not, however, have any right to terminate this Lease.
- 2.4 **Acknowledgements**. Lessee acknowledges that: (a) it has been given an opportunity to inspect and measure the Premises, (b) it has been advised by Lessor and/or Brokers to satisfy itself with respect to the size and condition of the Premises (including but not limited to the electrical, HVAC and fire sprinkler systems, security, environmental aspects, and compliance with Applicable Requirements and the Americans with Disabilities Act), and their suitability for Lessee's intended use, (c) Lessee has made such investigation as it deems necessary with reference to such matters and assumes all responsibility therefor as the same relate to its occupancy of the Premises, (d) it is not relying on any representation as to the size of the Premises made by Brokers or Lessor, (e) the square footage of the Premises was not material to Lessee's decision to lease the Premises and pay the Rent stated herein, and (f) neither Lessor, Lessor's agents, nor Brokers have made any oral or written representations or warranties with respect to said matters other than as set forth in this Lease. In addition, Lessor acknowledges that: (i) Brokers have made no representations, promises or warranties concerning Lessee's ability to honor the Lease or suitability to occupy the Premises, and (ii) it is Lessor's sole responsibility to investigate the financial capability and/or suitability of all proposed tenants.
- 2.5 **Lessee as Prior Owner/Occupant**. The warranties made by Lessor in Paragraph 2 shall be of no force or effect if immediately prior to the Start Date Lessee was the owner or occupant of the Premises. In such event, Lessee shall be responsible for any necessary corrective work.

3. Term.

- 3.1 Term. The Commencement Date, Expiration Date and Original Term of this Lease are as specified in Paragraph 1.3.
- 3.2 **Early Possession.** Any provision herein granting Lessee Early Possession of the Premises is subject to and conditioned upon the Premises being available for such possession prior to the Commencement Date. Any grant of Early Possession only conveys a non-exclusive right to occupy the Premises. If Lessee totally or partially occupies the Premises prior to the Commencement Date, the obligation to pay Base Rent shall be abated for the period of such Early Possession. All other terms of this Lease (including but not limited to the obligations to pay Real Property Taxes and insurance premiums and to maintain the Premises) shall be in effect during such period. Any such Early Possession shall not affect the Expiration Date.
- 3.3 **Delay In Possession**. Lessor agrees to use commercially reasonable efforts to deliver exclusive possession of the Premises to Lessee by the Commencement Date. If, despite said efforts, Lessor is unable to deliver possession by such date, Lessor shall not be subject to any liability therefor, nor shall such failure affect the validity of this Lease or change the Expiration Date. Lessee shall not, however, be obligated to pay Rent or perform its other obligations until Lessor delivers possession of the Premises and any period of rent abatement that Lessee would otherwise have enjoyed shall run from the date of delivery of possession and continue for a period equal to what Lessee would otherwise have enjoyed under the terms hereof, but minus any days of delay caused by the acts or omissions of Lessee. If possession is not delivered within 60 days after the Commencement Date, as the same may be extended under the terms of any Work Letter executed by Parties, Lessee may, at its option, by notice in writing within 10 days after the end of such 60 day period, cancel this Lease, in which event the Parties shall be discharged from all obligations hereunder. If such written notice is not received by Lessor within said 10 day period, Lessee's right to cancel shall terminate. If possession of the Premises is not delivered within 120 days after the Commencement Date, this Lease shall terminate unless other agreements are reached between Lessor and Lessee, in writing.
- 3.4 Lessee Compliance. Lessor shall not be required to tender possession of the Premises to Lessee until Lessee complies with its obligation to provide evidence of insurance (Paragraph 8.5). Pending delivery of such evidence, Lessee shall be required to perform all of its obligations under this Lease from and after the Start Date, including the payment of Rent, notwithstanding Lessor's election to withhold possession pending receipt of such evidence of insurance. Further, if Lessee is required to perform any other conditions prior to or concurrent with the Start Date, the Start Date shall occur but Lessor may elect to withhold possession until such conditions are satisfied.

4. Rent.

- 4.1 Rent Defined. All monetary obligations of Lessee to Lessor under the terms of this Lease (except for the Security Deposit) are deemed to be rent ("Rent").
- 4.2 **Payment**. Lessee shall cause payment of Rent to be received by Lessor in lawful money of the United States, without offset or deduction (except as specifically permitted in this Lease), on or before the day on which it is due. All monetary amounts shall be rounded to the nearest whole dollar. In the event that any invoice prepared by Lessor is inaccurate such inaccuracy shall not constitute a waiver and Lessee shall be obligated to pay the amount set forth in this Lease. Rent for any period during the term hereof which is for less than one full calendar month shall be prorated based upon the actual number of days of said month. Payment of Rent shall be made to Lessor at its address stated herein or to such other persons or place as Lessor may from time to time designate in writing. Acceptance of a payment which is less than the amount then due shall not be a waiver of Lessor's rights to the balance of such Rent, regardless of Lessor's endorsement of any check so stating. In the event that any check, draft, or other instrument of payment given by Lessee to Lessor is dishonored for any reason, Lessee agrees to pay to Lessor the sum of \$25 in addition to any Late Charge and Lessor, at its option, may require all future payments to be made by Lessee to be by cashier's check. Payments will be applied first to accrued late charges and attorney's fees, second to accrued interest, then to Base Rent, Insurance and Real Property Taxes, and any remaining amount to any other outstanding charges or costs.
- amount to any other outstanding charges or costs.

 4.3 Association Fees. In addition to the Base Rent, Lessee shall pay to Lessor each month an amount equal to any owner's association or condominium fees levied or assessed against the Premises. Said monies shall be paid at the same time and in the same manner as the Base Rent.
- security Deposit. Lessee shall deposit with Lessor upon execution hereof the Security Deposit as security for Lessee's faithful performance of its obligations under this Lease. If Lessee fails to pay Rent, or otherwise Defaults under this Lease, Lessor may use, apply or retain all or any portion of said Security Deposit for the payment of any amount already due Lessor, for Rents which will be due in the future, and/ or to reimburse or compensate Lessor for any liability, expense, loss or damage which Lessor may suffer or incur by reason thereof. If Lessor uses or applies all or any portion of the Security Deposit, Lessee shall within 10 days after written request therefor deposit monies with Lessor sufficient to restore said Security Deposit to the full amount required by this Lease. If the Base Rent increases during the term of this Lease, Lessee shall, upon written request from Lessor, deposit additional monies with Lessor so that the total amount of the Security Deposit shall at all times bear the same proportion to the increased Base Rent as the initial Security Deposit bore to the initial Base Rent. Should the Agreed Use be amended to accommodate a material change in the business of Lessee or to accommodate a sublessee or assignee, Lessor shall have the right to increase the Security Deposit to the extent necessary, in Lessor's reasonable judgment, to account for any increased wear and tear that the Premises may suffer as a result thereof. If a change in control of Lessee occurs during this Lease and following such change the financial condition of Lessee is, in Lessor's reasonable judgment, significantly reduced, Lessee shall deposit such additional monies with Lessor as shall be sufficient to cause the Security Deposit to be at a commercially reasonable level based on such change in financial condition. Lessor shall not be required to keep the Security Deposit separate from its general accounts. Within 90 days after the expiration or termination of this Lease, Lessor shall return that portion of the Security Deposit not used or applied by Lessor. Lessor shall upon written request provide Lessee with an accounting showing how that portion of the Security Deposit that was not returned was applied. No part of the Security Deposit shall be considered to be held in trust, to bear interest or to be prepayment for any monies to be paid by Lessee under this Lease. THE SECURITY DEPOSIT SHALL NOT BE USED BY LESSEE IN LIEU OF PAYMENT OF THE LAST MONTH'S RENT.

6. Use

6.1 **Use.** Lessee shall use and occupy the Premises only for the Agreed Use, or any other legal use which is reasonably comparable thereto, and for no other purpose. Lessee shall not use or permit the use of the Premises in a manner that is unlawful, creates damage, waste or a nuisance, or that disturbs occupants of or

causes damage to neighboring premises or properties. Other than guide, signal and seeing eye dogs, Lessee shall not keep or allow in the Premises any pets, animals, birds, fish, or reptiles. Lessor shall not unreasonably withhold or delay its consent to any written request for a modification of the Agreed Use, so long as the same will not impair the structural integrity of the improvements on the Premises or the mechanical or electrical systems therein, and/or is not significantly more burdensome to the Premises. If Lessor elects to withhold consent, Lessor shall within 7 days after such request give written notification of same, which notice shall include an explanation of Lessor's objections to the change in the Agreed Use.

6.2 Hazardous Substances

- (a) Reportable Uses Require Consent. The term "Hazardous Substance" as used in this Lease shall mean any product, substance, or waste whose presence, use, manufacture, disposal, transportation, or release, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentially injurious to the public health, safety or welfare, the environment or the Premises, (ii) regulated or monitored by any governmental authority, or (iii) a basis for potential liability of Lessor to any governmental agency or third party under any applicable statute or common law theory. Hazardous Substances shall include, but not be limited to, hydrocarbons, petroleum, gasoline, and/or crude oil or any products, by-products or fractions thereof. Lessee shall not engage in any activity in or on the Premises which constitutes a Reportable Use of Hazardous Substances without the express prior written consent of Lessor and timely compliance (at Lessee's expense) with all Applicable Requirements. "Reportable Use" shall mean (i) the installation or use of any above or below ground storage tank, (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and/or (iii) the presence at the Premises of a Hazardous Substance with respect to which any Applicable Requirements requires that a notice be given to persons entering or occupying the Premises or neighboring properties. Notwithstanding the foregoing, Lessee may use any ordinary and customary materials reasonably required to be used in the normal course of the Agreed Use, ordinary office supplies (copier toner, liquid paper, glue, etc.) and common household cleaning materials, so long as such use is in compliance with all Applicable Requirements, is not a Reportable Use, and does not expose the Premises or neighboring property to any meaningful risk of contamination or damage or expose Lessor to any liability therefor. In addition, Lessor may condition its consent to any Reportable Use upon receiving such additional assurances as Lessor reasonably deems necessary to protect itself, the public, the Premises and/or the environment against damage, contamination, injury and/or liability, including, but not limited to, the installation (and removal on or before Lease expiration or termination) of protective modifications (such as concrete encasements) and/or increasing the Security Deposit.
- (b) **Duty to Inform Lessor**. If Lessee knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises, other than as previously consented to by Lessor, Lessee shall immediately give written notice of such fact to Lessor, and provide Lessor with a copy of any report, notice, claim or other documentation which it has concerning the presence of such Hazardous Substance.
- (c) Lessee Remediation. Lessee shall not cause or permit any Hazardous Substance to be spilled or released in, on, under, or about the Premises (including through the plumbing or sanitary sewer system) and shall promptly, at Lessee's expense, comply with all Applicable Requirements and take all investigatory and/or remedial action reasonably recommended, whether or not formally ordered or required, for the cleanup of any contamination of, and for the maintenance, security and/or monitoring of the Premises or neighboring properties, that was caused or materially contributed to by Lessee, or pertaining to or involving any Hazardous Substance brought onto the Premises during the term of this Lease, by or for Lessee, or any third party.
- (d) Lessee Indemnification. Lessee shall indemnify, defend and hold Lessor, its agents, employees, lenders and ground lessor, if any, harmless from and against any and all loss of rents and/or damages, liabilities, judgments, claims, expenses, penalties, and attorneys' and consultants' fees arising out of or involving any Hazardous Substance brought onto the Premises by or for Lessee, or any third party (provided, however, that Lessee shall have no liability under this Lease with respect to underground migration of any Hazardous Substance under the Premises from adjacent properties not caused or contributed to by Lessee). Lessee's obligations shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Lessee, and the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease. No termination, cancellation or release agreement entered into by Lessor and Lessee shall release Lessee from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Lessor in writing at the time of such agreement.
- (e) Lessor Indemnification. Except as otherwise provided in paragraph 8.7, Lessor and its successors and assigns shall indemnify, defend, reimburse and hold Lessee, its employees and lenders, harmless from and against any and all environmental damages, including the cost of remediation, which result from Hazardous Substances which existed on the Premises prior to Lessee's occupancy or which are caused by the gross negligence or willful misconduct of Lessor, its agents or employees. Lessor's obligations, as and when required by the Applicable Requirements, shall include, but not be limited to, the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease.

 (f) Investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by
- (f) Investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by governmental entities having jurisdiction with respect to the existence of Hazardous Substances on the Premises prior to Lessee's occupancy, unless such remediation measure is required as a result of Lessee's use (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises, in which event Lessee shall be responsible for such payment. Lessee shall cooperate fully in any such activities at the request of Lessor, including allowing Lessor and Lessor's agents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remedial responsibilities.
- (g) Lessor Termination Option. If a Hazardous Substance Condition (see Paragraph 9.1(e)) occurs during the term of this Lease, unless Lessee is legally responsible therefor (in which case Lessee shall make the investigation and remediation thereof required by the Applicable Requirements and this Lease shall continue in full force and effect, but subject to Lessor's rights under Paragraph 6.2(d) and Paragraph 13), Lessor may, at Lessor's option, either (i) investigate and remediate such Hazardous Substance Condition, if required, as soon as reasonably possible at Lessor's expense, in which event this Lease shall continue in full force and effect, or (ii) if the estimated cost to remediate such condition exceeds 12 times the then monthly Base Rent or \$100,000, whichever is greater, give written notice to Lessee, within 30 days after receipt by Lessor of knowledge of the occurrence of such Hazardous Substance Condition, of Lessor's desire to terminate this Lease as of the date 60 days following the date of such notice. In the event Lessor elects to give a termination notice, Lessee may, within 10 days thereafter, give written notice to Lessor of Lessee's commitment to pay the amount by which the cost of the remediation of such Hazardous Substance Condition exceeds an amount equal to 12 times the then monthly Base Rent or \$100,000, whichever is greater. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days following such commitment. In such event, this Lease shall continue in full force and effect, and Lessor shall proceed to make such remediation as soon as reasonably possible after the required funds are available. If Lessee does not give such notice and provide the required funds or assurance thereof within the time provided, this Lease shall terminate as of the date specified in Lessor's notice of termination.
- 6.3 Lessee's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lessee's sole expense, fully, diligently and in a timely manner, materially comply with all Applicable Requirements, the requirements of any applicable fire insurance underwriter or rating bureau, and the recommendations of Lessor's engineers and/or consultants which relate in any manner to the Premises, without regard to whether said Applicable Requirements are now in effect or become effective after the Start Date. Lessee shall, within 10 days after receipt of Lessor's written request, provide Lessor with copies of all permits and other documents, and other information evidencing Lessee's compliance with any Applicable Requirements specified by Lessor, and shall immediately upon receipt, notify Lessor in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving the failure of Lessee or the Premises to comply with any Applicable Requirements. Likewise, Lessee shall immediately give written notice to Lessor of: (i) any water damage to the Premises and any suspected seepage, pooling, dampness or other condition conducive to the production of mold; or (ii) any mustiness or other odors that might indicate the presence of mold in the Premises. In addition, Lessee shall provide Lessor with copies of its business license, certificate of occupancy and/or any similar document within 10 days of the receipt of a written request therefor.
- 6.4 Inspection; Compliance. Lessor and Lessor's "Lender" (as defined in Paragraph 30) and consultants authorized by Lessor shall have the right to enter into Premises at any time in the case of an emergency, and otherwise at reasonable times after reasonable notice, for the purpose of inspecting and/or testing the condition of the Premises and/or for verifying compliance by Lessee with this Lease. The cost of any such inspections shall be paid by Lessor, unless a violation of Applicable Requirements, or a Hazardous Substance Condition (see Paragraph 9.1(e)) is found to exist or be imminent, or the inspection is requested or ordered by a governmental authority. In such case, Lessee shall upon request reimburse Lessor for the cost of such inspection, so long as such inspection is reasonably related to the violation or contamination. In addition, Lessee shall provide copies of all relevant material safety data sheets (MSDS) to Lessor within 10 days of the receipt of a written request therefor. Lessee acknowledges that any failure on its part to allow such inspections or testing will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to allow such inspections and/or testing in a timely fashion the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for the remainder to the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to allow such inspection and/or testing. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to such failure nor prevent the exercise of any of the other rights and remedies granted hereunder.

7. Maintenance; Repairs; Utility Installations; Trade Fixtures and Alterations.

7.1 Lessee's Obligations.

(a) In General. Subject to the provisions of Paragraph 2.2 (Condition), 2.3 (Compliance), 6.3 (Lessee's Compliance with Applicable Requirements), 7.2 (Lessor's Obligations), 9 (Damage or Destruction), and 14 (Condemnation), Lessee shall, at Lessee's sole expense, keep the Premises, Utility Installations (intended for Lessee's exclusive use, no matter where located), and Alterations in good order, condition and repair (whether or not the portion of the Premises requiring repairs, or the means of repairing the same, are reasonably or readily accessible to Lessee, and whether or not the need for such repairs occurs as a result of Lessee's use, any

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

prior use, the elements or the age of such portion of the Premises), including, but not limited to, all equipment or facilities, such as plumbing, HVAC equipment, electrical, lighting facilities, boilers, pressure vessels, fire protection system, fixtures, walls (interior and exterior), ceilings, floors, stairs, windows, doors, plate glass, skylights, landscaping, driveways, parking lots, fences, retaining walls, signs, sidewalks and parkways located in, on, or adjacent to the Premises. Lessee is also responsible for keeping the roof and roof drainage clean and free of debris. Lessor shall keep the surface and structural elements of the roof, foundations, and bearing walls in good repair (see paragraph 7.2). Lessee, in keeping the Premises in good order, condition and repair, shall exercise and perform good maintenance practices, specifically including the procurement and maintenance of the service contracts required by Paragraph 7.1(b) below. Lessee's obligations shall include restorations, replacements or renewals when necessary to keep the Premises and all improvements thereon or a part thereof in good order, condition and state of repair. Lessee shall, during the term of this Lease, keep the exterior appearance of the Building in a first-class condition (including, e.g. graffiti removal) consistent with the exterior appearance of other similar facilities of comparable age and size in the vicinity, including, when necessary, the exterior repainting of the Building.

- (b) Service Contracts. Lessee shall, at Lessee's sole expense, procure and maintain contracts, with copies to Lessor, in customary form and substance for, and with contractors specializing and experienced in the maintenance of the following equipment and improvements, if any, if and when installed on the Premises: (i) HVAC equipment, (ii) boiler, and pressure vessels, (iii) fire extinguishing systems, including fire alarm and/or smoke detection, (iv) landscaping and irrigation systems, and (v) clarifiers. However, Lessor reserves the right, upon notice to Lessee, to procure and maintain any or all of such service contracts, and Lessee shall reimburse Lessor, upon demand, for the cost thereof.
- (c) Failure to Perform. If Lessee fails to perform Lessee's obligations under this Paragraph 7.1, Lessor may enter upon the Premises after 10 days' prior written notice to Lessee (except in the case of an emergency, in which case no notice shall be required), perform such obligations on Lessee's behalf, and put the Premises in good order, condition and repair, and Lessee shall promptly pay to Lessor a sum equal to 115% of the cost thereof.
- (d) **Replacement**. Subject to Lessee's indemnification of Lessor as set forth in Paragraph 8.7 below, and without relieving Lessee of liability resulting from Lessee's failure to exercise and perform good maintenance practices, if an item described in Paragraph 7.1(b) cannot be repaired other than at a cost which is in excess of 50% of the cost of replacing such item, then such item shall be replaced by Lessor, and the cost thereof shall be prorated between the Parties and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date on which Base Rent is due, an amount equal to the product of multiplying the cost of such replacement by a fraction, the numerator of which is one, and the denominator of which is 144 (i.e. 1/144th of the cost per month). Lessee shall pay Interest on the unamortized balance but may prepay its obligation at any time.
- 7.2 **Lessor's Obligations**. Subject to the provisions of Paragraphs 2.2 (Condition), 2.3 (Compliance), 9 (Damage or Destruction) and 14 (Condemnation), it is intended by the Parties hereto that Lessor have no obligation, in any manner whatsoever, to repair and maintain the Premises, or the equipment therein, all of which obligations are intended to be that of the Lessee, except for the surface and structural elements of the roof, foundations and bearing walls, the repair of which shall be the responsibility of Lessor upon receipt of written notice that such a repair is necessary. It is the intention of the Parties that the terms of this Lease govern the respective obligations of the Parties as to maintenance and repair of the Premises.

7.3 Utility Installations; Trade Fixtures; Alterations.

- (a) **Definitions**. The term "**Utility Installations**" refers to all floor and window coverings, air and/or vacuum lines, power panels, electrical distribution, security and fire protection systems, communication cabling, lighting fixtures, HVAC equipment, plumbing, and fencing in or on the Premises. The term "**Trade Fixtures**" shall mean Lessee's machinery and equipment that can be removed without doing material damage to the Premises. The term "Alterations" shall mean any modification of the improvements, other than Utility Installations or Trade Fixtures, whether by addition or deletion. "**Lessee Owned Alterations and/or Utility Installations**" are defined as Alterations and/or Utility Installations made by Lessee that are not yet owned by Lessor pursuant to Paragraph 7.4(a).
- (b) Consent. Lessee shall not make any Alterations or Utility Installations to the Premises without Lessor's prior written consent. Lessee may, however, make non-structural Alterations or Utility Installations to the interior of the Premises (excluding the roof) without such consent but upon notice to Lessor, as long as they are not visible from the outside, do not involve puncturing, relocating or removing the roof or any existing walls, will not affect the electrical, plumbing, HVAC, and/or life safety systems, do not trigger the requirement for additional modifications and/or improvements to the Premises resulting from Applicable Requirements, such as compliance with Title 24, and the cumulative cost thereof during this Lease as extended does not exceed a sum equal to 3 month's Base Rent in the aggregate or a sum equal to one month's Base Rent in any one year. Notwithstanding the foregoing, Lessee shall not make or permit any roof penetrations and/or install anything on the roof without the prior written approval of Lessor. Lessor may, as a precondition to granting such approval, require Lessee to utilize a contractor chosen and/or approved by Lessor. Any Alterations or Utility Installations that Lessee shall desire to make and which require the consent of the Lessor shall be presented to Lessor in written form with detailed plans. Consent shall be deemed conditioned upon Lessee's: (i) acquiring all applicable governmental permits, (ii) furnishing Lessor with copies of both the permits and the plans and specifications prior to commencement of the work, and (iii) compliance with all conditions of said permits and other Applicable Requirements in a prompt and expeditious manner. Any Alterations or Utility Installations shall be performed in a workmanlike manner with good and sufficient materials. Lessee shall promptly upon completion furnish Lessor with as-built plans and specifications. For work which costs an amount in excess of one month's Base Rent, Lessor may condition its consent upon Lessee's nosting an
- cost of such Alteration or Utility Installation and/or upon Lessee's posting an additional Security Deposit with Lessor.

 (c) Liens; Bonds. Lessee shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lessee at or for use on the Premises, which claims are or may be secured by any mechanic's or materialmen's lien against the Premises or any interest therein. Lessee shall give Lessor not less than 10 days notice prior to the commencement of any work in, on or about the Premises, and Lessor shall have the right to post notices of non-responsibility. If Lessee shall contest the validity of any such lien, claim or demand, then Lessee shall, at its sole expense defend and protect itself, Lessor and the Premises against the same and shall pay and satisfy any such adverse judgment that may be rendered thereon before the enforcement thereof. If Lessor shall require, Lessee shall furnish a surety bond in an amount equal to 150% of the amount of such contested lien, claim or demand, indemnifying Lessor against liability for the same. If Lessor elects to participate in any such action, Lessee shall pay Lessor's attorneys' fees and costs.

7.4 Ownership; Removal; Surrender; and Restoration.

- (a) **Ownership**. Subject to Lessor's right to require removal or elect ownership as hereinafter provided, all Alterations and Utility Installations made by Lessee shall be the property of Lessee, but considered a part of the Premises. Lessor may, at any time, elect in writing to be the owner of all or any specified part of the Lessee Owned Alterations and Utility Installations. Unless otherwise instructed per paragraph 7.4(b) hereof, all Lessee Owned Alterations and Utility Installations shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.
- (b) Removal. By delivery to Lessee of written notice from Lessor not earlier than 90 and not later than 30 days prior to the end of the term of this Lease, Lessor may require that any or all Lessee Owned Alterations or Utility Installations be removed by the expiration or termination of this Lease. Lessor may require the removal at any time of all or any part of any Lessee Owned Alterations or Utility Installations made without the required consent.

 (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements,
- (c) **Surrender; Restoration**. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements, parts and surfaces thereof broom clean and free of debris, and in good operating order, condition and state of repair, ordinary wear and tear excepted. "Ordinary wear and tear" shall not include any damage or deterioration that would have been prevented by good maintenance practice. Notwithstanding the foregoing and the provisions of Paragraph 7.1(a), if the Lessee occupies the Premises for 12 months or less, then Lessee shall surrender the Premises in the same condition as delivered to Lessee on the Start Date with NO allowance for ordinary wear and tear. Lessee shall repair any damage occasioned by the installation, maintenance or removal of Trade Fixtures, Lessee owned Alterations and/or Utility Installations, furnishings, and equipment as well as the removal of any storage tank installed by or for Lessee. Lessee shall also completely remove from the Premises any and all Hazardous Substances brought onto the Premises by or for Lessee, or any third party (except Hazardous Substances which were deposited via underground migration from areas outside of the Premises) to the level specified in Applicable Requirements. Trade Fixtures shall remain the property of Lessee and shall be removed by Lessee. Any personal property of Lessee not removed on or before the Expiration Date or any earlier termination date shall be deemed to have been abandoned by Lessee and may be disposed of or retained by Lessor as Lessor may desire. The failure by Lessee to timely vacate the Premises pursuant to this Paragraph 7.4(c) without the express written consent of Lessor shall constitute a holdover under the provisions of Paragraph 26 below.

Insurance; Indemnity.

8.1 Payment of Premium Increases.

- (a) Lessee shall pay to Lessor any insurance cost increase ("Insurance Cost Increase") occurring during the term of this Lease. Insurance Cost Increase is defined as any increase in the actual cost of the insurance required under Paragraph 8.2(b), 8.3(a) and 8.3(b), over and above the Base Premium as hereinafter defined calculated on an annual basis. Insurance Cost Increase shall include but not be limited to increases resulting from the nature of Lessee's occupancy, any act or omission of Lessee, requirements of the holder of mortgage or deed of trust covering the Premises, increased valuation of the Premises and/or a premium rate increase. The parties are encouraged to fill in the Base Premium in Paragraph 1.8 with a reasonable premium for the Required Insurance based on the Agreed Use of the Premises. If the parties fail to insert a dollar amount in Paragraph 1.8, then the Base Premium shall be the lowest annual premium reasonably obtainable for the Required Insurance as of the commencement of the Original Term for the Agreed Use of the Premises. In no event, however, shall Lessee be responsible for any portion of the increase in the premium cost attributable to liability insurance carried by Lessor under Paragraph 8.2(b) in excess of \$2,000,000 per occurrence.
- (b) Lessee shall pay any such Insurance Cost Increase to Lessor within 30 days after receipt by Lessee of a copy of the premium statement or other reasonable evidence of the amount due. If the insurance policies maintained hereunder cover other property besides the Premises, Lessor shall also deliver to Lessee a statement of the amount of such Insurance Cost Increase attributable only to the Premises showing in reasonable detail the manner in which such amount was computed. Premiums for policy periods commencing prior to, or extending beyond the term of this Lease, shall be prorated to correspond to the term of this Lease.

IN	IT	ΙΑΙ	S

8.2 Liability Insurance.

- (a) Carried by Lessee. Lessee shall obtain and keep in force a Commercial General Liability policy of insurance protecting Lessee and Lessor as an additional insured against claims for bodily injury, personal injury and property damage based upon or arising out of the ownership, use, occupancy or maintenance of the Premises and all areas appurtenant thereto. Such insurance shall be on an occurrence basis providing single limit coverage in an amount not less than \$1,000,000 per occurrence with an annual aggregate of not less than \$2,000,000. Lessee shall add Lessor as an additional insured by means of an endorsement at least as broad as the Insurance Service Organization's "Additional Insured-Managers or Lessors of Premises" Endorsement. The policy shall not contain any intra-insured exclusions as between insured persons or organizations, but shall include coverage for liability assumed under this Lease as an "insured contract" for the performance of Lessee's indemnity obligations under this Lease. The limits of said insurance shall not, however, limit the liability of Lessee nor relieve Lessee of any obligation hereunder. Lessee shall provide an endorsement on its liability policy(ies) which provides that its insurance shall be primary to and not contributory with any similar insurance carried by Lessor, whose insurance shall be considered excess insurance only.
- (b) Carried by Lessor. Lessor shall maintain liability insurance as described in Paragraph 8.2(a), in addition to, and not in lieu of, the insurance required to be maintained by Lessee. Lessee shall not be named as an additional insured therein.

8.3 Property Insurance - Building, Improvements and Rental Value.

- Building and Improvements. The Insuring Party shall obtain and keep in force a policy or policies in the name of Lessor, with loss payable to Lessor, any ground-lessor, and to any Lender insuring loss or damage to the Premises. The amount of such insurance shall be equal to the full insurable replacement cost of the Premises, as the same shall exist from time to time, or the amount required by any Lender, but in no event more than the commercially reasonable and available insurable value thereof. Lessee Owned Alterations and Utility Installations, Trade Fixtures, and Lessee's personal property shall be insured by Lessee not by Lessor. If the coverage is available and commercially appropriate, such policy or policies shall insure against all risks of direct physical loss or damage (except the perils of flood and/or earthquake unless required by a Lender or included in the Base Premium), including coverage for debris removal and the enforcement of any Applicable Requirements requiring the upgrading, demolition, reconstruction or replacement of any portion of the Premises as the result of a covered loss. Said policy or policies shall also contain an agreed valuation provision in lieu of any coinsurance clause, waiver of subrogation, and inflation guard protection causing an increase in the annual property insurance coverage amount by a factor of not less than the adjusted U.S. Department of Labor Consumer Price Index for All Urban Consumers for the city nearest to where the Premises are located. If such insurance coverage has a deductible clause, the deductible amount shall not exceed \$5,000 per occurrence, and Lessee shall be liable for such deductible amount in the event of an Insured Loss.
- (b) Rental Value. The Insuring Party shall obtain and keep in force a policy or policies in the name of Lessor with loss payable to Lessor and any Lender, insuring the loss of the full Rent for one year with an extended period of indemnity for an additional 180 days ("Rental Value insurance"). Said insurance shall contain an agreed valuation provision in lieu of any coinsurance clause, and the amount of coverage shall be adjusted annually to reflect the projected Rent otherwise payable by Lessee, for the next 12 month period. Lessee shall be liable for any deductible amount in the event of such loss.
- (c) Adjacent Premises. If the Premises are part of a larger building, or of a group of buildings owned by Lessor which are adjacent to the Premises, the Lessee shall pay for any increase in the premiums for the property insurance of such building or buildings if said increase is caused by Lessee's acts, omissions, use or occupancy of the Premises.

8.4 Lessee's Property; Business Interruption Insurance; Worker's Compensation Insurance.

- (a) Property Damage. Lessee shall obtain and maintain insurance coverage on all of Lessee's personal property, Trade Fixtures, and Lessee Owned Alterations and Utility Installations. Such insurance shall be full replacement cost coverage with a deductible of not to exceed \$1,000 per occurrence. The proceeds from any such insurance shall be used by Lessee for the replacement of personal property, Trade Fixtures and Lessee Owned Alterations and Utility Installations.
- (b) Business Interruption. Lessee shall obtain and maintain loss of income and extra expense insurance in amounts as will reimburse Lessee for direct or indirect loss of earnings attributable to all perils commonly insured against by prudent lessees in the business of Lessee or attributable to prevention of access to the Premises as a result of such perils.
- (c) Worker's Compensation Insurance. Lessee shall obtain and maintain Worker's Compensation Insurance in such amount as may be required by Applicable Requirements. Such policy shall include a 'Waiver of Subrogation' endorsement. Lessee shall provide Lessor with a copy of such endorsement along with the certificate of insurance or copy of the policy required by paragraph 8.5.
- (d) No Representation of Adequate Coverage. Lessor makes no representation that the limits or forms of coverage of insurance specified herein are
- adequate to cover Lessee's property, business operations or obligations under this Lease.

 8.5 Insurance Policies. Insurance required herein shall be by companies maintaining during the policy term a "General Policyholders Rating" of at least A-, VII, as set forth in the most current issue of "Best's Insurance Guide", or such other rating as may be required by a Lender. Lessee shall not do or permit to be done anything which invalidates the required insurance policies. Lessee shall, prior to the Start Date, deliver to Lessor certified copies of policies of such insurance or certificates with copies of the required endorsements evidencing the existence and amounts of the required insurance. No such policy shall be cancelable or subject to modification except after 30 days prior written notice to Lessor. Lessee shall, at least 10 days prior to the expiration of such policies, furnish Lessor with evidence of renewals or "insurance binders" evidencing renewal thereof, or Lessor may increase his liability insurance coverage and charge the cost thereof to Lessee, which amount shall be payable by Lessee to Lessor upon demand. Such policies shall be for a term of at least one year, or the length of the remaining term of this Lease, whichever is less. If either Party shall fail to procure and maintain the insurance required to be carried by it, the other Party may, but shall not be required to, procure and maintain the same.
- 8.6 Waiver of Subrogation. Without affecting any other rights or remedies, Lessee and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is not limited by the amount of insurance carried or required, or by any deductibles applicable hereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Lessee, as the case may be, so long as the insurance is not invalidated thereby.
- 8.7 Indemnity. Except for Lessor's gross negligence or willful misconduct, Lessee shall indemnify, protect, defend and hold harmless the Premises, Lessor and its agents, Lessor's master or ground lessor, partners and Lenders, from and against any and all claims, loss of rents and/or damages, liens, judgments, penalties, attorneys' and consultants' fees, expenses and/or liabilities arising out of, involving, or in connection with, a Breach of the Lease by Lessee and/or the use and/or occupancy of the Premises and/or Project by Lessee and/or by Lessee's employees, contractors or invitees. If any action or proceeding is brought against Lessor by reason of any of the foregoing matters, Lessee shall upon notice defend the same at Lessee's expense by counsel reasonably satisfactory to Lessor and Lessor shall cooperate with Lessee in such defense. Lessor need not have first paid any such claim in order to be defended or indemnified.
- 8.8 **Exemption of Lessor and its Agents from Liability**. Notwithstanding the negligence or breach of this Lease by Lessor or its agents, neither Lessor nor its agents shall be liable under any circumstances for: (i) injury or damage to the person or goods, wares, merchandise or other property of Lessee, Lessee's employees, contractors, invitees, customers, or any other person in or about the Premises, whether such damage or injury is caused by or results from fire, steam, electricity, gas, water or rain, indoor air quality, the presence of mold or from the breakage, leakage, obstruction or other defects of pipes, fire sprinklers, wires, appliances, plumbing, HVAC or lighting fixtures, or from any other cause, whether the said injury or damage results from conditions arising upon the Premises or upon other portions of the building of which the Premises are a part, or from other sources or places, (ii) any damages arising from any act or neglect of any other tenant of Lessor or from the failure of Lessor or its agents to enforce the provisions of any other lease in the Project, or (iii) injury to Lessee's business or for any loss of income or profit therefrom. Instead, it is intended that Lessee's sole recourse in the event of such damages or injury be to file a claim on the insurance policy(ies) that Lessee is required to maintain pursuant to the provisions of paragraph 8.
- failure to Provide Insurance. Lessee acknowledges that any failure on its part to obtain or maintain the insurance required herein will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, for any month or portion thereof that Lessee does not maintain the required insurance and/or does not provide Lessor with the required binders or certificates evidencing the existence of the required insurance, the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater. The parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to maintain the required insurance. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to maintain such insurance, prevent the exercise of any of the other rights and remedies granted hereunder, nor relieve Lessee of its obligation to maintain the insurance specified in this Lease.

Damage or Destruction.

9.1 Definitions.

INITIALS

- "Premises Partial Damage" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility (a) Installations, which can reasonably be repaired in 6 months or less from the date of the damage or destruction. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (b) "Premises Total Destruction" shall mean damage or destruction to the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which cannot reasonably be repaired in 6 months or less from the date of the damage or destruction. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
 - "Insured Loss" shall mean damage or destruction to improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and

INITIALS

Trade Fixtures, which was caused by an event required to be covered by the insurance described in Paragraph 8.3(a), irrespective of any deductible amounts or coverage limits involved.

- (d) "Replacement Cost" shall mean the cost to repair or rebuild the improvements owned by Lessor at the time of the occurrence to their condition existing immediately prior thereto, including demolition, debris removal and upgrading required by the operation of Applicable Requirements, and without deduction for depreciation.
- (e) "Hazardous Substance Condition" shall mean the occurrence or discovery of a condition involving the presence of, or a contamination by, a Hazardous Substance, in, on, or under the Premises which requires restoration.
- 9.2 Partial Damage Insured Loss. If a Premises Partial Damage that is an Insured Loss occurs, then Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonably possible and this Lease shall continue in full force and effect; provided, however, that Lessee shall, at Lessor's election, make the repair of any damage or destruction the total cost to repair of which is \$10,000 or less, and, in such event, Lessor shall make any applicable insurance proceeds available to Lessee on a reasonable basis for that purpose. Notwithstanding the foregoing, if the required insurance was not in force or the insurance proceeds are not sufficient to effect such repair, the Insuring Party shall promptly contribute the shortage in proceeds (except as to the deductible which is Lessee's responsibility) as and when required to complete said repairs. In the event, however, such shortage was due to the fact that, by reason of the unique nature of the improvements, full replacement cost insurance coverage was not commercially reasonable and available, Lessor shall have no obligation to pay for the shortage in insurance proceeds or to fully restore the unique aspects of the Premises unless Lessee provides Lessor with the funds to cover same, or adequate assurance thereof, within 10 days following receipt of written notice of such shortage and request therefor. If Lessor receives said funds or adequate assurance thereof within said 10 day period, the party responsible for making the repairs shall complete them as soon as reasonably possible and this Lease shall remain in full force and effect. If such funds or assurance are not received, Lessor may nevertheless elect by written notice to Lessee within 10 days thereafter to: (i) make such restoration and repair as is commercially reasonable with Lessor paying any shortage in proceeds, in which case this Lease shall remain in full force and effect. If such funds or assurance are not received, Lessor payin
- 9.3 **Partial Damage Uninsured Loss.** If a Premises Partial Damage that is not an Insured Loss occurs, unless caused by a negligent or willful act of Lessee (in which event Lessee shall make the repairs at Lessee's expense), Lessor may either: (i) repair such damage as soon as reasonably possible at Lessor's expense, in which event this Lease shall continue in full force and effect, or (ii) terminate this Lease by giving written notice to Lessee within 30 days after receipt by Lessor of knowledge of the occurrence of such damage. Such termination shall be effective 60 days following the date of such notice. In the event Lessor elects to terminate this Lease, Lessee shall have the right within 10 days after receipt of the termination notice to give written notice to Lessor of Lessee's commitment to pay for the repair of such damage without reimbursement from Lessor. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days after making such commitment. In such event this Lease shall continue in full force and effect, and Lessor shall proceed to make such repairs as soon as reasonably possible after the required funds are available. If Lessee does not make the required commitment, this Lease shall terminate as of the date specified in the termination notice.
- required funds are available. If Lessee does not make the required commitment, this Lease shall terminate as of the date specified in the termination notice.

 9.4 **Total Destruction**. Notwithstanding any other provision hereof, if a Premises Total Destruction occurs, this Lease shall terminate 60 days following such Destruction. If the damage or destruction was caused by the gross negligence or willful misconduct of Lessee, Lessor shall have the right to recover Lessor's damages from Lessee, except as provided in Paragraph 8.6.
- 9.5 **Damage Near End of Term.** If at any time during the last 6 months of this Lease there is damage for which the cost to repair exceeds one month's Base Rent, whether or not an Insured Loss, Lessor may terminate this Lease effective 60 days following the date of occurrence of such damage by giving a written termination notice to Lessee within 30 days after the date of occurrence of such damage. Notwithstanding the foregoing, if Lessee at that time has an exercisable option to extend this Lease or to purchase the Premises, then Lessee may preserve this Lease by, (a) exercising such option and (b) providing Lessor with any shortage in insurance proceeds (or adequate assurance thereof) needed to make the repairs on or before the earlier of (i) the date which is 10 days after Lessee's receipt of Lessor's written notice purporting to terminate this Lease, or (ii) the day prior to the date upon which such option expires. If Lessee duly exercises such option during such period and provides Lessor with funds (or adequate assurance thereof) to cover any shortage in insurance proceeds, Lessor shall, at Lessor's commercially reasonable expense, repair such damage as soon as reasonably possible and this Lease shall continue in full force and effect. If Lessee fails to exercise such option and provide such funds or assurance during such period, then this Lease shall terminate on the date specified in the termination notice and Lessee's option shall be extinguished.

9.6 Abatement of Rent; Lessee's Remedies.

- (a) **Abatement.** In the event of Premises Partial Damage or Premises Total Destruction or a Hazardous Substance Condition for which Lessee is not responsible under this Lease, the Rent payable by Lessee for the period required for the repair, remediation or restoration of such damage shall be abated in proportion to the degree to which Lessee's use of the Premises is impaired, but not to exceed the proceeds received from the Rental Value insurance. All other obligations of Lessee hereunder shall be performed by Lessee, and Lessor shall have no liability for any such damage, destruction, remediation, repair or restoration except as provided herein.
- (b) Remedies. If Lessor is obligated to repair or restore the Premises and does not commence, in a substantial and meaningful way, such repair or restoration within 90 days after such obligation shall accrue, Lessee may, at any time prior to the commencement of such repair or restoration, give written notice to Lessor and to any Lenders of which Lessee has actual notice, of Lessee's election to terminate this Lease on a date not less than 60 days following the giving of such notice. If Lessee gives such notice and such repair or restoration is not commenced within 30 days thereafter, this Lease shall terminate as of the date specified in said notice. If the repair or restoration is commenced within such 30 days, this Lease shall continue in full force and effect. "Commence" shall mean either the unconditional authorization of the preparation of the required plans, or the beginning of the actual work on the Premises, whichever first occurs.

 9.7 Termination; Advance Payments. Upon termination of this Lease pursuant to Paragraph 6.2(g) or Paragraph 9, an equitable adjustment shall be made
- 9.7 **Termination; Advance Payments.** Upon termination of this Lease pursuant to Paragraph 6.2(g) or Paragraph 9, an equitable adjustment shall be made concerning advance Base Rent and any other advance payments made by Lessee to Lessor. Lessor shall, in addition, return to Lessee so much of Lessee's Security Deposit as has not been, or is not then required to be, used by Lessor.

10. Real Property Taxes.

- 10.1 **Definition**. As used herein, the term "**Real Property Taxes**" shall include any form of assessment; real estate, general, special, ordinary or extraordinary, or rental levy or tax (other than inheritance, personal income or estate taxes); improvement bond; and/or license fee imposed upon or levied against any legal or equitable interest of Lessor in the Premises or the Project, Lessor's right to other income therefrom, and/or Lessor's business of leasing, by any authority having the direct or indirect power to tax and where the funds are generated with reference to the Building address. Real Property Taxes shall also include any tax, fee, levy, assessment or charge, or any increase therein: (i) imposed by reason of events occurring during the term of this Lease, including but not limited to, a change in the ownership of the Premises, and (ii) levied or assessed on machinery or equipment provided by Lessor to Lessee pursuant to this Lease.
- 10.2 **Payment of Taxes.** Lessor shall pay the Real Property Taxes applicable to the Premises provided, however, that Lessee shall pay to Lessor the amount, if any, by which Real Property Taxes applicable to the Premises increase over the fiscal tax year during which the Commencement Date Occurs ("Tax Increase"). Payment of any such Tax Increase shall be made by Lessee to Lessor within 30 days after receipt of Lessor's written statement setting forth the amount due and computation thereof. If any such taxes shall cover any period of time prior to or after the expiration or termination of this Lease, Lessee's share of such taxes shall be prorated to cover only that portion of the tax bill applicable to the period that this Lease is in effect. In the event Lessee incurs a late charge on any Rent payment, Lessor may estimate the current Real Property Taxes, and require that the Tax Increase be paid in advance to Lessor by Lessee monthly in advance with the payment of the Base Rent. Such monthly payment shall be an amount equal to the amount of the estimated installment of the Tax Increase divided by the number of months remaining before the month in which said installment becomes delinquent. When the actual amount of the applicable Tax Increase is known, the amount of such equal monthly advance payments shall be adjusted as required to provide the funds needed to pay the applicable Tax Increase. If the amount collected by Lessor is insufficient to pay the Tax Increase when due, Lessee shall pay Lessor, upon demand, such additional sums as are necessary to pay such obligations. Advance payments may be intermingled with other moneys of Lessor and shall not bear interest. In the event of a Breach by Lessee in the performance of its obligations under this Lease, then any such advance payments may be treated by Lessor as an additional Security Deposit.
- 10.3 Additional Improvements. Notwithstanding anything to the contrary in this Paragraph 10.2, Lessee shall pay to Lessor upon demand therefor the entirety of any increase in Real Property Taxes assessed by reason of Alterations or Utility Installations placed upon the Premises by Lessee or at Lessee's request or by reason of any alterations or improvements to the Premises made by Lessor subsequent to the execution of this Lease by the Parties.
- 10.4 **Joint Assessment**. If the Premises are not separately assessed, Lessee's liability shall be an equitable proportion of the Tax Increase for all of the land and improvements included within the tax parcel assessed, such proportion to be conclusively determined by Lessor from the respective valuations assigned in the assessor's work sheets or such other information as may be reasonably available.
- 10.5 **Personal Property Taxes**. Lessee shall pay, prior to delinquency, all taxes assessed against and levied upon Lessee Owned Alterations, Utility Installations, Trade Fixtures, furnishings, equipment and all personal property of Lessee. When possible, Lessee shall cause its Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all other personal property to be assessed and billed separately from the real property of Lessor. If any of Lessee's said property shall be assessed with Lessor's real property, Lessee shall pay Lessor the taxes attributable to Lessee's property within 10 days after receipt of a written statement setting forth the taxes applicable to Lessee's property.

INITIALS		INITIALS
© 2019 AIR CRE. All	Rights Reserved.	

11. Utilities and Services.

- 11.1 Lessee shall pay for all water, gas, heat, light, power, telephone, trash disposal and other utilities and services supplied to the Premises, together with any taxes thereon. If any such services are not separately metered or billed to Lessee, Lessee shall pay a reasonable proportion, to be determined by Lessor, of all charges jointly metered or billed. There shall be no abatement of rent and Lessor shall not be liable in any respect whatsoever for the inadequacy, stoppage, interruption or discontinuance of any utility or service due to riot, strike, labor dispute, breakdown, accident, repair or other cause beyond Lessor's reasonable control or in cooperation with governmental request or directions.
- 11.2 Within fifteen days of Lessor's written request, Lessee agrees to deliver to Lessor such information, documents and/or authorization as Lessor needs in order for Lessor to comply with new or existing Applicable Requirements relating to commercial building energy usage, ratings, and/or the reporting thereof.

12. Assignment and Subletting.

12.1 Lessor's Consent Required.

- (a) Lessee shall not voluntarily or by operation of law assign, transfer, mortgage or encumber (collectively, "assign or assignment") or sublet all or any part of Lessee's interest in this Lease or in the Premises without Lessor's prior written consent.
- (b) Unless Lessee is a corporation and its stock is publicly traded on a national stock exchange, a change in the control of Lessee shall constitute an assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of the voting control of Lessee shall constitute a change in control for this purpose.
- (c) The involvement of Lessee or its assets in any transaction, or series of transactions (by way of merger, sale, acquisition, financing, transfer, leveraged buy-out or otherwise), whether or not a formal assignment or hypothecation of this Lease or Lessee's assets occurs, which results or will result in a reduction of the Net Worth of Lessee by an amount greater than 25% of such Net Worth as it was represented at the time of the execution of this Lease or at the time of the most recent assignment to which Lessor has consented, or as it exists immediately prior to said transaction or transactions constituting such reduction, whichever was or is greater, shall be considered an assignment of this Lease to which Lessor may withhold its consent. "Net Worth of Lessee" shall mean the net worth of Lessee (excluding any guarantors) established under generally accepted accounting principles.
- (d) An assignment or subletting without consent shall, at Lessor's option, be a Default curable after notice per Paragraph 13.1(d), or a noncurable Breach without the necessity of any notice and grace period. If Lessor elects to treat such unapproved assignment or subletting as a noncurable Breach, Lessor may either: (i) terminate this Lease, or (ii) upon 30 days written notice, increase the monthly Base Rent to 110% of the Base Rent then in effect. Further, in the event of such Breach and rental adjustment, (i) the purchase price of any option to purchase the Premises held by Lessee shall be subject to similar adjustment to 110% of the price previously in effect, and (ii) all fixed and non-fixed rental adjustments scheduled during the remainder of the Lease term shall be increased to 110% of the scheduled
 - Lessee's remedy for any breach of Paragraph 12.1 by Lessor shall be limited to compensatory damages and/or injunctive relief.
 - (f) Lessor may reasonably withhold consent to a proposed assignment or subletting if Lessee is in Default at the time consent is requested.
- (g) Notwithstanding the foregoing, allowing a de minimis portion of the Premises, ie. 20 square feet or less, to be used by a third party vendor in connection with the installation of a vending machine or payphone shall not constitute a subletting.

12.2 Terms and Conditions Applicable to Assignment and Subletting.

- (a) Regardless of Lessor's consent, no assignment or subletting shall: (i) be effective without the express written assumption by such assignee or sublessee of the obligations of Lessee under this Lease, (ii) release Lessee of any obligations hereunder, or (iii) alter the primary liability of Lessee for the payment of Rent or for the performance of any other obligations to be performed by Lessee.
- (b) Lessor may accept Rent or performance of Lessee's obligations from any person other than Lessee pending approval or disapproval of an assignment. Neither a delay in the approval or disapproval of such assignment nor the acceptance of Rent or performance shall constitute a waiver or estoppel of Lessor's right to exercise its remedies for Lessee's Default or Breach.
 - (c) Lessor's consent to any assignment or subletting shall not constitute a consent to any subsequent assignment or subletting.
- (d) In the event of any Default or Breach by Lessee, Lessor may proceed directly against Lessee, any Guarantors or anyone else responsible for the performance of Lessee's obligations under this Lease, including any assignee or sublessee, without first exhausting Lessor's remedies against any other person or entity responsible therefor to Lessor, or any security held by Lessor.
- (e) Each request for consent to an assignment or subletting shall be in writing, accompanied by information relevant to Lessor's determination as to the financial and operational responsibility and appropriateness of the proposed assignee or sublessee, including but not limited to the intended use and/or required modification of the Premises, if any, together with a fee of \$500 as consideration for Lessor's considering and processing said request. Lessee agrees to provide Lessor with such other or additional information and/or documentation as may be reasonably requested. (See also Paragraph 36)
- (f) Any assignee of, or sublessee under, this Lease shall, by reason of accepting such assignment, entering into such sublease, or entering into possession of the Premises or any portion thereof, be deemed to have assumed and agreed to conform and comply with each and every term, covenant, condition and obligation herein to be observed or performed by Lessee during the term of said assignment or sublease, other than such obligations as are contrary to or inconsistent with provisions of an assignment or sublease to which Lessor has specifically consented to in writing.

 (g) Lessor's consent to any assignment or subletting shall not transfer to the assignee or sublessee any Option granted to the original Lessee by this Lease
- unless such transfer is specifically consented to by Lessor in writing. (See Paragraph 39.2)
- 12.3 Additional Terms and Conditions Applicable to Subletting. The following terms and conditions shall apply to any subletting by Lessee of all or any part of the Premises and shall be deemed included in all subleases under this Lease whether or not expressly incorporated therein:
- (a) Lessee hereby assigns and transfers to Lessor all of Lessee's interest in all Rent payable on any sublease, and Lessor may collect such Rent and apply same toward Lessee's obligations under this Lease; provided, however, that until a Breach shall occur in the performance of Lessee's obligations, Lessee may collect said Rent. In the event that the amount collected by Lessor exceeds Lessee's then outstanding obligations any such excess shall be refunded to Lessee. Lessor shall not, by reason of the foregoing or any assignment of such sublease, nor by reason of the collection of Rent, be deemed liable to the sublessee for any failure of Lessee to perform and comply with any of Lessee's obligations to such sublessee. Lessee hereby irrevocably authorizes and directs any such sublessee, upon receipt of a written notice from Lessor stating that a Breach exists in the performance of Lessee's obligations under this Lease, to pay to Lessor all Rent due and to become due under the sublease. Sublessee shall rely upon any such notice from Lessor and shall pay all Rents to Lessor without any obligation or right to inquire as to whether such Breach exists, notwithstanding any claim from Lessee to the contrary.

 (b) In the event of a Breach by Lessee, Lessor may, at its option, require sublessee to attorn to Lessor, in which event Lessor shall undertake the
- obligations of the sublessor under such sublease from the time of the exercise of said option to the expiration of such sublease; provided, however, Lessor shall not be liable for any prepaid rents or security deposit paid by such sublessee to such sublessor or for any prior Defaults or Breaches of such sublessor.
 - Any matter requiring the consent of the sublessor under a sublease shall also require the consent of Lessor.
 - No sublessee shall further assign or sublet all or any part of the Premises without Lessor's prior written consent.
- Lessor shall deliver a copy of any notice of Default or Breach by Lessee to the sublessee, who shall have the right to cure the Default of Lessee within the grace period, if any, specified in such notice. The sublessee shall have a right of reimbursement and offset from and against Lessee for any such Defaults cured by the sublessee.

Default; Breach; Remedies.

- 13.1 Default; Breach. A "Default" is defined as a failure by the Lessee to comply with or perform any of the terms, covenants, conditions or Rules and Regulations under this Lease. A "Breach" is defined as the occurrence of one or more of the following Defaults, and the failure of Lessee to cure such Default within any applicable grace period:
- (a) The abandonment of the Premises; the vacating of the Premises prior to the expiration or termination of this Lease without providing a commercially reasonable level of security, or where the coverage of the property insurance described in Paragraph 8.3 is jeopardized as a result thereof, or without providing reasonable assurances to minimize potential vandalism; or failure to deliver to Lessor exclusive possession of the entire Premises in accordance herewith prior to the expiration or termination of this Lease.
- (b) The failure of Lessee to make any payment of Rent or any Security Deposit required to be made by Lessee hereunder, whether to Lessor or to a third party, when due, to provide reasonable evidence of insurance or surety bond, or to fulfill any obligation under this Lease which endangers or threatens life or property, where such failure continues for a period of 3 business days following written notice to Lessee. THE ACCEPTANCE BY LESSOR OF A PARTIAL PAYMENT OF RENT OR SECURITY DEPOSIT SHALL NOT CONSTITUTE A WAIVER OF ANY OF LESSOR'S RIGHTS. INCLUDING LESSOR'S RIGHT TO RECOVER POSSESSION OF THE PREMISES.
- (c) The failure of Lessee to allow Lessor and/or its agents access to the Premises or the commission of waste, act or acts constituting public or private nuisance, and/or an illegal activity on the Premises by Lessee, where such actions continue for a period of 3 business days following written notice to Lessee. In the event that Lessee commits waste, a nuisance or an illegal activity a second time then, the Lessor may elect to treat such conduct as a non-curable Breach rather than a
- (d) The failure by Lessee to provide (i) reasonable written evidence of compliance with Applicable Requirements, (ii) the service contracts, (iii) the rescission of an unauthorized assignment or subletting, (iv) an Estoppel Certificate or financial statements, (v) a requested subordination, (vi) evidence concerning any

IIIIIIALS

guaranty and/or Guarantor, (vii) any document requested under Paragraph 42, (viii) material safety data sheets (MSDS), or (ix) any other documentation or information which Lessor may reasonably require of Lessee under the terms of this Lease, where any such failure continues for a period of 10 days following written notice to Lessee

- (e) A Default by Lessee as to the terms, covenants, conditions or provisions of this Lease, or of the rules adopted under Paragraph 40 hereof, other than those described in subparagraphs 13.1(a), (b), (c) or (d), above, where such Default continues for a period of 30 days after written notice; provided, however, that if the nature of Lessee's Default is such that more than 30 days are reasonably required for its cure, then it shall not be deemed to be a Breach if Lessee commences such cure within said 30 day period and thereafter diligently prosecutes such cure to completion.
- (f) The occurrence of any of the following events: (i) the making of any general arrangement or assignment for the benefit of creditors; (ii) becoming a "debtor" as defined in 11 U.S.C. § 101 or any successor statute thereto (unless, in the case of a petition filed against Lessee, the same is dismissed within 60 days); (iii) the appointment of a trustee or receiver to take possession of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where possession is not restored to Lessee within 30 days; or (iv) the attachment, execution or other judicial seizure of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where such seizure is not discharged within 30 days; provided, however, in the event that any provision of this subparagraph is contrary to any applicable law, such provision shall be of no force or effect, and not affect the validity of the remaining provisions.
 - (g) The discovery that any financial statement of Lessee or of any Guarantor given to Lessor was materially false.
- (h) If the performance of Lessee's obligations under this Lease is guaranteed: (i) the death of a Guarantor, (ii) the termination of a Guarantor's liability with respect to this Lease other than in accordance with the terms of such guaranty, (iii) a Guarantor's becoming insolvent or the subject of a bankruptcy filing, (iv) a Guarantor's refusal to honor the guaranty, or (v) a Guarantor's breach of its guaranty obligation on an anticipatory basis, and Lessee's failure, within 60 days following written notice of any such event, to provide written alternative assurance or security, which, when coupled with the then existing resources of Lessee, equals or exceeds the combined financial resources of Lessee and the Guarantors that existed at the time of execution of this Lease.
- 13.2 **Remedies**. If Lessee fails to perform any of its affirmative duties or obligations, within 10 days after written notice (or in case of an emergency, without notice), Lessor may, at its option, perform such duty or obligation on Lessee's behalf, including but not limited to the obtaining of reasonably required bonds, insurance policies, or governmental licenses, permits or approvals. Lessee shall pay to Lessor an amount equal to 115% of the costs and expenses incurred by Lessor in such performance upon receipt of an invoice therefor. In the event of a Breach, Lessor may, with or without further notice or demand, and without limiting Lessor in the exercise of any right or remedy which Lessor may have by reason of such Breach:
- (a) Terminate Lessee's right to possession of the Premises by any lawful means, in which case this Lease shall terminate and Lessee shall immediately surrender possession to Lessor. In such event Lessor shall be entitled to recover from Lessee: (i) the unpaid Rent which had been earned at the time of termination; (ii) the worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that the Lessee proves could have been reasonably avoided; (iii) the worth at the time of award of the amount by which the unpaid rent for the balance of the term after the time of award exceeds the amount of such rental loss that the Lessee proves could be reasonably avoided; and (iv) any other amount necessary to compensate Lessor for all the detriment proximately caused by the Lessee's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom, including but not limited to the cost of recovering possession of the Premises, expenses of reletting, including necessary renovation and alteration of the Premises, reasonable attorneys' fees, and that portion of any leasing commission paid by Lessor in connection with this Lease applicable to the unexpired term of this Lease. The worth at the time of award of the amount referred to in provision (iii) of the immediately preceding sentence shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of the District within which the Premises are located at the time of award plus one percent. Efforts by Lessor to mitigate damages caused by Lessee's Breach of this Lease shall not waive Lessor's right to recover any damages to which Lessor is otherwise entitled. If termination of this Lease is obtained through the provisional remedy of unlawful detainer, Lessor shall have the right to recover in such proceeding any unpaid Rent and damages as are recoverable therein, or Lessor may
- detainer statute shall run concurrently, and the failure of Lessee to cure the Default within the greater of the two such grace periods shall constitute both an unlawful detainer and a Breach of this Lease entitling Lessor to the remedies provided for in this Lease and/or by said statute.

 (b) Continue the Lease and Lessee's right to possession and recover the Rent as it becomes due, in which event Lessee may sublet or assign, subject only to reasonable limitations. Acts of maintenance, efforts to relet, and/or the appointment of a receiver to protect the Lessor's interests, shall not constitute a termination of the Lessee's right to possession.
- (c) Pursue any other remedy now or hereafter available under the laws or judicial decisions of the state wherein the Premises are located. The expiration or termination of this Lease and/or the termination of Lessee's right to possession shall not relieve Lessee from liability under any indemnity provisions of this Lease as to matters occurring or accruing during the term hereof or by reason of Lessee's occupancy of the Premises.
- 13.3 **Inducement Recapture.** Any agreement for free or abated rent or other charges, the cost of tenant improvements for Lessee paid for or performed by Lessor, or for the giving or paying by Lessor to or for Lessee of any cash or other bonus, inducement or consideration for Lessee's entering into this Lease, all of which concessions are hereinafter referred to as "**Inducement Provisions**," shall be deemed conditioned upon Lessee's full and faithful performance of all of the terms, covenants and conditions of this Lease. Upon Breach of this Lease by Lessee, any such Inducement Provision shall automatically be deemed deleted from this Lease and of no further force or effect, and any rent, other charge, bonus, inducement or consideration theretofore abated, given or paid by Lessor under such an Inducement Provision shall be immediately due and payable by Lessee to Lessor, notwithstanding any subsequent cure of said Breach by Lessee. The acceptance by Lessor of rent or the cure of the Breach which initiated the operation of this paragraph shall not be deemed a waiver by Lessor of the provisions of this paragraph unless specifically so stated in writing by Lessor at the time of such acceptance.
- 13.4 Late Charges. Lessee hereby acknowledges that late payment by Lessee of Rent will cause Lessor to incur costs not contemplated by this Lease, the exact amount of which will be extremely difficult to ascertain. Such costs include, but are not limited to, processing and accounting charges, and late charges which may be imposed upon Lessor by any Lender. Accordingly, if any Rent shall not be received by Lessor within 5 days after such amount shall be due, then, without any requirement for notice to Lessee, Lessee shall immediately pay to Lessor a one-time late charge equal to 10% of each such overdue amount or \$100, whichever is greater. The Parties hereby agree that such late charge represents a fair and reasonable estimate of the costs Lessor will incur by reason of such late payment. Acceptance of such late charge by Lessor shall in no event constitute a waiver of Lessee's Default or Breach with respect to such overdue amount, nor prevent the exercise of any of the other rights and remedies granted hereunder. In the event that a late charge is payable hereunder, whether or not collected, for 3 consecutive installments of Base Rent, then notwithstanding any provision of this Lease to the contrary, Base Rent shall, at Lessor's option, become due and payable quarterly in advance.
- 13.5 Interest. Any monetary payment due Lessor hereunder, other than late charges, not received by Lessor, when due shall bear interest from the 31st day after it was due. The interest ("Interest") charged shall be computed at the rate of 10% per annum but shall not exceed the maximum rate allowed by law. Interest is payable in addition to the potential late charge provided for in Paragraph 13.4.
 - 13.6 Breach by Lessor.
- (a) **Notice of Breach**. Lessor shall not be deemed in breach of this Lease unless Lessor fails within a reasonable time to perform an obligation required to be performed by Lessor. For purposes of this Paragraph, a reasonable time shall in no event be less than 30 days after receipt by Lessor, and any Lender whose name and address shall have been furnished to Lessee in writing for such purpose, of written notice specifying wherein such obligation of Lessor has not been performed; provided, however, that if the nature of Lessor's obligation is such that more than 30 days are reasonably required for its performance, then Lessor shall not be in breach if performance is commenced within such 30 day period and thereafter diligently pursued to completion.
- (b) Performance by Lessee on Behalf of Lessor. In the event that neither Lessor nor Lender cures said breach within 30 days after receipt of said notice, or if having commenced said cure they do not diligently pursue it to completion, then Lessee may elect to cure said breach at Lessee's expense and offset from Rent the actual and reasonable cost to perform such cure, provided however, that such offset shall not exceed an amount equal to the greater of one month's Base Rent or the Security Deposit, reserving Lessee's right to seek reimbursement from Lessor for any such expense in excess of such offset. Lessee shall document the cost of said cure and supply said documentation to Lessor.
- 14. Condemnation. If the Premises or any portion thereof are taken under the power of eminent domain or sold under the threat of the exercise of said power (collectively "Condemnation"), this Lease shall terminate as to the part taken as of the date the condemning authority takes title or possession, whichever first occurs. If more than 10% of the Building, or more than 25% of that portion of the Premises not occupied by any building, is taken by Condemnation, Lessee may, at Lessee's option, to be exercised in writing within 10 days after Lessor shall have given Lessee written notice of such taking (or in the absence of such notice, within 10 days after the condemning authority shall have taken possession) terminate this Lease as of the date the condemning authority takes such possession. If Lessee does not terminate this Lease in accordance with the foregoing, this Lease shall remain in full force and effect as to the portion of the Premises remaining, except that the Base Rent shall be reduced in proportion to the reduction in utility of the Premises caused by such Condemnation. Condemnation awards and/or payments shall be the property of Lessor, whether such award shall be made as compensation for diminution in value of the leasehold, the value of the part taken, or for severance damages; provided, however, that Lessee shall be entitled to any compensation paid by the condemnor for Lessee's relocation expenses, loss of business goodwill and/or Trade Fixtures, without regard to whether or not this Lease is terminated pursuant to the provisions of this Paragraph. All Alterations and Utility Installations made to the Premises by Lessee, for purposes of Condemnation only, shall be considered the property of the Lessee and Lessee shall be entitled to any and all

INITIALS

compensation which is payable therefor. In the event that this Lease is not terminated by reason of the Condemnation, Lessor shall repair any damage to the Premises caused by such Condemnation.

15. Brokerage Fees

- 15.1 **Additional Commission**. In addition to the payments owed pursuant to Paragraph 1.9 above, Lessor agrees that: (a) if Lessee exercises any Option, (b) if Lessee or anyone affiliated with Lessee acquires any rights to the Premises or other premises owned by Lessor and located within the same Project, if any, within which the Premises is located, (c) if Lessee remains in possession of the Premises, with the consent of Lessor, after the expiration of this Lease, or (d) if Base Rent is increased, whether by agreement or operation of an escalation clause herein, then, Lessor shall pay Brokers a fee in accordance with the fee schedule of the Brokers in effect at the time the Lease was executed. The provisions of this paragraph are intended to supersede the provisions of any earlier agreement to the contrary.
- 15.2 **Assumption of Obligations**. Any buyer or transferee of Lessor's interest in this Lease shall be deemed to have assumed Lessor's obligation hereunder. Brokers shall be third party beneficiaries of the provisions of Paragraphs 1.9, 15, 22 and 31. If Lessor fails to pay to Brokers any amounts due as and for brokerage fees pertaining to this Lease when due, then such amounts shall accrue Interest. In addition, if Lessor fails to pay any amounts to Lessee's Broker when due, Lessee's Broker may send written notice to Lessor and Lessee of such failure and if Lessor fails to pay such amounts within 10 days after said notice, Lessee shall pay said monies to its Broker and offset such amounts against Rent. In addition, Lessee's Broker shall be deemed to be a third party beneficiary of any commission agreement entered into by and/or between Lessor and Lessor's Broker for the limited purpose of collecting any brokerage fee owed.
- 15.3 Representations and Indemnities of Broker Relationships. Lessee and Lessor each represent and warrant to the other that it has had no dealings with any person, firm, broker, agent or finder (other than the Brokers and Agents, if any) in connection with this Lease, and that no one other than said named Brokers and Agents is entitled to any commission or finder's fee in connection herewith. Lessee and Lessor do each hereby agree to indemnify, protect, defend and hold the other harmless from and against liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any dealings or actions of the indemnifying Party, including any costs, expenses, attorneys' fees reasonably incurred with respect thereto.

16. Estoppel Certificates.

- (a) Each Party (as "Responding Party") shall within 10 days after written notice from the other Party (the "Requesting Party") execute, acknowledge and deliver to the Requesting Party a statement in writing in form similar to the then most current "Estoppel Certificate" form published by AIR CRE, plus such additional information, confirmation and/or statements as may be reasonably requested by the Requesting Party.
- (b) If the Responding Party shall fail to execute or deliver the Estoppel Certificate within such 10 day period, the Requesting Party may execute an Estoppel Certificate stating that: (i) the Lease is in full force and effect without modification except as may be represented by the Requesting Party, (ii) there are no uncured defaults in the Requesting Party's performance, and (iii) if Lessor is the Requesting Party, not more than one month's rent has been paid in advance. Prospective purchasers and encumbrancers may rely upon the Requesting Party's Estoppel Certificate, and the Responding Party shall be estopped from denying the truth of the facts contained in said Certificate. In addition, Lessee acknowledges that any failure on its part to provide such an Estoppel Certificate will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to execute and/or deliver a requested Estoppel Certificate in a timely fashion the monthly Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for remainder of the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to provide the Estoppel Certificate. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to provide the Estoppel Certificate nor prevent the exercise of any of the other rights and remedies granted hereunder.
- (c) If Lessor desires to finance, refinance, or sell the Premises, or any part thereof, Lessee and all Guarantors shall within 10 days after written notice from Lessor deliver to any potential lender or purchaser designated by Lessor such financial statements as may be reasonably required by such lender or purchaser, including but not limited to Lessee's financial statements for the past 3 years. All such financial statements shall be received by Lessor and such lender or purchaser in confidence and shall be used only for the purposes herein set forth.
- 17. Definition of Lessor. The term "Lessor" as used herein shall mean the owner or owners at the time in question of the fee title to the Premises, or, if this is a sublease, of the Lessee's interest in the prior lease. In the event of a transfer of Lessor's title or interest in the Premises or this Lease, Lessor shall deliver to the transferee or assignee (in cash or by credit) any unused Security Deposit held by Lessor. Upon such transfer or assignment and delivery of the Security Deposit, as aforesaid, the prior Lessor shall be relieved of all liability with respect to the obligations and/or covenants under this Lease thereafter to be performed by the Lessor. Subject to the foregoing, the obligations and/or covenants in this Lease to be performed by the Lessor shall be binding only upon the Lessor as hereinabove defined.
- 18. Severability. The invalidity of any provision of this Lease, as determined by a court of competent jurisdiction, shall in no way affect the validity of any other provision hereof.
- 19. Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Lease shall mean and refer to calendar days.
- 20. Limitation on Liability. The obligations of Lessor under this Lease shall not constitute personal obligations of Lessor, or its partners, members, directors, officers or shareholders, and Lessee shall look to the Premises, and to no other assets of Lessor, for the satisfaction of any liability of Lessor with respect to this Lease, and shall not seek recourse against Lessor's partners, members, directors, officers or shareholders, or any of their personal assets for such satisfaction.
- 21. Time of Essence. Time is of the essence with respect to the performance of all obligations to be performed or observed by the Parties under this Lease.
- 22. No Prior or Other Agreements; Broker Disclaimer. This Lease contains all agreements between the Parties with respect to any matter mentioned herein, and no other prior or contemporaneous agreement or understanding shall be effective. Lessor and Lessee each represents and warrants to the Brokers that it has made, and is relying solely upon, its own investigation as to the nature, quality, character and financial responsibility of the other Party to this Lease and as to the use, nature, quality and character of the Premises. Brokers have no responsibility with respect thereto or with respect to any default or breach hereof by either Party.

23. Notices.

- 23.1 Notice Requirements. All notices required or permitted by this Lease or applicable law shall be in writing and may be delivered in person (by hand or by courier) or may be sent by regular, certified or registered mail or U.S. Postal Service Express Mail, with postage prepaid, or by facsimile transmission, or by email, and shall be deemed sufficiently given if served in a manner specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specify a different address for notice, except that upon Lessee's taking possession of the Premises, the Premises shall constitute Lessee's address for notice. A copy of all notices to Lessor shall be concurrently transmitted to such party or parties at such addresses as Lessor may from time to time hereafter designate in writing
- party or parties at such addresses as Lessor may from time to time hereafter designate in writing.

 23.2 Date of Notice. Any notice sent by registered or certified mail, return receipt requested, shall be deemed given on the date of delivery shown on the receipt card, or if no delivery date is shown, the postmark thereon. If sent by regular mail the notice shall be deemed given 72 hours after the same is addressed as required herein and mailed with postage prepaid. Notices delivered by United States Express Mail or overnight courier that guarantees next day delivery shall be deemed given 24 hours after delivery of the same to the Postal Service or courier. Notices delivered by hand, or transmitted by facsimile transmission or by email shall be deemed delivered upon actual receipt. If notice is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.
- 23.3 **Options.** Notwithstanding the foregoing, in order to exercise any Options (see paragraph 39), the Notice must be sent by Certified Mail (return receipt requested), Express Mail (signature required), courier (signature required) or some other methodology that provides a receipt establishing the date the notice was received by the Lessor.

24. Waivers.

- (a) No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any other term, covenant or condition hereof. Lessor's consent to, or approval of, any act shall not be deemed to render unnecessary the obtaining of Lessor's consent to, or approval of, any subsequent or similar act by Lessee, or be construed as the basis of an estoppel to enforce the provision or provisions of this Lease requiring such consent.
- (b) The acceptance of Rent by Lessor shall not be a waiver of any Default or Breach by Lessee. Any payment by Lessee may be accepted by Lessor on account of monies or damages due Lessor, notwithstanding any qualifying statements or conditions made by Lessee in connection therewith, which such statements and/or conditions shall be of no force or effect whatsoever unless specifically agreed to in writing by Lessor at or before the time of deposit of such payment.
- (c) THE PARTIES AGREE THAT THE TERMS OF THIS LEASE SHALL GOVERN WITH REGARD TO ALL MATTERS RELATED THERETO AND HEREBY WAIVE THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE TO THE EXTENT THAT SUCH STATUTE IS INCONSISTENT WITH THIS LEASE.

25. Disclosures Regarding The Nature of a Real Estate Agency Relationship.

(a) When entering into a discussion with a real estate agent regarding a real estate transaction, a Lessor or Lessee should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Lessor and Lessee acknowledge being advised by the Brokers in this transaction, as follows:

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

- (i) <u>Lessor's Agent</u>. A Lessor's agent under a listing agreement with the Lessor acts as the agent for the Lessor only. A Lessor's agent or subagent has the following affirmative obligations: <u>To the Lessor</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessor. <u>To the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (ii) <u>Lessee's Agent</u>. An agent can agree to act as agent for the Lessee only. In these situations, the agent is not the Lessor's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Lessor. An agent acting only for a Lessee has the following affirmative obligations. <u>To the Lessee</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessee. <u>To the Lessee and the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (iii) Agent Representing Both Lessor and Lessee. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Lessor and the Lessee in a transaction, but only with the knowledge and consent of both the Lessor and the Lessee. In a dual agency situation, the agent has the following affirmative obligations to both the Lessor and the Lessee: (a) A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Lessor or the Lessee. (b) Other duties to the Lessor and the Lessee as stated above in subparagraphs (i) or (ii). In representing both Lessor and Lessee, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating to either Lessee's or Lessor's financial position, motivations, bargaining position, or other personal information that may impact rent, including Lessor's willingness to accept a rent less than the listing rent or Lessee's willingness to pay rent greater than the rent offered. The above duties of the agent in a real estate transaction do not relieve a Lessor or Lessee from the responsibility to protect their own interests. Lessor and Lessee should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Both Lessor and Lessee should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.
- (b) Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this Lease may be brought against Broker more than one year after the Start Date and that the liability (including court costs and attorneys' fees), of any Broker with respect to any such lawsuit and/or legal proceeding shall not exceed the fee received by such Broker pursuant to this Lease; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.
- (c) Lessor and Lessee agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- 26. No Right To Holdover. Lessee has no right to retain possession of the Premises or any part thereof beyond the expiration or termination of this Lease. At or prior to the expiration or termination of this Lease Lessee shall deliver exclusive possession of the Premises to Lessor. For purposes of this provision and Paragraph 13.1(a), exclusive possession shall mean that Lessee shall have vacated the Premises, removed all of its personal property therefrom and that the Premises have been returned in the condition specified in this Lease. In the event that Lessee does not deliver exclusive possession to Lessor as specified above, then Lessor's damages during any holdover period shall be computed at the amount of the Rent (as defined in Paragraph 4.1) due during the last full month before the expiration or termination of this Lease (disregarding any temporary abatement of Rent that may have been in effect), but with Base Rent being 150% of the Base Rent payable during such last full month. Nothing contained herein shall be construed as consent by Lessor to any holding over by Lessee.
- 27. Cumulative Remedies. No remedy or election hereunder shall be deemed exclusive but shall, wherever possible, be cumulative with all other remedies at law or in equity.
- 28. Covenants and Conditions; Construction of Agreement. All provisions of this Lease to be observed or performed by Lessee are both covenants and conditions. In construing this Lease, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Lease. Whenever required by the context, the singular shall include the plural and vice versa. This Lease shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.
- 29. Binding Effect; Choice of Law. This Lease shall be binding upon the Parties, their personal representatives, successors and assigns and be governed by the laws of the State in which the Premises are located. Any litigation between the Parties hereto concerning this Lease shall be initiated in the county in which the Premises are located. Signatures to this Lease accomplished by means of electronic signature or similar technology shall be legal and binding.

30. Subordination; Attornment; Non-Disturbance.

- 30.1 **Subordination**. This Lease and any Option granted hereby shall be subject and subordinate to any ground lease, mortgage, deed of trust, or other hypothecation or security device (collectively, "**Security Device**"), now or hereafter placed upon the Premises, to any and all advances made on the security thereof, and to all renewals, modifications, and extensions thereof. Lessee agrees that the holders of any such Security Devices (in this Lease together referred to as "**Lender**") shall have no liability or obligation to perform any of the obligations of Lessor under this Lease. Any Lender may elect to have this Lease and/or any Option granted hereby superior to the lien of its Security Device by giving written notice thereof to Lessee, whereupon this Lease and such Options shall be deemed prior to such Security Device, notwithstanding the relative dates of the documentation or recordation thereof.
- 30.2 **Attornment**. In the event that Lessor transfers title to the Premises, or the Premises are acquired by another upon the foreclosure or termination of a Security Device to which this Lease is subordinated (i) Lessee shall, subject to the non-disturbance provisions of Paragraph 30.3, attorn to such new owner, and upon request, enter into a new lease, containing all of the terms and provisions of this Lease, with such new owner for the remainder of the term hereof, or, at the election of the new owner, this Lease will automatically become a new lease between Lessee and such new owner, and (ii) Lessor shall thereafter be relieved of any further obligations hereunder and such new owner shall assume all of Lessor's obligations, except that such new owner shall not: (a) be liable for any act or omission of any prior lessor or with respect to events occurring prior to acquisition of ownership; (b) be subject to any offsets or defenses which Lessee might have against any prior lessor, (c) be bound by prepayment of more than one month's rent, or (d) be liable for the return of any security deposit paid to any prior lessor which was not paid or credited to such new owner.
- 30.3 **Non-Disturbance**. With respect to Security Devices entered into by Lessor after the execution of this Lease, Lessee's subordination of this Lease shall be subject to receiving a commercially reasonable non-disturbance agreement (a "**Non-Disturbance Agreement**") from the Lender which Non-Disturbance Agreement provides that Lessee's possession of the Premises, and this Lease, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises. Further, within 60 days after the execution of this Lease, Lessor shall, if requested by Lessee, use its commercially reasonable efforts to obtain a Non-Disturbance Agreement from the holder of any pre-existing Security Device which is secured by the Premises. In the event that Lessor is unable to provide the Non-Disturbance Agreement within said 60 days, then Lessee may, at Lessee's option, directly contact Lender and attempt to negotiate for the execution and delivery of a Non-Disturbance Agreement.
- 30.4 **Self-Executing**. The agreements contained in this Paragraph 30 shall be effective without the execution of any further documents; provided, however, that, upon written request from Lessor or a Lender in connection with a sale, financing or refinancing of the Premises, Lessee and Lessor shall execute such further writings as may be reasonably required to separately document any subordination, attornment and/or Non-Disturbance Agreement provided for herein.
- 31. Attorneys' Fees. If any Party or Broker brings an action or proceeding involving the Premises whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred. In addition, Lessor shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is subsequently commenced in connection with such Default or resulting Breach (\$200 is a reasonable minimum per occurrence for such services and consultation).
- **32.** Lessor's Access; Showing Premises; Repairs. Lessor and Lessor's agents shall have the right to enter the Premises at any time, in the case of an emergency, and otherwise at reasonable times after reasonable prior notice for the purpose of showing the same to prospective purchasers, lenders, or tenants, and making such alterations, repairs, improvements or additions to the Premises as Lessor may deem necessary or desirable and the erecting, using and maintaining of utilities, services, pipes and conduits through the Premises and/or other premises as long as there is no material adverse effect on Lessee's use of the Premises. All such activities shall be without abatement of rent or liability to Lessee.

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

- **33. Auctions.** Lessee shall not conduct, nor permit to be conducted, any auction upon the Premises without Lessor's prior written consent. Lessor shall not be obligated to exercise any standard of reasonableness in determining whether to permit an auction.
- **34. Signs.** Lessor may place on the Premises ordinary "For Sale" signs at any time and ordinary "For Lease" signs during the last 6 months of the term hereof. Except for ordinary "for sublease" signs, Lessee shall not place any sign upon the Premises without Lessor's prior written consent. All signs must comply with all Applicable Requirements.
- **35. Termination; Merger.** Unless specifically stated otherwise in writing by Lessor, the voluntary or other surrender of this Lease by Lessee, the mutual termination or cancellation hereof, or a termination hereof by Lessor for Breach by Lessee, shall automatically terminate any sublease or lesser estate in the Premises; provided, however, that Lessor may elect to continue any one or all existing subtenancies. Lessor's failure within 10 days following any such event to elect to the contrary by written notice to the holder of any such lesser interest, shall constitute Lessor's election to have such event constitute the termination of such interest.
- **36. Consents.** All requests for consent shall be in writing. Except as otherwise provided herein, wherever in this Lease the consent of a Party is required to an act by or for the other Party, such consent shall not be unreasonably withheld or delayed. Lessor's actual reasonable costs and expenses (including but not limited to architects', attorneys', engineers' and other consultants' fees) incurred in the consideration of, or response to, a request by Lessee for any Lessor consent, including but not limited to consents to an assignment, a subletting or the presence or use of a Hazardous Substance, shall be paid by Lessee upon receipt of an invoice and supporting documentation therefor. Lessor's consent to any act, assignment or subletting shall not constitute an acknowledgment that no Default or Breach by Lessee of this Lease exists, nor shall such consent be deemed a waiver of any then existing Default or Breach, except as may be otherwise specifically stated in writing by Lessor at the time of such consent. The failure to specify herein any particular condition to Lessor's consent shall not preclude the imposition by Lessor at the time of consent of such further or other conditions as are then reasonable with reference to the particular matter for which consent is being given. In the event that either Party disagrees with any determination made by the other hereunder and reasonably requests the reasons for such determination, the determining party shall furnish its reasons in writing and in reasonable detail within 10 business days following such request.

37. Guarantor.

- 37.1 Execution. The Guarantors, if any, shall each execute a guaranty in the form most recently published by AIR CRE.
- 37.2 **Default**. It shall constitute a Default of the Lessee if any Guarantor fails or refuses, upon request to provide: (a) evidence of the execution of the guaranty, including the authority of the party signing on Guarantor's behalf to obligate Guarantor, and in the case of a corporate Guarantor, a certified copy of a resolution of its board of directors authorizing the making of such guaranty, (b) current financial statements, (c) an Estoppel Certificate, or (d) written confirmation that the guaranty is still in effect
- **38. Quiet Possession.** Subject to payment by Lessee of the Rent and performance of all of the covenants, conditions and provisions on Lessee's part to be observed and performed under this Lease, Lessee shall have quiet possession and quiet enjoyment of the Premises during the term hereof.
- 39. Options. If Lessee is granted any Option, as defined below, then the following provisions shall apply.
- 39.1 **Definition**. "**Option**" shall mean: (a) the right to extend or reduce the term of or renew this Lease or to extend or reduce the term of or renew any lease that Lessee has on other property of Lessor; (b) the right of first refusal or first offer to lease either the Premises or other property of Lessor; (c) the right to purchase, the right of first offer to purchase or the right of first refusal to purchase the Premises or other property of Lessor.
- 39.2 **Options Personal To Original Lessee**. Any Option granted to Lessee in this Lease is personal to the original Lessee, and cannot be assigned or exercised by anyone other than said original Lessee and only while the original Lessee is in full possession of the Premises and, if requested by Lessor, with Lessee certifying that Lessee has no intention of thereafter assigning or subletting.
- 39.3 **Multiple Options**. In the event that Lessee has any multiple Options to extend or renew this Lease, a later Option cannot be exercised unless the prior Options have been validly exercised.

39.4 Effect of Default on Options.

- (a) Lessee shall have no right to exercise an Option: (i) during the period commencing with the giving of any notice of Default and continuing until said Default is cured, (ii) during the period of time any Rent is unpaid (without regard to whether notice thereof is given Lessee), (iii) during the time Lessee is in Breach of this Lease, or (iv) in the event that Lessee has been given 3 or more notices of separate Default, whether or not the Defaults are cured, during the 12 month period immediately preceding the exercise of the Option.
- (b) The period of time within which an Option may be exercised shall not be extended or enlarged by reason of Lessee's inability to exercise an Option because of the provisions of Paragraph 39.4(a).
- (c) An Option shall terminate and be of no further force or effect, notwithstanding Lessee's due and timely exercise of the Option, if, after such exercise and prior to the commencement of the extended term or completion of the purchase, (i) Lessee fails to pay Rent for a period of 30 days after such Rent becomes due (without any necessity of Lessor to give notice thereof), or (ii) if Lessee commits a Breach of this Lease.
- **40. Multiple Buildings.** If the Premises are a part of a group of buildings controlled by Lessor, Lessee agrees that it will abide by and conform to all reasonable rules and regulations which Lessor may make from time to time for the management, safety, and care of said properties, including the care and cleanliness of the grounds and including the parking, loading and unloading of vehicles, and to cause its employees, suppliers, shippers, customers, contractors and invitees to so abide and conform. Lessee also agrees to pay its fair share of common expenses incurred in connection with such rules and regulations.
- **41. Security Measures.** Lessee hereby acknowledges that the Rent payable to Lessor hereunder does not include the cost of guard service or other security measures, and that Lessor shall have no obligation whatsoever to provide same. Lessee assumes all responsibility for the protection of the Premises, Lessee, its agents and invitees and their property from the acts of third parties.
- **42. Reservations.** Lessor reserves to itself the right, from time to time, to grant, without the consent or joinder of Lessee, such easements, rights and dedications that Lessor deems necessary, and to cause the recordation of parcel maps and restrictions, so long as such easements, rights, dedications, maps and restrictions do not unreasonably interfere with the use of the Premises by Lessee. Lessee agrees to sign any documents reasonably requested by Lessor to effectuate any such easement rights, dedication, map or restrictions.
- **43. Performance Under Protest.** If at any time a dispute shall arise as to any amount or sum of money to be paid by one Party to the other under the provisions hereof, the Party against whom the obligation to pay the money is asserted shall have the right to make payment "under protest" and such payment shall not be regarded as a voluntary payment and there shall survive the right on the part of said Party to institute suit for recovery of such sum. If it shall be adjudged that there was no legal obligation on the part of said Party to pay such sum or any part thereof, said Party shall be entitled to recover such sum or so much thereof as it was not legally required to pay. A Party who does not initiate suit for the recovery of sums paid "under protest" within 6 months shall be deemed to have waived its right to protest such payment.

44. Authority; Multiple Parties; Execution.

- (a) If either Party hereto is a corporation, trust, limited liability company, partnership, or similar entity, each individual executing this Lease on behalf of such entity represents and warrants that he or she is duly authorized to execute and deliver this Lease on its behalf. Each Party shall, within 30 days after request, deliver to the other Party satisfactory evidence of such authority.
- (b) If this Lease is executed by more than one person or entity as "Lessee", each such person or entity shall be jointly and severally liable hereunder. It is agreed that any one of the named Lessees shall be empowered to execute any amendment to this Lease, or other document ancillary thereto and bind all of the named Lessees, and Lessor may rely on the same as if all of the named Lessees had executed such document.
- (c) This Lease may be executed by the Parties in counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument.
- **45. Conflict.** Any conflict between the printed provisions of this Lease and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions.
- **46. Offer**. Preparation of this Lease by either Party or their agent and submission of same to the other Party shall not be deemed an offer to lease to the other Party. This Lease is not intended to be binding until executed and delivered by all Parties hereto.
- **47. Amendments.** This Lease may be modified only in writing, signed by the Parties in interest at the time of the modification. As long as they do not materially change Lessee's obligations hereunder, Lessee agrees to make such reasonable non-monetary modifications to this Lease as may be reasonably required by a Lender in connection with the obtaining of normal financing or refinancing of the Premises.
- 48. Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY

OR ARISING OUT OF THIS LEASE.

49. Arbitration of Disputes. An Addendum requiring the Arbitration of all dispute attached to this Lease.	s between the Parties and/or Brokers arising out of this Lease is is not
50. Accessibility; Americans with Disabilities Act.	
(a) The Premises: have not undergone an inspection by a Certified Access Specialist (CASp). Note: A Certified Access Specialist (CASp) can inspect the subject premises and determine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not require a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection of the subject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the arrangements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to correct violations of construction-related accessibility standards within the premises.	
have undergone an inspection by a Certified Access Specialist (CASp) and it wa accessibility standards pursuant to California Civil Code §55.51 et seq. Lessee ackno executing this Lease and agrees to keep such report confidential.	• • • • • • • • • • • • • • • • • • • •
have undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises did not meet all applicable construction-related accessibility standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this Lease and agrees to keep such report confidential except as necessary to complete repairs and corrections of violations of construction related accessibility standards.	
In the event that the Premises have been issued an inspection report by a CASp the within 7 days of the execution of this Lease.	Lessor shall provide a copy of the disability access inspection certificate to Lessee
(b) Since compliance with the Americans with Disabilities Act (ADA) an use of the Premises, Lessor makes no warranty or representation as to whether or no Lessee's use of the Premises requires modifications or additions to the Premises in commake any such necessary modifications and/or additions at Lessee's expense.	
LESSOR AND LESSEE HAVE CAREFULLY READ AND REVIEWED THIS LEASE AND EACH LEASE SHOW THEIR INFORMED AND VOLUNTARY CONSENT THERETO. THE PARTIE THIS LEASE ARE COMMERCIALLY REASONABLE AND EFFECTUATE THE INTENT AND	S HEREBY AGREE THAT, AT THE TIME THIS LEASE IS EXECUTED, THE TERMS OF
ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE O CONSEQUENCES OF THIS LEASE OR THE TRANSACTION TO WHICH IT RELATES. THE 1. SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF TH 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONLIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING AND OPERATING SYSTEMS, AND THE SUITABILITY OF THE PREMISES FOR LESSEE'S	PARTIES ARE URGED TO: IIS LEASE. IDITION OF THE PREMISES. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE OF THE PREMISES, THE STRUCTURAL INTEGRITY, THE CONDITION OF THE ROOF
WARNING: IF THE PREMISES ARE LOCATED IN A STATE OTHER THAN CALIFORNIA, THE LAWS OF THE STATE IN WHICH THE PREMISES ARE LOCATED.	CERTAIN PROVISIONS OF THE LEASE MAY NEED TO BE REVISED TO COMPLY WITH
The parties hereto have executed this Lease at the place and on the dates specified	above their respective signatures.
Executed at: On:	Executed at:On:
By LESSOR:	ByLESSEE:
By:	Ву:
Name Printed: Title:	Name Printed: Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Ву:	Ву:
Name Printed:	Name Printed:
Title:	Title: Phone:
Phone:	Fax:
Email:	Email:
Address:	Address:
Federal ID No.:	Federal ID No.:
BROKER	BROKER
Attn:	Attn:
Title:	Title:
Address:	Address:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Federal ID No.: Broker DRE License #:	Federal ID No.: Broker DRE License #:
Agent DRE License #:	Agent DRE License #:
	242.007.0777 *
AIR CRE * https://www.aircre.com * 2 NOTICE: No part of these works may be reprodu	_
	,
INITIALS	INITIALS

INITIALS
© 2019 AIR CRE. All Rights Reserved.
STG-27.40, Revised 10-22-2020



STANDARD INDUSTRIAL/COMMERCIAL LAND LEASE - GROSS (DO NOT USE THIS FORM FOR BUILDINGS OR FOR LEASES PROVIDING FOR THE CONSTRUCTION OF BUILDINGS)

1. Basic Provisions ("Basic Provisions").		
1.1 Parties. This Lease ("Lease"), dated for reference purposes only	, is made by and between ("Les	ssor") and
("Lessee"), (collectively the "Parties", or individually a "Party").1.2 Premises: That certain real property, including all improvements thereon	or to be provided by Lessor under the terms of thi	s Lease commonly known as
(street address, city, state, zip): ("Premises"). The Premises are located the control of	•	•
the nature of the property and the improvements on the property, if any, and , if ap	plicable, the " Project ", if the property is located with	hin a Project):
(See also Paragraph 2)		
1.3 Term: years and months ("Original Term") ("Expiration Date"). (See also Paragraph 3)	commencing ("Commencement Date	e") and ending
1.4 Early Possession: If the Premises are available Lessee may have non-exc	lusive possession of the Premises commencing	("Early Possession
Date"). (See also Paragraphs 3.2 and 3.3)		(=====
1.5 Base Rent: per month ("Base Rent"), payable on the	day of each month commencing	. (See also Paragraph 4)
If this box is checked, there are provisions in this Lease for the Base Rent	to be adjusted. See Paragraph	
1.6 Base Rent and Other Monies Paid Upon Execution:		
(a) Base Rent: for the period		
(b) Security Deposit: ("Security Deposit"). (See also Pa	ragraph 5)	
(c) Other: for (d) Total Due Upon Execution of this Lease:	7	
1.7 Agreed Use: (See also Paragraph 6)		/ > '
1.8 Insuring Party. Lessor is the "Insuring Party". The annual "Base Premium	n" is (See also Paragraph 8)	V
1.9 Real Estate Brokers. (See also Paragraph 15 and 25)		
(a) Representation: Each Party acknowledges receiving a Disclosure R agency relationships in this Lease with the following real estate brokers ("Broker(s)"		and consents to the following
Lessor's Brokerage Firm License No Is the broke		Losson and Losson (dual agent)
Lessor's Agent License No is (check one): t Agent and the Lessor's Agent (dual agent).	he Lessor's Agent (salesperson or broker associate);	or both the Lessee's
		1
Lessee's Brokerage Firm License No Is the broke agent).	er of (check one): the Lessee; or both the	Lessee and Lessor (dual
Lessee's Agent License No is (check one):		hadada tarah
Agent and the Lessor's Agent (dual agent).	the Lessee's Agent (salesperson or broker associate)	; or both the Lessee's
(b) Payment to Brokers. Upon execution and delivery of this Lease by	both Parties, Lessor shall pay to the Brokers the bro	kerage fee agreed to in a
separate written agreement (or if there is no such agreement, the sum of	or % of the total Base Rent) for the	ne brokerage services rendered
by the Brokers.	("Consumator") (Consider D	
1.10 Guarantor. The obligations of the Lessee under this Lease are to be guar 1.11 Attachments. Attached hereto are the following, all of which constitute		aragrapii 37)
an Addendum consisting of Paragraphs through		
a plot plan depicting the Premises;		
a current set of the Rules and Regulations;		
a Work Letter;		
other (specify):		
2. Premises.		
2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from L		
covenants and conditions set forth in this Lease. While the approximate square foo purposes of comparison, the Base Rent stated herein is NOT tied to square footage	· · · · · · · · · · · · · · · · · ·	_
different. NOTE: Lessee is advised to verify the actual size prior to executing this I		ze se determined to se
2.2 Condition . Lessor shall deliver the Premises to Lessee free of debris on t		
("Start Date"), and warrants that the existing electrical, plumbing, lighting, if any, ar shall be in good operating condition on said date. If a non-compliance with said wa		
malfunction or fail within the warranty period, Lessor shall, as Lessor's sole obligation		
promptly after receipt of written notice from Lessee setting forth with specificity the	· · · · · · · · · · · · · · · · · · ·	
Lessor's expense. The warranty period shall be 30 days. If Lessee does not give Lessuch non-compliance, malfunction or failure shall be the obligation of Lessee at Less		
in writing, Lessor is unaware of (i) any recorded Notices of Default affecting the Prei		
(iii) any bankruptcy proceeding affecting the Premises.		
2.3 Compliance. Lessor warrants that to the best of its knowledge the improcovenants or restrictions of record, regulations, and ordinances ("Applicable Requirements or restrictions of record, regulations, and ordinances ("Applicable Requirements or restrictions of record, regulations, and ordinances or restrictions of record, regulations, and ordinances or restrictions."		
thereof, was constructed. Said warranty does not apply to the use to which Lessee		
Disabilities Act or any similar laws as a result of Lessee's use (see Paragraph 50), or to	to any Alterations or Utility Installations (as defined i	in Paragraph 7.3(a)) made or to
be made by Lessee. NOTE: Lessee is responsible for determining whether or not the Lessee's intended use, and acknowledges that past uses of the Premises may no lo		
except as otherwise provided, promptly after receipt of written notice from Lessee	_	· · · · · · · · · · · · · · · · · · ·
the same at Lessor's expense. If Lessee does not give Lessor written notice of a non		owing the Start Date,
correction of that non-compliance shall be the obligation of Lessee at Lessee's sole of 2.4 Acknowledgements . Lessee acknowledges that: (a) it has been given an		(h) it has been advised by
Lessor and/or Brokers to satisfy itself with respect to the size and condition of the P		
environmental aspects, and compliance with Applicable Requirements and the Ame		
Lessee has made such investigation as it deems necessary with reference to such matter the Bramises. (d) it is not relying on any representation as to the size of the Bramises.		
the Premises, (d) it is not relying on any representation as to the size of the Premise material to Lessee's decision to lease the Premises and pay the Rent stated herein, a		
representations or warranties with respect to said matters other than as set forthin	this Lease. In addition, Lessor acknowledges that: (i) Brokers have made no
representations, promises or warranties concerning Lessee's ability to honor the Lea investigate the financial capability and/or suitability of all proposed tenants.	ise or suitability to occupy the Premises, and (ii) it is	Lessor's sole responsibility to
2.5 Lessee as Prior Owner/Occupant . The warranties made by Lessor in Par	agraph 2 shall be of no force or effect if immediately	prior to the Start Date Lessee
•	,	
INITIALS	INITIALS	
© 2019 AIR CRE. All Rights Reserved.	Last Edi	ited: 9/30/2022 2:33 PM

LG-15.23, Revised 10-22-2020

was the owner or occupant of the Premises. In such event, Lessee shall be responsible for any necessary corrective work.

3. Term

- 3.1 Term. The Commencement Date, Expiration Date and Original Term of this Lease are as specified in Paragraph 1.3.
- 3.2 **Early Possession**. Any provision herein granting Lessee Early Possession of the Premises is subject to and conditioned upon the Premises being available for such possession prior to the Commencement Date. Any grant of Early Possession only conveys a non-exclusive right to occupy the Premises. If Lessee totally or partially occupies the Premises prior to the Commencement Date, the obligation to pay Base Rent shall be abated for the period of such Early Possession. All other terms of this Lease (including but not limited to the obligations to pay Real Property Taxes and insurance premiums and to maintain the Premises) shall be in effect during such period. Any such Early Possession shall not affect the Expiration Date.
- 3.3 **Delay in Possession**. Lessor agrees to use commercially reasonable efforts to deliver exclusive possession of the Premises to Lessee by the Commencement Date. If, despite said efforts, Lessor is unable to deliver possession by such date, Lessor shall not be subject to any liability therefor, nor shall such failure affect the validity of this Lease or change the Expiration Date. Lessee shall not, however, be obligated to pay Rent or perform its other obligations until Lessor delivers possession of the Premises and any period of rent abatement that Lessee would otherwise have enjoyed shall run from the date of delivery of possession and continue for a period equal to what Lessee would otherwise have enjoyed under the terms hereof, but minus any days of delay caused by the acts or omissions of Lessee. If possession is not delivered within 60 days after the Commencement Date, Lessee may, at its option, by notice in writing within 10 days after the end of such 60 day period, cancel this Lease, in which event the Parties shall be discharged from all obligations hereunder. If such written notice is not received by Lessor within said 10 day period, Lessee's right to cancel shall terminate. If possession of the Premises is not delivered within 120 days after the Commencement Date, this Lease shall terminate unless other agreements are reached between Lessor and Lessee, in writing.
- 3.4 Lessee Compliance. Lessor shall not be required to tender possession of the Premises to Lessee until Lessee complies with its obligation to provide evidence of insurance (Paragraph 8.5). Pending delivery of such evidence, Lessee shall be required to perform all of its obligations under this Lease from and after the Start Date, including the payment of Rent, notwithstanding Lessor's election to withhold possession pending receipt of such evidence of insurance. Further, if Lessee is required to perform any other conditions prior to or concurrent with the Start Date, the Start Date shall occur but Lessor may elect to withhold possession until such conditions are satisfied.

Rent

- 4.1 Rent Defined. All monetary obligations of Lessee to Lessor under the terms of this Lease (except for the Security Deposit) are deemed to be rent ("Rent").
- 4.2 **Payment**. Lessee shall cause payment of Rent to be received by Lessor in lawful money of the United States, without offset or deduction (except as specifically permitted in this Lease), on or before the day on which it is due. All monetary amounts shall be rounded to the nearest whole dollar. In the event that any invoice prepared by Lessor is inaccurate such inaccuracy shall not constitute a waiver and Lessee shall be obligated to pay the amount set forth in this Lease. Rent for any period during the term hereof which is for less than one full calendar month shall be prorated based upon the actual number of days of said month. Payment of Rent shall be made to Lessor at its address stated herein or to such other persons or place as Lessor may from time to time designate in writing. Acceptance of a payment which is less than the amount then due shall not be a waiver of Lessor's rights to the balance of such Rent, regardless of Lessor's endorsement of any check so stating. In the event that any check, draft, or other instrument of payment given by Lessee to Lessor is dishonored for any reason, Lessee agrees to pay to Lessor the sum of \$25 in addition to any Late Charge and Lessor, at its option, may require all future payments to be made by Lessee to be by cashier's check. Payments will be applied first to accrued late charges and attorney's fees, second to accrued interest, then to Base Rent and Operating Expense Increase, and any remaining amount to any other outstanding charges or costs.
- 5. Security Deposit. Lessee shall deposit with Lessor upon execution hereof the Security Deposit as security for Lessee's faithful performance of its obligations under this Lease. If Lessee fails to pay Rent, or otherwise Defaults under this Lease, Lessor may use, apply or retain all or any portion of said Security Deposit for the payment of any amount already due Lessor, for Rents which will be due in the future, and/or to reimburse or compensate Lessor for any liability, expense, loss or damage which Lessor may suffer or incur by reason thereof. If Lessor uses or applies all or any portion of the Security Deposit, Lessee shall within 10 days after written request therefor deposit monies with Lessor sufficient to restore said Security Deposit to the full amount required by this Lease. If the Base Rent increases during the term of this Lease, Lessee shall, upon written request from Lessor, deposit additional monies with Lessor so that the total amount of the Security Deposit shall at all times bear the same proportion to the increased Base Rent as the initial Security Deposit bore to the initial Base Rent. Should the Agreed Use be amended to accommodate a material change in the business of Lessee or to accommodate a sublessee or assignee, Lessor shall have the right to increase the Security Deposit to the extent necessary, in Lessor's reasonable judgment, to account for any increased wear and tear that the Premises may suffer as a result thereof. If a change in control of Lessee occurs during this Lease and following such change the financial condition of Lessee is, in Lessor's reasonable judgment, significantly reduced, Lessee shall deposit such additional monies with Lessor as shall be sufficient to cause the Security Deposit to be at a commercially reasonable level based on such change in financial condition. Lessor shall return that portion of the Security Deposit not used or applied by Lessor. Lessor shall upon written request provide Lessee with an accounting showing how that portion of the Security Depos

6. Use.

6.1 **Use**. Lessee shall use and occupy the Premises only for the Agreed Use, or any other legal use which is reasonably comparable thereto, and for no other purpose. Lessee shall not use or permit the use of the Premises in a manner that is unlawful, creates damage, waste or a nuisance, or that disturbs occupants of or causes damage to neighboring premises or properties. Other than guide, signal and seeing eye dogs, Lessee shall not keep or allow in the Premises any pets, animals, birds, fish, or reptiles. Lessor shall not unreasonably withhold or delay its consent to any request for a modification of the Agreed Use, so long as the same will not be significantly more burdensome to the Premises. If Lessor elects to withhold consent, Lessor shall within 7 days after such request give written notification of same, which notice shall include an explanation of Lessor's objections to the change in the Agreed Use.

6.2 Hazardous Substances.

- (a) Reportable Uses Require Consent. The term "Hazardous Substance" as used in this Lease shall mean any product, substance, or waste whose presence, use, manufacture, disposal, transportation, or release, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentially injurious to the public health, safety or welfare, the environment or the Premises, (ii) regulated or monitored by any governmental authority, or (iii) a basis for potential liability of Lessor to any governmental agency or third party under any applicable statute or common law theory. Hazardous Substances shall include, but not be limited to, hydrocarbons, petroleum, gasoline, and/or crude oil or any products, by-products or fractions thereof. Lessee shall not engage in any activity in or on the Premises which constitutes a Reportable Use of Hazardous Substances without the express prior written consent of Lessor and timely compliance (at Lessee's expense) with all Applicable Requirements. "Reportable Use" shall mean (i) the installation or use of any above or below ground storage tank, (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and/or (iii) the presence at the Premises of a Hazardous Substance with respect to which any Applicable Requirements requires that a notice be given to persons entering or occupying the Premises or neighboring properties. Notwithstanding the foregoing, Lessee may use any ordinary and customary materials reasonably required to be used in the normal course of the Agreed Use, ordinary office supplies (copier toner, liquid paper, glue, etc.) and common household cleaning materials, so long as such use is in compliance with all Applicable Requirements, is not a Reportable Use, and does not expose the Premises or neighboring property to any meaningful risk of cont
- (b) **Duty to Inform Lessor**. If Lessee knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises, other than as previously consented to by Lessor, Lessee shall immediately give written notice of such fact to Lessor, and provide Lessor with a copy of any report, notice, claim or other documentation which it has concerning the presence of such Hazardous Substance.
- (c) Lessee Remediation. Lessee shall not cause or permit any Hazardous Substance to be spilled or released in, on, under, or about the Premises (including through the plumbing or sanitary sewer system) and shall promptly, at Lessee's expense, comply with all Applicable Requirements and take all investigatory and/or remedial action reasonably recommended, whether or not formally ordered or required, for the cleanup of any contamination of, and for the maintenance, security and/or monitoring of the Premises or neighboring properties, that was caused or materially contributed to by Lessee, or pertaining to or involving any Hazardous Substance brought onto the Premises during the term of this Lease, by or for Lessee, or any third party.
- (d) Lessee Indemnification. Lessee shall indemnify, defend and hold Lessor, its agents, employees, lenders and ground lessor, if any, harmless from and against any and all loss of rents and/or damages, liabilities, judgments, claims, expenses, penalties, and attorneys' and consultants' fees arising out of or involving any Hazardous Substance brought onto the Premises by or for Lessee, or any third party (provided, however, that Lessee shall have no liability under this Lease with respect to underground migration of any Hazardous Substance under the Premises from adjacent properties not caused or contributed to by Lessee). Lessee's obligations shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Lessee, and

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease. **No termination**, cancellation or release agreement entered into by Lessor and Lessee shall release Lessee from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Lessor in writing at the time of such agreement.

- (e) Lessor Indemnification. Except as otherwise provided in paragraph 8.7, Lessor and its successors and assigns shall indemnify, defend, reimburse and hold Lessee, its employees and lenders, harmless from and against any and all environmental damages, including the cost of remediation, which result from Hazardous Substances which existed on the Premises prior to Lessee's occupancy or which are caused by the gross negligence or willful misconduct of Lessor, its agents or employees. Lessor's obligations, as and when required by the Applicable Requirements, shall include, but not be limited to, the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease.
- (f) Investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by governmental entities having jurisdiction with respect to the existence of Hazardous Substances on the Premises prior to Lessee's occupancy, unless such remediation measure is required as a result of Lessee's use (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises, in which event Lessee shall be responsible for such payment. Lessee shall cooperate fully in any such activities at the request of Lessor, including allowing Lessor and Lessor's agents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remedial responsibilities.
- (g) Lessor Termination Option. If a Hazardous Substance Condition (see Paragraph 9.3(a)) occurs during the term of this Lease, unless Lessee is legally responsible therefor (in which case Lessee shall make the investigation and remediation thereof required by the Applicable Requirements and this Lease shall continue in full force and effect, but subject to Lessor's rights under Paragraph 6.2(d) and Paragraph 13), Lessor may, at Lessor's option, either (i) investigate and remediate such Hazardous Substance Condition, if required, as soon as reasonably possible at Lessor's expense, in which event this Lease shall continue in full force and effect, or (ii) if the estimated cost to remediate such condition exceeds 12 times the then monthly Base Rent or \$100,000, whichever is greater, give written notice to Lessee, within 30 days after receipt by Lessor of knowledge of the occurrence of such Hazardous Substance Condition, of Lessor's desire to terminate this Lease as of the date 60 days following the date of such notice.
- 6.3 Lessee's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lessee's sole expense, fully, diligently and in a timely manner, materially comply with all Applicable Requirements, the requirements of any applicable fire insurance underwriter or rating bureau, and the recommendations of Lessor's engineers and/or consultants which relate in any manner to the Premises, without regard to whether said Applicable Requirements are now in effect or become effective after the Start Date. Lessee shall, within 10 days after receipt of Lessor's written request, provide Lessor with copies of all permits and other documents, and other information evidencing Lessee's compliance with any Applicable Requirements specified by Lessor, and shall immediately upon receipt, notify Lessor in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving the failure of Lessee or the Premises to comply with any Applicable Requirements.
- 6.4 Inspection; Compliance. Lessor and Lessor's "Lender" (as defined in Paragraph 30) and consultants authorized by Lessor shall have the right to enter onto Premises at any time in the case of an emergency, and otherwise at reasonable times after reasonable notice, for the purpose of inspecting and/or testing the condition of the Premises and/or for verifying compliance by Lessee with this Lease. The cost of any such inspections shall be paid by Lessor, unless a violation of Applicable Requirements, or a Hazardous Substance Condition (see Paragraph 9.3(a)) is found to exist or be imminent, or the inspection is requested or ordered by a governmental authority. In such case, Lessee shall upon request reimburse Lessor for the cost of such inspection, so long as such inspection is reasonably related to the violation or contamination. In addition, Lessee shall provide copies of all relevant material safety data sheets (MSDS) to Lessor within 10 days of the receipt of a written request therefor. Lessee acknowledges that any failure on its part to allow such inspections or testing will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to allow such inspections and/or testing in a timely fashion the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for the remainder to the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to allow such inspection and/or testing. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to such failure nor prevent the exercise of any of the other rights and remedies granted hereunder.

7. Maintenance; Repairs; Utility Installations; Trade Fixtures and Alterations.

7.1 Lessee's Obligations.

- (a) In General. Subject to the provisions of Paragraph 2.2 (Condition), 2.3 (Compliance), 6.3 (Lessee's Compliance with Applicable Requirements), 7.2 (Lessor's Obligations), 9 (Damage or Destruction), and 14 (Condemnation), Lessee shall, at Lessee's sole expense, keep the Premises, in good order, condition and repair (whether or not the need for such repairs occurs as a result of Lessee's use, any prior use, the elements or the age of such portion of the Premises), including, but not limited to, plumbing, lighting facilities, landscaping, driveways, parking lots, fences, retaining walls, signs, sidewalks and parkways located on, or adjacent to, the Premises. Lessee, in keeping the Premises in good order, condition and repair, shall exercise and perform good maintenance practices, specifically including the procurement and maintenance of the service contracts required by Paragraph 7.1(b) below. Lessee's obligations shall include restorations, replacements or renewals when necessary to keep the Premises and all improvements thereon or a part thereof in good order, condition and state of repair. Lessee shall, during the term of this Lease, keep the appearance of the Premises in a first-class condition (including, e.g. graffiti removal).
- (b) Service Contracts. Lessee shall, at Lessee's sole expense, procure and maintain a contract, with copies to Lessor, in customary form and substance for, and with a contractor specializing and experienced in the maintenance of the landscaping and irrigation systems, However, Lessor reserves the right, upon notice to Lessee, to procure and maintain such service contract, and Lessee shall reimburse Lessor, upon demand, for the cost thereof.
- (c) Failure to Perform. If Lessee fails to perform Lessee's obligations under this Paragraph 7.1, Lessor may enter upon the Premises after 10 days' prior written notice to Lessee (except in the case of an emergency, in which case no notice shall be required), perform such obligations on Lessee's behalf, and put the Premises in good order, condition and repair, and Lessee shall promptly pay to Lessor a sum equal to 115% of the cost thereof.
- 7.2 **Lessor's Obligations**. Subject to the provisions of Paragraphs 2.2 (Condition), 2.3 (Compliance), 9 (Damage or Destruction) and 14 (Condemnation), it is intended by the Parties hereto that Lessor have no obligation, in any manner whatsoever, to repair and maintain the Premises. It is the intention of the Parties that the terms of this Lease govern the respective obligations of the Parties as to maintenance and repair of the Premises, and they expressly waive the benefit of any statute now or hereafter in effect to the extent it is inconsistent with the terms of this Lease.

7.3 Utility Installations; Trade Fixtures; Alterations.

- (a) **Definitions.** The term "**Utility Installations**" refers to security and fire protection systems, lighting fixtures, plumbing, and fencing on the Premises. The term "**Trade Fixtures**" shall mean Lessee's machinery and equipment that can be removed without doing material damage to the Premises. The term "**Alterations**" shall mean any modification of the improvements, other than Utility Installations or Trade Fixtures, whether by addition or deletion. "**Lessee Owned Alterations and/or Utility Installations**" are defined as Alterations and/or Utility Installations made by Lessee that are not yet owned by Lessor pursuant to Paragraph 7.4(a).
- (b) Consent. Lessee shall not make any Alterations or Utility Installations to the Premises without Lessor's prior written consent. Lessee may, however, make non-structural Alterations or Utility Installations to the Premises without such consent but upon notice to Lessor, as long as the cumulative cost thereof during this Lease as extended does not exceed a sum equal to 3 month's Base Rent in the aggregate or a sum equal to one month's Base Rent in any one year. Lessor may, as a precondition to granting such approval, require Lessee to utilize a contractor chosen and/or approved by Lessor. Any Alterations or Utility Installations that Lessee shall desire to make and which require the consent of the Lessor shall be presented to Lessor in written form with detailed plans. Consent shall be deemed conditioned upon Lessee's: (i) acquiring all applicable governmental permits, (ii) furnishing Lessor with copies of both the permits and the plans and specifications prior to commencement of the work, and (iii) compliance with all conditions of said permits and other Applicable Requirements in a prompt and expeditious manner. Any Alterations or Utility Installations shall be performed in a workmanlike manner with good and sufficient materials. Lessee shall promptly upon completion furnish Lessor with as-built plans and specifications. For work which costs an amount in excess of one month's Base Rent, Lessor may condition its consent upon Lessee providing a lien and completion bond in an amount equal to 150% of the estimated cost of such Alteration or Utility Installation and/or upon Lessee's posting an additional Security Deposit with Lessor.
- (c) Liens; Bonds. Lessee shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lessee at or for use on the Premises, which claims are or may be secured by any mechanic's or materialmen's lien against the Premises or any interest therein. Lessee shall give Lessor not less than 10 days notice prior to the commencement of any work in, on or about the Premises, and Lessor shall have the right to post notices of non-responsibility. If Lessee shall contest the validity of any such lien, claim or demand, then Lessee shall, at its sole expense defend and protect itself, Lessor and the Premises against the same and shall pay and satisfy any such adverse judgment that may be rendered thereon before the enforcement thereof. If Lessor shall require, Lessee shall furnish a surety bond in an amount equal to 150% of the amount of such contested lien, claim or demand, indemnifying Lessor against liability for the same. If Lessor elects to participate in any such action, Lessee shall pay Lessor's attorneys' fees and costs.

7.4 Ownership; Removal; Surrender; and Restoration.

(a) **Ownership.** Subject to Lessor's right to require removal or elect ownership as hereinafter provided, all Alterations and Utility Installations made by Lessee shall be the property of Lessee, but considered a part of the Premises. Lessor may, at any time, elect in writing to be the owner of all or any specified part of the Lessee Owned Alterations and Utility Installations. Unless otherwise instructed per paragraph 7.4(b) hereof, all Lessee Owned Alterations and Utility Installations shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

- (b) **Removal**. By delivery to Lessee of written notice from Lessor not earlier than 90 and not later than 30 days prior to the end of the term of this Lease, Lessor may require that any or all Lessee Owned Alterations or Utility Installations be removed by the expiration or termination of this Lease. Lessor may require the removal at any time of all or any part of any Lessee Owned Alterations or Utility Installations made without the required consent.
- (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements, parts and surfaces thereof broom clean and free of debris, and in good operating order, condition and state of repair, ordinary wear and tear excepted. "Ordinary wear and tear" shall not include any damage or deterioration that would have been prevented by good maintenance practice. Notwithstanding the foregoing and the provisions of Paragraph 7.1(a), if the Lessee occupies the Premises for 12 months or less, then Lessee shall surrender the Premises in the same condition as delivered to Lessee on the Start Date with NO allowance for ordinary wear and tear. Lessee shall repair any damage occasioned by the installation, maintenance or removal of Trade Fixtures, Lessee owned Alterations and/or Utility Installations, furnishings, and equipment as well as the removal of any storage tank installed by or for Lessee. Lessee shall also completely remove from the Premises any and all Hazardous Substances brought onto the Premises by or for Lessee, or any third party (except Hazardous Substances which were deposited via underground migration from areas outside of the Premises) to the level specified in Applicable Requirements. Trade Fixtures shall remain the property of Lessee and shall be removed by Lessee. Any personal property of Lessee not removed on or before the Expiration Date or any earlier termination date shall be deemed to have been abandoned by Lessee and may be disposed of or retained by Lessor as Lessor may desire. The failure by Lessee to timely vacate the Premises pursuant to this Paragraph 7.4(c) without the express written consent of Lessor shall constitute a holdover under the provisions of Paragraph 26 below.

8. Insurance; Indemnity.

8.1 Payment of Premium Increases.

- (a) Lessee shall pay to Lessor any insurance cost increase ("Insurance Cost Increase") occurring during the term of this Lease. Insurance Cost Increase is defined as any increase in the actual cost of the insurance required under Paragraph 8.2(b), 8.3(a) and 8.3(b), over and above the Base Premium as hereinafter defined calculated on an annual basis. Insurance Cost Increase shall include but not be limited to increases resulting from the nature of Lessee's occupancy, any act or omission of Lessee, requirements of the holder of mortgage or deed of trust covering the Premises, increased valuation of the Premises and/or a premium rate increase. The parties are encouraged to fill in the Base Premium in paragraph 1.8 with a reasonable premium for such insurance based on the Agreed Use of the Premises. If the parties fail to insert a dollar amount in Paragraph 1.8, then the Base Premium shall be the lowest annual premium reasonably obtainable for the required insurance as of the commencement of the Original Term for the Agreed Use of the Premises. In no event, however, shall Lessee be responsible for any portion of the increase in the premium cost attributable to liability insurance carried by Lessor under Paragraph 8.2(b) in excess of \$2,000,000 per occurrence.
- (b) Lessee shall pay any such Insurance Cost Increase to Lessor within 30 days after receipt by Lessee of a copy of the premium statement or other reasonable evidence of the amount due. If the insurance policies maintained hereunder cover other property besides the Premises, Lessor shall also deliver to Lessee a statement of the amount of such Insurance Cost Increase attributable only to the Premises showing in reasonable detail the manner in which such amount was computed. Premiums for policy periods commencing prior to, or extending beyond the term of this Lease, shall be prorated to correspond to the term of this Lease.

8.2 Liability Insurance.

- (a) Carried by Lessee. Lessee shall obtain and keep in force a Commercial General Liability policy of insurance protecting Lessee and Lessor as an additional insured against claims for bodily injury, personal injury and property damage based upon or arising out of the ownership, use, occupancy or maintenance of the Premises and all areas appurtenant thereto. Such insurance shall be on an occurrence basis providing single limit coverage in an amount not less than \$1,000,000 per occurrence with an annual aggregate of not less than \$2,000,000. Lessee shall add Lessor as an additional insured by means of an endorsement at least as broad as the Insurance Service Organization's "Additional Insured-Managers or Lessors of Premises" Endorsement. The policy shall not contain any intra-insured exclusions as between insured persons or organizations, but shall include coverage for liability assumed under this Lease as an "insured contract" for the performance of Lessee's indemnity obligations under this Lease. The limits of said insurance shall not, however, limit the liability of Lessee nor relieve Lessee of any obligation hereunder. Lessee shall provide an endorsement on its liability policy(ies) which provides that its insurance shall be primary to and not contributory with any similar insurance carried by Lessor, whose insurance shall be considered excess insurance only.
- (b) **Carried by Lessor**. Lessor shall maintain liability insurance as described in Paragraph 8.2(a), in addition to, and not in lieu of, the insurance required to be maintained by Lessee. Lessee shall not be named as an additional insured therein.

8.3 Property Insurance - Improvements and Rental Value.

- (a) Improvements. The Insuring Party shall obtain and keep in force a policy or policies in the name of Lessor, with loss payable to Lessor, any ground-lessor, and to any Lender insuring loss or damage to the Premises. The amount of such insurance shall be equal to the full insurable replacement cost of the Premises, as the same shall exist from time to time, or the amount required by any Lender, but in no event more than the commercially reasonable and available insurable value thereof. Lessee Owned Alterations and Utility Installations, Trade Fixtures, and Lessee's personal property shall be insured by Lessee and not by Lessor. If the coverage is available and commercially appropriate, such policy or policies shall insure against all risks of direct physical loss or damage (except the perils of flood and/or earthquake unless required by a Lender or included in the Base Premium), including coverage for debris removal and the enforcement of any Applicable Requirements requiring the upgrading, demolition, reconstruction or replacement of any portion of the Premises as the result of a covered loss. Said policy or policies shall also contain an agreed valuation provision in lieu of any coinsurance clause, waiver of subrogation, and inflation guard protection causing an increase in the annual property insurance coverage amount by a factor of not less than the adjusted U.S. Department of Labor Consumer Price Index for All Urban Consumers for the city nearest to where the Premises are located. If such insurance coverage has a deductible clause, the deductible amount shall not exceed \$5,000 per occurrence, and Lessee shall be liable for such deductible amount in the event of an Insured Loss.
- (b) **Rental Value**. The Insuring Party shall obtain and keep in force a policy or policies in the name of Lessor with loss payable to Lessor and any Lender, insuring the loss of the full Rent for one year with an extended period of indemnity for an additional 180 days ("Rental Value insurance"). Said insurance shall contain an agreed valuation provision in lieu of any coinsurance clause, and the amount of coverage shall be adjusted annually to reflect the projected Rent otherwise payable by Lessee, for the next 12 month period. Lessee shall be liable for any deductible amount in the event of such loss.
- by Lessee, for the next 12 month period. Lessee shall be liable for any deductible amount in the event of such loss.

 (c) Adjacent Premises. If the Premises are part of a larger property owned by Lessor, the Lessee shall pay for any increase in the premiums for the property insurance of such adjacent property if said increase is caused by Lessee's acts, omissions, use or occupancy of the Premises.

8.4 Lessee's Property; Coverage; Worker's Compensation Insurance.

- (a) **Property Damage**. Lessee shall obtain and maintain insurance coverage on all of Lessee's personal property, Trade Fixtures, and Lessee Owned Alterations and Utility Installations. Such insurance shall be full replacement cost coverage with a deductible of not to exceed \$1,000 per occurrence. The proceeds from any such insurance shall be used by Lessee for the replacement of personal property, Trade Fixtures and Lessee Owned Alterations and Utility Installations.
- (b) Worker's Compensation Insurance. Lessee shall obtain and maintain Worker's Compensation Insurance in such amount as may be required by Applicable Requirements. Such policy shall include a "Waiver of Subrogation" endorsement. Lessee shall provide Lessor with a copy of such endorsement along with the certificate of insurance or copy of the policy required by paragraph 8.5.
- (c) No Representation of Adequate Coverage. Lessor makes no representation that the limits or forms of coverage of insurance specified herein are adequate to cover Lessee's property, business operations or obligations under this Lease.
- 8.5 **Insurance Policies**. Insurance required herein shall be by companies maintaining during the policy term a "General Policyholders Rating" of at least A-, VII, as set forth in the most current issue of "Best's Insurance Guide", or such other rating as may be required by a Lender. Lessee shall not do or permit to be done anything which invalidates the required insurance policies. Lessee shall, prior to the Start Date, deliver to Lessor certified copies of policies of such insurance or certificates with copies of the required endorsements evidencing the existence and amounts of the required insurance. No such policy shall be cancelable or subject to modification except after 30 days prior written notice to Lessor. Lessee shall, at least 10 days prior to the expiration of such policies, furnish Lessor with evidence of renewals or "insurance binders" evidencing renewal thereof, or Lessor may increase his liability insurance coverage and charge the cost thereof to Lessee, which amount shall be payable by Lessee to Lessor upon demand. Such policies shall be for a term of at least one year, or the length of the remaining term of this Lease, whichever is less. If either Party shall fail to procure and maintain the insurance required to be carried by it, the other Party may, but shall not be required to, procure and maintain the same
- 8.6 Waiver of Subrogation. Without affecting any other rights or remedies, Lessee and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is not limited by the amount of insurance carried or required, or by any deductibles applicable hereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Lessee, as the case may be, so long as the insurance is not invalidated thereby.
- 8.7 Indemnity. Except for Lessor's gross negligence or willful misconduct, Lessee shall indemnify, protect, defend and hold harmless the Premises, Lessor and its agents, Lessor's master or ground lessor, partners and Lenders, from and against any and all claims, loss of rents and/or damages, liens, judgments, penalties, attorneys' and consultants' fees, expenses and/or liabilities arising out of, involving, or in connection with, a Breach of the Lease by Lessee and/or the use and/or occupancy of the Premises and/or Project by Lessee and/or by Lessee's employees, contractors or invitees. If any action or proceeding is brought against Lessor by reason of any of the foregoing matters, Lessee shall upon notice defend the same at Lessee's expense by counsel reasonably satisfactory to Lessor and Lessor shall cooperate with Lessee in such defense. Lessor need not have first paid any such claim in order to be defended or indemnified.
 - 8.8 **Exemption of Lessor and its Agents from Liability**. Notwithstanding the negligence or breach of this Lease by Lessor or its agents, neither Lessor nor its

INITIALS

agents shall be liable under any circumstances for: (i) injury or damage to the person or goods, wares, merchandise or other property of Lessee, Lessee's employees, contractors, invitees, customers, or any other person in or about the Premises, from any cause, (ii) any damages arising from any act or neglect of any other tenant of Lessor or from the failure of Lessor or its agents to enforce the provisions of any other lease in the Project, or (iii) injury to Lessee's business or for any loss of income or profit therefrom. Instead, it is intended that Lessee's sole recourse in the event of such damages or injury be to file a claim on the insurance policy(ies) that Lessee is required to maintain pursuant to the provisions of paragraph 8.

8.9 Failure to Provide Insurance. Lessee acknowledges that any failure on its part to obtain or maintain the insurance required herein will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, for any month or portion thereof that Lessee does not maintain the required insurance and/or does not provide Lessor with the required binders or certificates evidencing the existence of the required insurance, the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater. The parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to maintain the required insurance. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to maintain such insurance, prevent the exercise of any of the other rights and remedies granted hereunder, nor relieve Lessee of its obligation to maintain the insurance specified in this Lease.

Damage or Destruction.

- 9.1 Insured Loss. If the improvements on the Premises are damaged and such damage is an Insured Loss, then Lessor shall, at Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonably possible and this Lease shall continue in full force and effect; provided, however, that Lessee shall, at Lessor's election, make the repair of any damage or destruction the total cost to repair of which is \$10,000 or less. and, in such event, Lessor shall make any applicable insurance proceeds available to Lessee on a reasonable basis for that purpose. Notwithstanding the foregoing, if the required insurance was not in force or the insurance proceeds are not sufficient to effect such repair, the Insuring Party shall promptly contribute the shortage in proceeds (except as to the deductible which is Lessee's responsibility) as and when required to complete said repairs. In the event, however, such shortage was due to the fact that, by reason of the unique nature of the improvements, full replacement cost insurance coverage was not commercially reasonable and available, Lessor shall have no obligation to pay for the shortage in insurance proceeds or to fully restore the unique aspects of the Premises.
- 9.2 Uninsured Loss. If the improvements on the Premises are damaged and such damage is not an Insured Loss, unless caused by a negligent or willful act of Lessee (in which event Lessee shall make the repairs at Lessee's expense), Lessor may either: (i) repair such damage as soon as reasonably possible at Lessor's opense, in which event this Lease shall continue in full force and effect, or (ii) terminate this Lease by giving written notice to Lessee within 30 days after receipt by Lessor of knowledge of the occurrence of such damage. Such termination shall be effective 60 days following the date of such notice.

Abatement of Rent; Lessee's Remedies.

- (a) Abatement. In the event the improvements on the Premises are damaged or a Hazardous Substance Condition for which Lessee is not responsible under this Lease occurs ('Hazardous Substance Condition' shall mean the occurrence or discovery of a condition involving the presence of, or a contamination by, a Hazardous Substance on or under the Premises which requires remediation), the Rent payable by Lessee for the period required for the repair, remediation or restoration of such damage shall be abated in proportion to the degree to which Lessee's use of the Premises is impaired, but not to exceed the proceeds received from the Rental Value insurance. All other obligations of Lessee hereunder shall be performed by Lessee, and Lessor shall have no liability for any such damage, destruction, remediation, repair or restoration except as provided herein.
- Remedies. If Lessor is obligated to repair or restore the Premises and does not commence, in a substantial and meaningful way, such repair or restoration within 90 days after such obligation shall accrue, Lessee may, at any time prior to the commencement of such repair or restoration, give written notice to Lessor and to any Lenders of which Lessee has actual notice, of Lessee's election to terminate this Lease on a date not less than 60 days following the giving of such notice. If Lessee gives such notice and such repair or restoration is not commenced within 30 days thereafter, this Lease shall terminate as of the date specified in said notice. If the repair or restoration is commenced within such 30 days, this Lease shall continue in full force and effect. "Commence" shall mean either the unconditional authorization of the preparation of the required plans, or the beginning of the actual work on the Premises, whichever first occurs.
- 9.4 **Termination; Advance Payments**. Upon termination of this Lease pursuant to Paragraph 6.2(g) or Paragraph 9, an equitable adjustment shall be made concerning advance Base Rent and any other advance payments made by Lessee to Lessor. Lessor shall, in addition, return to Lessee so much of Lessee's Security Deposit as has not been, or is not then required to be, used by Lessor.
- 9.5 Waive Statutes. Lessor and Lessee agree that the terms of this Lease shall govern the effect of any damage to or destruction of the Premises with respect to the termination of this Lease and hereby waive the provisions of any present or future statute to the extent inconsistent herewith.

- 10.1 **Definition**. As used herein, the term "Real Property Taxes" shall include any form of assessment; real estate, general, special, ordinary or extraordinary, or rental levy or tax (other than inheritance, personal income or estate taxes); improvement bond; and/or license fee imposed upon or levied against any legal or equitable interest of Lessor in the Premises or the Project, Lessor's right to other income therefrom, and/or Lessor's business of leasing, by any authority having the direct or indirect power to tax and where the funds are generated with reference to the Building address. Real Property Taxes shall also include any tax, fee, levy, assessment or charge, or any increase therein: (i) imposed by reason of events occurring during the term of this Lease, including but not limited to, a change in the ownership of the Premises, and (ii) levied or assessed on machinery or equipment provided by Lessor to Lessee pursuant to this Lease.
- 10.2 Payment of Taxes. Lessor shall pay the Real Property Taxes applicable to the Premises provided, however, that Lessee shall pay to Lessor the amount, if any, by which Real Property Taxes applicable to the Premises increase over the fiscal tax year during which the Commencement Date Occurs ("Tax Increase"). Payment of any such Tax Increase shall be made by Lessee to Lessor within 30 days after receipt of Lessor's written statement setting forth the amount due and computation thereof. If any such taxes shall cover any period of time prior to or after the expiration or termination of this Lease, Lessee's share of such taxes shall be prorated to cover only that portion of the tax bill applicable to the period that this Lease is in effect. In the event lessee incurs a late charge on any Rent payment, Lessor may estimate the current Real Property Taxes, and require that the Tax Increase be paid in advance to Lessor by Lessee monthly in advance with the payment of the Base Rent. Such monthly payment shall be an amount equal to the amount of the estimated installment of the Tax Increase divided by the number of months remaining before the month in which said installment becomes delinquent. When the actual amount of the applicable Tax Increase is known, the amount of such equal monthly advance payments shall be adjusted as required to provide the funds needed to pay the applicable Tax Increase. If the amount collected by Lessor is insufficient to pay the Tax Increase when due, Lessee shall pay Lessor, upon demand, such additional sums as are necessary to pay such obligations. Advance payments may be intermingled with other moneys of Lessor and shall not bear interest. In the event of a Breach by Lessee in the performance of its obligations under this Lease, then any such advance payments may be treated by Lessor as an additional Security Deposit.
- 10.3 Additional Improvements. Notwithstanding anything to the contrary in this Paragraph 10.2, Lessee shall pay to Lessor upon demand therefor the entirety of any increase in Real Property Taxes assessed by reason of Alterations or Utility Installations placed upon the Premises by Lessee or at Lessee's request or by reason of any alterations or improvements to the Premises made by Lessor subsequent to the execution of this Lease by the Parties.
- 10.4 Joint Assessment. If the Premises are not separately assessed, Lessee's liability shall be an equitable proportion of the Tax Increase for all of the land and improvements included within the tax parcel assessed, such proportion to be conclusively determined by Lessor from the respective valuations assigned in the assessor's work sheets or such other information as may be reasonably available.
- 10.5 Personal Property Taxes. Lessee shall pay, prior to delinquency, all taxes assessed against and levied upon Lessee Owned Alterations, Utility Installations, Trade Fixtures, furnishings, equipment and all personal property of Lessee. When possible, Lessee shall cause its Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all other personal property to be assessed and billed separately from the real property of Lessor. If any of Lessee's said property shall be assessed with Lessor's real property, Lessee shall pay Lessor the taxes attributable to Lessee's property within 10 days after receipt of a written statement setting forth the taxes applicable to Lessee's property.

Utilities and Services.

- 11.1 Lessee shall pay for all water, gas, heat, light, power, telephone, trash disposal and other utilities and services supplied to the Premises, together with any taxes thereon. If any such services are not separately metered or billed to Lessee, Lessee shall pay a reasonable proportion, to be determined by Lessor, of all charges jointly metered or billed. There shall be no abatement of rent and Lessor shall not be liable in any respect whatsoever for the inadequacy, stoppage, interruption or discontinuance of any utility or service due to riot, strike, labor dispute, breakdown, accident, repair or other cause beyond Lessor's reasonable control or in cooperation with governmental request or directions.
- 11.2 Within fifteen days of Lessor's written request, Lessee agrees to deliver to Lessor such information, documents and/or authorization as Lessor needs in order for Lessor to comply with new or existing Applicable Requirements relating to commercial building energy usage, ratings, and/or the reporting thereof.

Assignment and Subletting.

12.1 Lessor's Consent Required.

- (a) Lessee shall not voluntarily or by operation of law assign, transfer, mortgage or encumber (collectively, "assign or assignment") or sublet all or any part of Lessee's interest in this Lease or in the Premises without Lessor's prior written consent.
 - (b) Unless Lessee is a corporation and its stock is publicly traded on a national stock exchange, a change in the control of Lessee shall constitute an

INITIALS

Last Edited: 9/30/2022 2:33 PM

INITIALS

assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of the voting control of Lessee shall constitute a change in control for this purpose.

- (c) The involvement of Lessee or its assets in any transaction, or series of transactions (by way of merger, sale, acquisition, financing, transfer, leveraged buy-out or otherwise), whether or not a formal assignment or hypothecation of this Lease or Lessee's assets occurs, which results or will result in a reduction of the Net Worth of Lessee by an amount greater than 25% of such Net Worth as it was represented at the time of the execution of this Lease or at the time of the most recent assignment to which Lessor has consented, or as it exists immediately prior to said transaction or transactions constituting such reduction, whichever was or is greater, shall be considered an assignment of this Lease to which Lessor may withhold its consent. "Net Worth of Lessee" shall mean the net worth of Lessee (excluding any guarantors) established under generally accepted accounting principles.
- (d) An assignment or subletting without consent shall, at Lessor's option, be a Default curable after notice per Paragraph 13.1(d), or a non-curable Breach without the necessity of any notice and grace period. If Lessor elects to treat such unapproved assignment or subletting as a non-curable Breach, Lessor may either: (i) terminate this Lease, or (ii) upon 30 days written notice, increase the monthly Base Rent to 110% of the Base Rent then in effect. Further, in the event of such Breach and rental adjustment, (i) the purchase price of any option to purchase the Premises held by Lessee shall be subject to similar adjustment to 110% of the price previously in effect, and (ii) all fixed and non-fixed rental adjustments scheduled during the remainder of the Lease term shall be increased to 110% of the scheduled adjusted rent.
 - Lessee's remedy for any breach of Paragraph 12.1 by Lessor shall be limited to compensatory damages and/or injunctive relief. (e)
 - Lessor may reasonably withhold consent to a proposed assignment or subletting if Lessee is in Default at the time consent is requested.
- Notwithstanding the foregoing, allowing a de minimis portion of the Premises, ie. 20 square feet or less, to be used by a third party vendor in connection with the installation of a vending machine or pay phone shall not constitute a subletting.

12.2 Terms and Conditions Applicable to Assignment and Subletting.

- (a) Regardless of Lessor's consent, no assignment or subletting shall: (i) be effective without the express written assumption by such assignee or sublessee of the obligations of Lessee under this Lease, (ii) release Lessee of any obligations hereunder, or (iii) alter the primary liability of Lessee for the payment of Rent or for the performance of any other obligations to be performed by Lessee.
- (b) Lessor may accept Rent or performance of Lessee's obligations from any person other than Lessee pending approval or disapproval of an assignment. Neither a delay in the approval or disapproval of such assignment nor the acceptance of Rent or performance shall constitute a waiver or estoppel of Lessor's right to exercise its remedies for Lessee's Default or Breach.
 - Lessor's consent to any assignment or subletting shall not constitute a consent to any subsequent assignment or subletting.
- (d) In the event of any Default or Breach by Lessee, Lessor may proceed directly against Lessee, any Guarantors or anyone else responsible for the performance of Lessee's obligations under this Lease, including any assignee or sublessee, without first exhausting Lessor's remedies against any other person or entity responsible therefor to Lessor, or any security held by Lessor.
- (e) Each request for consent to an assignment or subletting shall be in writing, accompanied by information relevant to Lessor's determination as to the financial and operational responsibility and appropriateness of the proposed assignee or sublessee, including but not limited to the intended use and/or required modification of the Premises, if any, together with a fee of \$500 as consideration for Lessor's considering and processing said request. Lessee agrees to provide Lessor with such other or additional information and/or documentation as may be reasonably requested. (See also Paragraph 36)
- (f) Any assignee of, or sublessee under, this Lease shall, by reason of accepting such assignment, entering into such sublease, or entering into possession of the Premises or any portion thereof, be deemed to have assumed and agreed to conform and comply with each and every term, covenant, condition and obligation herein to be observed or performed by Lessee during the term of said assignment or sublease, other than such obligations as are contrary to or inconsistent with provisions of an assignment or sublease to which Lessor has specifically consented to in writing.
- (g) Lessor's consent to any assignment or subletting shall not transfer to the assignee or sublessee any Option granted to the original Lessee by this Lease unless such transfer is specifically consented to by Lessor in writing. (See Paragraph 39.2)
- 12.3 Additional Terms and Conditions Applicable to Subletting. The following terms and conditions shall apply to any subletting by Lessee of all or any part of $the \ Premises \ and \ shall \ be \ deemed \ included \ in \ all \ subleases \ under \ this \ Lease \ whether \ or \ not \ expressly \ incorporated \ therein:$
- (a) Lessee hereby assigns and transfers to Lessor all of Lessee's interest in all Rent payable on any sublease, and Lessor may collect such Rent and apply same toward Lessee's obligations under this Lease; provided, however, that until a Breach shall occur in the performance of Lessee's obligations, Lessee may collect said Rent. In the event that the amount collected by Lessor exceeds Lessee's then outstanding obligations any such excess shall be refunded to Lessee. Lessor shall not, by reason of the foregoing or any assignment of such sublease, nor by reason of the collection of Rent, be deemed liable to the sublessee for any failure of Lessee to perform and comply with any of Lessee's obligations to such sublessee. Lessee hereby irrevocably authorizes and directs any such sublessee, upon receipt of a written notice from Lessor stating that a Breach exists in the performance of Lessee's obligations under this Lease, to pay to Lessor all Rent due and to become due under the sublease. Sublessee shall rely upon any such notice from Lessor and shall pay all Rents to Lessor without any obligation or right to inquire as to whether such Breach exists, notwithstanding any claim from Lessee to the contrary.
- (b) In the event of a Breach by Lessee, Lessor may, at its option, require sublessee to attorn to Lessor, in which event Lessor shall undertake the obligations of the sublessor under such sublease from the time of the exercise of said option to the expiration of such sublease; provided, however, Lessor shall not be liable for any prepaid rents or security deposit paid by such sublessee to such sublessor or for any prior Defaults or Breaches of such sublessor.

 (c) Any matter requiring the consent of the sublessor under a sublease shall also require the consent of Lessor.

 (d) No sublessee shall further assign or sublet all or any part of the Premises without Lessor's prior written consent.
- Lessor shall deliver a copy of any notice of Default or Breach by Lessee to the sublessee, who shall have the right to cure the Default of Lessee within (e) the grace period, if any, specified in such notice. The sublessee shall have a right of reimbursement and offset from and against Lessee for any such Defaults cured by

13. Default; Breach; Remedies.

- 13.1 Default; Breach. A "Default" is defined as a failure by the Lessee to comply with or perform any of the terms, covenants, conditions or Rules and Regulations under this Lease. A "Breagh" is defined as the occurrence of one or more of the following Defaults, and the failure of Lessee to cure such Default within any applicable grace period:
- (a) The abandonment of the Premises; the vacating of the Premises prior to the expiration or termination of this Lease without providing a commercially reasonable level of security, or where the coverage of the property insurance described in Paragraph 8.3 is jeopardized as a result thereof, or without providing reasonable assurances to minimize potential vandalism; or failure to deliver to Lessor exclusive possession of the entire Premises in accordance herewith prior to the expiration or termination of this Lease.
- (b) The failure of Lessee to make any payment of Rent or any Security Deposit required to be made by Lessee hereunder, whether to Lessor or to a third party, when due, to provide reasonable evidence of insurance or surety bond, or to fulfill any obligation under this Lease which endangers or threatens life or property, where such failure continues for a period of 3 business days following written notice to Lessee. THE ACCEPTANCE BY LESSOR OF A PARTIAL PAYMENT OF RENT OR SECURITY DEPOSIT SHALL NOT CONSTITUTE A WAIVER OF ANY OF LESSOR'S RIGHTS, INCLUDING LESSOR'S RIGHT TO RECOVER POSSESSION OF THE **PREMISES**
- The failure of Lessee to allow Lessor and/or its agents access to the Premises or the commission of waste, act or acts constituting public or private nuisance, and/or an illegal activity on the Premises by Lessee, where such actions continue for a period of 3 business days following written notice to Lessee. In the event that Lessee commits waste, a nuisance or an illegal activity a second time then, the Lessor may elect to treat such conduct as a non-curable Breach rather than a
- (d) The failure by Lessee to provide (i) reasonable written evidence of compliance with Applicable Requirements, (ii) the service contracts, (iii) the rescission of an unauthorized assignment or subletting, (iv) an Estoppel Certificate or financial statements, (v) a requested subordination, (vi) evidence concerning any guaranty and/or Guarantor, (vii) any document requested under Paragraph 42, (viii) material safety data sheets (MSDS), or (ix) any other documentation or information which Lessor may reasonably require of Lessee under the terms of this Lease, where any such failure continues for a period of 10 days following written
- (e) A Default by Lessee as to the terms, covenants, conditions or provisions of this Lease, or of the rules adopted under Paragraph 40 hereof, other than those described in subparagraphs 13.1(a), (b), (c) or (d), above, where such Default continues for a period of 30 days after written notice; provided, however, that if the nature of Lessee's Default is such that more than 30 days are reasonably required for its cure, then it shall not be deemed to be a Breach if Lessee commences such cure within said 30 day period and thereafter diligently prosecutes such cure to completion.
- The occurrence of any of the following events: (i) the making of any general arrangement or assignment for the benefit of creditors; (ii) becoming a "debtor" as defined in 11 U.S.C. § 101 or any successor statute thereto (unless, in the case of a petition filed against Lessee, the same is dismissed within 60 days); (iii) the appointment of a trustee or receiver to take possession of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where possession is not restored to Lessee within 30 days; or (iv) the attachment, execution or other judicial seizure of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where such seizure is not discharged within 30 days; provided, however, in the event that any provision of this subparagraph is contrary to any applicable law, such provision shall be of no force or effect, and not affect the validity of the remaining provisions.
 - (g) The discovery that any financial statement of Lessee or of any Guarantor given to Lessor was materially false.

- (h) If the performance of Lessee's obligations under this Lease is guaranteed: (i) the death of a Guarantor, (ii) the termination of a Guarantor's liability with respect to this Lease other than in accordance with the terms of such guaranty, (iii) a Guarantor's becoming insolvent or the subject of a bankruptcy filing, (iv) a Guarantor's refusal to honor the guaranty, or (v) a Guarantor's breach of its guaranty obligation on an anticipatory basis, and Lessee's failure, within 60 days following written notice of any such event, to provide written alternative assurance or security, which, when coupled with the then existing resources of Lessee, equals or exceeds the combined financial resources of Lessee and the Guarantors that existed at the time of execution of this Lease.
- 13.2 **Remedies**. If Lessee fails to perform any of its affirmative duties or obligations, within 10 days after written notice (or in case of an emergency, without notice), Lessor may, at its option, perform such duty or obligation on Lessee's behalf, including but not limited to the obtaining of reasonably required bonds, insurance policies, or governmental licenses, permits or approvals. Lessee shall pay to Lessor an amount equal to 115% of the costs and expenses incurred by Lessor in such performance upon receipt of an invoice therefor. In the event of a Breach, Lessor may, with or without further notice or demand, and without limiting Lessor in the exercise of any right or remedy which Lessor may have by reason of such Breach:
- (a) Terminate Lessee's right to possession of the Premises by any lawful means, in which case this Lease shall terminate and Lessee shall immediately surrender possession to Lessor. In such event Lessor shall be entitled to recover from Lessee: (i) the unpaid Rent which had been earned at the time of termination; (ii) the worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that the Lessee proves could have been reasonably avoided; (iii) the worth at the time of award of the amount by which the unpaid rent for the balance of the term after the time of award exceeds the amount of such rental loss that the Lessee proves could be reasonably avoided; and (iv) any other amount necessary to compensate Lessor for all the detriment proximately caused by the Lessee's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom, including but not limited to the cost of recovering possession of the Premises, expenses of reletting, including necessary renovation and alteration of the Premises, reasonable attorneys' fees, and that portion of any leasing commission paid by Lessor in connection with this Lease applicable to the unexpired term of this Lease. The worth at the time of award of the amount referred to in provision (iii) of the immediately preceding sentence shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of the District within which the Premises are located at the time of award plus one percent. Efforts by Lessor to mitigate damages caused by Lessee's Breach of this Lease shall not waive Lessor's right to recover any damages to which Lessor is otherwise entitled. If termination of this Lease is obtained through the provisional remedy of unlawful detainer, Lessor shall have the right to recover in such proceeding any unpaid Rent and damages as are recoverable therein, or Lessor may reserve the right to recover all or any part thereof in a separate suit. If a notice and grace period required under Paragraph 13.1 was not previously given, a notice to pay rent or quit, or to perform or quit given to Lessee under the unlawful detainer statute shall also constitute the notice required by Paragraph 13.1. In such case, the applicable grace period required by Paragraph 13.1 and the unlawful detainer statute shall run concurrently, and the failure of Lessee to cure the Default within the greater of the two such grace periods shall constitute both an unlawful detainer and a Breach of this Lease entitling Lessor to the remedies provided for in this Lease and/or by said statute.
- (b) Continue the Lease and Lessee's right to possession and recover the Rent as it becomes due, in which event Lessee may sublet or assign, subject only to reasonable limitations. Acts of maintenance, efforts to relet, and/or the appointment of a receiver to protect the Lessor's interests, shall not constitute a termination of the Lessee's right to possession.
- (c) Pursue any other remedy now or hereafter available under the laws or judicial decisions of the state wherein the Premises are located. The expiration or termination of this Lease and/or the termination of Lessee's right to possession shall not relieve Lessee from liability under any indemnity provisions of this Lease as to matters occurring or accruing during the term hereof or by reason of Lessee's occupancy of the Premises.
- 13.3 Inducement Recapture. Any agreement for free or abated rent or other charges, the cost of tenant improvements for Lessee paid for or performed by Lessor, or for the giving or paying by Lessor to or for Lessee of any cash or other bonus, inducement or consideration for Lessee's entering into this Lease, all of which concessions are hereinafter referred to as "Inducement Provisions," shall be deemed conditioned upon Lessee's full and faithful performance of all of the terms, covenants and conditions of this Lease. Upon Breach of this Lease by Lessee, any such Inducement Provision shall automatically be deemed deleted from this Lease and of no further force or effect, and any rent, other charge, bonus, inducement or consideration theretofore abated, given or paid by Lessor under such an Inducement Provision shall be immediately due and payable by Lessee to Lessor, notwithstanding any subsequent cure of said Breach by Lessee. The acceptance by Lessor of rent or the cure of the Breach which initiated the operation of this paragraph shall not be deemed a waiver by Lessor of the provisions of this paragraph unless specifically so stated in writing by Lessor at the time of such acceptance.
- 13.4 Late Charges. Lessee hereby acknowledges that late payment by Lessee of Rent will cause Lessor to incur costs not contemplated by this Lease, the exact amount of which will be extremely difficult to ascertain. Such costs include, but are not limited to, processing and accounting charges, and late charges which may be imposed upon Lessor by any Lender. Accordingly, if any Rent shall not be received by Lessor within 5 days after such amount shall be due, then, without any requirement for notice to Lessee, Lessee shall immediately pay to Lessor a one-time late charge equal to 10% of each such overdue amount or \$100, whichever is greater. The Parties hereby agree that such late charge represents a fair and reasonable estimate of the costs Lessor will incur by reason of such late payment. Acceptance of such late charge by Lessor shall in no event constitute a waiver of Lessee's Default or Breach with respect to such overdue amount, nor prevent the exercise of any of the other rights and remedies granted hereunder. In the event that a late charge is payable hereunder, whether or not collected, for 3 consecutive installments of Base Rent, then notwithstanding any provision of this Lease to the contrary, Base Rent shall, at Lessor's option, become due and payable quarterly in advance.
- 13.5 Interest. Any monetary payment due Lessor hereunder, other than late charges, not received by Lessor, when due shall bear interest from the 31st day after it was due. The interest ("Interest") charged shall be computed at the rate of 10% per annum but shall not exceed the maximum rate allowed by law. Interest is payable in addition to the potential late charge provided for in Paragraph 13.4.

$13.6 \ \, \textbf{Breach by Lessor}.$

- (a) Notice of Breach. Lessor shall not be deemed in breach of this Lease unless Lessor fails within a reasonable time to perform an obligation required to be performed by Lessor. For purposes of this Paragraph, a reasonable time shall in no event be less than 30 days after receipt by Lessor, and any Lender whose name and address shall have been furnished to Lessee in writing for such purpose, of written notice specifying wherein such obligation of Lessor has not been performed; provided, however, that if the nature of Lessor's obligation is such that more than 30 days are reasonably required for its performance, then Lessor shall not be in breach if performance is commenced within such 30 day period and thereafter diligently pursued to completion.
- (b) **Performance by Lessee on Behalf of Lessor**. In the event that neither Lessor nor Lender cures said breach within 30 days after receipt of said notice, or if having commenced said cure they do not diligently pursue it to completion, then Lessee may elect to cure said breach at Lessee's expense and offset from Rent the actual and reasonable cost to perform such cure, provided however, that such offset shall not exceed an amount equal to the greater of one month's Base Rent or the Security Deposit, reserving Lessee's right to seek reimbursement from Lessor for any such expense in excess of such offset. Lessee shall document the cost of said cure and supply said documentation to Lessor.
- 14. Condemnation. If the Premises or any portion thereof are taken under the power of eminent domain or sold under the threat of the exercise of said power (collectively "Condemnation"), this Lease shall terminate as to the part taken as of the date the condemning authority takes title or possession, whichever first occurs. If more than 25% of the Premises, is taken by Condemnation, Lessee may, at Lessee's option, to be exercised in writing within 10 days after Lessor shall have given Lessee written notice of such taking (or in the absence of such notice, within 10 days after the condemning authority shall have taken possession) terminate this Lease as of the date the condemning authority takes such possession. If Lessee does not terminate this Lease in accordance with the foregoing, this Lease shall remain in full force and effect as to the portion of the Premises remaining, except that the Base Rent shall be reduced in proportion to the reduction in utility of the Premises caused by such Condemnation. Condemnation awards and/or payments shall be the property of Lessor, whether such award shall be made as compensation for diminution in value of the leasehold, the value of the part taken, or for severance damages; provided, however, that Lessee shall be entitled to any compensation paid by the condemnor for Lessee's relocation expenses, loss of business goodwill and/or Trade Fixtures, without regard to whether or not this Lease is terminated pursuant to the provisions of this Paragraph. All Alterations and Utility Installations made to the Premises by Lessee, for purposes of Condemnation only, shall be considered the property of the Lessee and Lessee shall be entitled to any and all compensation which is payable therefor. In the event that this Lease is not terminated by reason of the Condemnation, Lessor shall repair any damage to the Premises caused by such Condemnation.

15. Brokerage Fees.

- 15.1 Additional Commission. In addition to the payments owed pursuant to Paragraph 1.9 above, Lessor agrees that: (a) if Lessee exercises any Option, (b) if Lessee or anyone affiliated with Lessee acquires any rights to the Premises or other premises owned by Lessor and located within the same Project, if any, within which the Premises is located, (c) if Lessee remains in possession of the Premises, with the consent of Lessor, after the expiration of this Lease, or (d) if Base Rent is increased, whether by agreement or operation of an escalation clause herein, then, Lessor shall pay Brokers a fee in accordance with the fee schedule of the Brokers in effect at the time the Lease was executed. The provisions of this paragraph are intended to supersede the provisions of any earlier agreement to the contrary.
- 15.2 **Assumption of Obligations**. Any buyer or transferee of Lessor's interest in this Lease shall be deemed to have assumed Lessor's obligation hereunder. Brokers shall be third party beneficiaries of the provisions of Paragraphs 1.9, 15, 22 and 31.
- 15.3 Representations and Indemnities of Broker Relationships. Lessee and Lessor each represent and warrant to the other that it has had no dealings with any person, firm, broker, agent or finder (other than the Brokers and Agents, if any) in connection with this Lease, and that no one other than said named Brokers and Agents is entitled to any commission or finder's fee in connection herewith. Lessee and Lessor do each hereby agree to indemnify, protect, defend and hold the other harmless from and against liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

dealings or actions of the indemnifying Party, including any costs, expenses, attorneys' fees reasonably incurred with respect thereto.

16. Estoppel Certificates.

- (a) Each Party (as "Responding Party") shall within 10 days after written notice from the other Party (the "Requesting Party") execute, acknowledge and deliver to the Requesting Party a statement in writing in form similar to the then most current "Estoppel Certificate" form published by AIR CRE, plus such additional information, confirmation and/or statements as may be reasonably requested by the Requesting Party.
- (b) If the Responding Party shall fail to execute or deliver the Estoppel Certificate within such 10 day period, the Requesting Party may execute an Estoppel Certificate stating that: (i) the Lease is in full force and effect without modification except as may be represented by the Requesting Party, (ii) there are no uncured defaults in the Requesting Party's performance, and (iii) if Lessor is the Requesting Party, not more than one month's rent has been paid in advance. Prospective purchasers and encumbrancers may rely upon the Requesting Party's Estoppel Certificate, and the Responding Party shall be estopped from denying the truth of the facts contained in said Certificate. In addition, Lessee acknowledges that any failure on its part to provide such an Estoppel Certificate will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to execute and/or deliver a requested Estoppel Certificate in a timely fashion the monthly Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for remainder of the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to provide the Estoppel Certificate. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to provide the Estoppel Certificate nor prevent the exercise of any of the other rights and remedies granted hereunder.
- (c) If Lessor desires to finance, refinance, or sell the Premises, or any part thereof, Lessee and all Guarantors shall within 10 days after written notice from Lessor deliver to any potential lender or purchaser designated by Lessor such financial statements as may be reasonably required by such lender or purchaser, including but not limited to Lessee's financial statements for the past 3 years. All such financial statements shall be received by Lessor and such lender or purchaser in confidence and shall be used only for the purposes herein set forth.
- 17. Definition of Lessor. The term "Lessor" as used herein shall mean the owner or owners at the time in question of the fee title to the Premises, or, if this is a sublease, of the Lessee's interest in the prior lease. In the event of a transfer of Lessor's title or interest in the Premises or this Lease, Lessor shall deliver to the transferee or assignee (in cash or by credit) any unused Security Deposit held by Lessor. Upon such transfer or assignment and delivery of the Security Deposit, as aforesaid, the prior Lessor shall be relieved of all liability with respect to the obligations and/or covenants under this Lease thereafter to be performed by the Lessor. Subject to the foregoing, the obligations and/or covenants in this Lease to be performed by the Lessor shall be binding only upon the Lessor as hereinabove defined.
- 18. Severability. The invalidity of any provision of this Lease, as determined by a court of competent jurisdiction, shall in no way affect the validity of any other provision hereof.
- 19. Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Lease shall mean and refer to calendar days.
- 20. Limitation on Liability. The obligations of Lessor under this Lease shall not constitute personal obligations of Lessor, or its partners, members, directors, officers or shareholders, and Lessee shall look to the Premises, and to no other assets of Lessor, for the satisfaction of any liability of Lessor with respect to this Lease, and shall not seek recourse against Lessor's partners, members, directors, officers or shareholders, or any of their personal assets for such satisfaction.
- 21. Time of Essence. Time is of the essence with respect to the performance of all obligations to be performed or observed by the Parties under this Lease.
- 22. No Prior or Other Agreements; Broker Disclaimer. This Lease contains all agreements between the Parties with respect to any matter mentioned herein, and no other prior or contemporaneous agreement or understanding shall be effective. Lessor and Lessee each represents and warrants to the Brokers that it has made, and is relying solely upon, its own investigation as to the nature, quality, character and financial responsibility of the other Party to this Lease and as to the use, nature, quality and character of the Premises. Brokers have no responsibility with respect thereto or with respect to any default or breach hereof by either Party.

23. Notices.

- 23.1 Notice Requirements. All notices required or permitted by this Lease or applicable law shall be in writing and may be delivered in person (by hand or by courier) or may be sent by regular, certified or registered mail or U.S. Postal Service Express Mail, with postage prepaid, or by facsimile transmission, or by email, and shall be deemed sufficiently given if served in a manner specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specify a different address for notice, except that upon Lessee's taking possession of the Premises, the Premises shall constitute Lessee's address for notice. A copy of all notices to Lessor shall be concurrently transmitted to such party or parties at such addresses as Lessor may from time to time hereafter designate in writing.
- 23.2 **Date of Notice**. Any notice sent by registered or certified mail, return receipt requested, shall be deemed given on the date of delivery shown on the receipt card, or if no delivery date is shown, the postmark thereon. If sent by regular mail the notice shall be deemed given 72 hours after the same is addressed as required herein and mailed with postage prepaid. Notices delivered by United States Express Mail or overnight courier that guarantees next day delivery shall be deemed given 24 hours after delivery of the same to the Postal Service or courier. Notices delivered by hand, or transmitted by facsimile transmission or by email shall be deemed delivered upon actual receipt. If notice is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.
- 23.3 **Options.** Notwithstanding the foregoing, in order to exercise any Options (see paragraph 39), the Notice must be sent by Certified Mail (return receipt requested), Express Mail (signature required), courier (signature required) or some other methodology that provides a receipt establishing the date the notice was received by the Lessor.

24. Waivers.

- (a) No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any other term, covenant or condition hereof. Lessor's consent to, or approval of, any act shall not be deemed to render unnecessary the obtaining of Lessor's consent to, or approval of, any subsequent or similar act by Lessee, or be construed as the basis of an estoppel to enforce the provision or provisions of this Lease requiring such consent.
- (b) The acceptance of Rent by Lessor shall not be a waiver of any Default or Breach by Lessee. Any payment by Lessee may be accepted by Lessor on account of monies or damages due Lessor, notwithstanding any qualifying statements or conditions made by Lessee in connection therewith, which such statements and/or conditions shall be of no force or effect whatsoever unless specifically agreed to in writing by Lessor at or before the time of deposit of such payment.
- (c) THE PARTIES AGREE THAT THE TERMS OF THIS LEASE SHALL GOVERN WITH REGARD TO ALL MATTERS RELATED THERETO AND HEREBY WAIVE THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE TO THE EXTENT THAT SUCH STATUTE IS INCONSISTENT WITH THIS LEASE.

25. Disclosures Regarding The Nature of a Real Estate Agency Relationship.

- (a) When entering into a discussion with a real estate agent regarding a real estate transaction, a Lessor or Lessee should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Lessor and Lessee acknowledge being advised by the Brokers in this transaction, as follows:
- (i) <u>Lessor's Agent</u>. A Lessor's agent under a listing agreement with the Lessor acts as the agent for the Lessor only. A Lessor's agent or subagent has the following affirmative obligations: <u>To the Lessor</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessor. <u>To the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (ii) <u>Lessee's Agent</u>. An agent can agree to act as agent for the Lessee only. In these situations, the agent is not the Lessor's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Lessor. An agent acting only for a Lessee has the following affirmative obligations. <u>To the Lessee</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessee. <u>To the Lessee and the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (iii) <u>Agent Representing Both Lessor and Lessee</u>. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Lessor and the Lessee in a transaction, but only with the knowledge and consent of both the Lessor and the Lessee. In a dual agency situation, the agent has the following affirmative obligations to both the Lessor and the Lessee: (a) A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Lessor or the Lessee. (b) Other duties to the Lessor and the Lessee as stated above in subparagraphs (i) or (ii). In representing both Lessor and Lessee, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

to either Lessee's or Lessor's financial position, motivations, bargaining position, or other personal information that may impact rent, including Lessor's willingness to accept a rent less than the listing rent or Lessee's willingness to pay rent greater than the rent offered. The above duties of the agent in a real estate transaction do not relieve a Lessor or Lessee from the responsibility to protect their own interests. Lessor and Lessee should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Both Lessor and Lessee should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.

- (b) Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this Lease may be brought against Broker more than one year after the Start Date and that the liability (including court costs and attorneys' fees), of any Broker with respect to any such lawsuit and/or legal proceeding shall not exceed the fee received by such Broker pursuant to this Lease; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.
- (c) Lessor and Lessee agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- 26. No Right To Holdover. Lessee has no right to retain possession of the Premises or any part thereof beyond the expiration or termination of this Lease. At or prior to the expiration or termination of this Lease Lessee shall deliver exclusive possession of the Premises to Lessor. For purposes of this provision and Paragraph 13.1(a), exclusive possession shall mean that Lessee shall have vacated the Premises, removed all of its personal property therefrom and that the Premises have been returned in the condition specified in this Lease. In the event that Lessee does not deliver exclusive possession to Lessor as specified above, then Lessor's damages during any holdover period shall be computed at the amount of the Rent (as defined in Paragraph 4.1) due during the last full month before the expiration or termination of this Lease (disregarding any temporary abatement of Rent that may have been in effect), but with Base Rent being 150% of the Base Rent payable during such last full month. Nothing contained herein shall be construed as consent by Lessor to any holding over by Lessee.
- 27. Cumulative Remedies. No remedy or election hereunder shall be deemed exclusive but shall, wherever possible, be cumulative with all other remedies at law or in equity.
- 28. Covenants and Conditions; Construction of Agreement. All provisions of this Lease to be observed or performed by Lessee are both covenants and conditions. In construing this Lease, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Lease. Whenever required by the context, the singular shall include the plural and vice versa. This Lease shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.
- 29. Binding Effect; Choice of Law. This Lease shall be binding upon the Parties, their personal representatives, successors and assigns and be governed by the laws of the State in which the Premises are located. Any litigation between the Parties hereto concerning this Lease shall be initiated in the county in which the Premises are located. Signatures to this Lease accomplished by means of electronic signature or similar technology shall be legal and binding.

30. Subordination; Attornment; Non-Disturbance.

- 30.1 **Subordination**. This Lease and any Option granted hereby shall be subject and subordinate to any ground lease, mortgage, deed of trust, or other hypothecation or security device (collectively, "Security Device"), now or hereafter placed upon the Premises, to any and all advances made on the security thereof, and to all renewals, modifications, and extensions thereof. Lessee agrees that the holders of any such Security Devices (in this Lease together referred to as "Lender" shall have no liability or obligation to perform any of the obligations of Lessor under this Lease. Any Lender may elect to have this Lease and/or any Option granted hereby superior to the lien of its Security Device by giving written notice thereof to Lessee, whereupon this Lease and such Options shall be deemed prior to such Security Device, notwithstanding the relative dates of the documentation or recordation thereof.
- 30.2 **Attornment**. In the event that Lessor transfers title to the Premises, or the Premises are acquired by another upon the foreclosure or termination of a Security Device to which this Lease is subordinated (i) Lessee shall, subject to the non-disturbance provisions of Paragraph 30.3, attorn to such new owner, and upon request, enter into a new lease, containing all of the terms and provisions of this Lease, with such new owner for the remainder of the term hereof, or, at the election of the new owner, this Lease will automatically become a new lease between Lessee and such new owner, and (ii) Lessor shall thereafter be relieved of any further obligations hereunder and such new owner shall assume all of Lessor's obligations, except that such new owner shall not: (a) be liable for any act or omission of any prior lessor or with respect to events occurring prior to acquisition of ownership; (b) be subject to any offsets or defenses which Lessee might have against any prior lessor, (c) be bound by prepayment of more than one month's rent, or (d) be liable for the return of any security deposit paid to any prior lessor which was not paid or credited to such new owner.
- 30.3 **Non-Disturbance**. With respect to Security Devices entered into by Lessor after the execution of this Lease, Lessee's subordination of this Lease shall be subject to receiving a commercially reasonable non-disturbance agreement (a "Non-Disturbance Agreement") from the Lender which Non-Disturbance Agreement provides that Lessee's possession of the Premises, and this Lease, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises.
- 30.4 **Self-Executing**. The agreements contained in this Paragraph 30 shall be effective without the execution of any further documents; provided, however, that, upon written request from Lessor or a Lender in connection with a sale, financing or refinancing of the Premises, Lessee and Lessor shall execute such further writings as may be reasonably required to separately document any subordination, attornment and/or Non-Disturbance Agreement provided for herein.
- 31. Attorneys' Fees. If any Party or Broker brings an action or proceeding involving the Premises whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred. In addition, Lessor shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is subsequently commenced in connection with such Default or resulting Breach (\$200 is a reasonable minimum per occurrence for such services and consultation).
- 32. Lessor's Access; Showing Premises; Repairs. Lessor and Lessor's agents shall have the right to enter the Premises at any time, in the case of an emergency, and otherwise at reasonable times after reasonable prior notice for the purpose of showing the same to prospective purchasers, lenders, or tenants, and making such alterations, repairs, improvements or additions to the Premises as Lessor may deem necessary or desirable and the erecting, using and maintaining of utilities, services, pipes and conduits through the Premises and/or other premises as long as there is no material adverse effect on Lessee's use of the Premises. All such activities shall be without abatement of rent or liability to Lessee.
- **33. Auctions.** Lessee shall not conduct, nor permit to be conducted, any auction upon the Premises without Lessor's prior written consent. Lessor shall not be obligated to exercise any standard of reasonableness in determining whether to permit an auction.
- **34. Signs.** Lessor may place on the Premises ordinary "For Sale" signs at any time and ordinary "For Lease" signs during the last 6 months of the term hereof. Except for ordinary "For Sublease" signs, Lessee shall not place any sign upon the Premises without Lessor's prior written consent. All signs must comply with all Applicable Requirements.
- **35. Termination; Merger.** Unless specifically stated otherwise in writing by Lessor, the voluntary or other surrender of this Lease by Lessee, the mutual termination or cancellation hereof, or a termination hereof by Lessor for Breach by Lessee, shall automatically terminate any sublease or lesser estate in the Premises; provided, however, that Lessor may elect to continue any one or all existing subtenancies. Lessor's failure within 10 days following any such event to elect to the contrary by written notice to the holder of any such lesser interest, shall constitute Lessor's election to have such event constitute the termination of such interest.
- **36. Consents.** All requests for consent shall be in writing. Except as otherwise provided herein, wherever in this Lease the consent of a Party is required to an act by or for the other Party, such consent shall not be unreasonably withheld or delayed. Lessor's actual reasonable costs and expenses (including but not limited to architects', attorneys', engineers' and other consultants' fees) incurred in the consideration of, or response to, a request by Lessee for any Lessor consent, including but not limited to consents to an assignment, a subletting or the presence or use of a Hazardous Substance, shall be paid by Lessee upon receipt of an invoice and supporting documentation therefor. Lessor's consent to any act, assignment or subletting shall not constitute an acknowledgment that no Default or Breach by Lessee of this Lease exists, nor shall such consent be deemed a waiver of any then existing Default or Breach, except as may be otherwise specifically stated in writing by Lessor at the time of such consent. The failure to specify herein any particular condition to Lessor's consent shall not preclude the imposition by Lessor at the time of consent of such further or other conditions as are then reasonable with reference to the particular matter for which consent is being given. In the event that either Party disagrees with any determination made by the other hereunder and reasonably requests the reasons for such determination, the determining party shall furnish its reasons in writing and in reasonable detail within 10 business days following such request.

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

37. Guarantor.

- 37.1 Execution. The Guarantors, if any, shall each execute a guaranty in the form most recently published by AIR CRE.
- 37.2 **Default**. It shall constitute a Default of the Lessee if any Guarantor fails or refuses, upon request to provide: (a) evidence of the execution of the guaranty, including the authority of the party signing on Guarantor's behalf to obligate Guarantor, and in the case of a corporate Guarantor, a certified copy of a resolution of its board of directors authorizing the making of such guaranty, (b) current financial statements, (c) an Estoppel Certificate, or (d) written confirmation that the guaranty is still in effect.
- **38.** Quiet Possession. Subject to payment by Lessee of the Rent and performance of all of the covenants, conditions and provisions on Lessee's part to be observed and performed under this Lease. Lessee shall have quiet possession and quiet enjoyment of the Premises during the term hereof.
- 39. Options. If Lessee is granted any Option, as defined below, then the following provisions shall apply.
- 39.1 **Definition**. "**Option**" shall mean: (a) the right to extend or reduce the term of or renew this Lease or to extend or reduce the term of or renew any lease that Lessee has on other property of Lessor; (b) the right of first refusal or first offer to lease either the Premises or other property of Lessor; (c) the right to purchase, the right of first offer to purchase or the right of first refusal to purchase the Premises or other property of Lessor.
- 39.2 **Options Personal To Original Lessee**. Any Option granted to Lessee in this Lease is personal to the original Lessee, and cannot be assigned or exercised by anyone other than said original Lessee and only while the original Lessee is in full possession of the Premises and, if requested by Lessor, with Lessee certifying that Lessee has no intention of thereafter assigning or subletting.
- 39.3 **Multiple Options**. In the event that Lessee has any multiple Options to extend or renew this Lease, a later Option cannot be exercised unless the prior Options have been validly exercised.

39.4 Effect of Default on Options.

- (a) Lessee shall have no right to exercise an Option: (i) during the period commencing with the giving of any notice of Default and continuing until said Default is cured, (ii) during the period of time any Rent is unpaid (without regard to whether notice thereof is given Lessee), (iii) during the time Lessee is in Breach of this Lease, or (iv) in the event that Lessee has been given 3 or more notices of separate Default, whether or not the Defaults are cured, during the 12 month period immediately preceding the exercise of the Option.
- (b) The period of time within which an Option may be exercised shall not be extended or enlarged by reason of Lessee's inability to exercise an Option because of the provisions of Paragraph 39.4(a).
- (c) An Option shall terminate and be of no further force or effect, notwithstanding Lessee's due and timely exercise of the Option, if, after such exercise and prior to the commencement of the extended term or completion of the purchase, (i) Lessee fails to pay Rent for a period of 30 days after such Rent becomes due (without any necessity of Lessor to give notice thereof), or (ii) if Lessee commits a Breach of this Lease.
- **40. Multiple Properties.** If the Premises are a part of a group of properties controlled by Lessor, Lessee agrees that it will abide by and conform to all reasonable rules and regulations which Lessor may make from time to time for the management, safety, and care of said properties, including the care and cleanliness of the grounds and including the parking, loading and unloading of vehicles, and to cause its employees, suppliers, shippers, customers, contractors and invitees to so abide and conform. Lessee also agrees to pay its fair share of common expenses incurred in connection with such rules and regulations.
- **41. Security Measures.** Lessee hereby acknowledges that the Rent payable to Lessor hereunder does not include the cost of guard service or other security measures, and that Lessor shall have no obligation whatsoever to provide same. Lessee assumes all responsibility for the protection of the Premises, Lessee, its agents and invitees and their property from the acts of third parties.
- **42. Reservations.** Lessor reserves to itself the right, from time to time, to grant, without the consent or joinder of Lessee, such easements, rights and dedications that Lessor deems necessary, and to cause the recordation of parcel maps and restrictions, so long as such easements, rights, dedications, maps and restrictions do not unreasonably interfere with the use of the Premises by Lessee. Lessee agrees to sign any documents reasonably requested by Lessor to effectuate any such easement rights, dedication, map or restrictions.
- **43. Performance Under Protest.** If at any time a dispute shall arise as to any amount or sum of money to be paid by one Party to the other under the provisions hereof, the Party against whom the obligation to pay the money is asserted shall have the right to make payment "under protest" and such payment shall not be regarded as a voluntary payment and there shall survive the right on the part of said Party to institute suit for recovery of such sum. If it shall be adjudged that there was no legal obligation on the part of said Party to pay such sum or any part thereof, said Party shall be entitled to recover such sum or so much thereof as it was not legally required to pay. A Party who does not initiate suit for the recovery of sums paid "under protest" within 6 months shall be deemed to have waived its right to protest such payment.

44. Authority; Multiple Parties; Execution.

- (a) If either Party hereto is a corporation, trust, limited liability company, partnership, or similar entity, each individual executing this Lease on behalf of such entity represents and warrants that he or she is duly authorized to execute and deliver this Lease on its behalf. Each Party shall, within 30 days after request, deliver to the other Party satisfactory evidence of such authority.
- (b) If this Lease is executed by more than one person or entity as "Lessee", each such person or entity shall be jointly and severally liable hereunder. It is agreed that any one of the named Lessees shall be empowered to execute any amendment to this Lease, or other document ancillary thereto and bind all of the named Lessees, and Lessor may rely on the same as if all of the named Lessees had executed such document.
- (c) This Lease may be executed by the Parties in counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument
- **45. Conflict.** Any conflict between the printed provisions of this Lease and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions.
- **46. Offer**. Preparation of this Lease by either Party or their agent and submission of same to the other Party shall not be deemed an offer to lease to the other Party. This Lease is not intended to be binding until executed and delivered by all Parties hereto.
- **47. Amendments.** This Lease may be modified only in writing, signed by the Parties in interest at the time of the modification. As long as they do not materially change Lessee's obligations hereunder, Lessee agrees to make such reasonable non-monetary modifications to this Lease as may be reasonably required by a Lender in connection with the obtaining of normal financing or refinancing of the Premises.
- 48. Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY OR ARISING OUT OF THIS AGREEMENT.

	itration of Disputes. An Addendum requiring the Arbitration of all disputes between the Parties and/or Brokers arising out of this Lease 🗀 is 🗀 is not to this Lease.
50. Acc	essibility; Americans with Disabilities Act. (a) The Premises:
determin require a the subje arrangem	not undergone an inspection by a Certified Access Specialist (CASp). Note: A Certified Access Specialist (CASp) can inspect the subject premises and e whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection of ct premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the lents for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to olations of construction-related accessibility standards within the premises.
standards	undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises met all applicable construction-related accessibility pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this diagrees to keep such report confidential.
accessibil executing	undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises did not meet all applicable construction-related ity standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to this Lease and agrees to keep such report confidential except as necessary to complete repairs and corrections of violations of construction related ity standards.

INITIALS

INITIALS
© 2019 AIR CRE. All Rights Reserved.
LG-15.23, Revised 10-22-2020

In the event that the Premises have been issued an inspection report by a CASp the Lessor shall provide a copy of the disability access inspection certificate to Lessee within 7 days of the execution of this Lease.

Since compliance with the Americans with Disabilities Act (ADA) and other state and local accessibility statutes are dependent upon Lessee's specific use of the Premises, Lessor makes no warranty or representation as to whether or not the Premises comply with ADA or any similar legislation. In the event that Lessee's use of the Premises requires modifications or additions to the Premises in order to be in compliance with ADA or other accessibility statutes, Lessee agrees to make any such necessary modifications and/or additions at Lessee's expense.

LESSOR AND LESSEE HAVE CAREFULLY READ AND REVIEWED THIS LEASE AND EACH TERM AND PROVISION CONTAINED HEREIN, AND BY THE EXECUTION OF THIS LEASE SHOW THEIR INFORMED AND VOLUNTARY CONSENT THERETO. THE PARTIES HEREBY AGREE THAT, AT THE TIME THIS LEASE IS EXECUTED, THE TERMS OF THIS LEASE ARE COMMERCIALLY REASONABLE AND EFFECTUATE THE INTENT AND PURPOSE OF LESSOR AND LESSEE WITH RESPECT TO THE PREMISES.

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS LEASE OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

1. SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS LEASE.

- RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PREMISES. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING OF THE PREMISES, THE CONDITION AND THE SUITABILITY OF THE PREMISES FOR LESSEE'S INTENDED USE.

WARNING: IF THE PREMISES ARE LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THE LEASE MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PREMISES ARE LOCATED.

The parties hereto have executed this Lease at the place and on the dates specifie	d above their respective signatures.
Executed at:	Executed at:
On:	On:
By LESSOR:	By LESSEE:
By:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
By:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Address:	Address:
Federal ID No.:	Federal ID No.:
	X
BROKER	BROKER
BROKEN	BROKER
< 1	
Attn:	Attn:
Title:	Title:
Address:	Address:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Federal ID No.:	Federal ID No.:
Broker DRE License #:	Broker DRE License #:
Agent DRE License #:	Agent DRE License #:
	· · · · · · · · · · · · · · · · · · ·
< /	
AIR CRE * https://www.aircre.com *	213-687-8777 * contracts@aircre.com
	duced in any form without permission in writing.
\	,
\sim \vee	

INITIALS

INITIALS © 2019 AIR CRE. All Rights Reserved.



STANDARD INDUSTRIAL/COMMERCIAL MULTI-TENANT LEASE - NET

1. Basic Provisions ("Basic Provisions").	
1.1 Parties. This Lease ("Lease"), dated for reference purposes only	, is made by and between ("Lessor") and
("Lessee"), (collectively the "Parties", or individually a "Party"). 1.2(a) Premises: That certain real property, including all improvements the	erein or to be provided by Lessor under the terms of this Lease, commonly known
as (street address, unit/suite, city, state, zip): ("Premises"). The Prem	
(describe briefly the nature of the Premises and the "Project"):	
Lessee shall have non-exclusive rights to any utility raceways of the building containing	
2.7 below), but shall not have any rights to the roof, or exterior walls of the Building	or to any other buildings in the Project. The Premises, the Building, the Common
Areas, the land upon which they are located, along with all other buildings and impro	ovements thereon, are herein collectively referred to as the "Project." (See also
Paragraph 2)	
1.2(b) Parking: unreserved vehicle parking spaces. (See also	
1.3 Term: years and months ("Original Term") o	ommencing ("Commencement Date") and ending
("Expiration Date"). (See also Paragraph 3) 1.4 Early Possession: If the Premises are available Lessee may have non-exclusion.	usive pessession of the Dremises commencing ("Early Descession
Date"). (See also Paragraphs 3.2 and 3.3)	isive possession of the Premises commencing (Early Possession
, ,	day of each month commencing (See also Paragraph 4)
Lifthis box is checked, there are provisions in this Lease for the Base Rent to 1.6 Lessee's Share of Common Area Operating Expenses: per	
Premises and/or the Project are modified during the term of this Lease, Lessor shall re	
1.7 Base Rent and Other Monies Paid Upon Execution:	eculculate respect 3 share to reflect sacri modification.
(a) Base Rent: for the period	
(b) Common Area Operating Expenses: The current estimate for the pe	eriod is .
(c) Security Deposit: ("Security Deposit"). (See also Par	
(d) Other: for	
(e) Total Due Upon Execution of this Lease:	
1.8 Agreed Use: (See also Paragraph 6)	
1.9 Insuring Party. Lessor is the "Insuring Party". (See also Paragraph 8)	
1.10 Real Estate Brokers. (See also Paragraphs 15 and 25)	
	garding Real Estate Agency Relationship, confirms and consents to the following
agency relationships in this Lease with the following real estate brokers ("Broker(s)")	
Lessor's Brokerage Firm License No Is the broker	of (check one): the Lessor; or both the Lessee and Lessor (dual agent).
Lessor's Agent License No is (check one): the	e Lessor's Agent (salesperson or broker associate); or both the Lessee's
Agent and the Lessor's Agent (dual agent).	
Lessee's Brokerage Firm License No Is the broke	r of (check one). the Lessee or hoth the Lessee and Lessor (dual
agent).	solvenie zessee und zesser (dad.
Lessee's Agent License No is (check one):t	had become Agent (calesparson or broker accepiate); or heath the Leccomic
Agent and the Lessor's Agent (dual agent).	the Lessee's Agent (salesperson of broker associate), of both the Lessee's
	ooth Parties Lessor shall hav to the Brokers the brokerage fee agreed to in a
(b) Payment to Brokers. Upon execution and delivery of this Lease by b	ooth Parties, Lessor shall pay to the Brokers the brokerage fee agreed to in a
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers.	or% of the total Base Rent) for the brokerage services rendered
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37)
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed attachments. Attached hereto are the following, all of which constitute as	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37)
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37)
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed and Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan depicting the Premises;	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37)
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37)
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed and Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan depicting the Premises;	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37)
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:;
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarantered at a Attachments. Attached hereto are the following, all of which constitute at an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:;
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:;
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarantered at a Attachments. Attached hereto are the following, all of which constitute at an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:;
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:;
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarantered at a state plan depicting of Paragraphs	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:; ion;
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarantered at a state plan depicting of Paragraphs through a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footed.	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:; ion; assor, the Premises, for the term, at the rental, and upon all of the terms, age of the Premises may have been used in the marketing of the Premises for
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarantered at a state plan depicting of Paragraphs through a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage and proposes of comparison, the Base Rent stated herein is NOT tied to square footage as	or
(b) Payment to Brokers. Upon execution and delivery of this Lease by the separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarantered at the following, all of which constitute a land an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter: other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lessee.	or
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed and the sum of such agreement, the sum of by the Brokers. 1.12 Attachments. Attached hereto are the following, all of which constitute a sum and an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter: other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. Condition. Lessor shall deliver that portion of the Premises contained with the sum of the premises contained with the premises contained with the premises contained with the premises contained with the premise contained with the p	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:; ion; issor, the Premises, for the term, at the rental, and upon all of the terms, age of the Premises may have been used in the marketing of the Premises for and is not subject to adjustment should the actual size be determined to be ease. hin the Building ("Unit") to Lessee broom clean and free of debris on the
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed and the sum of such agreement, the sum of by the Brokers. 1.12 Attachments. Attached hereto are the following, all of which constitute a sum and an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter: other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. Condition. Lessor shall deliver that portion of the Premises contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date of the Paramater of the Premises contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date of the Paramater of t	or% of the total Base Rent) for the brokerage services rendered nteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:; ion; issor, the Premises, for the term, at the rental, and upon all of the terms, age of the Premises may have been used in the marketing of the Premises for and is not subject to adjustment should the actual size be determined to be ease. hin the Building ("Unit") to Lessee broom clean and free of debris on the te"), and, so long as the required service contracts described in Paragraph 7.1(b)
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed and the sum of such agreement, the sum of by the Brokers. 1.12 Attachments. Attached hereto are the following, all of which constitute a sum and an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter: other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. Condition. Lessor shall deliver that portion of the Premises contained with the sum of the premises contained with the premises contained with the premises contained with the premises contained with the premise contained with the p	or
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed at 1.12 Attachments. Attached hereto are the following, all of which constitute a land an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage a different. NOTE: Lessee is advised to verify the actual size prior to executing this Lector 2.2 Condition. Lessor shall deliver that portion of the Premises contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the	or
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed at 1.12 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter: other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage a different. NOTE: Lessee is advised to verify the actual size prior to executing this Lector 2.2 Condition. Lessor shall deliver that portion of the Premises contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungil defined as the sum of the proper in the sum of the proper in th	or
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.12 Attachments. Attached hereto are the following, all of which constitute a land an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage a different. NOTE: Lessee is advised to verify the actual size prior to executing this Lector 2.2 Condition. Lessor shall deliver that portion of the Premises contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date below are obtained by Lessee and in effect within thirty days following the Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as twarranty exists as of the Start Date, or if one of such systems or elements should ma	or
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarantered and the sum of such a step land depicting the Premises; a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. Condition. Lessor shall deliver that portion of the Premises contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date below are obtained by Lessee and in effect within thirty days following the Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as twarranty exists as of the Start Date, or if one of such systems or elements should masole obligation with respect to such matter, except as otherwise provided in this Lease.	or
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranter. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. Work actual size prior to executing this Lease with the continuous contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date below are obtained by Lessee and in effect within thirty days following the Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as twarranty exists as of the Start Date, or if one of such systems or elements should ma sole obligation with respect to such matter, except as otherwise provided in this Lease specificity the nature and extent of such non-compliance, malfunction or failure, received as the such contain paragraphs.	mteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:; inion; assor, the Premises, for the term, at the rental, and upon all of the terms, age of the Premises may have been used in the marketing of the Premises for and is not subject to adjustment should the actual size be determined to be tease. hin the Building ("Unit") to Lessee broom clean and free of debris on the tee"), and, so long as the required service contracts described in Paragraph 7.1(b) et and in the the suit of the the suit of the the subject to adjustment should the actual size be determined to be tee"), and, so long as the required service contracts described in Paragraph 7.1(b) et and in the the subject to the the subject to the the subject to adjustment should the actual size be determined to be tee"), and, so long as the required service contracts described in Paragraph 7.1(b) et and in Paragra
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarantered and the sum of such a step land depicting the Premises; a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. Condition. Lessor shall deliver that portion of the Premises contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date below are obtained by Lessee and in effect within thirty days following the Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as twarranty exists as of the Start Date, or if one of such systems or elements should masole obligation with respect to such matter, except as otherwise provided in this Lease.	mteed by ("Guarantor"). (See also Paragraph 37) part of this Lease:; issor, the Premises, for the term, at the rental, and upon all of the terms, age of the Premises may have been used in the marketing of the Premises for and is not subject to adjustment should the actual size be determined to be ease. hin the Building ("Unit") to Lessee broom clean and free of debris on the ter"), and, so long as the required service contracts described in Paragraph 7.1(b) e, warrants that the existing electrical, plumbing, fire sprinkler, lighting, heating, and all other such elements in the Unit, other than those constructed by Lessee, roof, bearing walls and foundation of the Unit shall be free of material defects, oxic under applicable state or federal law. If a non-compliance with such lfunction or fail within the appropriate warranty period, Lessor shall, as Lessor's e, promptly after receipt of written notice from Lessee setting forth with iffy same at Lessor's expense. The warranty periods shall be as follows: (i) 6 er elements of the Unit. If Lessee does not give Lessor the required notice within
(b) Payment to Brokers. Upon execution and delivery of this Lease by the separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guara 1.12 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Premises; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lecovenants and conditions set forth in this Lease. While the approximate square footage a different. NOTE: Lessee is advised to verify the actual size prior to executing this Lesse. 2.2 Condition. Lessor shall deliver that portion of the Premises contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as twarranty exists as of the Start Date, or if one of such systems or elements should ma sole obligation with respect to such matter, except as otherwise provided in this Leas specificity the nature and extent of such non-compliance, malfunction or failure, recements as to the HVAC systems, and (ii) 30 days as to the remaining systems and oth the appropriate warranty period, correction of any such non-compliance, malfunction (except for the repairs to the fire sprinkler systems, roof, foundations, and/or bearing except for the repairs to the fire sprinkler systems, roof, foundations, and/or bearing except for the repairs to the fire sprinkler systems, roof, foundations, and/or bearing except for the repairs to the fire sprinkler systems.	or
(b) Payment to Brokers. Upon execution and delivery of this Lease by the Separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed to be guaranteed. It is a site plan depicting the Premises; a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' association as Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lector and conditions. Lessor shall deliver that portion of the Premises contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date below are obtained by Lessee and in effect within thirty days following the Start Date below are obtained by Lessee and in effect within thirty days following the Start Date wentilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as the warranty exists as of the Start Date, or if one of such systems or elements should masole obligation with respect to such matter, except as otherwise provided in this Leas specificity the nature and extent of such non-compliance, malfunction or failure, remonths as to the HVAC Systems, and (ii) 30 days as to the remaining systems and oth the appropriate warranty period, correction of any such non-compliance, malfunction (except for the repairs to the fire sprinkler systems, roof, foundations, and/or bearing writing, Lessor is unaware of (i) any recorded Notices of Default affecting the Premise.	or
(b) Payment to Brokers. Upon execution and delivery of this Lease by the Separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guara 1.12 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Premises; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): . 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing the Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing the Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing the Commencement Date or the Early Possession Date, whichever first occurs ("Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as twarranty exists as of the Start Date, or if one of such systems or elements should ma sole obligation with respect to such matter, except as otherwise provided in this Leas specificity the nature and extent of such non-compliance, malfunction or failure, rect months as to the HVAC systems, and (ii) 30 days as to the remaining systems and oth the appropriate warranty period, correction of any such non-compliance, malfunction (except for the repairs to the fire sprinkler syst	or
(b) Payment to Brokers. Upon execution and delivery of this Lease by the separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.12 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs	"("Guarantor"). (See also Paragraph 37) part of this Lease: ; geo of the Premises, for the term, at the rental, and upon all of the terms, age of the Premises may have been used in the marketing of the Premises for and is not subject to adjustment should the actual size be determined to be ease. hin the Building ("Unit") to Lessee broom clean and free of debris on the tet"), and, so long as the required service contracts described in Paragraph 7.1(b) by warrants that the existing electrical, plumbing, fire sprinkler, lighting, heating, and all other such elements in the Unit, other than those constructed by Lessee, roof, bearing walls and foundation of the Unit shall be free of material defects, oxic under applicable state or federal law. If a non-compliance with such lifunction or fail within the appropriate warranty period, Lessor shall, as Lessor's e, promptly after receipt of written notice from Lessee setting forth with iffy same at Lessor's expense. The warranty periods shall be as follows: (i) 6 er elements of the Unit. If Lessee does not give Lessor the required notice within no r failure shall be the obligation of Lessee at Lessee's sole cost and expense (walls - see Paragraph 7). Lessor also warrants, that unless otherwise specified in expense sements on the Premises comply with the building codes, applicable laws,
(b) Payment to Brokers. Upon execution and delivery of this Lease by the Separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guara 1.12 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Premises; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): . 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing the Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing the Lectovenants and conditions set forth in this Lease. While the approximate square footage addifferent. NOTE: Lessee is advised to verify the actual size prior to executing the Commencement Date or the Early Possession Date, whichever first occurs ("Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as twarranty exists as of the Start Date, or if one of such systems or elements should ma sole obligation with respect to such matter, except as otherwise provided in this Leas specificity the nature and extent of such non-compliance, malfunction or failure, rect months as to the HVAC systems, and (ii) 30 days as to the remaining systems and oth the appropriate warranty period, correction of any such non-compliance, malfunction (except for the repairs to the fire sprinkler syst	"Guarantor"). (See also Paragraph 37) part of this Lease: ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ;
(b) Payment to Brokers. Upon execution and delivery of this Lease by the separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.12 Attachments. Attached hereto are the following, all of which constitute a land an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' association a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lecovenants and conditions set forth in this Lease. While the approximate square footagurposes of comparison, the Base Rent stated herein is NOT tied to square footage a different. NOTE: Lessee is advised to verify the actual size prior to executing this Lease of the content of the premises contained with Commencement Date or the Early Possession Date, whichever first occurs ("Start Date below are obtained by Lessee and in effect within thirty days following the Start Date below are obtained by Lessee and in effect within thirty days following the Start Date wentilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as twarranty exists as of the Start Date, or if one of such systems or elements should massle obligation with respect to such matter, except as otherwise provided in this Leas specificity the nature and extent of such non-compliance, malfunction or failure, recoments as to the HVAC systems, and (ii) 30 days as to the remaining systems and oth the appropriate warranty period, correction of any such non-compliance, malfunction (except for the repairs to the fire sprinkler systems, roof, foundations, and/or bearing writin	"Guarantor"). (See also Paragraph 37) part of this Lease: ; ; ; ; ; ; ; ; ; ; ; ; ;
(b) Payment to Brokers. Upon execution and delivery of this Lease by the separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.12 Attachments. Attached hereto are the following, all of which constitute a land an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lecovenants and conditions set forth in this Lease. While the approximate square foote purposes of comparison, the Base Rent stated herein is NOT tied to square footage different. NOTE: Lessee is advised to verify the actual size prior to executing this Lecumentary of the commencement Date or the Early Possession Date, whichever first occurs ("Start Date below are obtained by Lessee and in effect within thirty days following the Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as a warranty exists as of the Start Date, or if one of such systems or elements should ma sole obligation with respect to such matter, except as otherwise provided in this Leas specificity the nature and extent of such non-compliance, malfunction or failure, recompniance, was an or the HVAC systems, and (ii) 30 days as to the remaining systems and other appropriate warranty period, correction of any such non-compl	"Guarantor"). (See also Paragraph 37) part of this Lease: ; ; ; ; ; ; ; ; ; ; ; ; ;
(b) Payment to Brokers. Upon execution and delivery of this Lease by the separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.12 Attachments. Attached hereto are the following, all of which constitute a land an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Project; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lecovenants and conditions set forth in this Lease. While the approximate square foote purposes of comparison, the Base Rent stated herein is NOT tied to square footage different. NOTE: Lessee is advised to verify the actual size prior to executing this Lecumentary of the commencement Date or the Early Possession Date, whichever first occurs ("Start Date below are obtained by Lessee and in effect within thirty days following the Start Date ventilating and air conditioning systems ("HVAC"), loading doors, sump pumps, if any shall be in good operating condition on said date, that the structural elements of the and that the Unit does not contain hazardous levels of any mold or fungi defined as a warranty exists as of the Start Date, or if one of such systems or elements should ma sole obligation with respect to such matter, except as otherwise provided in this Leas specificity the nature and extent of such non-compliance, malfunction or failure, recompniance, was an or the HVAC systems, and (ii) 30 days as to the remaining systems and other appropriate warranty period, correction of any such non-compl	"Guarantor"). (See also Paragraph 37) part of this Lease: ; ; ; ; ; ; ; ; ; ; ; ; ;
(b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guara 1.12 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan depicting the Premises; a site plan depicting the Premises; a current set of the Rules and Regulations for the Project; a current set of the Rules and Regulations adopted by the owners' associated a Work Letter other (specify):	"Guarantor"). (See also Paragraph 37) part of this Lease: ; ; ; ; ; ; ; ; ; ; ; ; ;

MTN-26.30, Revised 10-22-2020

be made by Lessee. NOTE: Lessee is responsible for determining whether or not the Applicable Requirements, and especially the zoning are appropriate for Lessee's intended use, and acknowledges that past uses of the Premises may no longer be allowed. If the Premises do not comply with said warranty, Lessor shall, except as otherwise provided, promptly after receipt of written notice from Lessee setting forth with specificity the nature and extent of such non-compliance, rectify the same at Lessor's expense. If Lessee does not give Lessor written notice of a non-compliance with this warranty within 6 months following the Start Date, correction of that non-compliance shall be the obligation of Lessee at Lessee's sole cost and expense. If the Applicable Requirements are hereafter changed so as to require during the term of this Lease the construction of an addition to or an alteration of the Unit, Premises and/or Building, the remediation of any Hazardous Substance, or the reinforcement or other physical modification of the Unit, Premises and/or Building ("Capital Expenditure"), Lessor and Lessee shall allocate the cost of such work as follows:

- (a) Subject to Paragraph 2.3(c) below, if such Capital Expenditures are required as a result of the specific and unique use of the Premises by Lessee as compared with uses by tenants in general. Lessee shall be fully responsible for the cost thereof, provided, however, that if such Capital Expenditure is required during the last 2 years of this Lease and the cost thereof exceeds 6 months' Base Rent, Lessee may instead terminate this Lease unless Lessor notifies Lessee, in writing, within 10 days after receipt of Lessee's termination notice that Lessor has elected to pay the difference between the actual cost thereof and the amount equal to 6 months' Base Rent. If Lessee elects termination, Lessee shall immediately cease the use of the Premises which requires such Capital Expenditure and deliver to Lessor written notice specifying a termination date at least 90 days thereafter. Such termination date shall, however, in no event be earlier than the last day that Lessee could legally utilize the Premises without commencing such Capital Expenditure.
- (b) If such Capital Expenditure is not the result of the specific and unique use of the Premises by Lessee (such as, governmentally mandated seismic modifications), then Lessor shall pay for such Capital Expenditure and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date that on which the Base Rent is due, an amount equal to 1/144th of the portion of such costs reasonably attributable to the Premises. Lessee shall pay Interest on the balance but may prepay its obligation at any time. If, however, such Capital Expenditure is required during the last 2 years of this Lease or if Lessor reasonably determines that it is not economically feasible to pay its share thereof. Lessor shall have the option to terminate this Lease upon 90 days prior written notice to Lessee unless Lessee notifies Lessor, in writing, within 10 days after receipt of Lessor's termination notice that Lessee will pay for such Capital Expenditure. If Lessor does not elect to terminate, and fails to tender its share of any such Capital Expenditure, Lessee may advance such funds and deduct same, with Interest, from Rent until Lessor's share of such costs have been fully paid. If Lessee is unable to finance Lessor's share, or if the balance of the Rent due and payable for the remainder of this Lease is not sufficient to fully reimburse Lessee on an offset basis, Lessee shall have the right to terminate this Lease upon 30 days written notice to Lessor.
- (c) Notwithstanding the above, the provisions concerning Capital Expenditures are intended to apply only to non-voluntary, unexpected, and new Applicable Requirements. If the Capital Expenditures are instead triggered by Lessee as a result of an actual or proposed change in use, change in intensity of use, or modification to the Premises then, and in that event, Lessee shall either: (i) immediately cease such changed use or intensity of use and/or take such other steps as may be necessary to eliminate the requirement for such Capital Expenditure, or (ii) complete such Capital Expenditure at its own expense. Lessee shall not have any right to terminate this Lease.
- 2.4 Acknowledgements. Lessee acknowledges that: (a) it has been given an opportunity to inspect and measure the Premises; (b) it has been advised by Lessor and/or Brokers to satisfy itself with respect to the size and condition of the Premises (including but not limited to the electrical, HVAC and fire sprinkler systems, security, environmental aspects, and compliance with Applicable Requirements and the Americans with Disabilities Act), and their suitability for Lessee's intended use; (c) Lessee has made such investigation as it deems necessary with reference to such matters and assumes all responsibility therefor as the same relate to its occupancy of the Premises; (d) it is not relying on any representation as to the size of the Premises made by Brokers or Lessor; (e) the square footage of the Premises was not material to Lessee's decision to lease the Premises and pay the Rent stated herein; and (f) neither Lessor, Lessor's agents, nor Brokers have made any oral or written representations or warranties with respect to said matters other than as set forth in this Lease. In addition, Lessor acknowledges that: (i) Brokers have made no representations, promises or warranties concerning Lessee's ability to honor the Lease or suitability to occupy the Premises and (ii) it is Lessor's sole responsibility to investigate the financial capability and/or suitability of all proposed tenants.
- 2.5 Lessee as Prior Owner/Occupant. The warranties made by Lessor in Paragraph 2 shall be of no force or effect if immediately prior to the Start Date Lessee was the owner or occupant of the Premises. In such event, Lessee shall be responsible for any necessary corrective work.

 2.6 Vehicle Parking. Lessee shall be entitled to use the number of Parking Spaces specified in Paragraph 1.2(b) on those portions of the Common Areas
- designated from time to time by Lessor for parking. Lessee shall not use more parking spaces than said number. Said parking spaces shall be used for parking by vehicles no larger than full-size passenger automobiles or pick-up trucks, herein called "Permitted Size Vehicles." Lessor may regulate the loading and unloading of vehicles by adopting Rules and Regulations as provided in Paragraph 2.9. No vehicles other than Permitted Size Vehicles may be parked in the Common Area without the prior written permission of Lessor. In addition:
- (a) Lessee shall not permit or allow any vehicles that belong to or are controlled by Lessee or Lessee's employees, suppliers, shippers, customers, contractors or invitees to be loaded, unloaded, or parked in areas other than those designated by Lessor for such activities.
 - (b) Lessee shall not service or store any vehicles in the Common Areas.
- If Lessee permits or allows any of the prohibited activities described in this Paragraph 2.6, then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove or tow away the vehicle involved and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.
- 2.7 Common Areas Definition. The term "Common Areas" is defined as all areas and facilities outside the Premises and within the exterior boundary line of the Project and interior utility raceways and installations within the Unit that are provided and designated by the Lessor from time to time for the general non-exclusive use of Lessor, Lessee and other tenants of the Project and their respective employees, suppliers, shippers, customers, contractors and invitees, including parking areas, loading and unloading areas, trash areas, roofs, roadways, walkways, driveways and landscaped areas.
- 2.8 Common Areas Lessee's Rights. Lessor grants to Lessee, for the benefit of Lessee and its employees, suppliers, shippers, contractors, customers and invitees, during the term of this Lease, the non-exclusive right to use, in common with others entitled to such use, the Common Areas as they exist from time to time, subject to any rights, powers, and privileges reserved by Lessor under the terms hereof or under the terms of any rules and regulations or restrictions governing the use of the Project. Under no circumstances shall the right herein granted to use the Common Areas be deemed to include the right to store any property, temporarily or permanently, in the Common Areas. Any such storage shall be permitted only by the prior written consent of Lessor or Lessor's designated agent, which consent may be revoked at any time. In the event that any unauthorized storage shall occur, then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove the property and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.

 2.9 **Common Areas - Rules and Regulations**. Lessor or such other person(s) as Lessor may appoint shall have the exclusive control and management of the
- Common Areas and shall have the right, from time to time, to establish, modify, amend and enforce reasonable rules and regulations ("Rules and Regulations") for the management, safety, care, and cleanliness of the grounds, the parking and unloading of vehicles and the preservation of good order, as well as for the convenience of other occupants or tenants of the Building and the Project and their invitees. Lessee agrees to abide by and conform to all such Rules and Regulations, and shall use its best efforts to cause its employees, suppliers, shippers, customers, contractors and invitees to so abide and conform. Lessor shall not be responsible to Lessee for the non-compliance with said Rules and Regulations by other tenants of the Project.
 - 2.10 Common Areas Changes. Lessor shall have the right, in Lessor's sole discretion, from time to time:
- (a) To make changes to the Common Areas, including, without limitation, changes in the location, size, shape and number of driveways, entrances, parking spaces, parking areas, loading and unloading areas, ingress, egress, direction of traffic, landscaped areas, walkways and utility raceways;
 - (b) To close temporarily any of the Common Areas for maintenance purposes so long as reasonable access to the Premises remains available;
 - To designate other land outside the boundaries of the Project to be a part of the Common Areas; (c)
 - (d) To add additional buildings and improvements to the Common Areas;
 - To use the Common Areas while engaged in making additional improvements, repairs or alterations to the Project, or any portion thereof; and (e)
- (f) To do and perform such other acts and make such other changes in, to or with respect to the Common Areas and Project as Lessor may, in the exercise of sound business judgment, deem to be appropriate.

Term

- 3.1 Term. The Commencement Date, Expiration Date and Original Term of this Lease are as specified in Paragraph 1.3.
- 3.2 Early Possession. Any provision herein granting Lessee Early Possession of the Premises is subject to and conditioned upon the Premises being available for such possession prior to the Commencement Date. Any grant of Early Possession only conveys a non-exclusive right to occupy the Premises. If Lessee totally or partially occupies the Premises prior to the Commencement Date, the obligation to pay Base Rent shall be abated for the period of such Early Possession. All other terms of this Lease (including but not limited to the obligations to pay Lessee's Share of Common Area Operating Expenses, Real Property Taxes and insurance premiums and to maintain the Premises) shall be in effect during such period. Any such Early Possession shall not affect the Expiration Date.
- 3.3 Delay In Possession. Lessor agrees to use commercially reasonable efforts to deliver exclusive possession of the Premises to Lessee by the Commencement Date. If, despite said efforts, Lessor is unable to deliver possession by such date, Lessor shall not be subject to any liability therefor, nor shall such failure affect the validity of this Lease or change the Expiration Date. Lessee shall not, however, be obligated to pay Rent or perform its other obligations until Lesson delivers possession of the Premises and any period of rent abatement that Lessee would otherwise have enjoyed shall run from the date of delivery of possession and

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

continue for a period equal to what Lessee would otherwise have enjoyed under the terms hereof, but minus any days of delay caused by the acts or omissions of Lessee. If possession is not delivered within 60 days after the Commencement Date, as the same may be extended under the terms of any Work Letter executed by Parties, Lessee may, at its option, by notice in writing within 10 days after the end of such 60 day period, cancel this Lease, in which event the Parties shall be discharged from all obligations hereunder. If such written notice is not received by Lessor within said 10 day period, Lessee's right to cancel shall terminate. If possession of the Premises is not delivered within 120 days after the Commencement Date, this Lease shall terminate unless other agreements are reached between Lessor and Lessee, in writing.

3.4 **Lessee Compliance**. Lessor shall not be required to tender possession of the Premises to Lessee until Lessee complies with its obligation to provide evidence of insurance (Paragraph 8.5). Pending delivery of such evidence, Lessee shall be required to perform all of its obligations under this Lease from and after the Start Date, including the payment of Rent, notwithstanding Lessor's election to withhold possession pending receipt of such evidence of insurance. Further, if Lessee is required to perform any other conditions prior to or concurrent with the Start Date, the Start Date shall occur but Lessor may elect to withhold possession until such conditions are satisfied.

4. Rent

- 4.1 Rent Defined. All monetary obligations of Lessee to Lessor under the terms of this Lease (except for the Security Deposit) are deemed to be rent ("Rent").
- 4.2 **Common Area Operating Expenses.** Lessee shall pay to Lessor during the term hereof, in addition to the Base Rent, Lessee's Share (as specified in Paragraph 1.6) of all Common Area Operating Expenses, as hereinafter defined, during each calendar year of the term of this Lease, in accordance with the following provisions:
- (a) "Common Area Operating Expenses" are defined, for purposes of this Lease, as all costs relating to the ownership and operation of the Project, including, but not limited to, the following:
 - (i) The operation, repair and maintenance, in neat, clean, good order and condition, and if necessary the replacement, of the following:
- (aa) The Common Areas and Common Area improvements, including parking areas, loading and unloading areas, trash areas, roadways, parkways, walkways, driveways, landscaped areas, bumpers, irrigation systems, Common Area lighting facilities, fences and gates, elevators, roofs, exterior walls of the buildings, building systems and roof drainage systems.
 - (bb) Exterior signs and any tenant directories.
 - (cc) Any fire sprinkler systems.
- (dd) All other areas and improvements that are within the exterior boundaries of the Project but outside of the Premises and/or any other space occupied by a tenant.
 - (ii) The cost of water, gas, electricity and telephone to service the Common Areas and any utilities not separately metered.
- (iii) The cost of trash disposal, pest control services, property management, security services, owners' association dues and fees, the cost to repaint the exterior of any structures and the cost of any environmental inspections.
 - (iv) Reserves set aside for maintenance, repair and/or replacement of Common Area improvements and equipment.
 - (v) Real Property Taxes (as defined in Paragraph 10).
 - (vi) The cost of the premiums for the insurance maintained by Lessor pursuant to Paragraph 8.
 - (vii) Any deductible portion of an insured loss concerning the Building or the Common Areas.
 - (viii) Auditors', accountants' and attorneys' fees and costs related to the operation, maintenance, repair and replacement of the Project.
- (ix) The cost of any capital improvement to the Building or the Project not covered under the provisions of Paragraph 2.3 provided; however, that Lessor shall allocate the cost of any such capital improvement over a 12 year period and Lessee shall not be required to pay more than Lessee's Share of 1/144th of the cost of such capital improvement in any given month. Lessee shall pay Interest on the unamortized balance but may prepay its obligation at any time.
 - (x) The cost of any other services to be provided by Lessor that are stated elsewhere in this Lease to be a Common Area Operating Expense.
- (b) Any Common Area Operating Expenses and Real Property Taxes that are specifically attributable to the Unit, the Building or to any other building in the Project or to the operation, repair and maintenance thereof, shall be allocated entirely to such Unit, Building, or other building. However, any Common Area Operating Expenses and Real Property Taxes that are not specifically attributable to the Building or to any other building or to the operation, repair and maintenance thereof, shall be equitably allocated by Lessor to all buildings in the Project.
- (c) The inclusion of the improvements, facilities and services set forth in Subparagraph 4.2(a) shall not be deemed to impose an obligation upon Lessor to either have said improvements or facilities or to provide those services unless the Project already has the same, Lessor already provides the services, or Lessor has agreed elsewhere in this Lease to provide the same or some of them.
- (d) Lessee's Share of Common Area Operating Expenses is payable monthly on the same day as the Base Rent is due hereunder. The amount of such payments shall be based on Lessor's estimate of the annual Common Area Operating Expenses. Within 60 days after written request (but not more than once each year) Lessor shall deliver to Lessee a reasonably detailed statement showing Lessee's Share of the actual Common Area Operating Expenses for the preceding year. If Lessee's payments during such year exceed Lessee's Share, Lessor shall credit the amount of such over-payment against Lessee's future payments. If Lessee's payments during such year were less than Lessee's Share, Lessee shall pay to Lessor the amount of the deficiency within 10 days after delivery by Lessor to Lessee of the statement.
- (e) Common Area Operating Expenses shall not include any expenses paid by any tenant directly to third parties, or as to which Lessor is otherwise reimbursed by any third party, other tenant, or insurance proceeds.
- 4.3 **Payment**. Lessee shall cause payment of Rent to be received by Lessor in lawful money of the United States, without offset or deduction (except as specifically permitted in this Lease), on or before the day on which it is due. All monetary amounts shall be rounded to the nearest whole dollar. In the event that any statement or invoice prepared by Lessor is inaccurate such inaccuracy shall not constitute a waiver and Lessee shall be obligated to pay the amount set forth in this Lease. Rent for any period during the term hereof which is for less than one full calendar month shall be prorated based upon the actual number of days of said month. Payment of Rent shall be made to Lessor at its address stated herein or to such other persons or place as Lessor may from time to time designate in writing. Acceptance of a payment which is less than the amount then due shall not be a waiver of Lessor's rights to the balance of such Rent, regardless of Lessor's endorsement of any check so stating. In the event that any check, draft, or other instrument of payment given by Lessee to Lessor is dishonored for any reason, Lessee agrees to pay to Lessor the sum of \$25 in addition to any Late Charge to compensate Lessor for additional time and expenses incurred in handling the dishonored payment and Lessor, at its option, may require all future Rent be paid by cashier's check. Payments will be applied first to accrued late charges and attorney's fees, second to accrued interest, then to Base Rent and Common Area Operating Expenses, and any remaining amount to any other outstanding charges or
- 5. Security Deposit. Lessee shall deposit with Lessor upon execution hereof the Security Deposit as security for Lessee's faithful performance of its obligations under this Lease. If Lessee fails to pay Rent, or otherwise Defaults under this Lease, Lessor may use, apply or retain all or any portion of said Security Deposit for the payment of any amount already due Lessor, for Rents which will be due in the future, and/ or to reimburse or compensate Lessor for any liability, expense, loss or damage which Lessor may suffer or incur by reason thereof. If Lessor uses or applies all or any portion of the Security Deposit, Lessee shall within 10 days after written request therefor deposit monies with Lessor sufficient to restore said Security Deposit to the full amount required by this Lease. If the Base Rent increases during the term of this Lease, Lessee shall, upon written request from Lessor, deposit additional monies with Lessor so that the total amount of the Security Deposit shall at all times bear the same proportion to the increased Base Rent as the initial Security Deposit bore to the initial Base Rent. Should the Agreed Use be amended to accommodate a material change in the business of Lessee or to accommodate a sublessee or assignee, Lessor shall have the right to increase the Security Deposit to the extent necessary, in Lessor's reasonable judgment, to account for any increased wear and tear that the Premises may suffer as a result thereof. If a change in control of Lessee occurs during this Lease and following such change the financial condition of Lesser's reasonable judgment, significantly reduced, Lessee shall deposit such additional monies with Lessor as shall be sufficient to cause the Security Deposit to be at a commercially reasonable level based on such change in financial condition. Lessor shall not be required to keep the Security Deposit separate from its general accounts. Within 90 days after the expiration or termination of this Lease, Lessor shall return that portion of the Security Deposi

6. Use.

6.1 **Use**. Lessee shall use and occupy the Premises only for the Agreed Use, or any other legal use which is reasonably comparable thereto, and for no other purpose. Lessee shall not use or permit the use of the Premises in a manner that is unlawful, creates damage, waste or a nuisance, or that disturbs occupants of or causes damage to neighboring premises or properties. Other than guide, signal and seeing eye dogs, Lessee shall not keep or allow in the Premises any pets, animals, birds, fish, or reptiles. Lessor shall not unreasonably withhold or delay its consent to any written request for a modification of the Agreed Use, so long as the same will not impair the structural integrity of the Building or the mechanical or electrical systems therein, and/or is not significantly more burdensome to the Project. If Lessor elects to withhold consent, Lessor shall within 7 days after such request give written notification of same, which notice shall include an explanation of Lessor's objections to the change in the Agreed Use.

6.2 Hazardous Substances.

ı	N	ΙT	ΊA	LS	
۰	٠.	٠.	., .		

- (a) Reportable Uses Require Consent. The term "Hazardous Substance" as used in this Lease shall mean any product, substance, or waste whose presence, use, manufacture, disposal, transportation, or release, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentially injurious to the public health, safety or welfare, the environment or the Premises, (ii) regulated or monitored by any governmental authority, or (iii) a basis for potential liability of Lessor to any governmental agency or third party under any applicable statute or common law theory. Hazardous Substances shall include, but not be limited to, hydrocarbons, petroleum, gasoline, and/or crude oil or any products, by-products or fractions thereof. Lessee shall not engage in any activity in or on the Premises which constitutes a Reportable Use of Hazardous Substances without the express prior written consent of Lessor and timely compliance (at Lessee's expense) with all Applicable Requirements. "Reportable Use" shall mean (i) the installation or use of any above or below ground storage tank, (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and/or (iii) the presence at the Premises of a Hazardous Substance with respect to which any Applicable Requirements requires that a notice be given to persons entering or occupying the Premises or neighboring properties. Notwithstanding the foregoing. Lessee may use any ordinary and customary materials reasonably required to be used in the normal course of the Agreed Use, ordinary office supplies (copier toner, liquid paper, glue, etc.) and common household cleaning materials, so long as such use is in compliance with all Applicable Requirements, is not a Reportable Use, and does not expose the Premises or neighboring property to any meaningful risk of contamination or damage or expose Lessor to any liability therefor. In addition, Lessor may condition its consent to any Reportable Use upon receiving such additional assurances as Lessor reasonably deems necessary to protect itself, the public, the Premises and/or the environment against damage, contamination, injury and/or liability, including, but not limited to, the installation (and removal on or before Lease expiration or termination) of protective modifications (such as concrete encasements) and/or increasing the Security Deposit.
- (b) **Duty to Inform Lessor**. If Lessee knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises, other than as previously consented to by Lessor, Lessee shall immediately give written notice of such fact to Lessor, and provide Lessor with a copy of any report, notice, claim or other documentation which it has concerning the presence of such Hazardous Substance.
- (c) Lessee Remediation. Lessee shall not cause or permit any Hazardous Substance to be spilled or released in, on, under, or about the Premises (including through the plumbing or sanitary sewer system) and shall promptly, at Lessee's expense, comply with all Applicable Requirements and take all investigatory and/or remedial action reasonably recommended, whether or not formally ordered or required, for the cleanup of any contamination of, and for the maintenance, security and/or monitoring of the Premises or neighboring properties, that was caused or materially contributed to by Lessee, or pertaining to or involving any Hazardous Substance brought onto the Premises during the term of this Lease, by or for Lessee, or any third party.
- (d) Lessee Indemnification. Lessee shall indemnify, defend and hold Lessor, its agents, employees, lenders and ground lessor, if any, harmless from and against any and all loss of rents and/or damages, liabilities, judgments, claims, expenses, penalties, and attorneys' and consultants' fees arising out of or involving any Hazardous Substance brought onto the Premises by or for Lessee, or any third party (provided, however, that Lessee shall have no liability under this Lease with respect to underground migration of any Hazardous Substance under the Premises from areas outside of the Project not caused or contributed to by Lessee). Lessee's obligations shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Lessee, and the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease. No termination, cancellation or release agreement entered into by Lessor and Lessee shall release Lessee from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Lessor in writing at the time of such agreement.
- (e) Lessor Indemnification. Except as otherwise provided in paragraph 8.7, Lessor and its successors and assigns shall indemnify, defend, reimburse and hold Lessee, its employees and lenders, harmless from and against any and all environmental damages, including the cost of remediation, which are suffered as a direct result of Hazardous Substances on the Premises prior to Lessee taking possession or which are caused by the gross negligence or willful misconduct of Lessor, its agents or employees. Lessor's obligations, as and when required by the Applicable Requirements, shall include, but not be limited to, the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease.
- (f) Investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by governmental entities having jurisdiction with respect to the existence of Hazardous Substances on the Premises prior to the Lessee taking possession, unless such remediation measure is required as a result of Lessee's use (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises, in which event Lessee shall be responsible for such payment. Lesser's agents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remedial responsibilities.
- (g) Lessor Termination Option. If a Hazardous Substance Condition (see Paragraph 9.1(e)) occurs during the term of this Lease, unless Lessee is legally responsible therefor (in which case Lessee shall make the investigation and remediation thereof required by the Applicable Requirements and this Lease shall continue in full force and effect, but subject to Lessor's rights under Paragraph 6.2(d) and Paragraph 13), Lessor may, at Lessor's option, either (i) investigate and remediate such Hazardous Substance Condition, if required, as soon as reasonably possible at Lessor's expense, in which event this Lease shall continue in full force and effect, or (ii) if the estimated cost to remediate such condition exceeds 12 times the then monthly Base Rent or \$100,000, whichever is greater, give written notice to Lessee, within 30 days after receipt by Lessor of knowledge of the occurrence of such Hazardous Substance Condition, of Lessor's desire to terminate this Lease as of the date 60 days following the date of such notice. In the event Lessor elects to give a termination notice, Lessee may, within 10 days thereafter, give written notice to Lessor of Lessee's commitment to pay the amount by which the cost of the remediation of such Hazardous Substance Condition exceeds an amount equal to 12 times the then monthly Base Rent or \$100,000, whichever is greater. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days following such commitment. In such event, this Lease shall continue in full force and effect, and Lessor shall proceed to make such remediation as soon as reasonably possible after the required funds are available. If Lessee does not give such notice and provide the required funds or assurance thereof within the time provided, this Lease shall terminate as of the date specified in Lessor's notice of termination.

 6.3 Lessee's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lessee's sole expense, fully, diligently and
- 6.3 Lessee's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lessee's sole expense, fully, diligently and in a timely manner, materially comply with all Applicable Requirements, the requirements of any applicable fire insurance underwriter or rating bureau, and the recommendations of Lessor's engineers and/or consultants which relate in any manner to the Premises, without regard to whether said Applicable Requirements are now in effect or become effective after the Start Date. Lessee shall, within 10 days after receipt of Lessor's written request, provide Lessor with copies of all permits and other documents, and other information evidencing Lessee's compliance with any Applicable Requirements specified by Lessor, and shall immediately upon receipt, notify Lessor in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving the failure of Lessee or the Premises to comply with any Applicable Requirements. Likewise, Lessee shall immediately give written notice to Lessor of: (i) any water damage to the Premises and any suspected seepage, pooling, dampness or other condition conducive to the production of mold; or (ii) any mustiness or other odors that might indicate the presence of mold in the Premises.

 6.4 Inspection; Compliance. Lessor and Lessor's "Lender" (as defined in Paragraph 30) and consultants authorized by Lessor shall have the right to enter into
- 6.4 Inspection; Compliance. Lessor and Lessor's "Lender" (as defined in Paragraph 30) and consultants authorized by Lessor shall have the right to enter into Premises at any time in the case of an emergency, and otherwise at reasonable times after reasonable notice, for the purpose of inspecting and/or testing the condition of the Premises and/or for verifying compliance by Lessee with this Lease. The cost of any such inspections shall be paid by Lessor, unless a violation of Applicable Requirements, or a Hazardous Substance Condition (see Paragraph 9.1(e)) is found to exist or be imminent, or the inspection is requested or ordered by a governmental authority. In such case, Lessee shall upon request reimburse Lessor for the cost of such inspection, so long as such inspection is reasonably related to the violation or contamination. In addition, Lessee shall provide copies of all relevant material safety data sheets (MSDS) to Lessor within 10 days of the receipt of written request therefor. Lessee acknowledges that any failure on its part to allow such inspections or testing will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to allow such inspections and/or testing in a timely fashion the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for the remainder to the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to allow such inspection and/or testing. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to such failure nor prevent the exercise of any of the other rights and remedies granted hereunder.

7. Maintenance; Repairs; Utility Installations; Trade Fixtures and Alterations.

7.1 Lessee's Obligations.

- (a) In General. Subject to the provisions of Paragraph 2.2 (Condition), 2.3 (Compliance), 6.3 (Lessee's Compliance with Applicable Requirements), 7.2 (Lessor's Obligations), 9 (Damage or Destruction), and 14 (Condemnation), Lessee shall, at Lessee's sole expense, keep the Premises, Utility Installations (intended for Lessee's exclusive use, no matter where located), and Alterations in good order, condition and repair (whether or not the portion of the Premises requiring repairs, or the means of repairing the same, are reasonably or readily accessible to Lessee, and whether or not the need for such repairs occurs as a result of Lessee's use, any prior use, the elements or the age of such portion of the Premises), including, but not limited to, all equipment or facilities, such as plumbing, HVAC equipment, electrical, lighting facilities, boilers, pressure vessels, fixtures, interior walls, interior surfaces of exterior walls, ceilings, floors, windows, doors, plate glass, and skylights but excluding any items which are the responsibility of Lessor pursuant to Paragraph 7.2. Lessee, in keeping the Premises in good order, condition and repair, shall exercise and perform good maintenance practices, specifically including the procurement and maintenance of the service contracts required by Paragraph 7.1(b) below. Lessee's obligations shall include restorations, replacements or renewals when necessary to keep the Premises and all improvements thereon or a part thereof in good order, condition and state of repair.
 - (b) Service Contracts. Lessee shall, at Lessee's sole expense, procure and maintain contracts, with copies to Lessor, in customary form and substance for,

and with contractors specializing and experienced in the maintenance of the following equipment and improvements, if any, if and when installed on the Premises: (i) HVAC equipment, (ii) boiler and pressure vessels, and (iii) clarifiers. However, Lessor reserves the right, upon notice to Lessee, to procure and maintain any or all of such service contracts, and Lessee shall reimburse Lessor, upon demand, for the cost thereof.

- (c) Failure to Perform. If Lessee fails to perform Lessee's obligations under this Paragraph 7.1, Lessor may enter upon the Premises after 10 days' prior written notice to Lessee (except in the case of an emergency, in which case no notice shall be required), perform such obligations on Lessee's behalf, and put the Premises in good order, condition and repair, and Lessee shall promptly pay to Lessor a sum equal to 115% of the cost thereof.
- (d) **Replacement**. Subject to Lessee's indemnification of Lessor as set forth in Paragraph 8.7 below, and without relieving Lessee of liability resulting from Lessee's failure to exercise and perform good maintenance practices, if an item described in Paragraph 7.1(b) cannot be repaired other than at a cost which is in excess of 50% of the cost of replacing such item, then such item shall be replaced by Lessor, and the cost thereof shall be prorated between the Parties and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date on which Base Rent is due, an amount equal to the product of multiplying the cost of such replacement by a fraction, the numerator of which is one, and the denominator of which is 144 (i.e. 1/144th of the cost per month). Lessee shall pay Interest on the unamortized balance but may prepay its obligation at any time.
- 7.2 **Lessor's Obligations**. Subject to the provisions of Paragraphs 2.2 (Condition), 2.3 (Compliance), 4.2 (Common Area Operating Expenses), 6 (Use), 7.1 (Lessee's Obligations), 9 (Damage or Destruction) and 14 (Condemnation), Lessor, subject to reimbursement pursuant to Paragraph 4.2, shall keep in good order, condition and repair the foundations, exterior walls, structural condition of interior bearing walls, exterior roof, fire sprinkler system, Common Area fire alarm and/or smoke detection systems, fire hydrants, parking lots, walkways, parkways, driveways, landscaping, fences, signs and utility systems serving the Common Areas and all parts thereof, as well as providing the services for which there is a Common Area Operating Expense pursuant to Paragraph 4.2. Lessor shall not be obligated to paint the exterior or interior surfaces of exterior walls nor shall Lessor be obligated to maintain, repair or replace windows, doors or plate glass of the Premises.

7.3 Utility Installations: Trade Fixtures: Alterations.

- (a) **Definitions.** The term "**Utility Installations**" refers to all floor and window coverings, air and/or vacuum lines, power panels, electrical distribution, security and fire protection systems, communication cabling, lighting fixtures, HVAC equipment, plumbing, and fencing in or on the Premises. The term "**Trade Fixtures**" shall mean Lessee's machinery and equipment that can be removed without doing material damage to the Premises. The term "Alterations" shall mean any modification of the improvements, other than Utility Installations or Trade Fixtures, whether by addition or deletion. "**Lessee Owned Alterations and/or Utility Installations**" are defined as Alterations and/or Utility Installations made by Lessee that are not yet owned by Lessor pursuant to Paragraph 7.4(a).
- (b) Consent. Lessee shall not make any Alterations or Utility Installations to the Premises without Lessor's prior written consent. Lessee may, however, make non-structural Alterations or Utility Installations to the interior of the Premises (excluding the roof) without such consent but upon notice to Lessor, as long as they are not visible from the outside, do not involve puncturing, relocating or removing the roof or any existing walls, will not affect the electrical, plumbing, HVAC, and/or life safety systems, do not trigger the requirement for additional modifications and/or improvements to the Premises resulting from Applicable Requirements, such as compliance with Title 24, and/or life safety systems, and the cumulative cost thereof during this Lease as extended does not exceed a sum equal to 3 month's Base Rent in the aggregate or a sum equal to one month's Base Rent in any one year. Notwithstanding the foregoing, Lessee shall not make or permit any roof penetrations and/or install anything on the roof without the prior written approval of Lessor. Lessor may, as a precondition to granting such approval, require Lessee to utilize a contractor chosen and/or approved by Lessor. Any Alterations or Utility Installations that Lessee shall desire to make and which require the consent of the Lessor shall be presented to Lessor in written form with detailed plans. Consent shall be deemed conditioned upon Lessee's. (i) acquiring all applicable governmental permits, (ii) furnishing Lessor with copies of both the permits and the plans and specifications prior to commencement of the work, and (iii) compliance with all conditions of said permits and other Applicable Requirements in a prompt and expeditious manner. Any Alterations or Utility Installations shall be performed in a workmanlike manner with good and sufficient materials. Lessee shall promptly upon completion furnish Lessor with as-built plans and specifications. For work which costs an amount in excess of one month's Base Rent, Lessor may condition its cons
- (c) Liens; Bonds. Lessee shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lessee at or for use on the Premises, which claims are or may be secured by any mechanic's or materialmen's lien against the Premises or any interest therein. Lessee shall give Lessor not less than 10 days notice prior to the commencement of any work in, on or about the Premises, and Lessor shall have the right to post notices of non-responsibility. If Lessee shall contest the validity of any such lien, claim or demand, then Lessee shall, at its sole expense defend and protect itself, Lessor and the Premises against the same and shall pay and satisfy any such adverse judgment that may be rendered thereon before the enforcement thereof. If Lessor shall require, Lessee shall furnish a surety bond in an amount equal to 150% of the amount of such contested lien, claim or demand, indemnifying Lessor against liability for the same. If Lessor elects to participate in any such action, Lessee shall pay Lessor's attorneys' fees and costs.

7.4 Ownership; Removal; Surrender; and Restoration.

- (a) **Ownership**. Subject to Lessor's right to require removal or elect ownership as hereinafter provided, all Alterations and Utility Installations made by Lessee shall be the property of Lessee, but considered a part of the Premises. Lessor may, at any time, elect in writing to be the owner of all or any specified part of the Lessee Owned Alterations and Utility Installations. Unless otherwise instructed per paragraph 7.4(b) hereof, all Lessee Owned Alterations and Utility Installations shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.
- shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.

 (b) Removal. By delivery to Lessee of written notice from Lessor not earlier than 90 and not later than 30 days prior to the end of the term of this Lease, Lessor may require that any or all Lessee Owned Alterations or Utility Installations be removed by the expiration or termination of this Lease. Lessor may require the removal at any time of all or any part of any Lessee Owned Alterations or Utility Installations made without the required consent.

 (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements,
- (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements, parts and surfaces thereof broom clean and free of debris, and in good operating order, condition and state of repair, ordinary wear and tear excepted. "Ordinary wear and tear" shall not include any damage or deterioration that would have been prevented by good maintenance practice. Notwithstanding the foregoing and the provisions of Paragraph 7.1(a), if the Lessee occupies the Premises for 12 months or less, then Lessee shall surrender the Premises in the same condition as delivered to Lessee on the Start Date with NO allowance for ordinary wear and tear. Lessee shall repair any damage occasioned by the installation, maintenance or removal of Trade Fixtures, Lessee owned Alterations and/or Utility Installations, furnishings, and equipment as well as the removal of any storage tank installed by or for Lessee. Lessee shall also remove from the Premises any and all Hazardous Substances brought onto the Premises by or for Lessee, or any third party (except Hazardous Substances which were deposited via underground migration from areas outside of the Project) to the level specified in Applicable Requirements. Trade Fixtures shall remain the property of Lessee and shall be removed by Lessee. Any personal property of Lessee not removed on or before the Expiration Date or any earlier termination date shall be deemed to have been abandoned by Lessee and may be disposed of or retained by Lessor as Lessor may desire. The failure by Lessee to timely vacate the Premises pursuant to this Paragraph 7.4(c) without the express written consent of Lessor shall constitute a holdover under the provisions of Paragraph 26 below.

8. Insurance; Indemnity.

8.1 **Payment of Premiums.** The cost of the premiums for the insurance policies required to be carried by Lessor, pursuant to Paragraphs 8.2(b), 8.3(a) and 8.3(b), shall be a Common Area Operating Expense. Premiums for policy periods commencing prior to, or extending beyond, the term of this Lease shall be prorated to coincide with the corresponding Start Date or Expiration Date.

8.2 Liability Insurance.

- (a) Carried by Lessee. Lessee shall obtain and keep in force a Commercial General Liability policy of insurance protecting Lessee and Lessor as an additional insured against claims for bodily injury, personal injury and property damage based upon or arising out of the ownership, use, occupancy or maintenance of the Premises and all areas appurtenant thereto. Such insurance shall be on an occurrence basis providing single limit coverage in an amount not less than \$1,000,000 per occurrence with an annual aggregate of not less than \$2,000,000. Lessee shall add Lessor as an additional insured by means of an endorsement at least as broad as the Insurance Service Organization's "Additional Insured-Managers or Lessors of Premises" Endorsement. The policy shall not contain any intra-insured exclusions as between insured persons or organizations, but shall include coverage for liability assumed under this Lease as an "insured contract" for the performance of Lessee's indemnity obligations under this Lease. The limits of said insurance shall not, however, limit the liability of Lessee nor relieve Lessee of any obligation hereunder. Lessee shall provide an endorsement on its liability policy(ies) which provides that its insurance shall be primary to and not contributory with any similar insurance carried by Lessor, whose insurance shall be considered excess insurance only.
- (b) **Carried by Lessor**. Lessor shall maintain liability insurance as described in Paragraph 8.2(a), in addition to, and not in lieu of, the insurance required to be maintained by Lessee. Lessee shall not be named as an additional insured therein.

8.3 Property Insurance - Building, Improvements and Rental Value.

(a) **Building and Improvements.** Lessor shall obtain and keep in force a policy or policies of insurance in the name of Lessor, with loss payable to Lessor, any ground-lessor, and to any Lender insuring loss or damage to the Premises. The amount of such insurance shall be equal to the full insurable replacement cost of the Premises, as the same shall exist from time to time, or the amount required by any Lender, but in no event more than the commercially reasonable and available insurable value thereof. Lessee Owned Alterations and Utility Installations, Trade Fixtures, and Lessee's personal property shall be insured by Lessee not by Lessor. If the coverage is available and commercially appropriate, such policy or policies shall insure against all risks of direct physical loss or damage (except the perils of flood and/or earthquake unless required by a Lender), including coverage for debris removal and the enforcement of any Applicable Requirements requiring the upgrading, demolition, reconstruction or replacement of any portion of the Premises as the result of a covered loss. Said policy or policies shall also contain an agreed valuation

ll l	NITIALS

Last Edited: 9/30/2022 2:33 PM

INITIALS

provision in lieu of any coinsurance clause, waiver of subrogation, and inflation guard protection causing an increase in the annual property insurance coverage amount by a factor of not less than the adjusted U.S. Department of Labor Consumer Price Index for All Urban Consumers for the city nearest to where the Premises are located. If such insurance coverage has a deductible clause, the deductible amount shall not exceed \$5,000 per occurrence.

- (b) **Rental Value**. Lessor shall also obtain and keep in force a policy or policies in the name of Lessor with loss payable to Lessor and any Lender, insuring the loss of the full Rent for one year with an extended period of indemnity for an additional 180 days ("Rental Value insurance"). Said insurance shall contain an agreed valuation provision in lieu of any coinsurance clause, and the amount of coverage shall be adjusted annually to reflect the projected Rent otherwise payable by Lessee, for the next 12 month period.
- (c) Adjacent Premises. Lessee shall pay for any increase in the premiums for the property insurance of the Building and for the Common Areas or other buildings in the Project if said increase is caused by Lessee's acts, omissions, use or occupancy of the Premises.
- (d) Lessee's Improvements. Since Lessor is the Insuring Party, Lessor shall not be required to insure Lessee Owned Alterations and Utility Installations unless the item in question has become the property of Lessor under the terms of this Lease.

8.4 Lessee's Property; Business Interruption Insurance; Worker's Compensation Insurance.

- (a) **Property Damage**. Lessee shall obtain and maintain insurance coverage on all of Lessee's personal property, Trade Fixtures, and Lessee Owned Alterations and Utility Installations. Such insurance shall be full replacement cost coverage with a deductible of not to exceed \$1,000 per occurrence. The proceeds from any such insurance shall be used by Lessee for the replacement of personal property, Trade Fixtures and Lessee Owned Alterations and Utility Installations.
- (b) **Business Interruption**. Lessee shall obtain and maintain loss of income and extra expense insurance in amounts as will reimburse Lessee for direct or indirect loss of earnings attributable to all perils commonly insured against by prudent lessees in the business of Lessee or attributable to prevention of access to the Premises as a result of such perils.
- (c) **Worker's Compensation Insurance**. Lessee shall obtain and maintain Worker's Compensation Insurance in such amount as may be required by Applicable Requirements. Such policy shall include a 'Waiver of Subrogation' endorsement. Lessee shall provide Lessor with a copy of such endorsement along with the certificate of insurance or copy of the policy required by paragraph 8.5.
- (d) No Representation of Adequate Coverage. Lessor makes no representation that the limits or forms of coverage of insurance specified herein are adequate to cover Lessee's property, business operations or obligations under this Lease.
- 8.5 **Insurance Policies**. Insurance required herein shall be by companies maintaining during the policy term a "General Policyholders Rating" of at least A-, VII, as set forth in the most current issue of "Best's Insurance Guide", or such other rating as may be required by a Lender. Lessee shall not do or permit to be done anything which invalidates the required insurance policies. Lessee shall, prior to the Start Date, deliver to Lessor certified copies of policies of such insurance or certificates with copies of the required endorsements evidencing the existence and amounts of the required insurance. No such policy shall be cancelable or subject to modification except after 30 days prior written notice to Lessor. Lessee shall, at least 10 days prior to the expiration of such policies, furnish Lessor with evidence of renewals or "insurance binders" evidencing renewal thereof, or Lessor may increase his liability insurance coverage and charge the cost thereof to Lessee, which amount shall be payable by Lessee to Lessor upon demand. Such policies shall be for a term of at least one year, or the length of the remaining term of this Lease, whichever is less. If either Party shall fail to procure and maintain the insurance required to be carried by it, the other Party may, but shall not be required to, procure and maintain the same.
- 8.6 Waiver of Subrogation. Without affecting any other rights or remedies, Lessee and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is not limited by the amount of insurance carried or required, or by any deductibles applicable hereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Lessee, as the case may be, so long as the insurance is not invalidated thereby.
- 8.7 Indemnity. Except for Lessor's gross negligence or willful misconduct, Lessee shall indemnify, protect, defend and hold harmless the Premises, Lessor and its agents, Lessor's master or ground lessor, partners and Lenders, from and against any and all claims, loss of rents and/or damages, liens, judgments, penalties, attorneys' and consultants' fees, expenses and/or liabilities arising out of, involving, or in connection with, a Breach of the Lease by Lessee and/or the use and/or occupancy of the Premises and/or Project by Lessee and/or by Lessee's employees, contractors or invitees. If any action or proceeding is brought against Lessor by reason of any of the foregoing matters, Lessee shall upon notice defend the same at Lessee's expense by counsel reasonably satisfactory to Lessor and Lessor shall cooperate with Lessee in such defense. Lessor need not have first paid any such claim in order to be defended or indemnified.
- 8.8 Exemption of Lessor and its Agents from Liability. Notwithstanding the negligence or breach of this Lease by Lessor or its agents, neither Lessor nor its agents shall be liable under any circumstances for: (i) injury or damage to the person or goods, wares, merchandise or other property of Lessee, Lessee's employees, contractors, invitees, customers, or any other person in or about the Premises, whether such damage or injury is caused by or results from fire, steam, electricity, gas, water or rain, indoor air quality, the presence of mold or from the breakage, leakage, obstruction or other defects of pipes, fire sprinklers, wires, appliances, plumbing, HVAC or lighting fixtures, or from any other cause, whether the said injury or damage results from conditions arising upon the Premises or upon other portions of the Building, or from other sources or places; (ii) any damages arising from any act or neglect of any other tenant of Lessor or from the failure of Lessor or its agents to enforce the provisions of any other lease in the Project; or (iii) injury to Lessee's business or for any loss of income or profit therefrom. Instead, it is intended that Lessee's sole recourse in the event of such damages or injury be to file a claim on the insurance policy(ies) that Lessee is required to maintain pursuant to the provisions of paragraph 8.
- 8.9 Failure to Provide Insurance. Lessee acknowledges that any failure on its part to obtain or maintain the insurance required herein will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, for any month or portion thereof that Lessee does not maintain the required insurance and/or does not provide Lessor with the required binders or certificates evidencing the existence of the required insurance, the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater. The parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to maintain the required insurance. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to maintain such insurance, prevent the exercise of any of the other rights and remedies granted hereunder, nor relieve Lessee of its obligation to maintain the insurance specified in this Lease.

9. Damage or Destruction.

9.1 **Definitions**.

- (a) "Premises Partial Damage" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations, which can reasonably be repaired in 3 months or less from the date of the damage or destruction, and the cost thereof does not exceed a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (b) "Premises Total Destruction" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which cannot reasonably be repaired in 3 months or less from the date of the damage or destruction and/or the cost thereof exceeds a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (c) "Insured Loss" shall mean damage or destruction to improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which was caused by an event required to be covered by the insurance described in Paragraph 8.3(a), irrespective of any deductible amounts or coverage limits involved.
- (d) "Replacement Cost" shall mean the cost to repair or rebuild the improvements owned by Lessor at the time of the occurrence to their condition existing immediately prior thereto, including demolition, debris removal and upgrading required by the operation of Applicable Requirements, and without deduction for depreciation.
- (e) "Hazardous Substance Condition" shall mean the occurrence or discovery of a condition involving the presence of, or a contamination by, a Hazardous Substance, in, on, or under the Premises which requires restoration.
- 9.2 Partial Damage Insured Loss. If a Premises Partial Damage that is an Insured Loss occurs, then Lessor shall, at Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonably possible and this Lease shall continue in full force and effect; provided, however, that Lessee shall, at Lessor's election, make the repair of any damage or destruction the total cost to repair of which is \$10,000 or less, and, in such event, Lessor shall make any applicable insurance proceeds available to Lessee on a reasonable basis for that purpose. Notwithstanding the foregoing, if the required insurance was not in force or the insurance proceeds are not sufficient to effect such repair, the Insuring Party shall promptly contribute the shortage in proceeds as and when required to complete said repairs. In the event, however, such shortage was due to the fact that, by reason of the unique nature of the improvements, full replacement cost insurance coverage was not commercially reasonable and available, Lessor shall have no obligation to pay for the shortage in insurance proceeds or to fully restore the unique aspects of the Premises unless Lessee provides Lessor with the funds to cover same, or adequate assurance thereof, within 10 days following receipt of written notice of such shortage and request therefor. If Lessor receives said funds or adequate assurance thereof within said 10 day period, the party responsible for making the repairs shall complete them as soon as reasonably possible and this Lease shall remain in full force and effect. If such funds or assurance are not received, Lessor may nevertheless elect by written notice to Lessee within 10 days thereafter to: (i) make such restoration and repair as is commercially reasonable with Lessor paying any shortage in proceeds, in which case this Lease shall remain in full force and effect, or (ii) have this Lease terminate 30

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

days thereafter. Lessee shall not be entitled to reimbursement of any funds contributed by Lessee to repair any such damage or destruction. Premises Partial Damage due to flood or earthquake shall be subject to Paragraph 9.3, notwithstanding that there may be some insurance coverage, but the net proceeds of any such insurance shall be made available for the repairs if made by either Party.

- 9.3 Partial Damage Uninsured Loss. If a Premises Partial Damage that is not an Insured Loss occurs, unless caused by a negligent or willful act of Lessee (in which event Lessee shall make the repairs at Lessee's expense), Lessor may either: (i) repair such damage as soon as reasonably possible at Lessor's expense (subject to reimbursement pursuant to Paragraph 4.2), in which event this Lease shall continue in full force and effect, or (ii) terminate this Lease by giving written notice to Lessee within 30 days after receipt by Lessor of knowledge of the occurrence of such damage. Such termination shall be effective 60 days following the date of such notice. In the event Lessor elects to terminate this Lease, Lessee shall have the right within 10 days after receipt of the termination notice to give written notice to Lessor of Lessee's commitment to pay for the repair of such damage without reimbursement from Lessor. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days after making such commitment. In such event this Lease shall continue in full force and effect, and Lessor shall proceed to make such repairs as soon as reasonably possible after the required funds are available. If Lessee does not make the required commitment, this Lease shall terminate as of the date specified in the termination notice.
- 9.4 Total Destruction. Notwithstanding any other provision hereof, if a Premises Total Destruction occurs, this Lease shall terminate 60 days following such Destruction. If the damage or destruction was caused by the gross negligence or willful misconduct of Lessee, Lessor shall have the right to recover Lessor's damages from Lessee, except as provided in Paragraph 8.6.
- 9.5 Damage Near End of Term. If at any time during the last 6 months of this Lease there is damage for which the cost to repair exceeds one month's Base Rent, whether or not an Insured Loss, Lessor may terminate this Lease effective 60 days following the date of occurrence of such damage by giving a written termination notice to Lessee within 30 days after the date of occurrence of such damage. Notwithstanding the foregoing, if Lessee at that time has an exercisable option to extend this Lease or to purchase the Premises, then Lessee may preserve this Lease by (a) exercising such option and (b) providing Lessor with any shortage in insurance proceeds (or adequate assurance thereof) needed to make the repairs on or before the earlier of (i) the date which is 10 days after Lessee's receipt of Lessor's written notice purporting to terminate this Lease, or (ii) the day prior to the date upon which such option expires. If Lessee duly exercises such option during such period and provides Lessor with funds (or adequate assurance thereof) to cover any shortage in insurance proceeds, Lessor shall, at Lessor's commercially reasonable expense, repair such damage as soon as reasonably possible and this Lease shall continue in full force and effect. If Lessee fails to exercise such option and provide such funds or assurance during such period, then this Lease shall terminate on the date specified in the termination notice and Lessee's option shall be extinguished.

9.6 Abatement of Rent; Lessee's Remedies.

- (a) Abatement. In the event of Premises Partial Damage or Premises Total Destruction or a Hazardous Substance Condition for which Lessee is not responsible under this Lease, the Rent payable by Lessee for the period required for the repair, remediation or restoration of such damage shall be abated in proportion to the degree to which Lessee's use of the Premises is impaired, but not to exceed the proceeds received from the Rental Value insurance. All other obligations of Lessee hereunder shall be performed by Lessee, and Lessor shall have no liability for any such damage, destruction, remediation, repair or restoration except as provided herein.
- (b) Remedies. If Lessor is obligated to repair or restore the Premises and does not commence, in a substantial and meaningful way, such repair or restoration within 90 days after such obligation shall accrue, Lessee may, at any time prior to the commencement of such repair or restoration, give written notice to Lessor and to any Lenders of which Lessee has actual notice, of Lessee's election to terminate this Lease on a date not less than 60 days following the giving of such notice. If Lessee gives such notice and such repair or restoration is not commenced within 30 days thereafter, this Lease shall terminate as of the date specified in said notice. If the repair or restoration is commenced within such 30 days, this Lease shall continue in full force and effect. "Commence" shall mean either the unconditional authorization of the preparation of the required plans, or the beginning of the actual work on the Premises, whichever first occurs.
- 9.7 Termination; Advance Payments. Upon termination of this Lease pursuant to Paragraph 6.2(g) or Paragraph 9, an equitable adjustment shall be made concerning advance Base Rent and any other advance payments made by Lessee to Lessor. Lessor shall, in addition, return to Lessee so much of Lessee's Security Deposit as has not been, or is not then required to be, used by Lessor.

Real Property Taxes.

- 10.1 Definition. As used herein, the term "Real Property Taxes" shall include any form of assessment; real estate, general, special, ordinary or extraordinary, or rental levy or tax (other than inheritance, personal income or estate taxes); improvement bond; and/or license fee imposed upon or levied against any legal or equitable interest of Lessor in the Project, Lessor's right to other income therefrom, and/or Lessor's business of leasing, by any authority having the direct or indirect power to tax and where the funds are generated with reference to the Project address. The term "Real Property Taxes" shall also include any tax, fee, levy, assessment or charge, or any increase therein: (i) imposed by reason of events occurring during the term of this Lease, including but not limited to, a change in the ownership of the Project, (ii) a change in the improvements thereon, and/or (iii) levied or assessed on machinery or equipment provided by Lessor to Lessee pursuant to this Lease. In calculating Real Property Taxes for any calendar year, the Real Property Taxes for any real estate tax year shall be included in the calculation of Real Property Taxes for such calendar year based upon the number of days which such calendar year and tax year have in common.

 10.2 Payment of Taxes. Except as otherwise provided in Paragraph 10.3, Lessor shall pay the Real Property Taxes applicable to the Project, and said payments shall be included in the calculation of Common Area Operating Expenses in accordance with the provisions of Paragraph 4.2.
- 10.3 Additional Improvements. Common Area Operating Expenses shall not include Real Property Taxes specified in the tax assessor's records and work sheets as being caused by additional improvements placed upon the Project by other lessees or by Lessor for the exclusive enjoyment of such other lessees. Notwithstanding Paragraph 10.2 hereof, Lessee shall, however, pay to Lessor at the time Common Area Operating Expenses are payable under Paragraph 4.2, the entirety of any increase in Real Property Taxes if assessed solely by reason of Alterations, Trade Fixtures or Utility Installations placed upon the Premises by Lessee or at Lessee's request or by reason of any alterations or improvements to the Premises made by Lessor subsequent to the execution of this Lease by the Parties.
- 10.4 Joint Assessment. If the Building is not separately assessed, Real Property Taxes allocated to the Building shall be an equitable proportion of the Real Property Taxes for all of the land and improvements included within the tax parcel assessed, such proportion to be determined by Lessor from the respective valuations assigned in the assessor's work sheets or such other information as may be reasonably available. Lessor's reasonable determination thereof, in good faith, shall be conclusive
- 10.5 Personal Property Taxes. Lessee shall pay prior to delinquency all taxes assessed against and levied upon Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all personal property of Lessee contained in the Premises. When possible, Lessee shall cause its Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all other personal property to be assessed and billed separately from the real property of Lessor. If any of Lessee's said property shall be assessed with Lessor's real property, Lessee shall pay Lessor the taxes attributable to Lessee's property within 10 days after receipt of a written statement setting forth the taxes applicable to Lessee's property.

11. Utilities and Services.

- 11.1 Lessee shall pay for all water, gas, heat, light, power, telephone, trash disposal and other utilities and services supplied to the Premises, together with any taxes thereon. Notwithstanding the provisions of Paragraph 4.2, if at any time in Lessor's sole judgment, Lessor determines that Lessee is using a disproportionate amount of water, electricity or other commonly metered utilities, or that Lessee is generating such a large volume of trash as to require an increase in the size of the trash receptacle and/or an increase in the number of times per month that it is emptied, then Lessor may increase Lessee's Base Rent by an amount equal to such increased costs. There shall be no abatement of Rent and Lessor shall not be liable in any respect whatsoever for the inadequacy, stoppage, interruption or discontinuance of any utility or service due to riot, strike, labor dispute, breakdown, accident, repair or other cause beyond Lessor's reasonable control or in cooperation with governmental request or directions.
- 11.2 Within fifteen days of Lessor's written request, Lessee agrees to deliver to Lessor such information, documents and/or authorization as Lessor needs in order for Lessor to comply with new or existing Applicable Requirements relating to commercial building energy usage, ratings, and/or the reporting thereof.

Assignment and Subletting. 12.

12.1 Lessor's Consent Required.

- (a) Lessee shall not voluntarily or by operation of law assign, transfer, mortgage or encumber (collectively, "assign or assignment") or sublet all or any part of Lessee's interest in this Lease or in the Premises without Lessor's prior written consent.
- (b) Unless Lessee is a corporation and its stock is publicly traded on a national stock exchange, a change in the control of Lessee shall constitute an assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of the voting control of Lessee shall constitute a change in control for this purpose.
- (c) The involvement of Lessee or its assets in any transaction, or series of transactions (by way of merger, sale, acquisition, financing, transfer, leveraged buy-out or otherwise), whether or not a formal assignment or hypothecation of this Lease or Lessee's assets occurs, which results or will result in a reduction of the Net Worth of Lessee by an amount greater than 25% of such Net Worth as it was represented at the time of the execution of this Lease or at the time of the most recent assignment to which Lessor has consented, or as it exists immediately prior to said transaction or transactions constituting such reduction, whichever was or is greater, shall be considered an assignment of this Lease to which Lessor may withhold its consent. "Net Worth of Lessee" shall mean the net worth of Lessee (excluding any guarantors) established under generally accepted accounting principles.
 - An assignment or subletting without consent shall, at Lessor's option, be a Default curable after notice per Paragraph 13.1(d), or a non-curable Breach (d)

without the necessity of any notice and grace period. If Lessor elects to treat such unapproved assignment or subletting as a non-curable Breach, Lessor may either: (i) terminate this Lease, or (ii) upon 30 days written notice, increase the monthly Base Rent to 110% of the Base Rent then in effect. Further, in the event of such Breach and rental adjustment, (i) the purchase price of any option to purchase the Premises held by Lessee shall be subject to similar adjustment to 110% of the price previously in effect, and (ii) all fixed and non-fixed rental adjustments scheduled during the remainder of the Lease term shall be increased to 110% of the scheduled adjusted rent.

- Lessee's remedy for any breach of Paragraph 12.1 by Lessor shall be limited to compensatory damages and/or injunctive relief. (e)
- (f) Lessor may reasonably withhold consent to a proposed assignment or subletting if Lessee is in Default at the time consent is requested.
- Notwithstanding the foregoing, allowing a de minimis portion of the Premises, i.e. 20 square feet or less, to be used by a third party vendor in (g) connection with the installation of a vending machine or payphone shall not constitute a subletting.

12.2 Terms and Conditions Applicable to Assignment and Subletting.

- (a) Regardless of Lessor's consent, no assignment or subletting shall: (i) be effective without the express written assumption by such assignee or sublessee of the obligations of Lessee under this Lease, (ii) release Lessee of any obligations hereunder, or (iii) alter the primary liability of Lessee for the payment of Rent or for the performance of any other obligations to be performed by Lessee.
- Lessor may accept Rent or performance of Lessee's obligations from any person other than Lessee pending approval or disapproval of an assignment. Neither a delay in the approval or disapproval of such assignment nor the acceptance of Rent or performance shall constitute a waiver or estoppel of Lessor's right to exercise its remedies for Lessee's Default or Breach.
 - Lessor's consent to any assignment or subletting shall not constitute a consent to any subsequent assignment or subletting.
- In the event of any Default or Breach by Lessee, Lessor may proceed directly against Lessee, any Guarantors or anyone else responsible for the performance of Lessee's obligations under this Lease, including any assignee or sublessee, without first exhausting Lessor's remedies against any other person or entity responsible therefor to Lessor, or any security held by Lessor.
- (e) Each request for consent to an assignment or subletting shall be in writing, accompanied by information relevant to Lessor's determination as to the financial and operational responsibility and appropriateness of the proposed assignee or sublessee, including but not limited to the intended use and/or required modification of the Premises, if any, together with a fee of \$500 as consideration for Lessor's considering and processing said request. Lessee agrees to provide Lessor with such other or additional information and/or documentation as may be reasonably requested. (See also Paragraph 36)
- Any assignee of, or sublessee under, this Lease shall, by reason of accepting such assignment, entering into such sublease, or entering into possession of the Premises or any portion thereof, be deemed to have assumed and agreed to conform and comply with each and every term, covenant, condition and obligation herein to be observed or performed by Lessee during the term of said assignment or sublease, other than such obligations as are contrary to or inconsistent with provisions of an assignment or sublease to which Lessor has specifically consented to in writing.
- (g) Lessor's consent to any assignment or subletting shall not transfer to the assignee or sublessee any Option granted to the original Lessee by this Lease unless such transfer is specifically consented to by Lessor in writing. (See Paragraph 39.2)
- 12.3 Additional Terms and Conditions Applicable to Subletting. The following terms and conditions shall apply to any subletting by Lessee of all or any part of the Premises and shall be deemed included in all subleases under this Lease whether or not expressly incorporated therein
- (a) Lessee hereby assigns and transfers to Lessor all of Lessee's interest in all Rent payable on any sublease, and Lessor may collect such Rent and apply same toward Lessee's obligations under this Lease; provided, however, that until a Breach shall occur in the performance of Lessee's obligations, Lessee may collect said Rent. In the event that the amount collected by Lessor exceeds Lessee's then outstanding obligations any such excess shall be refunded to Lessee. Lessor shall not, by reason of the foregoing or any assignment of such sublease, nor by reason of the collection of Rent, be deemed liable to the sublessee for any failure of Less to perform and comply with any of Lessee's obligations to such sublessee. Lessee hereby irrevocably authorizes and directs any such sublessee, upon receipt of a written notice from Lessor stating that a Breach exists in the performance of Lessee's obligations under this Lease, to pay to Lessor all Rent due and to become due under the sublease. Sublessee shall rely upon any such notice from Lessor and shall pay all Rents to Lessor without any obligation or right to inquire as to whether such Breach exists, notwithstanding any claim from Lessee to the contrary.
- (b) In the event of a Breach by Lessee, Lessor may, at its option, require sublessee to attorn to Lessor, in which event Lessor shall undertake the obligations of the sublessor under such sublease from the time of the exercise of said option to the expiration of such sublease; provided, however, Lessor shall not be liable for any prepaid rents or security deposit paid by such sublessee to such sublessor or for any prior Defaults or Breaches of such sublessor.

 (c) Any matter requiring the consent of the sublessor under a sublease shall also require the consent of Lessor.

 - No sublessee shall further assign or sublet all or any part of the Premises without Lessor's prior written consent. (d)
- Lessor shall deliver a copy of any notice of Default or Breach by Lessee to the sublessee, who shall have the right to cure the Default of Lessee within (e) the grace period, if any, specified in such notice. The sublessee shall have a right of reimbursement and offset from and against Lessee for any such Defaults cured by

Default; Breach; Remedies.

- 13.1 **Default; Breach**. A "**Default**" is defined as a failure by the Lessee to comply with or perform any of the terms, covenants, conditions or Rules and Regulations under this Lease. A "**Breach**" is defined as the occurrence of one or more of the following Defaults, and the failure of Lessee to cure such Default within any applicable grace period:
- (a) The abandonment of the Premises; the vacating of the Premises prior to the expiration or termination of this Lease without providing a commercially reasonable level of security, or where the coverage of the property insurance described in Paragraph 8.3 is jeopardized as a result thereof, or without providing reasonable assurances to minimize potential vandalism; or failure to deliver to Lessor exclusive possession of the entire Premises in accordance herewith prior to the expiration or termination of this Lease.
- (b) The failure of Lessee to (i) make any payment of Rent or any Security Deposit required to be made by Lessee hereunder, whether to Lessor or to a third party, when due, (ii) to provide reasonable evidence of insurance or surety bond, or (iii) to fulfill any obligation under this Lease which endangers or threatens life or property, where such failure continues for a period of 3 business days following written notice to Lessee. THE ACCEPTANCE BY LESSOR OF A PARTIAL PAYMENT OF RENT OR SECURITY DEPOSIT SHALL NOT CONSTITUTE A WAIVER OF ANY OF LESSOR'S RIGHTS, INCLUDING LESSOR'S RIGHT TO RECOVER POSSESSION OF THE PREMISES.
- The failure of Lessee to allow Lessor and/or its agents access to the Premises or the commission of waste, act or acts constituting public or private (c) nuisance, and/or an illegal activity on the Premises by Lessee, where such actions continue for a period of 3 business days following written notice to Lessee. In the event that Lessee commits waste, a nuisance or an illegal activity a second time then, the Lessor may elect to treat such conduct as a non-curable Breach rather than a
- The failure by Lessee to provide (i) reasonable written evidence of compliance with Applicable Requirements, (ii) the service contracts, (iii) the rescission of an unauthorized assignment or subletting, (iv) an Estoppel Certificate or financial statements, (v) a requested subordination, (vi) evidence concerning any guaranty and/or Guarantor, (vii) any document requested under Paragraph 41, (viii) material safety data sheets (MSDS), or (ix) any other documentation or information which Lessor may reasonably require of Lessee under the terms of this Lease, where any such failure continues for a period of 10 days following written notice to Lessee
- (e) A Default by Lessee as to the terms, covenants, conditions or provisions of this Lease, or of the rules adopted under Paragraph 2.9 hereof, other than those described in subparagraphs 13.1(a), (b), (c) or (d), above, where such Default continues for a period of 30 days after written notice; provided, however, that if the nature of Lessee's Default is such that more than 30 days are reasonably required for its cure, then it shall not be deemed to be a Breach if Lessee commences such cure within said 30 day period and thereafter diligently prosecutes such cure to completion.
- The occurrence of any of the following events: (i) the making of any general arrangement or assignment for the benefit of creditors; (ii) becoming a debton as defined in 11 U.S.C. § 101 or any successor statute thereto (unless, in the case of a petition filed against Lessee, the same is dismissed within 60 days); (iii) the appointment of a trustee or receiver to take possession of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where possession is not restored to Lessee within 30 days; or (iv) the attachment, execution or other judicial seizure of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where such seizure is not discharged within 30 days; provided, however, in the event that any provision of this subparagraph is contrary to any applicable law, such provision shall be of no force or effect, and not affect the validity of the remaining provisions.
 - $The\ discovery\ that\ any\ financial\ statement\ of\ Lessee\ or\ of\ any\ Guarantor\ given\ to\ Lessor\ was\ materially\ false.$
- (h) If the performance of Lessee's obligations under this Lease is guaranteed: (i) the death of a Guarantor, (ii) the termination of a Guarantor's liability with respect to this Lease other than in accordance with the terms of such guaranty, (iii) a Guarantor's becoming insolvent or the subject of a bankruptcy filing, (iv) a Guarantor's refusal to honor the guaranty, or (v) a Guarantor's breach of its guaranty obligation on an anticipatory basis, and Lessee's failure, within 60 days following written notice of any such event, to provide written alternative assurance or security, which, when coupled with the then existing resources of Lessee, equals or exceeds the combined financial resources of Lessee and the Guarantors that existed at the time of execution of this Lease.
- 13.2 Remedies. If Lessee fails to perform any of its affirmative duties or obligations, within 10 days after written notice (or in case of an emergency, without notice), Lessor may, at its option, perform such duty or obligation on Lessee's behalf, including but not limited to the obtaining of reasonably required bonds, insurance policies, or governmental licenses, permits or approvals. Lessee shall pay to Lessor an amount equal to 115% of the costs and expenses incurred by Lessor

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

in such performance upon receipt of an invoice therefor. In the event of a Breach, Lessor may, with or without further notice or demand, and without limiting Lessor in the exercise of any right or remedy which Lessor may have by reason of such Breach:

- Terminate Lessee's right to possession of the Premises by any lawful means, in which case this Lease shall terminate and Lessee shall immediately surrender possession to Lessor. In such event Lessor shall be entitled to recover from Lessee: (i) the unpaid Rent which had been earned at the time of termination; (ii) the worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that the Lessee proves could have been reasonably avoided; (iii) the worth at the time of award of the amount by which the unpaid rent for the balance of the term after the time of award exceeds the amount of such rental loss that the Lessee proves could be reasonably avoided; and (iv) any other amount necessary to compensate Lessor for all the detriment proximately caused by the Lessee's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom, including but not limited to the cost of recovering possession of the Premises, expenses of reletting, including necessary renovation and alteration of the Premises, reasonable attorneys' fees, and that portion of any leasing commission paid by Lessor in connection with this Lease applicable to the unexpired term of this Lease. Lessor and Lessee agree that the damages to be incurred by the Lessor in the event of Lessee's default of the Lease would be difficult or impossible to calculate and the parties therefore intend to provide by the foregoing for liquidated damages and not a penalty and agree that the sum provided is a reasonable pre-estimate of the probable loss. The worth at the time of award of the amount referred to in provision (iii) of the immediately preceding sentence shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of the District within which the Premises are located at the time of award plus one percent. Efforts by Lessor to mitigate damages caused by Lessee's Breach of this Lease shall not waive Lessor's right to recover any damages to which Lessor is otherwise entitled. If termination of this Lease is obtained through the provisional remedy of unlawful detainer, Lessor shall have the right to recover in such proceeding any unpaid Rent and damages as are recoverable therein, or Lessor may reserve the right to recover all or any part thereof in a separate suit. If a notice and grace period required under Paragraph 13.1 was not previously given, a notice to pay rent or quit, or to perform or quit given to Lessee under the unlawful detainer statute shall also constitute the notice required by Paragraph 13.1. In such case, the applicable grace period required by Paragraph 13.1 and the unlawful detainer statute shall run concurrently, and the failure of Lessee to cure the Default within the greater of the two such grace periods shall constitute both an unlawful detainer and a Breach of this Lease entitling Lessor to the remedies provided for in this Lease and/or by said statute.
- (b) Continue the Lease and Lessee's right to possession and recover the Rent as it becomes due, in which event Lessee may sublet or assign, subject only to reasonable limitations. Acts of maintenance, efforts to relet, and/or the appointment of a receiver to protect the Lessor's interests, shall not constitute a termination of the Lessee's right to possession.
- (c) Pursue any other remedy now or hereafter available under the laws or judicial decisions of the state wherein the Premises are located. The expiration or termination of this Lease and/or the termination of Lessee's right to possession shall not relieve Lessee from liability under any indemnity provisions of this Lease as to matters occurring or accruing during the term hereof or by reason of Lessee's occupancy of the Premises.
- 13.3 Inducement Recapture. Any agreement for free or abated rent or other charges, the cost of tenant improvements for Lessee paid for or performed by Lessor, or for the giving or paying by Lessor to or for Lessee of any cash or other bonus, inducement or consideration for Lessee's entering into this Lease, all of which concessions are hereinafter referred to as "Inducement Provisions," shall be deemed conditioned upon Lessee's full and faithful performance of all of the terms, covenants and conditions of this Lease. Upon Breach of this Lease by Lessee, any such Inducement Provision shall automatically be deemed deleted from this Lease and of no further force or effect, and any rent, other charge, bonus, inducement or consideration theretofore abated, given or paid by Lessor under such an Inducement Provision shall be immediately due and payable by Lessee to Lessor, notwithstanding any subsequent cure of said Breach by Lessee. The acceptance by Lessor of rent or the cure of the Breach which initiated the operation of this paragraph shall not be deemed a waiver by Lessor of the provisions of this paragraph unless specifically so stated in writing by Lessor at the time of such acceptance.
- 13.4 Late Charges. Lessee hereby acknowledges that late payment by Lessee of Rent will cause Lessor to incur costs not contemplated by this Lease, the exact amount of which will be extremely difficult to ascertain. Such costs include, but are not limited to, processing and accounting charges, and late charges which may be imposed upon Lessor by any Lender. Accordingly, if any Rent shall not be received by Lessor within 5 days after such amount shall be due, then, without any requirement for notice to Lessee, Lessee shall immediately pay to Lessor a one-time late charge equal to 10% of each such overdue amount or \$100, whichever is greater. The parties hereby agree that such late charge represents a fair and reasonable estimate of the costs Lessor will incur by reason of such late payment. Acceptance of such late charge by Lessor shall in no event constitute a waiver of Lessee's Default or Breach with respect to such overdue amount, nor prevent the exercise of any of the other rights and remedies granted hereunder. In the event that a late charge is payable hereunder, whether or not collected, for 3 consecutive installments of Base Rent, then notwithstanding any provision of this Lease to the contrary, Base Rent shall, at Lessor's option, become due and payable quarterly in advance.
- 13.5 Interest. Any monetary payment due Lessor hereunder, other than late charges, not received by Lessor, when due shall bear interest from the 31st day after it was due. The interest ("Interest") charged shall be computed at the rate of 10% per annum but shall not exceed the maximum rate allowed by law. Interest is payable in addition to the potential late charge provided for in Paragraph 13.4.

13.6 Breach by Lessor.

- (a) **Notice of Breach**. Lessor shall not be deemed in breach of this Lease unless Lessor fails within a reasonable time to perform an obligation required to be performed by Lessor. For purposes of this Paragraph, a reasonable time shall in no event be less than 30 days after receipt by Lessor, and any Lender whose name and address shall have been furnished to Lessee in writing for such purpose, of written notice specifying wherein such obligation of Lessor has not been performed; provided, however, that if the nature of Lessor's obligation is such that more than 30 days are reasonably required for its performance, then Lessor shall not be in breach if performance is commenced within such 30 day period and thereafter diligently pursued to completion.
- (b) **Performance by Lessee on Behalf of Lessor.** In the event that neither Lessor nor Lender cures said breach within 30 days after receipt of said notice, or if having commenced said cure they do not diligently pursue it to completion, then Lessee may elect to cure said breach at Lessee's expense and offset from Rent the actual and reasonable cost to perform such cure, provided however, that such offset shall not exceed an amount equal to the greater of one month's Base Rent or the Security Deposit, reserving Lessee's right to reimbursement from Lessor for any such expense in excess of such offset. Lessee shall document the cost of said cure and supply said documentation to Lessor.
- 14. Condemnation. If the Premises or any portion thereof are taken under the power of eminent domain or sold under the threat of the exercise of said power (collectively "Condemnation"), this Lease shall terminate as to the part taken as of the date the condemning authority takes title or possession, whichever first occurs. If more than 10% of the floor area of the Unit, or more than 25% of the parking spaces is taken by Condemnation, Lessee may, at Lessee's option, to be exercised in writing within 10 days after Lessor shall have given Lessee written notice of such taking (or in the absence of such notice, within 10 days after the condemning authority shall have taken possession) terminate this Lease as of the date the condemning authority takes such possession. If Lessee does not terminate this Lease in accordance with the foregoing, this Lease shall remain in full force and effect as to the portion of the Premises remaining, except that the Base Rent shall be reduced in proportion to the reduction in utility of the Premises caused by such Condemnation. Condemnation awards and/or payments shall be the property of Lessor, whether such award shall be made as compensation for diminution in value of the leasehold, the value of the part taken, or for severance damages; provided, however, that Lessee shall be entitled to any compensation paid by the condemnor for Lessee's relocation expenses, loss of business goodwill and/or Trade Fixtures, without regard to whether or not this Lease is terminated pursuant to the provisions of this Paragraph. All Alterations and Utility Installations made to the Premises by Lessee, for purposes of Condemnation only, shall be considered the property of the Lessee and Lessee shall be entitled to any and all compensation which is payable therefor. In the event that this Lease is not terminated by reason of the Condemnation, Lessor shall repair any damage to the Premises caused by such Condemnation.

15. Brokerage Fees.

- 15.1 Additional Commission. In addition to the payments owed pursuant to Paragraph 1.10 above, Lessor agrees that: (a) if Lessee exercises any Option, (b) if Lessee or anyone affiliated with Lessee acquires from Lessor any rights to the Premises or other premises owned by Lessor and located within the Project, (c) if Lessee remains in possession of the Premises, with the consent of Lessor, after the expiration of this Lease, or (d) if Base Rent is increased, whether by agreement or operation of an escalation clause herein, then, Lessor shall pay Brokers a fee in accordance with the fee schedule of the Brokers in effect at the time the Lease was executed. The provisions of this paragraph are intended to supersede the provisions of any earlier agreement to the contrary.
- 15.2 **Assumption of Obligations**. Any buyer or transferee of Lessor's interest in this Lease shall be deemed to have assumed Lessor's obligation hereunder. Brokers shall be third party beneficiaries of the provisions of Paragraphs 1.10, 15, 22 and 31. If Lessor fails to pay to Brokers any amounts due as and for brokerage fees pertaining to this Lease when due, then such amounts shall accrue Interest. In addition, if Lessor fails to pay any amounts to Lessee's Broker when due, Lessee's Broker may send written notice to Lessor and Lessee of such failure and if Lessor fails to pay such amounts within 10 days after said notice, Lessee shall pay said monies to its Broker and offset such amounts against Rent. In addition, Lessee's Broker shall be deemed to be a third party beneficiary of any commission agreement entered into by and/or between Lessor and Lessor's Broker for the limited purpose of collecting any brokerage fee owed.
- 15.3 Representations and Indemnities of Broker Relationships. Lessee and Lessor each represent and warrant to the other that it has had no dealings with any person, firm, broker, agent or finder (other than the Brokers and Agents, if any) in connection with this Lease, and that no one other than said named Brokers and Agents is entitled to any commission or finder's fee in connection herewith. Lessee and Lessor do each hereby agree to indemnify, protect, defend and hold the other harmless from and against liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any dealings or actions of the indemnifying Party, including any costs, expenses, attorneys' fees reasonably incurred with respect thereto.

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

16. Estoppel Certificates.

- (a) Each Party (as "Responding Party") shall within 10 days after written notice from the other Party (the "Requesting Party") execute, acknowledge and deliver to the Requesting Party a statement in writing in form similar to the then most current "Estoppel Certificate" form published by AIR CRE, plus such additional information, confirmation and/or statements as may be reasonably requested by the Requesting Party.
- (b) If the Responding Party shall fail to execute or deliver the Estoppel Certificate within such 10 day period, the Requesting Party may execute an Estoppel Certificate stating that: (i) the Lease is in full force and effect without modification except as may be represented by the Requesting Party, (ii) there are no uncured defaults in the Requesting Party's performance, and (iii) if Lessor is the Requesting Party, not more than one month's rent has been paid in advance. Prospective purchasers and encumbrancers may rely upon the Requesting Party's Estoppel Certificate, and the Responding Party shall be estopped from denying the truth of the facts contained in said Certificate. In addition, Lessee acknowledges that any failure on its part to provide such an Estoppel Certificate will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to execute and/or deliver a requested Estoppel Certificate in a timely fashion the monthly Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for remainder of the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to provide the Estoppel Certificate. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to provide the Estoppel Certificate nor prevent the exercise of any of the other rights and remedies granted hereunder.
- (c) If Lessor desires to finance, refinance, or sell the Premises, or any part thereof, Lessee and all Guarantors shall within 10 days after written notice from Lessor deliver to any potential lender or purchaser designated by Lessor such financial statements as may be reasonably required by such lender or purchaser, including but not limited to Lessee's financial statements for the past 3 years. All such financial statements shall be received by Lessor and such lender or purchaser in confidence and shall be used only for the purposes herein set forth.
- 17. Definition of Lessor. The term "Lessor" as used herein shall mean the owner or owners at the time in question of the fee title to the Premises, or, if this is a sublease, of the Lessee's interest in the prior lease. In the event of a transfer of Lessor's title or interest in the Premises or this Lease, Lessor shall deliver to the transferee or assignee (in cash or by credit) any unused Security Deposit held by Lessor. Upon such transfer or assignment and delivery of the Security Deposit, as aforesaid, the prior Lessor shall be relieved of all liability with respect to the obligations and/or covenants under this Lease thereafter to be performed by the Lessor. Subject to the foregoing, the obligations and/or covenants in this Lease to be performed by the Lessor shall be binding only upon the Lessor as hereinabove defined.
- 18. Severability. The invalidity of any provision of this Lease, as determined by a court of competent jurisdiction, shall in no way affect the validity of any other provision hereof.
- 19. Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Lease shall mean and refer to calendar days
- Limitation on Liability. The obligations of Lessor under this Lease shall not constitute personal obligations of Lessor, or its partners, members, directors, officers or shareholders, and Lessee shall look to the Premises, and to no other assets of Lessor, for the satisfaction of any liability of Lessor with respect to this Lease, and shall not seek recourse against Lessor's partners, members, directors, officers or shareholders, or any of their personal assets for such satisfaction.
- 21. Time of Essence. Time is of the essence with respect to the performance of all obligations to be performed or observed by the Parties under this Lease.
- 22. No Prior or Other Agreements; Broker Disclaimer. This Lease contains all agreements between the Parties with respect to any matter mentioned herein, and no other prior or contemporaneous agreement or understanding shall be effective. Lessor and Lessee each represents and warrants to the Brokers that it has made, and is relying solely upon, its own investigation as to the nature, quality, character and financial responsibility of the other Party to this Lease and as to the use, nature, quality and character of the Premises. Brokers have no responsibility with respect thereto or with respect to any default or breach hereof by either Party.

- 23.1 Notice Requirements. All notices required or permitted by this Lease or applicable law shall be in writing and may be delivered in person (by hand or by courier) or may be sent by regular, certified or registered mail or U.S. Postal Service Express Mail, with postage prepaid, or by facsimile transmission, or by email, and shall be deemed sufficiently given if served in a manner specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specify a different address for notice, except that upon Lessee's taking possession of the Premises, the Premises shall constitute Lessee's address for notice. A copy of all notices to Lessor shall be concurrently transmitted to such party or parties at such addresses as Lessor may from time to time hereafter designate in writing.
- 23.2 Date of Notice. Any notice sent by registered or certified mail, return receipt requested, shall be deemed given on the date of delivery shown on the receipt card, or if no delivery date is shown, the postmark thereon. If sent by regular mail the notice shall be deemed given 72 hours after the same is addressed as required herein and mailed with postage prepaid. Notices delivered by United States Express Mail or overnight courier that guarantees next day delivery shall be deemed given 24 hours after delivery of the same to the Postal Service or courier. Notices delivered by hand, or transmitted by facsimile transmission or by email shall be deemed delivered upon actual receipt. If notice is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.

 23.3 **Options.** Notwithstanding the foregoing, in order to exercise any Options (see paragraph 39), the Notice must be sent by Certified Mail (return receipt
- requested), Express Mail (signature required), courier (signature required) or some other methodology that provides a receipt establishing the date the notice was received by the Lessor.

Waivers

- (a) No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any other term, covenant or condition hereof. Lessor's consent to, or approval of, any act shall not be deemed to render unnecessary the obtaining of Lessor's consent to, or approval of, any subsequent or similar act by Lessee, or be construed as the basis of an estoppel to enforce the provision or provisions of this Lease requiring such consent.
- (b) The acceptance of Rent by Lessor shall not be a waiver of any Default or Breach by Lessee. Any payment by Lessee may be accepted by Lessor on account of monies or damages due Lessor, notwithstanding any qualifying statements or conditions made by Lessee in connection therewith, which such statements and/or conditions shall be of no force or effect whatsoever unless specifically agreed to in writing by Lessor at or before the time of deposit of such payment.
- THE PARTIES AGREE THAT THE TERMS OF THIS LEASE SHALL GOVERN WITH REGARD TO ALL MATTERS RELATED THERETO AND HEREBY WAIVE THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE TO THE EXTENT THAT SUCH STATUTE IS INCONSISTENT WITH THIS LEASE.

25. Disclosures Regarding The Nature of a Real Estate Agency Relationship.

- what type of agency relationship or representation it has with the agent or agents in the transaction. Lessor and Lessee acknowledge being advised by the Brokers in this transaction, as follows:
- (i) Lessor's Agent. A Lessor's agent under a listing agreement with the Lessor acts as the agent for the Lessor only. A Lessor's agent or subagent has the following affirmative obligations: To the Lessor: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessor. To the Lessoe and the Lesson: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of; the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (ii) <u>Lessee's Agent</u>. An agent can agree to act as agent for the Lessee only. In these situations, the agent is not the Lessor's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Lessor. An agent acting only for a Lessee has the following affirmative obligations. To the Lessee: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessee. To the Lessee and the Lessor: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (iii) Agent Representing Both Lessor and Lessee. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Lessor and the Lessee in a transaction, but only with the knowledge and consent of both the Lessor and the Lessee. In a dual agency situation, the agent has the following affirmative obligations to both the Lessor and the Lessee: (a) A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Lessor or the Lessee. (b) Other duties to the Lessor and the Lessee as stated above in subparagraphs (i) or (ii). In representing both Lessor and Lessee, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating to either Lessee's or Lessor's financial position, motivations, bargaining position, or other personal information that may impact rent, including Lessor's willingness to accept a rent less than the listing rent or Lessee's willingness to pay rent greater than the rent offered. The above duties of the agent in a real estate transaction do

ΓIALS

not relieve a Lessor or Lessee from the responsibility to protect their own interests. Lessor and Lessee should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Both Lessor and Lessee should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.

- (b) Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this Lease may be brought against Broker more than one year after the Start Date and that the liability (including court costs and attorneys' fees), of any Broker with respect to any such lawsuit and/or legal proceeding shall not exceed the fee received by such Broker pursuant to this Lease; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.
- (c) Lessor and Lessee agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- 26. No Right To Holdover. Lessee has no right to retain possession of the Premises or any part thereof beyond the expiration or termination of this Lease. At or prior to the expiration or termination of this Lease Lessee shall deliver exclusive possession of the Premises to Lessor. For purposes of this provision and Paragraph 13.1(a), exclusive possession shall mean that Lessee shall have vacated the Premises, removed all of its personal property therefrom and that the Premises have been returned in the condition specified in this Lease. In the event that Lessee does not deliver exclusive possession to Lessor as specified above, then Lessor's damages during any holdover period shall be computed at the amount of the Rent (as defined in Paragraph 4.1) due during the last full month before the expiration or termination of this Lease (disregarding any temporary abatement of Rent that may have been in effect), but with Base Rent being 150% of the Base Rent payable during such last full month. Nothing contained herein shall be construed as consent by Lessor to any holding over by Lessee.
- 27. Cumulative Remedies. No remedy or election hereunder shall be deemed exclusive but shall, wherever possible, be cumulative with all other remedies at law or in equity.
- 28. Covenants and Conditions; Construction of Agreement. All provisions of this Lease to be observed or performed by Lessee are both covenants and conditions. In construing this Lease, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Lease. Whenever required by the context, the singular shall include the plural and vice versa. This Lease shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.
- 29. Binding Effect; Choice of Law. This Lease shall be binding upon the Parties, their personal representatives, successors and assigns and be governed by the laws of the State in which the Premises are located. Any litigation between the Parties hereto concerning this Lease shall be initiated in the county in which the Premises are located. Signatures to this Lease accomplished by means of electronic signature or similar technology shall be legal and binding.

30. Subordination; Attornment; Non-Disturbance.

- 30.1 **Subordination**. This Lease and any Option granted hereby shall be subject and subordinate to any ground lease, mortgage, deed of trust, or other hypothecation or security device (collectively, "Security Device"), now or hereafter placed upon the Premises, to any and all advances made on the security thereof, and to all renewals, modifications, and extensions thereof. Lessee agrees that the holders of any such Security Devices (in this Lease together referred to as "Lender") shall have no liability or obligation to perform any of the obligations of Lessor under this Lease. Any Lender may elect to have this Lease and/or any Option granted hereby superior to the lien of its Security Device by giving written notice thereof to Lessee, whereupon this Lease and such Options shall be deemed prior to such Security Device, notwithstanding the relative dates of the documentation or recordation thereof.
- 30.2 **Attornment**. In the event that Lessor transfers title to the Premises, or the Premises are acquired by another upon the foreclosure or termination of a Security Device to which this Lease is subordinated (i) Lessee shall, subject to the non-disturbance provisions of Paragraph 30.3, attorn to such new owner, and upon request, enter into a new lease, containing all of the terms and provisions of this Lease, with such new owner for the remainder of the term hereof, or, at the election of the new owner, this Lease will automatically become a new lease between Lessee and such new owner, for the remainder of the term hereof and (ii) Lessor shall thereafter be relieved of any further obligations hereunder and such new owner shall assume all of Lessor's obligations, except that such new owner shall not: (a) be liable for any act or omission of any prior lessor or with respect to events occurring prior to acquisition of ownership; (b) be subject to any offsets or defenses which Lessee might have against any prior lessor, (c) be bound by prepayment of more than one month's rent, or (d) be liable for the return of any security deposit paid to any prior lessor which was not paid or credited to such new owner.
- 30.3 **Non-Disturbance**. With respect to Security Devices entered into by Lessor after the execution of this Lease, Lessee's subordination of this Lease shall be subject to receiving a commercially reasonable non-disturbance agreement (a "**Non-Disturbance Agreement**") from the Lender which Non-Disturbance Agreement provides that Lessee's possession of the Premises, and this Lease, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises. Further, within 60 days after the execution of this Lease, Lessor shall, if requested by Lessee, use its commercially reasonable efforts to obtain a Non-Disturbance Agreement from the holder of any pre-existing Security Device which is secured by the Premises. In the event that Lessor is unable to provide the Non-Disturbance Agreement within said 60 days, then Lessee may, at Lessee's option, directly contact Lender and attempt to negotiate for the execution and delivery of a Non-Disturbance Agreement.
- 30.4 **Self-Executing**. The agreements contained in this Paragraph 30 shall be effective without the execution of any further documents; provided, however, that, upon written request from Lessor or a Lender in connection with a sale, financing or refinancing of the Premises, Lessee and Lessor shall execute such further writings as may be reasonably required to separately document any subordination, attornment and/or Non-Disturbance Agreement provided for herein.
- 31. Attorneys' Fees. If any Party or Broker brings an action or proceeding involving the Premises whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred. In addition, Lessor shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is subsequently commenced in connection with such Default or resulting Breach (\$200 is a reasonable minimum per occurrence for such services and consultation).
- 32. Lessor's Access; Showing Premises; Repairs. Lessor and Lessor's agents shall have the right to enter the Premises at any time, in the case of an emergency, and otherwise at reasonable times after reasonable prior notice for the purpose of showing the same to prospective purchasers, lenders, or tenants, and making such alterations, repairs, improvements or additions to the Premises as Lessor may deem necessary or desirable and the erecting, using and maintaining of utilities, services, pipes and conduits through the Premises and/or other premises as long as there is no material adverse effect on Lessee's use of the Premises. All such activities shall be without abatement of rent or liability to Lessee.
- **33. Auctions.** Lessee shall not conduct, nor permit to be conducted, any auction upon the Premises without Lessor's prior written consent. Lessor shall not be obligated to exercise any standard of reasonableness in determining whether to permit an auction.
- **34. Signs.** Lessor may place on the Premises ordinary "For Sale" signs at any time and ordinary "For Lease" signs during the last 6 months of the term hereof. Except for ordinary "For Sublease" signs which may be placed only on the Premises, Lessee shall not place any sign upon the Project without Lessor's prior written consent. All signs must comply with all Applicable Requirements.
- **35. Termination; Merger.** Unless specifically stated otherwise in writing by Lessor, the voluntary or other surrender of this Lease by Lessee, the mutual termination or cancellation hereof, or a termination hereof by Lessor for Breach by Lessee, shall automatically terminate any sublease or lesser estate in the Premises; provided, however, that Lessor may elect to continue any one or all existing subtenancies. Lessor's failure within 10 days following any such event to elect to the contrary by written notice to the holder of any such lesser interest, shall constitute Lessor's election to have such event constitute the termination of such interest.
- **36. Consents.** All requests for consent shall be in writing. Except as otherwise provided herein, wherever in this Lease the consent of a Party is required to an act by or for the other Party, such consent shall not be unreasonably withheld or delayed. Lessor's actual reasonable costs and expenses (including but not limited to architects', attorneys', engineers' and other consultants' fees) incurred in the consideration of, or response to, a request by Lessee for any Lessor consent, including but not limited to consents to an assignment, a subletting or the presence or use of a Hazardous Substance, shall be paid by Lessee upon receipt of an invoice and supporting documentation therefor. Lessor's consent to any act, assignment or subletting shall not constitute an acknowledgment that no Default or Breach by Lessee of this Lease exists, nor shall such consent be deemed a waiver of any then existing Default or Breach, except as may be otherwise specifically stated in writing by Lessor at the time of such consent. The failure to specify herein any particular condition to Lessor's consent shall not preclude the imposition by Lessor at the time of consent of such further or other conditions as are then reasonable with reference to the particular matter for which consent is being given. In the event that either Party disagrees with any determination made by the other hereunder and reasonably requests the reasons for such determination, the determining party shall furnish

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

its reasons in writing and in reasonable detail within 10 business days following such request.

37 Guarantor

- 37.1 Execution. The Guarantors, if any, shall each execute a guaranty in the form most recently published by AIR CRE.
- 37.2 **Default**. It shall constitute a Default of the Lessee if any Guarantor fails or refuses, upon request to provide: (a) evidence of the execution of the guaranty, including the authority of the party signing on Guarantor's behalf to obligate Guarantor, and in the case of a corporate Guarantor, a certified copy of a resolution of its board of directors authorizing the making of such guaranty, (b) current financial statements, (c) an Estoppel Certificate, or (d) written confirmation that the guaranty is still in effect
- **38. Quiet Possession.** Subject to payment by Lessee of the Rent and performance of all of the covenants, conditions and provisions on Lessee's part to be observed and performed under this Lease, Lessee shall have quiet possession and quiet enjoyment of the Premises during the term hereof.
- 39. Options. If Lessee is granted any option, as defined below, then the following provisions shall apply.
- 39.1 **Definition**. "**Option**" shall mean: (a) the right to extend or reduce the term of or renew this Lease or to extend or reduce the term of or renew any lease that Lessee has on other property of Lessor; (b) the right of first refusal or first offer to lease either the Premises or other property of Lessor; (c) the right to purchase, the right of first offer to purchase or the right of first refusal to purchase the Premises or other property of Lessor.
- 39.2 **Options Personal To Original Lessee**. Any Option granted to Lessee in this Lease is personal to the original Lessee, and cannot be assigned or exercised by anyone other than said original Lessee and only while the original Lessee is in full possession of the Premises and, if requested by Lessor, with Lessee certifying that Lessee has no intention of thereafter assigning or subletting.
- 39.3 **Multiple Options**. In the event that Lessee has any multiple Options to extend or renew this Lease, a later Option cannot be exercised unless the prior Options have been validly exercised.

39.4 Effect of Default on Options.

- (a) Lessee shall have no right to exercise an Option: (i) during the period commencing with the giving of any notice of Default and continuing until said Default is cured, (ii) during the period of time any Rent is unpaid (without regard to whether notice thereof is given Lessee), (iii) during the time Lessee is in Breach of this Lease, or (iv) in the event that Lessee has been given 3 or more notices of separate Default, whether or not the Defaults are cured, during the 12 month period immediately preceding the exercise of the Option.
- (b) The period of time within which an Option may be exercised shall not be extended or enlarged by reason of Lessee's inability to exercise an Option because of the provisions of Paragraph 39.4(a).
- (c) An Option shall terminate and be of no further force or effect, notwithstanding Lessee's due and timely exercise of the Option, if, after such exercise and prior to the commencement of the extended term or completion of the purchase, (i) Lessee fails to pay Rent for a period of 30 days after such Rent becomes due (without any necessity of Lessor to give notice thereof), or (ii) if Lessee commits a Breach of this Lease.
- **40. Security Measures.** Lessee hereby acknowledges that the Rent payable to Lessor hereunder does not include the cost of guard service or other security measures, and that Lessor shall have no obligation whatsoever to provide same. Lessee assumes all responsibility for the protection of the Premises, Lessee, its agents and invitees and their property from the acts of third parties.
- 41. Reservations. Lessor reserves the right: (i) to grant, without the consent or joinder of Lessee, such easements, rights and dedications that Lessor deems necessary; (ii) to cause the recordation of parcel maps and restrictions; and (iii) to create and/or install new utility raceways, so long as such easements, rights, dedications, maps, restrictions, and utility raceways do not unreasonably interfere with the use of the Premises by Lessee. Lessee agrees to sign any documents reasonably requested by Lessor to effectuate such rights.
- **42. Performance Under Protest.** If at any time a dispute shall arise as to any amount or sum of money to be paid by one Party to the other under the provisions hereof, the Party against whom the obligation to pay the money is asserted shall have the right to make payment "under protest" and such payment shall not be regarded as a voluntary payment and there shall survive the right on the part of said Party to institute suit for recovery of such sum. If it shall be adjudged that there was no legal obligation on the part of said Party to pay such sum or any part thereof, said Party shall be entitled to recover such sum or so much thereof as it was not legally required to pay. A Party who does not initiate suit for the recovery of sums paid "under protest" within 6 months shall be deemed to have waived its right to protest such payment.

43. Authority; Multiple Parties; Execution.

- (a) If either Party hereto is a corporation, trust, limited liability company, partnership, or similar entity, each individual executing this Lease on behalf of such entity represents and warrants that he or she is duly authorized to execute and deliver this Lease on its behalf. Each Party shall, within 30 days after request, deliver to the other Party satisfactory evidence of such authority.
- (b) If this Lease is executed by more than one person or entity as "Lessee", each such person or entity shall be jointly and severally liable hereunder. It is agreed that any one of the named Lessees shall be empowered to execute any amendment to this Lease, or other document ancillary thereto and bind all of the named Lessees, and Lessor may rely on the same as if all of the named Lessees had executed such document.
- (c) This Lease may be executed by the Parties in counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument.
- **44. Conflict.** Any conflict between the printed provisions of this Lease and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions.
- **45. Offer**. Preparation of this Lease by either party or their agent and submission of same to the other Party shall not be deemed an offer to lease to the other Party. This Lease is not intended to be binding until executed and delivered by all Parties hereto.
- **46. Amendments.** This Lease may be modified only in writing, signed by the Parties in interest at the time of the modification. As long as they do not materially change Lessee's obligations hereunder, Lessee agrees to make such reasonable non-monetary modifications to this Lease as may be reasonably required by a Lender in connection with the obtaining of normal financing or refinancing of the Premises.
- 47. Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY OR ARISING OUT OF THIS AGREEMENT.

48.	Arbitration of Disputes. An Addendum requiring the Arbitration of all disputes between the Parties and/or Brokers arising out of this Lease is not
atta	ached to this Lease.
49.	Accessibility; Americans with Disabilities Act.
	(a) The Premises:
detereq the arra	have not undergone an inspection by a Certified Access Specialist (CASp). Note: A Certified Access Specialist (CASp) can inspect the subject premises and termine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not juire a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection of subject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the angements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to rect violations of construction-related accessibility standards within the premises.
	have undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises met all applicable construction-related accessibility indards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this
acc	ise and agrees to keep such report confidential. have undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises did not meet all applicable construction-related ressibility standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to recuting this Lease and agrees to keep such report confidential except as necessary to complete repairs and corrections of violations of construction related ressibility standards.

In the event that the Premises have been issued an inspection report by a CASp the Lessor shall provide a copy of the disability access inspection certificate to Lessee within 7 days of the execution of this Lease.

NITIALS	INITIA	LS
2019 AIR CRE. All Rights Reserved.		

(b) Since compliance with the Americans with Disabilities Act (ADA) and other state and local accessibility statutes are dependent upon Lessee's specific use of the Premises, Lessor makes no warranty or representation as to whether or not the Premises comply with ADA or any similar legislation. In the event that Lessee's use of the Premises requires modifications or additions to the Premises in order to be in compliance with ADA or other accessibility statutes, Lessee agrees to make any such necessary modifications and/or additions at Lessee's expense.

LESSOR AND LESSEE HAVE CAREFULLY READ AND REVIEWED THIS LEASE AND EACH TERM AND PROVISION CONTAINED HEREIN, AND BY THE EXECUTION OF THIS LEASE SHOW THEIR INFORMED AND VOLUNTARY CONSENT THERETO. THE PARTIES HEREBY AGREE THAT, AT THE TIME THIS LEASE IS EXECUTED, THE TERMS OF THIS LEASE ARE COMMERCIALLY REASONABLE AND EFFECTUATE THE INTENT AND PURPOSE OF LESSOR AND LESSEE WITH RESPECT TO THE PREMISES.

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS LEASE OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

- SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS LEASE.
- 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PREMISES. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING OF THE PREMISES, THE STRUCTURAL INTEGRITY, THE CONDITION OF THE ROOF AND OPERATING SYSTEMS, COMPLIANCE WITH THE AMERICANS WITH DISABILITIES ACT AND THE SUITABILITY OF THE PREMISES FOR LESSEE'S INTENDED USE.

WARNING: IF THE PREMISES ARE LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THE LEASE MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PREMISES ARE LOCATED.

The parties hereto have executed this Lease at the place and on the dates specified above their respective signatures.

Executed at:	Executed at:
On:	On:
By LESSOR:	By LESSEE:
<u>·</u>	
By:	Ву:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
By:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Address:	Address:
Federal ID No.:	Federal ID No.:
	'
BROKER	BROKER
Attn:	Attn:
Title:	Title:
Address:	Address:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Federal ID No.:	Federal ID No.:
Broker DRE License #:	Broker DRE License #:
Agent DRE License #:	Agent DRE License #:

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS INITIALS



STANDARD INDUSTRIAL/COMMERCIAL MULTI-TENANT LEASE - GROSS

1. Basic Provisions ("Basic Provisions").	
1.1 Parties. This Lease ("Lease"), dated for reference purposes only	, is made by and between ("Lessor") and
("Lessee"), (collectively the "Parties", or individually a "Party"). 1.2(a) Premises: That certain real property, including all improvements	therein or to be provided by Lessor under the terms of this Lease, commonly known
as (street address, unit/suite, city, state, zip): ("Premises"). The Pre	emises are located in the County of, and are generally described as
	addition to Lessee's rights to use and occupy the Premises as hereinafter specified,
2.7 below), but shall not have any rights to the roof, or exterior walls of the Buildin	ining the Premises ("Building") and to the Common Areas (as defined in Paragraphing or to any other buildings in the Project. The Premises, the Building, the Common provements thereon, are herein collectively referred to as the "Project." (See also
Paragraph 2)	provenients thereon, are never concentrely referred to as the Troject. (See also
1.2(b) Parking: unreserved vehicle parking spaces. (See	also Paragraph 2.6)
1.3 Term: years and months ("Original Term"	") commencing ("Commencement Date") and ending
("Expiration Date"). (See also Paragraph 3) 1.4 Early Possession: If the Premises are available Lessee may have non-experience.	xclusive possession of the Premises commencing ("Early Possession
Date"). (See also Paragraphs 3.2 and 3.3)	
	day of each month commencing (See also Paragraph 4)
If this box is checked, there are provisions in this Lease for the Base Ren	
Premises and/or the Project are modified during the term of this Lease, Lessor sha	percent (
1.7 Base Rent and Other Monies Paid Upon Execution:	The second secon
(a) Base Rent: for the period	
(b) Common Area Operating Expenses: The current estimate for the	e period is
(c) Security Deposit: ("Security Deposit"). (See also	Paragraph 5)
(d) Other: for	
(e) Total Due Upon Execution of this Lease:	
1.8 Agreed Use: (See also Paragraph 6)1.9 Insuring Party. Lessor is the "Insuring Party". (See also Paragraph 8)	
1.9 Insuring Party. Lesson's the Insuring Party . (See also Paragraph 8) 1.10 Real Estate Brokers. (See also Paragraphs 15 and 25)	
, , , ,	Regarding Real Estate Agency Relationship, confirms and consents to the following
agency relationships in this Lease with the following real estate brokers ("Broker(s	s)") and/or their agents ("Agent(s)"):
Lessor's Brokerage Firm License No is the brok	ker of (check one): the Lessor; or both the Lessee and Lessor (dual agent).
Lessor's Agent License No. is (check one):	the Lessor's Agent (salesperson or broker associate); or both the Lessee's
Agent and the Lessor's Agent (dual agent).	
Lessee's Brokerage Firm License No Is the brol	ker of (check one): $oxdot$ the Lessee; or $oxdot$ both the Lessee and Lessor (dual agent).
	the Lessee's Agent (salesperson or broker associate); or both the Lessee's
Agent and the Lessor's Agent (dual agent).	G. Harris and A.
	by both Parties, Lessor shall pay to the Brokers the brokerage fee agreed to in a
	or% of the total Base Rent) for the brokerage services rendered
by the Brokers.	(IIC
1.11 Guarantor. The obligations of the Lessee under this Lease are to be guing 1.12 Attachments. Attached hereto are the following, all of which constitutes:	
an Addendum consisting of Paragraphs through	
	
a site plan depicting the Premises;	
a site plan depicting the Project;	
a current set of the Rules and Regulations for the Project;	
a current set of the Rules and Regulations adopted by the owners' associated by the owners associated as a current set of the Rules and Regulations adopted by the owners as a current set of the Rules and Regulations adopted by the owners as a current set of the Rules and Regulations adopted by the owners as a current set of the Rules and Regulations adopted by the owners as a current set of the Rules and Regulations adopted by the owners as a current set of the Rules and Regulations adopted by the owners as a current set of the Rules and Regulations adopted by the owners as a current set of the Rules and Regulations adopted by the owners are current set of the Rules and Regulations adopted by the owners are current set of the Rules and Regulations and Regulations are current set of the Rules and Regulations and Regulations are current set of the Rules and Regulations and Regulations are current set of the Rules and Regulations and Regulations and Regulations are current set of the Rules and Regulations and Regulations are current set of the Rules and Regulations and Regulations are current set of the Rules and Regulations and Regulations are current set of the Rules and Regulations and Regulations are current set of the Rules and Regulations and Regulations are current set of the Rules and Regulations and Regulations are current set of the Rules and	ciation;
a Work Letter;	
other (specify);	
 Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from 	a Lessor, the Premises, for the term, at the rental, and upon all of the terms
covenants and conditions set forth in this Lease. While the approximate square for	·
purposes of comparison, the Base Rent stated herein is NOT tied to square footage	e and is not subject to adjustment should the actual size be determined to be
different. NOTE: Lessee is advised to verify the actual size prior to executing this	
2.2 Condition. Lessor shall deliver that portion of the Premises contained	within the Building (" Unit ") to Lessee broom clean and free of debris on the Date "), and, so long as the required service contracts described in Paragraph 7.1(b)
	Date, warrants that the existing electrical, plumbing, fire sprinkler, lighting, heating,
	any, and all other such elements in the Unit, other than those constructed by Lessee,
	the roof, bearing walls and foundation of the Unit shall be free of material defects,
and that the Unit does not contain hazardous levels of any mold or fungi defined a	
sole obligation with respect to such matter, except as otherwise provided in this Le	malfunction or fail within the appropriate warranty period, Lessor shall, as Lessor's
specificity the nature and extent of such non-compliance, malfunction or failure, r	
months as to the HVAC systems, and (ii) 30 days as to the remaining systems and of	other elements of the Unit. If Lessee does not give Lessor the required notice within
the appropriate warranty period, correction of any such non-compliance, malfunc	·
	ring walls - see Paragraph 7). Lessor also warrants, that unless otherwise specified in nise; (ii) any delinquent amounts due under any loan secured by the Premises; and
(iii) any bankruptcy proceeding affecting the Premises.	ilse, (ii) any demiquent amounts due diluer any loan secured by the Frenises, and
	provements on the Premises and the Common Areas comply with the building codes,
• • • • • • • • • • • • • • • • • • • •	Applicable Requirements") that were in effect at the time that each improvement,
or portion thereof, was constructed. Said warranty does not apply to the use to w	
7.3(a)) made or to be made by Lessee. NOTE: Lessee is responsible for determini	Paragraph 49), or to any Alterations or Utility Installations (as defined in Paragraphing whether or not the Applicable Requirements, and especially the zoning are
appropriate for Lessee's intended use, and acknowledges that past uses of the P	
<u>-</u> .	• •
INITIALS	INITIALS
INTIALS	Last Edited: 9/30/2022 2:33 PM

warranty, Lessor shall, except as otherwise provided, promptly after receipt of written notice from Lessee setting forth with specificity the nature and extent of such non-compliance, rectify the same at Lessor's expense. If Lessee does not give Lessor written notice of a non-compliance with this warranty within 6 months following the Start Date, correction of that non-compliance shall be the obligation of Lessee at Lessee's sole cost and expense. If the Applicable Requirements are hereafter changed so as to require during the term of this Lease the construction of an addition to or an alteration of the Unit, Premises and/or Building, the remediation of any Hazardous Substance, or the reinforcement or other physical modification of the Unit, Premises and/or Building ("Capital Expenditure"), Lessor and Lessee shall allocate the cost of such work as follows:

- (a) Subject to Paragraph 2.3(c) below, if such Capital Expenditures are required as a result of the specific and unique use of the Premises by Lessee as compared with uses by tenants in general, Lessee shall be fully responsible for the cost thereof, provided, however, that if such Capital Expenditure is required during the last 2 years of this Lease and the cost thereof exceeds 6 months' Base Rent, Lessee may instead terminate this Lease unless Lessor notifies Lessee, in writing, within 10 days after receipt of Lessee's termination notice that Lessor has elected to pay the difference between the actual cost thereof and the amount equal to 6 months' Base Rent. If Lessee elects termination, Lessee shall immediately cease the use of the Premises which requires such Capital Expenditure and deliver to Lessor written notice specifying a termination date at least 90 days thereafter. Such termination date shall, however, in no event be earlier than the last day that Lessee could legally utilize the Premises without commencing such Capital Expenditure.
- (b) If such Capital Expenditure is not the result of the specific and unique use of the Premises by Lessee (such as, governmentally mandated seismic modifications), then Lessor shall pay for such Capital Expenditure and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date that on which the Base Rent is due, an amount equal to 1/144th of the portion of such costs reasonably attributable to the Premises. Lessee shall pay Interest on the balance but may prepay its obligation at any time. If, however, such Capital Expenditure is required during the last 2 years of this Lease or if Lessor reasonably determines that it is not economically feasible to pay its share thereof, Lessor shall have the option to terminate this Lease upon 90 days prior written notice to Lessee unless Lessee notifies Lessor, in writing, within 10 days after receipt of Lessor's termination notice that Lessee will pay for such Capital Expenditure. If Lessor does not elect to terminate, and fails to tender its share of any such Capital Expenditure, Lessee may advance such funds and deduct same, with Interest, from Rent until Lessor's share of such costs have been fully paid. If Lessee is unable to finance Lessor's share, or if the balance of the Rent due and payable for the remainder of this Lease is not sufficient to fully reimburse Lessee on an offset basis, Lessee shall have the right to terminate this Lease upon 30 days written notice to Lessor.
- (c) Notwithstanding the above, the provisions concerning Capital Expenditures are intended to apply only to non-voluntary, unexpected, and new Applicable Requirements. If the Capital Expenditures are instead triggered by Lessee as a result of an actual or proposed change in use, change in intensity of use, or modification to the Premises then, and in that event, Lessee shall either: (i) immediately cease such changed use or intensity of use and/or take such other steps as may be necessary to eliminate the requirement for such Capital Expenditure, or (ii) complete such Capital Expenditure at its own expense. Lessee shall not have any right to terminate this Lease.
- 2.4 **Acknowledgements**. Lessee acknowledges that: (a) it has been given an opportunity to inspect and measure the Premises; (b) it has been advised by Lessor and/or Brokers to satisfy itself with respect to the size and condition of the Premises (including but not limited to the electrical, HVAC and fire sprinkler systems, security, environmental aspects, and compliance with Applicable Requirements and the Americans with Disabilities Act), and their suitability for Lessee's intended use; (c) Lessee has made such investigation as it deems necessary with reference to such matters and assumes all responsibility therefor as the same relate to its occupancy of the Premises; (d) it is not relying on any representation as to the size of the Premises made by Brokers or Lessor; (e) the square footage of the Premises was not material to Lessee's decision to lease the Premises and pay the Rent stated herein; and (f) neither Lessor, Lessor's agents, nor Brokers have made any oral or written representations or warranties with respect to said matters other than as set forth in this Lease. In addition, Lessor acknowledges that: (i) Brokers have made no representations, promises or warranties concerning Lessee's ability to honor the Lease or suitability to occupy the Premises, and (ii) it is Lessor's sole responsibility to investigate the financial capability and/or suitability of all proposed tenants.
- 2.5 Lessee as Prior Owner/Occupant. The warranties made by Lessor in Paragraph 2 shall be of no force or effect if immediately prior to the Start Date Lessee was the owner or occupant of the Premises. In such event, Lessee shall be responsible for any necessary corrective work.
- 2.6 **Vehicle Parking**. Lessee shall be entitled to use the number of Parking Spaces specified in Paragraph 1.2(b) on those portions of the Common Areas designated from time to time by Lessor for parking. Lessee shall not use more parking spaces than said number. Said parking spaces shall be used for parking by vehicles no larger than full-size passenger automobiles or pick-up trucks, herein called "**Permitted Size Vehicles**." Lessor may regulate the loading and unloading of vehicles by adopting Rules and Regulations as provided in Paragraph 2.9. No vehicles other than Permitted Size Vehicles may be parked in the Common Area without the prior written permission of Lessor. In addition:
- (a) Lessee shall not permit or allow any vehicles that belong to or are controlled by Lessee or Lessee's employees, suppliers, shippers, customers, contractors or invitees to be loaded, unloaded, or parked in areas other than those designated by Lessor for such activities.
 - (b) Lessee shall not service or store any vehicles in the Common Areas.
- (c) If Lessee permits or allows any of the prohibited activities described in this Paragraph 2.6, then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove or tow away the vehicle involved and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.
- 2.7 **Common Areas Definition**. The term "**Common Areas**" is defined as all areas and facilities outside the Premises and within the exterior boundary line of the Project and interior utility raceways and installations within the Unit that are provided and designated by the Lessor from time to time for the general non-exclusive use of Lessor, Lessee and other tenants of the Project and their respective employees, suppliers, shippers, customers, contractors and invitees, including parking areas, loading and unloading areas, trash areas, roofs, roadways, walkways, driveways and landscaped areas.
- parking areas, loading and unloading areas, trash areas, roofs, roadways, walkways, driveways and landscaped areas.

 2.8 Common Areas Lessee's Rights. Lessor grants to Lessee, for the benefit of Lessee and its employees, suppliers, shippers, contractors, customers and invitees, during the term of this Lease, the non-exclusive right to use, in common with others entitled to such use, the Common Areas as they exist from time to time, subject to any rights, powers, and privileges reserved by Lessor under the terms hereof or under the terms of any rules and regulations or restrictions governing the use of the Project. Under no circumstances shall the right herein granted to use the Common Areas be deemed to include the right to store any property, temporarily or permanently, in the Common Areas. Any such storage shall be permitted only by the prior written consent of Lessor or Lessor's designated agent, which consent may be revoked at any time. In the event that any unauthorized storage shall occur, then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove the property and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.
- 2.9 Common Areas Rules and Regulations. Lessor or such other person(s) as Lessor may appoint shall have the exclusive control and management of the Common Areas and shall have the right, from time to time, to establish, modify, amend and enforce reasonable rules and regulations ("Rules and Regulations") for the management, safety, care, and cleanliness of the grounds, the parking and unloading of vehicles and the preservation of good order, as well as for the convenience of other occupants or tenants of the Building and the Project and their invitees. Lessee agrees to abide by and conform to all such Rules and Regulations, and shall use its best efforts to cause its employees, suppliers, shippers, customers, contractors and invitees to so abide and conform. Lessor shall not be responsible to Lessee for the non-compliance with said Rules and Regulations by other tenants of the Project.
 - 2.10 **Common Areas Changes**. Lessor shall have the right, in Lessor's sole discretion, from time to time:
- (a) To make changes to the Common Areas, including, without limitation, changes in the location, size, shape and number of driveways, entrances, parking spaces, parking areas, loading and unloading areas, ingress, egress, direction of traffic, landscaped areas, walkways and utility raceways;
 - (b) To close temporarily any of the Common Areas for maintenance purposes so long as reasonable access to the Premises remains available;
 - (c) To designate other land outside the boundaries of the Project to be a part of the Common Areas;
 - d) To add additional buildings and improvements to the Common Areas;
 - (e) To use the Common Areas while engaged in making additional improvements, repairs or alterations to the Project, or any portion thereof; and
- (f) To do and perform such other acts and make such other changes in, to or with respect to the Common Areas and Project as Lessor may, in the exercise of sound business judgment, deem to be appropriate.
- 3. Term.

INI

- 3.1 Term. The Commencement Date, Expiration Date and Original Term of this Lease are as specified in Paragraph 1.3.
- 3.2 **Early Possession.** Any provision herein granting Lessee Early Possession of the Premises is subject to and conditioned upon the Premises being available for such possession prior to the Commencement Date. Any grant of Early Possession only conveys a non-exclusive right to occupy the Premises. If Lessee totally or partially occupies the Premises prior to the Commencement Date, the obligation to pay Base Rent shall be abated for the period of such Early Possession. All other terms of this Lease (including but not limited to the obligations to pay Lessee's Share of Common Area Operating Expenses, Real Property Taxes and insurance premiums and to maintain the Premises) shall be in effect during such period. Any such Early Possession shall not affect the Expiration Date.
- 3.3 **Delay In Possession**. Lessor agrees to use commercially reasonable efforts to deliver exclusive possession of the Premises to Lessee by the Commencement Date. If, despite said efforts, Lessor is unable to deliver possession by such date, Lessor shall not be subject to any liability therefor, nor shall such failure affect the validity of this Lease or change the Expiration Date. Lessee shall not, however, be obligated to pay Rent or perform its other obligations until Lessor delivers possession of the Premises and any period of rent abatement that Lessee would otherwise have enjoyed shall run from the date of delivery of possession and continue for a period equal to what Lessee would otherwise have enjoyed under the terms hereof, but minus any days of delay caused by the acts or omissions of Lessee. If possession is not delivered within 60 days after the Commencement Date, as the same may be extended under the terms of any Work Letter executed by

INITIAI

Parties, Lessee may, at its option, by notice in writing within 10 days after the end of such 60 day period, cancel this Lease, in which event the Parties shall be discharged from all obligations hereunder. If such written notice is not received by Lessor within said 10 day period, Lessee's right to cancel shall terminate. If possession of the Premises is not delivered within 120 days after the Commencement Date, this Lease shall terminate unless other agreements are reached between Lessor and Lessee, in writing.

3.4 Lessee Compliance. Lessor shall not be required to tender possession of the Premises to Lessee until Lessee complies with its obligation to provide evidence of insurance (Paragraph 8.5). Pending delivery of such evidence, Lessee shall be required to perform all of its obligations under this Lease from and after the Start Date, including the payment of Rent, notwithstanding Lessor's election to withhold possession pending receipt of such evidence of insurance. Further, if Lessee is required to perform any other conditions prior to or concurrent with the Start Date, the Start Date shall occur but Lessor may elect to withhold possession until such conditions are satisfied.

4. Rent.

- 4.1 Rent Defined. All monetary obligations of Lessee to Lessor under the terms of this Lease (except for the Security Deposit) are deemed to be rent ("Rent").
- 4.2 **Common Area Operating Expenses.** Lessee shall pay to Lessor during the term hereof, in addition to the Base Rent, Lessee's Share (as specified in Paragraph 1.6) of all Common Area Operating Expenses, as hereinafter defined, during each calendar year of the term of this Lease, in accordance with the following provisions:
 - (a) The following costs relating to the ownership and operation of the Project are defined as "Common Area Operating Expenses":
- (i) Costs relating to the operation, repair and maintenance, in neat, clean, good order and condition, but not the replacement (see subparagraph (e)), of the following:
- (aa) The Common Areas and Common Area improvements, including parking areas, loading and unloading areas, trash areas, roadways, parkways, walkways, driveways, landscaped areas, bumpers, irrigation systems, Common Area lighting facilities, fences and gates, elevators, roofs, exterior walls of the buildings, building systems and roof drainage systems.
 - (bb) Exterior signs and any tenant directories.
 - (cc) Any fire sprinkler systems.
- (dd) All other areas and improvements that are within the exterior boundaries of the Project but outside of the Premises and/or any other space occupied by a tenant.
 - (ii) The cost of water, gas, electricity and telephone to service the Common Areas and any utilities not separately metered.
- (iii) The cost of trash disposal, pest control services, property management, security services, owners' association dues and fees, the cost to repaint the exterior of any structures and the cost of any environmental inspections.
 - (iv) Reserves set aside for maintenance and repair of Common Areas and Common Area equipment.
 - (v) Any increase above the Base Real Property Taxes (as defined in Paragraph 10).
 - (vi) Any "Insurance Cost Increase" (as defined in Paragraph 8).
 - (vii) Any deductible portion of an insured loss concerning the Building or the Common Areas.
 - (viii) Auditors', accountants' and attorneys' fees and costs related to the operation, maintenance, repair and replacement of the Project.
- (ix) The cost of any capital improvement to the Building or the Project not covered under the provisions of Paragraph 2.3 provided; however, that Lessor shall allocate the cost of any such capital improvement over a 12 year period and Lessee shall not be required to pay more than Lessee's Share of 1/144th of the cost of such capital improvement in any given month. Lessee shall pay Interest on the unamortized balance but may prepay its obligation at any time.
 - (x) The cost of any other services to be provided by Lessor that are stated elsewhere in this Lease to be a Common Area Operating Expense.
- (b) Any Common Area Operating Expenses and Real Property Taxes that are specifically attributable to the Unit, the Building or to any other building in the Project or to the operation, repair and maintenance thereof, shall be allocated entirely to such Unit, Building, or other building. However, any Common Area Operating Expenses and Real Property Taxes that are not specifically attributable to the Building or to any other building or to the operation, repair and maintenance thereof, shall be equitably allocated by Lessor to all buildings in the Project.
- (c) The inclusion of the improvements, facilities and services set forth in Subparagraph 4.2(a) shall not be deemed to impose an obligation upon Lessor to either have said improvements or facilities or to provide those services unless the Project already has the same, Lessor already provides the services, or Lessor has agreed elsewhere in this Lease to provide the same or some of them.
- (d) Lessee's Share of Common Area Operating Expenses is payable monthly on the same day as the Base Rent is due hereunder. The amount of such payments shall be based on Lessor's estimate of the annual Common Area Operating Expenses. Within 60 days after written request (but not more than once each year) Lessor shall deliver to Lessee a reasonably detailed statement showing Lessee's Share of the actual Common Area Operating Expenses for the preceding year. If Lessee's payments during such year exceed Lessee's Share, Lessor shall credit the amount of such over-payment against Lessee's future payments. If Lessee's payments during such year were less than Lessee's Share, Lessee shall pay to Lessor the amount of the deficiency within 10 days after delivery by Lessor to Lessee of the statement.
- (e) Common Area Operating Expenses shall not include the cost of replacing equipment or capital components such as the roof, foundations, exterior walls or Common Area capital improvements, such as the parking lot paving, elevators, fences that have a useful life for accounting purposes of 5 years or more.
- (f) Common Area Operating Expenses shall not include any expenses paid by any tenant directly to third parties, or as to which Lessor is otherwise reimbursed by any third party, other tenant, or insurance proceeds.
- 4.3 **Payment.** Lessee shall cause payment of Rent to be received by Lessor in lawful money of the United States, without offset or deduction (except as specifically permitted in this Lease), on or before the day on which it is due. All monetary amounts shall be rounded to the nearest whole dollar. In the event that any statement or invoice prepared by Lessor is inaccurate such inaccuracy shall not constitute a waiver and Lessee shall be obligated to pay the amount set forth in this Lease. Rent for any period during the term hereof which is for less than one full calendar month shall be prorated based upon the actual number of days of said month. Payment of Rent shall be made to Lessor at its address stated herein or to such other persons or place as Lessor may from time to time designate in writing. Acceptance of a payment which is less than the amount then due shall not be a waiver of Lessor's rights to the balance of such Rent, regardless of Lessor's endorsement of any check so stating. In the event that any check, draft, or other instrument of payment given by Lessee to Lessor is dishonored for any reason, Lessee agrees to pay to Lessor the sum of \$25 in addition to any Late Charge and Lessor, at its option, may require all future Rent be paid by cashier's check. Payments will be applied first to accrued late charges and attorney's fees, second to accrued interest, then to Base Rent and Common Area Operating Expenses, and any remaining amount to any other outstanding charges or costs.
- Security Deposit. Lessee shall deposit with Lessor upon execution hereof the Security Deposit as security for Lessee's faithful performance of its obligations under this Lease. If Lessee fails to pay Rent, or otherwise Defaults under this Lease, Lessor may use, apply or retain all or any portion of said Security Deposit for the payment of any amount already due Lessor, for Rents which will be due in the future, and/or to reimburse or compensate Lessor for any liability, expense, loss or damage which Lessor may suffer or incur by reason thereof. If Lessor uses or applies all or any portion of the Security Deposit, Lessee shall within 10 days after written request therefor deposit monies with Lessor sufficient to restore said Security Deposit to the full amount required by this Lease. If the Base Rent increases during the term of this Lease. Lessee shall, upon written request from Lessor, deposit additional monies with Lessor so that the total amount of the Security Deposit shall at all times bear the same proportion to the increased Base Rent as the initial Security Deposit bore to the initial Base Rent. Should the Agreed Use be amended to accommodate a material change in the business of Lessee or to accommodate a sublessee or assignee, Lessor shall have the right to increase the Security Deposit to the extent necessary, in Lessor's reasonable judgment, to account for any increased wear and tear that the Premises may suffer as a result thereof. If a change in control of Lessee occurs during this Lease and following such change the financial condition of Lessee is, in Lessor's reasonable judgment, significantly reduced, Lessee shall deposit such additional monies with Lessor as shall be sufficient to cause the Security Deposit to be at a commercially reasonable level based on such change in financial condition. Lessor shall not be required to keep the Security Deposit separate from its general accounts. Within 90 days after the expiration or termination of this Lease, Lessor shall return that portion of the Security Deposit not used or applied by Lessor. Lessor shall upon written request provide Lessee with an accounting showing how that portion of the Security Deposit that was not returned was applied. No part of the Security Deposit shall be considered to be held in trust, to bear interest or to be prepayment for any monies to be paid by Lessee under this Lease. THE SECURITY DEPOSIT SHALL NOT BE USED BY LESSEE IN LIEU OF PAYMENT OF THE LAST MONTH'S RENT.

6. Use.

6.1 **Use**. Lessee shall use and occupy the Premises only for the Agreed Use, or any other legal use which is reasonably comparable thereto, and for no other purpose. Lessee shall not use or permit the use of the Premises in a manner that is unlawful, creates damage, waste or a nuisance, or that disturbs occupants of or causes damage to neighboring premises or properties. Other than guide, signal and seeing eye dogs, Lessee shall not keep or allow in the Premises any pets, animals, birds, fish, or reptiles. Lessor shall not unreasonably withhold or delay its consent to any written request for a modification of the Agreed Use, so long as the same will not impair the structural integrity of the Building or the mechanical or electrical systems therein, and/or is not significantly more burdensome to the Project. If Lessor elects to withhold consent, Lessor shall within 7 days after such request give written notification of same, which notice shall include an explanation of Lessor's objections to the change in the Agreed Use.

6.2 Hazardous Substances.

INITIALS	5	LS	Α	1	ΙT	٧	11	
----------	---	----	---	---	----	---	----	--

- (a) Reportable Uses Require Consent. The term "Hazardous Substance" as used in this Lease shall mean any product, substance, or waste whose presence, use, manufacture, disposal, transportation, or release, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentially injurious to the public health, safety or welfare, the environment or the Premises; (ii) regulated or monitored by any governmental authority, or (iii) a basis for potential liability of Lessor to any governmental agency or third party under any applicable statute or common law theory. Hazardous Substances shall include, but not be limited to, hydrocarbons, petroleum, gasoline, and/or crude oil or any products, by-products or fractions thereof. Lessee shall not engage in any activity in or on the Premises which constitutes a Reportable Use of Hazardous Substances without the express prior written consent of Lessor and timely compliance (at Lessee's expense) with all Applicable Requirements. "Reportable Use" shall mean (i) the installation or use of any above or below ground storage tank; (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and/or (iii) the presence at the Premises of a Hazardous Substance with respect to which any Applicable Requirements requires that a notice be given to persons entering or occupying the Premises or neighboring properties. Notwithstanding the foregoing. Lessee may use any ordinary and customary materials reasonably required to be used in the normal course of the Agreed Use, ordinary office supplies (copier toner, liquid paper, glue, etc.) and common household cleaning materials, so long as such use is in compliance with all Applicable Requirements, is not a Reportable Use, and does not expose the Premises or neighboring property to any meaningful risk of contamination or damage or expose Lessor to any liability therefor. In addition, Lessor may condition its consent to any Reportable Use upon receiving such additional assurances as Lessor reasonably deems necessary to protect itself, the public, the Premises and/or the environment against damage, contamination, injury and/or liability, including, but not limited to, the installation (and removal on or before Lease expiration or termination) of protective modifications (such as concrete encasements) and/or increasing the Security Deposit.
- (b) **Duty to Inform Lessor**. If Lessee knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises, other than as previously consented to by Lessor, Lessee shall immediately give written notice of such fact to Lessor, and provide Lessor with a copy of any report, notice, claim or other documentation which it has concerning the presence of such Hazardous Substance.
- (c) Lessee Remediation. Lessee shall not cause or permit any Hazardous Substance to be spilled or released in, on, under, or about the Premises (including through the plumbing or sanitary sewer system) and shall promptly, at Lessee's expense, comply with all Applicable Requirements and take all investigatory and/or remedial action reasonably recommended, whether or not formally ordered or required, for the cleanup of any contamination of, and for the maintenance, security and/or monitoring of the Premises or neighboring properties, that was caused or materially contributed to by Lessee, or pertaining to or involving any Hazardous Substance brought onto the Premises during the term of this Lease, by or for Lessee, or any third party.
- (d) Lessee Indemnification. Lessee shall indemnify, defend and hold Lessor, its agents, employees, lenders and ground lessor, if any, harmless from and against any and all loss of rents and/or damages, liabilities, judgments, claims, expenses, penalties, and attorneys' and consultants' fees arising out of or involving any Hazardous Substance brought onto the Premises by or for Lessee, or any third party (provided, however, that Lessee shall have no liability under this Lease with respect to underground migration of any Hazardous Substance under the Premises from areas outside of the Project not caused or contributed to by Lessee). Lessee's obligations shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Lessee, and the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease. No termination, cancellation or release agreement entered into by Lessor and Lessee shall release Lessee from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Lessor in writing at the time of such agreement.
- (e) Lessor Indemnification. Except as otherwise provided in paragraph 8.7, Lessor and its successors and assigns shall indemnify, defend, reimburse and hold Lessee, its employees and lenders, harmless from and against any and all environmental damages, including the cost of remediation, which are suffered as a direct result of Hazardous Substances on the Premises prior to Lessee taking possession or which are caused by the gross negligence or willful misconduct of Lessor, its agents or employees. Lessor's obligations, as and when required by the Applicable Requirements, shall include, but not be limited to, the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease.
- (f) Investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by governmental entities having jurisdiction with respect to the existence of Hazardous Substances on the Premises prior to the Lessee taking possession, unless such remediation measure is required as a result of Lessee's use (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises, in which event Lessee shall be responsible for such payment. Lesser's agents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remedial responsibilities.
- (g) Lessor Termination Option. If a Hazardous Substance Condition (see Paragraph 9.1(e)) occurs during the term of this Lease, unless Lessee is legally responsible therefor (in which case Lessee shall make the investigation and remediation thereof required by the Applicable Requirements and this Lease shall continue in full force and effect, but subject to Lessor's rights under Paragraph 6.2(d) and Paragraph 13), Lessor may, at Lessor's option, either (i) investigate and remediate such Hazardous Substance Condition, if required, as soon as reasonably possible at Lessor's expense, in which event this Lease shall continue in full force and effect, or (ii) if the estimated cost to remediate such condition exceeds 12 times the then monthly Base Rent or \$100,000, whichever is greater, give written notice to Lessee, within 30 days after receipt by Lessor of knowledge of the occurrence of such Hazardous Substance Condition, of Lessor's desire to terminate this Lease as of the date 60 days following the date of such notice. In the event Lessor elects to give a termination notice, Lessee may, within 10 days thereafter, give written notice to Lessor of Lessee's commitment to pay the amount by which the cost of the remediation of such Hazardous Substance Condition exceeds an amount equal to 12 times the then monthly Base Rent or \$100,000, whichever is greater. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days following such commitment. In such event, this Lease shall continue in full force and effect, and Lessor shall proceed to make such remediation as soon as reasonably possible after the required funds are available. If Lessee does not give such notice and provide the required funds or assurance thereof within the time provided, this Lease shall terminate as of the date specified in Lessor's notice of termination.

 6.3 Lessee's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lessee's sole expense, fully, diligently and
- 6.3 Lessee's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lessee's sole expense, fully, diligently and in a timely manner, materially comply with all Applicable Requirements, the requirements of any applicable fire insurance underwriter or rating bureau, and the recommendations of Lessor's engineers and/or consultants which relate in any manner to the Premises, without regard to whether said Applicable Requirements are now in effect or become effective after the Start Date. Lessee shall, within 10 days after receipt of Lessor's written request, provide Lessor with copies of all permits and other documents, and other information evidencing Lessee's compliance with any Applicable Requirements specified by Lessor, and shall immediately upon receipt, notify Lessor in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving the failure of Lessee or the Premises to comply with any Applicable Requirements. Likewise, Lessee shall immediately give written notice to Lessor of: (i) any water damage to the Premises and any suspected seepage, pooling, dampness or other condition conducive to the production of mold; or (ii) any mustiness or other odors that might indicate the presence of mold in the Premises.
- other odors that might indicate the presence of mold in the Premises.

 6.4 Inspection; Compliance. Lessor and Lessor's "Lender" (as defined in Paragraph 30) and consultants authorized by Lessor shall have the right to enter into Premises at any time in the case of an emergency, and otherwise at reasonable times after reasonable notice, for the purpose of inspecting and/or testing the condition of the Premises and/or for verifying compliance by Lessee with this Lease. The cost of any such inspections shall be paid by Lessor, unless a violation of Applicable Requirements, or a Hazardous Substance Condition (see Paragraph 9.1(e)) is found to exist or be imminent, or the inspection is requested or ordered by a governmental authority. In such case, Lessee shall upon request reimburse Lessor for the cost of such inspection, so long as such inspection is reasonably related to the violation or contamination. In addition, Lessee shall provide copies of all relevant material safety data sheets (MSDS) to Lessor within 10 days of the receipt of written request therefor. Lessee acknowledges that any failure on its part to allow such inspections or testing will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to allow such inspections and/or testing in a timely fashion the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for the remainder to the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to allow such inspection and/or testing. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to such failure nor prevent the exercise of any of the other rights and remedies gr

7. Maintenance; Repairs; Utility Installations; Trade Fixtures and Alterations.

7.1 Lessee's Obligations

- (a) In General. Subject to the provisions of Paragraph 2.2 (Condition), 2.3 (Compliance), 6.3 (Lessee's Compliance with Applicable Requirements), 7.2 (Lessor's Obligations), 9 (Damage or Destruction), and 14 (Condemnation), Lessee shall, at Lessee's sole expense, keep the Premises, Utility Installations (intended for Lessee's exclusive use, no matter where located), and Alterations in good order, condition and repair (whether or not the portion of the Premises requiring repairs, or the means of repairing the same, are reasonably or readily accessible to Lessee, and whether or not the need for such repairs occurs as a result of Lessee's use, any prior use, the elements or the age of such portion of the Premises), including, but not limited to, all equipment or facilities, such as plumbing, HVAC equipment, electrical, lighting facilities, boilers, pressure vessels, fixtures, interior walls, interior surfaces of exterior walls, ceilings, floors, windows, doors, plate glass, and skylights but excluding any items which are the responsibility of Lessor pursuant to Paragraph 7.2. Lessee, in keeping the Premises in good order, condition and repair, shall exercise and perform good maintenance practices, specifically including the procurement and maintenance of the service contracts required by Paragraph 7.1(b) below. Lessee's obligations shall include restorations, replacements or renewals when necessary to keep the Premises and all improvements thereon or a part thereof in good order, condition and state of repair.
 - (b) Service Contracts. Lessee shall, at Lessee's sole expense, procure and maintain contracts, with copies to Lessor, in customary form and substance for,

and with contractors specializing and experienced in the maintenance of the following equipment and improvements, if any, if and when installed on the Premises: (i) HVAC equipment, (ii) boiler and pressure vessels, and (iii) clarifiers. However, Lessor reserves the right, upon notice to Lessee, to procure and maintain any or all of such service contracts, and Lessee shall reimburse Lessor, upon demand, for the cost thereof.

- (c) Failure to Perform. If Lessee fails to perform Lessee's obligations under this Paragraph 7.1, Lessor may enter upon the Premises after 10 days' prior written notice to Lessee (except in the case of an emergency, in which case no notice shall be required), perform such obligations on Lessee's behalf, and put the Premises in good order, condition and repair, and Lessee shall promptly pay to Lessor a sum equal to 115% of the cost thereof.
- (d) **Replacement**. Subject to Lessee's indemnification of Lessor as set forth in Paragraph 8.7 below, and without relieving Lessee of liability resulting from Lessee's failure to exercise and perform good maintenance practices, if an item described in Paragraph 7.1(b) cannot be repaired other than at a cost which is in excess of 50% of the cost of replacing such item, then such item shall be replaced by Lessor, and the cost thereof shall be prorated between the Parties and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date on which Base Rent is due, an amount equal to the product of multiplying the cost of such replacement by a fraction, the numerator of which is one, and the denominator of which is 144 (i.e. 1/144th of the cost per month). Lessee shall pay Interest on the unamortized balance but may prepay its obligation at any time.
- 7.2 **Lessor's Obligations**. Subject to the provisions of Paragraphs 2.2 (Condition), 2.3 (Compliance), 4.2 (Common Area Operating Expenses), 6 (Use), 7.1 (Lessee's Obligations), 9 (Damage or Destruction) and 14 (Condemnation), Lessor, subject to reimbursement pursuant to Paragraph 4.2, shall keep in good order, condition and repair the foundations, exterior walls, structural condition of interior bearing walls, exterior roof, fire sprinkler system, Common Area fire alarm and/or smoke detection systems, fire hydrants, parking lots, walkways, parkways, driveways, landscaping, fences, signs and utility systems serving the Common Areas and all parts thereof, as well as providing the services for which there is a Common Area Operating Expense pursuant to Paragraph 4.2. Lessor shall not be obligated to paint the exterior or interior surfaces of exterior walls nor shall Lessor be obligated to maintain, repair or replace windows, doors or plate glass of the Premises.

7.3 Utility Installations: Trade Fixtures: Alterations.

- (a) **Definitions.** The term "**Utility Installations**" refers to all floor and window coverings, air and/or vacuum lines, power panels, electrical distribution, security and fire protection systems, communication cabling, lighting fixtures, HVAC equipment, plumbing, and fencing in or on the Premises. The term "**Trade Fixtures**" shall mean Lessee's machinery and equipment that can be removed without doing material damage to the Premises. The term "Alterations" shall mean any modification of the improvements, other than Utility Installations or Trade Fixtures, whether by addition or deletion. "**Lessee Owned Alterations and/or Utility Installations**" are defined as Alterations and/or Utility Installations made by Lessee that are not yet owned by Lessor pursuant to Paragraph 7.4(a).
- (b) Consent. Lessee shall not make any Alterations or Utility Installations to the Premises without Lessor's prior written consent. Lessee may, however, make non-structural Alterations or Utility Installations to the interior of the Premises (excluding the roof) without such consent but upon notice to Lessor, as long as they are not visible from the outside, do not involve puncturing, relocating or removing the roof or any existing walls, will not affect the electrical, plumbing, HVAC, and/or life safety systems, do not trigger the requirement for additional modifications and/or improvements to the Premises resulting from Applicable Requirements, such as compliance with Title 24, and the cumulative cost thereof during this Lease as extended does not exceed a sum equal to 3 month's Base Rent in the aggregate or a sum equal to one month's Base Rent in any one year. Notwithstanding the foregoing, Lessee shall not make or permit any roof penetrations and/or install anything on the roof without the prior written approval of Lessor. Lessor may, as a precondition to granting such approval, require Lessee to utilize a contractor chosen and/or approved by Lessor. Any Alterations or Utility Installations that Lessee shall desire to make and which require the consent of the Lessor shall be presented to Lessor in written form with detailed plans. Consent shall be deemed conditioned upon Lessee's: (i) acquiring all applicable governmental permits, (ii) furnishing Lessor with copies of both the permits and the plans and specifications prior to commencement of the work, and (iii) compliance with all conditions of said permits and other Applicable Requirements in a prompt and expeditious manner. Any Alterations or Utility Installations shall be performed in a workmanlike manner with good and sufficient materials. Lessee shall promptly upon completion furnish Lessor with as-built plans and specifications. For work which costs an amount in excess of one month's Base Rent, Lessor may condition its consent upon Lessee's posting an
- (c) Liens; Bonds. Lessee shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lessee at or for use on the Premises, which claims are or may be secured by any mechanic's or materialmen's lien against the Premises or any interest therein. Lessee shall give Lessor not less than 10 days notice prior to the commencement of any work in, on or about the Premises, and Lessor shall have the right to post notices of non-responsibility. If Lessee shall contest the validity of any such lien, claim or demand, then Lessee shall, at its sole expense defend and protect itself, Lessor and the Premises against the same and shall pay and satisfy any such adverse judgment that may be rendered thereon before the enforcement thereof. If Lessor shall require, Lessee shall furnish a surety bond in an amount equal to 150% of the amount of such contested lien, claim or demand, indemnifying Lessor against liability for the same. If Lessor elects to participate in any such action, Lessee shall pay Lessor's attorneys' fees and costs.

7.4 Ownership; Removal; Surrender; and Restoration.

- (a) **Ownership**. Subject to Lessor's right to require removal or elect ownership as hereinafter provided, all Alterations and Utility Installations made by Lessee shall be the property of Lessee, but considered a part of the Premises. Lessor may, at any time, elect in writing to be the owner of all or any specified part of the Lessee Owned Alterations and Utility Installations. Unless otherwise instructed per paragraph 7.4(b) hereof, all Lessee Owned Alterations and Utility Installations shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.
- shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.

 (b) Removal. By delivery to Lessee of written notice from Lessor not earlier than 90 and not later than 30 days prior to the end of the term of this Lease, Lessor may require that any or all Lessee Owned Alterations or Utility Installations be removed by the expiration or termination of this Lease. Lessor may require the removal at any time of all or any part of any Lessee Owned Alterations or Utility Installations made without the required consent.

 (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements,
- (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements, parts and surfaces thereof broom clean and free of debris, and in good operating order, condition and state of repair, ordinary wear and tear excepted. "Ordinary wear and tear" shall not include any damage or deterioration that would have been prevented by good maintenance practice. Notwithstanding the foregoing and the provisions of Paragraph 7.1(a), if the Lessee occupies the Premises for 12 months or less, then Lessee shall surrender the Premises in the same condition as delivered to Lessee on the Start Date with NO allowance for ordinary wear and tear. Lessee shall repair any damage occasioned by the installation, maintenance or removal of Trade Fixtures, Lessee owned Alterations and/or Utility Installations, furnishings, and equipment as well as the removal of any storage tank installed by or for Lessee. Lessee shall also remove from the Premises any and all Hazardous Substances brought onto the Premises by or for Lessee, or any third party (except Hazardous Substances which were deposited via underground migration from areas outside of the Premises) to the level specified in Applicable Requirements. Trade Fixtures shall remain the property of Lessee and shall be removed by Lessee. Any personal property of Lessee not removed on or before the Expiration Date or any earlier termination date shall be deemed to have been abandoned by Lessee and may be disposed of or retained by Lessor as Lessor may desire. The failure by Lessee to timely vacate the Premises pursuant to this Paragraph 7.4(c) without the express written consent of Lessor shall constitute a holdover under the provisions of Paragraph 26 below.

8. Insurance; Indemnity.

8.1 Payment of Premium Increases.

- (a) As used herein, the term "Insurance Cost Increase" is defined as any increase in the actual cost of the insurance applicable to the Building and/or the Project and required to be carried by Lessor, pursuant to Paragraphs 8.2(b), 8.3(a) and 8.3(b), over and above the Base Premium, as hereinafter defined, calculated on an annual basis. Insurance Cost Increase shall include, but not be limited to, requirements of the holder of a mortgage or deed of trust covering the Premises, Building and/or Project, increased valuation of the Premises, Building and/or Project, and/or a general premium rate increase. The term Insurance Cost Increase shall not, however, include any premium increases resulting from the nature of the occupancy of any other tenant of the Building. The "Base Premium" shall be the annual premium applicable to the 12 month period immediately preceding the Start Date. If, however, the Project was not insured for the entirety of such 12 month period, then the Base Premium shall be the lowest annual premium reasonably obtainable for the Required Insurance as of the Start Date, assuming the most nominal use possible of the Building. In no event, however, shall Lessee be responsible for any portion of the premium cost attributable to liability insurance coverage in excess of \$2,000,000 procured under Paragraph 8.2(b).
- (b) Lessee shall pay any Insurance Cost Increase to Lessor pursuant to Paragraph 4.2. Premiums for policy periods commencing prior to, or extending beyond, the term of this Lease shall be prorated to coincide with the corresponding Start Date or Expiration Date.

8.2 Liability Insurance.

- (a) Carried by Lessee. Lessee shall obtain and keep in force a Commercial General Liability policy of insurance protecting Lessee and Lessor as an additional insured against claims for bodily injury, personal injury and property damage based upon or arising out of the ownership, use, occupancy or maintenance of the Premises and all areas appurtenant thereto. Such insurance shall be on an occurrence basis providing single limit coverage in an amount not less than \$1,000,000 per occurrence with an annual aggregate of not less than \$2,000,000. Lessee shall add Lessor as an additional insured by means of an endorsement at least as broad as the Insurance Service Organization's "Additional Insured-Managers or Lessors of Premises" Endorsement. The policy shall not contain any intra-insured exclusions as between insured persons or organizations, but shall include coverage for liability assumed under this Lease as an "insured contract" for the performance of Lessee's indemnity obligations under this Lease. The limits of said insurance shall not, however, limit the liability of Lessee nor relieve Lessee of any obligation hereunder. Lessee shall provide an endorsement on its liability policy(ies) which provides that its insurance shall be primary to and not contributory with any similar insurance carried by Lessor, whose insurance shall be considered excess insurance only.
 - (b) Carried by Lessor. Lessor shall maintain liability insurance as described in Paragraph 8.2(a), in addition to, and not in lieu of, the insurance required to

be maintained by Lessee. Lessee shall not be named as an additional insured therein.

8.3 Property Insurance - Building, Improvements and Rental Value.

- (a) **Building and Improvements.** Lessor shall obtain and keep in force a policy or policies of insurance in the name of Lessor, with loss payable to Lessor, any ground-lessor, and to any Lender insuring loss or damage to the Premises. The amount of such insurance shall be equal to the full insurable replacement cost of the Premises, as the same shall exist from time to time, or the amount required by any Lender, but in no event more than the commercially reasonable and available insurable value thereof. Lessee Owned Alterations and Utility Installations, Trade Fixtures, and Lessee's personal property shall be insured by Lessee not by Lessor. If the coverage is available and commercially appropriate, such policy or policies shall insure against all risks of direct physical loss or damage (except the perils of flood and/or earthquake unless required by a Lender), including coverage for debris removal and the enforcement of any Applicable Requirements requiring the upgrading, demolition, reconstruction or replacement of any portion of the Premises as the result of a covered loss. Said policy or policies shall also contain an agreed valuation provision in lieu of any coinsurance clause, waiver of subrogation, and inflation guard protection causing an increase in the annual property insurance coverage amount by a factor of not less than the adjusted U.S. Department of Labor Consumer Price Index for All Urban Consumers for the city nearest to where the Premises are located. If such insurance coverage has a deductible clause, the deductible amount shall not exceed \$5,000 per occurrence.
- (b) Rental Value. Lessor shall also obtain and keep in force a policy or policies in the name of Lessor with loss payable to Lessor and any Lender, insuring the loss of the full Rent for one year with an extended period of indemnity for an additional 180 days ("Rental Value insurance"). Said insurance shall contain an agreed valuation provision in lieu of any coinsurance clause, and the amount of coverage shall be adjusted annually to reflect the projected Rent otherwise payable by Lessee, for the next 12 month period.
- (c) Adjacent Premises. Lessee shall pay for any increase in the premiums for the property insurance of the Building and for the Common Areas or other buildings in the Project if said increase is caused by Lessee's acts, omissions, use or occupancy of the Premises.
- (d) Lessee's Improvements. Since Lessor is the Insuring Party, Lessor shall not be required to insure Lessee Owned Alterations and Utility Installations unless the item in question has become the property of Lessor under the terms of this Lease.

8.4 Lessee's Property; Business Interruption Insurance; Worker's Compensation Insurance.

- (a) **Property Damage**. Lessee shall obtain and maintain insurance coverage on all of Lessee's personal property, Trade Fixtures, and Lessee Owned Alterations and Utility Installations. Such insurance shall be full replacement cost coverage with a deductible of not to exceed \$1,000 per occurrence. The proceeds from any such insurance shall be used by Lessee for the replacement of personal property, Trade Fixtures and Lessee Owned Alterations and Utility Installations.
- (b) **Business Interruption**. Lessee shall obtain and maintain loss of income and extra expense insurance in amounts as will reimburse Lessee for direct or indirect loss of earnings attributable to all perils commonly insured against by prudent lessees in the business of Lessee or attributable to prevention of access to the Premises as a result of such perils.
- (c) Worker's Compensation Insurance. Lessee shall obtain and maintain Worker's Compensation Insurance in such amount as may be required by Applicable Requirements. Such policy shall include a 'Waiver of Subrogation' endorsement. Lessee shall provide Lessor with a copy of such endorsement along with the certificate of insurance or copy of the policy required by paragraph 8.5.
- (d) No Representation of Adequate Coverage. Lessor makes no representation that the limits or forms of coverage of insurance specified herein are adequate to cover Lessee's property, business operations or obligations under this Lease.
- 8.5 Insurance Policies. Insurance required herein shall be by companies maintaining during the policy term a "General Policyholders Rating" of at least A-, VII, as set forth in the most current issue of "Best's Insurance Guide", or such other rating as may be required by a Lender. Lessee shall not do or permit to be done anything which invalidates the required insurance policies. Lessee shall, prior to the Start Date, deliver to Lessor certified copies of policies of such insurance or certificates with copies of the required endorsements evidencing the existence and amounts of the required insurance. No such policy shall be cancelable or subject to modification except after 30 days prior written notice to Lessor. Lessee shall, at least 10 days prior to the expiration of such policies, furnish Lessor with evidence of renewals or "insurance binders" evidencing renewal thereof, or Lessor may increase his liability insurance coverage and charge the cost thereof to Lessee, which amount shall be payable by Lessee to Lessor upon demand. Such policies shall be for a term of at least one year, or the length of the remaining term of this Lease, whichever is less. If either Party shall fail to procure and maintain the insurance required to be carried by it, the other Party may, but shall not be required to, procure and maintain the same.
- 8.6 Waiver of Subrogation. Without affecting any other rights or remedies, Lessee and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is not limited by the amount of insurance carried or required, or by any deductibles applicable hereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Lessee, as the case may be, so long as the insurance is not invalidated thereby.
- 8.7 Indemnity. Except for Lessor's gross negligence or willful misconduct, Lessee shall indemnify, protect, defend and hold harmless the Premises, Lessor and its agents, Lessor's master or ground lessor, partners and Lenders, from and against any and all claims, loss of rents and/or damages, liens, judgments, penalties, attorneys' and consultants' fees, expenses and/or liabilities arising out of, involving, or in connection with, a Breach of the Lease by Lessee and/or the use and/or occupancy of the Premises and/or Project by Lessee and/or by Lessee's employees, contractors or invitees. If any action or proceeding is brought against Lessor by reason of any of the foregoing matters, Lessee shall upon notice defend the same at Lessee's expense by counsel reasonably satisfactory to Lessor and Lessor shall cooperate with Lessee in such defense. Lessor need not have first paid any such claim in order to be defended or indemnified.

 8.8 Exemption of Lessor and its Agents from Liability. Notwithstanding the negligence or breach of this Lease by Lessor or its agents, neither Lessor nor its agents shall be liable under any circumstances for: (i) injury or damage to the person or goods, wares, merchandise or other property of Lessee, Lessee's employees,
- 8.8 **Exemption of Lessor and its Agents from Liability.** Notwithstanding the negligence or breach of this Lease by Lessor or its agents, neither Lessor nor its agents shall be liable under any circumstances for: (i) injury or damage to the person or goods, wares, merchandise or other property of Lessee, Lessee's employees, contractors, invitees, customers, or any other person in or about the Premises, whether such damage or injury is caused by or results from fire, steam, electricity, gas, water or rain, indoor air quality, the presence of mold or from the breakage, leakage, obstruction or other defects of pipes, fire sprinklers, wires, appliances, plumbing, HVAC or lighting fixtures, or from any other cause, whether the said injury or damage results from conditions arising upon the Premises or upon other portions of the Building, or from other sources or places; (ii) any damages arising from any act or neglect of any other tenant of Lessor or from the failure of Lessor or its agents to enforce the provisions of any other lease in the Project; or (iii) injury to Lessee's business or for any loss of income or profit therefrom. Instead, it is intended that Lessee's sole recourse in the event of such damages or injury be to file a claim on the insurance policy(ies) that Lessee is required to maintain pursuant to the provisions of paragraph 8.
- 8.9 **Failure to Provide Insurance**. Lessee acknowledges that any failure on its part to obtain or maintain the insurance required herein will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, for any month or portion thereof that Lessee does not maintain the required insurance and/or does not provide Lessor with the required binders or certificates evidencing the existence of the required insurance, the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater. The parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to maintain the required insurance. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to maintain such insurance, prevent the exercise of any of the other rights and remedies granted hereunder, nor relieve Lessee of its obligation to maintain the insurance specified in this Lease.

9. Damage or Destruction.

9.1 Definitions.

- (a) "Premises Partial Damage" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations, which can reasonably be repaired in 3 months or less from the date of the damage or destruction, and the cost thereof does not exceed a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (b) "Premises Total Destruction" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which cannot reasonably be repaired in 3 months or less from the date of the damage or destruction and/or the cost thereof exceeds a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (c) "Insured Loss" shall mean damage or destruction to improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which was caused by an event required to be covered by the insurance described in Paragraph 8.3(a), irrespective of any deductible amounts or coverage limits involved.
- coverage limits involved.

 (d) "Replacement Cost" shall mean the cost to repair or rebuild the improvements owned by Lessor at the time of the occurrence to their condition existing immediately prior thereto, including demolition, debris removal and upgrading required by the operation of Applicable Requirements, and without deduction for depreciation.
- (e) "Hazardous Substance Condition" shall mean the occurrence or discovery of a condition involving the presence of, or a contamination by, a Hazardous Substance, in, on, or under the Premises which requires restoration.
- 9.2 Partial Damage Insured Loss. If a Premises Partial Damage that is an Insured Loss occurs, then Lessor shall, at Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonably possible and this Lease shall continue in full force and effect; provided, however, that Lessee shall, at Lessor's election, make the repair of any damage or destruction the total cost to repair of which is \$10,000 or less, and, in such

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

event, Lessor shall make any applicable insurance proceeds available to Lessee on a reasonable basis for that purpose. Notwithstanding the foregoing, if the required insurance was not in force or the insurance proceeds are not sufficient to effect such repair, the Insuring Party shall promptly contribute the shortage in proceeds as and when required to complete said repairs. In the event, however, such shortage was due to the fact that, by reason of the unique nature of the improvements, full replacement cost insurance coverage was not commercially reasonable and available, Lessor shall have no obligation to pay for the shortage in insurance proceeds or to fully restore the unique aspects of the Premises unless Lessee provides Lessor with the funds to cover same, or adequate assurance thereof, within 10 days following receipt of written notice of such shortage and request therefor. If Lessor receives said funds or adequate assurance thereof within said 10 day period, the party responsible for making the repairs shall complete them as soon as reasonably possible and this Lease shall remain in full force and effect. If such funds or assurance are not received, Lessor may nevertheless elect by written notice to Lessee within 10 days thereafter to: (i) make such restoration and repair as is commercially reasonable with Lessor paying any shortage in proceeds, in which case this Lease shall remain in full force and effect, or (ii) have this Lease terminate 30 days thereafter. Lessee shall not be entitled to reimbursement of any funds contributed by Lessee to repair any such damage or destruction. Premises Partial Damage due to flood or earthquake shall be subject to Paragraph 9.3, notwithstanding that there may be some insurance coverage, but the net proceeds of any such insurance shall be made available for the repairs if made by either Party.

- 9.3 **Partial Damage Uninsured Loss**. If a Premises Partial Damage that is not an Insured Loss occurs, unless caused by a negligent or willful act of Lessee (in which event Lessee shall make the repairs at Lessee's expense), Lessor may either: (i) repair such damage as soon as reasonably possible at Lessor's expense (subject to reimbursement pursuant to Paragraph 4.2), in which event this Lease shall continue in full force and effect, or (ii) terminate this Lease by giving written notice to Lessee within 30 days after receipt by Lessor of knowledge of the occurrence of such damage. Such termination shall be effective 60 days following the date of such notice. In the event Lessor elects to terminate this Lease, Lessee shall have the right within 10 days after receipt of the termination notice to give written notice to Lessor of Lessee's commitment to pay for the repair of such damage without reimbursement from Lessor. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days after making such commitment. In such event this Lease shall continue in full force and effect, and Lessor shall proceed to make such repairs as soon as reasonably possible after the required funds are available. If Lessee does not make the required commitment, this Lease shall terminate as of the date specified in the termination notice.
- 9.4 **Total Destruction**. Notwithstanding any other provision hereof, if a Premises Total Destruction occurs, this Lease shall terminate 60 days following such Destruction. If the damage or destruction was caused by the gross negligence or willful misconduct of Lessee, Lessor shall have the right to recover Lessor's damages from Lessee, except as provided in Paragraph 8.6.
- 9.5 Damage Near End of Term. If at any time during the last 6 months of this Lease there is damage for which the cost to repair exceeds one month's Base Rent, whether or not an Insured Loss, Lessor may terminate this Lease effective 60 days following the date of occurrence of such damage by giving a written termination notice to Lessee within 30 days after the date of occurrence of such damage. Notwithstanding the foregoing, if Lessee at that time has an exercisable option to extend this Lease or to purchase the Premises, then Lessee may preserve this Lease by (a) exercising such option and (b) providing Lessor with any shortage in insurance proceeds (or adequate assurance thereof) needed to make the repairs on or before the earlier of (i) the date which is 10 days after Lessee's receipt of Lessor's written notice purporting to terminate this Lease, or (ii) the day prior to the date upon which such option expires. If Lessee duly exercises such option during such period and provides Lessor with funds (or adequate assurance thereof) to cover any shortage in insurance proceeds, Lessor shall, at Lessor's commercially reasonable expense, repair such damage as soon as reasonably possible and this Lease shall continue in full force and effect. If Lessee fails to exercise such option and provide such funds or assurance during such period, then this Lease shall terminate on the date specified in the termination notice and Lessee's option shall be extinguished.

9.6 Abatement of Rent; Lessee's Remedies.

- (a) Abatement. In the event of Premises Partial Damage or Premises Total Destruction or a Hazardous Substance Condition for which Lessee is not responsible under this Lease, the Rent payable by Lessee for the period required for the repair, remediation or restoration of such damage shall be abated in proportion to the degree to which Lessee's use of the Premises is impaired, but not to exceed the proceeds received from the Rental Value insurance. All other obligations of Lessee hereunder shall be performed by Lessee, and Lessor shall have no liability for any such damage, destruction, remediation, repair or restoration except as provided herein.
- (b) Remedies. If Lessor is obligated to repair or restore the Premises and does not commence, in a substantial and meaningful way, such repair or restoration within 90 days after such obligation shall accrue, Lessee may, at any time prior to the commencement of such repair or restoration, give written notice to Lessor and to any Lenders of which Lessee has actual notice, of Lessee's election to terminate this Lease on a date not less than 60 days following the giving of such notice. If Lessee gives such notice and such repair or restoration is not commenced within 30 days thereafter, this Lease shall terminate as of the date specified in said notice. If the repair or restoration is commenced within such 30 days, this Lease shall continue in full force and effect. "Commence" shall mean either the unconditional authorization of the preparation of the required plans, or the beginning of the actual work on the Premises, whichever first occurs.
- 9.7 **Termination; Advance Payments**. Upon termination of this Lease pursuant to Paragraph 6.2(g) or Paragraph 9, an equitable adjustment shall be made concerning advance Base Rent and any other advance payments made by Lessee to Lessor. Lessor shall, in addition, return to Lessee so much of Lessee's Security Deposit as has not been, or is not then required to be, used by Lessor.

10. Real Property Taxes.

10.1 Definitions.

- (a) "Real Property Taxes." As used herein, the term "Real Property Taxes" shall include any form of assessment; real estate, general, special, ordinary or extraordinary, or rental levy or tax (other than inheritance, personal income or estate taxes); improvement bond; and/or license fee imposed upon or levied against any legal or equitable interest of Lessor in the Project, Lessor's right to other income therefrom, and/or Lessor's business of leasing, by any authority having the direct or indirect power to tax and where the funds are generated with reference to the Project address. The term "Real Property Taxes" shall also include any tax, fee, levy, assessment or charge, or any increase therein: (i) imposed by reason of events occurring during the term of this Lease, including but not limited to, a change in the ownership of the Project; (ii) a change in the improvements thereon; and/or (iii) levied or assessed on machinery or equipment provided by Lessor to Lessee pursuant to this Lease.
- (b) "Base Real Property Taxes." As used herein, the term "Base Real Property Taxes" shall be the amount of Real Property Taxes, which are assessed against the Project, during the entire calendar year in which the Lease is executed.
- 10.2 **Payment of Taxes**. Except as otherwise provided in Paragraph 10.3, Lessor shall pay the Real Property Taxes applicable to the Project, and said payments shall be included in the calculation of Common Area Operating Expenses in accordance with the provisions of Paragraph 4.2.
- 10.3 Additional Improvements. Common Area Operating Expenses shall not include Real Property Taxes specified in the tax assessor's records and work sheets as being caused by additional improvements placed upon the Project by other tenants or by Lessor for the exclusive enjoyment of such other Tenants. Notwithstanding Paragraph 10.2 hereof, Lessee shall, however, pay to Lessor at the time Common Area Operating Expenses are payable under Paragraph 4.2, the entirety of any increase in Real Property Taxes if assessed solely by reason of Alterations, Trade Fixtures or Utility Installations placed upon the Premises by Lessee or at Lessee's request or by reason of any alterations or improvements to the Premises made by Lessor subsequent to the execution of this Lease by the Parties.
- 10.4 **Joint Assessment**. If the Building is not separately assessed, Real Property Taxes allocated to the Building shall be an equitable proportion of the Real Property Taxes for all of the land and improvements included within the tax parcel assessed, such proportion to be determined by Lessor from the respective valuations assigned in the assessor's work sheets or such other information as may be reasonably available. Lessor's reasonable determination thereof, in good faith, shall be conclusive.
- 10.5 **Personal Property Taxes**. Lessee shall pay prior to delinquency all taxes assessed against and levied upon Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all personal property of Lessee contained in the Premises. When possible, Lessee shall cause its Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all other personal property to be assessed and billed separately from the real property of Lessor. If any of Lessee's said property shall be assessed with Lessor's real property, Lessee shall pay Lessor the taxes attributable to Lessee's property within 10 days after receipt of a written statement setting forth the taxes applicable to Lessee's property.

11. Utilities and Services.

- 11.1 Lessee shall pay for all water, gas, heat, light, power, telephone, trash disposal and other utilities and services supplied to the Premises, together with any taxes thereon. Notwithstanding the provisions of Paragraph 4.2, if at any time in Lessor's sole judgment, Lessor determines that Lessee is using a disproportionate amount of water, electricity or other commonly metered utilities, or that Lessee is generating such a large volume of trash as to require an increase in the size of the trash receptacle and/or an increase in the number of times per month that it is emptied, then Lessor may increase Lessee's Base Rent by an amount equal to such increased costs. There shall be no abatement of Rent and Lessor shall not be liable in any respect whatsoever for the inadequacy, stoppage, interruption or discontinuance of any utility or service due to riot, strike, labor dispute, breakdown, accident, repair or other cause beyond Lessor's reasonable control or in cooperation with governmental request or directions.
- 11.2 Within fifteen days of Lessor's written request, Lessee agrees to deliver to Lessor such information, documents and/or authorization as Lessor needs in order for Lessor to comply with new or existing Applicable Requirements relating to commercial building energy usage, ratings, and/or the reporting thereof.

12. Assignment and Subletting.

12.1 Lessor's Consent Required.

INITIALS		INITIALS
© 2019 AIR CRE.	All Rights Reserved.	

- (a) Lessee shall not voluntarily or by operation of law assign, transfer, mortgage or encumber (collectively, "assign or assignment") or sublet all or any part of Lessee's interest in this Lease or in the Premises without Lessor's prior written consent.
- (b) Unless Lessee is a corporation and its stock is publicly traded on a national stock exchange, a change in the control of Lessee shall constitute an assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of the voting control of Lessee shall constitute a change in control for this purpose.
- (c) The involvement of Lessee or its assets in any transaction, or series of transactions (by way of merger, sale, acquisition, financing, transfer, leveraged buy-out or otherwise), whether or not a formal assignment or hypothecation of this Lease or Lessee's assets occurs, which results or will result in a reduction of the Net Worth of Lessee by an amount greater than 25% of such Net Worth as it was represented at the time of the execution of this Lease or at the time of the most recent assignment to which Lessor has consented, or as it exists immediately prior to said transaction or transactions constituting such reduction, whichever was or is greater, shall be considered an assignment of this Lease to which Lessor may withhold its consent. "Net Worth of Lessee" shall mean the net worth of Lessee (excluding any guarantors) established under generally accepted accounting principles.
- (d) An assignment or subletting without consent shall, at Lessor's option, be a Default curable after notice per Paragraph 13.1(d), or a non-curable Breach without the necessity of any notice and grace period. If Lessor elects to treat such unapproved assignment or subletting as a non-curable Breach, Lessor may either: (i) terminate this Lease, or (ii) upon 30 days written notice, increase the monthly Base Rent to 110% of the Base Rent then in effect. Further, in the event of such Breach and rental adjustment, (i) the purchase price of any option to purchase the Premises held by Lessee shall be subject to similar adjustment to 110% of the price previously in effect, and (ii) all fixed and non-fixed rental adjustments scheduled during the remainder of the Lease term shall be increased to 110% of the scheduled adjusted rent.
 - (e) Lessee's remedy for any breach of Paragraph 12.1 by Lessor shall be limited to compensatory damages and/or injunctive relief.
 - (f) Lessor may reasonably withhold consent to a proposed assignment or subletting if Lessee is in Default at the time consent is requested.
- (g) Notwithstanding the foregoing, allowing a de minimis portion of the Premises, ie. 20 square feet or less, to be used by a third party vendor in connection with the installation of a vending machine or pay phone shall not constitute a subletting.

12.2 Terms and Conditions Applicable to Assignment and Subletting.

- (a) Regardless of Lessor's consent, no assignment or subletting shall: (i) be effective without the express written assumption by such assignee or sublessee of the obligations of Lessee under this Lease, (ii) release Lessee of any obligations hereunder, or (iii) alter the primary liability of Lessee for the payment of Rent or for the performance of any other obligations to be performed by Lessee.
- (b) Lessor may accept Rent or performance of Lessee's obligations from any person other than Lessee pending approval or disapproval of an assignment. Neither a delay in the approval or disapproval of such assignment nor the acceptance of Rent or performance shall constitute a waiver or estoppel of Lessor's right to exercise its remedies for Lessee's Default or Breach.
 - (c) Lessor's consent to any assignment or subletting shall not constitute a consent to any subsequent assignment or subletting.
- (d) In the event of any Default or Breach by Lessee, Lessor may proceed directly against Lessee, any Guarantors or anyone else responsible for the performance of Lessee's obligations under this Lease, including any assignee or sublessee, without first exhausting Lessor's remedies against any other person or entity responsible therefor to Lessor, or any security held by Lessor.
- (e) Each request for consent to an assignment or subletting shall be in writing, accompanied by information relevant to Lessor's determination as to the financial and operational responsibility and appropriateness of the proposed assignee or sublessee, including but not limited to the intended use and/or required modification of the Premises, if any, together with a fee of \$500 as consideration for Lessor's considering and processing said request. Lessee agrees to provide Lessor with such other or additional information and/or documentation as may be reasonably requested. (See also Paragraph 36)
- (f) Any assignee of, or sublessee under, this Lease shall, by reason of accepting such assignment, entering into such sublease, or entering into possession of the Premises or any portion thereof, be deemed to have assumed and agreed to conform and comply with each and every term, covenant, condition and obligation herein to be observed or performed by Lessee during the term of said assignment or sublease, other than such obligations as are contrary to or inconsistent with provisions of an assignment or sublease to which Lessor has specifically consented to in writing.
- (g) Lessor's consent to any assignment or subletting shall not transfer to the assignee or sublessee any Option granted to the original Lessee by this Lease unless such transfer is specifically consented to by Lessor in writing. (See Paragraph 39.2)
- 12.3 Additional Terms and Conditions Applicable to Subletting. The following terms and conditions shall apply to any subletting by Lessee of all or any part of the Premises and shall be deemed included in all subleases under this Lease whether or not expressly incorporated therein:

 (a) Lessee hereby assigns and transfers to Lessor all of Lessee's interest in all Rent payable on any sublease, and Lessor may collect such Rent and apply
- same toward Lessee's obligations under this Lease; provided, however, that until a Breach shall occur in the performance of Lessee's obligations, Lessee may collect said Rent. In the event that the amount collected by Lessor exceeds Lessee's then outstanding obligations any such excess shall be refunded to Lessee. Lessor shall not, by reason of the foregoing or any assignment of such sublease, nor by reason of the collection of Rent, be deemed liable to the sublessee for any failure of Lessee to perform and comply with any of Lessee's obligations to such sublessee. Lessee hereby irrevocably authorizes and directs any such sublessee, upon receipt of a written notice from Lessor stating that a Breach exists in the performance of Lessee's obligations under this Lease, to pay to Lessor all Rent due and to become due under the sublease. Sublessee shall rely upon any such notice from Lessor and shall pay all Rents to Lessor without any obligation or right to inquire as to whether such Breach exists, notwithstanding any claim from Lessee to the contrary.
- (b) In the event of a Breach by Lessee, Lessor may, at its option, require sublessee to attorn to Lessor, in which event Lessor shall undertake the obligations of the sublessor under such sublease from the time of the exercise of said option to the expiration of such sublease; provided, however, Lessor shall not be liable for any prepaid rents or security deposit paid by such sublessee to such sublessor or for any prior Defaults or Breaches of such sublessor.
 - (c) Any matter requiring the consent of the sublessor under a sublease shall also require the consent of Lessor.
 - (d) No sublessee shall further assign or sublet all or any part of the Premises without Lessor's prior written consent.
- (e) Lessor shall deliver a copy of any notice of Default or Breach by Lessee to the sublessee, who shall have the right to cure the Default of Lessee within the grace period, if any, specified in such notice. The sublessee shall have a right of reimbursement and offset from and against Lessee for any such Defaults cured by the sublessee.

13. Default; Breach; Remedies.

- 13.1 **Default; Breach**. A "**Default**" is defined as a failure by the Lessee to comply with or perform any of the terms, covenants, conditions or Rules and Regulations under this Lease. A "**Breach**" is defined as the occurrence of one or more of the following Defaults, and the failure of Lessee to cure such Default within any applicable grace period:
- (a) The abandonment of the Premises; the vacating of the Premises prior to the expiration or termination of this Lease without providing a commercially reasonable level of security, or where the coverage of the property insurance described in Paragraph 8.3 is jeopardized as a result thereof, or without providing reasonable assurances to minimize potential vandalism; or failure to deliver to Lessor exclusive possession of the entire Premises in accordance herewith prior to the expiration or termination of this Lease.
- (b) The failure of Lessee to make any payment of Rent or any Security Deposit required to be made by Lessee hereunder, whether to Lessor or to a third party, when due, to provide reasonable evidence of insurance or surety bond, or to fulfill any obligation under this Lease which endangers or threatens life or property, where such failure continues for a period of 3 business days following written notice to Lessee. THE ACCEPTANCE BY LESSOR OF A PARTIAL PAYMENT OF RENT OR SECURITY DEPOSIT SHALL NOT CONSTITUTE A WAIVER OF ANY OF LESSOR'S RIGHTS, INCLUDING LESSOR'S RIGHT TO RECOVER POSSESSION OF THE PREMISES.
- (c) The failure of Lessee to allow Lessor and/or its agents access to the Premises or the commission of waste, act or acts constituting public or private nuisance, and/or an illegal activity on the Premises by Lessee, where such actions continue for a period of 3 business days following written notice to Lessee. In the event that Lessee commits waste, a nuisance or an illegal activity a second time then, the Lessor may elect to treat such conduct as a non-curable Breach rather than a Default.
- (d) The failure by Lessee to provide (i) reasonable written evidence of compliance with Applicable Requirements, (ii) the service contracts, (iii) the rescission of an unauthorized assignment or subletting, (iv) an Estoppel Certificate or financial statements, (v) a requested subordination, (vi) evidence concerning any guaranty and/or Guarantor, (vii) any document requested under Paragraph 41, (viii) material safety data sheets (MSDS), or (ix) any other documentation or information which Lessor may reasonably require of Lessee under the terms of this Lease, where any such failure continues for a period of 10 days following written notice to Lessee
- (e) A Default by Lessee as to the terms, covenants, conditions or provisions of this Lease, or of the rules adopted under Paragraph 2.9 hereof, other than those described in subparagraphs 13.1(a), (b), (c) or (d), above, where such Default continues for a period of 30 days after written notice; provided, however, that if the nature of Lessee's Default is such that more than 30 days are reasonably required for its cure, then it shall not be deemed to be a Breach if Lessee commences such cure within said 30 day period and thereafter diligently prosecutes such cure to completion.
- (f) The occurrence of any of the following events: (i) the making of any general arrangement or assignment for the benefit of creditors; (ii) becoming a "debtor" as defined in 11 U.S.C. § 101 or any successor statute thereto (unless, in the case of a petition filed against Lessee, the same is dismissed within 60 days); (iii) the appointment of a trustee or receiver to take possession of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where possession is not restored to Lessee within 30 days; or (iv) the attachment, execution or other judicial seizure of substantially all of Lessee's assets located at the

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

Premises or of Lessee's interest in this Lease, where such seizure is not discharged within 30 days; provided, however, in the event that any provision of this subparagraph is contrary to any applicable law, such provision shall be of no force or effect, and not affect the validity of the remaining provisions.

- (g) The discovery that any financial statement of Lessee or of any Guarantor given to Lessor was materially false.
- (h) If the performance of Lessee's obligations under this Lease is guaranteed: (i) the death of a Guarantor, (ii) the termination of a Guarantor's liability with respect to this Lease other than in accordance with the terms of such guaranty, (iii) a Guarantor's becoming insolvent or the subject of a bankruptcy filing, (iv) a Guarantor's refusal to honor the guaranty, or (v) a Guarantor's breach of its guaranty obligation on an anticipatory basis, and Lessee's failure, within 60 days following written notice of any such event, to provide written alternative assurance or security, which, when coupled with the then existing resources of Lessee, equals or exceeds the combined financial resources of Lessee and the Guarantors that existed at the time of execution of this Lease.
- 13.2 Remedies. If Lessee fails to perform any of its affirmative duties or obligations, within 10 days after written notice (or in case of an emergency, without notice), Lessor may, at its option, perform such duty or obligation on Lessee's behalf, including but not limited to the obtaining of reasonably required bonds, insurance policies, or governmental licenses, permits or approvals. Lessee shall pay to Lessor an amount equal to 115% of the costs and expenses incurred by Lessor in such performance upon receipt of an invoice therefor. In the event of a Breach, Lessor may, with or without further notice or demand, and without limiting Lessor in the exercise of any right or remedy which Lessor may have by reason of such Breach:
- Terminate Lessee's right to possession of the Premises by any lawful means, in which case this Lease shall terminate and Lessee shall immediately surrender possession to Lessor. In such event Lessor shall be entitled to recover from Lessee: (i) the unpaid Rent which had been earned at the time of termination; (ii) the worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that the Lessee proves could have been reasonably avoided; (iii) the worth at the time of award of the amount by which the unpaid rent for the balance of the term after the time of award exceeds the amount of such rental loss that the Lessee proves could be reasonably avoided; and (iv) any other amount necessary to compensate Lessor for all the detriment proximately caused by the Lessee's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom, including but not limited to the cost of recovering possession of the Premises, expenses of reletting, including necessary renovation and alteration of the Premises, reasonable attorneys' fees, and that portion of any leasing commission paid by Lessor in connection with this Lease applicable to the unexpired term of this Lease. The worth at the time of award of the amount referred to in provision (iii) of the immediately preceding sentence shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of the District within which the Premises are located at the time of award plus one percent. Efforts by Lessor to mitigate damages caused by Lessee's Breach of this Lease shall not waive Lessor's right to recover any damages to which Lessor is otherwise entitled. If termination of this Lease is obtained through the provisional remedy of unlawful detainer, Lessor shall have the right to recover in such proceeding any unpaid Rent and damages as are recoverable therein, or Lessor may reserve the right to recover all or any part thereof in a separate suit. If a notice and grace period required under Paragraph 13.1 was not previously given, a notice to pay rent or quit, or to perform or quit given to Lessee under the unlawful detainer statute shall also constitute the notice required by Paragraph 13.1. In such case, the applicable grace period required by Paragraph 13.1 and the unlawful detainer statute shall run concurrently, and the failure of Lessee to cure the Default within the greater of the two such grace periods shall constitute both an unlawful detainer and a Breach of this Lease entitling Lessor to the remedies provided for in this Lease and/or by said statute.
- (b) Continue the Lease and Lessee's right to possession and recover the Rent as it becomes due, in which event Lessee may sublet or assign, subject only to reasonable limitations. Acts of maintenance, efforts to relet, and/or the appointment of a receiver to protect the Lessor's interests, shall not constitute a termination of the Lessee's right to possession.
- (c) Pursue any other remedy now or hereafter available under the laws or judicial decisions of the state wherein the Premises are located. The expiration or termination of this Lease and/or the termination of Lessee's right to possession shall not relieve Lessee from liability under any indemnity provisions of this Lease as to matters occurring or accruing during the term hereof or by reason of Lessee's occupancy of the Premises.
- 13.3 Inducement Recapture. Any agreement for free or abated rent or other charges, the cost of tenant improvements for Lessee paid for or performed by Lessor, or for the giving or paying by Lessor to or for Lessee of any cash or other bonus, inducement or consideration for Lessee's entering into this Lease, all of which concessions are hereinafter referred to as "Inducement Provisions," shall be deemed conditioned upon Lessee's full and faithful performance of all of the terms, covenants and conditions of this Lease. Upon Breach of this Lease by Lessee, any such Inducement Provision shall automatically be deemed deleted from this Lease and of no further force or effect, and any rent, other charge, bonus, inducement or consideration theretofore abated, given or paid by Lessor under such an Inducement Provision shall be immediately due and payable by Lessee to Lessor, notwithstanding any subsequent cure of said Breach by Lessee. The acceptance by Lessor of rent or the cure of the Breach which initiated the operation of this paragraph shall not be deemed a waiver by Lessor of the provisions of this paragraph unless specifically so stated in writing by Lessor at the time of such acceptance.
- 13.4 Late Charges. Lessee hereby acknowledges that late payment by Lessee of Rent will cause Lessor to incur costs not contemplated by this Lease, the exact amount of which will be extremely difficult to ascertain. Such costs include, but are not limited to, processing and accounting charges, and late charges which may be imposed upon Lessor by any Lender. Accordingly, if any Rent shall not be received by Lessor within 5 days after such amount shall be due, then, without any requirement for notice to Lessee, Lessee shall immediately pay to Lessor a one-time late charge equal to 10% of each such overdue amount or \$100, whichever is greater. The parties hereby agree that such late charge represents a fair and reasonable estimate of the costs Lessor will incur by reason of such late payment. Acceptance of such late charge by Lessor shall in no event constitute a waiver of Lessee's Default or Breach with respect to such overdue amount, nor prevent the exercise of any of the other rights and remedies granted hereunder. In the event that a late charge is payable hereunder, whether or not collected, for 3 consecutive installments of Base Rent, then notwithstanding any provision of this Lease to the contrary, Base Rent shall, at Lessor's option, become due and payable quarterly in advance.
- 13.5 Interest. Any monetary payment due Lessor hereunder, other than late charges, not received by Lessor, when due shall bear interest from the 31st day after it was due. The interest ("Interest") charged shall be computed at the rate of 10% per annum but shall not exceed the maximum rate allowed by law. Interest is payable in addition to the potential late charge provided for in Paragraph 13.4.

13.6 Breach by Lessor.

- (a) Notice of Breach. Lessor shall not be deemed in breach of this Lease unless Lessor fails within a reasonable time to perform an obligation required to be performed by Lessor. For purposes of this Paragraph, a reasonable time shall in no event be less than 30 days after receipt by Lessor, and any Lender whose name and address shall have been furnished to Lessee in writing for such purpose, of written notice specifying wherein such obligation of Lessor has not been performed; provided, however, that if the nature of Lessor's obligation is such that more than 30 days are reasonably required for its performance, then Lessor shall not be in breach if performance is commenced within such 30 day period and thereafter diligently pursued to completion.
- (b) **Performance by Lessee on Behalf of Lessor**. In the event that neither Lessor nor Lender cures said breach within 30 days after receipt of said notice, or if having commenced said cure they do not diligently pursue it to completion, then Lessee may elect to cure said breach at Lessee's expense and offset from Rent the actual and reasonable cost to perform such cure, provided however, that such offset shall not exceed an amount equal to the greater of one month's Base Rent or the Security Deposit, reserving Lessee's right to reimbursement from Lessor for any such expense in excess of such offset. Lessee shall document the cost of said cure and supply said documentation to Lessor.
- 14. Condemnation. If the Premises or any portion thereof are taken under the power of eminent domain or sold under the threat of the exercise of said power (collectively "Condemnation"), this Lease shall terminate as to the part taken as of the date the condemning authority takes title or possession, whichever first occurs. If more than 10% of the floor area of the Unit, or more than 25% of the parking spaces is taken by Condemnation, Lessee may, at Lessee's option, to be exercised in writing within 10 days after Lessor shall have given Lessee written notice of such taking (or in the absence of such notice, within 10 days after the condemning authority shall have taken possession) terminate this Lease as of the date the condemning authority takes such possession. If Lessee does not terminate this Lease in accordance with the foregoing, this Lease shall remain in full force and effect as to the portion of the Premises remaining, except that the Base Rent shall be reduced in proportion to the reduction in utility of the Premises caused by such Condemnation. Condemnation awards and/or payments shall be the property of Lessor, whether such award shall be made as compensation for diminution in value of the leasehold, the value of the part taken, or for severance damages; provided, however, that Lessee shall be entitled to any compensation paid by the condemnor for Lessee's relocation expenses, loss of business goodwill and/or Trade Fixtures, without regard to whether or not this Lease is terminated pursuant to the provisions of this Paragraph. All Alterations and Utility Installations made to the Premises by Lessee, for purposes of Condemnation only, shall be considered the property of the Lessee and Lessee shall be entitled to any and all compensation which is payable therefor. In the event that this Lease is not terminated by reason of the Condemnation, Lessor shall repair any damage to the Premises caused by such Condemnation.

15. Brokerage Fees.

- 15.1 Additional Commission. In addition to the payments owed pursuant to Paragraph 1.10 above, Lessor agrees that: (a) if Lessee exercises any Option, (b) if Lessee or anyone affiliated with Lessee acquires from Lessor any rights to the Premises or other premises owned by Lessor and located within the Project, (c) if Lessee remains in possession of the Premises, with the consent of Lessor, after the expiration of this Lease, or (d) if Base Rent is increased, whether by agreement or operation of an escalation clause herein, then, Lessor shall pay Brokers a fee in accordance with the fee schedule of the Brokers in effect at the time the Lease was executed. The provisions of this paragraph are intended to supersede the provisions of any earlier agreement to the contrary.
- 15.2 **Assumption of Obligations**. Any buyer or transferee of Lessor's interest in this Lease shall be deemed to have assumed Lessor's obligation hereunder. Brokers shall be third party beneficiaries of the provisions of Paragraphs 1.10, 15, 22 and 31. If Lessor fails to pay to Brokers any amounts due as and for brokerage

INITIAL

fees pertaining to this Lease when due, then such amounts shall accrue Interest. In addition, if Lessor fails to pay any amounts to Lessee's Broker when due, Lessee's Broker may send written notice to Lessor and Lessor failure and if Lessor fails to pay such amounts within 10 days after said notice, Lessee shall pay said monies to its Broker and offset such amounts against Rent. In addition, Lessee's Broker shall be deemed to be a third party beneficiary of any commission agreement entered into by and/or between Lessor and Lessor's Broker for the limited purpose of collecting any brokerage fee owed.

15.3 Representations and Indemnities of Broker Relationships. Lessee and Lessor each represent and warrant to the other that it has had no dealings with any person, firm, broker, agent or finder (other than the Brokers and Agents, if any) in connection with this Lease, and that no one other than said named Brokers and Agents is entitled to any commission or finder's fee in connection herewith. Lessee and Lessor do each hereby agree to indemnify, protect, defend and hold the other harmless from and against liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any dealings or actions of the indemnifying Party, including any costs, expenses, attorneys' fees reasonably incurred with respect thereto.

16. Estoppel Certificates.

- (a) Each Party (as "Responding Party") shall within 10 days after written notice from the other Party (the "Requesting Party") execute, acknowledge and deliver to the Requesting Party a statement in writing in form similar to the then most current "Estoppel Certificate" form published by AIR CRE, plus such additional information, confirmation and/or statements as may be reasonably requested by the Requesting Party.
- (b) If the Responding Party shall fail to execute or deliver the Estoppel Certificate within such 10 day period, the Requesting Party may execute an Estoppel Certificate stating that: (i) the Lease is in full force and effect without modification except as may be represented by the Requesting Party, (ii) there are no uncured defaults in the Requesting Party's performance, and (iii) if Lessor is the Requesting Party, not more than one month's rent has been paid in advance. Prospective purchasers and encumbrancers may rely upon the Requesting Party's Estoppel Certificate, and the Responding Party shall be estopped from denying the truth of the facts contained in said Certificate. In addition, Lessee acknowledges that any failure on its part to provide such an Estoppel Certificate will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to execute and/or deliver a requested Estoppel Certificate in a timely fashion the monthly Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for remainder of the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to provide the Estoppel Certificate. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to provide the Estoppel Certificate nor prevent the exercise of any of the other rights and remedies granted hereunder.
- (c) If Lessor desires to finance, refinance, or sell the Premises, or any part thereof, Lessee and all Guarantors shall within 10 days after written notice from Lessor deliver to any potential lender or purchaser designated by Lessor such financial statements as may be reasonably required by such lender or purchaser, including but not limited to Lessee's financial statements for the past 3 years. All such financial statements shall be received by Lessor and such lender or purchaser in confidence and shall be used only for the purposes herein set forth.
- 17. Definition of Lessor. The term "Lessor" as used herein shall mean the owner or owners at the time in question of the fee title to the Premises, or, if this is a sublease, of the Lessee's interest in the prior lease. In the event of a transfer of Lessor's title or interest in the Premises or this Lease, Lessor shall deliver to the transferee or assignee (in cash or by credit) any unused Security Deposit held by Lessor. Upon such transfer or assignment and delivery of the Security Deposit, as aforesaid, the prior Lessor shall be relieved of all liability with respect to the obligations and/or covenants under this Lease thereafter to be performed by the Lessor. Subject to the foregoing, the obligations and/or covenants in this Lease to be performed by the Lessor shall be binding only upon the Lessor as hereinabove defined.
- 18. Severability. The invalidity of any provision of this Lease, as determined by a court of competent jurisdiction, shall in no way affect the validity of any other provision hereof.
- 19. Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Lease shall mean and refer to calendar days.
- 20. Limitation on Liability. The obligations of Lessor under this Lease shall not constitute personal obligations of Lessor, or its partners, members, directors, officers or shareholders, and Lessee shall look to the Premises, and to no other assets of Lessor, for the satisfaction of any liability of Lessor with respect to this Lease, and shall not seek recourse against Lessor's partners, members, directors, officers or shareholders, or any of their personal assets for such satisfaction.
- 21. Time of Essence. Time is of the essence with respect to the performance of all obligations to be performed or observed by the Parties under this Lease.
- 22. No Prior or Other Agreements; Broker Disclaimer. This Lease contains all agreements between the Parties with respect to any matter mentioned herein, and no other prior or contemporaneous agreement or understanding shall be effective. Lessor and Lessee each represents and warrants to the Brokers that it has made, and is relying solely upon, its own investigation as to the nature, quality, character and financial responsibility of the other Party to this Lease and as to the use, nature, quality and character of the Premises. Brokers have no responsibility with respect thereto or with respect to any default or breach hereof by either Party.

23. Notices.

- 23.1 Notice Requirements. All notices required or permitted by this Lease or applicable law shall be in writing and may be delivered in person (by hand or by courier) or may be sent by regular, certified or registered mail or U.S. Postal Service Express Mail, with postage prepaid, or by facsimile transmission, or by email, and shall be deemed sufficiently given if served in a manner specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specify a different address for notice, except that upon Lessee's taking possession of the Premises, the Premises shall constitute Lessee's address for notice. A copy of all notices to Lessor shall be concurrently transmitted to such party or parties at such addresses as Lessor may from time to time hereafter designate in writing.
- 23.2 Date of Notice. Any notice sent by registered or certified mail, return receipt requested, shall be deemed given on the date of delivery shown on the receipt card, or if no delivery date is shown, the postmark thereon. If sent by regular mail the notice shall be deemed given 72 hours after the same is addressed as required herein and mailed with postage prepaid. Notices delivered by United States Express Mail or overnight courier that guarantees next day delivery shall be deemed given 24 hours after delivery of the same to the Postal Service or courier. Notices delivered by hand, or transmitted by facsimile transmission or by email shall be deemed delivered upon actual receipt. If notice is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.
- 23.3 **Options.** Notwithstanding the foregoing, in order to exercise any Options (see paragraph 39), the Notice must be sent by Certified Mail (return receipt requested), Express Mail (signature required), courier (signature required) or some other methodology that provides a receipt establishing the date the notice was received by the Lessor.

24. Waivers.

- (a) No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any other term, covenant or condition hereof. Lessor's consent to, or approval of, any act shall not be deemed to render unnecessary the obtaining of Lessor's consent to, or approval of, any subsequent or similar act by Lessee, or be construed as the basis of an estoppel to enforce the provision or provisions of this Lease requiring such consent.
- (b) The acceptance of Rent by Lessor shall not be a waiver of any Default or Breach by Lessee. Any payment by Lessee may be accepted by Lessor on account of monies or damages due Lessor, notwithstanding any qualifying statements or conditions made by Lessee in connection therewith, which such statements and/or conditions shall be of no force or effect whatsoever unless specifically agreed to in writing by Lessor at or before the time of deposit of such payment.
- (c) THE PARTIES AGREE THAT THE TERMS OF THIS LEASE SHALL GOVERN WITH REGARD TO ALL MATTERS RELATED THERETO AND HEREBY WAIVE THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE TO THE EXTENT THAT SUCH STATUTE IS INCONSISTENT WITH THIS LEASE.

25. Disclosures Regarding The Nature of a Real Estate Agency Relationship.

- (a) When entering into a discussion with a real estate agent regarding a real estate transaction, a Lessor or Lessee should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Lessor and Lessee acknowledge being advised by the Brokers in this transaction, as follows:
- (i) <u>Lessor's Agent</u>. A Lessor's agent under a listing agreement with the Lessor acts as the agent for the Lessor only. A Lessor's agent or subagent has the following affirmative obligations: <u>To the Lessor</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessor. <u>To the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (ii) <u>Lessee's Agent</u>. An agent can agree to act as agent for the Lessee only. In these situations, the agent is not the Lessor's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Lessor. An agent acting only for a Lessee has the following affirmative obligations. <u>To the Lessee</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessee. <u>To the Lessee and the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts

INITIALS

Last Edited: 9/30/2022 2:33 PM

INITIALS

known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth

- (iii) Agent Representing Both Lessor and Lessee. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Lessor and the Lessee in a transaction, but only with the knowledge and consent of both the Lessor and the Lessee. In a dual agency situation, the agent has the following affirmative obligations to both the Lessor and the Lessee: (a) A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Lessor or the Lessee. (b) Other duties to the Lessor and the Lessee as stated above in subparagraphs (i) or (ii). In representing both Lessor and Lessee, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating to either Lessee's or Lessor's financial position, motivations, bargaining position, or other personal information that may impact rent, including Lessor's willingness to accept a rent less than the listing rent or Lessee's willingness to pay rent greater than the rent offered. The above duties of the agent in a real estate transaction do not relieve a Lessor or Lessee from the responsibility to protect their own interests. Lessor and Lessee should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Both Lessor and Lessee should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.
- (b) Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this Lease may be brought against Broker more than one year after the Start Date and that the liability (including court costs and attorneys' fees), of any Broker with respect to any such lawsuit and/or legal proceeding shall not exceed the fee received by such Broker pursuant to this Lease; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.
- (c) Lessor and Lessee agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- 26. No Right To Holdover. Lessee has no right to retain possession of the Premises or any part thereof beyond the expiration or termination of this Lease. At or prior to the expiration or termination of this Lease Lessee shall deliver exclusive possession of the Premises to Lessor. For purposes of this provision and Paragraph 13.1(a), exclusive possession shall mean that Lessee shall have vacated the Premises, removed all of its personal property therefrom and that the Premises have been returned in the condition specified in this Lease. In the event that Lessee does not deliver exclusive possession to Lessor as specified above, then Lessor's damages during any holdover period shall be computed at the amount of the Rent (as defined in Paragraph 4.1) due during the last full month before the expiration or termination of this Lease (disregarding any temporary abatement of Rent that may have been in effect), but with Base Rent being 150% of the Base Rent payable during such last full month. Nothing contained herein shall be construed as consent by Lessor to any holding over by Lessee.
- 27. Cumulative Remedies. No remedy or election hereunder shall be deemed exclusive but shall, wherever possible, be cumulative with all other remedies at law or in equity.
- 28. Covenants and Conditions; Construction of Agreement. All provisions of this Lease to be observed or performed by Lessee are both covenants and conditions. In construing this Lease, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Lease. Whenever required by the context, the singular shall include the plural and vice versa. This Lease shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.
- 29. Binding Effect; Choice of Law. This Lease shall be binding upon the Parties, their personal representatives, successors and assigns and be governed by the laws of the State in which the Premises are located. Any litigation between the Parties hereto concerning this Lease shall be initiated in the county in which the Premises are located. Signatures to this Lease accomplished by means of electronic signature or similar technology shall be legal and binding.

30. Subordination; Attornment; Non-Disturbance.

- 30.1 **Subordination**. This Lease and any Option granted hereby shall be subject and subordinate to any ground lease, mortgage, deed of trust, or other hypothecation or security device (collectively, "**Security Device**"), now or hereafter placed upon the Premises, to any and all advances made on the security thereof, and to all renewals, modifications, and extensions thereof. Lessee agrees that the holders of any such Security Devices (in this Lease together referred to as "**Lender**") shall have no liability or obligation to perform any of the obligations of Lessor under this Lease. Any Lender may elect to have this Lease and/or any Option granted hereby superior to the lien of its Security Device by giving written notice thereof to Lessee, whereupon this Lease and such Options shall be deemed prior to such Security Device, notwithstanding the relative dates of the documentation or recordation thereof.

 30.2 **Attornment**. In the event that Lessor transfers title to the Premises, or the Premises are acquired by another upon the foreclosure or termination of a
- 30.2 Attornment. In the event that Lessor transfers title to the Premises, or the Premises are acquired by another upon the foreclosure or termination of a Security Device to which this Lease is subordinated (i) Lessee shall, subject to the non-disturbance provisions of Paragraph 30.3, attorn to such new owner, and upon request, enter into a new lease, containing all of the terms and provisions of this Lease, with such new owner for the remainder of the term hereof, or, at the election of the new owner, this Lease will automatically become a new lease between Lessee and such new owner, and (ii) Lessor shall thereafter be relieved of any further obligations hereunder and such new owner shall assume all of Lessor's obligations, except that such new owner shall not: (a) be liable for any act or omission of any prior lessor or with respect to events occurring prior to acquisition of ownership; (b) be subject to any offsets or defenses which Lessee might have against any prior lessor, (c) be bound by prepayment of more than one month's rent, or (d) be liable for the return of any security deposit paid to any prior lessor which was not paid or credited to such new owner.
- 30.3 **Non-Disturbance**. With respect to Security Devices entered into by Lessor after the execution of this Lease, Lessee's subordination of this Lease shall be subject to receiving a commercially reasonable non-disturbance agreement (a "**Non-Disturbance Agreement**") from the Lender which Non-Disturbance Agreement provides that Lessee's possession of the Premises, and this Lease, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises. Further, within 60 days after the execution of this Lease, Lessor shall, if requested by Lessee, use its commercially reasonable efforts to obtain a Non-Disturbance Agreement from the holder of any pre-existing Security Device which is secured by the Premises. In the event that Lessor is unable to provide the Non-Disturbance Agreement within said 60 days, then Lessee may, at Lessee's option, directly contact Lender and attempt to negotiate for the execution and delivery of a Non-Disturbance Agreement.
- 30.4 **Self-Executing**. The agreements contained in this Paragraph 30 shall be effective without the execution of any further documents; provided, however, that, upon written request from Lessor or a Lender in connection with a sale, financing or refinancing of the Premises, Lessee and Lessor shall execute such further writings as may be reasonably required to separately document any subordination, attornment and/or Non-Disturbance Agreement provided for herein.
- 31. Attorneys' Fees. If any Party or Broker brings an action or proceeding involving the Premises whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred. In addition, Lessor shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is subsequently commenced in connection with such Default or resulting Breach (\$200 is a reasonable minimum per occurrence for such services and consultation).
- 32. Lessor's Access; Showing Premises; Repairs. Lessor and Lessor's agents shall have the right to enter the Premises at any time, in the case of an emergency, and otherwise at reasonable times after reasonable prior notice for the purpose of showing the same to prospective purchasers, lenders, or tenants, and making such alterations, repairs, improvements or additions to the Premises as Lessor may deem necessary or desirable and the erecting, using and maintaining of utilities, services, pipes and conduits through the Premises and/or other premises as long as there is no material adverse effect on Lessee's use of the Premises. All such activities shall be without abatement of rent or liability to Lessee.
- **33. Auctions.** Lessee shall not conduct, nor permit to be conducted, any auction upon the Premises without Lessor's prior written consent. Lessor shall not be obligated to exercise any standard of reasonableness in determining whether to permit an auction.
- **34. Signs.** Lessor may place on the Premises ordinary "For Sale" signs at any time and ordinary "For Lease" signs during the last 6 months of the term hereof. Except for ordinary "For Sublease" signs which may be placed only on the Premises, Lessee shall not place any sign upon the Project without Lessor's prior written consent. All signs must comply with all Applicable Requirements.
- **35. Termination; Merger.** Unless specifically stated otherwise in writing by Lessor, the voluntary or other surrender of this Lease by Lessee, the mutual termination or cancellation hereof, or a termination hereof by Lessor for Breach by Lessee, shall automatically terminate any sublease or lesser estate in the Premises; provided, however, that Lessor may elect to continue any one or all existing subtenancies. Lessor's failure within 10 days following any such event to elect to the contrary by

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

written notice to the holder of any such lesser interest, shall constitute Lessor's election to have such event constitute the termination of such interest.

36. Consents. All requests for consent shall be in writing. Except as otherwise provided herein, wherever in this Lease the consent of a Party is required to an act by or for the other Party, such consent shall not be unreasonably withheld or delayed. Lessor's actual reasonable costs and expenses (including but not limited to architects', attorneys', engineers' and other consultants' fees) incurred in the consideration of, or response to, a request by Lessee for any Lessor consent, including but not limited to consents to an assignment, a subletting or the presence or use of a Hazardous Substance, shall be paid by Lessee upon receipt of an invoice and supporting documentation therefor. Lessor's consent to any act, assignment or subletting shall not constitute an acknowledgment that no Default or Breach by Lessee of this Lease exists, nor shall such consent be deemed a waiver of any then existing Default or Breach, except as may be otherwise specifically stated in writing by Lessor at the time of such consent. The failure to specify herein any particular condition to Lessor's consent shall not preclude the imposition by Lessor at the time of consent of such further or other conditions as are then reasonable with reference to the particular matter for which consent is being given. In the event that either Party disagrees with any determination made by the other hereunder and reasonably requests the reasons for such determination, the determining party shall furnish its reasons in writing and in reasonable detail within 10 business days following such request.

37. Guarantor.

- 37.1 Execution. The Guarantors, if any, shall each execute a guaranty in the form most recently published by AIR CRE.
- 37.2 **Default**. It shall constitute a Default of the Lessee if any Guarantor fails or refuses, upon request to provide: (a) evidence of the execution of the guaranty, including the authority of the party signing on Guarantor's behalf to obligate Guarantor, and in the case of a corporate Guarantor, a certified copy of a resolution of its board of directors authorizing the making of such guaranty, (b) current financial statements, (c) an Estoppel Certificate, or (d) written confirmation that the guaranty is still in effect.
- **38. Quiet Possession.** Subject to payment by Lessee of the Rent and performance of all of the covenants, conditions and provisions on Lessee's part to be observed and performed under this Lease, Lessee shall have quiet possession and quiet enjoyment of the Premises during the term hereof.
- 39. Options. If Lessee is granted any option, as defined below, then the following provisions shall apply.
- 39.1 **Definition**. "**Option**" shall mean: (a) the right to extend or reduce the term of or renew this Lease or to extend or reduce the term of or renew any lease that Lessee has on other property of Lessor; (b) the right of first refusal or first offer to lease either the Premises or other property of Lessor; (c) the right to purchase, the right of first offer to purchase or the right of first refusal to purchase the Premises or other property of Lessor.
- 39.2 **Options Personal To Original Lessee**. Any Option granted to Lessee in this Lease is personal to the original Lessee, and cannot be assigned or exercised by anyone other than said original Lessee and only while the original Lessee is in full possession of the Premises and, if requested by Lessor, with Lessee certifying that Lessee has no intention of thereafter assigning or subletting.
- 39.3 Multiple Options. In the event that Lessee has any multiple Options to extend or renew this Lease, a later Option cannot be exercised unless the prior Options have been validly exercised.

39.4 Effect of Default on Options.

- (a) Lessee shall have no right to exercise an Option: (i) during the period commencing with the giving of any notice of Default and continuing until said Default is cured, (ii) during the period of time any Rent is unpaid (without regard to whether notice thereof is given Lessee), (iii) during the time Lessee is in Breach of this Lease, or (iv) in the event that Lessee has been given 3 or more notices of separate Default, whether or not the Defaults are cured, during the 12 month period immediately preceding the exercise of the Option.
- (b) The period of time within which an Option may be exercised shall not be extended or enlarged by reason of Lessee's inability to exercise an Option because of the provisions of Paragraph 39.4(a).
- (c) An Option shall terminate and be of no further force or effect, notwithstanding Lessee's due and timely exercise of the Option, if, after such exercise and prior to the commencement of the extended term or completion of the purchase, (i) Lessee fails to pay Rent for a period of 30 days after such Rent becomes due (without any necessity of Lessor to give notice thereof), or (ii) if Lessee commits a Breach of this Lease.
- **40. Security Measures.** Lessee hereby acknowledges that the Rent payable to Lessor hereunder does not include the cost of guard service or other security measures, and that Lessor shall have no obligation whatsoever to provide same. Lessee assumes all responsibility for the protection of the Premises, Lessee, its agents and invitees and their property from the acts of third parties.
- **41. Reservations.** Lessor reserves the right: (i) to grant, without the consent or joinder of Lessee, such easements, rights and dedications that Lessor deems necessary; (ii) to cause the recordation of parcel maps and restrictions; and (iii) to create and/or install new utility raceways, so long as such easements, rights, dedications, maps, restrictions, and utility raceways do not unreasonably interfere with the use of the Premises by Lessee. Lessee agrees to sign any documents reasonably requested by Lessor to effectuate such rights.
- **42. Performance Under Protest.** If at any time a dispute shall arise as to any amount or sum of money to be paid by one Party to the other under the provisions hereof, the Party against whom the obligation to pay the money is asserted shall have the right to make payment "under protest" and such payment shall not be regarded as a voluntary payment and there shall survive the right on the part of said Party to institute suit for recovery of such sum. If it shall be adjudged that there was no legal obligation on the part of said Party to pay such sum or any part thereof, said Party shall be entitled to recover such sum or so much thereof as it was not legally required to pay. A Party who does not initiate suit for the recovery of sums paid "under protest" within 6 months shall be deemed to have waived its right to protest such payment.

43. Authority; Multiple Parties; Execution.

- (a) If either Party hereto is a corporation, trust, limited liability company, partnership, or similar entity, each individual executing this Lease on behalf of such entity represents and warrants that he or she is duly authorized to execute and deliver this Lease on its behalf. Each Party shall, within 30 days after request, deliver to the other Party satisfactory evidence of such authority.
- (b) If this Lease is executed by more than one person or entity as "Lessee", each such person or entity shall be jointly and severally liable hereunder. It is agreed that any one of the named Lessees shall be empowered to execute any amendment to this Lease, or other document ancillary thereto and bind all of the named Lessees, and Lessor may rely on the same as if all of the named Lessees had executed such document.
- (c) This Lease may be executed by the Parties in counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument.
- **44. Conflict.** Any conflict between the printed provisions of this Lease and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions.
- **45. Offer.** Preparation of this Lease by either party or their agent and submission of same to the other Party shall not be deemed an offer to lease to the other Party. This Lease is not intended to be binding until executed and delivered by all Parties hereto.
- **46. Amendments.** This Lease may be modified only in writing, signed by the Parties in interest at the time of the modification. As long as they do not materially change Lessee's obligations hereunder, Lessee agrees to make such reasonable non-monetary modifications to this Lease as may be reasonably required by a Lender in connection with the obtaining of normal financing or refinancing of the Premises.
- 47. Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY OR ARISING OUT OF THIS AGREEMENT.

	Arbitration of Disputes. An Addendum requiring the Arbitration of all disputes between the Parties and/or Brokers arising out of this Lease is is not
aπacr	ned to this Lease.
49.	Accessibility; Americans with Disabilities Act.
	(a) The Premises:
deter requii the su arran	nave not undergone an inspection by a Certified Access Specialist (CASp). Note: A Certified Access Specialist (CASp) can inspect the subject premises and mine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection of ubject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the gements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to ct violations of construction-related accessibility standards within the premises.
L r	nave undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises met all applicable construction-related accessibility
	<u>—</u>

INITIALS

INITIALS © 2019 AIR CRE. All Rights Reserved.

MTG-24.40, Revised 10-22-2020

Lease and agrees to keep such report confidential.	lactic received a copy of the inspection report acteast 48 hours prior to executing this
• • • • • • • • • • • • • • • • • • • •	as determined that the Premises did not meet all applicable construction-related nowledges that it received a copy of the inspection report at least 48 hours prior to ary to complete repairs and corrections of violations of construction related
In the event that the Premises have been issued an inspection report by a CASp th within 7 days of the execution of this Lease.	e Lessor shall provide a copy of the disability access inspection certificate to Lessee
use of the Premises, Lessor makes no warranty or representation as to whether or	and other state and local accessibility statutes are dependent upon Lessee's specific not the Premises comply with ADA or any similar legislation. In the event that n order to be in compliance with ADA or other accessibility statutes, Lessee agrees to
LESSOR AND LESSEE HAVE CAREFULLY READ AND REVIEWED THIS LEASE AND EAL LEASE SHOW THEIR INFORMED AND VOLUNTARY CONSENT THERETO. THE PART THIS LEASE ARE COMMERCIALLY REASONABLE AND EFFECTUATE THE INTENT AN	
	HE PARTIES ARE URGED TO: ITHIS LEASE. INDITION OF THE PREMISES. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE INDITION OF THE PREMISES, THE STRUCTURAL INTEGRITY, THE CONDITION OF THE ROOF
WARNING: IF THE PREMISES ARE LOCATED IN A STATE OTHER THAN CALIFORNIA THE LAWS OF THE STATE IN WHICH THE PREMISES ARE LOCATED.	A, CERTAIN PROVISIONS OF THE LEASE MAY NEED TO BE REVISED TO COMPLY WITH
The parties hereto have executed this Lease at the place and on the dates specifie	d above their respective signatures.
Executed at:	Executed at:
On:	On:
By LESSOR:	By LESSEE:
By:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
By:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
<i>(</i>) \	
Address: Federal ID No.:	Address: Federal ID No.:
redefall b No.:	redefailb No
BROKER	BROKER
Attn:	Attn:
Title:	Title:
Address	Address
Address:Phone:	Address: Phone:
Fax:	Fax:
Email:	Email:
Federal ID No.:	Federal ID No.:
Broker DRE License #:	Broker DRE License #:
Agent DRE License #:	Agent DRE License #:
	213-687-8777 * contracts@aircre.com duced in any form without permission in writing.
NOTICE. NO part of these works may be repro	duced in any form without permission in writing.
*	
INITIALS	INITIALS



STANDARD MULTI-TENANT OFFICE LEASE - NET

1.1 Dantina Thia Lagar (III agan) data different formation and a second a second and a second	
	ade by and between ("Lessor") and
("Lessee"), (collectively the "Parties", or individually a "Party"). 1.2(a) Premises: That certain portion of the Project (as defined below), commonly kn	nown as (street address suite sity state): ("Dremises")
The Premises are located in the County of, and consist of approximately	
square feet. In addition to Lessee's rights to use and occupy the Premises as hereinafter specifi	
defined in Paragraph 2.7 below) as hereinafter specified, but shall not have any rights to the root	
raceways of the building containing the Premises ("Building") or to any other buildings in the Premises (buildings and improvements thereon, are herein contained in the premise of the buildings and improvements thereon, are herein contained in the premise of the buildings and improvements thereon, are herein contained in the premise of the buildings and improvements thereon, are herein to be also be als	
approximately rentable square feet. (See also Paragraph 2)	bilectively referred to as the Project. The Project consists of
1.2(b) Parking: unreserved and reserved vehicle parking	g spaces at a monthly cost of per unreserved space and
per reserved space. (See Paragraph 2.6)	
1.3 Term: years and months ("Original Term") commencing	ng ("Commencement Date") and ending
("Expiration Date"). (See also Paragraph 3)	ossion of the Dromices commencing ("Fault Decretion
1.4 Early Possession: If the Premises are available Lessee may have non-exclusive posses Date"). (See also Paragraphs 3.2 and 3.3)	ession of the Premises commencing (Early Possession
1.5 Base Rent: per month ("Base Rent"), payable on the d	ay of each month commencing . (See also Paragraph 4)
If this box is checked, there are provisions in this Lease for the Base Rent to be adjust	
1.6 Lessee's Share of Operating Expenses percent (%) ("Le	
Project are modified during the term of this Lease, Lessor shall recalculate Lessee's Share to ref	
1.7 Base Rent and Other Monies Paid Upon Execution:	\sim \vee
(a) Base Rent: for the period	<) \
(b) Operating Expenses: The current estimate for the period is	
(d) Parking: for the period	
(e) Other: for .	() '
(f) Total Due Upon Execution of this Lease:	
1.8 Agreed Use: (See also Paragraph 6)	
1.9 Insuring Party. Lessor is the "Insuring Party". (See also Paragraph 8)	
1.10 Real Estate Brokers. (See also Paragraph 15 and 25)(a) Representation: Each Party acknowledges receiving a Disclosure Regarding Re	al Estate Agency Relationship, confirms and consents to the following
agency relationships in this Lease with the following real estate brokers ("Broker(s)") and/or the	
Lessor's Brokerage Firm License No Is the broker of (check	one): the Lessor; or both the Lessee and Lessor (dual agent).
Lessor's Agent License No. is (check one): the Lessor's	▼
Agent and the Lessor's Agent (dual agent).	Agent (salesperson of broker associate), or — both the lessee's
Lessee's Brokerage Firm License No Is the broker of (check	one). the Lessee: or hoth the Lessee and Lessor (dual
agent).	one, and ecosee, or an according to the ecose and ecosor (dual)
Lessee's Agent License No. is (check one): the Lessee's	s Agent (salesperson or broker associate); or hoth the Lessee's
Agent and the Lessor's Agent (dual agent).	
(b) Payment to Brokers. Upon execution and delivery of this Lease by both Partie	
separate written agreement (or if there is no such agreement, the sum of or	% of the total Base Rent) for the brokerage services rendered
by the Brokers. 1.11 Guarantor. The obligations of the Lessee under this Lease are to be guaranteed by	("Guarantar") (Soo also Paragraph 27)
1.12 Business Hours for the Building: a.m. to p.m., Monday	
1.12 Business Hours for the Building: a.m. to p.m., Monday p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the	
1.12 Business Hours for the Building: a.m. to p.m., Monday p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and	
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the	dates of observation of New Year's Day, President's Day, Memorial Day,
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and	dates of observation of New Year's Day, President's Day, Memorial Day,
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso	dates of observation of New Year's Day, President's Day, Memorial Day,
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity	dates of observation of New Year's Day, President's Day, Memorial Day,
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises:
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify): 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises:
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify): 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through ;	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises:
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify): 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through ; a plot plan depicting the Premises;	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises:
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify): 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through; a plot plan depicting the Premises; a current set of the Rules and Regulations;	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises:
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify): 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through ; a plot plan depicting the Premises;	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises:
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify): 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through; a plot plan depicting the Premises; a current set of the Rules and Regulations;	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises:
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify). 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through; a plot plan depicting the Premises; a current set of the Rules and Regulations; a Work Letter;	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises:
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify): 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through; a plot plan depicting the Premises; a current set of the Rules and Regulations; a Work Letter; a janitorial schedule;	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises:
	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises: is Lease: Premises, for the term, at the rental, and upon all of the terms,
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify). 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through; a plot plan depicting the Premises; a current set of the Rules and Regulations; a Work Letter; a janitorial schedule; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lessor, the Province of the Rules and conditions set forth in this Lease. While the approximate square footage of the	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises: is Lease: Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for
	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises: is Lease: Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for
	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises: is Lease: Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for ubject to adjustment should the actual size be determined to be commencement Date or the Early Possession Date, whichever first occurs
	Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for ubject to adjustment should the actual size be determined to be commencement Date or the Early Possession Date, whichever first occurs us, ventilating and air conditioning systems ("HVAC"), and all other items
	Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for ubject to adjustment should the actual size be determined to be commencement Date or the Early Possession Date, whichever first occurs ug, ventilating and air conditioning systems ("HVAC"), and all other items other than those constructed by Lessee, shall be in good operating
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Janitorial services Janitorial services	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises: by Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for subject to adjustment should the actual size be determined to be commencement Date or the Early Possession Date, whichever first occurs ag, ventilating and air conditioning systems ("HVAC"), and all other items of the Unit shall be free of material defects, and that the Premises do not law. Lessor also warrants, that unless otherwise specified in writing,
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify) 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through; a plot plan depicting the Premises; a current set of the Rules and Regulations; a Work Letter; a janitorial schedule; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lessor, the Proposes of comparison, the Base Rent stated herein is NOT tied to square footage and is not stifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. 2.2 Condition. Lessor shall deliver the Premises to Lessee in a clean condition on the Condition on said date, that the existing electrical, plumbing, fire sprinkler, lighting, heating which the Lessor is obligated to construct pursuant to the Work Letter attached hereto, if any, condition on said date, that the structural elements of the roof, bearing walls and foundation or contain hazardous levels of any mold or fungi defined as toxic under applicable state or federal Lessor is unaware of (i) any recorded Notices of Default affecting the Premise; (ii) any delinquent	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises: by Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for subject to adjustment should the actual size be determined to be commencement Date or the Early Possession Date, whichever first occurs ag, ventilating and air conditioning systems ("HVAC"), and all other items of the Unit shall be free of material defects, and that the Premises do not law. Lessor also warrants, that unless otherwise specified in writing,
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and	dates of observation of New Year's Day, President's Day, Memorial Day, r is NOT obligated to provide the following within the Premises: Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for ubject to adjustment should the actual size be determined to be ommencement Date or the Early Possession Date, whichever first occurs ag, ventilating and air conditioning systems ("HVAC"), and all other items of the Unit shall be free of material defects, and that the Premises do not law. Lessor also warrants, that unless otherwise specified in writing, it amounts due under any loan secured by the Premises; and (iii) any
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify) 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through; a plot plan depicting the Premises; a current set of the Rules and Regulations; a Work Letter; a janitorial schedule; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lessor, the Proposes of comparison, the Base Rent stated herein is NOT tied to square footage and is not stifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. 2.2 Condition. Lessor shall deliver the Premises to Lessee in a clean condition on the Condition on said date, that the existing electrical, plumbing, fire sprinkler, lighting, heating which the Lessor is obligated to construct pursuant to the Work Letter attached hereto, if any, condition on said date, that the structural elements of the roof, bearing walls and foundation or contain hazardous levels of any mold or fungi defined as toxic under applicable state or federal Lessor is unaware of (i) any recorded Notices of Default affecting the Premise; (ii) any delinquent	r is NOT obligated to provide the following within the Premises: Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for ubject to adjustment should the actual size be determined to be Commencement Date or the Early Possession Date, whichever first occurs ag, ventilating and air conditioning systems ("HVAC"), and all other items of the Unit shall be free of material defects, and that the Premises do not law. Lessor also warrants, that unless otherwise specified in writing, it amounts due under any loan secured by the Premises; and (iii) any in the Premises and the Common Areas comply with the building codes,
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and	r is NOT obligated to provide the following within the Premises: Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for ubject to adjustment should the actual size be determined to be Commencement Date or the Early Possession Date, whichever first occurs ag, ventilating and air conditioning systems ("HVAC"), and all other items of the Unit shall be free of material defects, and that the Premises do not law. Lessor also warrants, that unless otherwise specified in writing, it amounts due under any loan secured by the Premises; and (iii) any in the Premises and the Common Areas comply with the building codes,
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and 1.13 Lessor Supplied Services. Notwithstanding the provisions of Paragraph 11.1, Lesso Janitorial services Electricity Other (specify). 1.14 Attachments. Attached hereto are the following, all of which constitute a part of the an Addendum consisting of Paragraphs through; a plot plan depicting the Premises; a current set of the Rules and Regulations; a Work Letter; a janitorial schedule; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lessor, the Province of the purposes of comparison, the Base Rent stated herein is NOT tied to square footage and is not still different. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. 2.2 Condition. Lessor shall deliver the Premises to Lessee in a clean condition on the Condition	r is NOT obligated to provide the following within the Premises: Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for ubject to adjustment should the actual size be determined to be Commencement Date or the Early Possession Date, whichever first occurs ag, ventilating and air conditioning systems ("HVAC"), and all other items of the Unit shall be free of material defects, and that the Premises do not law. Lessor also warrants, that unless otherwise specified in writing, it amounts due under any loan secured by the Premises; and (iii) any in the Premises and the Common Areas comply with the building codes,
p.m. on Saturdays (except Building Holidays). "Building Holidays" shall mean the Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and	r is NOT obligated to provide the following within the Premises: Premises, for the term, at the rental, and upon all of the terms, Premises may have been used in the marketing of the Premises for ubject to adjustment should the actual size be determined to be Commencement Date or the Early Possession Date, whichever first occurs ag, ventilating and air conditioning systems ("HVAC"), and all other items of the Unit shall be free of material defects, and that the Premises do not law. Lessor also warrants, that unless otherwise specified in writing, it amounts due under any loan secured by the Premises; and (iii) any in the Premises and the Common Areas comply with the building codes,

MTON-20.20, Revised 10-22-2020

or portion thereof, was constructed. Said warranty does not apply to the use to which Lessee will put the Premises, modifications which may be required by the Americans with Disabilities Act or any similar laws as a result of Lessee's use (see Paragraph 49), or to any Alterations or Utility Installations (as defined in Paragraph 7.3(a)) made or to be made by Lessee. NOTE: Lessee is responsible for determining whether or not the zoning and other Applicable Requirements are appropriate for Lessee's intended use, and acknowledges that past uses of the Premises may no longer be allowed. If the Premises do not comply with said warranty, Lessor shall, except as otherwise provided, promptly after receipt of written notice from Lessee setting forth with specificity the nature and extent of such non-compliance, rectify the same. If the Applicable Requirements are hereafter changed so as to require during the term of this Lease the construction of an addition to or an alteration of the Premises, the remediation of any Hazardous Substance, or the reinforcement or other physical modification of the Premises ("Capital Expenditure"), Lessor and Lessee shall allocate the cost of such work as follows:

- (a) Subject to Paragraph 2.3(c) below, if such Capital Expenditures are required as a result of the specific and unique use of the Premises by Lessee as compared with uses by tenants in general, Lessee shall be fully responsible for the cost thereof, provided, however, that if such Capital Expenditure is required during the last 2 years of this Lease and the cost thereof exceeds 6 months' Base Rent, Lessee may instead terminate this Lease unless Lessor notifies Lessee, in writing, within 10 days after receipt of Lessee's termination notice that Lessor has elected to pay the difference between the actual cost thereof and the amount equal to 6 months' Base Rent. If Lessee elects termination, Lessee shall immediately cease the use of the Premises which requires such Capital Expenditure and deliver to Lessor written notice specifying a termination date at least 90 days thereafter. Such termination date shall, however, in no event be earlier than the last day that Lessee could legally utilize the Premises without commencing such Capital Expenditure.
- (b) If such Capital Expenditure is not the result of the specific and unique use of the Premises by Lessee (such as, governmentally mandated seismic modifications), then Lessor shall pay for such Capital Expenditure and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date that on which the Base Rent is due, an amount equal to 1/144th of the portion of such costs reasonably attributable to the Premises. Lessee shall pay Interest on the balance but may prepay its obligation at any time. If, however, such Capital Expenditure is required during the last 2 years of this Lease or if Lessor reasonably determines that it is not economically feasible to pay its share thereof, Lessor shall have the option to terminate this Lease upon 90 days prior written notice to Lessee unless Lessee notifies Lessor, in writing, within 10 days after receipt of Lessor's termination notice that Lessee will pay for such Capital Expenditure. If Lessor does not elect to terminate, and fails to tender its share of any such Capital Expenditure, Lessee may advance such funds and deduct same, with Interest, from Rent until Lessor's share of such costs have been fully paid. If Lessee is unable to finance Lessor's share, or if the balance of the Rent due and payable for the remainder of this Lease is not sufficient to fully reimburse Lessee on an offset basis, Lessee shall have the right to terminate this Lease upon 30 days written notice to Lessor.
- (c) Notwithstanding the above, the provisions concerning Capital Expenditures are intended to apply only to non-voluntary, unexpected, and new Applicable Requirements. If the Capital Expenditures are instead triggered by Lessee as a result of an actual or proposed change in use, change in intensity of use, or modification to the Premises then, and in that event, Lessee shall either: (i) immediately cease such changed use or intensity of use and/or take such other steps as may be necessary to eliminate the requirement for such Capital Expenditure, or (ii) complete such Capital Expenditure at its own expense. Lessee shall not have any right to terminate this Lease.
- 2.4 **Acknowledgements.** Lessee acknowledges that: (a) it has been given an opportunity to inspect and measure the Premises, (b) Lessee has been advised by Lessor and/or Brokers to satisfy itself with respect to the size and condition of the Premises (including but not limited to the electrical, HVAC and fire sprinkler systems, security, environmental aspects, and compliance with Applicable Requirements), and their suitability for Lessee's intended use, (c) Lessee has made such investigation as it deems necessary with reference to such matters and assumes all responsibility therefor as the same relate to its occupancy of the Premises, (d) it is not relying on any representation as to the size of the Premises made by Brokers or Lessor, (e) the square footage of the Premises was not material to Lessee's decision to lease the Premises and pay the Rent stated herein, and (f) neither Lessor, Lessor's agents, nor Brokers have made any oral or written representations or warranties with respect to said matters other than as set forth in this Lease. In addition, Lessor acknowledges that: (i) Brokers have made no representations, promises or warranties concerning Lessee's ability to honor the Lease or suitability to occupy the Premises, and (ii) it is Lessor's sole responsibility to investigate the financial capability and/or suitability of all proposed tenants.
- 2.5 **Lessee as Prior Owner/Occupant**. The warranties made by Lessor in Paragraph 2 shall be of no force or effect if immediately prior to the Start Date, Lessee was the owner or occupant of the Premises. In such event, Lessee shall be responsible for any necessary corrective work.
- 2.6 **Vehicle Parking**. So long as Lessee is not in default, and subject to the Rules and Regulations attached hereto, and as established by Lessor from time to time, Lessee shall be entitled to rent and use the number of parking spaces specified in Paragraph 1.2(b) at the rental rate applicable from time to time for monthly parking as set by Lessor and/or its licensee.
- (a) If Lessee commits, permits or allows any of the prohibited activities described in the Lease or the rules then in effect, then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove or tow away the vehicle involved and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.
- (b) The monthly rent per parking space specified in Paragraph 1.2(b) is subject to change upon 30 days prior written notice to Lessee. The rent for the parking is payable one month in advance prior to the first day of each calendar month.
- 2.7 Common Areas Definition. The term "Common Areas" is defined as all areas and facilities outside the Premises and within the exterior boundary line of the Project and interior utility raceways and installations within the Premises that are provided and designated by the Lessor from time to time for the general nonexclusive use of Lessor, Lessee and other tenants of the Project and their respective employees, suppliers, shippers, customers, contractors and invitees, including, but not limited to, common entrances, lobbies, corridors, stairwells, public restrooms, elevators, parking areas, loading and unloading areas, trash areas, roadways, walkways, driveways and landscaped areas.
- 2.8 **Common Areas Lessee's Rights**. Lessor grants to Lessee, for the benefit of Lessee and its employees, suppliers, shippers, contractors, customers and invitees, during the term of this Lease, the non-exclusive right to use, in common with others entitled to such use, the Common Areas as they exist from time to time, subject to any rights, powers, and privileges reserved by Lessor under the terms hereof or under the terms of any rules and regulations or restrictions governing the use of the Project. Under no circumstances shall the right herein granted to use the Common Areas be deemed to include the right to store any property, temporarily or permanently, in the Common Areas. Any such storage shall be permitted only by the prior written consent of Lessor or Lessor's designated agent, which consent may be revoked at any time. In the event that any unauthorized storage shall occur, then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove the property and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.
- 2.9 **Common Areas Rules and Regulations**. Lessor or such other person(s) as Lessor may appoint shall have the exclusive control and management of the Common Areas and shall have the right, from time to time, to adopt, modify, amend and enforce reasonable rules and regulations ("**Rules and Regulations**") for the management, safety, care, and cleanliness of the grounds, the parking and unloading of vehicles and the preservation of good order, as well as for the convenience of other occupants or tenants of the Building and the Project and their invitees. The Lessee agrees to abide by and conform to all such Rules and Regulations, and shall use its best efforts to cause its employees, suppliers, shippers, customers, contractors and invitees to so abide and conform. Lessor shall not be responsible to Lessee for the noncompliance with said Rules and Regulations by other tenants of the Project.
 - 2.10 **Common Areas Changes**. Lessor shall have the right, in Lessor's sole discretion, from time to time:
- (a) To make changes to the Common Areas, including, without limitation, changes in the location, size, shape and number of the lobbies, windows, stairways, air shafts, elevators, escalators, restrooms, driveways, entrances, parking spaces, parking areas, loading and unloading areas, ingress, egress, direction of traffic, landscaped areas, walkways and utility raceways;
 - (b) To close temporarily any of the Common Areas for maintenance purposes so long as reasonable access to the Premises remains available;
 - c) To designate other land outside the boundaries of the Project to be a part of the Common Areas;
 - (d) To add additional buildings and improvements to the Common Areas;
 - (e) To use the Common Areas while engaged in making additional improvements, repairs or alterations to the Project, or any portion thereof; and
- (f) To do and perform such other acts and make such other changes in, to or with respect to the Common Areas and Project as Lessor may, in the exercise of sound business judgment, deem to be appropriate.

3. Term.

INITIALS

- 3.1 **Term**. The Commencement Date, Expiration Date and Original Term of this Lease are as specified in Paragraph 1.3.
- 3.2 **Early Possession**. Any provision herein granting Lessee Early Possession of the Premises is subject to and conditioned upon the Premises being available for such possession prior to the Commencement Date. Any grant of Early Possession only conveys a non-exclusive right to occupy the Premises. If Lessee totally or partially occupies the Premises prior to the Commencement Date, the obligation to pay Base Rent shall be abated for the period of such Early Possession. All other terms of this Lease (including but not limited to the obligations to pay Lessee's Share of the Operating Expenses) shall be in effect during such period. Any such Early Possession shall not affect the Expiration Date.
- 3.3 **Delay In Possession**. Lessor agrees to use commercially reasonable efforts to deliver exclusive possession of the Premises to Lessee by the Commencement Date. If, despite said efforts, Lessor is unable to deliver possession by such date, Lessor shall not be subject to any liability therefor, nor shall such failure affect the validity of this Lease or change the Expiration Date. Lessee shall not, however, be obligated to pay Rent or perform its other obligations until Lessor delivers possession of the Premises and any period of rent abatement that Lessee would otherwise have enjoyed shall run from the date of delivery of possession and continue for a period equal to what Lessee would otherwise have enjoyed under the terms hereof, but minus any days of delay caused by the acts or omissions of

INITIALS

Lessee. If possession is not delivered within 60 days after the Commencement Date, as the same may be extended under the terms of any Work Letter executed by Parties, Lessee may, at its option, by notice in writing within 10 days after the end of such 60 day period, cancel this Lease, in which event the Parties shall be discharged from all obligations hereunder. If such written notice is not received by Lessor within said 10 day period, Lessee's right to cancel shall terminate. If possession of the Premises is not delivered within 120 days after the Commencement Date, this Lease shall terminate unless other agreements are reached between Lessor and Lessee, in writing.

3.4 Lessee Compliance. Lessor shall not be required to deliver possession of the Premises to Lessee until Lessee complies with its obligation to provide evidence of insurance (Paragraph 8.5). Pending delivery of such evidence, Lessee shall be required to perform all of its obligations under this Lease from and after the Start Date, including the payment of Rent, notwithstanding Lessor's election to withhold possession pending receipt of such evidence of insurance. Further, if Lessee is required to perform any other conditions prior to or concurrent with the Start Date, the Start Date shall occur but Lessor may elect to withhold possession until such conditions are satisfied.

Rent.

- 4.1 Rent Defined. All monetary obligations of Lessee to Lessor under the terms of this Lease (except for the Security Deposit) are deemed to be rent ("Rent").
- 4.2 Operating Expenses. Lessee shall pay to Lessor during the term hereof, in addition to the Base Rent, Lessee's Share of all Operating Expenses, as hereinafter defined, during each calendar year of the term of this Lease, in accordance with the following provisions:
- "Operating Expenses" include all costs relating to the ownership and operation of the Project, calculated as if the Project was at least 95% occupied, including, but not limited to, the following:
 - The operation, repair, and maintenance in neat, clean, safe, good order and condition, of the following: (i)
- (aa) The Common Areas, including their surfaces, coverings, decorative items, carpets, drapes and window coverings, and including parking areas, loading and unloading areas, trash areas, roadways, sidewalks, walkways, stairways, parkways, driveways, landscaped areas, striping, bumpers, irrigation systems, Common Area lighting facilities, building exteriors and roofs, fences and gates;
- (bb) All heating, air conditioning, plumbing, electrical systems, life safety equipment, communication systems and other equipment used in common by, or for the benefit of, lessees or occupants of the Project, including elevators and escalators, tenant directories, fire detection systems including sprinkler system maintenance and repair.
 - (cc) The Premises and/or any other space occupied by a tenant.
 - The cost of trash disposal, janitorial and security services, pest control services, and the costs of any environmental inspections;
 - The cost of any other service to be provided by Lessor that is elsewhere in this Lease stated to be an "Operating Expense"
- The cost of the premiums for the insurance policies maintained by Lessor pursuant to paragraph 8 and any deductible portion of an insured loss concerning the Building or the Common Areas;
 - The amount of the Real Property Taxes payable by Lessor pursuant to paragraph 10; (v)
 - (vi) The cost of water, sewer, gas, electricity, and other publicly mandated services not separately metered,
- (vii) Labor, salaries, and applicable fringe benefits and costs, materials, supplies and tools, used in maintaining and/or cleaning the Project and accounting and management fees attributable to the operation of the Project;
- (viii) The cost to replace equipment or capital components such as the roof, foundations, or exterior walls, the cost to replace a Common Area capital improvement, such as the parking lot paving, elevators or fences, and/or the cost of any capital improvement to the Building or the Project not covered under the provisions of Paragraph 2.3. Provided however, that if such equipment or capital component has a useful life for accounting purposes of 5 years or more that Lessor shall allocate the cost of any such capital improvement over a 12 year period and Lessee shall not be required to pay more than Lessee's Share of 1/144th of the cost of such capital improvement in any given month;
 - (ix) The cost to replace equipment or improvements that have a useful life for accounting purposes of 5 years or less.
 - Reserves set aside for maintenance, repair and/or replacement of Common Area improvements and equipment.
- Any item of Operating Expense that is specifically attributable to the Premises, the Building or to any other building in the Project or to the operation, repair and maintenance thereof, shall be allocated entirely to such Premises, Building, or other building. However, any such item that is not specifically attributable to the Building or to any other building or to the operation, repair and maintenance thereof, shall be equitably allocated by Lessor to all buildings in the Project.
- The inclusion of the improvements, facilities and services set forth in Subparagraph 4.2(a) shall not be deemed to impose an obligation upon Lessor to either have said improvements or facilities or to provide those services unless the Project already has the same, Lessor already provides the services, or Lessor has agreed elsewhere in this Lease to provide the same or some of them.
- (d) Lessee's Share of Operating Expenses is payable monthly on the same day as the Base Rent is due hereunder. The amount of such payments shall be based on Lessor's estimate of the Operating Expenses. Within 60 days after written request (but not more than once each year) Lessor shall deliver to Lessee a reasonably detailed statement showing Lessee's Share of the actual Operating Expenses for the preceding year. If Lessee's payments during such year exceed Lessee's Share, Lessor shall credit the amount of such over-payment against Lessee's future payments. If Lessee's payments during such year were less than Lessee's Share, Lessee shall pay to Lessor the amount of the deficiency within 10 days after delivery by Lessor to Lessee of the statement.

 (e) Operating Expenses shall not include any expenses paid by any tenant directly to third parties, or as to which Lessor is otherwise reimbursed by any
- third party, other tenant, or by insurance proceeds.
- 4.3 Payment. Lessee shall cause payment of Rent to be received by Lessor in lawful money of the United States, without offset or deduction (except as specifically permitted in this Lease), on or before the day on which it is due. All monetary amounts shall be rounded to the nearest whole dollar. In the event that any invoice prepared by Lessor is inaccurate such inaccuracy shall not constitute a waiver and Lessee shall be obligated to pay the amount set forth in this Lease. Rent for any period during the term hereof which is for less than one full calendar month shall be prorated based upon the actual number of days of said month. Payment of Rent shall be made to Lessor at its address stated herein or to such other persons or place as Lessor may from time to time designate in writing. Acceptance of a payment which is less than the amount then due shall not be a waiver of Lessor's rights to the balance of such Rent, regardless of Lessor's endorsement of any check so stating. In the event that any check, draft, or other instrument of payment given by Lessee to Lessor is dishonored for any reason, Lessee agrees to pay to Lessor the sum of \$25 in addition to any Late Charge and Lessor, at its option, may require all future Rent be paid by cashier's check. Payments will be applied first to accrued late charges and attorney's fees, second to accrued interest, then to Base Rent and Operating Expenses, and any remaining amount to any other outstanding charges
- Security Deposit. Lessee shall deposit with Lessor upon execution hereof the Security Deposit as security for Lessee's faithful performance of its obligations under this Lease. If Lessee fails to pay Rent, or otherwise Defaults under this Lease, Lessor may use, apply or retain all or any portion of said Security Deposit for the payment of any amount already due Lessor, for Rents which will be due in the future, and/or to reimburse or compensate Lessor for any liability, expense, loss or damage which Lessor may suffer or incur by reason thereof. If Lessor uses or applies all or any portion of the Security Deposit, Lessee shall within 10 days after written request therefor deposit monies with Lessor sufficient to restore said Security Deposit to the full amount required by this Lease. If the Base Rent increases during the term of this Lease, Lessee shall, upon written request from Lessor, deposit additional monies with Lessor so that the total amount of the Security Deposit shall at all times bear the same proportion to the increased Base Rent as the initial Security Deposit bore to the initial Base Rent. Should the Agreed Use be amended to accommodate a material change in the business of Lessee or to accommodate a sublessee or assignee, Lessor shall have the right to increase the Security Deposit to the extent necessary, in Lessor's reasonable judgment, to account for any increased wear and tear that the Premises may suffer as a result thereof. If a change in control of Lessee occurs during this Lease and following such change the financial condition of Lessee is, in Lessor's reasonable judgment, significantly reduced, Lessee shall deposit such additional monies with Lessor as shall be sufficient to cause the Security Deposit to be at a commercially reasonable level based on such change in financial condition. Lessor shall not be required to keep the Security Deposit separate from its general accounts. Within 90 days after the expiration or termination of this Lease, Lessor shall return that portion of the Security Deposit not used or applied by Lessor. Lessor shall upon written request provide Lessee with an accounting showing how that portion of the Security Deposit that was not returned was applied. No part of the Security Deposit shall be considered to be held in trust, to bear interest or to be prepayment for any monies to be paid by Lessee under this Lease. THE SECURITY DEPOSIT SHALL NOT BE USED BY LESSEE IN LIEU OF PAYMENT OF THE LAST MONTH'S RENT.

6.1 Use. Lessee shall use and occupy the Premises only for the Agreed Use, or any other legal use which is reasonably comparable thereto, and for no other purpose. Lessee shall not use or permit the use of the Premises in a manner that is unlawful, creates damage, waste or a nuisance, or that disturbs occupants of or causes damage to neighboring premises or properties. Other than guide, signal and seeing eye dogs, Lessee shall not keep or allow in the Premises any pets, animals, birds, fish, or reptiles. Lessor shall not unreasonably withhold or delay its consent to any written request for a modification of the Agreed Use, so long as the same will not impair the structural integrity of the improvements of the Building, will not adversely affect the mechanical, electrical, HVAC, and other systems of the Building, and/or will not affect the exterior appearance of the Building. If Lessor elects to withhold consent, Lessor shall within 7 days after such request give written notification of same, which notice shall include an explanation of Lessor's objections to the change in the Agreed Use.

6.2 Hazardous Substances

IN	ΙT	ΙΑΙ	S

- (a) Reportable Uses Require Consent. The term "Hazardous Substance" as used in this Lease shall mean any product, substance, or waste whose presence, use, manufacture, disposal, transportation, or release, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentially injurious to the public health, safety or welfare, the environment or the Premises, (ii) regulated or monitored by any governmental authority, or (iii) a basis for potential liability of Lessor to any governmental agency or third party under any applicable statute or common law theory. Hazardous Substances shall include, but not be limited to, hydrocarbons, petroleum, gasoline, and/or crude oil or any products, byproducts or fractions thereof. Lessee shall not engage in any activity in or the Premises which constitutes a Reportable Use of Hazardous Substances without the express prior written consent of Lessor and timely compliance (at Lessee's expense) with all Applicable Requirements. "Reportable Use" shall mean (i) the installation or use of any above or below ground storage tank, (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and/or (iii) the presence at the Premises of a Hazardous Substance with respect to which any Applicable Requirements requires that a notice be given to persons entering or occupying the Premises or neighboring properties. Notwithstanding the foregoing. Lessee may use any ordinary and customary materials reasonably required to be used in the normal course of the Agreed Use such as ordinary office supplies (copier toner, liquid paper, glue, etc.) and common household cleaning materials, so long as such use is in compliance with all Applicable Requirements, is not a Reportable Use, and does not expose the Premises or neighboring property to any meaningful risk of contamination or damage or expose Lessor to any liability therefor. In addition, Lessor may condition its consent to any Reportable Use upon receiving such additional assurances as Lessor reasonably deems necessary to protect itself, the public, the Premises and/or the environment against damage, contamination, injury and/or liability, including, but not limited to, the installation (and removal on or before Lease expiration or termination) of protective modifications (such as concrete encasements) and/or increasing the Security Deposit.
- (b) **Duty to Inform Lessor**. If Lessee knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises, other than as previously consented to by Lessor, Lessee shall immediately give written notice of such fact to Lessor, and provide Lessor with a copy of any report, notice, claim or other documentation which it has concerning the presence of such Hazardous Substance.
- (c) Lessee Remediation. Lessee shall not cause or permit any Hazardous Substance to be spilled or released in, on, under, or about the Premises (including through the plumbing or sanitary sewer system) and shall promptly, at Lessee's expense, comply with all Applicable Requirements and take all investigatory and/or remedial action reasonably recommended, whether or not formally ordered or required, for the cleanup of any contamination of, and for the maintenance, security and/or monitoring of the Premises or neighboring properties, that was caused or materially contributed to by Lessee, or pertaining to or involving any Hazardous Substance brought onto the Premises during the term of this Lease, by or for Lessee, or any third party.
- (d) Lessee Indemnification. Lessee shall indemnify, defend and hold Lessor, its agents, employees, lenders and ground lessor, if any, harmless from and against any and all loss of rents and/or damages, liabilities, judgments, claims, expenses, penalties, and attorneys' and consultants' fees arising out of or involving any Hazardous Substance brought onto the Premises by or for Lessee, or any third party (provided, however, that Lessee shall have no liability under this Lease with respect to underground migration of any Hazardous Substance under the Premises from areas outside of the Project not caused or contributed to by Lessee). Lessee's obligations shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Lessee, and the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease. No termination, cancellation or release agreement entered into by Lessor and Lessee shall release Lessee from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Lessor in writing at the time of such agreement.
- (e) Lessor Indemnification. Except as otherwise provided in paragraph 8.7, Lessor and its successors and assigns shall indemnify, defend, reimburse and hold Lessee, its employees and lenders, harmless from and against any and all environmental damages, including the cost of remediation, which result from Hazardous Substances which existed on the Premises prior to Lessee's occupancy or which are caused by the gross negligence or willful misconduct of Lessor, its agents or employees. Lessor's obligations, as and when required by the Applicable Requirements, shall include, but not be limited to, the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease.
- (f) Investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by governmental entities having jurisdiction with respect to the existence of Hazardous Substances on the Premises prior to Lessee's occupancy, unless such remediation measure is required as a result of Lessee's use (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises, in which event Lessee shall be responsible for such payment. Lessee shall cooperate fully in any such activities at the request of Lessor, including allowing Lessor and Lessor's agents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remedial responsibilities.
- (g) Lessor Termination Option. If a Hazardous Substance Condition (see Paragraph 9.1(e)) occurs during the term of this Lease, unless Lessee is legally responsible therefor (in which case Lessee shall make the investigation and remediation thereof required by the Applicable Requirements and this Lease shall continue in full force and effect, but subject to Lessor's rights under Paragraph 6.2(d) and Paragraph 13), Lessor may, at Lessor's option, either (i) investigate and remediate such Hazardous Substance Condition, if required, as soon as reasonably possible at Lessor's expense, in which event this Lease shall continue in full force and effect, or (ii) if the estimated cost to remediate such condition exceeds 12 times the then monthly Base Rent or \$100,000, whichever is greater, give written notice to Lessee, within 30 days after receipt by Lessor of knowledge of the occurrence of such Hazardous Substance Condition, of Lessor's desire to terminate this Lease as of the date 60 days following the date of such notice. In the event Lessor elects to give a termination notice, Lessee may, within 10 days thereafter, give written notice to Lessor of Lessee's commitment to pay the amount by which the cost of the remediation of such Hazardous Substance Condition exceeds an amount equal to 12 times the then monthly Base Rent or \$100,000, whichever is greater. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days following such commitment. In such event, this Lease shall continue in full force and effect, and Lessor shall proceed to make such remediation as soon as reasonably possible after the required funds are available. If Lessee does not give such notice and provide the required funds or assurance thereof within the time provided, this Lease shall terminate as of the date specified in Lessor's notice of termination.
- 6.3 Lessee's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lessee's sole expense, fully, diligently and in a timely manner, materially comply with all Applicable Requirements, the requirements of any applicable fire insurance underwriter or rating bureau, and the recommendations of Lessor's engineers and/or consultants which relate in any manner to the Premises, without regard to whether said Applicable Requirements are now in effect or become effective after the Start Date. Lessee shall, within 10 days after receipt of Lessor's written request, provide Lessor with copies of all permits and other documents, and other information evidencing Lessee's compliance with any Applicable Requirements specified by Lessor, and shall immediately upon receipt, notify Lessor in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving the failure of Lessee or the Premises to comply with any Applicable Requirements. Likewise, Lessee shall immediately give written notice to Lessor of: (i) any water damage to the Premises and any suspected seepage, pooling, dampness or other condition conducive to the production of mold; or (ii) any mustiness or other odors that might indicate the presence of mold in the Premises.
- other odors that might indicate the presence of mold in the Premises.

 6.4 Inspection; Compliance. Lessor and Lessor's "Lender" (as defined in Paragraph 30) and consultants authorized by Lessor shall have the right to enter into Premises at any time, in the case of an emergency, and otherwise at reasonable times, after reasonable notice, for the purpose of inspecting and/or testing the condition of the Premises and/or for verifying compliance by Lessee with this Lease. The cost of any such inspections shall be paid by Lessor, unless a violation of Applicable Requirements, or a Hazardous Substance Condition (see Paragraph 9.1(e)) is found to exist or be imminent, or the inspection is requested or ordered by a governmental authority. In such case, Lessee shall upon request reimburse Lessor for the cost of such inspection, so long as such inspection is reasonably related to the violation or contamination. In addition, Lessee shall provide copies of all relevant material safety data sheets (MSDS) to Lessor within 10 days of the receipt of written request therefor. Lessee acknowledges that any failure on its part to allow such inspections or testing will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to allow such inspections and/or testing in a timely fashion the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for the remainder to the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to allow such inspection and/or testing. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to such failure nor prevent the exercise of any of the other rights and remedies
- 7. Maintenance; Repairs; Utility Installations; Trade Fixtures and Alterations.
- 7.1 **Lessee's Obligations**. Notwithstanding Lessor's obligation to keep the Premises in good condition and repair, Lessee shall be responsible for the cost of painting, repairing or replacing wall coverings, and to repair or replace any improvements within the Premises.
- 7.2 **Lessor's Obligations**. Subject to the provisions of Paragraphs 2.2 (Condition), 2.3 (Compliance), 4.2 (Operating Expenses), 6 (Use), 7.1 (Lessee's Obligations), 9 (Damage or Destruction) and 14 (Condemnation), Lessor, subject to reimbursement pursuant to Paragraph 4.2, shall keep in good order, condition and repair the Premises, the foundations, exterior walls, structural condition of interior bearing walls, exterior roof, fire sprinkler system, fire alarm and/or smoke detection systems. fire hydrants, and the Common Areas.
 - 7.3 Utility Installations; Trade Fixtures; Alterations.
- (a) **Definitions**. The term "**Utility Installations**" refers to all floor and window coverings, air lines, vacuum lines, power panels, electrical distribution, security and fire protection systems, communication cabling, lighting fixtures, HVAC equipment, and plumbing in or on the Premises. The term "**Trade Fixtures**" shall mean Lessee's machinery and equipment that can be removed without doing material damage to the Premises. The term "**Alterations**" shall mean any modification of the improvements, other than Utility Installations or Trade Fixtures, whether by addition or deletion. "**Lessee Owned Alterations and/or Utility Installations**" are defined as Alterations and/or Utility Installations made by Lessee that are not yet owned by Lessor pursuant to Paragraph 7.4(a).

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

- (b) Consent. Lessee shall not make any Alterations or Utility Installations to the Premises without Lessor's prior written consent. Lessee may, however, make non-structural Alterations or Utility Installations to the interior of the Premises (excluding the roof) without such consent but upon notice to Lessor, as long as they are not visible from the outside, do not involve puncturing, relocating or removing the roof, ceilings, floors or any existing walls, will not affect the electrical, plumbing, HVAC, and/or life safety systems, do not trigger the requirement for additional modifications and/or improvements to the Premises resulting from Applicable Requirements, such as compliance with Title 24, and the cumulative cost thereof during this Lease as extended does not exceed \$2000. Notwithstanding the foregoing, Lessee shall not make or permit any roof penetrations and/or install anything on the roof without the prior written approval of Lessor. Lessor may, as a precondition to granting such approval, require Lessee to utilize a contractor chosen and/or approved by Lessor. Any Alterations or Utility Installations that Lessee shall desire to make and which require the consent of the Lessor shall be presented to Lessor in written form with detailed plans. Consent shall be deemed conditioned upon Lessee's: (i) acquiring all applicable governmental permits, (ii) furnishing Lessor with copies of both the permits and the plans and specifications prior to commencement of the work, and (iii) compliance with all conditions of said permits and other Applicable Requirements in a prompt and expeditious manner. Any Alterations or Utility Installations shall be performed in a workmanlike manner with good and sufficient materials. Lessee shall promptly upon completion furnish Lessor with as-built plans and specifications. For work which costs an amount in excess of one month's Base Rent, Lessor may condition its consent upon Lessee's posting an additional Security Deposit with Lessor.
- (c) Liens; Bonds. Lessee shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lessee at or for use on the Premises, which claims are or may be secured by any mechanic's or materialmen's lien against the Premises or any interest therein. Lessee shall give Lessor not less than 10 days notice prior to the commencement of any work in, on or about the Premises, and Lessor shall have the right to post notices of non-responsibility. If Lessee shall contest the validity of any such lien, claim or demand, then Lessee shall, at its sole expense defend and protect itself, Lessor and the Premises against the same and shall pay and satisfy any such adverse judgment that may be rendered thereon before the enforcement thereof. If Lessor shall require, Lessee shall furnish a surety bond in an amount equal to 150% of the amount of such contested lien, claim or demand, indemnifying Lessor against liability for the same. If Lessor elects to participate in any such action, Lessee shall pay Lessor's attorneys' fees and costs.

7.4 Ownership; Removal; Surrender; and Restoration.

- (a) **Ownership.** Subject to Lessor's right to require removal or elect ownership as hereinafter provided, all Alterations and Utility Installations made by Lessee shall be the property of Lessee, but considered a part of the Premises. Lessor may, at any time, elect in writing to be the owner of all or any specified part of the Lessee Owned Alterations and Utility Installations. Unless otherwise instructed per paragraph 7.4(b) hereof, all Lessee Owned Alterations and Utility Installations shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.
- (b) **Removal**. By delivery to Lessee of written notice from Lessor not earlier than 90 and not later than 30 days prior to the end of the term of this Lease, Lessor may require that any or all Lessee Owned Alterations or Utility Installations be removed by the expiration or termination of this Lease. Lessor may require the removal at any time of all or any part of any Lessee Owned Alterations or Utility Installations made without the required consent.
- (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements, parts and surfaces thereof clean and free of debris, and in good operating order, condition and state of repair, ordinary wear and tear excepted. "Ordinary wear and tear" shall not include any damage or deterioration that would have been prevented by good maintenance practice. Notwithstanding the foregoing and the provisions of Paragraph 7.1(a), if the Lessee occupies the Premises for 12 months or less, then Lessee shall surrender the Premises in the same condition as delivered to Lessee on the Start Date with NO allowance for ordinary wear and tear. Lessee shall repair any damage occasioned by the installation, maintenance or removal of Trade Fixtures, Lessee owned Alterations and/or Utility Installations, furnishings, and equipment as well as the removal of any storage tank installed by or for Lessee. Lessee shall also remove from the Premises any and all Hazardous Substances brought onto the Premises by or for Lessee, or any third party (except Hazardous Substances which were deposited via underground migration from areas outside of the Project) to the level specified in Applicable Requirements. Trade Fixtures shall remain the property of Lessee and shall be removed by Lessee. Any personal property of Lessee not removed on or before the Expiration Date or any earlier termination date shall be deemed to have been abandoned by Lessee and may be disposed of or retained by Lessor as Lessor may desire. The failure by Lessee to timely vacate the Premises pursuant to this Paragraph 7.4(c) without the express written consent of Lessor shall constitute a holdover under the provisions of Paragraph 26 below.

8. Insurance; Indemnity.

8.1 Insurance Premiums. The cost of the premiums for the insurance policies maintained by Lessor pursuant to paragraph 8 are included as Operating Expenses (see paragraph 4.2 (a)(iv)). Said costs shall include increases in the premiums resulting from additional coverage related to requirements of the holder of a mortgage or deed of trust covering the Premises, Building and/or Project, increased valuation of the Premises, Building and/or Project, and/or a general premium rate increase. Said costs shall not, however, include any premium increases resulting from the nature of the occupancy of any other tenant of the Building. In no event, however, shall Lessee be responsible for any portion of the premium cost attributable to liability insurance coverage in excess of \$2,000,000 procured under Paragraph 8.2(b).

8.2 Liability Insurance.

- (a) Carried by Lessee. Lessee shall obtain and keep in force a Commercial General Liability policy of insurance protecting Lessee and Lessor as an additional insured against claims for bodily injury, personal injury and property damage based upon or arising out of the ownership, use, occupancy or maintenance of the Premises and all areas appurtenant thereto. Such insurance shall be on an occurrence basis providing single limit coverage in an amount not less than \$1,000,000 per occurrence with an annual aggregate of not less than \$2,000,000. Lessee shall add Lessor as an additional insured by means of an endorsement at least as broad as the Insurance Service Organization's "Additional Insured-Managers or Lessors of Premises" Endorsement. The policy shall not contain any intra-insured exclusions as between insured persons or organizations, but shall include coverage for liability assumed under this Lease as an "insured contract" for the performance of Lessee's indemnity obligations under this Lease. The limits of said insurance shall not, however, limit the liability of Lessee nor relieve Lessee of any obligation hereunder. Lessee shall provide an endorsement on its liability policy(ies) which provides that its insurance shall be primary to and not contributory with any similar insurance carried by Lessor, whose insurance shall be considered excess insurance only.
- (b) **Carried by Lessor**. Lessor shall maintain liability insurance as described in Paragraph 8.2(a), in addition to, and not in lieu of, the insurance required to be maintained by Lessee. Lessee shall not be named as an additional insured therein.

8.3 Property Insurance - Building, Improvements and Rental Value.

- (a) **Building and Improvements**. Lessor shall obtain and keep in force a policy or policies of insurance in the name of Lessor, with loss payable to Lessor, any ground-lessor, and to any Lender insuring loss or damage to the Building and/or Project. The amount of such insurance shall be equal to the full insurable replacement cost of the Building and/or Project, as the same shall exist from time to time, or the amount required by any Lender, but in no event more than the commercially reasonable and available insurable value thereof. Lessee Owned Alterations and Utility Installations, Trade Fixtures, and Lessee's personal property shall be insured by Lessee not by Lessor. If the coverage is available and commercially appropriate, such policy or policies shall insure against all risks of direct physical loss or damage (except the perils of flood and/or earthquake unless required by a Lender), including coverage for debris removal and the enforcement of any Applicable Requirements requiring the upgrading, demolition, reconstruction or replacement of any portion of the Premises as the result of a covered loss. Said policy or policies shall also contain an agreed valuation provision in lieu of any coinsurance clause, waiver of subrogation, and inflation guard protection causing an increase in the annual property insurance coverage amount by a factor of not less than the adjusted U.S. Department of Labor Consumer Price Index for All Urban Consumers for the city nearest to where the Premises are located. If such insurance coverage has a deductible clause, the deductible amount shall not exceed \$5,000 per occurrence.
- (b) Rental Value. Lessor shall also obtain and keep in force a policy or policies in the name of Lessor with loss payable to Lessor and any Lender, insuring the loss of the full Rent for one year with an extended period of indemnity for an additional 180 days ("Rental Value insurance"). Said insurance shall contain an agreed valuation provision in lieu of any coinsurance clause, and the amount of coverage shall be adjusted annually to reflect the projected Rent otherwise payable by Lessee, for the next 12 month period.
- (c) Adjacent Premises. Lessee shall pay for any increase in the premiums for the property insurance of the Building and for the Common Areas or other buildings in the Project if said increase is caused by Lessee's acts, omissions, use or occupancy of the Premises.
- (d) Lessee's Improvements. Since Lessor is the Insuring Party, Lessor shall not be required to insure Lessee Owned Alterations and Utility Installations unless the item in question has become the property of Lessor under the terms of this Lease.

8.4 Lessee's Property; Business Interruption Insurance; Worker's Compensation Insurance.

- (a) **Property Damage**. Lessee shall obtain and maintain insurance coverage on all of Lessee's personal property, Trade Fixtures, and Lessee Owned Alterations and Utility Installations. Such insurance shall be full replacement cost coverage with a deductible of not to exceed \$1,000 per occurrence. The proceeds from any such insurance shall be used by Lessee for the replacement of personal property, Trade Fixtures and Lessee Owned Alterations and Utility Installations.
- (b) **Business Interruption**. Lessee shall obtain and maintain loss of income and extra expense insurance in amounts as will reimburse Lessee for direct or indirect loss of earnings attributable to all perils commonly insured against by prudent lessees in the business of Lessee or attributable to prevention of access to the Premises as a result of such perils.
- (c) Worker's Compensation Insurance. Lessee shall obtain and maintain Worker's Compensation Insurance in such amount as may be required by Applicable Requirements. Such policy shall include a 'Waiver of Subrogation' endorsement. Lessee shall provide Lessor with a copy of such endorsement along with the certificate of insurance or copy of the policy required by paragraph 8.5.
 - (d) No Representation of Adequate Coverage. Lessor makes no representation that the limits or forms of coverage of insurance specified herein are

adequate to cover Lessee's property, business operations or obligations under this Lease.

- 8.5 **Insurance Policies**. Insurance required herein shall be by companies maintaining during the policy term a "General Policyholders Rating" of at least A-, VII, as set forth in the most current issue of "Best's Insurance Guide", or such other rating as may be required by a Lender. Lessee shall not do or permit to be done anything which invalidates the required insurance policies. Lessee shall, prior to the Start Date, deliver to Lessor certified copies of policies of such insurance or certificates with copies of the required endorsements evidencing the existence and amounts of the required insurance. No such policy shall be cancelable or subject to modification except after 30 days prior written notice to Lessor. Lessee shall, at least 10 days prior to the expiration of such policies, furnish Lessor with evidence of renewals or "insurance binders" evidencing renewal thereof, or Lessor may increase his liability insurance coverage and charge the cost thereof to Lessee, which amount shall be payable by Lessee to Lessor upon demand. Such policies shall be for a term of at least one year, or the length of the remaining term of this Lease, whichever is less. If either Party shall fail to procure and maintain the insurance required to be carried by it, the other Party may, but shall not be required to, procure and maintain the same.
- 8.6 Waiver of Subrogation. Without affecting any other rights or remedies, Lessee and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is not limited by the amount of insurance carried or required, or by any deductibles applicable hereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Lessee, as the case may be, so long as the insurance is not invalidated thereby.
- 8.7 **Indemnity**. Except for Lessor's gross negligence or willful misconduct, Lessee shall indemnify, protect, defend and hold harmless the Premises, Lessor and its agents, Lessor's master or ground lessor, partners and Lenders, from and against any and all claims, loss of rents and/or damages, liens, judgments, penalties, attorneys' and consultants' fees, expenses and/or liabilities arising out of, involving, or in connection with, a Breach of the Lease by Lessee and/or the use and/or occupancy of the Premises and/or Project by Lessee and/or by Lessee's employees, contractors or invitees. If any action or proceeding is brought against Lessor by reason of any of the foregoing matters, Lessee shall upon notice defend the same at Lessee's expense by counsel reasonably satisfactory to Lessor and Lessor shall cooperate with Lessee in such defense. Lessor need not have first paid any such claim in order to be defended or indemnified.
- 8.8 **Exemption of Lessor and its Agents from Liability**. Notwithstanding the negligence or breach of this Lease by Lessor or its agents, neither Lessor nor its agents shall be liable under any circumstances for: (i) injury or damage to the person or goods, wares, merchandise or other property of Lessee, Lessee's employees, contractors, invitees, customers, or any other person in or about the Premises, whether such damage or injury is caused by or results from fire, steam, electricity, gas, water or rain, indoor air quality, the presence of mold or from the breakage, leakage, obstruction or other defects of pipes, fire sprinklers, wires, appliances, plumbing, HVAC or lighting fixtures, or from any other cause, whether the said injury or damage results from conditions arising upon the Premises or upon other portions of the Building, or from other sources or places, (ii) any damages arising from any act or neglect of any other tenant of Lessor or from the failure of Lessor or its agents to enforce the provisions of any other lease in the Project, or (iii) injury to Lessee's business or for any loss of income or profit therefrom. Instead, it is intended that Lessee's sole recourse in the event of such damages or injury be to file a claim on the insurance policy(ies) that Lessee is required to maintain pursuant to the provisions of paragraph 8.
- 8.9 **Failure to Provide Insurance**. Lessee acknowledges that any failure on its part to obtain or maintain the insurance required herein will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, for any month or portion thereof that Lessee does not maintain the required insurance and/or does not provide Lessor with the required binders or certificates evidencing the existence of the required insurance, the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater. The parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to maintain the required insurance. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to maintain such insurance, prevent the exercise of any of the other rights and remedies granted hereunder, nor relieve Lessee of its obligation to maintain the insurance specified in this Lease.

9. Damage or Destruction.

9.1 Definitions

- (a) "Premises Partial Damage" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations, which can reasonably be repaired in 3 months or less from the date of the damage or destruction, and the cost thereof does not exceed a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (b) "Premises Total Destruction" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which cannot reasonably be repaired in 3 months or less from the date of the damage or destruction and/or the cost thereof exceeds a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (c) "Insured Loss" shall mean damage or destruction to improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which was caused by an event required to be covered by the insurance described in Paragraph 8.3(a), irrespective of any deductible amounts or coverage limits involved.
- (d) "Replacement Cost" shall mean the cost to repair or rebuild the improvements owned by Lessor at the time of the occurrence to their condition existing immediately prior thereto, including demolition, debris removal and upgrading required by the operation of Applicable Requirements, and without deduction for depreciation.
- (e) "Hazardous Substance Condition" shall mean the occurrence or discovery of a condition involving the presence of, or a contamination by, a Hazardous Substance, in, on, or under the Premises which requires restoration.
- 9.2 Partial Damage Insured Loss. If a Premises Partial Damage that is an Insured Loss occurs, then Lessor shall, at Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonably possible and this Lease shall continue in full force and effect; provided, however, that Lessee shall, at Lessor's election, make the repair of any damage or destruction the total cost to repair of which is \$5,000 or less, and, in such event, Lessor shall make any applicable insurance proceeds available to Lessee on a reasonable basis for that purpose. Notwithstanding the foregoing, if the required insurance was not in force or the insurance proceeds are not sufficient to effect such repair, the Insuring Party shall promptly contribute the shortage in proceeds as and when required to complete said repairs. In the event, however, such shortage was due to the fact that, by reason of the unique nature of the improvements, full replacement cost insurance coverage was not commercially reasonable and available, Lessor shall have no obligation to pay for the shortage in insurance proceeds or to fully restore the unique aspects of the Premises unless Lessee provides Lessor with the funds to cover same, or adequate assurance thereof, within 10 days following receipt of written notice of such shortage and request therefor. If Lessor receives said funds or adequate assurance thereof within said 10 day period, the party responsible for making the repairs shall complete them as soon as reasonably possible and this Lease shall remain in full force and effect. If such funds or assurance are not received, Lessor may nevertheless elect by written notice to Lessee within 10 days thereafter to: (i) make such restoration and repair as is commercially reasonable with Lessor paying any shortage in proceeds, in which case this Lease shall remain in full force and effect, or (ii) have this Lease terminate 30 days thereafter. Lessee shall not be entitled to reimbursement of a
- shall be made available for the repairs if made by either Party.

 9.3 Partial Damage Uninsured Loss. If a Premises Partial Damage that is not an Insured Loss occurs, unless caused by a negligent or willful act of Lessee (in which event Lessee shall make the repairs at Lessee's expense), Lessor may either: (i) repair such damage as soon as reasonably possible at Lessor's expense (subject to reimbursement pursuant to Paragraph 4.2), in which event this Lease shall continue in full force and effect, or (ii) terminate this Lease by giving written notice to Lessee within 30 days after receipt by Lessor of knowledge of the occurrence of such damage. Such termination shall be effective 60 days following the date of such notice. In the event Lessor elects to terminate this Lease, Lessee shall have the right within 10 days after receipt of the termination notice to give written notice to Lessor of Lessee's commitment to pay for the repair of such damage without reimbursement from Lessor. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days after making such commitment. In such event this Lease shall continue in full force and effect, and Lessor shall proceed to make such repairs as soon as reasonably possible after the required funds are available. If Lessee does not make the required commitment, this Lease shall terminate as of the date specified in the termination notice.
- 9.4 **Total Destruction**. Notwithstanding any other provision hereof, if a Premises Total Destruction occurs, this Lease shall terminate 60 days following such Destruction. If the damage or destruction was caused by the gross negligence or willful misconduct of Lessee, Lessor shall have the right to recover Lessor's damages from Lessee, except as provided in Paragraph 8.6.
- 9.5 Damage Near End of Term. If at any time during the last 6 months of this Lease there is damage for which the cost to repair exceeds one month's Base Rent, whether or not an Insured Loss, Lessor may terminate this Lease effective 60 days following the date of occurrence of such damage by giving a written termination notice to Lessee within 30 days after the date of occurrence of such damage. Notwithstanding the foregoing, if Lessee at that time has an exercisable option to extend this Lease or to purchase the Premises, then Lessee may preserve this Lease by, (a) exercising such option and (b) providing Lessor with any shortage in insurance proceeds (or adequate assurance thereof) needed to make the repairs on or before the earlier of (i) the date which is 10 days after Lessee's receipt of Lessor's written notice purporting to terminate this Lease, or (ii) the day prior to the date upon which such option expires. If Lessee duly exercises such option during such period and provides Lessor with funds (or adequate assurance thereof) to cover any shortage in insurance proceeds, Lessor shall, at Lessor's commercially

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

reasonable expense, repair such damage as soon as reasonably possible and this Lease shall continue in full force and effect. If Lessee fails to exercise such option and provide such funds or assurance during such period, then this Lease shall terminate on the date specified in the termination notice and Lessee's option shall be extinguished.

9.6 Abatement of Rent; Lessee's Remedies.

- Abatement. In the event of Premises Partial Damage or Premises Total Destruction or a Hazardous Substance Condition for which Lessee is not responsible under this Lease, the Rent payable by Lessee for the period required for the repair, remediation or restoration of such damage shall be abated in proportion to the degree to which Lessee's use of the Premises is impaired, but not to exceed the proceeds received from the Rental Value insurance. All other obligations of Lessee hereunder shall be performed by Lessee, and Lessor shall have no liability for any such damage, destruction, remediation, repair or restoration except as provided herein.
- (b) Remedies. If Lessor shall be obligated to repair or restore the Premises and does not commence, in a substantial and meaningful way, such repair or restoration within 90 days after such obligation shall accrue, Lessee may, at any time prior to the commencement of such repair or restoration, give written notice to Lessor and to any Lenders of which Lessee has actual notice, of Lessee's election to terminate this Lease on a date not less than 60 days following the giving of such notice. If Lessee gives such notice and such repair or restoration is not commenced within 30 days thereafter, this Lease shall terminate as of the date specified in said notice. If the repair or restoration is commenced within such 30 days, this Lease shall continue in full force and effect. "Commence" shall mean either the unconditional authorization of the preparation of the required plans, or the beginning of the actual work on the Premises, whichever first occurs.
- 9.7 Termination; Advance Payments. Upon termination of this Lease pursuant to Paragraph 6.2(g) or Paragraph 9, an equitable adjustment shall be made concerning advance Base Rent and any other advance payments made by Lessee to Lessor. Lessor shall, in addition, return to Lessee so much of Lessee's Security Deposit as has not been, or is not then required to be, used by Lessor.

Real Property Taxes.

- 10.1 Definitions. As used herein, the term "Real Property Taxes" shall include any form of assessment; real estate, general, special, ordinary or extraordinary, or rental levy or tax (other than inheritance, personal income or estate taxes); improvement bond; and/or license fee imposed upon or levied against any legal or equitable interest of Lessor in the Project, Lessor's right to other income therefrom, and/or Lessor's business of leasing, by any authority having the direct or indirect power to tax and where the funds are generated with reference to the Project address. "Real Property Taxes" shall also include any tax, fee, levy, assessment or charge, or any increase therein: (i) imposed by reason of events occurring during the term of this Lease, including but not limited to, a change in the ownership of the Project, (ii) a change in the improvements thereon, and/or (iii) levied or assessed on machinery or equipment provided by Lessor to Lessee pursuant to this Lease.
- 10.2 Payment of Taxes. Except as otherwise provided in Paragraph 10.3, Lessor shall pay the Real Property Taxes applicable to the Project, and said payments shall be included in the calculation of Operating Expenses in accordance with the provisions of Paragraph 4.2.
- 10.3 Additional Improvements. Operating Expenses shall not include Real Property Taxes specified in the tax assessor's records and work sheets as being caused by additional improvements placed upon the Project by other lessees or by Lessor for the exclusive enjoyment of such other lessees. Notwithstanding Paragraph 10.2 hereof, Lessee shall, however, pay to Lessor at the time Operating Expenses are payable under Paragraph 4.2, the entirety of any increase in Real Property Taxes if assessed solely by reason of Alterations, Trade Fixtures or Utility Installations placed upon the Premises by Lessee or at Lessee's request or by reason of any alterations or improvements to the Premises made by Lessor subsequent to the execution of this Lease by the Parties,
- 10.4 Joint Assessment. If the Building is not separately assessed, Real Property Taxes allocated to the Building shall be an equitable proportion of the Real Property Taxes for all of the land and improvements included within the tax parcel assessed, such proportion to be determined by Lessor from the respective valuations assigned in the assessor's work sheets or such other information as may be reasonably available. Lessor's reasonable determination thereof, in good faith, shall be conclusive
- 10.5 Personal Property Taxes. Lessee shall pay prior to delinquency all taxes assessed against and levied upon Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all personal property of Lessee contained in the Premises. When possible, Lessee shall cause its Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all other personal property to be assessed and billed separately from the real property of Lessor. If any of Lessee's said property shall be assessed with Lessor's real property, Lessee shall pay Lessor the taxes attributable to Lessee's property within 10 days ofter receipt of a maintain attributable to Lessee's property within 10 days after receipt of a written statement setting forth the taxes applicable to Lessee's property.

- 11.1 Services Provided by Lessor. Lessor shall provide heating, ventilation, air conditioning, reasonable amounts of electricity for normal lighting and office machines, water for reasonable and normal drinking and lavatory use in connection with an office, and replacement light bulbs and/or fluorescent tubes and ballasts for standard overhead fixtures. Lessor shall also provide janitorial services to the Premises and Common Areas 5 times per week, excluding Building Holidays, or pursuant to the attached janitorial schedule, if any. Lessor shall not, however, be required to provide janitorial services to kitchens or storage areas included within the Premises.
- 11.2 Services Exclusive to Lessee. Notwithstanding the provision of paragraph 11.1, Lessee shall pay for all water, gas, heat, light, power, telephone and other utilities and services specially or exclusively supplied and/or metered exclusively to the Premises or to Lessee, together with any taxes thereon. If a service is deleted by Paragraph 1.13 and such service is not separately metered to the Premises, Lessee shall pay at Lessor's option, either Lessee's Share or a reasonable proportion to be determined by Lessor of all charges for such jointly metered service.
- 11.3 Hours of Service. Said services and utilities shall be provided during times set forth in Paragraph 1.12. Utilities and services required at other times shall be subject to advance request and reimbursement by Lessee to Lessor of the cost thereof.
- 11.4 Excess Usage by Lessee. Lessee shall not make connection to the utilities except by or through existing outlets and shall not install or use machinery or equipment in or about the Premises that uses excess water, lighting or power, or suffer or permit any act that causes extra burden upon the utilities or services, including but not limited to security and trash services, over standard office usage for the Project. Lessor shall require Lessee to reimburse Lessor for any excess expenses or costs that may arise out of a breach of this subparagraph by Lessee. Lessor may, in its sole discretion, install at Lessee's expense supplemental equipment and/or separate metering applicable to Lessee's excess usage or loading.
- 11.5 Interruptions. There shall be no abatement of rent and Lessor shall not be liable in any respect whatsoever for the inadequacy, stoppage, interruption or discontinuance of any utility or service due to riot, strike, labor dispute, breakdown, accident, repair or other cause beyond Lessor's reasonable control or in cooperation with governmental request or directions.
- 11.6 Within fifteen days of Lessor's written request, Lessee agrees to deliver to Lessor such information, documents and/or authorization as Lessor needs in order for Lessor to comply with new or existing Applicable Requirements relating to commercial building energy usage, ratings, and/or the reporting thereof.

Assignment and Subletting. 12.1 Lessor's Consent Required.

- (a) Lessee shall not voluntarily or by operation of law assign, transfer, mortgage or encumber (collectively, "assign or assignment") or sublet all or any part of Lessee's interest in this Lease or in the Premises without Lessor's prior written consent.
- (b) Unless Lessee is a corporation and its stock is publicly traded on a national stock exchange, a change in the control of Lessee shall constitute an assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of the voting control of Lessee shall constitute a change in control for this purpose. The involvement of Lessee or its assets in any transaction, or series of transactions (by way of merger, sale, acquisition, financing, transfer, leveraged buyout or otherwise), whether or not a formal assignment or hypothecation of this Lease or Lessee's assets occurs, which results or will result in a reduction of the Net Worth of Lessee by an amount greater than 25% of such Net Worth as it was represented at the time of the execution of this Lease or at the time of the most recent assignment to which Lessor has consented, or as it exists immediately prior to said transaction or transactions constituting such reduction, whichever was or is greater, shall be considered an assignment of this Lease to which Lessor may withhold its consent. "Net Worth of Lessee" shall mean the net worth of Lessee $(excluding\ any\ guarantors)\ established\ under\ generally\ accepted\ accounting\ principles.$
- (d) An assignment or subletting without consent shall, at Lessor's option, be a Default curable after notice per Paragraph 13.1(d), or a noncurable Breach without the necessity of any notice and grace period. If Lessor elects to treat such unapproved assignment or subletting as a noncurable Breach, Lessor may either: (i) terminate this Lease, or (ii) upon 30 days written notice, increase the monthly Base Rent to 110% of the Base Rent then in effect. Further, in the event of such Breach and rental adjustment, (i) the purchase price of any option to purchase the Premises held by Lessee shall be subject to similar adjustment to 110% of the price previously in effect, and (ii) all fixed and non-fixed rental adjustments scheduled during the remainder of the Lease term shall be increased to 110% of the scheduled
 - Lessee's remedy for any breach of Paragraph 12.1 by Lessor shall be limited to compensatory damages and/or injunctive relief.
 - (f) Lessor may reasonably withhold consent to a proposed assignment or subletting if Lessee is in Default at the time consent is requested.
- Notwithstanding the foregoing, allowing a de minimis portion of the Premises, ie. 20 square feet or less, to be used by a third party vendor in connection with the installation of a vending machine or payphone shall not constitute a subletting.

12.2 Terms and Conditions Applicable to Assignment and Subletting.

Regardless of Lessor's consent, no assignment or subletting shall: (i) be effective without the express written assumption by such assignee or

		INITIAL

Last Edited: 9/30/2022 2:33 PM

INITIALS

sublessee of the obligations of Lessee under this Lease, (ii) release Lessee of any obligations hereunder, or (iii) alter the primary liability of Lessee for the payment of Rent or for the performance of any other obligations to be performed by Lessee.

- (b) Lessor may accept Rent or performance of Lessee's obligations from any person other than Lessee pending approval or disapproval of an assignment. Neither a delay in the approval or disapproval of such assignment nor the acceptance of Rent or performance shall constitute a waiver or estoppel of Lessor's right to exercise its remedies for Lessee's Default or Breach.
 - (c) Lessor's consent to any assignment or subletting shall not constitute a consent to any subsequent assignment or subletting.
- (d) In the event of any Default or Breach by Lessee, Lessor may proceed directly against Lessee, any Guarantors or anyone else responsible for the performance of Lessee's obligations under this Lease, including any assignee or sublessee, without first exhausting Lessor's remedies against any other person or entity responsible therefor to Lessor, or any security held by Lessor.
- (e) Each request for consent to an assignment or subletting shall be in writing, accompanied by information relevant to Lessor's determination as to the financial and operational responsibility and appropriateness of the proposed assignee or sublessee, including but not limited to the intended use and/or required modification of the Premises, if any, together with a fee of \$500 as consideration for Lessor's considering and processing said request. Lessee agrees to provide Lessor with such other or additional information and/or documentation as may be reasonably requested. (See also Paragraph 36)
- (f) Any assignee of, or sublessee under, this Lease shall, by reason of accepting such assignment, entering into such sublease, or entering into possession of the Premises or any portion thereof, be deemed to have assumed and agreed to conform and comply with each and every term, covenant, condition and obligation herein to be observed or performed by Lessee during the term of said assignment or sublease, other than such obligations as are contrary to or inconsistent with provisions of an assignment or sublease to which Lessor has specifically consented to in writing.
- (g) Lessor's consent to any assignment or subletting shall not transfer to the assignee or sublessee any Option granted to the original Lessee by this Lease unless such transfer is specifically consented to by Lessor in writing. (See Paragraph 39.2)
- 12.3 Additional Terms and Conditions Applicable to Subletting. The following terms and conditions shall apply to any subletting by Lessee of all or any part of the Premises and shall be deemed included in all subleases under this Lease whether or not expressly incorporated therein:
- (a) Lessee hereby assigns and transfers to Lessor all of Lessee's interest in all Rent payable on any sublease, and Lessor may collect such Rent and apply same toward Lessee's obligations under this Lease; provided, however, that until a Breach shall occur in the performance of Lessee's obligations, Lessee may collect said Rent. In the event that the amount collected by Lessor exceeds Lessee's then outstanding obligations any such excess shall be refunded to Lessee. Lessor shall not, by reason of the foregoing or any assignment of such sublease, nor by reason of the collection of Rent, be deemed liable to the sublessee for any failure of Lessee to perform and comply with any of Lessee's obligations to such sublessee. Lessee hereby irrevocably authorizes and directs any such sublessee, upon receipt of a written notice from Lessor stating that a Breach exists in the performance of Lessee's obligations under this Lease, to pay to Lessor all Rent due and to become due under the sublease. Sublessee shall rely upon any such notice from Lessor and shall pay all Rents to Lessor without any obligation or right to inquire as to whether such Breach exists, notwithstanding any claim from Lessee to the contrary.
- (b) In the event of a Breach by Lessee, Lessor may, at its option, require sublessee to attorn to Lessor, in which event Lessor shall undertake the obligations of the sublessor under such sublease from the time of the exercise of said option to the expiration of such sublease; provided, however, Lessor shall not be liable for any prepaid rents or security deposit paid by such sublessee to such sublessor or for any prior Defaults or Breaches of such sublessor.
 - (c) Any matter requiring the consent of the sublessor under a sublease shall also require the consent of Lessor.
 - (d) No sublessee shall further assign or sublet all or any part of the Premises without Lessor's prior written consent.
- (e) Lessor shall deliver a copy of any notice of Default or Breach by Lessee to the sublessee, who shall have the right to cure the Default of Lessee within the grace period, if any, specified in such notice. The sublessee shall have a right of reimbursement and offset from and against Lessee for any such Defaults cured by the sublessee.

13. Default; Breach; Remedies.

- 13.1 **Default; Breach**. A "**Default**" is defined as a failure by the Lessee to comply with or perform any of the terms, covenants, conditions or Rules and Regulations under this Lease. A "**Breach**" is defined as the occurrence of one or more of the following Defaults, and the failure of Lessee to cure such Default within any applicable grace period:
- (a) The abandonment of the Premises; the vacating of the Premises prior to the expiration or termination of this Lease without providing a commercially reasonable level of security, or where the coverage of the property insurance described in Paragraph 8.3 is jeopardized as a result thereof, or without providing reasonable assurances to minimize potential vandalism; or failure to deliver to Lessor exclusive possession of the entire Premises in accordance herewith prior to the expiration or termination of this Lease.
- (b) The failure of Lessee to make any payment of Rent or any Security Deposit required to be made by Lessee hereunder, whether to Lessor or to a third party, when due, to provide reasonable evidence of insurance or surety bond, or to fulfill any obligation under this Lease which endangers or threatens life or property, where such failure continues for a period of 3 business days following written notice to Lessee. THE ACCEPTANCE BY LESSOR OF A PARTIAL PAYMENT OF RENT OR SECURITY DEPOSIT SHALL NOT CONSTITUTE A WAIVER OF ANY OF LESSOR'S RIGHTS, INCLUDING LESSOR'S RIGHT TO RECOVER POSSESSION OF THE PREMISES.
- (c) The failure of Lessee to allow Lessor and/or its agents access to the Premises or the commission of waste, act or acts constituting public or private nuisance, and/or an illegal activity on the Premises by Lessee, where such actions continue for a period of 3 business days following written notice to Lessee. In the event that Lessee commits waste, a nuisance or an illegal activity a second time then, the Lessor may elect to treat such conduct as a non-curable Breach rather than a Default
- (d) The failure by Lessee to provide (i) reasonable written evidence of compliance with Applicable Requirements, (ii) the service contracts, (iii) the rescission of an unauthorized assignment or subletting, (iv) an Estoppel Certificate or financial statements, (v) a requested subordination, (vi) evidence concerning any guaranty and/or Guarantor, (vii) any document requested under Paragraph 41, (viii) material safety data sheets (MSDS), or (ix) any other documentation or information which Lessor may reasonably require of Lessee under the terms of this Lease, where any such failure continues for a period of 10 days following written notice to Lessee.
- (e) A Default by Lessee as to the terms, covenants, conditions or provisions of this Lease, or of the rules adopted under Paragraph 2.9 hereof, other than those described in subparagraphs 13.1(a), (b), (c) or (d), above, where such Default continues for a period of 30 days after written notice; provided, however, that if the nature of Lessee's Default is such that more than 30 days are reasonably required for its cure, then it shall not be deemed to be a Breach if Lessee commences such cure within said 30 day period and thereafter diligently prosecutes such cure to completion.
- (f) The occurrence of any of the following events: (i) the making of any general arrangement or assignment for the benefit of creditors; (ii) becoming a "debtor" as defined in 11 U.S.C. § 101 or any successor statute thereto (unless, in the case of a petition filed against Lessee, the same is dismissed within 60 days); (iii) the appointment of a trustee or receiver to take possession of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where possession is not restored to Lessee within 30 days; or (iv) the attachment, execution or other judicial seizure of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where such seizure is not discharged within 30 days; provided, however, in the event that any provision of this subparagraph (e) is contrary to any applicable law, such provision shall be of no force or effect, and not affect the validity of the remaining provisions.
 - (g) The discovery that any financial statement of Lessee or of any Guarantor given to Lessor was materially false.
- (h) If the performance of Lessee's obligations under this Lease is guaranteed: (i) the death of a Guarantor, (ii) the termination of a Guarantor's liability with respect to this Lease other than in accordance with the terms of such guaranty, (iii) a Guarantor's becoming insolvent or the subject of a bankruptcy filing, (iv) a Guarantor's refusal to honor the guaranty, or (v) a Guarantor's breach of its guaranty obligation on an anticipatory basis, and Lessee's failure, within 60 days following written notice of any such event, to provide written alternative assurance or security, which, when coupled with the then existing resources of Lessee, equals or exceeds the combined financial resources of Lessee and the Guarantors that existed at the time of execution of this Lease.
- 13.2 Remedies. If Lessee fails to perform any of its affirmative duties or obligations, within 10 days after written notice (or in case of an emergency, without notice), Lessor may, at its option, perform such duty or obligation on Lessee's behalf, including but not limited to the obtaining of reasonably required bonds, insurance policies, or governmental licenses, permits or approvals. Lessee shall pay to Lessor an amount equal to 115% of the costs and expenses incurred by Lessor in such performance upon receipt of an invoice therefor. In the event of a Breach, Lessor may, with or without further notice or demand, and without limiting Lessor in the exercise of any right or remedy which Lessor may have by reason of such Breach:
- (a) Terminate Lessee's right to possession of the Premises by any lawful means, in which case this Lease shall terminate and Lessee shall immediately surrender possession to Lessor. In such event Lessor shall be entitled to recover from Lessee: (i) the unpaid Rent which had been earned at the time of termination; (ii) the worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that the Lessee proves could have been reasonably avoided; (iii) the worth at the time of award of the amount by which the unpaid rent for the balance of the term after the time of award exceeds the amount of such rental loss that the Lessee proves could be reasonably avoided; and (iv) any other amount necessary to compensate Lessor for all the detriment proximately caused by the Lessee's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom, including but not limited to the cost of recovering possession of the Premises, expenses of reletting, including necessary renovation and alteration of the Premises, reasonable attorneys' fees, and that portion of any leasing commission paid by Lessor in connection with this Lease applicable to the unexpired term of this Lease. The worth at the time of award of the amount referred to in provision (iii) of the immediately preceding

INITIALS

sentence shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of the District within which the Premises are located at the time of award plus one percent. Efforts by Lessor to mitigate damages caused by Lessee's Breach of this Lease shall not waive Lessor's right to recover any damages to which Lessor is otherwise entitled. If termination of this Lease is obtained through the provisional remedy of unlawful detainer, Lessor shall have the right to recover in such proceeding any unpaid Rent and damages as are recoverable therein, or Lessor may reserve the right to recover all or any part thereof in a separate suit. If a notice and grace period required under Paragraph 13.1 was not previously given, a notice to pay rent or quit, or to perform or quit given to Lessee under the unlawful detainer statute shall also constitute the notice required by Paragraph 13.1. In such case, the applicable grace period required by Paragraph 13.1 and the unlawful detainer statute shall run concurrently, and the failure of Lessee to cure the Default within the greater of the two such grace periods shall constitute both an unlawful detainer and a Breach of this Lease entitling Lessor to the remedies provided for in this Lease and/or by said statute.

- (b) Continue the Lease and Lessee's right to possession and recover the Rent as it becomes due, in which event Lessee may sublet or assign, subject only to reasonable limitations. Acts of maintenance, efforts to relet, and/or the appointment of a receiver to protect the Lessor's interests, shall not constitute a termination of the Lessee's right to possession.
- (c) Pursue any other remedy now or hereafter available under the laws or judicial decisions of the state wherein the Premises are located. The expiration or termination of this Lease and/or the termination of Lessee's right to possession shall not relieve Lessee from liability under any indemnity provisions of this Lease as to matters occurring or accruing during the term hereof or by reason of Lessee's occupancy of the Premises.
- 13.3 Inducement Recapture. Any agreement for free or abated rent or other charges, the cost of tenant improvements for Lessee paid for or performed by Lessor, or for the giving or paying by Lessor to or for Lessee of any cash or other bonus, inducement or consideration for Lessee's entering into this Lease, all of which concessions are hereinafter referred to as "Inducement Provisions," shall be deemed conditioned upon Lessee's full and faithful performance of all of the terms, covenants and conditions of this Lease. Upon Breach of this Lease by Lessee, any such Inducement Provision shall automatically be deemed deleted from this Lease and of no further force or effect, and any rent, other charge, bonus, inducement or consideration theretofore abated, given or paid by Lessor under such an Inducement Provision shall be immediately due and payable by Lessee to Lessor, notwithstanding any subsequent cure of said Breach by Lessee. The acceptance by Lessor of rent or the cure of the Breach which initiated the operation of this paragraph shall not be deemed a waiver by Lessor of the provisions of this paragraph unless specifically so stated in writing by Lessor at the time of such acceptance.
- 13.4 Late Charges. Lessee hereby acknowledges that late payment by Lessee of Rent will cause Lessor to incur costs not contemplated by this Lease, the exact amount of which will be extremely difficult to ascertain. Such costs include, but are not limited to, processing and accounting charges, and late charges which may be imposed upon Lessor by any Lender. Accordingly, if any Rent shall not be received by Lessor within 5 days after such amount shall be due, then, without any requirement for notice to Lessee, Lessee shall immediately pay to Lessor a one-time late charge equal to 10% of each such overdue amount or \$100, whichever is greater. The parties hereby agree that such late charge represents a fair and reasonable estimate of the costs Lessor will incur by reason of such late payment. Acceptance of such late charge by Lessor shall in no event constitute a waiver of Lessee's Default or Breach with respect to such overdue amount, nor prevent the exercise of any of the other rights and remedies granted hereunder. In the event that a late charge is payable hereunder, whether or not collected, for 3 consecutive installments of Base Rent, then notwithstanding any provision of this Lease to the contrary, Base Rent shall, at Lessor's option, become due and payable quarterly in advance.
- 13.5 Interest. Any monetary payment due Lessor hereunder, other than late charges, not received by Lessor, when due shall bear interest from the 31st day after it was due. The interest ("Interest") charged shall be computed at the rate of 10% per annum but shall not exceed the maximum rate allowed by law. Interest is payable in addition to the potential late charge provided for in Paragraph 13.4.

13.6 Breach by Lessor.

- (a) Notice of Breach. Lessor shall not be deemed in breach of this Lease unless Lessor fails within a reasonable time to perform an obligation required to be performed by Lessor. For purposes of this Paragraph, a reasonable time shall in no event be less than 30 days after receipt by Lessor, and any Lender whose name and address shall have been furnished to Lessee in writing for such purpose, of written notice specifying wherein such obligation of Lessor has not been performed; provided, however, that if the nature of Lessor's obligation is such that more than 30 days are reasonably required for its performance, then Lessor shall not be in breach if performance is commenced within such 30 day period and thereafter diligently pursued to completion.
- (b) Performance by Lessee on Behalf of Lessor. In the event that neither Lessor nor Lender cures said breach within 30 days after receipt of said notice, or if having commenced said cure they do not diligently pursue it to completion, then Lessee may elect to cure said breach at Lessee's expense and offset from Rent the actual and reasonable cost to perform such cure, provided, however, that such offset shall not exceed an amount equal to the greater of one month's Base Rent or the Security Deposit, reserving Lessee's right to seek reimbursement from Lessor for any such expense in excess of such offset. Lessee shall document the cost of said cure and supply said documentation to Lessor.
- 14. Condemnation. If the Premises or any portion thereof are taken under the power of eminent domain or sold under the threat of the exercise of said power (collectively "Condemnation"), this Lease shall terminate as to the part taken as of the date the condemning authority takes title or possession, whichever first occurs. If more than 10% of the rentable floor area of the Premises, or more than 25% of Lessee's Reserved Parking Spaces, if any, are taken by Condemnation, Lessee may, at Lessee's option, to be exercised in writing within 10 days after Lessor shall have given Lessee written notice of such taking (or in the absence of such notice, within 10 days after the condemning authority shall have taken possession) terminate this Lease as of the date the condemning authority takes such possession. If Lessee does not terminate this Lease in accordance with the foregoing, this Lease shall remain in full force and effect as to the portion of the Premises remaining, except that the Base Rent shall be reduced in proportion to the reduction in utility of the Premises caused by such Condemnation. Condemnation awards and/or payments shall be the property of Lessor, whether such award shall be made as compensation for diminution in value of the leasehold, the value of the part taken, or for severance damages; provided, however, that Lessee shall be entitled to any compensation paid by the condemnor for Lessee's relocation expenses, loss of business goodwill and/or Trade Fixtures, without regard to whether or not this Lease is terminated pursuant to the provisions of this Paragraph. All Alterations and Utility Installations made to the Premises by Lessee, for purposes of Condemnation only, shall be considered the property of the Lessee and Lessee shall be entitled to any and all compensation which is payable therefor. In the event that this Lease is not terminated by reason of the Condemnation, Lessor shall repair any damage to the Premises caused by such Condemnation.

15. Brokerage Fees.

- 15.1 Additional Commission. In addition to the payments owed pursuant to Paragraph 1.10 above, Lessor agrees that: (a) if Lessee exercises any Option, (b) if Lessee or anyone affiliated with Lessee acquires from Lessor any rights to the Premises or other premises owned by Lessor and located within the Project, (c) if Lessee remains in possession of the Premises, with the consent of Lessor, after the expiration of this Lease, or (d) if Base Rent is increased, whether by agreement or operation of an escalation clause herein, then, Lessor shall pay Brokers a fee in accordance with the fee schedule of the Brokers in effect at the time the Lease was executed. The provisions of this paragraph are intended to supersede the provisions of any earlier agreement to the contrary.
- 15.2 **Assumption of Obligations**. Any buyer or transferee of Lessor's interest in this Lease shall be deemed to have assumed Lessor's obligation hereunder. Brokers shall be third party beneficiaries of the provisions of Paragraphs 1.10, 15, 22 and 31. If Lessor fails to pay to Brokers any amounts due as and for brokerage fees pertaining to this Lease when due, then such amounts shall accrue Interest. In addition, if Lessor fails to pay any amounts to Lessee's Broker when due, Lessee's Broker may send written notice to Lessor and Lessee of such failure and if Lessor fails to pay such amounts within 10 days after said notice, Lessee shall pay said monies to its Broker and offset such amounts against Rent. In addition, Lessee's Broker shall be deemed to be a third party beneficiary of any commission agreement entered into by and/or between Lessor and Lessor's Broker for the limited purpose of collecting any brokerage fee owed.
- 15.3 Representations and Indemnities of Broker Relationships. Lessee and Lessor each represent and warrant to the other that it has had no dealings with any person, firm, broker, agent or finder (other than the Brokers and Agents, if any) in connection with this Lease, and that no one other than said named Brokers and Agents is entitled to any commission or finder's fee in connection herewith. Lessee and Lessor do each hereby agree to indemnify, protect, defend and hold the other harmless from and against liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any dealings or actions of the indemnifying Party, including any costs, expenses, attorneys' fees reasonably incurred with respect thereto.

16. Estoppel Certificates.

- (a) Each Party (as "Responding Party") shall within 10 days after written notice from the other Party (the "Requesting Party") execute, acknowledge and deliver to the Requesting Party a statement in writing in form similar to the then most current "Estoppel Certificate" form published by AIR CRE, plus such additional information, confirmation and/or statements as may be reasonably requested by the Requesting Party.
- (b) If the Responding Party shall fail to execute or deliver the Estoppel Certificate within such 10 day period, the Requesting Party may execute an Estoppel Certificate stating that: (i) the Lease is in full force and effect without modification except as may be represented by the Requesting Party, (ii) there are no uncured defaults in the Requesting Party's performance, and (iii) if Lessor is the Requesting Party, not more than one month's rent has been paid in advance. Prospective purchasers and encumbrancers may rely upon the Requesting Party's Estoppel Certificate, and the Responding Party shall be estopped from denying the truth of the facts contained in said Certificate. In addition, Lessee acknowledges that any failure on its part to provide such an Estoppel Certificate will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to execute and/or deliver a requested Estoppel Certificate in a timely fashion the monthly Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for remainder of the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to

	INITIAL

INITIALS

provide the Estoppel Certificate. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to provide the Estoppel Certificate nor prevent the exercise of any of the other rights and remedies granted hereunder.

- (c) If Lessor desires to finance, refinance, or sell the Premises, or any part thereof, Lessee and all Guarantors shall within 10 days after written notice from Lessor deliver to any potential lender or purchaser designated by Lessor such financial statements as may be reasonably required by such lender or purchaser, including but not limited to Lessee's financial statements for the past 3 years. All such financial statements shall be received by Lessor and such lender or purchaser in confidence and shall be used only for the purposes herein set forth.
- 17. Definition of Lessor. The term "Lessor" as used herein shall mean the owner or owners at the time in question of the fee title to the Premises, or, if this is a sublease, of the Lessee's interest in the prior lease. In the event of a transfer of Lessor's title or interest in the Premises or this Lease, Lessor shall deliver to the transferee or assignee (in cash or by credit) any unused Security Deposit held by Lessor. Upon such transfer or assignment and delivery of the Security Deposit, as aforesaid, the prior Lessor shall be relieved of all liability with respect to the obligations and/or covenants under this Lease thereafter to be performed by the Lessor. Subject to the foregoing, the obligations and/or covenants in this Lease to be performed by the Lessor shall be binding only upon the Lessor as hereinabove defined.
- 18. Severability. The invalidity of any provision of this Lease, as determined by a court of competent jurisdiction, shall in no way affect the validity of any other provision hereof.
- 19. Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Lease shall mean and refer to calendar days.
- 20. Limitation on Liability. The obligations of Lessor under this Lease shall not constitute personal obligations of Lessor, or its partners, members, directors, officers or shareholders, and Lessee shall look to the Project, and to no other assets of Lessor, for the satisfaction of any liability of Lessor with respect to this Lease, and shall not seek recourse against Lessor's partners, members, directors, officers or shareholders, or any of their personal assets for such satisfaction.
- 21. Time of Essence. Time is of the essence with respect to the performance of all obligations to be performed or observed by the Parties under this Lease.
- 22. No Prior or Other Agreements; Broker Disclaimer. This Lease contains all agreements between the Parties with respect to any matter mentioned herein, and no other prior or contemporaneous agreement or understanding shall be effective. Lessor and Lessee each represents and warrants to the Brokers that it has made, and is relying solely upon, its own investigation as to the nature, quality, character and financial responsibility of the other Party to this Lease and as to the use, nature, quality and character of the Premises. Brokers have no responsibility with respect thereto or with respect to any default or breach hereof by either Party.

23. Notices.

- 23.1 **Notice Requirements**. All notices required or permitted by this Lease or applicable law shall be in writing and may be delivered in person (by hand or by courier) or may be sent by regular, certified or registered mail or U.S. Postal Service Express Mail, with postage prepaid, or by facsimile transmission, or by email, and shall be deemed sufficiently given if served in a manner specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specify a different address for notice, except that upon Lessee's taking possession of the Premises, the Premises shall constitute Lessee's address for notice. A copy of all notices to Lessor shall be concurrently transmitted to such party or parties at such addresses as Lessor may from time to time hereafter designate in writing.
- 23.2 Date of Notice. Any notice sent by registered or certified mail, return receipt requested, shall be deemed given on the date of delivery shown on the receipt card, or if no delivery date is shown, the postmark thereon. If sent by regular mail the notice shall be deemed given 72 hours after the same is addressed as required herein and mailed with postage prepaid. Notices delivered by United States Express Mail or overnight courier that guarantees next day delivery shall be deemed given 24 hours after delivery of the same to the Postal Service or courier. Notices delivered by hand, or transmitted by facsimile transmission or by email shall be deemed delivered upon actual receipt. If notice is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.
- 23.3 **Options.** Notwithstanding the foregoing, in order to exercise any Options (see paragraph 39), the Notice must be sent by Certified Mail (return receipt requested), Express Mail (signature required), courier (signature required) or some other methodology that provides a receipt establishing the date the notice was received by the Lessor.

24. Waivers

- (a) No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any other term, covenant or condition hereof. Lessor's consent to, or approval of, any act shall not be deemed to render unnecessary the obtaining of Lessor's consent to, or approval of, any subsequent or similar act by Lessee, or be construed as the basis of an estoppel to enforce the provision or provisions of this Lease requiring such consent.
- (b) The acceptance of Rent by Lessor shall not be a waiver of any Default or Breach by Lessee. Any payment by Lessee may be accepted by Lessor on account of monies or damages due Lessor, notwithstanding any qualifying statements or conditions made by Lessee in connection therewith, which such statements and/or conditions shall be of no force or effect whatsoever unless specifically agreed to in writing by Lessor at or before the time of deposit of such payment.
- (c) THE PARTIES AGREE THAT THE TERMS OF THIS LEASE SHALL GOVERN WITH REGARD TO ALL MATTERS RELATED THERETO AND HEREBY WAIVE THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE TO THE EXTENT THAT SUCH STATUTE IS INCONSISTENT WITH THIS LEASE.

25. Disclosures Regarding The Nature of a Real Estate Agency Relationship.

- (a) When entering into a discussion with a real estate agent regarding a real estate transaction, a Lessor or Lessee should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Lessor and Lessee acknowledge being advised by the Brokers in this transaction, as follows:
- (i) <u>Lessor's Agent</u>. A Lessor's agent under a listing agreement with the Lessor acts as the agent for the Lessor only. A Lessor's agent or subagent has the following affirmative obligations: <u>To the Lessor</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessor. <u>To the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (ii) <u>Lessee's Agent</u>. An agent can agree to act as agent for the Lessee only. In these situations, the agent is not the Lessor's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Lessor. An agent acting only for a Lessee has the following affirmative obligations. <u>To the Lessee</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessee. <u>To the Lessee and the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (iii) Agent Representing Both Lessor and Lessee. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Lessor and the Lessee in a transaction, but only with the knowledge and consent of both the Lessor and the Lessee. In a dual agency situation, the agent has the following affirmative obligations to both the Lessor and the Lessee: (a) A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Lessor or the Lessee. (b) Other duties to the Lessor and the Lessee as stated above in subparagraphs (i) or (ii). In representing both Lessor and Lessee, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating to either Lessee's or Lessor's financial position, motivations, bargaining position, or other personal information that may impact rent, including Lessor's willingness to accept a rent less than the listing rent or Lessee's willingness to pay rent greater than the rent offered. The above duties of the agent in a real estate transaction do not relieve a Lessor or Lessee from the responsibility to protect their own interests. Lessor and Lessee should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Both Lessor and Lessee should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.
- (b) Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this Lease may be brought against Broker more than one year after the Start Date and that the liability (including court costs and attorneys' fees), of any Broker with respect to any such lawsuit and/or legal proceeding shall not exceed the fee received by such Broker pursuant to this Lease; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.
- (c) Lessor and Lessee agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- 26. No Right To Holdover. Lessee has no right to retain possession of the Premises or any part thereof beyond the expiration or termination of this Lease. At or

|--|

prior to the expiration or termination of this Lease Lessee shall deliver exclusive possession of the Premises to Lessor. For purposes of this provision and Paragraph 13.1(a), exclusive possession shall mean that Lessee shall have vacated the Premises, removed all of its personal property therefrom and that the Premises have been returned in the condition specified in this Lease. In the event that Lessee does not deliver exclusive possession to Lessor as specified above, then Lessor's damages during any holdover period shall be computed at the amount of the Rent (as defined in Paragraph 4.1) due during the last full month before the expiration or termination of this Lease (disregarding any temporary abatement of Rent that may have been in effect), but with Base Rent being 150% of the Base Rent payable during such last full month. Nothing contained herein shall be construed as consent by Lessor to any holding over by Lessee.

- 27. Cumulative Remedies. No remedy or election hereunder shall be deemed exclusive but shall, wherever possible, be cumulative with all other remedies at law or in equity.
- Covenants and Conditions; Construction of Agreement. All provisions of this Lease to be observed or performed by Lessee are both covenants and conditions. In construing this Lease, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Lease. Whenever required by the context, the singular shall include the plural and vice versa. This Lease shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.
- 29. Binding Effect; Choice of Law. This Lease shall be binding upon the Parties, their personal representatives, successors and assigns and be governed by the laws of the State in which the Premises are located. Any litigation between the Parties hereto concerning this Lease shall be initiated in the county in which the Premises are located. Signatures to this Lease accomplished by means of electronic signature or similar technology shall be legal and binding.

Subordination; Attornment; Non-Disturbance.

- 30.1 Subordination. This Lease and any Option granted hereby shall be subject and subordinate to any ground lease, mortgage, deed of trust, or other hypothecation or security device (collectively, "Security Device"), now or hereafter placed upon the Premises, to any and all advances made on the security thereof, and to all renewals, modifications, and extensions thereof. Lessee agrees that the holders of any such Security Devices (in this Lease together referred to as "Lender") shall have no liability or obligation to perform any of the obligations of Lessor under this Lease. Any Lender may elect to have this Lease and/or any Option granted hereby superior to the lien of its Security Device by giving written notice thereof to Lessee, whereupon this Lease and such Options shall be deemed prior to such Security Device, notwithstanding the relative dates of the documentation or recordation thereof.
- 30.2 Attornment. In the event that Lessor transfers title to the Premises, or the Premises are acquired by another upon the foreclosure or termination of a Security Device to which this Lease is subordinated (i) Lessee shall, subject to the non-disturbance provisions of Paragraph 30.3, attorn to such new owner, and upon request, enter into a new lease, containing all of the terms and provisions of this Lease, with such new owner for the remainder of the term hereof, or, at the election of the new owner, this Lease will automatically become a new lease between Lessee and such new owner, and (ii) Lessor shall thereafter be relieved of any further obligations hereunder and such new owner shall assume all of Lessor's obligations, except that such new owner shall not: (a) be liable for any act or omission of any prior lessor or with respect to events occurring prior to acquisition of ownership; (b) be subject to any offsets or defenses which Lessee might have against any prior lessor, (c) be bound by prepayment of more than one month's rent, or (d) be liable for the return of any security deposit paid to any prior lessor.
- 30.3 Non-Disturbance. With respect to Security Devices entered into by Lessor after the execution of this Lease, Lessee's subordination of this Lease shall be subject to receiving a commercially reasonable non-disturbance agreement (a "Non-Disturbance Agreement") from the Lender which Non-Disturbance Agreement provides that Lessee's possession of the Premises, and this Lease, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises. Further, within 60 days after the execution of this Lease, Lessor shall, if requested by Lessee, use its commercially reasonable efforts to obtain a Non-Disturbance Agreement from the holder of any pre-existing Security Device which is secured by the Premises. In the event that Lessor is unable to provide the Non-Disturbance Agreement within said 60 days, then Lessee may, at Lessee's option, directly contact Lender and attempt to negotiate for the execution and delivery of a Non-Disturbance Agreement.
- 30.4 Self-Executing. The agreements contained in this Paragraph 30 shall be effective without the execution of any further documents; provided, however, that, upon written request from Lessor or a Lender in connection with a sale, financing of refinancing of the Premises, Lessee and Lessor shall execute such further writings as may be reasonably required to separately document any subordination, attornment and/or Non-Disturbance Agreement provided for herein.
- 31. Attorneys' Fees. If any Party or Broker brings an action or proceeding involving the Premises whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred. In addition, Lessor shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is subsequently commenced in connection with such Default or resulting Breach (\$200 is a reasonable minimum per occurrence for such services and consultation).
- 32. Lessor's Access; Showing Premises; Repairs. Lessor and Lessor's agents shall have the right to enter the Premises at any time, in the case of an emergency, and otherwise at reasonable times after reasonable prior notice for the purpose of showing the same to prospective purchasers, lenders, or tenants, and making such alterations, repairs, improvements or additions to the Premises as Lessor may deem necessary or desirable and the erecting, using and maintaining of utilities, services, pipes and conduits through the Premises and/or other premises as long as there is no material adverse effect to Lessee's use of the Premises. All such activities shall be without abatement of rent or liability to Lessee. In addition, Lessor shall have the right to retain keys to the Premises and to unlock all doors in or upon the Premises other than to files, vaults and safes, and in the case of emergency to enter the Premises by any reasonably appropriate means, and any such entry shall not be deemed a forcible or unlawful entry or detainer of the Premises or an eviction. Lessee waives any charges for damages or injuries or interference with Lessee's property or business in connection therewith.
- 33. Auctions. Lessee shall not conduct, nor permit to be conducted, any auction upon the Premises without Lessor's prior written consent. Lessor shall not be obligated to exercise any standard of reasonableness in determining whether to permit an auction.
- Signs. Lessor may place on the Premises ordinary "For Sale" signs at any time and ordinary "For Lease" signs during the last 6 months of the term hereof. Lessor may not place any sign on the exterior of the Building that covers any of the windows of the Premises. Except for ordinary "For Sublease" signs which may be placed only on the Premises, Lessee shall not place any sign upon the Project without Lessor's prior written consent. All signs must comply with all Applicable Requirements.
- 35. Termination; Merger. Unless specifically stated otherwise in writing by Lessor, the voluntary or other surrender of this Lease by Lessee, the mutual termination or cancellation hereof, or a termination hereof by Lessor for Breach by Lessee, shall automatically terminate any sublease or lesser estate in the Premises; provided, however, that Lessor may elect to continue any one or all existing subtenancies. Lessor's failure within 10 days following any such event to elect to the contrary by written notice to the holder of any such lesser interest, shall constitute Lessor's election to have such event constitute the termination of such interest.
- 36. Consents. All requests for consent shall be in writing. Except as otherwise provided herein, wherever in this Lease the consent of a Party is required to an act by or for the other Party, such consent shall not be unreasonably withheld or delayed. Lessor's actual reasonable costs and expenses (including but not limited to architects', attorneys', engineers' and other consultants' fees) incurred in the consideration of, or response to, a request by Lessee for any Lessor consent, including but not limited to consents to an assignment, a subletting or the presence or use of a Hazardous Substance, shall be paid by Lessee upon receipt of an invoice and supporting documentation therefor. Lessor's consent to any act, assignment or subletting shall not constitute an acknowledgment that no Default or Breach by Lessee of this Lease exists, nor shall such consent be deemed a waiver of any then existing Default or Breach, except as may be otherwise specifically stated in writing by Lessor at the time of such consent. The failure to specify herein any particular condition to Lessor's consent shall not preclude the imposition by Lessor at the time of consent of such further or other conditions as are then reasonable with reference to the particular matter for which consent is being given. In the event that either Party disagrees with any determination made by the other hereunder and reasonably requests the reasons for such determination, the determining party shall furnish its reasons in writing and in reasonable detail within 10 business days following such request.

- 37.1 Execution. The Guarantors, if any, shall each execute a guaranty in the form most recently published by AIR CRE.
- 37.2 Default. It shall constitute a Default of the Lessee if any Guarantor fails or refuses, upon request to provide: (a) evidence of the execution of the guaranty, including the authority of the party signing on Guarantor's behalf to obligate Guarantor, and in the case of a corporate Guarantor, a certified copy of a resolution of its board of directors authorizing the making of such guaranty, (b) current financial statements, (c) an Estoppel Certificate, or (d) written confirmation that the guaranty is
- 38. Quiet Possession. Subject to payment by Lessee of the Rent and performance of all of the covenants, conditions and provisions on Lessee's part to be observed and performed under this Lease, Lessee shall have quiet possession and quiet enjoyment of the Premises during the term hereof.

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

- **Options.** If Lessee is granted any option, as defined below, then the following provisions shall apply. 39.
- 39.1 Definition. "Option" shall mean: (a) the right to extend or reduce the term of or renew this Lease or to extend or reduce the term of or renew any lease that Lessee has on other property of Lessor; (b) the right of first refusal or first offer to lease either the Premises or other property of Lessor; (c) the right to purchase, the right of first offer to purchase or the right of first refusal to purchase the Premises or other property of Lessor.
- 39.2 Options Personal To Original Lessee. Any Option granted to Lessee in this Lease is personal to the original Lessee, and cannot be assigned or exercised by anyone other than said original Lessee and only while the original Lessee is in full possession of the Premises and, if requested by Lessor, with Lessee certifying that Lessee has no intention of thereafter assigning or subletting.
- 39.3 Multiple Options. In the event that Lessee has any multiple Options to extend or renew this Lease, a later Option cannot be exercised unless the prior Options have been validly exercised.

39.4 Effect of Default on Options.

- (a) Lessee shall have no right to exercise an Option: (i) during the period commencing with the giving of any notice of Default and continuing until said Default is cured, (ii) during the period of time any Rent is unpaid (without regard to whether notice thereof is given Lessee), (iii) during the time Lessee is in Breach of this Lease, or (iv) in the event that Lessee has been given 3 or more notices of separate Default, whether or not the Defaults are cured, during the 12 month period immediately preceding the exercise of the Option.
- (b) The period of time within which an Option may be exercised shall not be extended or enlarged by reason of Lessee's inability to exercise an Option because of the provisions of Paragraph 39.4(a).
- (c) An Option shall terminate and be of no further force or effect, notwithstanding Lessee's due and timely exercise of the Option, if, after such exercise and prior to the commencement of the extended term or completion of the purchase, (i) Lessee fails to pay Rent for a period of 30 days after such Rent becomes due (without any necessity of Lessor to give notice thereof), or (ii) if Lessee commits a Breach of this Lease.
- 40. Security Measures. Lessee hereby acknowledges that the Rent payable to Lessor hereunder does not include the cost of guard service or other security measures, and that Lessor shall have no obligation whatsoever to provide same. Lessee assumes all responsibility for the protection of the Premises, Lessee, its agents and invitees and their property from the acts of third parties. In the event, however, that Lessor should elect to provide security services, then the cost thereof shall be an Operating Expense.

41. Reservations.

- (a) Lessor reserves the right: (i) to grant, without the consent or joinder of Lessee, such easements, rights and dedications that Lessor deems necessary, (ii) to cause the recordation of parcel maps and restrictions, (iii) to create and/or install new utility raceways, so long as such easements, rights, dedications, maps, restrictions, and utility raceways do not unreasonably interfere with the use of the Premises by Lessee. Lessor may also: change the name, address or title of the Building or Project upon at least 90 days prior written notice; provide and install, at Lessee's expense, Building standard graphics on the door of the Premises and such portions of the Common Areas as Lessor shall reasonably deem appropriate; grant to any lessee the exclusive right to conduct any business as long as such exclusive right does not conflict with any rights expressly given herein; and to place such signs, notices or displays as Lessor reasonably deems necessary or advisable upon the roof, exterior of the Building or the Project or on pole signs in the Common Areas. Lessee agrees to sign any documents reasonably requested by Lessor to effectuate such rights. The obstruction of Lessee's view, air, or light by any structure erected in the vicinity of the Building, whether by Lessor or third parties, shall in no way affect this Lease or impose any liability upon Lessor.
- (b) Lessor also reserves the right to move Lessee to other space of comparable size in the Building or Project. Lessor must provide at least 45 days prior written notice of such move, and the new space must contain improvements of comparable quality to those contained within the Premises. Lessor shall pay the reasonable out of pocket costs that Lessee incurs with regard to such relocation, including the expenses of moving and necessary stationary revision costs. In no event, however, shall Lessor be required to pay an amount in excess of two months Base Rent. Lessee may not be relocated more than once during the term of this Lease.
- (c) Lessee shall not: (i) use a representation (photographic or otherwise) of the Building or Project or their name(s) in connection with Lessee's business; or (ii) suffer or permit anyone, except in emergency, to go upon the roof of the Building.
- 42. Performance Under Protest. If at any time a dispute shall arise as to any amount or sum of money to be paid by one Party to the other under the provisions hereof, the Party against whom the obligation to pay the money is asserted shall have the right to make payment "under protest" and such payment shall not be regarded as a voluntary payment and there shall survive the right on the part of said Party to institute suit for recovery of such sum. If it shall be adjudged that there was no legal obligation on the part of said Party to pay such sum or any part thereof, said Party shall be entitled to recover such sum or so much thereof as it was not legally required to pay. A Party who does not initiate suit for the recovery of sums paid "under protest" within 6 months shall be deemed to have waived its right to protest such payment.

43. Authority; Multiple Parties; Execution.

- (a) If either Party hereto is a corporation, trust, limited liability company, partnership, or similar entity, each individual executing this Lease on behalf of such entity represents and warrants that he or she is duly authorized to execute and deliver this Lease on its behalf. Each Party shall, within 30 days after request, deliver to the other Party satisfactory evidence of such authority.
- (b) If this Lease is executed by more than one person or entity as "Lessee", each such person or entity shall be jointly and severally liable hereunder. It is agreed that any one of the named Lessees shall be empowered to execute any amendment to this Lease, or other document ancillary thereto and bind all of the named Lessees, and Lessor may rely on the same as if all of the named Lessees had executed such document.

 (c) This Lease may be executed by the Parties in counterparts, each of which shall be deemed an original and all of which together shall constitute one
- and the same instrument.
- 44. Conflict. Any conflict between the printed provisions of this Lease and the typewritten or handwritten provisions shall be controlled by the typewritten or
- **45. Offer**. Preparation of this Lease by either party or their agent and submission of same to the other Party shall not be deemed an offer to lease to the other Party. This Lease is not intended to be binding until executed and delivered by all Parties hereto.
- Amendments. This lease may be modified only in writing, signed by the Parties in interest at the time of the modification. As long as they do not materially change Lessee's obligations hereunder, Lessee agrees to make such reasonable non-monetary modifications to this Lease as may be reasonably required by a Lender in connection with the obtaining of normal financing or refinancing of the Premises.
- 47. Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY OR ARISING OUT OF THIS AGREEMENT.

48. Arbitration of Disputes. An Addendum requiring the Arbitration of all disputes between the Parties and/or Brokers arising out of this Lease is not attached to this Lease.
49. Accessibility; Americans with Disabilities Act. (a) The Premises:
have not undergone an inspection by a Certified Access Specialist (CASp). Note: A Certified Access Specialist (CASp) can inspect the subject premises and determine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not require a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection the subject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the arrangements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to correct violations of construction-related accessibility standards within the premises.
have undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises met all applicable construction-related accessibility standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this Lease and agrees to keep such report confidential.
have undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises did not meet all applicable construction-related accessibility standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this Lease and agrees to keep such report confidential except as necessary to complete repairs and corrections of violations of construction related accessibility standards.

INITIALS

In the event that the Premises have been issued an inspection report by a CASp the Lessor shall provide a copy of the disability access inspection certificate to Lessee within 7 days of the execution of this Lease.

Since compliance with the Americans with Disabilities Act (ADA) and other state and local accessibility statutes are dependent upon Lessee's specific use of the Premises, Lessor makes no warranty or representation as to whether or not the Premises comply with ADA or any similar legislation. In the event that Lessee's use of the Premises requires modifications or additions to the Premises in order to be in compliance with ADA or other accessibility statutes, Lessee agrees to make any such necessary modifications and/or additions at Lessee's expense.

LESSOR AND LESSEE HAVE CAREFULLY READ AND REVIEWED THIS LEASE AND EACH TERM AND PROVISION CONTAINED HEREIN, AND BY THE EXECUTION OF THIS LEASE SHOW THEIR INFORMED AND VOLUNTARY CONSENT THERETO. THE PARTIES HEREBY AGREE THAT, AT THE TIME THIS LEASE IS EXECUTED, THE TERMS OF THIS LEASE ARE COMMERCIALLY REASONABLE AND EFFECTUATE THE INTENT AND PURPOSE OF LESSOR AND LESSEE WITH RESPECT TO THE PREMISES.

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS LEASE OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

1. SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS LEASE.

- RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PREMISES. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING AND SIZE OF THE PREMISES, THE STRUCTURAL INTEGRITY, THE CONDITION OF THE ROOF AND OPERATING SYSTEMS, COMPLIANCE WITH THE AMERICANS WITH DISABILITIES ACT AND THE SUITABILITY OF THE PREMISES FOR LESSEE'S

WARNING: IF THE PREMISES ARE LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THE LEASE MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PREMISES ARE LOCATED.

The parties hereto have executed this Lease at the place and on the dates specified	above their respective signatures.
Executed at:	Executed at:
On:	On:
By LESSOR:	By LESSEE:
by LESSOR.	by lessee.
By:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
	~
Ву:	Ву:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Address:	Address:
Federal ID No.:	Federal ID No.:
() \	
BROKER	BROKER
Attn:	Attn:
Title:	Title:
Address:	Address:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Federal ID No.:	Federal ID No.:
Broker DRE License #:	Broker DRE License #:
Agent DRE License #:	Agent DRE License #:

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS © 2019 AIR CRE. All Rights Reserved. MTON-20.20, Revised 10-22-2020



STANDARD MULTI-TENANT OFFICE LEASE - GROSS

1. Basic Provisions ("Basic Provisions").		
1.1 Parties. This Lease ("Lease"), dated for reference purp	ooses only, is made by and	between ("Lessor") and
("Lessee"), (collectively the "Parties", or individually a "Party"). 1.2(a) Premises: That certain Portion of the Project (as	defined below) commonly known as (str	eet address, suite, city, state): ("Premises").
The Premises are located in the County of, and cor		taran da antara da a
square feet. In addition to Lessee's rights to use and occupy the		
defined in Paragraph 2.7 below) as hereinafter specified, but shal	I not have any rights to the roof, the exter	ior walls, the area above the dropped ceilings, or the utility
raceways of the building containing the Premises ("Building") or t		
which they are located, along with all other buildings and improve	·	ferred to as the " Project ." The Project consists of
approximately rentable square feet. (See also Para 1.2(b) Parking: unreserved and	= : :	a monthly cost of per unreserved space and
per reserved space. (See Paragraph 2.6)	Teserved verifice parking spaces at	a monthly cost of per unreserved space and
1.3 Term: years and months	("Original Term") commencing	("Commencement Date") and ending
("Expiration Date"). (See also Paragraph 3)		
1.4 Early Possession: If the Premises are available Lessee	may have non-exclusive possession of the	Premises commencing ("Early Possession
Date"). (See also Paragraphs 3.2 and 3.3) 1.5 Base Rent: per month ("Base Rent"), p	avable on the day of each r	nonth commencing (See also Paragraph 4)
If this box is checked, there are provisions in this Lease 1.6 Lessee's Share of Operating Expense Increase.		
and/or the Project are modified during the term of this Lease, Le		
1.7 Base Rent and Other Monies Paid Upon Execution:	soli sitali recalculate Lessee's sitale to reli	ect such modification.
(a) Base Rent: for the period		
(b) Security Deposit: ("Security Depo	osit"). (See also Paragraph 5)	
(c) Parking: for the period	·	
(d) Other: for		
(e) Total Due Upon Execution of this Lease:	·	
1.8 Agreed Use: (See also Paragraph 6)		
1.9 Base Year; Insuring Party. The Base Year is	Lessor is the " Insuring Party ". (See a	lso Paragraphs 4.2 and 8)
1.10 Real Estate Brokers. (See also Paragraphs 15 and 25)(a) Representation: Each Party acknowledges received	ving a Disclosure Regarding Real Estate Ag	ency Relationship, confirms and consents to the following
agency relationships in this Lease with the following real estate b		
Lessor's Brokerage Firm License No	Is the broker of (check one):	the Lessor; or both the Lessee and Lessor (dual agent).
Lessor's Agent License No is (check one): the Lessor's Agent (sale:	sperson or broker associate); or both the Lessee's
Agent and the Lessor's Agent (dual agent).	January, Jan	
Lessee's Brokerage Firm License No	Is the broker of (check one):	the Lessee: or both the Lessee and Lessor (dual
agent).		
Lessee's Agent License No is	(check one): the Lessee's Agent (sale	esperson or broker associate); or both the Lessee's
Agent and the Lessor's Agent (dual agent).	()	
(b) Payment to Brokers. Upon execution and delive		
separate written agreement (or if there is no such agreement, the	e sum of or%	of the total Base Rent) for the brokerage services rendered
by the Brokers.		(0 1 1 1 1 1 1 1
1.11 Guarantor. The obligations of the Lessee under this Le		
1.12 Business Hours for the Building: a.m. t		ridays (except Building Holidays) and a.m. to servation of New Year's Day, President's Day, Memorial Day,
Independence Day, Labor Day, Thanksgiving Day, Christmas Day, a		servation of New Year's Day, Trestactics Day, Wellional Day,
1.13 Lessor Supplied Services. Notwithstanding the provise		gated to provide the following within the Premises:
Janitorial services		
Electricity		
Other (specify):1.14 Attachments. Attached hereto are the following, all o	f which constitute a part of this Lease:	
an Addendum consisting of Paragraphs	•	
	, , , , , , , , , , , , , , , , , , ,	
a plot plan depicting the Premises;		
a current set of the Rules and Regulations;		
a Work Letter;		
a janitorial schedule;		
other (specify):		
\		
 Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee he 	reby leases from Lessor. the Premises. for	the term, at the rental, and upon all of the terms.
covenants and conditions set forth in this Lease. While the appro		
purposes of comparison, the Base Rent stated herein is NOT tied	-	justment should the actual size be determined to be
different. NOTE: Lessee is advised to verify the actual size prior	_	ent Date or the Early Possession Date, whichever first occurs
("Start Date"), and warrants that the existing electrical, plumbing		
which the Lessor is obligated to construct pursuant to the Work L		
condition on said date, that the structural elements of the roof, b	=	
contain hazardous levels of any mold or fungi defined as toxic un Lessor is unaware of (i) any recorded Notices of Default affecting		
bankruptcy proceeding affecting the Premises.	and remise, (ii) any demiquent amounts	and ander any loan secured by the Fremises, dilu (III) dily
2.3 Compliance. Lessor warrants that to the best of its kn		
applicable laws, covenants or restrictions of record, regulations, a	,	· · · · · · · · · · · · · · · · · · ·
or portion thereof, was constructed. Said warranty does not apple	y to the use to which Lessee will put the F	remises, modifications which may be required by the
INITIALS	INITIALS	
© 2019 AIR CRE. All Rights Reserved.		Last Edited: 9/30/2022 2:33 PM

Americans with Disabilities Act or any similar laws as a result of Lessee's use (see Paragraph 49), or to any Alterations or Utility Installations (as defined in Paragraph 7.3(a)) made or to be made by Lessee. NOTE: Lessee is responsible for determining whether or not the zoning and other Applicable Requirements are appropriate for Lessee's intended use, and acknowledges that past uses of the Premises may no longer be allowed. If the Premises do not comply with said warranty, Lesson shall, except as otherwise provided, promptly after receipt of written notice from Lessee setting forth with specificity the nature and extent of such non-compliance, rectify the same. If the Applicable Requirements are hereafter changed so as to require during the term of this Lease the construction of an addition to or an alteration of the Premises, the remediation of any Hazardous Substance, or the reinforcement or other physical modification of the Premises ("Capital Expenditure"), Lessor and Lessee shall allocate the cost of such work as follows:

- (a) Subject to Paragraph 2.3(c) below, if such Capital Expenditures are required as a result of the specific and unique use of the Premises by Lessee as compared with uses by tenants in general, Lessee shall be fully responsible for the cost thereof, provided, however, that if such Capital Expenditure is required during the last 2 years of this Lease and the cost thereof exceeds 6 months' Base Rent, Lessee may instead terminate this Lease unless Lessor notifies Lessee, in writing, within 10 days after receipt of Lessee's termination notice that Lessor has elected to pay the difference between the actual cost thereof and the amount equal to 6 months' Base Rent. If Lessee elects termination, Lessee shall immediately cease the use of the Premises which requires such Capital Expenditure and deliver to Lessor written notice specifying a termination date at least 90 days thereafter. Such termination date shall, however, in no event be earlier than the last day that Lessee could legally utilize the Premises without commencing such Capital Expenditure.
- (b) If such Capital Expenditure is not the result of the specific and unique use of the Premises by Lessee (such as, governmentally mandated seismic modifications), then Lessor shall pay for such Capital Expenditure and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date that on which the Base Rent is due, an amount equal to 1/144th of the portion of such costs reasonably attributable to the Premises. Lessee shall pay Interest on the balance but may prepay its obligation at any time. If, however, such Capital Expenditure is required during the last 2 years of this Lease or if Lessor reasonably determines that it is not economically feasible to pay its share thereof, Lessor shall have the option to terminate this Lease upon 90 days prior written notice to Lessee unless Lessee notifies Lessor, in writing, within 10 days after receipt of Lessor's termination notice that Lessee will pay for such Capital Expenditure. If Lessor does not elect to terminate, and fails to tender its share of any such Capital Expenditure, Lessee may advance such funds and deduct same, with Interest, from Rent until Lessor's share of such costs have been fully paid. If Lessee is unable to finance Lessor's share, or if the balance of the Rent due and payable for the remainder of this Lease is not sufficient to fully reimburse Lessee on an offset basis, Lessee shall have the right to terminate this Lease upon 30
- Notwithstanding the above, the provisions concerning Capital Expenditures are intended to apply only to non-voluntary, unexpected, and new Applicable Requirements. If the Capital Expenditures are instead triggered by Lessee as a result of an actual or proposed change in use, change in intensity of use, or modification to the Premises then, and in that event, Lessee shall either: (i) immediately cease such changed use or intensity of use and/or take such other steps as may be necessary to eliminate the requirement for such Capital Expenditure, or (ii) complete such Capital Expenditure at its own expense. Lessee shall not have any right to terminate this Lease.
- 2.4 **Acknowledgements**. Lessee acknowledges that: (a) it has been given an opportunity to inspect and measure the Premises, (b) Lessee has been advised by Lessor and/or Brokers to satisfy itself with respect to the size and condition of the Premises (including but not limited to the electrical, HVAC and fire sprinkler systems, security, environmental aspects, and compliance with Applicable Requirements), and their suitability for Lessee's intended use, (c) Lessee has made such investigation as it deems necessary with reference to such matters and assumes all responsibility therefor as the same relate to its occupancy of the Premises, (d) it is not relying on any representation as to the size of the Premises made by Brokers or Lessor, (e) the square footage of the Premises was not material to Lessee's decision to lease the Premises and pay the Rent stated herein, and (f) neither Lessor, Lessor's agents, nor Brokers have made any oral or written representations or warranties with respect to said matters other than as set forth in this Lease. In addition, Lessor acknowledges that: (i) Brokers have made no representations, promises or warranties concerning Lessee's ability to honor the Lease or suitability to occupy the Premises, and (ii) it is Lessor's sole responsibility to investigate the financial capability and/or suitability of all proposed tenants.
- 2.5 Lessee as Prior Owner/Occupant. The warranties made by Lessor in Paragraph 2 shall be of no force or effect if immediately prior to the Start Date, Lessee
- was the owner or occupant of the Premises. In such event, Lessee shall be responsible for any necessary corrective work.

 2.6 Vehicle Parking. So long as Lessee is not in default, and subject to the Rules and Regulations attached hereto, and as established by Lessor from time to time, Lessee shall be entitled to rent and use the number of parking spaces specified in Paragraph 1.2(b) at the rental rate applicable from time to time for monthly parking as set by Lessor and/or its licensee.
- (a) If Lessee commits, permits or allows any of the prohibited activities described in the Lease or the rules then in effect, then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove or tow away the vehicle involved and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.
- (b) The monthly rent per parking space specified in Paragraph 1.2(b) is subject to change upon 30 days prior written notice to Lessee. The rent for the parking is payable one month in advance prior to the first day of each calendar month.
- 2.7 Common Areas Definition. The term "Common Areas" is defined as all areas and facilities outside the Premises and within the exterior boundary line of the Project and interior utility raceways and installations within the Premises that are provided and designated by the Lessor from time to time for the general non-exclusive use of Lessor, Lessee and other tenants of the Project and their respective employees, suppliers, shippers, customers, contractors and invitees, including parking areas, loading and unloading areas, trash areas, roofs, roadways, walkways, driveways and landscaped areas.

 2.8 Common Areas - Lessee's Rights. Lessor grants to Lessee, for the benefit of Lessee and its employees, suppliers, shippers, contractors, customers and invitees, during the term of this Lease, the non-exclusive right to use, in common with others entitled to such use, the Common Areas as they exist from time to time,
- subject to any rights, powers, and privileges reserved by Lessor under the terms hereof or under the terms of any rules and regulations or restrictions governing the use of the Project. Under no circumstances shall the right herein granted to use the Common Areas be deemed to include the right to store any property, temporarily or permanently, in the Common Areas. Any such storage shall be permitted only by the prior written consent of Lessor or Lessor's designated agent, which consent may be revoked at any time. In the event that any unauthorized storage shall occur, then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove the property and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.
- 2.9 Common Areas Rules and Regulations. Lessor or such other person(s) as Lessor may appoint shall have the exclusive control and management of the Common Areas and shall have the right, from time to time, to adopt, modify, amend and enforce reasonable rules and regulations ("Rules and Regulations") for the management, safety, care, and cleanliness of the grounds, the parking and unloading of vehicles and the preservation of good order, as well as for the convenience of other occupants or tenants of the Building and the Project and their invitees. The Lessee agrees to abide by and conform to all such Rules and Regulations, and shall use its best efforts to cause its employees, suppliers, shippers, customers, contractors and invitees to so abide and conform. Lessor shall not be responsible to Lessee for the noncompliance with said Rules and Regulations by other tenants of the Project.
 - 2.10 Common Areas Changes. Lessor shall have the right, in Lessor's sole discretion, from time to time:
- To make changes to the Common Areas, including, without limitation, changes in the location, size, shape and number of the lobbies, windows, stairways, air shafts, elevators, escalators, restrooms, driveways, entrances, parking spaces, parking areas, loading and unloading areas, ingress, egress, direction of traffic, landscaped areas, walkways and utility raceways;
 - To close temporarily any of the Common Areas for maintenance purposes so long as reasonable access to the Premises remains available;
 - (c) To designate other land outside the boundaries of the Project to be a part of the Common Areas;
 - To add additional buildings and improvements to the Common Areas;
 - To use the Common Areas while engaged in making additional improvements, repairs or alterations to the Project, or any portion thereof; and
- To do and perform such other acts and make such other changes in, to or with respect to the Common Areas and Project as Lessor may, in the exercise of sound business judgment, deem to be appropriate.

Term.

- 3.1 Term. The Commencement Date, Expiration Date and Original Term of this Lease are as specified in Paragraph 1.3.
- Early Possession. Any provision herein granting Lessee Early Possession of the Premises is subject to and conditioned upon the Premises being available for such possession prior to the Commencement Date. Any grant of Early Possession only conveys a non-exclusive right to occupy the Premises. If Lessee totally or partially occupies the Premises prior to the Commencement Date, the obligation to pay Base Rent shall be abated for the period of such Early Possession. All other terms of this Lease (including but not limited to the obligations to pay Lessee's Share of the Operating Expense Increase) shall be in effect during such period. Any such Early Possession shall not affect the Expiration Date.
- 3.3 **Delay In Possession.** Lessor agrees to use commercially reasonable efforts to deliver exclusive possession of the Premises to Lessee by the Commencement Date. If, despite said efforts, Lessor is unable to deliver possession by such date, Lessor shall not be subject to any liability therefor, nor shall such failure affect the validity of this Lease or change the Expiration Date. Lessee shall not, however, be obligated to pay Rent or perform its other obligations until Lessor delivers possession of the Premises and any period of rent abatement that Lessee would otherwise have enjoyed shall run from the date of delivery of possession and continue for a period equal to what Lessee would otherwise have enjoyed under the terms hereof, but minus any days of delay caused by the acts or omissions of Lessee. If possession is not delivered within 60 days after the Commencement Date, as the same may be extended under the terms of any Work Letter executed by Parties, Lessee may, at its option, by notice in writing within 10 days after the end of such 60 day period, cancel this Lease, in which event the Parties shall be

INITIALS INITIALS

discharged from all obligations hereunder. If such written notice is not received by Lessor within said 10 day period, Lessee's right to cancel shall terminate. If possession of the Premises is not delivered within 120 days after the Commencement Date, this Lease shall terminate unless other agreements are reached between Lessor and Lessee, in writing.

3.4 Lessee Compliance. Lessor shall not be required to deliver possession of the Premises to Lessee until Lessee complies with its obligation to provide evidence of insurance (Paragraph 8.5). Pending delivery of such evidence, Lessee shall be required to perform all of its obligations under this Lease from and after the Start Date, including the payment of Rent, notwithstanding Lessor's election to withhold possession pending receipt of such evidence of insurance. Further, if Lessee is required to perform any other conditions prior to or concurrent with the Start Date, the Start Date shall occur but Lessor may elect to withhold possession until such conditions are satisfied.

4. Rent.

- 4.1 Rent Defined. All monetary obligations of Lessee to Lessor under the terms of this Lease (except for the Security Deposit) are deemed to be rent ("Rent").
- 4.2 Operating Expense Increase. Lessee shall pay to Lessor during the term hereof, in addition to the Base Rent, Lessee's Share of the amount by which all Operating Expenses for each Comparison Year exceeds the amount of all Operating Expenses for the Base Year, such excess being hereinafter referred to as the "Operating Expense Increase", in accordance with the following provisions:
 - (a) "Base Year" is as specified in Paragraph 1.9.
- (b) "Comparison Year" is defined as each calendar year during the term of this Lease subsequent to the Base Year; provided, however, Lessee shall have no obligation to pay a share of the Operating Expense Increase applicable to the first 12 months of the Lease Term (other than such as are mandated by a governmental authority, as to which government mandated expenses Lessee shall pay Lessee's Share, notwithstanding they occur during the first twelve (12) months). Lessee's Share of the Operating Expense Increase for the first and last Comparison Years of the Lease Term shall be prorated according to that portion of such Comparison Year as to which Lessee is responsible for a share of such increase.
- (c) The following costs relating to the ownership and operation of the Project, calculated as if the Project was at least 95% occupied, are defined as "Operating Expenses":
- (i) Costs relating to the operation, repair, and maintenance in neat, clean, safe, good order and condition, but not the replacement (see subparagraph (g)), of the following:
- (aa) The Common Areas, including their surfaces, coverings, decorative items, carpets, drapes and window coverings, and including parking areas, loading and unloading areas, trash areas, roadways, sidewalks, walkways, stairways, parkways, driveways, landscaped areas, striping, bumpers, irrigation systems, Common Area lighting facilities, building exteriors and roofs, fences and gates;
- (bb) All heating, air conditioning, plumbing, electrical systems, life safety equipment, communication systems and other equipment used in common by, or for the benefit of, tenants or occupants of the Project, including elevators and escalators, tenant directories, fire detection systems including sprinkler system maintenance and repair.
- (cc) All other areas and improvements that are within the exterior boundaries of the Project but outside of the Premises and/or any other space occupied by a tenant.
 - (ii) The cost of trash disposal, janitorial and security services, pest control services, and the costs of any environmental inspections;
 - (iii) The cost of any other service to be provided by Lessor that is elsewhere in this Lease stated to be an "Operating Expense";
- (iv) The cost of the premiums for the insurance policies maintained by Lessor pursuant to paragraph 8 and any deductible portion of an insured loss concerning the Building or the Common Areas;
 - (v) The amount of the Real Property Taxes payable by Lessor pursuant to paragraph 10;
 - (vi) The cost of water, sewer, gas, electricity, and other publicly mandated services not separately metered;
- (vii) Labor, salaries, and applicable fringe benefits and costs, materials, supplies and tools, used in maintaining and/or cleaning the Project and accounting and management fees attributable to the operation of the Project;
- (viii) The cost of any capital improvement to the Building or the Project not covered under the provisions of Paragraph 2.3 provided; however, that Lessor shall allocate the cost of any such capital improvement over a 12 year period and Lessee shall not be required to pay more than Lessee's Share of 1/144th of the cost of such Capital Expenditure in any given month;
 - (ix) The cost to replace equipment or improvements that have a useful life for accounting purposes of 5 years or less.
 - (x) Reserves set aside for maintenance, repair and/or replacement of Common Area improvements and equipment.
- (d) Any item of Operating Expense that is specifically attributable to the Premises, the Building or to any other building in the Project or to the operation, repair and maintenance thereof, shall be allocated entirely to such Premises, Building, or other building. However, any such item that is not specifically attributable to the Building or to any other building or to the operation, repair and maintenance thereof shall be equitably allocated by lessor to all buildings in the Project.
- the Building or to any other building or to the operation, repair and maintenance thereof, shall be equitably allocated by Lessor to all buildings in the Project.

 (e) The inclusion of the improvements, facilities and services set forth in Subparagraph 4.2(c) shall not be deemed to impose an obligation upon Lessor to either have said improvements or facilities or to provide those services unless the Project already has the same, Lessor already provides the services, or Lessor has agreed elsewhere in this Lease to provide the same or some of them.
- (f) Lessee's Share of Operating Expense Increase is payable monthly on the same day as the Base Rent is due hereunder. The amount of such payments shall be based on Lessor's estimate of the Operating Expenses. Within 60 days after written request (but not more than once each year) Lessor shall deliver to Lessee a reasonably detailed statement showing Lessee's Share of the actual Common Area Operating Expenses for the preceding year. If Lessee's payments during such Year exceed Lessee's Share, Lessee shall credit the amount of such over-payment against Lessee's future payments. If Lessee's payments during such Year were less than Lessee's Share, Lessee shall pay to Lessor the amount of the deficiency within 10 days after delivery by Lessor to Lessee of said statement. Lessor and Lessee shall forthwith adjust between them by cash payment any balance determined to exist with respect to that portion of the last Comparison Year for which Lessee is responsible as to Operating Expense Increases, notwithstanding that the Lease term may have terminated before the end of such Comparison Year.
- (g) Operating Expenses shall not include the costs of replacement for equipment or capital components such as the roof, foundations, exterior walls or a Common Area capital improvement, such as the parking lot paying, elevators, fences that have a useful life for accounting purposes of 5 years or more.
- Common Area capital improvement, such as the parking lot paving, elevators, fences that have a useful life for accounting purposes of 5 years or more.

 (h) Operating Expenses shall not include any expenses paid by any tenant directly to third parties, or as to which Lessor is otherwise reimbursed by any third party, other tenant, or by insurance proceeds.
- 4.3 **Payment.** Lessee shall cause payment of Rent to be received by Lessor in lawful money of the United States on or before the day on which it is due, without offset or deduction (except as specifically permitted in this Lease). All monetary amounts shall be rounded to the nearest whole dollar. In the event that any invoice prepared by Lessor is inaccurate such inaccuracy shall not constitute a waiver and Lessee shall be obligated to pay the amount set forth in this Lease. Rent for any period during the term hereof which is for less than one full calendar month shall be prorated based upon the actual number of days of said month. Payment of Rent shall be made to Lessor at its address stated herein or to such other persons or place as Lessor may from time to time designate in writing. Acceptance of a payment which is less than the amount then due shall not be a waiver of Lessor's rights to the balance of such Rent, regardless of Lessor's endorsement of any check so stating. In the event that any check, draft, or other instrument of payment given by Lessee to Lessor is dishonored for any reason, Lessee agrees to pay to Lessor the sum of \$25 in addition to any Late Charge and Lessor, at its option, may require all future Rent be paid by cashier's check. Payments will be applied first to accrued late charges and attorney's fees, second to accrued interest, then to Base Rent and Common Area Operating Expenses, and any remaining amount to any other outstanding charges or costs.
- Security Deposit. Lessee shall deposit with Lessor upon execution hereof the Security Deposit as security for Lessee's faithful performance of its obligations under this Lease. If Lessee fails to pay Rent, or otherwise Defaults under this Lease, Lessor may use, apply or retain all or any portion of said Security Deposit for the payment of any amount already due Lessor, for Rents which will be due in the future, and/or to reimburse or compensate Lessor for any liability, expense, loss or damage which Lessor may suffer or incur by reason thereof. If Lessor uses or applies all or any portion of the Security Deposit, Lessee shall within 10 days after written request therefor deposit monies with Lessor sufficient to restore said Security Deposit to the full amount required by this Lease. If the Base Rent increases during the term of this Lease, Lessee shall, upon written request from Lessor, deposit additional monies with Lessor so that the total amount of the Security Deposit shall at all times bear the same proportion to the increased Base Rent as the initial Security Deposit bore to the initial Base Rent. Should the Agreed Use be amended to accommodate a material change in the business of Lessee or to accommodate a sublessee or assignee, Lessor shall have the right to increase the Security Deposit to the extent necessary, in Lessor's reasonable judgment, to account for any increased wear and tear that the Premises may suffer as a result thereof. If a change in control of Lessee occurs during this Lease and following such change the financial condition of Lessee is, in Lessor's reasonable judgment, significantly reduced, Lessee shall deposit such additional monies with Lessor as shall be sufficient to cause the Security Deposit to be at a commercially reasonable level based on such change in financial condition. Lessor shall not be required to keep the Security Deposit separate from its general accounts. Within 90 days after the expiration or termination of this Lease, Lessor shall return that portion of the Security Deposit not used or applied by Lessor. Lessor shall upon written request provide Lessee with an accounting showing how that portion of the Security Deposit that was not returned was applied. No part of the Security Deposit shall be considered to be held in trust, to bear interest or to be prepayment for any monies to be paid by Lessee under this Lease. THE SECURITY DEPOSIT SHALL NOT BE USED BY LESSEE IN LIEU OF PAYMENT OF THE LAST MONTH'S RENT.

6. Use.	
	
INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

OFG-21.30, Revised 10-22-2020

6.1 **Use**. Lessee shall use and occupy the Premises only for the Agreed Use, or any other legal use which is reasonably comparable thereto, and for no other purpose. Lessee shall not use or permit the use of the Premises in a manner that is unlawful, creates damage, waste or a nuisance, or that disturbs occupants of or causes damage to neighboring premises or properties. Other than guide, signal and seeing eye dogs, Lessee shall not keep or allow in the Premises any pets, animals, birds, fish, or reptiles. Lessor shall not unreasonably withhold or delay its consent to any written request for a modification of the Agreed Use, so long as the same will not impair the structural integrity of the improvements of the Building, will not adversely affect the mechanical, electrical, HVAC, and other systems of the Building, and/or will not affect the exterior appearance of the Building. If Lessor elects to withhold consent, Lessor shall within 7 days after such request give written notification of same, which notice shall include an explanation of Lessor's objections to the change in the Agreed Use.

6.2 Hazardous Substances

- (a) Reportable Uses Require Consent. The term "Hazardous Substance" as used in this Lease shall mean any product, substance, or waste whose presence, use, manufacture, disposal, transportation, or release, either by itself or in combination with other materials expected to be on the Premises, is either; (i) potentially injurious to the public health, safety or welfare, the environment or the Premises, (ii) regulated or monitored by any governmental authority, or (iii) a basis for potential liability of Lessor to any governmental agency or third party under any applicable statute or common law theory. Hazardous Substances shall include, but not be limited to, hydrocarbons, petroleum, gasoline, and/or crude oil or any products, byproducts or fractions thereof. Lessee shall not engage in any activity in or on the Premises which constitutes a Reportable Use of Hazardous Substances without the express prior written consent of Lessor and timely compliance (at Lessee's expense) with all Applicable Requirements. "Reportable Use" shall mean (i) the installation or use of any above or below ground storage tank, (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and/or (iii) the presence at the Premises of a Hazardous Substance with respect to which any Applicable Requirements requires that a notice be given to persons entering or occupying the Premises or neighboring properties. Notwithstanding the foregoing, Lessee may use any ordinary and customary materials reasonably required to be used in the normal course of the Agreed Use such as ordinary office supplies (copier toner, liquid paper, glue, etc.) and common household cleaning materials, so long as such use is in compliance with all Applicable Requirements, is not a Reportable Use, and does not expose the Premises or neighboring property to any meaningful risk of contamination or damage or expose Lessor to any liability therefore. In addition, Lessor may condition its consent to any Reportable Use upon receiving such additional assurances as Lessor reasonably deems necessary to protect itself, the public, the Premises and/or the environment against damage, contamination, injury and/or liability, including, but not limited to, the installation (and removal on or before Lease expiration or termination) of protective modifications (such as concrete encasements) and/or increasing the Security Deposit.
- (b) **Duty to Inform Lessor**. If Lessee knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises, other than as previously consented to by Lessor, Lessee shall immediately give written notice of such fact to Lessor, and provide Lessor with a copy of any report, notice, claim or other documentation which it has concerning the presence of such Hazardous Substance.
- (c) Lessee Remediation. Lessee shall not cause or permit any Hazardous Substance to be spilled or released in, on, under, or about the Premises (including through the plumbing or sanitary sewer system) and shall promptly, at Lessee's expense, comply with all Applicable Requirements and take all investigatory and/or remedial action reasonably recommended, whether or not formally ordered or required, for the cleanup of any contamination of, and for the maintenance, security and/or monitoring of the Premises or neighboring properties, that was caused or materially contributed to by Lessee, or pertaining to or involving any Hazardous Substance brought onto the Premises during the term of this Lease, by or for Lessee, or any third party.
- (d) Lessee Indemnification. Lessee shall indemnify, defend and hold Lessor, its agents, employees, lenders and ground lessor, if any, harmless from and against any and all loss of rents and/or damages, liabilities, judgments, claims, expenses, penalties, and attorneys' and consultants' fees arising out of or involving any Hazardous Substance brought onto the Premises by or for Lessee, or any third party (provided, however, that Lessee shall have no liability under this Lease with respect to underground migration of any Hazardous Substance under the Premises from areas outside of the Project not caused or contributed to by Lessee). Lessee's obligations shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Lessee, and the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease. No termination, cancellation or release agreement entered into by Lessor and Lessee shall release Lessee from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Lessor in writing at the time of such agreement.
- (e) Lessor Indemnification. Except as otherwise provided in paragraph 8.7, Lessor and its successors and assigns shall indemnify, defend, reimburse and hold Lessee, its employees and lenders, harmless from and against any and all environmental damages, including the cost of remediation, which result from Hazardous Substances which existed on the Premises prior to Lessee's occupancy or which are caused by the gross negligence or willful misconduct of Lessor, its agents or employees. Lessor's obligations, as and when required by the Applicable Requirements, shall include, but not be limited to, the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease.
- (f) Investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by governmental entities having jurisdiction with respect to the existence of Hazardous Substances on the Premises prior to Lessee's occupancy, unless such remediation measure is required as a result of Lessee's use (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises, in which event Lessee shall be responsible for such payment. Lessee shall cooperate fully in any such activities at the request of Lessor, including allowing Lessor and Lessor's agents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remedial responsibilities.
- responsible for such payment. Lessee shall cooperate fully in any such activities at the request of Lessor, including allowing Lessor and Lessor's agents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remedial responsibilities.

 (g) Lessor Termination Option. If a Hazardous Substance Condition (see Paragraph 9.1(e)) occurs during the term of this Lease, unless Lessee is legally responsible therefor (in which case Lessee shall make the investigation and remediation thereof required by the Applicable Requirements and this Lease shall continue in full force and effect, but subject to Lessor's rights under Paragraph 6.2(d) and Paragraph 13), Lessor may, at Lessor's option, either (i) investigate and remediate such Hazardous Substance Condition, if required, as soon as reasonably possible at Lessor's expense, in which event this Lease shall continue in full force and effect, or (ii) if the estimated cost to remediate such condition exceeds 12 times the then monthly Base Rent or \$100,000, whichever is greater, give written notice to Lessee, within 30 days after receipt by Lessor of knowledge of the occurrence of such Hazardous Substance Condition, of Lessor's desire to terminate this Lease as of the date 60 days following the date of such notice. In the event Lessor elects to give a termination notice, Lessee may, within 10 days thereafter, give written notice to Lessor of Lessee's commitment to pay the amount by which the cost of the remediation of such Hazardous Substance Condition exceeds an amount equal to 12 times the then monthly Base Rent or \$100,000, whichever is greater. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days following such commitment. In such event, this Lease shall continue in full force and effect, and Lessor shall proceed to make such remediation as soon as reasonably possible after the required funds are available. If Lessee does not give such notice and provide the required funds or ass
- 6.3 Lessee's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lessee's sole expense, fully, diligently and in a timely manner, materially comply with all Applicable Requirements, the requirements of any applicable fire insurance underwriter or rating bureau, and the recommendations of Lessor's engineers and/or consultants which relate in any manner to the Premises, without regard to whether said Applicable Requirements are now in effect or become effective after the Start Date. Lessee shall, within 10 days after receipt of Lessor's written request, provide Lessor with copies of all permits and other documents, and other information evidencing Lessee's compliance with any Applicable Requirements specified by Lessor, and shall immediately upon receipt, notify Lessor in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving the failure of Lessee or the Premises to comply with any Applicable Requirements. Likewise, Lessee shall immediately give written notice to Lessor of: (i) any water damage to the Premises and any suspected seepage, pooling, dampness or other condition conducive to the production of mold; or (ii) any mustiness or other odors that might indicate the presence of mold in the Premises.
- 6.4 Inspection; Compliance. Lessor and Lessor's "Lender" (as defined in Paragraph 30) and consultants authorized by Lessor shall have the right to enter into Premises at any time, in the case of an emergency, and otherwise at reasonable times, after reasonable notice, for the purpose of inspecting and/or testing the condition of the Premises and/or for verifying compliance by Lessee with this Lease. The cost of any such inspections shall be paid by Lessor, unless a violation of Applicable Requirements, or a Hazardous Substance Condition (see Paragraph 9.1(e)) is found to exist or be imminent, or the inspection is requested or ordered by a governmental authority. In such case, Lessee shall upon request reimburse Lessor for the cost of such inspection, so long as such inspection is reasonably related to the violation or contamination. In addition, Lessee shall provide copies of all relevant material safety data sheets (MSDS) to Lessor within 10 days of the receipt of written request therefor. Lessee acknowledges that any failure on its part to allow such inspections or testing will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to allow such inspections and/or testing in a timely fashion the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for the remainder to the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to allow such inspection and/or testing. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to such failure nor prevent the exercise of any of the other rights and remedies granted hereunder.

7. Maintenance; Repairs; Utility Installations; Trade Fixtures and Alterations.

7.1 **Lessee's Obligations**. Notwithstanding Lessor's obligation to keep the Premises in good condition and repair, Lessee shall be responsible for payment of the cost thereof to Lessor as additional rent for that portion of the cost of any maintenance and repair of the Premises, or any equipment (wherever located) that serves only Lessee or the Premises, to the extent such cost is attributable to abuse or misuse. In addition, Lessee rather than the Lessor shall be responsible for the cost of painting, repairing or replacing wall coverings, and to repair or replace any similar improvements within the Premises. Lessor may, at its option, upon

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

reasonable notice, elect to have Lessee perform any particular such maintenance or repairs the cost of which is otherwise Lessee's responsibility hereunder."

7.2 **Lessor's Obligations**. Subject to the provisions of Paragraphs 2.2 (Condition), 2.3 (Compliance), 4.2 (Operating Expenses), 6 (Use), 7.1 (Lessee's Obligations), 9 (Damage or Destruction) and 14 (Condemnation), Lessor, subject to reimbursement pursuant to Paragraph 4.2, shall keep in good order, condition and repair the foundations, exterior walls, structural condition of interior bearing walls, exterior roof, fire sprinkler system, fire alarm and/or smoke detection systems, fire hydrants, and the Common Areas.

7.3 Utility Installations; Trade Fixtures; Alterations.

- (a) **Definitions**. The term "**Utility Installations**" refers to all floor and window coverings, air lines, vacuum lines, power panels, electrical distribution, security and fire protection systems, communication cabling, lighting fixtures, HVAC equipment, and plumbing in or on the Premises. The term "**Trade Fixtures**" shall mean Lessee's machinery and equipment that can be removed without doing material damage to the Premises. The term "**Alterations**" shall mean any modification of the improvements, other than Utility Installations or Trade Fixtures, whether by addition or deletion. "**Lessee Owned Alterations and/or Utility Installations**" are defined as Alterations and/or Utility Installations made by Lessee that are not yet owned by Lessor pursuant to Paragraph 7.4(a).
- (b) Consent. Lessee shall not make any Alterations or Utility Installations to the Premises without Lessor's prior written consent. Lessee may, however, make non-structural Alterations or Utility Installations to the interior of the Premises (excluding the roof) without such consent but upon notice to Lessor, as long as they are not visible from the outside, do not involve puncturing, relocating or removing the roof, ceilings, floors or any existing walls, will not affect the electrical, plumbing, HVAC, and/or life safety systems, do not trigger the requirement for additional modifications and/or improvements to the Premises resulting from Applicable Requirements, such as compliance with Title 24, and the cumulative cost thereof during this Lease as extended does not exceed \$2000. Notwithstanding the foregoing, Lessee shall not make or permit any roof penetrations and/or install anything on the roof without the prior written approval of Lessor. Lessor may, as a precondition to granting such approval, require Lessee to utilize a contractor chosen and/or approved by Lessor. Any Alterations or Utility Installations that Lessee shall desire to make and which require the consent of the Lessor shall be presented to Lessor in written form with detailed plans. Consent shall be deemed conditioned upon Lessee's: (i) acquiring all applicable governmental permits, (ii) furnishing Lessor with copies of both the permits and the plans and specifications prior to commencement of the work, and (iii) compliance with all conditions of said permits and other Applicable Requirements in a prompt and expeditions manner. Any Alterations or Utility Installations shall be performed in a workmanlike manner with good and sufficient materials. Lessee shall promptly upon completion furnish Lessor with as-built plans and specifications. For work which costs an amount in excess of one month's Base Rent, Lessor may condition its consent upon Lessee's posting an additional Security Deposit with Lessor.
- (c) Liens; Bonds. Lessee shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lessee at or for use on the Premises, which claims are or may be secured by any mechanic's or materialmen's lien against the Premises or any interest therein. Lessee shall give Lessor not less than 10 days notice prior to the commencement of any work in, on or about the Premises, and Lessor shall have the right to post notices of non-responsibility. If Lessee shall contest the validity of any such lien, claim or demand, then Lessee shall, at its sole expense defend and protect itself, Lessor and the Premises against the same and shall pay and satisfy any such adverse judgment that may be rendered thereon before the enforcement thereof. If Lessor shall require, Lessee shall furnish a surety bond in an amount equal to 150% of the amount of such contested lien, claim or demand, indemnifying Lessor against liability for the same. If Lessor elects to participate in any such action, Lessee shall pay Lessor's attorneys' fees and costs.

7.4 Ownership; Removal; Surrender; and Restoration.

- (a) **Ownership.** Subject to Lessor's right to require removal or elect ownership as hereinafter provided, all Alterations and Utility Installations made by Lessee shall be the property of Lessee, but considered a part of the Premises. Lessor may, at any time, elect in writing to be the owner of all or any specified part of the Lessee Owned Alterations and Utility Installations. Unless otherwise instructed per paragraph 7.4(b) hereof, all Lessee Owned Alterations and Utility Installations shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.
- (b) **Removal**. By delivery to Lessee of written notice from Lessor not earlier than 90 and not later than 30 days prior to the end of the term of this Lease, Lessor may require that any or all Lessee Owned Alterations or Utility Installations be removed by the expiration or termination of this Lease. Lessor may require the removal at any time of all or any part of any Lessee Owned Alterations or Utility Installations made without the required consent.
- (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements, parts and surfaces thereof clean and free of debris, and in good operating order, condition and state of repair, ordinary wear and tear excepted. "Ordinary wear and tear" shall not include any damage or deterioration that would have been prevented by good maintenance practice. Notwithstanding the foregoing and the provisions of Paragraph 7.1(a), if the Lessee occupies the Premises for 12 months or less, then Lessee shall surrender the Premises in the same condition as delivered to Lessee on the Start Date with NO allowance for ordinary wear and tear. Lessee shall repair any damage occasioned by the installation, maintenance or removal of Trade Fixtures, Lessee owned Alterations and/or Utility Installations, furnishings, and equipment as well as the removal of any storage tank installed by or for Lessee. Lessee shall also remove from the Premises any and all Hazardous Substances brought onto the Premises by or for Lessee, or any third party (except Hazardous Substances which were deposited via underground migration from areas outside of the Premises) to the level specified in Applicable Requirements. Trade Fixtures shall remain the property of Lessee and shall be removed by Lessee. Any personal property of Lessee not removed on or before the Expiration Date or any earlier termination date shall be deemed to have been abandoned by Lessee and may be disposed of or retained by Lessor as Lessor may desire. The failure by Lessee to timely vacate the Premises pursuant to this Paragraph 7.4(c) without the express written consent of Lessor shall constitute a holdover under the provisions of Paragraph 26 below.

8. Insurance; Indemnity.

8.1 Insurance Premiums. The cost of the premiums for the insurance policies maintained by Lessor pursuant to paragraph 8 are included as Operating Expenses (see paragraph 4.2 (c)(iv)). Said costs shall include increases in the premiums resulting from additional coverage related to requirements of the holder of a mortgage or deed of trust covering the Premises, Building and/or Project, increased valuation of the Premises, Building and/or Project, and/or a general premium rate increase. Said costs shall not, however, include any premium increases resulting from the nature of the occupancy of any other tenant of the Building. If the Project was not insured for the entirety of the Base Year, then the base premium shall be the lowest annual premium reasonably obtainable for the required insurance as of the Start Date, assuming the most nominal use possible of the Building and/or Project. In no event, however, shall Lessee be responsible for any portion of the premium cost attributable to liability insurance coverage in excess of \$2,000,000 procured under Paragraph 8.2(b).

8.2 Liability Insurance.

- (a) Carried by Lessee. Lessee shall obtain and keep in force a Commercial General Liability policy of insurance protecting Lessee and Lessor as an additional insured against claims for bodily injury, personal injury and property damage based upon or arising out of the ownership, use, occupancy or maintenance of the Premises and all areas appurtenant thereto. Such insurance shall be on an occurrence basis providing single limit coverage in an amount not less than \$1,000,000 per occurrence with an annual aggregate of not less than \$2,000,000. Lessee shall add Lessor as an additional insured by means of an endorsement at least as broad as the Insurance Service Organization's "Additional Insured-Managers or Lessors of Premises" Endorsement and coverage shall also be extended to include damage caused by heat, smoke or fumes from a hostile fire. The policy shall not contain any intra-insured exclusions as between insured persons or organizations, but shall include coverage for liability assumed under this Lease as an "insured contract" for the performance of Lessee's indemnity obligations under this Lease. The limits of said insurance shall not, however, limit the liability of Lessee nor relieve Lessee of any obligation hereunder. Lessee shall provide an endorsement on its liability policy(ies) which provides that its insurance shall be primary to and not contributory with any similar insurance carried by Lessor, whose insurance shall be considered excess insurance only.
- (b) Carried by Lessor. Lessor shall maintain liability insurance as described in Paragraph 8.2(a), in addition to, and not in lieu of, the insurance required to be maintained by Lessee. Lessee shall not be named as an additional insured therein.

8.3 Property Insurance - Building, Improvements and Rental Value.

- (a) Building and Improvements. Lessor shall obtain and keep in force a policy or policies of insurance in the name of Lessor, with loss payable to Lessor, any ground-lessor, and to any Lender insuring loss or damage to the Building and/or Project. The amount of such insurance shall be equal to the full insurable replacement cost of the Building and/or Project, as the same shall exist from time to time, or the amount required by any Lender, but in no event more than the commercially reasonable and available insurable value thereof. Lessee Owned Alterations and Utility Installations, Trade Fixtures, and Lessee's personal property shall be insured by Lessor. If the coverage is available and commercially appropriate, such policy or policies shall insure against all risks of direct physical loss or damage (except the perils of flood and/or earthquake unless required by a Lender), including coverage for debris removal and the enforcement of any Applicable Requirements requiring the upgrading, demolition, reconstruction or replacement of any portion of the Premises as the result of a covered loss. Said policy or policies shall also contain an agreed valuation provision in lieu of any coinsurance clause, waiver of subrogation, and inflation guard protection causing an increase in the annual property insurance coverage amount by a factor of not less than the adjusted U.S. Department of Labor Consumer Price Index for All Urban Consumers for the city nearest to where the Premises are located. If such insurance coverage has a deductible clause, the deductible amount shall not exceed \$5,000 per occurrence.
- (b) Rental Value. Lessor shall also obtain and keep in force a policy or policies in the name of Lessor with loss payable to Lessor and any Lender, insuring the loss of the full Rent for one year with an extended period of indemnity for an additional 180 days ("Rental Value insurance"). Said insurance shall contain an agreed valuation provision in lieu of any coinsurance clause, and the amount of coverage shall be adjusted annually to reflect the projected Rent otherwise payable by Lessee, for the next 12 month period.
- (c) Adjacent Premises. Lessee shall pay for any increase in the premiums for the property insurance of the Building and for the Common Areas or other buildings in the Project if said increase is caused by Lessee's acts, omissions, use or occupancy of the Premises.

INITIALS
INITIALS

Last Edited: 9/30/2022 2:33 PM

INITIALS

- (d) Lessee's Improvements. Since Lessor is the Insuring Party, Lessor shall not be required to insure Lessee Owned Alterations and Utility Installations unless the item in question has become the property of Lessor under the terms of this Lease.
 - 8.4 Lessee's Property; Business Interruption Insurance; Worker's Compensation Insurance.
- (a) **Property Damage**. Lessee shall obtain and maintain insurance coverage on all of Lessee's personal property, Trade Fixtures, and Lessee Owned Alterations and Utility Installations. Such insurance shall be full replacement cost coverage with a deductible of not to exceed \$1,000 per occurrence. The proceeds from any such insurance shall be used by Lessee for the replacement of personal property, Trade Fixtures and Lessee Owned Alterations and Utility Installations.
- (b) Worker's Compensation Insurance. Lessee shall obtain and maintain Worker's Compensation Insurance in such amount as may be required by Applicable Requirements. Such policy shall include a 'Waiver of Subrogation' endorsement. Lessee shall provide Lessor with a copy of such endorsement along with the certificate of insurance or copy of the policy required by paragraph 8.5.
- (c) **Business Interruption**. Lessee shall obtain and maintain loss of income and extra expense insurance in amounts as will reimburse Lessee for direct or indirect loss of earnings attributable to all perils commonly insured against by prudent lessees in the business of Lessee or attributable to prevention of access to the Premises as a result of such perils.
- (d) No Representation of Adequate Coverage. Lessor makes no representation that the limits or forms of coverage of insurance specified herein are adequate to cover Lessee's property, business operations or obligations under this Lease.
- 8.5 **Insurance Policies**. Insurance required herein shall be by companies maintaining during the policy term a "General Policyholders Rating" of at least A-, VII, as set forth in the most current issue of "Best's Insurance Guide", or such other rating as may be required by a Lender. Lessee shall not do or permit to be done anything which invalidates the required insurance policies. Lessee shall, prior to the Start Date, deliver to Lessor certified copies of policies of such insurance or certificates with copies of the required endorsements evidencing the existence and amounts of the required insurance. No such policy shall be cancelable or subject to modification except after 10 days prior written notice to Lessor. Lessee shall, at least 30 days prior to the expiration of such policies, furnish Lessor with evidence of renewals or "insurance binders" evidencing renewal thereof, or Lessor may increase his liability insurance coverage and charge the cost thereof to Lessee, which amount shall be payable by Lessee to Lessor upon demand. Such policies shall be for a term of at least one year, or the length of the remaining term of this Lease, whichever is less. If either Party shall fail to procure and maintain the insurance required to be carried by it, the other Party may, but shall not be required to, procure and maintain the same.
- 8.6 Waiver of Subrogation. Without affecting any other rights or remedies, Lessee and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is not limited by the amount of insurance carried or required, or by any deductibles applicable hereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Lessee, as the case may be, so long as the insurance is not invalidated thereby.
- 8.7 **Indemnity**. Except for Lessor's gross negligence or willful misconduct, Lessee shall indemnify, protect, defend and hold harmless the Premises, Lessor and its agents, Lessor's master or ground lessor, partners and Lenders, from and against any and all claims, loss of rents and/or damages, liens, judgments, penalties, attorneys' and consultants' fees, expenses and/or liabilities arising out of, involving, or in connection with, a Breach of the Lease by Lessee and/or the use and/or occupancy of the Premises and/or Project by Lessee and/or by Lessee's employees, contractors or invitees. If any action or proceeding is brought against Lessor by reason of any of the foregoing matters, Lessee shall upon notice defend the same at Lessee's expense by counsel reasonably satisfactory to Lessor and Lessor shall cooperate with Lessee in such defense. Lessor need not have first paid any such claim in order to be defended or indemnified.
- 8.8 **Exemption of Lessor and its Agents from Liability**. Notwithstanding the negligence or breach of this Lease by Lessor or its agents, neither Lessor nor its agents shall be liable under any circumstances for: (i) injury or damage to the person or goods, wares, merchandise or other property of Lessee, Lessee's employees, contractors, invitees, customers, or any other person in or about the Premises, whether such damage or injury is caused by or results from fire, steam, electricity, gas, water or rain, indoor air quality, the presence of mold or from the breakage, leakage, obstruction or other defects of pipes, fire sprinklers, wires, appliances, plumbing, HVAC or lighting fixtures, or from any other cause, whether the said injury or damage results from conditions arising upon the Premises or upon other portions of the Building, or from other sources or places, (ii) any damages arising from any act or neglect of any other tenant of Lessor or from the failure of Lessor or its agents to enforce the provisions of any other lease in the Project, or (iii) injury to Lessee's business or for any loss of income or profit therefrom. Instead, it is intended that Lessee's sole recourse in the event of such damages or injury be to file a claim on the insurance policy(ies) that Lessee is required to maintain pursuant to the provisions of paragraph 8.
- 8.9 **Failure to Provide Insurance**. Lessee acknowledges that any failure on its part to obtain or maintain the insurance required herein will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, for any month or portion thereof that Lessee does not maintain the required insurance and/or does not provide Lessor with the required binders or certificates evidencing the existence of the required insurance, the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater. The parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to maintain the required insurance. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to maintain such insurance, prevent the exercise of any of the other rights and remedies granted hereunder, nor relieve Lessee of its obligation to maintain the insurance specified in this Lease.

9. Damage or Destruction.

9.1 **Definitions**.

- (a) "Premises Partial Damage" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations, which can reasonably be repaired in 3 months or less from the date of the damage or destruction, and the cost thereof does not exceed a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (b) "Premises Total Destruction" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which cannot reasonably be repaired in 3 months or less from the date of the damage or destruction and/or the cost thereof exceeds a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (c) "Insured Loss" shall mean damage or destruction to improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which was caused by an event required to be covered by the insurance described in Paragraph 8.3(a), irrespective of any deductible amounts or coverage limits involved.
- (d) "Replacement Cost" shall mean the cost to repair or rebuild the improvements owned by Lessor at the time of the occurrence to their condition existing immediately prior thereto, including demolition, debris removal and upgrading required by the operation of Applicable Requirements, and without deduction for depreciation.
- for depreciation.

 (e) "Hazardous Substance Condition" shall mean the occurrence or discovery of a condition involving the presence of, or a contamination by, a Hazardous Substance, in, on, or under the Premises which requires restoration.
- 9.2 Partial Damage Insured Loss. If a Premises Partial Damage that is an Insured Loss occurs, then Lessor shall, at Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonably possible and this Lease shall continue in full force and effect; provided, however, that Lessee shall, at Lessor's election, make the repair of any damage or destruction the total cost to repair of which is \$5,000 or less, and, in such event, Lessor shall make any applicable insurance proceeds available to Lessee on a reasonable basis for that purpose. Notwithstanding the foregoing, if the required insurance was not in force or the insurance proceeds are not sufficient to effect such repair, the Insuring Party shall promptly contribute the shortage in proceeds as and when required to complete said repairs. In the event, however, such shortage was due to the fact that, by reason of the unique nature of the improvements, full replacement cost insurance coverage was not commercially reasonable and available, Lessor shall have no obligation to pay for the shortage in insurance proceeds or to fully restore the unique aspects of the Premises unless Lessee provides Lessor with the funds to cover same, or adequate assurance thereof, within 10 days following receipt of written notice of such shortage and request therefor. If Lessor receives said funds or adequate assurance thereof within said 10 day period, the party responsible for making the repairs shall complete them as soon as reasonably possible and this Lease shall remain in full force and effect. If such funds or assurance are not received, Lessor may nevertheless elect by written notice to Lessee within 10 days thereafter to: (i) make such restoration and repair as is commercially reasonable with Lessor paying any shortage in proceeds, in which case this Lease shall remain in full force and effect, or (ii) have this Lease terminate 30 days thereafter. Lessee shall not be entitled to reimbursement of a
- 9.3 Partial Damage Uninsured Loss. If a Premises Partial Damage that is not an Insured Loss occurs, unless caused by a negligent or willful act of Lessee (in which event Lessee shall make the repairs at Lessee's expense), Lessor may either: (i) repair such damage as soon as reasonably possible at Lessor's expense (subject to reimbursement pursuant to Paragraph 4.2), in which event this Lease shall continue in full force and effect, or (ii) terminate this Lease by giving written notice to Lessee within 30 days after receipt by Lessor of knowledge of the occurrence of such damage. Such termination shall be effective 60 days following the date of such notice. In the event Lessor elects to terminate this Lease, Lessee shall have the right within 10 days after receipt of the termination notice to give written notice to Lessor of Lessee's commitment to pay for the repair of such damage without reimbursement from Lessor. Lessee shall provide Lessor with said funds or satisfactory

INITIALS

© 2019 AIR CRE. All Rights Reserved.

assurance thereof within 30 days after making such commitment. In such event this Lease shall continue in full force and effect, and Lessor shall proceed to make such repairs as soon as reasonably possible after the required funds are available. If Lessee does not make the required commitment, this Lease shall terminate as of the date specified in the termination notice.

- 9.4 **Total Destruction**. Notwithstanding any other provision hereof, if a Premises Total Destruction occurs, this Lease shall terminate 60 days following such Destruction. If the damage or destruction was caused by the gross negligence or willful misconduct of Lessee, Lessor shall have the right to recover Lessor's damages from Lessee, except as provided in Paragraph 8.6.
- 9.5 **Damage Near End of Term.** If at any time during the last 6 months of this Lease there is damage for which the cost to repair exceeds one month's Base Rent, whether or not an Insured Loss, Lessor may terminate this Lease effective 60 days following the date of occurrence of such damage by giving a written termination notice to Lessee within 30 days after the date of occurrence of such damage. Notwithstanding the foregoing, if Lessee at that time has an exercisable option to extend this Lease or to purchase the Premises, then Lessee may preserve this Lease by, (a) exercising such option and (b) providing Lessor with any shortage in insurance proceeds (or adequate assurance thereof) needed to make the repairs on or before the earlier of (i) the date which is 10 days after Lessee's receipt of Lessor's written notice purporting to terminate this Lease, or (ii) the day prior to the date upon which such option expires. If Lessee duly exercises such option during such period and provides Lessor with funds (or adequate assurance thereof) to cover any shortage in insurance proceeds, Lessor shall, at Lessor's commercially reasonable expense, repair such damage as soon as reasonably possible and this Lease shall continue in full force and effect. If Lessee fails to exercise such option and provide such funds or assurance during such period, then this Lease shall terminate on the date specified in the termination notice and Lessee's option shall be extinguished.

9.6 Abatement of Rent; Lessee's Remedies.

- (a) Abatement. In the event of Premises Partial Damage or Premises Total Destruction or a Hazardous Substance Condition for which Lessee is not responsible under this Lease, the Rent payable by Lessee for the period required for the repair, remediation or restoration of such damage shall be abated in proportion to the degree to which Lessee's use of the Premises is impaired, but not to exceed the proceeds received from the Rental Value insurance. All other obligations of Lessee hereunder shall be performed by Lessee, and Lessor shall have no liability for any such damage, destruction, remediation, repair or restoration except as provided herein.
- (b) Remedies. If Lessor is obligated to repair or restore the Premises and does not commence, in a substantial and meaningful way, such repair or restoration within 90 days after such obligation shall accrue, Lessee may, at any time prior to the commencement of such repair or restoration, give written notice to Lessor and to any Lenders of which Lessee has actual notice, of Lessee's election to terminate this Lease on a date not less than 60 days following the giving of such notice. If Lessee gives such notice and such repair or restoration is not commenced within 30 days thereafter, this Lease shall terminate as of the date specified in said notice. If the repair or restoration is commenced within such 30 days, this Lease shall continue in full force and effect. "Commence" shall mean either the unconditional authorization of the preparation of the required plans, or the beginning of the actual work on the Premises, whichever first occurs.
- 9.7 **Termination; Advance Payments**. Upon termination of this Lease pursuant to Paragraph 6.2(g) or Paragraph 9, an equitable adjustment shall be made concerning advance Base Rent and any other advance payments made by Lessee to Lessor. Lessor shall, in addition, return to Lessee so much of Lessee's Security Deposit as has not been, or is not then required to be, used by Lessor.

10. Real Property Taxes.

- 10.1 **Definitions**. As used herein, the term "Real Property Taxes" shall include any form of assessment; real estate, general, special, ordinary or extraordinary, or rental levy or tax (other than inheritance, personal income or estate taxes); improvement bond; and/or license fee imposed upon or levied against any legal or equitable interest of Lessor in the Project, Lessor's right to other income therefrom, and/or Lessor's business of leasing, by any authority having the direct or indirect power to tax and where the funds are generated with reference to the Project address. "Real Property Taxes" shall also include any tax, fee, levy, assessment or charge, or any increase therein: (i) imposed by reason of events occurring during the term of this Lease, including but not limited to, a change in the ownership of the Project, (ii) a change in the improvements thereon, and/or (iii) levied or assessed on machinery or equipment provided by Lessor to Lessee pursuant to this Lease.
- 10.2 **Payment of Taxes**. Except as otherwise provided in Paragraph 10.3, Lessor shall pay the Real Property Taxes applicable to the Project, and said payments shall be included in the calculation of Operating Expenses in accordance with the provisions of Paragraph 4.2.
- 10.3 Additional Improvements. Operating Expenses shall not include Real Property Taxes specified in the tax assessor's records and work sheets as being caused by additional improvements placed upon the Project by other lessees or by Lessor for the exclusive enjoyment of such other lessees. Notwithstanding Paragraph 10.2 hereof, Lessee shall, however, pay to Lessor at the time Operating Expenses are payable under Paragraph 4.2, the entirety of any increase in Real Property Taxes if assessed solely by reason of Alterations, Trade Fixtures or Utility Installations placed upon the Premises by Lessee or at Lessee's request or by reason of any alterations or improvements to the Premises made by Lessor subsequent to the execution of this Lease by the Parties.
- 10.4 **Joint Assessment**. If the Building is not separately assessed, Real Property Taxes allocated to the Building shall be an equitable proportion of the Real Property Taxes for all of the land and improvements included within the tax parcel assessed, such proportion to be determined by Lessor from the respective valuations assigned in the assessor's work sheets or such other information as may be reasonably available. Lessor's reasonable determination thereof, in good faith, shall be conclusive.
- 10.5 **Personal Property Taxes**. Lessee shall pay prior to delinquency all taxes assessed against and levied upon Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all personal property of Lessee contained in the Premises. When possible, Lessee shall cause its Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all other personal property to be assessed and billed separately from the real property of Lessor. If any of Lessee's said property shall be assessed with Lessor's real property, Lessee shall pay Lessor the taxes attributable to Lessee's property within 10 days after receipt of a written statement setting forth the taxes applicable to Lessee's property.

11. Utilities and Services.

- 11.1 Services Provided by Lessor. Lessor shall provide heating, ventilation, air conditioning, reasonable amounts of electricity for normal lighting and office machines, water for reasonable and normal drinking and lavatory use in connection with an office, and replacement light bulbs and/or fluorescent tubes and ballasts for standard overhead fixtures. Lessor shall also provide janitorial services to the Premises and Common Areas 5 times per week, excluding Building Holidays, or pursuant to the attached janitorial schedule, if any. Lessor shall not, however, be required to provide janitorial services to kitchens or storage areas included within the Premises.
- 11.2 Services Exclusive to Lessee. Notwithstanding the provisions of paragraph 11.1, Lessee shall pay for all water, gas, light, power, telephone and other utilities and services specially or exclusively supplied and/or metered exclusively to the Premises or to Lessee, together with any taxes thereon. Notwithstanding the provisions of Paragraph 4.2(vi), if a service is deleted by Paragraph 1.13 and such service is not separately metered to the Premises, Lessee shall pay at Lessor's option, either Lessee's Share or a reasonable proportion to be determined by Lessor of all charges for such jointly metered service.
- 11.3 Hours of Service. Said services and utilities shall be provided during times set forth in Paragraph 1.12. Utilities and services required at other times shall be subject to advance request and reimbursement by Lessee to Lessor of the cost thereof.
- 11.4 Excess Usage by Lessee. Lessee shall not make connection to the utilities except by or through existing outlets and shall not install or use machinery or equipment in or about the Premises that uses excess water, lighting or power, or suffer or permit any act that causes extra burden upon the utilities or services, including but not limited to security and trash services, over standard office usage for the Project. Lessor shall require Lessee to reimburse Lessor for any excess expenses or costs that may arise out of a breach of this subparagraph by Lessee. Lessor may, in its sole discretion, install at Lessee's expense supplemental equipment and/or separate metering applicable to Lessee's excess usage or loading.
- 11.5 Interruptions. There shall be no abatement of rent and Lessor shall not be liable in any respect whatsoever for the inadequacy, stoppage, interruption or discontinuance of any utility or service due to riot, strike, labor dispute, breakdown, accident, repair or other cause beyond Lessor's reasonable control or in cooperation with governmental request or directions.
- 11.6 Within fifteen days of Lessor's written request, Lessee agrees to deliver to Lessor such information, documents and/or authorization as Lessor needs in order for Lessor to comply with new or existing Applicable Requirements relating to commercial building energy usage, ratings, and/or the reporting thereof.

12. Assignment and Subletting.

12.1 Lessor's Consent Required.

- (a) Lessee shall not voluntarily or by operation of law assign, transfer, mortgage or encumber (collectively, "assign or assignment") or sublet all or any part of Lessee's interest in this Lease or in the Premises without Lessor's prior written consent.
- (b) Unless Lessee is a corporation and its stock is publicly traded on a national stock exchange, a change in the control of Lessee shall constitute an assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of the voting control of Lessee shall constitute a change in control for this purpose.
- (c) The involvement of Lessee or its assets in any transaction, or series of transactions (by way of merger, sale, acquisition, financing, transfer, leveraged buyout or otherwise), whether or not a formal assignment or hypothecation of this Lease or Lessee's assets occurs, which results or will result in a reduction of the Net Worth of Lessee by an amount greater than 25% of such Net Worth as it was represented at the time of the execution of this Lease or at the time of the most recent assignment to which Lessor has consented, or as it exists immediately prior to said transaction or transactions constituting such reduction, whichever was or is greater, shall be considered an assignment of this Lease to which Lessor may withhold its consent. "Net Worth of Lessee" shall mean the net worth of Lessee

INITIALS

(excluding any guarantors) established under generally accepted accounting principles.

- (d) An assignment or subletting without consent shall, at Lessor's option, be a Default curable after notice per Paragraph 13.1(d), or a noncurable Breach without the necessity of any notice and grace period. If Lessor elects to treat such unapproved assignment or subletting as a noncurable Breach, Lessor may either: (i) terminate this Lease, or (ii) upon 30 days written notice, increase the monthly Base Rent to 110% of the Base Rent then in effect. Further, in the event of such Breach and rental adjustment, (i) the purchase price of any option to purchase the Premises held by Lessee shall be subject to similar adjustment to 110% of the price previously in effect, and (ii) all fixed and non-fixed rental adjustments scheduled during the remainder of the Lease term shall be increased to 110% of the scheduled adjusted rent.
 - (e) Lessee's remedy for any breach of Paragraph 12.1 by Lessor shall be limited to compensatory damages and/or injunctive relief.
 - (f) Lessor may reasonably withhold consent to a proposed assignment or subletting if Lessee is in Default at the time consent is requested.
- (g) Notwithstanding the foregoing, allowing a de minimis portion of the Premises, ie. 20 square feet or less, to be used by a third party vendor in connection with the installation of a vending machine or payphone shall not constitute a subletting.

12.2 Terms and Conditions Applicable to Assignment and Subletting.

- (a) Regardless of Lessor's consent, no assignment or subletting shall: (i) be effective without the express written assumption by such assignee or sublessee of the obligations of Lessee under this Lease, (ii) release Lessee of any obligations hereunder, or (iii) alter the primary liability of Lessee for the payment of Rent or for the performance of any other obligations to be performed by Lessee.
- (b) Lessor may accept Rent or performance of Lessee's obligations from any person other than Lessee pending approval or disapproval of an assignment. Neither a delay in the approval or disapproval of such assignment nor the acceptance of Rent or performance shall constitute a waiver or estoppel of Lessor's right to exercise its remedies for Lessee's Default or Breach.
 - (c) Lessor's consent to any assignment or subletting shall not constitute a consent to any subsequent assignment or subletting.
- (d) In the event of any Default or Breach by Lessee, Lessor may proceed directly against Lessee, any Guarantors or anyone else responsible for the performance of Lessee's obligations under this Lease, including any assignee or sublessee, without first exhausting Lessor's remedies against any other person or entity responsible therefor to Lessor, or any security held by Lessor.
- (e) Each request for consent to an assignment or subletting shall be in writing, accompanied by information relevant to Lessor's determination as to the financial and operational responsibility and appropriateness of the proposed assignee or sublessee, including but not limited to the intended use and/or required modification of the Premises, if any, together with a fee of \$500 as consideration for Lessor's considering and processing said request. Lessee agrees to provide Lessor with such other or additional information and/or documentation as may be reasonably requested. (See also Paragraph 36)
- (f) Any assignee of, or sublessee under, this Lease shall, by reason of accepting such assignment, entering into such sublease, or entering into possession of the Premises or any portion thereof, be deemed to have assumed and agreed to conform and comply with each and every term, covenant, condition and obligation herein to be observed or performed by Lessee during the term of said assignment or sublease, other than such obligations as are contrary to or inconsistent with provisions of an assignment or sublease to which Lessor has specifically consented to in writing.
- (g) Lessor's consent to any assignment or subletting shall not transfer to the assignee or sublessee any Option granted to the original Lessee by this Lease unless such transfer is specifically consented to by Lessor in writing. (See Paragraph 39.2)
- 12.3 Additional Terms and Conditions Applicable to Subletting. The following terms and conditions shall apply to any subletting by Lessee of all or any part of the Premises and shall be deemed included in all subleases under this Lease whether or not expressly incorporated therein:
- (a) Lessee hereby assigns and transfers to Lessor all of Lessee's interest in all Rent payable on any sublease, and Lessor may collect such Rent and apply same toward Lessee's obligations under this Lease; provided, however, that until a Breach shall occur in the performance of Lessee's obligations, Lessee may collect said Rent. In the event that the amount collected by Lessor exceeds Lessee's then outstanding obligations any such excess shall be refunded to Lessee. Lessor shall not, by reason of the foregoing or any assignment of such sublease, nor by reason of the collection of Rent, be deemed liable to the sublessee for any failure of Lessee to perform and comply with any of Lessee's obligations to such sublessee. Lessee hereby irrevocably authorizes and directs any such sublessee, upon receipt of a written notice from Lessor stating that a Breach exists in the performance of Lessee's obligations under this Lease, to pay to Lessor all Rent due and to become due under the sublease. Sublessee shall rely upon any such notice from Lessor and shall pay all Rents to Lessor without any obligation or right to inquire as to whether such Breach exists, notwithstanding any claim from Lessee to the contrary.
- (b) In the event of a Breach by Lessee, Lessor may, at its option, require sublessee to attorn to Lessor, in which event Lessor shall undertake the obligations of the sublessor under such sublease from the time of the exercise of said option to the expiration of such sublease; provided, however, Lessor shall not be liable for any prepaid rents or security deposit paid by such sublessee to such sublessor or for any prior Defaults or Breaches of such sublessor.
 - (c) Any matter requiring the consent of the sublessor under a sublease shall also require the consent of Lessor.
 - (d) No sublessee shall further assign or sublet all or any part of the Premises without Lessor's prior written consent.
- (e) Lessor shall deliver a copy of any notice of Default or Breach by Lessee to the sublessee, who shall have the right to cure the Default of Lessee within the grace period, if any, specified in such notice. The sublessee shall have a right of reimbursement and offset from and against Lessee for any such Defaults cured by the sublessee.

13. Default; Breach; Remedies.

- 13.1 **Default; Breach**. A "**Default**" is defined as a failure by the Lessee to comply with or perform any of the terms, covenants, conditions or Rules and Regulations under this Lease. A "**Breach**" is defined as the occurrence of one or more of the following Defaults, and the failure of Lessee to cure such Default within any applicable grace period:
- (a) The abandonment of the Premises; the vacating of the Premises prior to the expiration or termination of this Lease without providing a commercially reasonable level of security, or where the coverage of the property insurance described in Paragraph 8.3 is jeopardized as a result thereof, or without providing reasonable assurances to minimize potential vandalism; or failure to deliver to Lessor exclusive possession of the entire Premises in accordance herewith prior to the expiration or termination of this Lease.
- (b) The failure of Lessee to make any payment of Rent or any Security Deposit required to be made by Lessee hereunder, whether to Lessor or to a third party, when due, to provide reasonable evidence of insurance or surety bond, or to fulfill any obligation under this Lease which endangers or threatens life or property, where such failure continues for a period of 3 business days following written notice to Lessee. THE ACCEPTANCE BY LESSOR OF A PARTIAL PAYMENT OF RENT OR SECURITY DEPOSIT SHALL NOT CONSTITUTE A WAIVER OF ANY OF LESSOR'S RIGHTS, INCLUDING LESSOR'S RIGHT TO RECOVER POSSESSION OF THE
- (c) The failure of Lessee to allow Lessor and/or its agents access to the Premises or the commission of waste, act or acts constituting public or private nuisance, and/or an illegal activity on the Premises by Lessee, where such actions continue for a period of 3 business days following written notice to Lessee. In the event that Lessee commits waste, a nuisance or an illegal activity a second time then, the Lessor may elect to treat such conduct as a non-curable Breach rather than a
- (d) The failure by Lessee to provide (i) reasonable written evidence of compliance with Applicable Requirements, (ii) the service contracts, (iii) the rescission of an unauthorized assignment or subletting, (iv) an Estoppel Certificate or financial statements, (v) a requested subordination, (vi) evidence concerning any guaranty and/or Guarantor, (vii) any document requested under Paragraph 41, (viii) material safety data sheets (MSDS), or (ix) any other documentation or information which Lessor may reasonably require of Lessee under the terms of this Lease, where any such failure continues for a period of 10 days following written notice to Lessee.
- (e) A Default by Lessee as to the terms, covenants, conditions or provisions of this Lease, or of the rules adopted under Paragraph 2.9 hereof, other than those described in subparagraphs 13.1(a), (b) or (c), above, where such Default continues for a period of 30 days after written notice; provided, however, that if the nature of Lessee's Default is such that more than 30 days are reasonably required for its cure, then it shall not be deemed to be a Breach if Lessee commences such cure within said 30 day period and thereafter diligently prosecutes such cure to completion.
- (f) The occurrence of any of the following events: (i) the making of any general arrangement or assignment for the benefit of creditors; (ii) becoming a "debtor" as defined in 11 U.S.C. § 101 or any successor statute thereto (unless, in the case of a petition filed against Lessee, the same is dismissed within 60 days); (iii) the appointment of a trustee or receiver to take possession of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where possession is not restored to Lessee within 30 days; or (iv) the attachment, execution or other judicial seizure of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where such seizure is not discharged within 30 days; provided, however, in the event that any provision of this subparagraph is contrary to any applicable law, such provision shall be of no force or effect, and not affect the validity of the remaining provisions.
 - (g) The discovery that any financial statement of Lessee or of any Guarantor given to Lessor was materially false.
- (h) If the performance of Lessee's obligations under this Lease is guaranteed: (i) the death of a Guarantor, (ii) the termination of a Guarantor's liability with respect to this Lease other than in accordance with the terms of such guaranty, (iii) a Guarantor's becoming insolvent or the subject of a bankruptcy filing, (iv) a Guarantor's refusal to honor the guaranty, or (v) a Guarantor's breach of its guaranty obligation on an anticipatory basis, and Lessee's failure, within 60 days following written notice of any such event, to provide written alternative assurance or security, which, when coupled with the then existing resources of Lessee, equals or exceeds the combined financial resources of Lessee and the Guarantors that existed at the time of execution of this Lease.
 - 13.2 Remedies. If Lessee fails to perform any of its affirmative duties or obligations, within 10 days after written notice (or in case of an emergency, without

|--|

notice), Lessor may, at its option, perform such duty or obligation on Lessee's behalf, including but not limited to the obtaining of reasonably required bonds, insurance policies, or governmental licenses, permits or approvals. Lessee shall pay to Lessor an amount equal to 115% of the costs and expenses incurred by Lessor in such performance upon receipt of an invoice therefor. In the event of a Breach, Lessor may, with or without further notice or demand, and without limiting Lessor in the exercise of any right or remedy which Lessor may have by reason of such Breach:

- (a) Terminate Lessee's right to possession of the Premises by any lawful means, in which case this Lease shall terminate and Lessee shall immediately surrender possession to Lessor. In such event Lessor shall be entitled to recover from Lessee: (i) the unpaid Rent which had been earned at the time of termination; (ii) the worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that the Lessee proves could have been reasonably avoided; (iii) the worth at the time of award of the amount by which the unpaid rent for the balance of the term after the time of award exceeds the amount of such rental loss that the Lessee proves could be reasonably avoided; and (iv) any other amount necessary to compensate Lessor for all the detriment proximately caused by the Lessee's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom, including but not limited to the cost of recovering possession of the Premises, expenses of reletting, including necessary renovation and alteration of the Premises, reasonable attorneys' fees, and that portion of any leasing commission paid by Lessor in connection with this Lease applicable to the unexpired term of this Lease. The worth at the time of award of the amount referred to in provision (iii) of the immediately preceding sentence shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of the District within which the Premises are located at the time of award plus one percent. Efforts by Lessor to mitigate damages caused by Lessee's Breach of this Lease shall not waive Lessor's right to recover any damages to which Lessor is otherwise entitled. If termination of this Lease is obtained through the provisional remedy of unlawful detainer, Lessor shall have the right to recover in such proceeding any unpaid Rent and damages as are recoverable therein, or Lessor may reserve the right to recover all or any part thereof in a separate suit. If a notice and grace period required under Paragraph 13.1 was not previously given, a notice to pay rent or quit, or to perform or quit given to Lessee under the unlawful detainer statute shall also constitute the notice required by Paragraph 13.1. In such case, the applicable grace period required by Paragraph 13.1 and the unlawful detainer statute shall run concurrently, and the failure of Lessee to cure the Default within the greater of the two such grace periods shall constitute both an unlawful detainer and a Breach of this Lease entitling Lessor to the remedies provided for in this Lease and/or by said statute.
- (b) Continue the Lease and Lessee's right to possession and recover the Rent as it becomes due, in which event Lessee may sublet or assign, subject only to reasonable limitations. Acts of maintenance, efforts to relet, and/or the appointment of a receiver to protect the Lessor's interests, shall not constitute a termination of the Lessee's right to possession.
- (c) Pursue any other remedy now or hereafter available under the laws or judicial decisions of the state wherein the Premises are located. The expiration or termination of this Lease and/or the termination of Lessee's right to possession shall not relieve Lessee from liability under any indemnity provisions of this Lease as to matters occurring or accruing during the term hereof or by reason of Lessee's occupancy of the Premises.
- 13.3 Inducement Recapture. Any agreement for free or abated rent or other charges, the cost of tenant improvements for Lessee paid for or performed by Lessor, or for the giving or paying by Lessor to or for Lessee of any cash or other bonus, inducement or consideration for Lessee's entering into this Lease, all of which concessions are hereinafter referred to as "Inducement Provisions," shall be deemed conditioned upon Lessee's full and faithful performance of all of the terms, covenants and conditions of this Lease. Upon Breach of this Lease by Lessee, any such Inducement Provision shall automatically be deemed deleted from this Lease and of no further force or effect, and any rent, other charge, bonus, inducement or consideration theretofore abated, given or paid by Lessor under such an Inducement Provision shall be immediately due and payable by Lessee to Lessor, notwithstanding any subsequent cure of said Breach by Lessee. The acceptance by Lessor of rent or the cure of the Breach which initiated the operation of this paragraph shall not be deemed a waiver by Lessor of the provisions of this paragraph unless specifically so stated in writing by Lessor at the time of such acceptance.
- 13.4 Late Charges. Lessee hereby acknowledges that late payment by Lessee of Rent will cause Lessor to incur costs not contemplated by this Lease, the exact amount of which will be extremely difficult to ascertain. Such costs include, but are not limited to, processing and accounting charges, and late charges which may be imposed upon Lessor by any Lender. Accordingly, if any Rent shall not be received by Lessor within 5 days after such amount shall be due, then, without any requirement for notice to Lessee, Lessee shall immediately pay to Lessor a one-time late charge equal to 10% of each such overdue amount or \$100, whichever is greater. The parties hereby agree that such late charge represents a fair and reasonable estimate of the costs Lessor will incur by reason of such late payment. Acceptance of such late charge by Lessor shall in no event constitute a waiver of Lessee's Default or Breach with respect to such overdue amount, nor prevent the exercise of any of the other rights and remedies granted hereunder. In the event that a late charge is payable hereunder, whether or not collected, for 3 consecutive installments of Base Rent, then notwithstanding any provision of this Lease to the contrary, Base Rent shall, at Lessor's option, become due and payable quarterly in advance.
- 13.5 Interest. Any monetary payment due Lessor hereunder, other than late charges, not received by Lessor, when due shall bear interest from the 31st day after it was due. The interest ("Interest") charged shall be computed at the rate of 10% per annum but shall not exceed the maximum rate allowed by law. Interest is payable in addition to the potential late charge provided for in Paragraph 13.4.

13.6 Breach by Lessor.

- (a) **Notice of Breach**. Lessor shall not be deemed in breach of this Lease unless Lessor fails within a reasonable time to perform an obligation required to be performed by Lessor. For purposes of this Paragraph, a reasonable time shall in no event be less than 30 days after receipt by Lessor, and any Lender whose name and address shall have been furnished to Lessee in writing for such purpose, of written notice specifying wherein such obligation of Lessor has not been performed; provided, however, that if the nature of Lessor's obligation is such that more than 30 days are reasonably required for its performance, then Lessor shall not be in breach if performance is commenced within such 30 day period and thereafter diligently pursued to completion.
- (b) **Performance by Lessee on Behalf of Lessor.** In the event that neither Lessor nor Lender cures said breach within 30 days after receipt of said notice, or if having commenced said cure they do not diligently pursue it to completion, then Lessee may elect to cure said breach at Lessee's expense and offset from Rent the actual and reasonable cost to perform such cure, provided, however, that such offset shall not exceed an amount equal to the greater of one month's Base Rent or the Security Deposit, reserving Lessee's right to seek reimbursement from Lessor for any such expense in excess of such offset. Lessee shall document the cost of said cure and supply said documentation to Lessor.
- 14. Condemnation. If the Premises or any portion thereof are taken under the power of eminent domain or sold under the threat of the exercise of said power (collectively "Condemnation"), this Lease shall terminate as to the part taken as of the date the condemning authority takes title or possession, whichever first occurs. If more than 10% of the rentable floor area of the Premises, or more than 25% of Lessee's Reserved Parking Spaces, if any, are taken by Condemnation, Lessee may, at Lessee's option, to be exercised in writing within 10 days after Lessor shall have given Lessee written notice of such taking (or in the absence of such notice, within 10 days after the condemning authority shall have taken possession) terminate this Lease as of the date the condemning authority takes such possession. If Lessee does not terminate this Lease in accordance with the foregoing, this Lease shall remain in full force and effect as to the portion of the Premises remaining, except that the Base Rent shall be reduced in proportion to the reduction in utility of the Premises caused by such Condemnation. Condemnation awards and/or payments shall be the property of Lessor, whether such award shall be made as compensation for diminution in value of the leasehold, the value of the part taken, or for severance damages; provided, however, that Lessee shall be entitled to any compensation paid by the condemnor for Lessee's relocation expenses, loss of business goodwill and/or Trade Fixtures, without regard to whether or not this Lease is terminated pursuant to the provisions of this Paragraph. All Alterations and Utility Installations made to the Premises by Lessee, for purposes of Condemnation only, shall be considered the property of the Lessee and Lessee shall be entitled to any and all compensation which is payable therefor. In the event that this Lease is not terminated by reason of the Condemnation, Lessor shall repair any damage to the Premises caused by such Condemnation.

15. Brokerage Fees.

- 15.1 Additional Commission. In addition to the payments owed pursuant to Paragraph 1.10 above, Lessor agrees that: (a) if Lessee exercises any Option, (b) if Lessee or anyone affiliated with Lessee acquires from Lessor any rights to the Premises or other premises owned by Lessor and located within the Project, (c) if Lessee remains in possession of the Premises, with the consent of Lessor, after the expiration of this Lease, or (d) if Base Rent is increased, whether by agreement or operation of an escalation clause herein, then, Lessor shall pay Brokers a fee in accordance with the fee schedule of the Brokers in effect at the time the Lease was executed. The provisions of this paragraph are intended to supersede the provisions of any earlier agreement to the contrary.
- 15.2 **Assumption of Obligations**. Any buyer or transferee of Lessor's interest in this Lease shall be deemed to have assumed Lessor's obligation hereunder. Brokers shall be third party beneficiaries of the provisions of Paragraphs 1.10, 15, 22 and 31. If Lessor fails to pay to Brokers any amounts due as and for brokerage fees pertaining to this Lease when due, then such amounts shall accrue Interest. In addition, if Lessor fails to pay any amounts to Lessee's Broker when due, Lessee's Broker may send written notice to Lessor and Lessee of such failure and if Lessor fails to pay such amounts within 10 days after said notice, Lessee shall pay said monies to its Broker and offset such amounts against Rent. In addition, Lessee's Broker shall be deemed to be a third party beneficiary of any commission agreement entered into by and/or between Lessor and Lessor's Broker for the limited purpose of collecting any brokerage fee owed.
- 15.3 Representations and Indemnities of Broker Relationships. Lessee and Lessor each represent and warrant to the other that it has had no dealings with any person, firm, broker, agent or finder (other than the Brokers and Agents, if any) in connection with this Lease, and that no one other than said named Brokers and Agents is entitled to any commission or finder's fee in connection herewith. Lessee and Lessor do each hereby agree to indemnify, protect, defend and hold the other harmless from and against liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any dealings or actions of the indemnifying Party, including any costs, expenses, attorneys' fees reasonably incurred with respect thereto.

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

16. Estoppel Certificates.

- (a) Each Party (as "Responding Party") shall within 10 days after written notice from the other Party (the "Requesting Party") execute, acknowledge and deliver to the Requesting Party a statement in writing in form similar to the then most current "Estoppel Certificate" form published by AIR CRE, plus such additional information, confirmation and/or statements as may be reasonably requested by the Requesting Party.
- (b) If the Responding Party shall fail to execute or deliver the Estoppel Certificate within such 10 day period, the Requesting Party may execute an Estoppel Certificate stating that: (i) the Lease is in full force and effect without modification except as may be represented by the Requesting Party, (ii) there are no uncured defaults in the Requesting Party's performance, and (iii) if Lessor is the Requesting Party, not more than one month's rent has been paid in advance. Prospective purchasers and encumbrancers may rely upon the Requesting Party's Estoppel Certificate, and the Responding Party shall be estopped from denying the truth of the facts contained in said Certificate. In addition, Lessee acknowledges that any failure on its part to provide such an Estoppel Certificate will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to execute and/or deliver a requested Estoppel Certificate in a timely fashion the monthly Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for remainder of the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to provide the Estoppel Certificate. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to provide the Estoppel Certificate nor prevent the exercise of any of the other rights and remedies granted hereunder.
- (c) If Lessor desires to finance, refinance, or sell the Premises, or any part thereof, Lessee and all Guarantors shall within 10 days after written notice from Lessor deliver to any potential lender or purchaser designated by Lessor such financial statements as may be reasonably required by such lender or purchaser, including but not limited to Lessee's financial statements for the past 3 years. All such financial statements shall be received by Lessor and such lender or purchaser in confidence and shall be used only for the purposes herein set forth.
- 17. Definition of Lessor. The term "Lessor" as used herein shall mean the owner or owners at the time in question of the fee title to the Premises, or, if this is a sublease, of the Lessee's interest in the prior lease. In the event of a transfer of Lessor's title or interest in the Premises or this Lease, Lessor shall deliver to the transferee or assignee (in cash or by credit) any unused Security Deposit held by Lessor. Upon such transfer or assignment and delivery of the Security Deposit, as aforesaid, the prior Lessor shall be relieved of all liability with respect to the obligations and/or covenants under this Lease thereafter to be performed by the Lessor. Subject to the foregoing, the obligations and/or covenants in this Lease to be performed by the Lessor shall be binding only upon the Lessor as hereinabove defined.
- **18. Severability.** The invalidity of any provision of this Lease, as determined by a court of competent jurisdiction, shall in no way affect the validity of any other provision hereof.
- 19. Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Lease shall mean and refer to calendar days
- 20. Limitation on Liability. The obligations of Lessor under this Lease shall not constitute personal obligations of Lessor, or its partners, members, directors, officers or shareholders, and Lessee shall look to the Project, and to no other assets of Lessor, for the satisfaction of any liability of Lessor with respect to this Lease, and shall not seek recourse against Lessor's partners, members, directors, officers or shareholders, or any of their personal assets for such satisfaction.
- 21. Time of Essence. Time is of the essence with respect to the performance of all obligations to be performed or observed by the Parties under this Lease.
- 22. No Prior or Other Agreements; Broker Disclaimer. This Lease contains all agreements between the Parties with respect to any matter mentioned herein, and no other prior or contemporaneous agreement or understanding shall be effective. Lessor and Lessee each represents and warrants to the Brokers that it has made, and is relying solely upon, its own investigation as to the nature, quality, character and financial responsibility of the other Party to this Lease and as to the use, nature, quality and character of the Premises. Brokers have no responsibility with respect thereto or with respect to any default or breach hereof by either Party.

23. Notices.

- 23.1 Notice Requirements. All notices required or permitted by this Lease or applicable law shall be in writing and may be delivered in person (by hand or by courier) or may be sent by regular, certified or registered mail or U.S. Postal Service Express Mail, with postage prepaid, or by facsimile transmission, or by email, and shall be deemed sufficiently given if served in a manner specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specify a different address for notice, except that upon Lessee's taking possession of the Premises, the Premises shall constitute Lessee's address for notice. A copy of all notices to Lessor shall be concurrently transmitted to such party or parties at such addresses as Lessor may from time to time hereafter designate in writing.
- 23.2 Date of Notice. Any notice sent by registered or certified mail, return receipt requested, shall be deemed given on the date of delivery shown on the receipt card, or if no delivery date is shown, the postmark thereon. If sent by regular mail the notice shall be deemed given 72 hours after the same is addressed as required herein and mailed with postage prepaid. Notices delivered by United States Express Mail or overnight courier that guarantees next day delivery shall be deemed given 24 hours after delivery of the same to the Postal Service or courier. Notices delivered by hand, or transmitted by facsimile transmission or by email shall be deemed delivered upon actual receipt. If notice is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.

 23.3 Options. Notwithstanding the foregoing, in order to exercise any Options (see paragraph 39), the Notice must be sent by Certified Mail (return receipt
- 23.3 **Options.** Notwithstanding the foregoing, in order to exercise any Options (see paragraph 39), the Notice must be sent by Certified Mail (return receipt requested), Express Mail (signature required), courier (signature required) or some other methodology that provides a receipt establishing the date the notice was received by the Lessor.

24. Waivers

- (a) No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any other term, covenant or condition hereof. Lessor's consent to, or approval of, any act shall not be deemed to render unnecessary the obtaining of Lessor's consent to, or approval of, any subsequent or similar act by Lessee, or be construed as the basis of an estoppel to enforce the provision or provisions of this Lease requiring such consent.
- (b) The acceptance of Rent by Lessor shall not be a waiver of any Default or Breach by Lessee. Any payment by Lessee may be accepted by Lessor on account of monies or damages due Lessor, notwithstanding any qualifying statements or conditions made by Lessee in connection therewith, which such statements and/or conditions shall be of no force or effect whatsoever unless specifically agreed to in writing by Lessor at or before the time of deposit of such payment.
- (c) THE PARTIES AGREE THAT THE TERMS OF THIS LEASE SHALL GOVERN WITH REGARD TO ALL MATTERS RELATED THERETO AND HEREBY WAIVE THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE TO THE EXTENT THAT SUCH STATUTE IS INCONSISTENT WITH THIS LEASE.

25. Disclosures Regarding The Nature of a Real Estate Agency Relationship.

- (a) When entering into a discussion with a real estate agent regarding a real estate transaction, a Lessor or Lessee should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Lessor and Lessee acknowledge being advised by the Brokers in this transaction, as follows:
- (i) Lessor's Agent. A Lessor's agent under a listing agreement with the Lessor acts as the agent for the Lessor only. A Lessor's agent or subagent has the following affirmative obligations: To the Lessor. A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessor. To the Lessee and the Lessor. (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (ii) <u>Lessee's Agent</u>. An agent can agree to act as agent for the Lessee only. In these situations, the agent is not the Lessor's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Lessor. An agent acting only for a Lessee has the following affirmative obligations. <u>To the Lessee</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessee. <u>To the Lessee and the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (iii) Agent Representing Both Lessor and Lessee. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Lessor and the Lessee in a transaction, but only with the knowledge and consent of both the Lessor and the Lessee. In a dual agency situation, the agent has the following affirmative obligations to both the Lessor and the Lessee: (a) A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Lessor or the Lessee. (b) Other duties to the Lessor and the Lessee as stated above in subparagraphs (i) or (ii). In representing both Lessor and Lessee, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating to either Lessee's or Lessor's financial position, motivations, bargaining position, or other personal information that may impact rent, including Lessor's willingness to accept a rent less than the listing rent or Lessee's willingness to pay rent greater than the rent offered. The above duties of the agent in a real estate transaction do

INITIALS	INITIAL
© 2019 AIR CRE. All Rights Reserved.	

not relieve a Lessor or Lessee from the responsibility to protect their own interests. Lessor and Lessee should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Both Lessor and Lessee should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.

- (b) Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this Lease may be brought against Broker more than one year after the Start Date and that the liability (including court costs and attorneys' fees), of any Broker with respect to any such lawsuit and/or legal proceeding shall not exceed the fee received by such Broker pursuant to this Lease; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.
- (c) Lessor and Lessee agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- 26. No Right To Holdover. Lessee has no right to retain possession of the Premises or any part thereof beyond the expiration or termination of this Lease. At or prior to the expiration or termination of this Lease Lessee shall deliver exclusive possession of the Premises to Lessor. For purposes of this provision and Paragraph 13.1(a), exclusive possession shall mean that Lessee shall have vacated the Premises, removed all of its personal property therefrom and that the Premises have been returned in the condition specified in this Lease. In the event that Lessee does not deliver exclusive possession to Lessor as specified above, then Lessor's damages during any holdover period shall be computed at the amount of the Rent (as defined in Paragraph 4.1) due during the last full month before the expiration or termination of this Lease (disregarding any temporary abatement of Rent that may have been in effect), but with Base Rent being 150% of the Base Rent payable during such last full month. Nothing contained herein shall be construed as consent by Lessor to any holding over by Lessee.
- 27. Cumulative Remedies. No remedy or election hereunder shall be deemed exclusive but shall, wherever possible, be cumulative with all other remedies at law or in equity.
- 28. Covenants and Conditions; Construction of Agreement. All provisions of this Lease to be observed or performed by Lessee are both covenants and conditions. In construing this Lease, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Lease. Whenever required by the context, the singular shall include the plural and vice versa. This Lease shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.
- 29. Binding Effect; Choice of Law. This Lease shall be binding upon the Parties, their personal representatives, successors and assigns and be governed by the laws of the State in which the Premises are located. Any litigation between the Parties hereto concerning this Lease shall be initiated in the county in which the Premises are located. Signatures to this Lease accomplished by means of electronic signature or similar technology shall be legal and binding.

30. Subordination; Attornment; Non-Disturbance.

- 30.1 **Subordination**. This Lease and any Option granted hereby shall be subject and subordinate to any ground lease, mortgage, deed of trust, or other hypothecation or security device (collectively, "Security Device"), now or hereafter placed upon the Premises, to any and all advances made on the security thereof, and to all renewals, modifications, and extensions thereof. Lessee agrees that the holders of any such Security Devices (in this Lease together referred to as "Lender") shall have no liability or obligation to perform any of the obligations of Lessor under this Lease. Any Lender may elect to have this Lease and/or any Option granted hereby superior to the lien of its Security Device by giving written notice thereof to Lessee, whereupon this Lease and such Options shall be deemed prior to such Security Device, notwithstanding the relative dates of the documentation or recordation thereof.
- 30.2 **Attornment**. In the event that Lessor transfers title to the Premises, or the Premises are acquired by another upon the foreclosure or termination of a Security Device to which this Lease is subordinated (i) Lessee shall, subject to the non-disturbance provisions of Paragraph 30.3, attorn to such new owner, and upon request, enter into a new lease, containing all of the terms and provisions of this Lease, with such new owner for the remainder of the term hereof, or, at the election of the new owner, this Lease will automatically become a new lease between Lessee and such new owner, and (ii) Lessor shall thereafter be relieved of any further obligations hereunder and such new owner shall assume all of Lessor's obligations, except that such new owner shall not: (a) be liable for any act or omission of any prior lessor or with respect to events occurring prior to acquisition of ownership; (b) be subject to any offsets or defenses which Lessee might have against any prior lessor, (c) be bound by prepayment of more than one month's rent, or (d) be liable for the return of any security deposit paid to any prior lessor which was not paid or credited to such new owner.
- 30.3 **Non-Disturbance**. With respect to Security Devices entered into by Lessor after the execution of this Lease, Lessee's subordination of this Lease shall be subject to receiving a commercially reasonable non-disturbance agreement (a "**Non-Disturbance Agreement**") from the Lender which Non-Disturbance Agreement provides that Lessee's possession of the Premises, and this Lease, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises. Further, within 60 days after the execution of this Lease, Lessor shall, if requested by Lessee, use its commercially reasonable efforts to obtain a Non-Disturbance Agreement from the holder of any pre-existing Security Device which is secured by the Premises. In the event that Lessor is unable to provide the Non-Disturbance Agreement within said 60 days, then Lessee may, at Lessee's option, directly contact Lender and attempt to negotiate for the execution and delivery of a Non-Disturbance Agreement.
- 30.4 **Self-Executing**. The agreements contained in this Paragraph 30 shall be effective without the execution of any further documents; provided, however, that, upon written request from Lessor or a Lender in connection with a sale, financing or refinancing of the Premises, Lessee and Lessor shall execute such further writings as may be reasonably required to separately document any subordination, attornment and/or Non-Disturbance Agreement provided for herein.
- 31. Attorneys' Fees. If any Party or Broker brings an action or proceeding involving the Premises whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred. In addition, Lessor shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is subsequently commenced in connection with such Default or resulting Breach (\$200 is a reasonable minimum per occurrence for such services and consultation).
- 32. Lessor's Access; Showing Premises; Repairs. Lessor and Lessor's agents shall have the right to enter the Premises at any time, in the case of an emergency, and otherwise at reasonable times after reasonable prior notice for the purpose of showing the same to prospective purchasers, lenders, or tenants, and making such alterations, repairs, improvements or additions to the Premises as Lessor may deem necessary or desirable and the erecting, using and maintaining of utilities, services, pipes and conduits through the Premises and/or other premises as long as there is no material adverse effect on Lessee's use of the Premises. All such activities shall be without abatement of rent or liability to Lessee.
- **33. Auctions.** Lessee shall not conduct, nor permit to be conducted, any auction upon the Premises without Lessor's prior written consent. Lessor shall not be obligated to exercise any standard of reasonableness in determining whether to permit an auction.
- 34. Signs. Lessor may place on the Premises ordinary "For Sale" signs at any time and ordinary "For Lease" signs during the last 6 months of the term hereof. Lessor may not place any sign on the exterior of the Building that covers any of the windows of the Premises. Except for ordinary "For Sublease" signs which may be placed only on the Premises, Lessee shall not place any sign upon the Project without Lessor's prior written consent. All signs must comply with all Applicable Requirements.
- **35. Termination; Merger.** Unless specifically stated otherwise in writing by Lessor, the voluntary or other surrender of this Lease by Lessee, the mutual termination or cancellation hereof, or a termination hereof by Lessor for Breach by Lessee, shall automatically terminate any sublease or lesser estate in the Premises; provided, however, that Lessor may elect to continue any one or all existing subtenancies. Lessor's failure within 10 days following any such event to elect to the contrary by written notice to the holder of any such lesser interest, shall constitute Lessor's election to have such event constitute the termination of such interest.
- **36. Consents.** All requests for consent shall be in writing. Except as otherwise provided herein, wherever in this Lease the consent of a Party is required to an act by or for the other Party, such consent shall not be unreasonably withheld or delayed. Lessor's actual reasonable costs and expenses (including but not limited to architects', attorneys', engineers' and other consultants' fees) incurred in the consideration of, or response to, a request by Lessee for any Lessor consent, including but not limited to consents to an assignment, a subletting or the presence or use of a Hazardous Substance, shall be paid by Lessee upon receipt of an invoice and supporting documentation therefor. Lessor's consent to any act, assignment or subletting shall not constitute an acknowledgment that no Default or Breach by Lessee of this Lease exists, nor shall such consent be deemed a waiver of any then existing Default or Breach, except as may be otherwise specifically stated in writing by Lessor at the time of such consent. The failure to specify herein any particular condition to Lessor's consent shall not preclude the imposition by Lessor at the time of consent of such further or other conditions as are then reasonable with reference to the particular matter for which consent is being given. In the event that either Party disagrees with any determination made by the other hereunder and reasonably requests the reasons for such determination, the determining party shall furnish

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

its reasons in writing and in reasonable detail within 10 business days following such request.

37 Guarantor

- 37.1 Execution. The Guarantors, if any, shall each execute a guaranty in the form most recently published by AIR CRE.
- 37.2 **Default**. It shall constitute a Default of the Lessee if any Guarantor fails or refuses, upon request to provide: (a) evidence of the execution of the guaranty, including the authority of the party signing on Guarantor's behalf to obligate Guarantor, and in the case of a corporate Guarantor, a certified copy of a resolution of its board of directors authorizing the making of such guaranty, (b) current financial statements, (c) an Estoppel Certificate, or (d) written confirmation that the guaranty is still in effect
- **38. Quiet Possession.** Subject to payment by Lessee of the Rent and performance of all of the covenants, conditions and provisions on Lessee's part to be observed and performed under this Lease, Lessee shall have quiet possession and quiet enjoyment of the Premises during the term hereof.
- 39. Options. If Lessee is granted any option, as defined below, then the following provisions shall apply.
- 39.1 **Definition**. "**Option**" shall mean: (a) the right to extend or reduce the term of or renew this Lease or to extend or reduce the term of or renew any lease that Lessee has on other property of Lessor; (b) the right of first refusal or first offer to lease either the Premises or other property of Lessor; (c) the right to purchase, the right of first offer to purchase or the right of first refusal to purchase the Premises or other property of Lessor.
- 39.2 **Options Personal To Original Lessee**. Any Option granted to Lessee in this Lease is personal to the original Lessee, and cannot be assigned or exercised by anyone other than said original Lessee and only while the original Lessee is in full possession of the Premises and, if requested by Lessor, with Lessee certifying that Lessee has no intention of thereafter assigning or subletting.
- 39.3 **Multiple Options**. In the event that Lessee has any multiple Options to extend or renew this Lease, a later Option cannot be exercised unless the prior Options have been validly exercised.

39.4 Effect of Default on Options.

- (a) Lessee shall have no right to exercise an Option: (i) during the period commencing with the giving of any notice of Default and continuing until said Default is cured, (ii) during the period of time any Rent is unpaid (without regard to whether notice thereof is given Lessee), (iii) during the time Lessee is in Breach of this Lease, or (iv) in the event that Lessee has been given 3 or more notices of separate Default, whether or not the Defaults are cured, during the 12 month period immediately preceding the exercise of the Option.
- (b) The period of time within which an Option may be exercised shall not be extended or enlarged by reason of Lessee's inability to exercise an Option because of the provisions of Paragraph 39.4(a).
- (c) An Option shall terminate and be of no further force or effect, notwithstanding Lessee's due and timely exercise of the Option, if, after such exercise and prior to the commencement of the extended term or completion of the purchase, (i) Lessee fails to pay Rent for a period of 30 days after such Rent becomes due (without any necessity of Lessor to give notice thereof), or (ii) if Lessee commits a Breach of this Lease.
- **40. Security Measures.** Lessee hereby acknowledges that the Rent payable to Lessor hereunder does not include the cost of guard service or other security measures, and that Lessor shall have no obligation whatsoever to provide same. Lessee assumes all responsibility for the protection of the Premises, Lessee, its agents and invitees and their property from the acts of third parties. In the event, however, that Lessor should elect to provide security services, then the cost thereof shall be an Operating Expense.

41. Reservations.

- (a) Lessor reserves the right: (i) to grant, without the consent or joinder of Lessee, such easements, rights and dedications that Lessor deems necessary, (ii) to cause the recordation of parcel maps and restrictions, (iii) to create and/or install new utility raceways, so long as such easements, rights, dedications, maps, restrictions, and utility raceways do not unreasonably interfere with the use of the Premises by Lessee. Lessor may also: change the name, address or title of the Building or Project upon at least 90 days prior written notice; provide and install, at Lessee's expense, Building standard graphics on the door of the Premises and such portions of the Common Areas as Lessor shall reasonably deem appropriate; grant to any lessee the exclusive right to conduct any business as long as such exclusive right does not conflict with any rights expressly given herein; and to place such signs, notices or displays as Lessor reasonably deems necessary or advisable upon the roof, exterior of the Building or the Project or on signs in the Common Areas. Lessee agrees to sign any documents reasonably requested by Lessor to effectuate such rights. The obstruction of Lessee's view, air, or light by any structure erected in the vicinity of the Building, whether by Lessor or third parties, shall in no way affect this Lease or impose any liability upon Lessor.
- (b) Lessor also reserves the right to move Lessee to other space of comparable size in the Building or Project. Lessor must provide at least 45 days prior written notice of such move, and the new space must contain improvements of comparable quality to those contained within the Premises. Lessor shall pay the reasonable out of pocket costs that Lessee incurs with regard to such relocation, including the expenses of moving and necessary stationary revision costs. In no event, however, shall Lessor be required to pay an amount in excess of two months Base Rent. Lessee may not be relocated more than once during the term of this lease.
- (c) Lessee shall not: (i) use a representation (photographic or otherwise) of the Building or Project or their name(s) in connection with Lessee's business; or (ii) suffer or permit anyone, except in emergency, to go upon the roof of the Building.
- **42. Performance Under Protest.** If at any time a dispute shall arise as to any amount or sum of money to be paid by one Party to the other under the provisions hereof, the Party against whom the obligation to pay the money is asserted shall have the right to make payment "under protest" and such payment shall not be regarded as a voluntary payment and there shall survive the right on the part of said Party to institute suit for recovery of such sum. If it shall be adjudged that there was no legal obligation on the part of said Party to pay such sum or any part thereof, said Party shall be entitled to recover such sum or so much thereof as it was not legally required to pay. A Party who does not initiate suit for the recovery of sums paid "under protest" within 6 months shall be deemed to have waived its right to protest such payment.

43. Authority; Multiple Parties; Execution.

- (a) If either Party hereto is a corporation, trust, limited liability company, partnership, or similar entity, each individual executing this Lease on behalf of such entity represents and warrants that he or she is duly authorized to execute and deliver this Lease on its behalf. Each Party shall, within 30 days after request, deliver to the other Party satisfactory evidence of such authority.
- (b) If this Lease is executed by more than one person or entity as "Lessee", each such person or entity shall be jointly and severally liable hereunder. It is agreed that any one of the named Lessees shall be empowered to execute any amendment to this Lease, or other document ancillary thereto and bind all of the named Lessees, and Lessor may rely on the same as if all of the named Lessees had executed such document.
- (c) This Lease may be executed by the Parties in counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument.
- **44. Conflict.** Any conflict between the printed provisions of this Lease and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions.
- **45. Offer**. Preparation of this Lease by either party or their agent and submission of same to the other Party shall not be deemed an offer to lease to the other Party. This Lease is not intended to be binding until executed and delivered by all Parties hereto.
- **46. Amendments.** This Lease may be modified only in writing, signed by the Parties in interest at the time of the modification. As long as they do not materially change Lessee's obligations hereunder, Lessee agrees to make such reasonable non-monetary modifications to this Lease as may be reasonably required by a Lender in connection with the obtaining of normal financing or refinancing of the Premises.
- 47. Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY OR ARISING OUT OF THIS AGREEMENT.

	Arbitration of Disputes. An Addendum requiring the Arbitration of all disputes between the Parties and/or Brokers arising out of this Lease is is is not be to this Lease.	ot
49.	Accessibility; Americans with Disabilities Act. (a) The Premises:	
dete requ	ave not undergone an inspection by a Certified Access Specialist (CASp). Note: A Certified Access Specialist (CASp) can inspect the subject premises and nine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection between the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the occupancy of the lessee or tenant, and the occupancy of the lessee or tenant.	ion o

INITIALS

© 2019 AIR CRE. All Rights Reserved.

arrangements for the time and manner of the CASp inspection, the payment of the f correct violations of construction-related accessibility standards within the premises		
have undergone an inspection by a Certified Access Specialist (CASp) and it was standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that Lease and agrees to keep such report confidential.	determined that the Premises met all applicable construction-related accessibility it received a copy of the inspection report at least 48 hours prior to executing this	
have undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises did not meet all applicable construction-related accessibility standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this Lease and agrees to keep such report confidential except as necessary to complete repairs and corrections of violations of construction related accessibility standards.		
In the event that the Premises have been issued an inspection report by a CASp the within 7 days of the execution of this Lease.	Lessor shall provide a copy of the disability access inspection certificate to Lessee	
(b) Since compliance with the Americans with Disabilities Act (ADA) and other state and local accessibility statutes are dependent upon Lessee's specific use of the Premises, Lessor makes no warranty or representation as to whether or not the Premises comply with ADA or any similar legislation. In the event that Lessee's use of the Premises requires modifications or additions to the Premises in order to be in compliance with ADA or other accessibility statutes, Lessee agrees to make any such necessary modifications and/or additions at Lessee's expense.		
LESSOR AND LESSEE HAVE CAREFULLY READ AND REVIEWED THIS LEASE AND EACH LEASE SHOW THEIR INFORMED AND VOLUNTARY CONSENT THERETO. THE PARTIES THIS LEASE ARE COMMERCIALLY REASONABLE AND EFFECTUATE THE INTENT AND	S HEREBY AGREE THAT, AT THE TIME THIS LEASE IS EXECUTED, THE TERMS OF	
ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS LEASE OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO: 1. SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS LEASE. 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PREMISES. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING AND SIZE OF THE PREMISES, THE STRUCTURAL INTEGRITY, THE CONDITION OF THE ROOF AND OPERATING SYSTEMS, COMPLIANCE WITH THE AMERICANS WITH DISABILITIES ACT AND THE SUITABILITY OF THE PREMISES FOR LESSEE'S INTENDED USE.		
WARNING: IF THE PREMISES ARE LOCATED IN A STATE OTHER THAN CALIFORNIA, OTHE LAWS OF THE STATE IN WHICH THE PREMISES ARE LOCATED.	CERTAIN PROVISIONS OF THE LEASE MAY NEED TO BE REVISED TO COMPLY WITH	
The parties hereto have executed this Lease at the place and on the dates specified a	above their respective signatures.	
Executed at: On:	Executed at:On:	
By LESSOR:	By LESSEE:	
		
By:	By:	
Name Printed:	Name Printed:	
Title:	Title:	
Phone:	Phone:	
Fax: Email:	Fax: Email:	
By:	By:	
Name Printed:	Name Printed:	
Title:	Title:	
Phone:	Phone:	
Fax: Email:	Fax: Email:	
Enidii		
Address: Federal ID No.:	Address: Federal ID No.:	
Teacher III III III III III III III III III I	receium no.:	
BROKER	BROKER	
Attn:Title:	Attn: Title:	

Address:Phone:	Address: Phone:	
Fax:	Fax:	
Email:	Email:	
Federal ID No.:	Federal ID No.:	
Broker DRE License #: Agent DRE License #:	Broker DRE License #:	
Agent Dre License #.	Agent DRE License #:	
AIR CRE * https://www.aircre.com * 2	:13-687-8777 * contracts@aircre.com	
NOTICE: No part of these works may be reproduced in any form without permission in writing.		

INITIALS



1. Basic Provisions ("Basic Provisions").

STANDARD INDUSTRIAL/COMMERCIAL MULTI-TENANT MONTH TO MONTH LEASE - GROSS

SHORT FORM

(Only to be used for transitory tenancies)

1.1 Parties. This Lease ("Lease"), dated for reference purposes on	y, is made by and between	(" Lessor ") and
("Lessee"), (collectively the "Parties", or individually a "Party").		
1.2(a) Premises: That certain portion of the Project (as defined		
this Lease, commonly known as (street address, unit/suite, city, state):	("Premises"). The Premises are located	in the County of, and are
generally described as (describe briefly the nature of the Premises):	. In addition to Lessee's rights to use and or	ccupy the Premises as hereinafter
specified, Lessee shall have non-exclusive rights to any utility raceways of	the building containing the Premises ("Building") an	d to the Common Areas. The Premises,
the Building, the Common Areas, the land upon which they are located, al	ong with all other buildings and improvements there	eon, are herein collectively referred to as
the "Project." (See also Paragraph 2)		
1.2(b) Parking: unreserved vehicle parking space	es. (See also Paragraph 2.6)	
1.3 Term: Month-to-month commencing ("Comme	ncement Date") and ending 30 days after written no	tice by either Party to the other.
1.4 Early Possession: If the Premises are available Lessee may hav	e non-exclusive possession of the Premises commen	ncing ("Early Possession
Date"). (See also Paragraphs 3.2 and 3.3)		
1.5 Base Rent: per month ("Base Rent"), payable or	n the day of each month commencing	(See also Paragraph 4)
1.6 Base Rent and Other Monies Paid Upon Execution:		, (Samuel and Brahma)
(a) Base Rent: for the period		
(b) Security Deposit: ("Security Deposit"). (S	on also Daragraph C\	/ > '
	ee also Paragraph 3)	\sim \vee
(c) Other: for) \
(d) Total Due Upon Execution of this Lease:		
1.7 Agreed Use: (See also Paragraph 6)		
1.8 Insuring Party. Lessor is the "Insuring Party". (See also Paragr	aph 8)	
1.9 Real Estate Brokers. (See also Paragraph 15 and 25)		/ ,
(a) Representation: Each Party acknowledges receiving a Dis		confirms and consents to the following
agency relationships in this Lease with the following real estate brokers ("	Broker(s)") and/or their agents ("Agent(s)"):	/
Lessor's Brokerage Firm License No Is	the broker of (check one): the Lessor; or	both the Lessee and Lessor (dual agent).
Lessor's Agent License No is (check or	ne): the Lessor's Agent (salesperson or broker a	ssociate); or both the Lessee's
Agent and the Lessor's Agent (dual agent).		
Lessee's Brokerage Firm License No Is	s the broker of (check one): the Lessee; or	both the Lessee and Lessor (dual
agent).		·
Lessee's Agent License No. is (check o	LAND THE RESERVE TO T	hath the Larred
	ne): the Lessee's Agent (salesperson or broker)	associate); or both the Lessee's
Agent and the Lessor's Agent (dual agent).		
(b) Payment to Brokers. Upon execution and delivery of this	s Lease by both Parties, Lessor shall pay to the Broke	rs the brokerage fee agreed to in a
separate written agreement or the sum of	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	
1.10 Attachments. Attached hereto are the following, all of which of	onstitute a part of this Lease:	
an Addendum consisting of Paragraphs through) \ ;	
a site plan depicting the Premises;		
a site plan depicting the Project;		
a current set of the Rules and Regulations for the Project;	•	
a current set of the Rules and Regulations for the Project,		
other (specify):		
2. Premises.		
 Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby lea 	cas from Lassar the Dramises for the term at the re	antal and upon all of the terms, set forth
herein. While the approximate square footage of the Premises may have		
stated herein is NOT tied to square footage and is not subject to adjustme		•
the actual size prior to executing this Lease.	int should the actual size be determined to be differen	ent. NOTE. Lessee is advised to verify
2.2 Condition . Lessor shall deliver the Premises to Lessee broom of	lean and free of dehris on the Commencement Date	or the Early Possession Date, whichever
first occurs (" Start Date "), and, so long as the required service contracts d		
following the Start Date, warrants that the existing electrical, plumbing, fin		
doors, sump pumps, if any, and all other such elements, other than those		
elements of the roof, bearing walls and foundation of the Premises shall be		
or fungi defined as toxic under applicable law. If a non-compliance with su		
malfunction or fail within the appropriate warranty period, Lessor shall, property of such page compliance, malfunction or failure, rectifus compliance, rectifus com		
and extent of such non-compliance, malfunction or failure, rectify same at systems, and (ii) 30 days as to the remaining systems and other elements.		
correction of any such non-compliance, malfunction or failure shall be the sprinkler systems, roof, foundations, and/or bearing walls - see Paragraph	=	
any recorded Notices of Default affecting the Premise; (ii) any delinquent		= ::
affecting the Premises.	aniounts due under any loan secured by the Premise	es, and (iii) any bankruptcy proceeding
2.3 Compliance . Lessor warrants that to the best of its knowledge	the improvements on the Premises and the Commo	on Areas comply with the huilding codes
applicable laws, covenants or restrictions of record, regulations, and ordin		
or portion thereof, was constructed. Said warranty does not apply to the		-
Americans with Disabilities Act or any similar laws as a result of Lessee's u		
7.3(a)) made or to be made by Lessee. NOTE: Lessee is responsible for de		
appropriate for Lessee's intended use, and acknowledges that past uses		
warranty, Lessor shall, except as otherwise provided, promptly after receip		* *
non-compliance, rectify the same at Lessor's expense. If Lessee does not		-
the Start Date, correction of that non-compliance shall be the obligation of	•	,
changed so as to require during the term of this Lease the construction of		
Hazardous Substance, or the reinforcement or other physical modification		
are triggered by Lessee as a result of an actual or proposed change in use,		
shall either: (i) immediately cease such use or intensity of use and/or take		
Expenditure, (ii) complete such additions and/or alterations at its own ex		
F. J. Landy (III) Compress Such additions directly of differential de 13 0WH CA	,, - ,, p 20 day date to 20000 to term	

- 2.4 **Acknowledgements**. Lessee acknowledges that: (a) it has been given an opportunity to inspect and measure the Premises; (b) it has been advised by Lessor and/or Brokers to satisfy itself with respect to the size and condition of the Premises (including but not limited to the electrical, HVAC and fire sprinkler systems, security, environmental aspects, and compliance with Applicable Requirements and the Americans with Disabilities Act), and their suitability for Lessee's intended use; (c) Lessee has made such investigation as it deems necessary with reference to such matters and assumes all responsibility therefor as the same relate to its occupancy of the Premises; (d) it is not relying on any representation as to the size of the Premises made by Brokers or Lessor; (e) the square footage of the Premises was not material to Lessee's decision to lease the Premises and pay the Rent stated herein, and (f) neither Lessor, Lessor's agents, nor Brokers have made any oral or written representations or warranties with respect to said matters other than as set forth in this Lease. In addition, Lessor acknowledges that: (i) Brokers have made no representations, promises or warranties concerning Lessee's ability to honor the Lease or suitability to occupy the Premises, and (ii) it is Lessor's sole responsibility to investigate the financial capability and/or suitability of all proposed tenants.
- 2.5 Lessee as Prior Owner/Occupant. The warranties made by Lessor in Paragraph 2 shall be of no force or effect if immediately prior to the Start Date Lessee was the owner or occupant of the Premises. In such event, Lessee shall be responsible for any necessary corrective work.
- 2.6 **Vehicle Parking**. Lessee shall be entitled to use the number of Parking Spaces specified in Paragraph 1.2(b) on those portions of the Common Areas designated from time to time by Lessor for parking. Lessee shall not use more parking spaces than said number. Said parking spaces shall be used for parking by vehicles no larger than full-size passenger automobiles or pick-up trucks, herein called "**Permitted Size Vehicles**." Lessor may regulate the loading and unloading of vehicles by adopting Rules and Regulations as provided in Paragraph 2.9. No vehicles other than Permitted Size Vehicles may be parked in the Common Area without the prior written permission of Lessor. In addition:
- (a) Lessee shall not permit or allow any vehicles that belong to or are controlled by Lessee or Lessee's employees, suppliers, shippers, customers, contractors or invitees to be loaded, unloaded, or parked in areas other than those designated by Lessor for such activities.
 - (b) Lessee shall not service or store any vehicles in the Common Areas.
- (c) If Lessee permits or allows any of the prohibited activities described in this Paragraph 2.6, then Lessor shall have the right, without notice, to remove or tow away the vehicle involved and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.
- 2.7 **Common Areas Definition**. The term "**Common Areas**" is defined as all areas and facilities outside the Premises and within the exterior boundary line of the Project and interior utility raceways and installations within the Premises that are designated for the general non-exclusive use of Lessor, Lessee and other tenants of the Project and their respective employees, suppliers, shippers, customers, contractors and invitees, including parking areas, loading and unloading areas, trash areas, roofs, roadways, walkways, driveways and landscaped areas.
- 2.8 **Common Areas Lessee's Rights.** Lessor grants to Lessee, for the benefit of Lessee and its employees, suppliers, shippers, contractors, customers and invitees, during the term of this Lease, the non-exclusive right to use, in common with others' use, the Common Areas, subject to any rules and regulations or restrictions governing the use of the Project. Under no circumstances shall the right herein granted to use the Common Areas be deemed to include the right to store any property, temporarily or permanently, in the Common Areas. In the event that any unauthorized storage shall occur, then Lessor shall have the right, without notice, to remove the property and charge the cost to Lessee.
- 2.9 **Common Areas Rules and Regulations**. Lessor shall have the exclusive control and management of the Common Areas and shall have the right to establish, modify, amend and enforce reasonable rules and regulations ("**Rules and Regulations**") for the management of the Common Areas. Lessee agrees to abide by and conform to all such Rules and Regulations, and shall use its best efforts to cause its employees, suppliers, shippers, customers, contractors and invitees to so abide and conform. Lessor shall not be responsible to Lessee for the non-compliance with said Rules and Regulations by other tenants of the Project.
 - 2.10 Common Areas Changes. Lessor shall have the right, in Lessor's sole discretion:
 - (a) To make changes to the Common Areas;
 - (b) To close temporarily any of the Common Areas for maintenance purposes so long as reasonable access to the Premises remains available;
 - (c) To use the Common Areas while engaged in making repairs or alterations to the Project; and
 - (d) To do and perform such other acts and make such other changes in, to or with respect to the Common Areas as Lessor may deem to be appropriate.

Term.

- 3.1 **Term**. This is a month-to-month Lease. The Commencement Date is specified in Paragraph 1.3.
- 3.2 **Early Possession**. Any provision herein granting Lessee Early Possession of the Premises is subject to and conditioned upon the Premises being available for such possession prior to the Commencement Date. Any grant of Early Possession only conveys a non-exclusive right to occupy the Premises. If Lessee totally or partially occupies the Premises prior to the Commencement Date, the obligation to provide insurance and to maintain the Premises shall be in effect during such period.
- 3.3 **Delay In Possession**. Lessor agrees to use commercially reasonable efforts to deliver exclusive possession of the Premises to Lessee by the Commencement Date. If, despite said efforts, Lessor is unable to deliver possession by such date, Lessor shall not be subject to any liability therefor. Lessee shall not, however, be obligated to pay Rent or perform its other obligations until Lessor delivers possession of the Premises and any period of rent abatement that Lessee would otherwise have enjoyed shall run from the date of delivery of possession and continue for a period equal to what Lessee would otherwise have enjoyed under the terms hereof, but minus any days of delay caused by the acts or omissions of Lessee. If possession is not delivered within 60 days after the Commencement Date, as the same may be extended under the terms of any Work Letter executed by Parties, Lessee may, at its option, by notice in writing within 10 days after the end of such 60 day period, cancel this Lease. If such written notice is not received by Lessor within said 10 day period, Lessee's right to cancel shall terminate. If possession of the Premises is not delivered within 120 days after the Commencement Date, this Lease shall terminate unless other agreements are reached between Lessor and Lessee, in writing.
- 3.4 Lessee Compliance. Lessor shall not be required to tender possession of the Premises to Lessee until Lessee complies with its obligation to provide evidence of insurance (Paragraph 8.4). Pending delivery of such evidence, Lessee shall be required to perform all of its obligations under this Lease from and after the Start Date, including the payment of Rent, notwithstanding Lessor's election to withhold possession pending receipt of such evidence of insurance.

4. Rent

- 4.1 Rent Defined. All monetary obligations of Lessee to Lessor under the terms of this Lease (except for the Security Deposit) are deemed to be rent ("Rent").
- 4.2 **Payment**. Lessee shall cause payment of Rent to be received by Lessor in lawful money of the United States, without offset or deduction (except as specifically permitted in this Lease), on or before the day on which it is due. All monetary amounts shall be rounded to the nearest whole dollar. In the event that any statement or invoice prepared by Lessor is inaccurate such inaccuracy shall not constitute a waiver and Lessee shall be obligated to pay the amount set forth in this Lease. Rent for any period during the term hereof which is for less than one full calendar month shall be prorated based upon the actual number of days of said month. Payment of Rent shall be made to Lessor at its address stated herein or to such other persons or place as Lessor may from time to time designate in writing. Acceptance of a payment which is less than the amount then due shall not be a waiver of Lessor's rights to the balance of such Rent, regardless of Lessor's endorsement of any check so stating. In the event that any check, draft, or other instrument of payment given by Lessee to Lessor is dishonored for any reason, Lessee agrees to pay to Lessor the sum of \$25 in addition to any Late Charge and Lessor, at its option, may require all future Rent be paid by cashier's check. Payments will be applied first to accrued late charges and attorney's fees, second to accrued interest, then to Base Rent and Common Area Operating Expenses, and any remaining amount to any other outstanding charges or costs.
- 5. Security Deposit. Lessee shall deposit with Lessor upon execution hereof the Security Deposit as security for Lessee's faithful performance of its obligations under this Lease. If Lessee fails to pay Rent, or otherwise Defaults under this Lease, Lessor may use, apply or retain all or any portion of said Security Deposit for the payment of any amount already due Lessor, for Rents which will be due in the future, and/or to reimburse or compensate Lessor for any liability, expense, loss or damage which Lessor may suffer or incur by reason thereof. If Lessor uses or applies all or any portion of the Security Deposit, Lessee shall within 10 days after written request therefor deposit monies with Lessor sufficient to restore said Security Deposit to the full amount required by this Lease. Lessor shall not be required to keep the Security Deposit separate from its general accounts. Within 90 days after the expiration or termination of this Lease, Lessor shall return that portion of the Security Deposit not used or applied by Lessor. Lessor shall upon written request provide Lessee with an accounting showing how that portion of the Security Deposit that was not returned was applied. No part of the Security Deposit shall be considered to be held in trust, to bear interest or to be prepayment for any monies to be paid by Lessee under this Lease. THE SECURITY DEPOSIT SHALL NOT BE USED BY LESSEE IN LIEU OF PAYMENT OF THE LAST MONTH'S RENT.

6. Use

- 6.1 **Use**. Lessee shall use and occupy the Premises only for the Agreed Use, and for no other purpose. Lessee shall not use or permit the use of the Premises in a manner that is unlawful, creates damage, waste or a nuisance, or that disturbs occupants of or causes damage to neighboring premises or properties. Other than guide, signal and seeing eye dogs, Lessee shall not keep or allow in the Premises any pets, animals, birds, fish, or reptiles.
 - 6.2 Hazardous Substances
- (a) **Reportable Uses Require Consent**. The term "Hazardous Substance" as used in this Lease shall mean any product, substance, or waste whose presence, use, manufacture, disposal, transportation, or release, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentially injurious to the public health, safety or welfare, the environment or the Premises; (ii) regulated or monitored by any governmental authority, or (iii) a basis for potential liability of Lessor to any governmental agency or third party under any applicable statute or common law theory. Hazardous Substances shall include, but

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

not be limited to, hydrocarbons, petroleum, gasoline, and/or crude oil or any products, by-products or fractions thereof. Lessee shall not engage in any activity in or on the Premises which constitutes a Reportable Use of Hazardous Substances without the express prior written consent of Lessor and timely compliance (at Lessee's expense) with all Applicable Requirements. "Reportable Use" shall mean (i) the installation or use of any above or below ground storage tank; (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and/or (iii) the presence at the Premises of a Hazardous Substance with respect to which any Applicable Requirements requires that a notice be given to persons entering or occupying the Premises or neighboring properties. Notwithstanding the foregoing, Lessee may use any ordinary and customary materials reasonably required to be used in the normal course of the Agreed Use, ordinary office supplies (copier toner, liquid paper, glue, etc.) and common household cleaning materials, so long as such use is in compliance with all Applicable Requirements, is not a Reportable Use, and does not expose the Premises or neighboring property to any meaningful risk of contamination or damage or expose Lessor to any liability therefor. In addition, Lessor may condition its consent to any Reportable Use upon receiving such additional assurances as Lessor reasonably deems necessary to protect itself, the public, the Premises and/or the environment against damage, contamination, injury and/or liability, including, but not limited to, the installation (and removal on or before Lease expiration or termination) of protective modifications (such as concrete encasements) and/or increasing the Security Deposit.

- (b) **Duty to Inform Lessor**. If Lessee knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises, other than as previously consented to by Lessor, Lessee shall immediately give written notice of such fact to Lessor, and provide Lessor with a copy of any report, notice, claim or other documentation which it has concerning the presence of such Hazardous Substance.
- (c) Lessee Remediation. Lessee shall not cause or permit any Hazardous Substance to be spilled or released in, on, under, or about the Premises (including through the plumbing or sanitary sewer system) and shall promptly, at Lessee's expense, comply with all Applicable Requirements and take all investigatory and/or remedial action reasonably recommended, whether or not formally ordered or required, for the cleanup of any contamination of, and for the maintenance, security and/or monitoring of the Premises or neighboring properties, that was caused or materially contributed to by Lessee, or pertaining to or involving any Hazardous Substance brought onto the Premises during the term of this Lease, by or for Lessee, or any third party.
- (d) Lessee Indemnification. Lessee shall indemnify, defend and hold Lessor, its agents, employees, lenders and ground lessor, if any, harmless from and against any and all loss of rents and/or damages, liabilities, judgments, claims, expenses, penalties, and attorneys' and consultants' fees arising out of or involving any Hazardous Substance brought onto the Premises by or for Lessee, or any third party (provided, however, that Lessee shall have no liability under this Lease with respect to underground migration of any Hazardous Substance under the Premises from areas outside of the Project not caused or contributed to by Lessee). Lessee's obligations shall survive the expiration or termination of this Lease. No termination, cancellation or release agreement entered into by Lessor and Lessee shall release Lessee from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Lessor in writing at the time of such agreement.
- (e) Lessor Indemnification. Except as otherwise provided in Paragraph 8.6, Lessor and its successors and assigns shall indemnify, defend, reimburse and hold Lessee, its employees and lenders, harmless from and against any and all environmental damages, including the cost of remediation, which are suffered as a direct result of Hazardous Substances on the Premises prior to Lessee taking possession or which are caused by the gross negligence or willful misconduct of Lessor, its agents or employees. Lessor's obligations, as and when required by the Applicable Requirements, shall include, but not be limited to, the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease.
- (f) Investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by governmental entities having jurisdiction with respect to the existence of Hazardous Substances on the Premises prior to the Lessee taking possession, unless such remediation measure is required as a result of Lessee's use (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises, in which event Lessee shall be responsible for such payment. Lessee shall cooperate fully in any such activities at the request of Lessor, including allowing Lessor and Lessor's agents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remedial responsibilities.
- 6.3 Lessee's Compliance with Applicable Requirements. Lessee shall, at Lessee's sole expense, fully, diligently and in a timely manner, materially comply with all Applicable Requirements, without regard to whether said Applicable Requirements are now in effect or become effective after the Start Date. Lessee shall, within 10 days after receipt of Lessor's written request, provide Lessor with copies of all permits and other documents, and other information evidencing Lessee's compliance with any Applicable Requirements specified by Lessor, and shall immediately upon receipt, notify Lessor in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving the failure of Lessee or the Premises to comply with any Applicable Requirements. Likewise, Lessee shall immediately give written notice to Lessor of: (i) any water damage to the Premises and any suspected seepage, pooling, dampness or other condition conducive to the production of mold; or (ii) any mustiness or other odors that might indicate the presence of mold in the Premises.

 6.4 Inspection; Compliance. Lessor and Lessor's consultants authorized by Lessor shall have the right to enter into Premises at any time in the case of an
- emergency, and otherwise at reasonable times after reasonable notice, for the purpose of inspecting and/or testing the condition of the Premises and/or for verifying compliance by Lessee with this Lease. The cost of any such inspections shall be paid by Lessor, unless a violation of Applicable Requirements, or a Hazardous Substance is found to exist or be imminent, or the inspection is requested or ordered by a governmental authority. In such case, Lessee shall upon request reimburse Lessor for the cost of such inspection, so long as such inspection is reasonably related to the violation or contamination. In addition, Lessee shall provide copies of all relevant material safety data sheets (MSDS) to Lessor within 10 days of the receipt of written request therefor. Lessee acknowledges that any failure on its part to allow such inspections or testing will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to allow such inspections and/or testing in a timely fashion the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for the remainder to the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to allow such inspection and/or testing. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to such failure nor prevent the exercise of any of the other rights and remedies granted hereunder.

7. Maintenance; Repairs; Utility Installations; Trade Fixtures and Alterations.

7.1 Lessee's Obligations.

- (a) In General. Subject to the provisions of Paragraph 2.2 (Condition), 2.3 (Compliance), 6.3 (Lessee's Compliance with Applicable Requirements), 7.2 (Lessor's Obligations), 9 (Damage or Destruction), and 14 (Condemnation), Lessee shall, at Lessee's sole expense, keep the Premises, Utility Installations (intended for Lessee's exclusive use, no matter where located), and Alterations in good order, condition and repair (whether or not the portion of the Premises requiring repairs, or the means of repairing the same, are reasonably or readily accessible to Lessee, and whether or not the need for such repairs occurs as a result of Lessee's use, any prior use, the elements or the age of such portion of the Premises), including, but not limited to, all equipment or facilities, such as plumbing, HVAC equipment, electrical, lighting facilities, fixtures, interior walls, interior surfaces of exterior walls, ceilings, floors, windows, doors, plate glass, and skylights but excluding any items which are the responsibility of Lessor pursuant to Paragraph 7.2. Lessee, in keeping the Premises in good order, condition and repair, shall exercise and perform good maintenance practices, specifically including the procurement and maintenance of the service contracts required by Paragraph 7.1(b) below. Lessee's obligations shall include restorations, replacements or renewals when necessary to keep the Premises and all improvements thereon or a part thereof in good order, condition and state of repair.
- (b) Service Contracts. Lessee shall, at Lessee's sole expense, procure and maintain contracts, with copies to Lessor, in customary form and substance for, and with contractors specializing and experienced in the maintenance of the following equipment and improvements, if any, if and when installed on the Premises: (i) HVAC equipment, and (ii) clarifiers. However, Lessor reserves the right, upon notice to Lessee, to procure and maintain any or all of such service contracts, and Lessee shall reimburse Lessor, upon demand, for the cost thereof.
- (c) Failure to Perform. If Lessee fails to perform Lessee's obligations under this Paragraph 7.1, Lessor may enter upon the Premises after 10 days' prior written notice to Lessee (except in the case of an emergency, in which case no notice shall be required), perform such obligations on Lessee's behalf, and put the Premises in good order, condition and repair, and Lessee shall promptly pay to Lessor a sum equal to 115% of the cost thereof.
- (d) **Replacement**. Subject to Lessee's indemnification of Lessor as set forth in Paragraph 8.6 below, and without relieving Lessee of liability resulting from Lessee's failure to exercise and perform good maintenance practices, if an item described in Paragraph 7.1(b) cannot be repaired other than at a cost which is in excess of 50% of the cost of replacing such item, then such item shall be replaced by Lessor.
- 7.2 **Lessor's Obligations**. Subject to the provisions of Paragraphs 2.2 (Condition), 2.3 (Compliance), 6 (Use), 7.1 (Lessee's Obligations), 9 (Damage or Destruction) and 14 (Condemnation), Lessor shall keep in good order, condition and repair the foundations, exterior walls, structural condition of interior bearing walls, exterior roof, fire sprinkler system, Common Area fire alarm and/or smoke detection systems, fire hydrants, parking lots, walkways, driveways, landscaping, fences, signs and utility systems serving the Common Areas and all parts thereof. Lessor shall not be obligated to paint the exterior or interior surfaces of exterior walls nor shall Lessor be obligated to maintain, repair or replace windows, doors or plate glass of the Premises.

7.3 Utility Installations; Trade Fixtures; Alterations.

- (a) **Definitions**. The term "**Utility Installations**" refers to all floor and window coverings, air and/or vacuum lines, power panels, electrical distribution, security and fire protection systems, communication cabling, lighting fixtures, HVAC equipment, plumbing, and fencing in or on the Premises. The term "**Trade Fixtures**" shall mean Lessee's machinery and equipment that can be removed without doing material damage to the Premises. The term "Alterations" shall mean any modification of the improvements, other than Utility Installations or Trade Fixtures, whether by addition or deletion. "**Lessee Owned Alterations and/or Utility Installations**" are defined as Alterations and/or Utility Installations made by Lessee that are not yet owned by Lessor pursuant to Paragraph 7.4(a).
 - (b) Consent. Lessee shall not make any Alterations or Utility Installations to the Premises without Lessor's prior written consent. Any Alterations or

Utility Installations that Lessee shall desire to make shall be presented to Lessor in written form with detailed plans. Consent shall be deemed conditioned upon Lessee's: (i) acquiring all applicable governmental permits, (ii) furnishing Lessor with copies of both the permits and the plans and specifications prior to commencement of the work, and (iii) compliance with all conditions of said permits and other Applicable Requirements in a prompt and expeditious manner. Any Alterations or Utility Installations shall be performed in a workmanlike manner with good and sufficient materials. For work which costs an amount in excess of one month's Base Rent, Lessor may condition its consent upon Lessee providing a lien and completion bond in an amount equal to 150% of the estimated cost of such Alteration or Utility Installation and/or upon Lessee's posting an additional Security Deposit with Lessor.

(c) Liens; Bonds. Lessee shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lessee at or for use on the Premises, which claims are or may be secured by any mechanic's or materialmen's lien against the Premises or any interest therein. Lessee shall give Lessor not less than 10 days notice prior to the commencement of any work in, on or about the Premises, and Lessor shall have the right to post notices of non-responsibility. If Lessee shall contest the validity of any such lien, claim or demand, then Lessee shall, at its sole expense defend and protect itself, Lessor and the Premises against the same and shall pay and satisfy any such adverse judgment that may be rendered thereon before the enforcement thereof. If Lessor shall require, Lessee shall furnish a surety bond in an amount equal to 150% of the amount of such contested lien, claim or demand, indemnifying Lessor against liability for the same. If Lessor elects to participate in any such action, Lessee shall pay Lessor's attorneys' fees and costs.

7.4 Ownership; Removal; Surrender; and Restoration.

- (a) **Ownership.** Subject to Lessor's right to require removal or elect ownership as hereinafter provided, all Alterations and Utility Installations made by Lessee shall be the property of Lessee, but considered a part of the Premises. Lessor may, at any time, elect in writing to be the owner of all or any specified part of the Lessee Owned Alterations and Utility Installations. Unless otherwise instructed per paragraph 7.4(b) hereof, all Lessee Owned Alterations and Utility Installations shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.
- (b) **Removal**. By delivery to Lessee of written notice from Lessor prior to the end of the term of this Lease, Lessor may require that any or all Lessee Owned Alterations or Utility Installations be removed by the expiration or termination of this Lease. Lessor may require the removal at any time of all or any part of any Lessee Owned Alterations or Utility Installations made without the required consent.
- (c) Surrender; Restoration. Lessee shall surrender the Premises in the same condition as delivered to Lessee on the Start Date with NO allowance for ordinary wear and tear. Lessee shall repair any damage occasioned by the installation, maintenance or removal of Trade Fixtures, Lessee owned Alterations and/or Utility Installations, furnishings, and equipment as well as the removal of any storage tank installed by or for Lessee. Lessee shall also remove from the Premises any and all Hazardous Substances brought onto the Premises by or for Lessee, or any third party (except Hazardous Substances which were deposited via underground migration from areas outside of the Premises) to the level specified in Applicable Requirements. Trade Fixtures shall remain the property of Lessee and shall be removed by Lessee. Any personal property of Lessee not removed on or before the Expiration Date or any earlier termination date shall be deemed to have been abandoned by Lessee and may be disposed of or retained by Lessor as Lessor may desire. The failure by Lessee to timely vacate the Premises pursuant to this Paragraph 7.4(c) without the express written consent of Lessor shall constitute a holdover under the provisions of Paragraph 26 below.

8. Insurance: Indemnity.

8.1 Liability Insurance.

- (a) Carried by Lessee. Lessee shall obtain and keep in force a Commercial General Liability policy of insurance protecting Lessee and Lessor as an additional insured against claims for bodily injury, personal injury and property damage based upon or arising out of the ownership, use, occupancy or maintenance of the Premises and all areas appurtenant thereto. Such insurance shall be on an occurrence basis providing single limit coverage in an amount not less than \$1,000,000 per occurrence with an annual aggregate of not less than \$2,000,000. Lessee shall add Lessor as an additional insured by means of an endorsement at least as broad as the Insurance Service Organization's "Additional Insured-Managers or Lessors of Premises" Endorsement. The policy shall not contain any intra-insured exclusions as between insured persons or organizations, but shall include coverage for liability assumed under this Lease as an "insured contract" for the performance of Lessee's indemnity obligations under this Lease. The limits of said insurance shall not, however, limit the liability of Lessee nor relieve Lessee of any obligation hereunder. Lessee shall provide an endorsement on its liability policy(ies) which provides that its insurance shall be primary to and not contributory with any similar insurance carried by Lessor, whose insurance shall be considered excess insurance only.
- (b) Carried by Lessor. Lessor shall maintain liability insurance, in addition to, and not in lieu of, the insurance required to be maintained by Lessee. Lessee shall not be named as an additional insured therein.

8.2 Property Insurance - Building, Improvements and Rental Value.

- (a) Building and Improvements. Lessor shall obtain and keep in force a policy or policies of insurance in the name of Lessor, with loss payable to Lessor, any ground-lessor, and to any Lender insuring loss or damage to the Premises. The amount of such insurance shall be equal to the full insurable replacement cost of the Premises, as the same shall exist from time to time, or the amount required by any Lender, but in no event more than the commercially reasonable and available insurable value thereof. Lessee Owned Alterations and Utility Installations, Trade Fixtures, and Lessee's personal property shall be insured by Lessee not by Lessor. If the coverage is available and commercially appropriate, such policy or policies shall insure against all risks of direct physical loss or damage (except the perils of flood and/or earthquake unless required by a Lender), including coverage for debris removal and the enforcement of any Applicable Requirements requiring the upgrading, demolition, reconstruction or replacement of any portion of the Premises as the result of a covered loss. Said policy or policies shall also contain an agreed valuation provision in lieu of any coinsurance clause, waiver of subrogation, and inflation guard protection causing an increase in the annual property insurance coverage amount by a factor of not less than the adjusted U.S. Department of Labor Consumer Price Index for All Urban Consumers for the city nearest to where the Premises are located. If such insurance coverage has a deductible clause, the deductible amount shall not exceed \$5,000 per occurrence.
- (b) **Rental Value**. Lessor shall also obtain and keep in force a policy or policies in the name of Lessor with loss payable to Lessor and any Lender, insuring the loss of the full Rent for one year with an extended period of indemnity for an additional 180 days ("Rental Value insurance"). Said insurance shall contain an agreed valuation provision in lieu of any coinsurance clause, and the amount of coverage shall be adjusted annually to reflect the projected Rent otherwise payable by Lessee, for the next 12 month period.
- (c) Adjacent Premises. Lessee shall pay for any increase in the premiums for the property insurance of the Building and for the Common Areas or other buildings in the Project if said increase is caused by Lessee's acts, omissions, use or occupancy of the Premises.
- (d) Lessee's Improvements. Since Lessor is the Insuring Party, Lessor shall not be required to insure Lessee Owned Alterations and Utility Installations unless the item in question has become the property of Lessor under the terms of this Lease.

8.3 Lessee's Property; Business Interruption Insurance; Worker's Compensation Insurance.

- (a) **Property Damage**. Lessee shall obtain and maintain insurance coverage on all of Lessee's personal property, Trade Fixtures, and Lessee Owned Alterations and Utility Installations. Such insurance shall be full replacement cost coverage with a deductible of not to exceed \$1,000 per occurrence. The proceeds from any such insurance shall be used by Lessee for the replacement of personal property, Trade Fixtures and Lessee Owned Alterations and Utility Installations.
- (b) **Business Interruption**. Lessee shall obtain and maintain loss of income and extra expense insurance in amounts as will reimburse Lessee for direct or indirect loss of earnings attributable to all perils commonly insured against by prudent lessees in the business of Lessee or attributable to prevention of access to the Premises as a result of such perils.
- (c) Worker's Compensation Insurance. Lessee shall obtain and maintain Worker's Compensation Insurance in such amount as may be required by Applicable Requirements. Such policy shall include a 'Waiver of Subrogation' endorsement. Lessee shall provide Lessor with a copy of such endorsement along with the certificate of insurance or copy of the policy required by paragraph 8.5.
- (d) No Representation of Adequate Coverage. Lessor makes no representation that the limits or forms of coverage of insurance specified herein are adequate to cover Lessee's property, business operations or obligations under this Lease.
- 8.4 Insurance Policies. Insurance required herein shall be by companies maintaining during the policy term a "General Policyholders Rating" of at least A-, VII, as set forth in the most current issue of "Best's Insurance Guide", or such other rating as may be required by a Lender. Lessee shall not do or permit to be done anything which invalidates the required insurance policies. Lessee shall, prior to the Start Date, deliver to Lessor certified copies of policies of such insurance or certificates with copies of the required endorsements evidencing the existence and amounts of the required insurance. No such policy shall be cancelable or subject to modification except after 30 days prior written notice to Lessor. Lessee shall, at least 10 days prior to the expiration of such policies, furnish Lessor with evidence of renewals or "insurance binders" evidencing renewal thereof, or Lessor may increase his liability insurance coverage and charge the cost thereof to Lessee, which amount shall be payable by Lessee to Lessor upon demand. Such policies shall be for a term of at least one year, or the length of the remaining term of this Lease, whichever is less. If either Party shall fail to procure and maintain the insurance required to be carried by it, the other Party may, but shall not be required to, procure and maintain the same.
- 8.5 **Waiver of Subrogation**. Without affecting any other rights or remedies, Lessee and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is not limited by the amount of insurance carried or required, or by any deductibles applicable hereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Lessee, as the case may be, so long as the insurance is not invalidated thereby.
- 8.6 **Indemnity**. Except for Lessor's gross negligence or willful misconduct, Lessee shall indemnify, protect, defend and hold harmless the Premises, Lessor and its agents, Lessor's master or ground lessor, partners and Lenders, from and against any and all claims, loss of rents and/or damages, liens, judgments, penalties,

INITIALS			

INITIALS

attorneys' and consultants' fees, expenses and/or liabilities arising out of, involving, or in connection with, a Breach of the Lease by Lessee and/or the use and/or occupancy of the Premises and/or Project by Lessee and/or by Lessee's employees, contractors or invitees. If any action or proceeding is brought against Lessor by reason of any of the foregoing matters, Lessee shall upon notice defend the same at Lessee's expense by counsel reasonably satisfactory to Lessor and Lessor shall cooperate with Lessee in such defense. Lessor need not have first paid any such claim in order to be defended or indemnified.

- 8.7 **Exemption of Lessor and its Agents from Liability**. Notwithstanding the negligence or breach of this Lease by Lessor or its agents, neither Lessor nor its agents shall be liable under any circumstances for: (i) injury or damage to the person or goods, wares, merchandise or other property of Lessee, Lessee's employees, contractors, invitees, customers, or any other person in or about the Premises, from any cause, whether the said injury or damage results from conditions arising upon the Premises or upon other portions of the Building, or from other sources or places; (ii) any damages arising from any act or neglect of any other tenant of Lessor or from the failure of Lessor or its agents to enforce the provisions of any other lease in the Project; or (iii) injury to Lessee's business or for any loss of income or profit therefrom. Instead, it is intended that Lessee's sole recourse in the event of such damages or injury be to file a claim on the insurance policy(ies) that Lessee is required to maintain pursuant to the provisions of paragraph 8.
- 8.8 **Failure to Provide Insurance**. Lessee acknowledges that any failure on its part to obtain or maintain the insurance required herein will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, for any month or portion thereof that Lessee does not maintain the required insurance and/or does not provide Lessor with the required binders or certificates evidencing the existence of the required insurance, the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater. The parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to maintain the required insurance. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to maintain such insurance, prevent the exercise of any of the other rights and remedies granted hereunder, nor relieve Lessee of its obligation to maintain the insurance specified in this Lease.
- 9. Damage or Destruction. If the premises are materially damaged or destroyed either party may terminate this Lease by giving at least 30 days prior written notice of such termination to the other Party.
- 10. Real Property Taxes. Lessor shall pay any Real Property Taxes.

11. Utilities and Services.

- 11.1 Lessee shall pay for all water, gas, heat, light, power, telephone, trash disposal and other utilities and services supplied to the Premises, together with any taxes thereon. If at any time in Lessor's sole judgment, Lessor determines that Lessee is using a disproportionate amount of water, electricity or other commonly metered utilities, or that Lessee is generating such a large volume of trash as to require an increase in the size of the trash receptacle and/or an increase in the number of times per month that it is emptied, then Lessor may increase Lessee's Base Rent by an amount equal to such increased costs. There shall be no abatement of Rent and Lessor shall not be liable in any respect whatsoever for the inadequacy, stoppage, interruption or discontinuance of any utility or service due to riot, strike, labor dispute, breakdown, accident, repair or other cause beyond Lessor's reasonable control or in cooperation with governmental request or directions.
- 11.2 Within fifteen days of Lessor's written request, Lessee agrees to deliver to Lessor such information, documents and/or authorization as Lessor needs in order for Lessor to comply with new or existing Applicable Requirements relating to commercial building energy usage, ratings, and/or the reporting thereof.
- 12. Assignment and Subletting. Lessee shall not voluntarily or by operation of law assign, transfer, mortgage or encumber (collectively, "assign or assignment") or sublet all or any part of Lessee's interest in this Lease or in the Premises without Lessor's prior written consent.

13. Default; Breach; Remedies.

- 13.1 **Default**; **Breach**. A "**Default**" is defined as a failure by the Lessee to comply with or perform any of the terms, covenants, conditions or Rules and Regulations under this Lease. A "**Breach**" is defined as the occurrence of one or more of the following Defaults, and the failure of Lessee to cure such Default within any applicable grace period:
- (a) The abandonment of the Premises; or the vacating of the Premises without providing a commercially reasonable level of security, or where the coverage of the property insurance described in Paragraph 8.3 is jeopardized as a result thereof, or without providing reasonable assurances to minimize potential vandalism.
- (b) The failure of Lessee to make any payment of Rent or any Security Deposit required to be made by Lessee hereunder, whether to Lessor or to a third party, when due, to provide reasonable evidence of insurance or surety bond, or to fulfill any obligation under this Lease which endangers or threatens life or property, where such failure continues for a period of 3 business days following written notice to Lessee. THE ACCEPTANCE BY LESSOR OF A PARTIAL PAYMENT OF RENT OR SECURITY DEPOSIT SHALL NOT CONSTITUTE A WAIVER OF ANY OF LESSOR'S RIGHTS, INCLUDING LESSOR'S RIGHT TO RECOVER POSSESSION OF THE PREMISES.
- (c) The failure of Lessee to allow Lessor and/or its agents access to the Premises or the commission of waste, act or acts constituting public or private nuisance, and/or an illegal activity on the Premises by Lessee, where such actions continue for a period of 3 business days following written notice to Lessee. In the event that Lessee commits waste, a nuisance or an illegal activity a second time then, the Lessor may elect to treat such conduct as a non-curable Breach rather than a Default.
- (d) The failure by Lessee to provide (i) reasonable written evidence of compliance with Applicable Requirements, (ii) the service contracts, (iii) the rescission of an unauthorized assignment or subletting, (iv) an Estoppel Certificate or financial statements, (v) a requested subordination, (vi) evidence concerning any guaranty and/or Guarantor, (vii) any document requested under Paragraph 41, (viii) material safety data sheets (MSDS), or (ix) any other documentation or information which Lessor may reasonably require of Lessee under the terms of this Lease, where any such failure continues for a period of 10 days following written notice to Lessee.
- (e) A Default by Lessee as to the terms, covenants, conditions or provisions of this Lease, or of the rules adopted under Paragraph 2.9 hereof, other than those described in subparagraphs 13.1(a), (b), (c) or (d), above, where such Default continues for a period of 30 days after written notice; provided, however, that if the nature of Lessee's Default is such that more than 30 days are reasonably required for its cure, then it shall not be deemed to be a Breach if Lessee commences such cure within said 30 day period and thereafter diligently prosecutes such cure to completion.
- (f) The occurrence of any of the following events: (i) the making of any general arrangement or assignment for the benefit of creditors; (ii) becoming a "debtor" as defined in 11 U.S.C. § 101 or any successor statute thereto (unless, in the case of a petition filed against Lessee, the same is dismissed within 60 days); (iii) the appointment of a trustee or receiver to take possession of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where possession is not restored to Lessee within 30 days; or (iv) the attachment, execution or other judicial seizure of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where such seizure is not discharged within 30 days; provided, however, in the event that any provision of this subparagraph is contrary to any applicable law, such provision shall be of no force or effect, and not affect the validity of the remaining provisions.
 - (g) The discovery that any financial statement of Lessee or of any Guarantor given to Lessor was materially false.
- (h) If the performance of Lessee's obligations under this Lease is guaranteed: (i) the death of a Guarantor, (ii) the termination of a Guarantor's liability with respect to this Lease other than in accordance with the terms of such guaranty, (iii) a Guarantor's becoming insolvent or the subject of a bankruptcy filing, (iv) a Guarantor's refusal to honor the guaranty, or (v) a Guarantor's breach of its guaranty obligation on an anticipatory basis, and Lessee's failure, within 60 days following written notice of any such event, to provide written alternative assurance or security, which, when coupled with the then existing resources of Lessee, equals or exceeds the combined financial resources of Lessee and the Guarantors that existed at the time of execution of this Lease.
- 13.2 Remedies. If Lessee fails to perform any of its affirmative duties or obligations, within 10 days after written notice (or in case of an emergency, without notice), Lessor may, at its option, perform such duty or obligation on Lessee's behalf, including but not limited to the obtaining of reasonably required bonds, insurance policies, or governmental licenses, permits or approvals. Lessee shall pay to Lessor an amount equal to 115% of the costs and expenses incurred by Lessor in such performance upon receipt of an invoice therefor. In the event of a Breach, Lessor may, with or without further notice or demand, and without limiting Lessor in the exercise of any right or remedy which Lessor may have by reason of such Breach:
- (a) Terminate Lessee's right to possession of the Premises by any lawful means, in which case this Lease shall terminate and Lessee shall immediately surrender possession to Lessor. In such event Lessor shall be entitled to recover from Lessee: (i) the unpaid Rent which had been earned at the time of termination; (ii) the worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that the Lessee proves could have been reasonably avoided; (iii) the worth at the time of award of the amount by which the unpaid rent for the balance of the term after the time of award exceeds the amount of such rental loss that the Lessee proves could be reasonably avoided; and (iv) any other amount necessary to compensate Lessor for all the detriment proximately caused by the Lessee's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom, including but not limited to the cost of recovering possession of the Premises, expenses of reletting, including necessary renovation and alteration of the Premises, reasonable attorneys' fees, and that portion of any leasing commission paid by Lessor in connection with this Lease applicable to the unexpired term of this Lease. The worth at the time of award of the amount referred to in provision (iii) of the immediately preceding sentence shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of the District within which the Premises are located at the time of award plus one percent. Efforts by Lessor to mitigate damages caused by Lessee's Breach of this Lease shall not waive Lessor's right to recover any damages to

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

which Lessor is otherwise entitled. If termination of this Lease is obtained through the provisional remedy of unlawful detainer, Lessor shall have the right to recover in such proceeding any unpaid Rent and damages as are recoverable therein, or Lessor may reserve the right to recover all or any part thereof in a separate suit. If a notice and grace period required under Paragraph 13.1 was not previously given, a notice to pay rent or quit, or to perform or quit given to Lessee under the unlawful detainer statute shall also constitute the notice required by Paragraph 13.1. In such case, the applicable grace period required by Paragraph 13.1 and the unlawful detainer statute shall run concurrently, and the failure of Lessee to cure the Default within the greater of the two such grace periods shall constitute both an unlawful detainer and a Breach of this Lease entitling Lessor to the remedies provided for in this Lease and/or by said statute.

- (b) Continue the Lease and Lessee's right to possession and recover the Rent as it becomes due, in which event Lessee may sublet or assign, subject only to reasonable limitations. Acts of maintenance, efforts to relet, and/or the appointment of a receiver to protect the Lessor's interests, shall not constitute a termination of the Lessee's right to possession.
- (c) Pursue any other remedy now or hereafter available under the laws or judicial decisions of the state wherein the Premises are located. The expiration or termination of this Lease and/or the termination of Lessee's right to possession shall not relieve Lessee from liability under any indemnity provisions of this Lease as to matters occurring or accruing during the term hereof or by reason of Lessee's occupancy of the Premises.
- 13.3 Inducement Recapture. Any agreement for free or abated rent or other charges, the cost of tenant improvements for Lessee paid for or performed by Lessor, or for the giving or paying by Lessor to or for Lessee of any cash or other bonus, inducement or consideration for Lessee's entering into this Lease, all of which concessions are hereinafter referred to as "Inducement Provisions", shall be deemed conditioned upon Lessee's full and faithful performance of all of the terms, covenants and conditions of this Lease. Upon Breach of this Lease by Lessee, any such Inducement Provision shall automatically be deemed deleted from this Lease and of no further force or effect, and any rent, other charge, bonus, inducement or consideration theretofore abated, given or paid by Lessor under such an Inducement Provision shall be immediately due and payable by Lessee to Lessor, notwithstanding any subsequent cure of said Breach by Lessee.
- 13.4 Late Charges. Lessee hereby acknowledges that late payment by Lessee of Rent will cause Lessor to incur costs not contemplated by this Lease, the exact amount of which will be extremely difficult to ascertain. Such costs include, but are not limited to, processing and accounting charges, and late charges which may be imposed upon Lessor by any Lender. Accordingly, if any Rent shall not be received by Lessor within 5 days after such amount shall be due, then, without any requirement for notice to Lessee, Lessee shall immediately pay to Lessor a one-time late charge equal to 10% of each such overdue amount or \$100, whichever is greater. The parties hereby agree that such late charge represents a fair and reasonable estimate of the costs Lessor will incur by reason of such late payment. Acceptance of such late charge by Lessor shall in no event constitute a waiver of Lessee's Default or Breach with respect to such overdue amount, nor prevent the exercise of any of the other rights and remedies granted hereunder.
- 13.5 Interest. Any monetary payment due Lessor hereunder, other than late charges, not received by Lessor, when due shall bear interest from the 31st day after it was due. The interest ("Interest") charged shall be computed at the rate of 10% per annum but shall not exceed the maximum rate allowed by law. Interest is payable in addition to the potential late charge provided for in Paragraph 13.4.
- 13.6 **Breach by Lessor**. Lessor shall not be deemed in breach of this Lease unless Lessor fails within a reasonable time to perform an obligation required to be performed by Lessor. For purposes of this Paragraph, a reasonable time shall in no event be less than 30 days after receipt by Lessor of written notice specifying wherein such obligation of Lessor has not been performed; provided, however, that if the nature of Lessor's obligation is such that more than 30 days are reasonably required for its performance, then Lessor shall not be in breach if performance is commenced within such 30 day period and thereafter diligently pursued to completion.
- 14. Condemnation. If the Premises or any portion thereof are taken under the power of eminent domain or sold under the threat of the exercise of said power (collectively "Condemnation"), this Lease shall terminate as to the part taken as of the date the condemning authority takes title or possession, whichever first occurs. If more than 10% of the floor area of the Unit, or more than 25% of the parking spaces is taken by Condemnation, Lessee may, at Lessee's option, to be exercised in writing within 10 days after Lessor shall have given Lessee written notice of such taking (or in the absence of such notice, within 10 days after the condemning authority shall have taken possession) terminate this Lease as of the date the condemning authority takes such possession. If Lessee does not terminate this Lease in accordance with the foregoing, this Lease shall remain in full force and effect as to the portion of the Premises remaining, except that the Base Rent shall be reduced in proportion to the reduction in utility of the Premises caused by such Condemnation. Condemnation awards and/or payments shall be the property of Lessor, whether such award shall be made as compensation for diminution in value of the leasehold, the value of the part taken, or for severance damages; provided, however, that Lessee shall be entitled to any compensation paid by the condemnor for Lessee's relocation expenses, loss of business goodwill and/or Trade Fixtures, without regard to whether or not this Lease is terminated pursuant to the provisions of this Paragraph. All Alterations and Utility Installations made to the Premises by Lessee, for purposes of Condemnation only, shall be considered the property of the Lessee shall be entitled to any and all compensation which is payable therefor. In the event that this Lease is not terminated by reason of the Condemnation, Lessor shall repair any damage to the Premises caused by such Condemnation.

15. Brokerage Fees.

- 15.1 Additional Commission. Lessor agrees that: (a) if Lessee or anyone affiliated with Lessee acquires from Lessor any rights to the Premises or other premises owned by Lessor and located within the Project, or (b) if Lessee remains in possession of the Premises, with the consent of Lessor, after the expiration of this Lease, then Lessor shall pay Brokers a fee in accordance with the schedule attached to such brokerage fee agreement. The provisions of this paragraph are intended to supersede the provisions of any earlier agreement to the contrary.
- 15.2 **Assumption of Obligations**. Any buyer or transferee of Lessor's interest in this Lease shall be deemed to have assumed Lessor's obligation hereunder. Brokers shall be third party beneficiaries of the provisions of Paragraphs 1.9, 15, 22 and 31.
- 15.3 **Representations and Indemnities of Broker Relationships**. Lessee and Lessor each represent and warrant to the other that it has had no dealings with any person, firm, broker, agent or finder (other than the Brokers and Agents, if any) in connection with this Lease, and that no one other than said named Brokers and Agents is entitled to any commission or finder's fee in connection herewith. Lessee and Lessor do each hereby agree to indemnify, protect, defend and hold the other harmless from and against liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any dealings or actions of the indemnifying Party, including any costs, expenses, attorneys' fees reasonably incurred with respect thereto.

16. Estoppel Certificates.

- (a) Each Party (as "Responding Party") shall within 10 days after written notice from the other Party (the "Requesting Party") execute, acknowledge and deliver to the Requesting Party a statement in writing in form similar to the then most current "Estoppel Certificate" form published by AIR CRE.
- (b) If the Responding Party shall fail to execute or deliver the Estoppel Certificate within such 10 day period, the Requesting Party may execute an Estoppel Certificate stating that: (i) the Lease is in full force and effect without modification except as may be represented by the Requesting Party, (ii) there are no uncured defaults in the Requesting Party's performance, and (iii) if Lessor is the Requesting Party, not more than one month's rent has been paid in advance. Prospective purchasers and encumbrancers may rely upon the Requesting Party's Estoppel Certificate, and the Responding Party shall be estopped from denying the truth of the facts contained in said Certificate. In addition, Lessee acknowledges that any failure on its part to provide such an Estoppel Certificate will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to execute and/or deliver a requested Estoppel Certificate in a timely fashion the monthly Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for remainder of the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to provide the Estoppel Certificate. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to provide the Estoppel Certificate nor prevent the exercise of any of the other rights and remedies granted hereunder.
- (c) If Lessor desires to finance, refinance, or sell the Premises, or any part thereof, Lessee and all Guarantors shall within 10 days after written notice from Lessor deliver to any potential lender or purchaser designated by Lessor such financial statements as may be reasonably required by such lender or purchaser, including but not limited to Lessee's financial statements for the past 3 years. All such financial statements shall be received by Lessor and such lender or purchaser in confidence and shall be used only for the purposes herein set forth.
- 17. Definition of Lessor. The term "Lessor" as used herein shall mean the owner or owners at the time in question of the fee title to the Premises, or, if this is a sublease, of the Lessee's interest in the prior lease. In the event of a transfer of Lessor's title or interest in the Premises or this Lease, Lessor shall deliver to the transferee or assignee (in cash or by credit) any unused Security Deposit held by Lessor. Upon such transfer or assignment and delivery of the Security Deposit, as aforesaid, the prior Lessor shall be relieved of all liability with respect to the obligations and/or covenants under this Lease thereafter to be performed by the Lessor. Subject to the foregoing, the obligations and/or covenants in this Lease to be performed by the Lessor shall be binding only upon the Lessor as hereinabove defined.
- 18. Severability. The invalidity of any provision of this Lease, as determined by a court of competent jurisdiction, shall in no way affect the validity of any other provision hereof.
- 19. Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Lease shall mean and refer to calendar days.
- 20. Limitation on Liability. The obligations of Lessor under this Lease shall not constitute personal obligations of Lessor, or its partners, members, directors, officers

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

or shareholders, and Lessee shall look to the Premises, and to no other assets of Lessor, for the satisfaction of any liability of Lessor with respect to this Lease, and shall not seek recourse against Lessor's partners, members, directors, officers or shareholders, or any of their personal assets for such satisfaction.

- 21. Time of Essence. Time is of the essence with respect to the performance of all obligations to be performed or observed by the Parties under this Lease.
- 22. No Prior or Other Agreements; Broker Disclaimer. This Lease contains all agreements between the Parties with respect to any matter mentioned herein, and no other prior or contemporaneous agreement or understanding shall be effective. Lessor and Lessee each represents and warrants to the Brokers that it has made, and is relying solely upon, its own investigation as to the nature, quality, character and financial responsibility of the other Party to this Lease and as to the use, nature, quality and character of the Premises. Brokers have no responsibility with respect thereto or with respect to any default or breach hereof by either Party.

23. Notices.

- 23.1 **Notice Requirements**. All notices required or permitted by this Lease or applicable law shall be in writing and may be delivered in person (by hand or by courier) or may be sent by regular, certified or registered mail or U.S. Postal Service Express Mail, with postage prepaid, or by facsimile transmission, or by email, and shall be deemed sufficiently given if served in a manner specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specify a different address for notice, except that upon Lessee's taking possession of the Premises, the Premises shall constitute Lessee's address for notice. A copy of all notices to Lessor shall be concurrently transmitted to such party or parties at such addresses as Lessor may from time to time hereafter designate in writing.
- 23.2 Date of Notice. Any notice sent by registered or certified mail, return receipt requested, shall be deemed given on the date of delivery shown on the receipt card, or if no delivery date is shown, the postmark thereon. If sent by regular mail the notice shall be deemed given 72 hours after the same is addressed as required herein and mailed with postage prepaid. Notices delivered by United States Express Mail or overnight courier that guarantees next day delivery shall be deemed given 24 hours after delivery of the same to the Postal Service or courier. Notices delivered by hand, or transmitted by facsimile transmission or by email shall be deemed delivered upon actual receipt. If notice is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.
- 23.3 **Options.** Notwithstanding the foregoing, in order to exercise any options, the Notice must be sent by Certified Mail (return receipt requested), Express Mail (signature required), courier (signature required) or some other methodology that provides a receipt establishing the date the notice was received by the Lesson.

24. Waivers.

- (a) No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any other term, covenant or condition hereof.
- (b) The acceptance of Rent by Lessor shall not be a waiver of any Default or Breach by Lessee. Any payment by Lessee may be accepted by Lessor on account of monies or damages due Lessor, notwithstanding any qualifying statements or conditions made by Lessee in connection therewith, which such statements and/or conditions shall be of no force or effect whatsoever unless specifically agreed to in writing by Lessor at or before the time of deposit of such payment.
- (c) THE PARTIES AGREE THAT THE TERMS OF THIS LEASE SHALL GOVERN WITH REGARD TO ALL MATTERS RELATED THERETO AND HEREBY WAIVE THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE TO THE EXTENT THAT SUCH STATUTE IS INCONSISTENT WITH THIS LEASE.

25. Disclosures Regarding The Nature of a Real Estate Agency Relationship.

- (a) When entering into a discussion with a real estate agent regarding a real estate transaction, a Lessor or Lessee should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Lessor and Lessee acknowledge being advised by the Brokers in this transaction, as follows:
- (i) <u>Lessor's Agent</u>. A Lessor's agent under a listing agreement with the Lessor acts as the agent for the Lessor only. A Lessor's agent or subagent has the following affirmative obligations: <u>To the Lessor</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessor. <u>To the Lessee and the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (ii) <u>Lessee's Agent</u>. An agent can agree to act as agent for the Lessee only. In these situations, the agent is not the Lessor's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Lessor. An agent acting only for a Lessee has the following affirmative obligations. <u>To the Lessee</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessee. <u>To the Lessee and the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (iii) Agent Representing Both Lessor and Lessee. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Lessor and the Lessee in a transaction, but only with the knowledge and consent of both the Lessor and the Lessee. In a dual agency situation, the agent has the following affirmative obligations to both the Lessor and the Lessee: (a) A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Lessor or the Lessee. (b) Other duties to the Lessor and the Lessee as stated above in subparagraphs (i) or (ii). In representing both Lessor and Lessee, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating to either Lessee's or Lessor's financial position, motivations, bargaining position, or other personal information that may impact rent, including Lessor's willingness to accept a rent less than the listing rent or Lessee's willingness to pay rent greater than the rent offered. The above duties of the agent in a real estate transaction do not relieve a Lessor or Lessee from the responsibility to protect their own interests. Lessor and Lessee should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Both Lessor and Lessee should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.
- (b) Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this Lease may be brought against Broker more than one year after the Start Date and that the liability (including court costs and attorneys' fees), of any Broker with respect to any such lawsuit and/or legal proceeding shall not exceed the fee received by such Broker pursuant to this Lease; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.
- (c) Lessor and Lessee agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- **26. No Right To Holdover**. Lessee has no right to retain possession of the Premises or any part thereof beyond the expiration or termination of this Lease. In the event that Lessee holds over, then the Base Rent shall be increased to 150% of the Base Rent applicable immediately preceding the expiration or termination. Holdover Base Rent shall be calculated on monthly basis. Nothing contained herein shall be construed as consent by Lessor to any holding over by Lessee.
- **27. Cumulative Remedies.** No remedy or election hereunder shall be deemed exclusive but shall, wherever possible, be cumulative with all other remedies at law or in equity.
- 28. Covenants and Conditions; Construction of Agreement. All provisions of this Lease to be observed or performed by Lessee are both covenants and conditions. In construing this Lease, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Lease. Whenever required by the context, the singular shall include the plural and vice versa. This Lease shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.
- 29. Binding Effect; Choice of Law. This Lease shall be binding upon the Parties, their personal representatives, successors and assigns and be governed by the laws of the State in which the Premises are located. Any litigation between the Parties hereto concerning this Lease shall be initiated in the county in which the Premises are located. Signatures to this Lease accomplished by means of electronic signature or similar technology shall be legal and binding.

30. Subordination; Attornment; Non-Disturbance.

- 30.1 **Subordination**. This Lease and any Option granted hereby shall be subject and subordinate to any ground lease, mortgage, deed of trust, or other hypothecation or security device (collectively, "Security Device"), now or hereafter placed upon the Premises, to any and all advances made on the security thereof, and to all renewals, modifications, and extensions thereof.
- 30.2 **Attornment**. In the event that Lessor transfers title to the Premises, or the Premises are acquired by another upon the foreclosure or termination of a Security Device to which this Lease is subordinated (i) Lessee shall, subject to the non-disturbance provisions of Paragraph 30.3, attorn to such new owner.
 - 30.3 Non-Disturbance. With respect to Security Devices entered into by Lessor after the execution of this Lease, Lessee's subordination of this Lease shall be

				INITIALS
 	_			

INITIALS

subject to receiving a commercially reasonable non-disturbance agreement (a "Non-Disturbance Agreement") from the Lender which Non-Disturbance Agreement provides that Lessee's possession of the Premises, and this Lease, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises.

30.4 **Self-Executing**. The agreements contained in this Paragraph 30 shall be effective without the execution of any further documents; provided, however, that, upon written request from Lessor or a Lender in connection with a sale, financing or refinancing of the Premises, Lessee and Lessor shall execute such further writings as may be reasonably required to separately document any subordination, attornment and/or Non-Disturbance Agreement provided for herein.

- 31. Attorneys' Fees. If any Party or Broker brings an action or proceeding involving the Premises whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. In addition, Lessor shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is subsequently commenced in connection with such Default or resulting Breach (\$200 is a reasonable minimum per occurrence for such services and consultation).
- **32.** Lessor's Access; Showing Premises; Repairs. Lessor and Lessor's agents shall have the right to enter the Premises at any time, in the case of an emergency, and otherwise at reasonable times after reasonable prior notice for the purpose of showing the same to prospective purchasers, lenders, or tenants, and making such alterations, repairs, improvements or additions to the Premises as Lessor may deem necessary or desirable and the erecting, using and maintaining of utilities, services, pipes and conduits through the Premises and/or other premises as long as there is no material adverse effect on Lessee's use of the Premises. All such activities shall be without abatement of rent or liability to Lessee.
- **33. Auctions.** Lessee shall not conduct, nor permit to be conducted, any auction upon the Premises without Lessor's prior written consent. Lessor shall not be obligated to exercise any standard of reasonableness in determining whether to permit an auction.
- **34. Signs.** Lessor may place on the Premises ordinary "For Sale" and/or "For Lease" signs at any time. Lessee shall not place any sign upon the Project without Lessor's prior written consent. All signs must comply with all Applicable Requirements.
- **35. Termination; Merger.** Unless specifically stated otherwise in writing by Lessor, the voluntary or other surrender of this Lease by Lessee, the mutual termination or cancellation hereof, or a termination hereof by Lessor for Breach by Lessee, shall automatically terminate any sublease or lesser estate in the Premises; provided, however, that Lessor may elect to continue any one or all existing subtenancies. Lessor's failure within 10 days following any such event to elect to the contrary by written notice to the holder of any such lesser interest, shall constitute Lessor's election to have such event constitute the termination of such interest.
- **36. Consents.** All requests for consent shall be in writing. Except as otherwise provided herein, wherever in this Lease the consent of a Party is required to an act by or for the other Party, such consent shall not be unreasonably withheld or delayed. Lessor's actual reasonable costs and expenses (including but not limited to architects', attorneys', engineers' and other consultants' fees) incurred in the consideration of, or response to, a request by Lessee for any Lessor consent, including but not limited to consents to an assignment, a subletting or the presence or use of a Hazardous Substance, shall be paid by Lessee upon receipt of an invoice and supporting documentation therefor.
- 37. Guarantor. The Guarantors, if any, shall each execute a guaranty in the form most recently published by AIR CRE.
- **38. Quiet Possession.** Subject to payment by Lessee of the Rent and performance of all of the covenants, conditions and provisions on Lessee's part to be observed and performed under this Lease, Lessee shall have quiet possession and quiet enjoyment of the Premises during the term hereof.
- **39. Security Measures.** Lessee hereby acknowledges that the Rent payable to Lessor hereunder does not include the cost of guard service or other security measures, and that Lessor shall have no obligation whatsoever to provide same. Lessee assumes all responsibility for the protection of the Premises, Lessee, its agents and invitees and their property from the acts of third parties.
- **40. Reservations.** Lessor reserves the right: (i) to grant, without the consent or joinder of Lessee, such easements, rights and dedications that Lessor deems necessary, (ii) to cause the recordation of parcel maps and restrictions, and (iii) to create and/or install new utility raceways, so long as such easements, rights, dedications, maps, restrictions, and utility raceways do not unreasonably interfere with the use of the Premises by Lessee. Lessee agrees to sign any documents reasonably requested by Lessor to effectuate such rights.

41. Authority; Multiple Parties; Execution.

- (a) If either Party hereto is a corporation, trust, limited liability company, partnership, or similar entity, each individual executing this Lease on behalf of such entity represents and warrants that he or she is duly authorized to execute and deliver this Lease on its behalf. Each Party shall, within 30 days after request, deliver to the other Party satisfactory evidence of such authority.
- (b) If this Lease is executed by more than one person or entity as "Lessee", each such person or entity shall be jointly and severally liable hereunder. It is agreed that any one of the named Lessees shall be empowered to execute any amendment to this Lease, or other document ancillary thereto and bind all of the named Lessees, and Lessor may rely on the same as if all of the named Lessees had executed such document.
- (c) This Lease may be executed by the Parties in counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument.
- **42. Conflict.** Any conflict between the printed provisions of this Lease and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions.
- **43. Offer**. Preparation of this Lease by either party or their agent and submission of same to the other Party shall not be deemed an offer to lease to the other Party. This Lease is not intended to be binding until executed and delivered by all Parties hereto.
- 44. Amendments. This Lease may be modified only in writing, signed by the Parties in interest at the time of the modification.
- 45. Accessibility; Americans with Disabilities Act.

45. Accessibility; Americans with disabilities Act.
(a) The Premises:
have not undergone an inspection by a Certified Access Specialist (CASp). Note: A Certified Access Specialist (CASp) can inspect the subject premises and
determine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not
require a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection of
the subject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the
arrangements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to
correct violations of construction-related accessibility standards within the premises.
have undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises met all applicable construction-related accessibilit standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this Lease and agrees to keep such report confidential.
have undergone an inspection by a Certified Access Specialist (CASp) and it was determined that the Premises did not meet all applicable construction-related accessibility standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this Lease and agrees to keep such report confidential except as necessary to complete repairs and corrections of violations of construction related accessibility standards.
In the event that the Premises have been issued an inspection report by a CASp the Lessor shall provide a copy of the disability access inspection certificate to Lessee

within 7 days of the execution of this Lease.

(b) Since compliance with the Americans with Disabilities Act (ADA) and other state and local accessibility statutes are dependent upon Lessee's specific use of the Premises, Lessor makes no warranty or representation as to whether or not the Premises comply with ADA or any similar legislation. In the event that Lessee's use of the Premises requires modifications or additions to the Premises in order to be in compliance with ADA or other accessibility statutes, Lessee agrees to make any such necessary modifications and/or additions at Lessee's expense.

LESSOR AND LESSEE HAVE CAREFULLY READ AND REVIEWED THIS LEASE AND EACH TERM AND PROVISION CONTAINED HEREIN, AND BY THE EXECUTION OF THIS

INITIALS		INITIALS
© 2019 AIR CRE.	All Rights Reserved.	

LEASE SHOW THEIR INFORMED AND VOLUNTARY CONSENT THERETO. THE PARTIES HEREBY AGREE THAT, AT THE TIME THIS LEASE IS EXECUTED, THE TERMS OF THIS LEASE ARE COMMERCIALLY REASONABLE AND EFFECTUATE THE INTENT AND PURPOSE OF LESSOR AND LESSEE WITH RESPECT TO THE PREMISES.

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS LEASE OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

- 1. SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS LEASE.
- 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PREMISES. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING OF THE PREMISES, THE STRUCTURAL INTEGRITY, THE CONDITION OF THE ROOF AND OPERATING SYSTEMS, COMPLIANCE WITH THE AMERICANS WITH DISABILITIES ACT AND THE SUITABILITY OF THE PREMISES FOR LESSEE'S INTENDED USE.

WARNING: IF THE PREMISES ARE LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THE LEASE MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PREMISES ARE LOCATED.

The parties hereto have executed this Lease at the place and on the dates specified above their respective signatures. Executed at: Executed at: By LESSOR: By LESSEE: By: By: Name Printed: Name Printed: Title: Title: Phone: Phone: Fax: Fax: Email: Ву: By: Name Printed: Name Printed: Title: Title: Phone: Phone: Fax: Fax: Email: Address: Address: Federal ID No.: Federal ID No.: BROKER BROKER Attn: Attn Title Address: Phone: Phone: Fax: Email: Email: Federal ID No.: Federal ID No.: Broker DRE License #: Broker DRE License #: Agent DRE License #: Agent DRE License #:

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com

NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS

INITIALS
© 2019 AIR CRE. All Rights Reserved.
MTMG-1.30, Revised 10-22-2020



STANDARD MULTI-TENANT SHOPPING CENTER LEASE - NET

1. Basic Provisions ("Basic Provisions").		
1.1 Parties. This Lease ("Lease"), dated for reference purposes only	, is made by and between	(" Lessor ") and
("Lessee") (collectively the "Parties", or individually a "Party").1.2 Premises: That certain portion of the Shopping Center (as defined below).	including all improvements therein or to be	a provided by Lassar under the terms
of this Lease, commonly known as (street address, unit/suite, city, state):		
generally described as (describe briefly the nature of the Premises): In		
specified, Lessee shall have non-exclusive rights to the Common Areas (as defined in I		
roof, the exterior walls, or the utility raceways of the building containing the Premises		
the Building are situated within the Shopping Center known as The Pr	emises, the Building, the Common Areas, a	nd all other buildings and
improvements within said Shopping Center, together with the land upon which they a	re located, are herein collectively referred t	to as the " Shopping Center ." (See
also Paragraph 2)		
1.3 Term: years and months ("Original Term") co	ommencing ("Commencement	nt Date") and ending
("Expiration Date"). (See also Paragraph 3) 1.4 Early Possession: If the Premises are available Lessee may have non-exclu	sive persection of the Promises commencing	og ("Forth Possession
Date"). (See also Paragraphs 3.2 and 3.3)	sive possession of the Fremises commencing	(Early Possession
1.5 Base Rent: per month ("Base Rent"), payable on the	day of each month commencing	. (See also Paragraph 4)
If this box is checked, there are provisions in this Lease for the Base Rent to 1.6 Percentage Rent Rate: percent (%) of Gross S		
provisions of the Percentage Rent Addendum, if any, attached hereto and made a par		vable in accordance with the
1.7 Lessee's Share of Common Area Operating Expenses: perc		the event that the size of the
Premises and/or the Shopping Center are modified during the term of this Lease, Less		
1.8 Merchants' Association Annual Dues: per year ("Merchant		
become a member of the Merchants' Association in accordance with the provisions o		/
1.9 Base Rent and Other Monies Paid Upon Execution:		
(a) Base Rent: for the period		
(b) Common Area Operating Expenses: The current estimate for the pe		*
(c) Security Deposit: ("Security Deposit"). (See also Para		
(d) Merchants' Association Dues: for the period		
(e) Other: for	*	
(f) Total Due Upon Execution of this Lease:		
1.10 Agreed Use: (See also Paragraph 6)		
1.11 Agreed Trade Name: (See also Paragraph 6)	/ \	
1.12 Insuring Party. Lessor is the "Insuring Party". (See also Paragraph 8)		
1.13 Real Estate Brokers. (See also Paragraph 15 and 25)(a) Representation: Each Party acknowledges receiving a Disclosure Reg	zarding Real Estate Agency Relationship, cor	ofirms and consents to the following
agency relationships in this Lease with the following real estate brokers ("Broker(s)")		ministra consents to the following
Lessor's Brokerage Firm License No Is the broker		th the Lessee and Lessor (dual agent)
Lessor's Agent License No is (check one): the	e Lessor's Agent (salesperson or broker asso	ociate); or 🗀 both the Lessee's
A nearly and the Languig Annut (dural annut)		
Agent and the Lessor's Agent (dual agent).		
Lessee's Brokerage Firm License No is the broker	of (check one): the Lessee; or bo	oth the Lessee and Lessor (dual
Lessee's Brokerage Firm License No is the broker agent).		
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): the		
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): Uth Agent and the Lessor's Agent (dual agent).	ne Lessee's Agent (salesperson or broker ass	sociate); or both the Lessee's
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): Uthe Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by both	ne Lessee's Agent (salesperson or broker ass oth Parties, Lessor shall pay to the Brokers t	sociate); or both the Lessee's the brokerage fee agreed to in a
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of	ne Lessee's Agent (salesperson or broker ass oth Parties, Lessor shall pay to the Brokers t	sociate); or both the Lessee's the brokerage fee agreed to in a
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers.	ne Lessee's Agent (salesperson or broker ass oth Parties, Lessor shall pay to the Brokers t or% of the total Base Ren	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of	oth Parties, Lessor shall pay to the Brokers t or	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a	oth Parties, Lessor shall pay to the Brokers to or% of the total Base Renoteed by ("Guarantor"). (See part of this Lease:	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through	oth Parties, Lessor shall pay to the Brokers to or% of the total Base Renoteed by ("Guarantor"). (See part of this Lease:	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises;	oth Parties, Lessor shall pay to the Brokers to or% of the total Base Renoteed by ("Guarantor"). (See part of this Lease:	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through	oth Parties, Lessor shall pay to the Brokers to or% of the total Base Renoteed by ("Guarantor"). (See part of this Lease:	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is the broker agent). Lessee's Agent License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarar 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center;	oth Parties, Lessor shall pay to the Brokers to or% of the total Base Renoteed by ("Guarantor"). (See part of this Lease:	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center;	oth Parties, Lessor shall pay to the Brokers to or% of the total Base Renoteed by ("Guarantor"). (See part of this Lease:	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarar 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center;	oth Parties, Lessor shall pay to the Brokers to or% of the total Base Renoteed by ("Guarantor"). (See part of this Lease:	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center;	oth Parties, Lessor shall pay to the Brokers to or% of the total Base Renoteed by ("Guarantor"). (See part of this Lease:	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarar 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center;	oth Parties, Lessor shall pay to the Brokers to or% of the total Base Renoteed by ("Guarantor"). (See part of this Lease:	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarar 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify):	oth Parties, Lessor shall pay to the Brokers to or% of the total Base Renoteed by ("Guarantor"). (See part of this Lease:	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered
Lessee's Brokerage Firm License No is (check one): the depicting the Premises; a current set of the Rules and Regulations for the Shopping Center; a Work Letter;	ne Lessee's Agent (salesperson or broker assorth Parties, Lessor shall pay to the Brokers to or	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered e also Paragraph 37)
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lescovenants and conditions set forth in this Lease. While the approximate square footal	ne Lessee's Agent (salesperson or broker assorth Parties, Lessor shall pay to the Brokers to or	both the Lessee's the brokerage fee agreed to in a at) for the brokerage services rendered also Paragraph 37)
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lescovenants and conditions set forth in this Lease. While the approximate square footapurposes of comparison, the Base Rent stated herein is NOT tied to square footage are to be guarant and conditions set forth in this Lease.	ne Lessee's Agent (salesperson or broker assorth Parties, Lessor shall pay to the Brokers to or	both the Lessee's the brokerage fee agreed to in a at) for the brokerage services rendered also Paragraph 37)
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lecovenants and conditions set forth in this Lease. While the approximate square footapurposes of comparison, the Base Rent stated herein is NOT tied to square footage ard different. NOTE: Lessee is advised to verify the actual size prior to executing this Lease.	ne Lessee's Agent (salesperson or broker assorth Parties, Lessor shall pay to the Brokers to or	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered e also Paragraph 37) al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage are different. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. Condition. Lessor shall deliver the Premises to Lessee broom clean and free	ne Lessee's Agent (salesperson or broker assorth Parties, Lessor shall pay to the Brokers to or	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered e also Paragraph 37) al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be r the Early Possession Date, whichever
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarar 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lescorenants and conditions set forth in this Lease. While the approximate square footage ard different. NOTE: Lessee is advised to verify the actual size prior to executing this Le 2.2 Condition. Lessor shall deliver the Premises to Lessee broom clean and frefirst occurs ("Start Date"), and, so long as the required service contracts described in	the Lessee's Agent (salesperson or broker assorth Parties, Lessor shall pay to the Brokers to or	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered e also Paragraph 37) al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be r the Early Possession Date, whichever see and in effect within 30 days
Lessee's Brokerage Firm License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage are different. NOTE: Lessee is advised to verify the actual size prior to executing this Lease. Condition. Lessor shall deliver the Premises to Lessee broom clean and free	the Lessee's Agent (salesperson or broker assorth Parties, Lessor shall pay to the Brokers to or	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered e also Paragraph 37) al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be r the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading
Lessee's Brokerage Firm License No is (check one): the broker agent). Lessee's Agent License No is (check one): the same of the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarante. It is an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square footage ard different. NOTE: Lessee is advised to verify the actual size prior to executing this Lectors ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises, other than those constructs structural elements of the roof, bearing walls and foundation of the Premises shall be	ssor, the Premises, for the term, at the rents ge of the Premises may have been used in the dis not subject to adjustment should the a ase. see of debris on the Commencement Date or Paragraph 7.1(b) below are obtained by Les lighting, heating, ventilating and air conditied by Lessee, shall be in good operating confree of material defects, and that the Premises of the Brokers as a see of material defects, and that the Premises of the Brokers as a see of material defects, and that the Premises of the Brokers as a see of material defects, and that the Premises of the Brokers as a see of material defects, and that the Premises of the Brokers as a see of the Brokers and that the Premises of the Brokers as a see of the Brokers and that the Premises of the Brokers as a see of the Brokers as a second and the B	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be r the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading addition on said date, that the lises do not contain hazardous levels
Lessee's Brokerage Firm License No is (check one): the broker agent). Lessee's Agent License No is (check one): the brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarar 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lescovenants and conditions set forth in this Lease. While the approximate square footage ard different. NOTE: Lessee is advised to verify the actual size prior to executing this Lecture 1. NOTE: Lessee is advised to verify the actual size prior to executing this Lecture 1. NOTE: Lessor shall deliver the Premises to Lessee broom clean and from first occurs ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises, other than those constructs structural elements of the roof, bearing walls and foundation of the Premises shall be of any mold or fungi defined as toxic under applicable state or federal law. If a non-contract can be a contracted and the premises are forced and the premises are forced and the premises are forced as toxic under applicable state or federal law. If a non-contract described in foundation of the premises and the premises are forced an	ssor, the Premises, for the term, at the rents die is not subject to adjustment should the a ase. see of debris on the Commencement Date or Paragraph 7.1(b) below are obtained by Les lighting, heating, ventilating and air conditied by Lessee, shall be in good operating confree of material defects, and that the Premomphiance with such warranty exists as of the premises was assessed in the commencement bate or paragraph 7.1(b) below are obtained by Les lighting, heating, ventilating and air condities of material defects, and that the Premomphiance with such warranty exists as of the	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading didition on said date, that the lises do not contain hazardous levels the Start Date, or if one of such systems
Lessee's Brokerage Firm License No. Is the broker agent). Lessee's Agent License No. is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant of the Brokers. 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through as ite plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lecovenants and conditions set forth in this Lease. While the approximate square footage and different. NOTE: Lessee is advised to verify the actual size prior to executing this Lecovenants and condition. Lessor shall deliver the Premises to Lessee broom clean and frest first occurs ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises, other than those constructs structural elements of the roof, bearing walls and foundation of the Premises shall be of any mold or fungi defined as toxic under applicable state or federal law. If a non-cor elements should malfunction or fail within the appropriate warranty period, Lesson or elements should malfunction or fail within the appropriate warranty period, Lesson or elements should malfunction or fail within the appropriate warranty period, Lesson or elements should malfunction or fail within the appropriate warranty period, Lesson or elements should malfunction or fail within the appropriate warranty period, Lesson or element	ssor, the Premises, for the term, at the rents of the Premises may have been used in the dis not subject to adjustment should the a ase. The of debris on the Commencement Date or Paragraph 7.1(b) below are obtained by Les lighting, heating, ventilating and air conditions of material defects, and that the Premises of the Premises of the premises of the commencement Date or Paragraph 7.1(b) below are obtained by Les lighting, heating, ventilating and air conditions of the premises of the pr	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be or the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading didition on said date, that the sites do not contain hazardous levels the Start Date, or if one of such systems text to such matter, except as otherwise
Lessee's Brokerage Firm License No is (check one): the broker agent). Lessee's Agent License No is (check one): the brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarar 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lescovenants and conditions set forth in this Lease. While the approximate square footage ard different. NOTE: Lessee is advised to verify the actual size prior to executing this Lecture 1. NOTE: Lessee is advised to verify the actual size prior to executing this Lecture 1. NOTE: Lessor shall deliver the Premises to Lessee broom clean and from first occurs ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises, other than those constructs structural elements of the roof, bearing walls and foundation of the Premises shall be of any mold or fungi defined as toxic under applicable state or federal law. If a non-contract can be a contracted and the premises are forced and the premises are forced and the premises are forced as toxic under applicable state or federal law. If a non-contract described in foundation of the premises and the premises are forced an	ssor, the Premises, for the term, at the rentinge of the Premises may have been used in the distribution of distribution of distribution of the Premises of th	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be or the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading didition on said date, that the sites do not contain hazardous levels the Start Date, or if one of such systems text to such matter, except as otherwise such non-compliance, malfunction or
Lessee's Brokerage Firm License No is (check one): the broker agent). Lessee's Agent License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by b separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2. 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lecovenants and conditions set forth in this Lease. While the approximate square footage and different. NOTE: Lessee is advised to verify the actual size prior to executing this Lecovenants and conditions set forth in the Premises to Lessee broom clean and frefirst occurs ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises, other than those constructs structural elements of the roof, bearing walls and foundation of the Premises shall be of any mold or fungi defined as toxic under applicable state or federal law. If a non-coor elements should malfunction or fail within the appropriate warranty periods. Lessoe rovided in this Lease, promptly after receipt of written notice from Lessee setting for failure, rectify same at Lessor's expense. The warranty periods shall be as follows: (i) and other elements of the Premises. If Lessee does not give Lessor the required	ssor, the Premises, for the term, at the rent: ge of the Premises may have been used in tid is not subject to adjustment should the a ase. Paragraph 7.1(b) below are obtained by Les lighting, heating, ventilating and air conditied by Lessee, shall be in good operating confree of material defects, and that the Premises may have been used in the shall, as Lessor's sole obligation with respect the with specificity the nature and extent of 6 months as to the HVAC systems, and (ii) 3 the within the appropriate warranty period, of the contraction of the appropriate warranty period, of the within the appropriate warranty period, of the contraction of the warranty period, of the contraction of the warranty period, of the within the appropriate warranty period, of the contraction of the contraction of the warranty period, of the warranty period.	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered a also Paragraph 37) al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be r the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading didion on said date, that the ises do not contain hazardous levels he Start Date, or if one of such systems ect to such matter, except as otherwise such non-compliance, malfunction or 0 days as to the remaining systems correction of any such
Lessee's Brokerage Firm License No. is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square foota purposes of comparison, the Base Rent stated herein is NOT tied to square footage and different. NOTE: Lessee is advised to verify the actual size prior to executing this Lector ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises to Lessee broom clean and from first occurs ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises to Lessee setting for any mold or fungi defined as toxic under applicable state or federal law. If a non-correlements should malfunction or fail within the appropriate warranty period, Lessor provided in this Lease, promptly after receipt of written notice from Lessee setting for failure, rectify same at Lessor's expense. The warranty periods shall be as follows: (i) and other elements of the Premises. If Lessee does not give Lessor the required notic no	ssor, the Premises, for the term, at the rentage of the Premises may have been used in today is not subject to adjustment should the asse. Lee of debris on the Commencement Date or Paragraph 7.1(b) below are obtained by Lesselighting, heating, ventilating and air conditions of the premises of the Commencement Date or Paragraph 7.1(b) below are obtained by Lesse of the Premises o	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered a also Paragraph 37) al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be r the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading addition on said date, that the ises do not contain hazardous levels he Start Date, or if one of such systems to to such matter, except as otherwise such non-compliance, malfunction or 0 days as to the remaining systems correction of any such to the fire sprinkler systems, roof,
Lessee's Brokerage Firm License No is (check one): the broker agent). Lessee's Agent License No is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by b separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2. 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lecovenants and conditions set forth in this Lease. While the approximate square footage and different. NOTE: Lessee is advised to verify the actual size prior to executing this Lecovenants and conditions set forth in the Premises to Lessee broom clean and frefirst occurs ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises, other than those constructs structural elements of the roof, bearing walls and foundation of the Premises shall be of any mold or fungi defined as toxic under applicable state or federal law. If a non-coor elements should malfunction or fail within the appropriate warranty periods. Lessoe rovided in this Lease, promptly after receipt of written notice from Lessee setting for failure, rectify same at Lessor's expense. The warranty periods shall be as follows: (i) and other elements of the Premises. If Lessee does not give Lessor the required	ssor, the Premises, for the term, at the rentage of the Premises may have been used in today is not subject to adjustment should the asse. Lee of debris on the Commencement Date or Paragraph 7.1(b) below are obtained by Lesselighting, heating, ventilating and air conditions of the premises of the Commencement Date or Paragraph 7.1(b) below are obtained by Lesse of the Premises o	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered a also Paragraph 37) al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be r the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading addition on said date, that the ises do not contain hazardous levels he Start Date, or if one of such systems to to such matter, except as otherwise such non-compliance, malfunction or 0 days as to the remaining systems correction of any such to the fire sprinkler systems, roof,
Lessee's Brokerage Firm License No. is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Premises; a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Sign Criteria for the Shopping Center; a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lectovenants and conditions set forth in this Lease. While the approximate square foota purposes of comparison, the Base Rent stated herein is NOT tied to square footage and different. NOTE: Lessee is advised to verify the actual size prior to executing this Lector ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises to Lessee broom clean and from first occurs ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises to Lessee setting for any mold or fungi defined as toxic under applicable state or federal law. If a non-correlements should malfunction or fail within the appropriate warranty period, Lessor provided in this Lease, promptly after receipt of written notice from Lessee setting for failure, rectify same at Lessor's expense. The warranty periods shall be as follows: (i) and other elements of the Premises. If Lessee does not give Lessor the required notic no	ssor, the Premises, for the term, at the rentage of the Premises may have been used in today is not subject to adjustment should the asse. Lee of debris on the Commencement Date or Paragraph 7.1(b) below are obtained by Lesselighting, heating, ventilating and air conditions of the premises of the Commencement Date or Paragraph 7.1(b) below are obtained by Lesse of the Premises o	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered a also Paragraph 37) al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be r the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading addition on said date, that the ises do not contain hazardous levels he Start Date, or if one of such systems to to such matter, except as otherwise such non-compliance, malfunction or 0 days as to the remaining systems correction of any such to the fire sprinkler systems, roof,
Lessee's Brokerage Firm License No. is (check one): the Lessee's Agent License No. is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lecovenants and conditions set forth in this Lease. While the approximate square footage ardifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lecovenants and condition. Lessor shall deliver the Premises to Lessee broom clean and from first occurs ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises, other than those construct structural elements of the roof, bearing walls and foundation of the Premises shall be foany mold or fungi defined as toxic under applicable state or federal law. If a non-coor elements should malfunction or fail within the appropriate warranty period, Lessor provided in this Lease, promptly after receipt of written notice from Lessee setting for failure, rectify same at Lessor's expense. The warranty periods shall be as follows: (i) and other elements of the Premises. If Lessee does not give Lessor the required notic non-compliance, malfunction or failure shall be the obligation of Lessee at Lessee's so foundations, and/or bearing walls). Lessor also warrants, that unless	ssor, the Premises, for the term, at the rents ge of the Premises may have been used in the dis not subject to adjustment should the a ase. see of debris on the Commencement Date or Paragraph 7.1(b) below are obtained by Les lighting, heating, ventilating and air conditied by Lessee, shall be in good operating confree of material defects, and that the Premompliance with such warranty exists as of the shall, as Lessor's sole obligation with respect the with specificity the nature and extent of 6 months as to the HVAC systems, and (ii) 3 the within the appropriate warranty period, colle cost and expense (except for the repairs fied in writing, Lessor is unaware of (ii) any respect to the response of the repairs fied in writing, Lessor is unaware of (ii) any respective the state of the repairs fied in writing, Lessor is unaware of (ii) any respective the response of the repairs fied in writing, Lessor is unaware of (ii) any respective the response of the repairs fied in writing, Lessor is unaware of (ii) any respective the response of the repairs fied in writing, Lessor is unaware of (ii) any respective the response of the repairs fied in writing, Lessor is unaware of (ii) any respective the response of the response of the repairs fied in writing, Lessor is unaware of (ii) any response of the response of the response of the repairs field in writing, Lessor is unaware of (ii) any response of the respon	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered a also Paragraph 37) al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be r the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading addition on said date, that the ises do not contain hazardous levels he Start Date, or if one of such systems to to such matter, except as otherwise such non-compliance, malfunction or 0 days as to the remaining systems correction of any such to the fire sprinkler systems, roof,
Lessee's Brokerage Firm License No. is (check one): the Lessee's Agent License No. is (check one): the Agent and the Lessor's Agent (dual agent). (b) Payment to Brokers. Upon execution and delivery of this Lease by be separate written agreement (or if there is no such agreement, the sum of by the Brokers. 1.14 Guarantor. The obligations of the Lessee under this Lease are to be guarant 1.15 Attachments. Attached hereto are the following, all of which constitute a an Addendum consisting of Paragraphs through a site plan marked Exhibit depicting the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a current set of the Rules and Regulations for the Shopping Center; a Work Letter; other (specify): 2. Premises. 2.1 Letting. Lessor hereby leases to Lessee, and Lessee hereby leases from Lecovenants and conditions set forth in this Lease. While the approximate square footage ardifferent. NOTE: Lessee is advised to verify the actual size prior to executing this Lecovenants and condition. Lessor shall deliver the Premises to Lessee broom clean and from first occurs ("Start Date"), and, so long as the required service contracts described in following the Start Date, warrants that the existing electrical, plumbing, fire sprinkler, doors, if any, and all other such elements in the Premises, other than those construct structural elements of the roof, bearing walls and foundation of the Premises shall be foany mold or fungi defined as toxic under applicable state or federal law. If a non-coor elements should malfunction or fail within the appropriate warranty period, Lessor provided in this Lease, promptly after receipt of written notice from Lessee setting for failure, rectify same at Lessor's expense. The warranty periods shall be as follows: (i) and other elements of the Premises. If Lessee does not give Lessor the required notic non-compliance, malfunction or failure shall be the obligation of Lessee at Lessee's so foundations, and/or bearing walls). Lessor also warrants, that unless	ssor, the Premises, for the term, at the rentage of the Premises may have been used in the dis not subject to adjustment should the a ase. ee of debris on the Commencement Date or Paragraph 7.1(b) below are obtained by Les lighting, heating, ventilating and air conditied by Lessee, shall be in good operating confree of material defects, and that the Premise the with specificity the nature and extent of 6 months as to the HVAC systems, and (ii) 3 see within the appropriate warranty period, cole cost and expense (except for the repairs field in writing, Lessor is unaware of (i) any reministration.	both the Lessee's the brokerage fee agreed to in a t) for the brokerage services rendered a also Paragraph 37) al, and upon all of the terms, the marketing of the Premises for ctual size be determined to be r the Early Possession Date, whichever see and in effect within 30 days oning systems ("HVAC"), loading addition on said date, that the ises do not contain hazardous levels he Start Date, or if one of such systems to to such matter, except as otherwise such non-compliance, malfunction or 0 days as to the remaining systems correction of any such to the fire sprinkler systems, roof,

SCLN-20.30, Revised 10-22-2020

the Premise; (ii) any delinquent amounts due under any loan secured by the Premises; and (iii) any bankruptcy proceeding affecting the Premises.

- 2.3 Compliance. Lessor warrants that to the best of its knowledge the improvements on the Premises and the Common Areas comply with the building codes, applicable laws, covenants or restrictions of record, regulations, and ordinances ("Applicable Requirements") that were in effect at the time that each improvement, or portion thereof, was constructed. Said warranty does not apply to the use to which Lessee will put the Premises, modifications which may be required by the Americans with Disabilities Act or any similar laws as a result of Lessee's use (see Paragraph 50), or to any Alterations or Utility Installations (as defined in Paragraph 7.3(a)) made or to be made by Lessee. NOTE: Lessee is responsible for determining whether or not the Applicable Requirements, and especially the zoning, are appropriate for Lessee's intended use, and acknowledges that past uses of the Premises may no longer be allowed. If the Premises do not comply with said warranty, Lessor shall, except as otherwise provided, promptly after receipt of written notice from Lessee setting forth with specificity the nature and extent of such non-compliance, rectify the same at Lessor's expense. If Lessee does not give Lessor written notice of a non-compliance with this warranty within 6 months following the Start Date, correction of that non-compliance shall be the obligation of Lessee at Lessee's sole cost and expense. If the Applicable Requirements are hereafter changed so as to require during the term of this Lease the construction of an addition to or an alteration of the Premises and/or Building, the remediation of any Hazardous Substance, or the reinforcement or other physical modification of the Premises and/or Building ("Capital Expenditure"), Lessor and Lessee shall allocate the cost of such work as follows:
- (a) Subject to Paragraph 2.3(c) below, if such Capital Expenditures are required as a result of the specific and unique use of the Premises by Lessee as compared with uses by tenants in general, Lessee shall be fully responsible for the cost thereof, provided, however, that if such Capital Expenditure is required during the last 2 years of this Lease and the cost thereof exceeds 6 months' Base Rent, Lessee may instead terminate this Lease unless Lessor notifies Lessee, in writing, within 10 days after receipt of Lessee's termination notice that Lessor has elected to pay the difference between the actual cost thereof and the amount equal to 6 months' Base Rent. If Lessee elects termination, Lessee shall immediately cease the use of the Premises which requires such Capital Expenditure and deliver to Lessor written notice specifying a termination date at least 90 days thereafter. Such termination date shall, however, in no event be earlier than the last day that Lessee could legally utilize the Premises without commencing such Capital Expenditure.
- (b) If such Capital Expenditure is not the result of the specific and unique use of the Premises by Lessee (such as, governmentally mandated seismic modifications), then Lessor shall pay for such Capital Expenditure and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date that on which the Base Rent is due, an amount equal to 1/144th of the portion of such costs reasonably attributable to the Premises. Lessee shall pay Interest on the balance but may prepay its obligation at any time. If, however, such Capital Expenditure is required during the last 2 years of this Lease or if Lessor reasonably determines that it is not economically feasible to pay its share thereof, Lessor shall have the option to terminate this Lease upon 90 days prior written notice to Lessee unless Lessee notifies Lessor, in writing, within 10 days after receipt of Lessor's termination notice that Lessee will pay for such Capital Expenditure. If Lessor does not elect to terminate, and fails to tender its share of any such Capital Expenditure, Lessee may advance such funds and deduct same, with Interest, from Rent until Lessor's share of such costs have been fully paid. If Lessee is unable to finance Lessor's share, or if the balance of the Rent due and payable for the remainder of this Lease is not sufficient to fully reimburse Lessee on an offset basis, Lessee shall have the right to terminate this Lease upon 30 days written notice to Lessor.
- (c) Notwithstanding the above, the provisions concerning Capital Expenditures are intended to apply only to non-voluntary, unexpected, and new Applicable Requirements. If the Capital Expenditures are instead triggered by Lessee as a result of an actual or proposed change in use, change in intensity of use, or modification to the Premises then, and in that event, Lessee shall either: (i) immediately cease such changed use or intensity of use and/or take such other steps as may be necessary to eliminate the requirement for such Capital Expenditure, or (ii) complete such Capital Expenditure at its own expense. Lessee shall not have any right to terminate this Lease.
- 2.4 **Acknowledgements**. Lessee acknowledges that: (a) it has been given an opportunity to inspect and measure the Premises; (b) it has been advised by Lessor and/or Brokers to satisfy itself with respect to the size and condition of the Premises (including but not limited to the electrical, HVAC and fire sprinkler systems, security, environmental aspects, and compliance with Applicable Requirements and the Americans with Disabilities Act) and their suitability for Lessee's intended use; (c) Lessee has made such investigation as it deems necessary with reference to such matters and assumes all responsibility therefor as the same relate to its occupancy of the Premises; (d) it is not relying on any representation as to the size of the Premises made by Brokers or Lessor; (e) the square footage of the Premises was not material to Lessee's decision to lease the Premises and pay the Rent stated herein; and (f) neither Lessor, Lessor's agents, nor Brokers have made any oral or written representations or warranties with respect to said matters other than as set forth in this Lease. In addition, Lessor acknowledges that: (i) Brokers have made no representations, promises or warranties concerning Lessee's ability to honor the Lease or suitability to occupy the Premises, and (ii) it is Lessor's sole responsibility to investigate the financial capability and/or suitability of all proposed tenants.
- 2.5 **Lessee as Prior Owner/Occupant**. The warranties made by Lessor in Paragraph 2 shall be of no force or effect if immediately prior to the Start Date Lessee was the owner or occupant of the Premises. In such event, Lessee shall be responsible for any necessary corrective work.
- 2.6 **Vehicle Parking**. Lessee shall not use and shall not permit its employees to use any parking spaces in the Shopping Center except for parking by vehicles that are no larger than full-size passenger automobiles or pick-up trucks, herein called "**Permitted Size Vehicles**." Lessee shall permit its employees to only occupy those parking spaces, if any, as depicted as employee parking spaces on the Shopping Center site plan. Lessor may regulate the loading and unloading of vehicles by adopting Rules and Regulations as provided in Paragraph 2.9. No vehicles other than Permitted Size Vehicles may be parked in the Common Area without the prior written permission of Lessor. In addition:
- (a) Lessee shall not permit or allow any vehicles that belong to or are controlled by Lessee or Lessee's employees, suppliers, shippers, contractors or invitees to be loaded, unloaded, or parked in areas other than those designated by Lessor for such activities.
 - (b) Lessee shall not service or store any vehicles in the Common Areas.
- (c) If Lessee permits or allows any of the prohibited activities described in this Paragraph 2.6, then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove or tow away the vehicle involved and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.
- 2.7 Common Areas Definition. The term "Common Areas" is defined as all areas and facilities outside the Premises and within the exterior boundary line of the Shopping Center and interior utility raceways and installations within the Premises that are provided and designated by the Lessor from time to time for the general non-exclusive use of Lessor, Lessee and other tenants of the Shopping Center and their respective employees, suppliers, shippers, customers, contractors and invitees, including parking areas, loading and unloading areas, trash areas, roadways, walkways, driveways and landscaped areas.
- 2.8 **Common Areas Lessee's Rights.** Lessor grants to Lessee, for the benefit of Lessee and its employees, suppliers, shippers, contractors, customers and invitees, during the term of this Lease, the non-exclusive right to use, in common with others entitled to such use, the Common Areas as they exist from time to time, subject to any rights, powers, and privileges reserved by Lessor under the terms hereof or under the terms of any rules and regulations or restrictions governing the use of the Shopping Center. Under no circumstances shall the right herein granted to use the Common Areas be deemed to include the right to store any property, temporarily or permanently, in the Common Areas, nor the right to display merchandise or conduct sales in the Common Areas. Any such storage, display or sales shall be permitted only by the prior written consent of Lessor or Lessor's designated agent, as exercised in Lessor's sole discretion, which consent may be revoked at any time. In the event that any unauthorized storage or displays shall occur then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove the property and charge the cost to Lessee, which cost shall be immediately payable upon demand by Lessor.
- 2.9 Common Areas Rules and Regulations. Lessor or such other person(s) as Lessor may appoint shall have the exclusive control and management of the Common Areas and shall have the right, from time to time, to establish, modify, amend and enforce reasonable rules and regulations ("Rules and Regulations") for the management, safety, care, and cleanliness of the grounds, the parking and unloading of vehicles and the preservation of good order, as well as for the convenience of other occupants or tenants of the Building and the Shopping Center and their invitees. Lessee agrees to abide by and conform to all such Rules and Regulations, and shall use its best efforts to cause its employees, suppliers, shippers, customers, contractors and invitees to so abide and conform. Lessor shall not be responsible to Lessee for the non-compliance with said Rules and Regulations by other tenants of the Shopping Center.
 - 2.10 **Common Areas Changes**. Lessor shall have the right, in Lessor's sole discretion, from time to time:
- (a) To make changes or additions to the Common Areas, including, without limitation, changes in the location, size, shape and number of driveways, entrances, parking spaces, parking areas, loading and unloading areas, ingress, egress, direction of traffic, elevations, landscaped areas, signage, walkways and utility rareways:
- (b) To use and close temporarily any of the Common Areas for the purpose of maintaining, repairing and altering the Shopping Center, so long as reasonable access to the Premises remains available, and to close temporarily any of the Common Areas to whatever extent is required in the opinion of Lessor's counsel to prevent a dedication of or the accrual of any rights of any persons or of the public to any of the Common Areas;
- (c) To designate other land outside the boundaries of the Shopping Center to be a part of the Common Areas or to be entitled to use the Common Areas on a reciprocal basis;
 - (d) To add additional buildings and improvements to the Common Areas; and
- (e) To do and perform such other acts and make such other changes in, to or with respect to the Common Areas and Shopping Center as Lessor may, in the exercise of sound business judgment, deem to be appropriate.
- 2.11 Common Areas Promotional Events; Sidewalk Sales. Lessor reserves the right, from time to time, in Lessor's sole discretion, to utilize portions of the Common Areas for promotional events, which may include but shall not be limited to entertainment. Lessor further reserves the right, in Lessor's sole discretion, to

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

permit any one or more tenants of the Shopping Center to conduct the display and/or sale of merchandise from the sidewalks immediately adjacent to such tenants' respective premises.

2.12 **Common Areas - Remodeling**. At any time during the Term, Lessor may remodel or expand, in any manner, the existing Shopping Center, which work may include, without limitation, the addition of shops and/or new buildings to the Shopping Center (collectively, "**Remodeled Center**"). If Lessor deems it necessary for construction personnel to enter the Premises in order to construct the Remodeled Center, Lessor shall give Lessee no less than 60 days prior notice and Lessee shall allow such entry. Lessor shall use reasonable efforts to complete any work affecting the Premises in an efficient manner so as not to interfere unreasonably with Lessee's business. Lessee shall not be entitled to any damages for any inconvenience or any disruption to Lessee's business caused by such work; provided, however, the Base Rent paid by Lessee for the period of the inconvenience shall be abated in proportion to the degree that Lessee's use of the Premises is impaired. Lessor shall have the right to use portions of the Premises to accommodate any structures required for the Remodeled Center, provided that if as a result thereof there is a permanent decrease in the floor area of the Premises of 3% or more, there shall be a proportionate downward adjustment of Base Rent and Lessee's Share.

3. Term.

- 3.1 Term. The Commencement Date, Expiration Date and Original Term of this Lease are as specified in Paragraph 1.3.
- 3.2 **Early Possession**. Any provision herein granting Lessee Early Possession of the Premises is subject to and conditioned upon the Premises being available for such possession prior to the Commencement Date. Any grant of Early Possession only conveys a non-exclusive right to occupy the Premises. If an Early Possession Date has been specified in Paragraph 1.4, the Parties intend that Lessee shall have access to the Premises as of the Early Possession Date for purposes of preparing and fixturizing the Premises for the conduct of Lessee's business. If Lessee totally or partially occupies the Premises prior to the Commencement Date for any reason (and for purposes hereof, "occupancy" shall include, without limitation, Lessee's entry onto the Premises for purposes of preparing and fixturizing the Premises for business), the obligation to pay Base Rent and Percentage Rent shall be abated for the period of such early possession. All other terms of this Lease (including but not limited to Lessee's obligations to carry insurance and to maintain the Premises) shall be in effect during such period, except that Lessee's obligations to pay Lessee's Share of Common Area Operating Expenses, Real Property Taxes and insurance premiums shall only be in effect prior to the Commencement Date if Lessee has opened for business in the Premises prior to the Commencement Date. Any such Early Possession shall not affect the Expiration Date.
- 3.3 **Delay In Possession**. Lessor agrees to use commercially reasonable efforts to deliver exclusive possession of the Premises to Lessee by the Commencement Date. If, despite said efforts, Lessor is unable to deliver possession by such date, Lessor shall not be subject to any liability therefor, nor shall such failure affect the validity of this Lease or change the Expiration Date. Lessee shall not, however, be obligated to pay Rent or perform its other obligations until Lessor delivers possession of the Premises and any period of rent abatement that Lessee would otherwise have enjoyed shall run from the date of delivery of possession and continue for a period equal to what Lessee would otherwise have enjoyed under the terms hereof, but minus any days of delay caused by the acts or omissions of Lessee. If possession is not delivered within 60 days after the Commencement Date, as the same may be extended under the terms of any Work Letter executed by Parties, Lessee may, at its option, by notice in writing within 10 days after the end of such 60 day period, cancel this Lease, in which event the Parties shall be discharged from all obligations hereunder. If such written notice is not received by Lessor within said 10 day period, Lessee's right to cancel shall terminate. If possession of the Premises is not delivered within 120 days after the Commencement Date, this Lease shall terminate unless other agreements are reached between Lessor and Lessee, in writing.
- 3.4 Lessee Compliance. Lessor shall not be required to tender possession of the Premises to Lessee until Lessee complies with its obligation to provide evidence of insurance (Paragraph 8.5). Pending delivery of such evidence, Lessee shall be required to perform all of its obligations under this Lease from and after the Start Date, including the payment of Rent, notwithstanding Lessor's election to withhold possession pending receipt of such evidence of insurance. Further, if Lessee is required to perform any other conditions prior to or concurrent with the Start Date, the Start Date shall occur but Lessor may elect to withhold possession until such conditions are satisfied.

4. Rent.

- 4.1 Rent Defined. All monetary obligations of Lessee to Lessor under the terms of this Lease (except for the Security Deposit) are deemed to be rent ("Rent").
- 4.2 **Common Area Operating Expenses.** Lessee shall pay to Lessor during the term hereof, in addition to the Base Rent and, if applicable, Percentage Rent, Lessee's Share (as specified in Paragraph 1.7) of all Common Area Operating Expenses, as hereinafter defined, during each calendar year of the term of this Lease, in accordance with the following provisions:
- (a) "Common Area Operating Expenses" are defined, for purposes of this Lease, as all costs relating to the ownership and operation of the Shopping Center, including, but not limited to, the following:
- (i) Costs relating to the operation, repair and maintenance, in neat, clean, good order and condition, and replacement as reasonably necessary, of the following:
- (aa) The Common Areas and Common Area improvements, including parking areas, loading and unloading areas, trash areas, roadways, parkways, walkways, driveways, landscaped areas, parking lot striping, bumpers, irrigation systems, Common Area lighting facilities, fences and gates, elevators, roofs, exterior walls of the buildings, building systems and roof drainage systems.
 - (bb) Exterior signs and any tenant directories.
 - (cc) Any fire detection and/or sprinkler systems.
 - (dd) Common electrical, plumbing and other utilities servicing any building in the Shopping Center and/or the Common Areas.
- (ee) All other areas and improvements that are within the exterior boundaries of the Project but outside of the Premises and/or any other space occupied by a tenant.
 - (ii) The cost of water, gas, electricity and telephone to service the Common Areas and any utilities not separately metered.
- (iii) The cost of trash disposal, pest control services, property management (including, but not be limited to, a property management fee to Lessor
- equal to 5% of Base Rent and Percentage Rent), security services, and the costs of any environmental inspections.

 (iv) Reserves set aside for equipment, maintenance, repair and replacement of Common Areas.
 - (v) Real Property Taxes (as defined in Paragraph 10).
 - (vi) The cost of the premiums for the insurance maintained by Lessor pursuant to Paragraph 8.
 - $\begin{tabular}{ll} \begin{tabular}{ll} \beg$
 - (viii) Auditors', accountants' and attorneys' fees and costs related to the operation of the Shopping Center.
- (ix) The cost of any capital improvement to the Building or the Shopping Center not covered under the provisions of Paragraph 2.3, provided, however, that Lessor shall allocate the cost of any such capital improvement over a 12 year period and Lessee shall not be required to pay more than Lessee's Share of 1/144th of the cost of such capital improvement in any given month. Lessee shall pay Interest on the unamortized balance but may prepay its obligation at any time.
 - (x) The cost of any other services to be provided by Lessor that are stated elsewhere in this Lease to be a Common Area Operating Expense.
- (b) If Lessor determines that the method of proration of any item included within Common Area Operating Expenses is inequitable, Lessor may prorate such item on the basis of usage or other equitable considerations. Any Common Area Operating Expenses and Real Property Taxes that are specifically attributable to the Premises, the Building or to any other premises or building in the Shopping Center or to the operation, repair and maintenance thereof, shall be allocated entirely to such premises or building. However, any Common Area Operating Expenses and Real Property Taxes that are not specifically attributable to any premises or building or to the operation, repair and maintenance thereof, shall be equitably allocated by Lessor to all buildings in the Shopping Center.
- (c) The inclusion of the improvements, facilities and services set forth in Subparagraph 4.2(a) shall not be deemed to impose an obligation upon Lessor to either have said improvements or facilities or to provide those services unless the Shopping Center already has the same, Lessor already provides the services, or Lessor has agreed elsewhere in this Lease to provide the same or some of them.
- (d) Lessee's Share of Common Area Operating Expenses is payable monthly on the same day as the Base Rent is due hereunder. The amount of such payments shall be based on Lessor's estimate of the annual Common Area Operating Expenses. Within 60 days after written request (but not more than once each year) Lessor shall deliver to Lessee a reasonably detailed statement showing Lessee's Share of the actual Common Area Operating Expenses for the preceding year. If Lessee's payments during such year exceed Lessee's Share, Lessor shall credit the amount of such over-payment against Lessee's future payments. If Lessee's payments during such year were less than Lessee's Share, Lessee shall pay to Lessor the amount of the deficiency within 10 days after delivery by Lessor to Lessee of the statement.
- (e) If there are one or more Major Tenants (as hereinafter defined) within the Shopping Center, then at Lessor's sole option, the amount to be reimbursed by such Major Tenants to Lessor for all or a portion of the Common Area Operating Expenses may be determined by alternative equitable methods (e.g., a Major Tenant may pay directly for its own security), and the actual amount paid by such Major Tenants shall be credited against the Common Area Operating Expenses allocated to other tenants of the Shopping Center; provided, however, that in such event the rentable area of the buildings leased to such Major Tenants shall be excluded from the rentable area of the Shopping Center for purposes of determining Lessee's Share of Common Area Operating Expenses for those specific items, notwithstanding the percentage set forth in Paragraph 1.7. As used herein, the term "Major Tenant" shall mean a tenant leasing at least 15,000 square feet of rentable area within the Shopping Center.
- (f) Common Area Operating Expenses shall not include any expenses paid by any tenant directly to third parties, or as to which Lessor is otherwise reimbursed by any third party, other tenant, or insurance proceeds.

INITIALS	
----------	--

- 4.3 **Payment.** Lessee shall cause payment of Rent to be received by Lessor in lawful money of the United States, without offset or deduction (except as specifically permitted in this Lease), on or before the day on which it is due. All monetary amounts shall be rounded to the nearest whole dollar. In the event that any invoice prepared by Lessor is inaccurate such inaccuracy shall not constitute a waiver and Lessee shall be obligated to pay the amount set forth in this Lease. Rent for any period during the term hereof which is for less than one full calendar month shall be prorated based upon the actual number of days of said month. Payment of Rent shall be made to Lessor at its address stated herein or to such other persons or place as Lessor may from time to time designate in writing. Acceptance of a payment which is less than the amount then due shall not be a waiver of Lessor's rights to the balance of such Rent, regardless of Lessor's endorsement of any check so stating. In the event that any check, draft, or other instrument of payment given by Lessee to Lessor is dishonored for any reason, Lessee agrees to pay to Lessor the sum of \$25 in addition to any Late Charge and Lessor, at its option, may require all future Rent be paid by cashier's check. Payments will be applied first to accrued late charges and attorney's fees, second to accrued interest, then to Base Rent and Common Area Operating Expenses, and any remaining amount to any other outstanding charges or costs.
- Security Deposit. Lessee shall deposit with Lessor upon execution hereof the Security Deposit as security for Lessee's faithful performance of its obligations under this Lease. If Lessee fails to pay Rent, or otherwise Defaults under this Lease, Lessor may use, apply or retain all or any portion of said Security Deposit for the payment of any amount already due Lessor, for Rents which will be due in the future, and/or to reimburse or compensate Lessor for any liability, expense, loss or damage which Lessor may suffer or incur by reason thereof. If Lessor uses or applies all or any portion of the Security Deposit, Lessee shall within 10 days after written request therefor deposit monies with Lessor sufficient to restore said Security Deposit to the full amount required by this Lease. If the Base Rent increases during the term of this Lease, Lessee shall, upon written request from Lessor, deposit additional monies with Lessor so that the total amount of the Security Deposit shall at all times bear the same proportion to the increased Base Rent as the initial Security Deposit bore to the initial Base Rent. Should the Agreed Use be amended to accommodate a material change in the business of Lessee or to accommodate a sublessee or assignee, Lessor shall have the right to increase the Security Deposit to the extent necessary, in Lessor's reasonable judgment, to account for any increased wear and tear that the Premises may suffer as a result thereof. If a change in control of Lessee occurs during this Lease and following such change the financial condition of Lessee is, in Lessor's reasonable judgment, significantly reduced, Lessee shall deposit such additional monies with Lessor as shall be sufficient to cause the Security Deposit to be at a commercially reasonable level based on such change in financial condition. Lessor shall not be required to keep the Security Deposit separate from its general accounts. Within 90 days after the expiration or termination of this Lease, Lessor shall return that portion of the Security Deposit not used or applied by Lessor. Lessor shall upon written request provide Lessee with an accounting showing how that portion of the Security Deposit that was not returned was applied. No part of the Security Deposit shall be considered to be held in trust, to bear interest or to be prepayment for any monies to be paid by Lessee under this Lease. THE SECURITY DEPOSIT SHALL NOT BE USED BY LESSEE IN LIEU OF PAYMENT OF THE LAST MONTH'S RENT.

Use.

6.1 Use.

- (a) Agreed Use; Agreed Trade Name. Lessee shall use and occupy the Premises only for the Agreed Use, and for no other purpose, and Lessee shall operate at the Premises only under the Agreed Trade Name and under no other trade name. Lessee shall not use or permit the use of the Premises in a manner that is unlawful, creates damage, waste or a nuisance, or that disturbs occupants of or causes damage to neighboring premises or properties. Other than guide, signal and seeing eye dogs, Lessee shall not keep or allow in the Premises any pets, animals, birds, fish, or reptiles. Lessor shall not unreasonably withhold or delay its consent to any written request for a modification of the Agreed Use, so long as the same will not impair the structural integrity of the improvement on the Premises or the mechanical or electrical systems therein, and/or is not significantly more burdensome to the Premises, and/or is not in conflict with or incompatible with the existing or proposed uses (whether or not exclusive) of other occupants of the Shopping Center. Lessor shall not unreasonably withhold or delay its consent to any written request for a modification of the Agreed Trade Name, so long as the same is not in conflict with or incompatible with the nature and character of the Shopping Center or other existing or proposed uses of other occupants of the Shopping Center. If Lessor elects to withhold consent, Lessor shall within 7 days after such request give written notification of same, which notice shall include an explanation of Lessor's objections to the change in the Agreed Use and/or Agreed Trade Name.
- (b) **Continuous Operation**. Lessee shall continuously (i) operate and conduct the Agreed Use under the Agreed Trade Name within the entire Premises in a reputable manner and in conformity with industry standards of practice prevailing in the field of business among merchants engaged in the same or similar business in the city in which the Premises are located, (ii) staff the Premises with sufficient sales personnel, stock the Premises with adequate merchandise and exercise sound business practices so as to maximize Gross Sales for the benefit of Lessor. At a minimum, Lessee shall keep the Premises continuously open for business Monday through Friday from 9:00 a.m. to 6:00 p.m., Saturday from 9:00 a.m. to 6:00 p.m., Saturday from 9:00 a.m. to 6:00 p.m. If Lessee fails to comply with the requirements of this Paragraph 6.1(b), then in addition to any and all other rights and remedies of Lessor, Lessee shall pay to Lessor an amount equal to 1/15th of the Base Rent for each day or portion thereof that Lessee fails to so comply. Such sum shall be in addition to, and not a part of, the Base Rent otherwise due under this Lesse.
- (c) Violations of Exclusive Use Rights. Lessee acknowledges that Lessor may grant, or may have previously granted, exclusive use rights to other tenants of the Shopping Center and agrees that a material consideration to Lessor in entering into this Lease is Lessee's covenant to limit its use of the Premises to the Agreed Use under the Agreed Trade Name as set forth above. Lessee's violation of exclusive use rights granted to other tenants of the Shopping Center will result in Lessor suffering irreparable harm and, therefore, in addition to all other rights and remedies available to Lessor, Lessor may seek to enjoin Lessee's breach of such covenant and Lessee shall be liable for any damages incurred or sustained by Lessor to such other tenants whose exclusive use rights are breached by Lessee. In no event shall Lessor be liable to Lessee for any failure of any other tenants of the Shopping Center to operate their businesses, or for any loss or damage that may be occasioned by or through the acts or omissions of other tenants or third parties.
- (d) Other Tenancies. Lessor, at its sole discretion, reserves the absolute right to establish procedures to control other tenancies in the Shopping Center. Regardless of whether any specific tenants are shown on any site plan attached hereto, Lessee does not rely on that fact, nor does Lessor represent that any specific tenant or number or type of tenants shall or shall not during the Term occupy any portion of the Shopping Center, nor does Lessee rely on any other tenant operating its business in the Shopping Center at any particular time or times. Further, no conduct by any tenant, subtenant or other occupant of, or any customer of, or any supplier to or use of any portion of the Shopping Center shall constitute an eviction, constructive or otherwise, of Lessee from the Premises, and Lessee hereby waives any and all claims that it might otherwise have against Lessor by reason thereof.

6.2 Hazardous Substances.

- (a) Reportable Uses Require Consent. The term "Hazardous Substance" as used in this Lease shall mean any product, substance, or waste whose presence, use, manufacture, disposal, transportation, or release, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentially injurious to the public health, safety or welfare, the environment or the Premises, (ii) regulated or monitored by any governmental authority, or (iii) a basis for potential liability of Lessor to any governmental agency or third party under any applicable statute or common law theory. Hazardous Substances shall include, but not be limited to, hydrocarbons, petroleum, gasoline, and/or crude oil or any products, by-products or fractions thereof. Lessee shall not engage in any activity in or on the Premises which constitutes a Reportable Use of Hazardous Substances without the express prior written consent of Lessor and timely compliance (at Lessee's expense) with all Applicable Requirements. "Reportable Use" shall mean (i) the installation or use of any above or below ground storage tank, (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and/or (iii) the presence at the Premises of a Hazardous Substance with respect to which any Applicable Requirements requires that a notice be given to persons entering or occupying the Premises or neighboring properties. Notwithstanding the foregoing, Lessee may use any ordinary and customary materials reasonably required to be used in the normal course of the Agreed Use, ordinary office supplies (copier toner, liquid paper, glue, etc.) and common household cleaning materials, so long as such use is in compliance with all Applicable Requirements, is not a Reportable Use, and does not expose the Premises or neighboring property to any meaningful risk of cont
- (b) **Duty to Inform Lessor**. If Lessee knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises, other than as previously consented to by Lessor, Lessee shall immediately give written notice of such fact to Lessor, and provide Lessor with a copy of any report, notice, claim or other documentation which it has concerning the presence of such Hazardous Substance.
- (c) Lessee Remediation. Lessee shall not cause or permit any Hazardous Substance to be spilled or released in, on, under, or about the Premises (including through the plumbing or sanitary sewer system) and shall promptly, at Lessee's expense, comply with all Applicable Requirements and take all investigatory and/or remedial action reasonably recommended, whether or not formally ordered or required, for the cleanup of any contamination of, and for the maintenance, security and/or monitoring of the Premises or neighboring properties, that was caused or materially contributed to by Lessee, or pertaining to or involving any Hazardous Substance brought onto the Premises during the term of this Lease, by or for Lessee, or any third party.
- (d) Lessee Indemnification. Lessee shall indemnify, defend and hold Lessor, its agents, employees, lenders and ground lessor, if any, harmless from and against any and all loss of rents and/or damages, liabilities, judgments, claims, expenses, penalties, and attorneys' and consultants' fees arising out of or involving any Hazardous Substance brought onto the Premises by or for Lessee, or any third party (provided, however, that Lessee shall have no liability under this Lease with respect to underground migration of any Hazardous Substance under the Premises from areas outside of the Shopping Center not caused or contributed to by Lessee).

	Ī	NI	ΤI	ΑI	S
--	---	----	----	----	---

Lessee's obligations shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Lessee, and the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease. No termination, cancellation or release agreement entered into by Lessor and Lessee shall release Lessee from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Lessor in writing at the time of such agreement.

- (e) Lessor Indemnification. Except as otherwise provided in paragraph 8.7, Lessor and its successors and assigns shall indemnify, defend, reimburse and hold Lessee, its employees and lenders, harmless from and against any and all environmental damages, including the cost of remediation, which are suffered as a direct result of Hazardous Substances on the Premises prior to Lessee taking possession or which are caused by the gross negligence or willful misconduct of Lessor, its agents or employees. Lessor's obligations, as and when required by the Applicable Requirements, shall include, but not be limited to, the cost of investigation, removal, remediation, restoration and/or abatement, and shall survive the expiration or termination of this Lease.
- (f) Investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by governmental entities having jurisdiction with respect to the existence of Hazardous Substances on the Premises prior to Lessee taking possession, unless such remediation measure is required as a result of Lessee's use (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises, in which event Lessee shall be responsible for such payment. Lessee shall cooperate fully in any such activities at the request of Lessor, including allowing Lessor and Lessor's agents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remedial responsibilities.
- (g) Lessor Termination Option. If a Hazardous Substance Condition (see Paragraph 9.1(e)) occurs during the term of this Lease, unless Lessee is legally responsible therefor (in which case Lessee shall make the investigation and remediation thereof required by the Applicable Requirements and this Lease shall continue in full force and effect, but subject to Lessor's rights under Paragraph 6.2(d) and Paragraph 13), Lessor may, at Lessor's option, either (i) investigate and remediate such Hazardous Substance Condition, if required, as soon as reasonably possible at Lessor's expense, in which event this Lease shall continue in full force and effect, or (ii) if the estimated cost to remediate such condition exceeds 12 times the then monthly Base Rent or \$100,000, whichever is greater, give written notice to Lessee, within 30 days after receipt by Lessor of knowledge of the occurrence of such Hazardous Substance Condition, of Lessor's desire to terminate this Lease as of the date 60 days following the date of such notice. In the event Lessor elects to give a termination notice, Lessee may, within 10 days thereafter, give written notice to Lessor of Lessee's commitment to pay the amount by which the cost of the remediation of such Hazardous Substance Condition exceeds an amount equal to 12 times the then monthly Base Rent or \$100,000, whichever is greater. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days following such commitment. In such event, this Lease shall continue in full force and effect, and Lessor shall proceed to make such remediation as soon as reasonably possible after the required funds are available. If Lessee does not give such notice and provide the required funds or assurance thereof within the time provided, this Lease shall terminate as of the date specified in Lessor's notice of termination.
- 6.3 Lessee's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lessee's sole expense, fully, diligently and in a timely manner, materially comply with all Applicable Requirements, the requirements of any applicable fire insurance underwriter or rating bureau, and the recommendations of Lessor's engineers and/or consultants which relate in any manner to the Premises, without regard to whether said Applicable Requirements are now in effect or become effective after the Start Date. Lessee shall, within 10 days after receipt of Lessor's written request, provide Lessor with copies of all permits and other documents, and other information evidencing Lessee's compliance with any Applicable Requirements specified by Lessor, and shall immediately upon receipt, notify Lessor in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving the failure of Lessee or the Premises to comply with any Applicable Requirements. Likewise, Lessee shall immediately give written notice to Lessor of: (i) any water damage to the Premises and any suspected seepage, pooling, dampness or other condition conducive to the production of mold; or (ii) any mustiness or other odors that might indicate the presence of mold in the Premises.
- 6.4 Inspection; Compliance. Lessor and Lessor's "Lender" (as defined in Paragraph 30) and consultants authorized by Lessor shall have the right to enter into Premises at any time in the case of an emergency, and otherwise at reasonable times after reasonable notice, for the purpose of inspecting and/or testing the condition of the Premises and/or for verifying compliance by Lessee with this Lease. The cost of any such inspections shall be paid by Lessor, unless a violation of Applicable Requirements, or a Hazardous Substance Condition (see Paragraph 9.1(e)) is found to exist or be imminent, or the inspection is requested or ordered by a governmental authority. In such case, Lessee shall upon request reimburse Lessor for the cost of such inspection, so long as such inspection is reasonably related to the violation or contamination. In addition, Lessee shall provide copies of all relevant material safety data sheets (MSDS) to Lessor within 10 days of the receipt of written request therefor. Lessee acknowledges that any failure on its part to allow such inspections or testing will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to allow such inspections and/or testing in a timely fashion the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for the remainder to the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to allow such inspection and/or testing. Such increase in Base Rent shall be remainded to such failure nor prevent the exercise of any of the other rights and remedies granted becaused to such failure on prevent the exercise of any of the other rights and remedies granted becaused to such failure on pr

7. Maintenance; Repairs; Utility Installations; Trade Fixtures and Alterations.

7.1 Lessee's Obligations.

- (a) In General. Subject to the provisions of Paragraph 2.2 (Condition), 2.3 (Compliance), 6.3 (Lessee's Compliance with Applicable Requirements), 7.2 (Lessor's Obligations), 9 (Damage or Destruction), and 14 (Condemnation), Lessee shall, at Lessee's sole expense, keep the Premises, Utility Installations (intended for Lessee's exclusive use, no matter where located), and Alterations in good order, condition and repair (whether or not the portion of the Premises requiring repairs, or the means of repairing the same, are reasonably or readily accessible to Lessee, and whether or not the need for such repairs occurs as a result of Lessee's use, any prior use, the elements or the age of such portion of the Premises), including, but not limited to, all equipment or facilities, such as plumbing, HVAC equipment, electrical, lighting facilities, boilers, pressure vessels, fixtures, interior walls, interior surfaces of exterior walls, ceilings, floors, windows, doors, plate glass, and skylights but excluding any items which are the responsibility of Lessor pursuant to Paragraph 7.2. Lessee, in keeping the Premises in good order, condition and repair, shall exercise and perform good maintenance practices, specifically including the procurement and maintenance of the service contracts required by Paragraph 7.1(b) below. Lessee's obligations shall include restorations, replacements or renewals when necessary to keep the Premises and all improvements thereon or a part thereof in good order, condition and state of repair.

 (b) Service Contracts. Lessee shall, at Lessee's sole expense, procure and maintain contracts, with copies to Lessor, in customary form and substance for,
- (b) Service Contracts. Lessee shall, at Lessee's sole expense, procure and maintain contracts, with copies to Lessor, in customary form and substance for, and with contractors specializing and experienced in the maintenance of the following equipment and improvements, if any, if and when installed on the Premises: (i) HVAC equipment, (ii) boiler and pressure vessels, and (iii) clarifiers. However, Lessor reserves the right, upon notice to Lessee, to procure and maintain any or all of such service contracts, and Lessee shall reimburse Lessor, upon demand, for the cost thereof.
- (c) Failure to Perform. If Lessee fails to perform Lessee's obligations under this Paragraph 7.1, Lessor may enter upon the Premises after 10 days' prior written notice to Lessee (except in the case of an emergency, in which case no notice shall be required), perform such obligations on Lessee's behalf, and put the Premises in good order, condition and repair, and Lessee shall promptly pay to Lessor a sum equal to 115% of the cost thereof.
- (d) **Replacement**. Subject to Lessee's indemnification of Lessor as set forth in Paragraph 8.7 below, and without relieving Lessee of liability resulting from Lessee's failure to exercise and perform good maintenance practices, if an item described in Paragraph 7.1(b) cannot be repaired other than at a cost which is in excess of 50% of the cost of replacing such item, then such item shall be replaced by Lessor, and the cost thereof shall be prorated between the Parties and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lease or any extension thereof, on the date on which Base Rent is due, an amount equal to the product of multiplying the cost of such replacement by a fraction, the numerator of which is one, and the denominator of which is 144 (i.e. 1/144th of the cost per month). Lessee shall pay Interest on the unamortized balance but may prepay its obligation at any time.
- 7.2 **Lessor's Obligations**. Subject to the provisions of Paragraphs 2.2 (Condition), 2.3 (Compliance), 4.2 (Common Area Operating Expenses), 6 (Use), 7.1 (Lessee's Obligations), 9 (Damage or Destruction) and 14 (Condemnation), Lessor, subject to reimbursement pursuant to Paragraph 4.2, shall keep in good order, condition and repair the foundations, exterior walls, structural condition of interior bearing walls, exterior roof, fire sprinkler system, Common Area fire alarm and/or smoke detection systems, fire hydrants, parking lots, walkways, parkways, driveways, landscaping, fences, signs and utility systems serving the Common Areas and all parts thereof, as well as providing the services for which there is a Common Area Operating Expense pursuant to Paragraph 4.2. Lessor shall not be obligated to paint the exterior or interior surfaces of exterior walls nor shall Lessor be obligated to maintain, repair or replace windows, doors or plate glass of the Premises.

7.3 Utility Installations; Trade Fixtures; Alterations.

- (a) **Definitions.** The term "**Utility Installations**" refers to all floor and window coverings, air and/or vacuum lines, power panels, electrical distribution, security and fire protection systems, communication cabling, lighting fixtures, HVAC equipment, plumbing, and fencing in or on the Premises. The term "**Trade Fixtures**" shall mean Lessee's machinery and equipment that can be removed without doing material damage to the Premises. The term "**Alterations**" shall mean any modification of the improvements, other than Utility Installations or Trade Fixtures, whether by addition or deletion. "**Lessee Owned Alterations and/or Utility Installations**" are defined as Alterations and/or Utility Installations made by Lessee that are not yet owned by Lessor pursuant to Paragraph 7.4(a).
- (b) **Consent**. Lessee shall not make any Alterations or Utility Installations to the Premises without Lessor's prior written consent. Lessee may, however, make non-structural Alterations or Utility Installations to the interior of the Premises (excluding the roof) without such consent but upon notice to Lessor, as long as they are not visible from the outside, do not involve puncturing, relocating or removing the roof or any existing walls, will not affect the electrical, plumbing, HVAC, and/or life safety systems, do not trigger the requirement for additional modifications and/or improvements to the Premises resulting from Applicable Requirements,

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

such as compliance with Title 24, and the cumulative cost thereof during this Lease as extended does not exceed a sum equal to 3 month's Base Rent in the aggregate or a sum equal to one month's Base Rent in any one year. Notwithstanding the foregoing, Lessee shall not make or permit any roof penetrations and/or install anything on the roof without the prior written approval of Lessor. Lessor may, as a precondition to granting such approval, require Lessee to utilize a contractor chosen and/or approved by Lessor. Any Alterations or Utility Installations that Lessee shall desire to make and which require the consent of the Lessor shall be presented to Lessor in written form with detailed plans. Consent shall be deemed conditioned upon Lessee's: (i) acquiring all applicable governmental permits, (ii) furnishing Lessor with copies of both the permits and the plans and specifications prior to commencement of the work, and (iii) compliance with all conditions of said permits and other Applicable Requirements in a prompt and expeditious manner. Any Alterations or Utility Installations shall be performed in a workmanlike manner with good and sufficient materials. Lessee shall promptly upon completion furnish Lessor with as-built plans and specifications. For work which costs an amount in excess of one month's Base Rent, Lessor may condition its consent upon Lessee providing a lien and completion bond in an amount equal to 150% of the estimated cost of such Alteration or Utility Installation and/or upon Lessee's posting an additional Security Deposit with Lessor.

(c) Liens; Bonds. Lessee shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lessee at or for use on the Premises, which claims are or may be secured by any mechanic's or materialmen's lien against the Premises or any interest therein. Lessee shall give Lessor not less than 10 days notice prior to the commencement of any work in, on or about the Premises, and Lessor shall have the right to post notices of non-responsibility. If Lessee shall contest the validity of any such lien, claim or demand, then Lessee shall, at its sole expense defend and protect itself, Lessor and the Premises against the same and shall pay and satisfy any such adverse judgment that may be rendered thereon before the enforcement thereof. If Lessor shall require, Lessee shall furnish a surety bond in an amount equal to 150% of the amount of such contested lien, claim or demand, indemnifying Lessor against liability for the same. If Lessor elects to participate in any such action, Lessee shall pay Lessor's attorneys' fees and costs.

7.4 Ownership; Removal; Surrender; and Restoration.

- (a) **Ownership**. Subject to Lessor's right to require removal or elect ownership as hereinafter provided, all Alterations and Utility Installations made by Lessee shall be the property of Lessee, but considered a part of the Premises. Lessor may, at any time, elect in writing to be the owner of all or any specified part of the Lessee Owned Alterations and Utility Installations. Unless otherwise instructed per paragraph 7.4(b) hereof, all Lessee Owned Alterations and Utility Installations shall, at the expiration or termination of this Lease, become the property of Lessor and be surrendered by Lessee with the Premises.
- (b) **Removal**. By delivery to Lessee of written notice from Lessor not earlier than 90 and not later than 30 days prior to the end of the term of this Lease, Lessor may require that any or all Lessee Owned Alterations or Utility Installations be removed by the expiration or termination of this Lease. Lessor may require the removal at any time of all or any part of any Lessee Owned Alterations or Utility Installations made without the required consent.
- (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the improvements, parts and surfaces thereof broom clean and free of debris, and in good operating order, condition and state of repair, ordinary wear and tear excepted. "Ordinary wear and tear" shall not include any damage or deterioration that would have been prevented by good maintenance practice. Notwithstanding the foregoing and the provisions of Paragraph 7.1(a), if the Lessee occupies the Premises for 12 months or less, then Lessee shall surrender the Premises in the same condition as delivered to Lessee on the Start Date with NO allowance for ordinary wear and tear. Lessee shall repair any damage occasioned by the installation, maintenance or removal of Trade Fixtures, Lessee owned Alterations and/or Utility Installations, furnishings, and equipment as well as the removal of any storage tank installed by or for Lessee. Lessee shall also remove from the Premises any and all Hazardous Substances brought onto the Premises by or for Lessee, or any third party (except Hazardous Substances which were deposited via underground migration from areas outside of the Premises) to the level specified in Applicable Requirements. Trade Fixtures shall remain the property of Lessee and shall be removed by Lessee. Any personal property of Lessee not removed on or before the Expiration Date or any earlier termination date shall be deemed to have been abandoned by Lessee and may be disposed of or retained by Lessor as Lessor may desire. The failure by Lessee to timely vacate the Premises pursuant to this Paragraph 7.4(c) without the express written consent of Lessor shall constitute a holdover under the provisions of Paragraph 26 below.

8. Insurance; Indemnity.

8.1 **Payment of Premiums**. The cost of the premiums for the insurance policies required to be carried by Lessor, pursuant to Paragraphs 8.2(b), 8.3(a) and 8.3(b), shall be a Common Area Operating Expense. Premiums for policy periods commencing prior to, or extending beyond, the term of this Lease shall be prorated to coincide with the corresponding Start Date or Expiration Date.

8.2 Liability Insurance.

- (a) Carried by Lessee. Lessee shall obtain and keep in force a Commercial General Liability policy of insurance protecting Lessee and Lessor as an additional insured against claims for bodily injury, personal injury and property damage based upon or arising out of the ownership, use, occupancy or maintenance of the Premises and all areas appurtenant thereto. Such insurance shall be on an occurrence basis providing single limit coverage in an amount not less than \$1,000,000 per occurrence with an annual aggregate of not less than \$2,000,000. Lessee shall add Lessor as an additional insured by means of an endorsement at least as broad as the Insurance Service Organization's "Additional Insured-Managers or Lessors of Premises" Endorsement. The policy shall not contain any intra-insured exclusions as between insured persons or organizations, but shall include coverage for liability assumed under this Lease as an "insured contract" for the performance of Lessee's indemnity obligations under this Lease. The limits of said insurance shall not, however, limit the liability of Lessee nor relieve Lessee of any obligation hereunder. Lessee shall provide an endorsement on its liability policy(ies) which provides that its insurance shall be primary to and not contributory with any similar insurance carried by Lessor, whose insurance shall be considered excess insurance only.
- (b) **Carried by Lessor**. Lessor shall maintain liability insurance as described in Paragraph 8.2(a), in addition to, and not in lieu of, the insurance required to be maintained by Lessee. Lessee shall not be named as an additional insured therein.

8.3 Property Insurance - Building, Improvements and Rental Value.

- (a) **Building and Improvements**. Lessor shall obtain and keep in force a policy or policies of insurance in the name of Lessor, with loss payable to Lessor, any ground-lessor, and to any Lender insuring loss or damage to the Premises. The amount of such insurance shall be equal to the full insurable replacement cost of the Premises, as the same shall exist from time to time, or the amount required by any Lender, but in no event more than the commercially reasonable and available insurable value thereof. Lessee Owned Alterations and Utility Installations, Trade Fixtures, and Lessee's personal property shall be insured by Lessee not by Lessor. If the coverage is available and commercially appropriate, such policy or policies shall insure against all risks of direct physical loss or damage (except the perils of flood and/or earthquake unless required by a Lender), including coverage for debris removal and the enforcement of any Applicable Requirements requiring the upgrading, demolition, reconstruction or replacement of any portion of the Premises as the result of a covered loss. Said policy or policies shall also contain an agreed valuation provision in lieu of any coinsurance clause, waiver of subrogation, and inflation guard protection causing an increase in the annual property insurance coverage amount by a factor of not less than the adjusted U.S. Department of Labor Consumer Price Index for All Urban Consumers for the city nearest to where the Premises are located. If such insurance coverage has a deductible clause, the deductible amount shall not exceed \$5,000 per occurrence.
- (b) Rental Value. Lessor shall also obtain and keep in force a policy or policies in the name of Lessor with loss payable to Lessor and any Lender, insuring the loss of the full Rent for one year with an extended period of indemnity for an additional 180 days ("Rental Value insurance"). Said insurance shall contain an agreed valuation provision in lieu of any coinsurance clause, and the amount of coverage shall be adjusted annually to reflect the projected Rent otherwise payable by Lessee, for the next 12 month period.
- (c) Adjacent Premises. Lessee shall pay for any increase in the premiums for the property insurance of the Building and for the Common Areas or other buildings in the Shopping Center if said increase is caused by Lessee's acts, omissions, use or occupancy of the Premises.
- (d) Lessee's Improvements. Since Lessor is the Insuring Party, Lessor shall not be required to insure Lessee Owned Alterations and Utility Installations unless the item in question has become the property of Lessor under the terms of this Lease.

8.4 Lessee's Property; Business Interruption Insurance; Worker's Compensation Insurance.

- (a) **Property Damage**. Lessee shall obtain and maintain insurance coverage on all of Lessee's personal property, Trade Fixtures, and Lessee Owned Alterations and Utility Installations. Such insurance shall be full replacement cost coverage with a deductible of not to exceed \$1,000 per occurrence. The proceeds from any such insurance shall be used by Lessee for the replacement of personal property, Trade Fixtures and Lessee Owned Alterations and Utility Installations.
- (b) **Business Interruption**. Lessee shall obtain and maintain loss of income and extra expense insurance in amounts as will reimburse Lessee for direct or indirect loss of earnings attributable to all perils commonly insured against by prudent lessees in the business of Lessee or attributable to prevention of access to the Premises as a result of such perils.
- (c) Worker's Compensation Insurance. Lessee shall obtain and maintain Worker's Compensation Insurance in such amount as may be required by Applicable Requirements. Such policy shall include a 'Waiver of Subrogation' endorsement. Lessee shall provide Lessor with a copy of such endorsement along with the certificate of insurance or copy of the policy required by paragraph 8.5.
- (d) **No Representation of Adequate Coverage**. Lessor makes no representation that the limits or forms of coverage of insurance specified herein are adequate to cover Lessee's property, business operations or obligations under this Lease.
- 8.5 **Insurance Policies**. Insurance required herein shall be by companies maintaining during the policy term a "General Policyholders Rating" of at least A-, VII, as set forth in the most current issue of "Best's Insurance Guide", or such other rating as may be required by a Lender. Lessee shall not do or permit to be done anything which invalidates the required insurance policies. Lessee shall, prior to the Start Date, deliver to Lessor certified copies of policies of such insurance or certificates with copies of the required endorsements evidencing the existence and amounts of the required insurance. No such policy shall be cancelable or subject to modification except after 30 days prior written notice to Lessor. Lessee shall, at least 10 days prior to the expiration of such policies, furnish Lessor with evidence of

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

renewals or "insurance binders" evidencing renewal thereof, or Lessor may increase his liability insurance coverage and charge the cost thereof to Lessee, which amount shall be payable by Lessee to Lessor upon demand. Such policies shall be for a term of at least one year, or the length of the remaining term of this Lease, whichever is less. If either Party shall fail to procure and maintain the insurance required to be carried by it, the other Party may, but shall not be required to, procure and maintain the same

- 8.6 Waiver of Subrogation. Without affecting any other rights or remedies, Lessee and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is not limited by the amount of insurance carried or required, or by any deductibles applicable hereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Lessee, as the case may be, so long as the insurance is not invalidated thereby.
- 8.7 Indemnity. Except for Lessor's gross negligence or willful misconduct, Lessee shall indemnify, protect, defend and hold harmless the Premises, Lessor and its agents, Lessor's master or ground lessor, partners and Lenders, from and against any and all claims, loss of rents and/or damages, liens, judgments, penalties, attorneys' and consultants' fees, expenses and/or liabilities arising out of, involving, or in connection with, a Breach of the Lease by Lessee and/or the use and/or occupancy of the Premises and/or Project by Lessee and/or by Lessee's employees, contractors or invitees. If any action or proceeding is brought against Lessor by reason of any of the foregoing matters, Lessee shall upon notice defend the same at Lessee's expense by counsel reasonably satisfactory to Lessor and Lessor shall cooperate with Lessee in such defense. Lessor need not have first paid any such claim in order to be defended or indemnified.
- 8.8 **Exemption of Lessor and its Agents from Liability**. Notwithstanding the negligence or breach of this Lease by Lessor or its agents, neither Lessor nor its agents shall be liable under any circumstances for: (i) injury or damage to the person or goods, wares, merchandise or other property of Lessee, Lessee's employees, contractors, invitees, customers, or any other person in or about the Premises, whether such damage or injury is caused by or results from fire, steam, electricity, gas, water or rain, indoor air quality, the presence of mold or from the breakage, leakage, obstruction or other defects of pipes, fire sprinklers, wires, appliances, plumbing, HVAC or lighting fixtures, or from any other cause, whether the said injury or damage results from conditions arising upon the Premises or upon other portions of the building of which the Premises are a part, or from other sources or places, (ii) any damages arising from any act or neglect of any other tenant of Lessor or from the failure of Lessor or its agents to enforce the provisions of any other lease in the Shopping Center, or (iii) injury to Lessee's business or for any loss of income or profit therefrom. Instead, it is intended that Lessee's sole recourse in the event of such damages or injury be to file a claim on the insurance policy(ies) that Lessee is required to maintain pursuant to the provisions of paragraph 8.
- 8.9 **Failure to Provide Insurance**. Lessee acknowledges that any failure on its part to obtain or maintain the insurance required herein will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, for any month or portion thereof that Lessee does not maintain the required insurance and/or does not provide Lessor with the required binders or certificates evidencing the existence of the required insurance, the Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater. The parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to maintain the required insurance. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to maintain such insurance, prevent the exercise of any of the other rights and remedies granted hereunder, nor relieve Lessee of its obligation to maintain the insurance specified in this Lease.

Damage or Destruction.

9.1 Definitions.

- (a) "Premises Partial Damage" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations, which can reasonably be repaired in 3 months or less from the date of the damage or destruction, and the cost thereof does not exceed a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (b) "Premises Total Destruction" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which cannot reasonably be repaired in 3 months or less from the date of the damage or destruction and/or the cost thereof exceeds a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.
- (c) "Insured Loss" shall mean damage or destruction to improvements on the Premises, other than Lessee Owned Alterations and Utility Installations and Trade Fixtures, which was caused by an event required to be covered by the insurance described in Paragraph 8.3(a), irrespective of any deductible amounts or coverage limits involved.
- (d) "Replacement Cost" shall mean the cost to repair or rebuild the improvements owned by Lessor at the time of the occurrence to their condition existing immediately prior thereto, including demolition, debris removal and upgrading required by the operation of Applicable Requirements, and without deduction for depreciation.
- (e) "Hazardous Substance Condition" shall mean the occurrence or discovery of a condition involving the presence of, or a contamination by, a Hazardous Substance, in, on, or under the Premises or Common Areas which requires restoration.
- 9.2 Partial Damage Insured Loss. If a Premises Partial Damage that is an Insured Loss occurs, then Lessor shall, at Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonably possible and this Lease shall continue in full force and effect; provided, however, that Lessee shall, at Lessor's election, make the repair of any damage or destruction the total Replacement Cost of which is \$10,000 or less, and, in such event, Lessor shall make any applicable insurance proceeds available to Lessee on a reasonable basis for that purpose. Notwithstanding the foregoing, if the required insurance was not in force or the insurance proceeds are not sufficient to effect such repair, the Insuring Party shall promptly contribute the shortage in proceeds as and when required to complete said repairs. In the event, however, such shortage was due to the fact that, by reason of the unique nature of the improvements, full Replacement Cost insurance coverage was not commercially reasonable and available, Lessor shall have no obligation to pay for the shortage in insurance proceeds or to fully restore the unique aspects of the Premises unless Lessee provides Lessor with the funds to cover same, or adequate assurance thereof, within 10 days following receipt of written notice of such shortage and request therefor. If Lessor receives said funds or adequate assurance thereof within said 10 day period, the party responsible for making the repairs shall complete them as soon as reasonably possible and this Lease shall remain in full force and effect. If such funds or assurance are not received, Lessor may nevertheless elect by written notice to Lessee within 10 days thereafter to: (i) make such restoration and repair as is commercially reasonable with Lessor paying any shortage in proceeds, in which case this Lease shall remain in full force and effect, or (ii) have this Lease terminate 30 days thereafter. Lessee shall pot be entitled to reimbursement o
- 9.3 **Partial Damage Uninsured Loss**. If a Premises Partial Damage that is not an Insured Loss occurs, unless caused by a negligent or willful act of Lessee (in which event Lessee shall make the repairs at Lessee's expense), Lessor may either: (i) repair such damage as soon as reasonably possible at Lessor's expense (subject to reimbursement pursuant to Paragraph 4.2), in which event this Lease shall continue in full force and effect, or (ii) terminate this Lease by giving written notice to Lessee within 30 days after receipt by Lessor of knowledge of the occurrence of such damage. Such termination shall be effective 60 days following the date of such notice. In the event Lessor elects to terminate this Lease, Lessee shall have the right within 10 days after receipt of the termination notice to give written notice to Lesser of Lessee's commitment to pay for the repair of such damage without reimbursement from Lessor. Lessee shall provide Lessor with said funds or satisfactory assurance thereof within 30 days after making such commitment. In such event this Lease shall continue in full force and effect, and Lessor shall proceed to make such repairs as soon as reasonably possible after the required funds are available. If Lessee does not make the required commitment, this Lease shall terminate as of the date specified in the termination notice.
- 9.4 **Total Destruction**. Notwithstanding any other provision hereof, if a Premises Total Destruction occurs, this Lease shall terminate 60 days following such Destruction. If the damage or destruction was caused by the gross negligence or willful misconduct of Lessee, Lessor shall have the right to recover Lessor's damages from Lessee, except as provided in Paragraph 8.6.
- 9.5 Damage Near End of Term. If at any time during the last 6 months of this Lease there is damage for which the cost to repair exceeds one month's Base Rent, whether or not an Insured Loss, Lessor may terminate this Lease effective 60 days following the date of occurrence of such damage by giving a written termination notice to Lessee within 30 days after the date of occurrence of such damage. Notwithstanding the foregoing, if Lessee at that time has an exercisable option to extend this Lease or to purchase the Premises, then Lessee may preserve this Lease by (a) exercising such option and (b) providing Lessor with any shortage in insurance proceeds (or adequate assurance thereof) needed to make the repairs on or before the earlier of (i) the date which is 10 days after Lessee's receipt of Lessor's written notice purporting to terminate this Lease, or (ii) the day prior to the date upon which such option expires. If Lessee duly exercises such option during such period and provides Lessor with funds (or adequate assurance thereof) to cover any shortage in insurance proceeds, Lessor shall, at Lessor's commercially reasonable expense, repair such damage as soon as reasonably possible and this Lease shall continue in full force and effect. If Lessee fails to exercise such option and provides such funds or assurance during such period, then this Lease shall terminate on the date specified in the termination notice and Lessee's option shall be extinguished.
- 9.6 **Damage to Shopping Center.** In the event of any damage or destruction to other portions of the Building or to any other buildings in the Shopping Center, whether insured or uninsured (and whether or not there is also damage or destruction to the Premises), which cannot reasonably be repaired in 6 months or less from the date of the damage or destruction, Lessor may either (i) repair such damage or destruction as soon as reasonably possible without expense to Lessee, in

which event this Lease shall continue in full force and effect, or (ii) terminate this Lease by giving written notice to Lessee within 30 days after receipt by Lessor of knowledge of the occurrence of such damage or destruction. Such termination shall be effective 60 days following the date of such notice.

9.7 Abatement of Rent; Lessee's Remedies.

- (a) Abatement. In the event of Premises Partial Damage or Premises Total Destruction or a Hazardous Substance Condition for which Lessee is not responsible under this Lease, the Base Rent payable by Lessee for the period required for the repair, remediation or restoration of such damage shall be abated in proportion to the degree to which Lessee's use of the Premises is impaired, but not to exceed the proceeds received from the Rental Value Insurance. All other obligations of Lessee hereunder shall be performed by Lessee, and Lessor shall have no liability for any such damage, destruction, remediation, repair or restoration except as provided herein.
- (b) Remedies. If Lessor shall be obligated to repair or restore the Premises and does not commence, in a substantial and meaningful way, such repair or restoration within 90 days after such obligation shall accrue, Lessee may, at any time prior to the commencement of such repair or restoration, give written notice to Lessor and to any Lenders of which Lessee has actual notice, of Lessee's election to terminate this Lease on a date not less than 60 days following the giving of such notice. If Lessee gives such notice and such repair or restoration is not commenced within 30 days thereafter, this Lease shall terminate as of the date specified in said notice. If the repair or restoration is commenced within such 30 days, this Lease shall continue in full force and effect. "Commence" shall mean either the unconditional authorization of the preparation of the required plans, or the beginning of the actual work on the Premises, whichever first occurs.
- 9.8 **Termination; Advance Payments.** Upon termination of this Lease pursuant to Paragraph 6.2(g) or Paragraph 9, an equitable adjustment shall be made concerning advance Base Rent and any other advance payments made by Lessee to Lessor. Lessor shall, in addition, return to Lessee so much of Lessee's Security Deposit as has not been, or is not then required to be, used by Lessor.

10. Real Property Taxes

- 10.1 **Definition**. As used herein, the term "Real Property Taxes" shall include any form of assessment; real estate, general, special, ordinary or extraordinary, or rental levy or tax (other than inheritance, personal income or estate taxes); improvement bond; and/or license fee imposed upon or levied against any legal or equitable interest of Lessor in the Shopping Center, Lessor's right to other income therefrom, and/or Lessor's business of leasing, by any authority having the direct or indirect power to tax and where the funds are generated with reference to the Shopping Center address. The term "Real Property Taxes" shall also include any tax, fee, levy, assessment or charge, or any increase therein: (i) imposed by reason of events occurring during the term of this Lease, including but not limited to, a change in the ownership of the Shopping Center, (ii) a change in the improvements thereon, and/or (iii) levied or assessed on machinery or equipment provided by Lessor to Lessee pursuant to this Lease. In calculating Real Property Taxes for any calendar year, the Real Property Taxes for any real estate tax year shall be included in the calculation of Real Property Taxes for such calendar year based upon the number of days which such calendar year and tax year have in common.
- 10.2 **Payment of Taxes**. Except as otherwise provided in Paragraph 10.3, Lessor shall pay the Real Property Taxes applicable to the Shopping Center, and said payments shall be included in the calculation of Common Area Operating Expenses in accordance with the provisions of Paragraph 4.2.
- 10.3 Additional Improvements. Common Area Operating Expenses shall not include Real Property Taxes specified in the tax assessor's records and work sheets as being caused by additional improvements placed upon the Shopping Center by other lessees or by Lessor for the exclusive enjoyment of such other lessees. Notwithstanding Paragraph 10.2 hereof, Lessee shall, however, pay to Lessor at the time Common Area Operating Expenses are payable under Paragraph 4.2, the entirety of any increase in Real Property Taxes if assessed solely by reason of Alterations, Trade Fixtures or Utility Installations placed upon the Premises by Lessee or at Lessee's request, or by reason of any alterations or improvements to the Premises made by Lessor subsequent to the execution of this Lease by the Parties.
- 10.4 **Joint Assessment**. If the Building is not separately assessed, Real Property Taxes allocated to the Building shall be an equitable proportion of the Real Property Taxes for all of the land and improvements included within the tax parcel assessed, such proportion to be determined by Lessor from the respective valuations assigned in the assessor's work sheets or such other information as may be reasonably available. Lessor's reasonable determination thereof, in good faith, shall be conclusive
- 10.5 **Personal Property Taxes**. Lessee shall pay prior to delinquency all taxes assessed against and levied upon Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all personal property of Lessee contained in the Premises. When possible, Lessee shall cause its Lessee Owned Alterations and Utility Installations, Trade Fixtures, furnishings, equipment and all other personal property to be assessed and billed separately from the real property of Lessor. If any of Lessee's said property shall be assessed with Lessor's real property, Lessee shall pay Lessor the taxes attributable to Lessee's property within 10 days after receipt of a written statement setting forth the taxes applicable to Lessee's property.

11. Utilities and Services.

- 11.1 Lessee shall pay for all water, gas, heat, light, power, telephone, trash disposal and other utilities and services supplied to the Premises, together with any taxes thereon. To the extent any such utilities and/or services are not separately metered, Lessee shall pay Lessee's Share thereof in accordance with Paragraph 4.2. Notwithstanding the provisions of Paragraph 4.2, if at any time in Lessor's sole judgment, Lessor determines that Lessee is using a disproportionate amount of water, electricity or other commonly metered utilities, or that Lessee is generating such a large volume of trash as to require an increase in the size of the trash receptacle and/or an increase in the number of times per month that it is emptied, then Lessor may increase Lessee's Base Rent by an amount equal to such increased costs. There shall be no abatement of Rent and Lessor shall not be liable in any respect whatsoever for the inadequacy, stoppage, interruption or discontinuance of any utility or service due to riot, strike, labor dispute, breakdown, accident, repair or other cause beyond Lessor's reasonable control or in cooperation with governmental request or directions.
- 11.2 Within fifteen days of Lessor's written request, Lessee agrees to deliver to Lessor such information, documents and/or authorization as Lessor needs in order for Lessor to comply with new or existing Applicable Requirements relating to commercial building energy usage, ratings, and/or the reporting thereof.

12. Assignment and Subletting.

12.1 Lessor's Consent Required.

- (a) Lessee shall not voluntarily or by operation of law assign, transfer, mortgage or encumber (collectively, "assign or assignment") or sublet all or any part of Lessee's interest in this Lease or in the Premises without Lessor's prior written consent.
- (b) Unless Lessee is a corporation and its stock is publicly traded on a national stock exchange, a change in the control of Lessee shall constitute an assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of the voting control of Lessee shall constitute a change in control for this purpose.
- (c) The involvement of Lessee or its assets in any transaction, or series of transactions (by way of merger, sale, acquisition, financing, transfer, leveraged buyout or otherwise), whether or not a formal assignment or hypothecation of this Lease or Lessee's assets occurs, which results or will result in a reduction of the Net Worth of Lessee by an amount greater than 25% of such Net Worth as it was represented at the time of the execution of this Lease or at the time of the most recent assignment to which Lessor has consented, or as it exists immediately prior to said transaction or transactions constituting such reduction, whichever was or is greater, shall be considered an assignment of this Lease to which Lessor may withhold its consent. "Net Worth of Lessee" shall mean the net worth of Lessee (excluding any guarantors) established under generally accepted accounting principles.
- (d) An assignment or subletting without consent shall, at Lessor's option, be a Default curable after notice per Paragraph 13.1(d), or a noncurable Breach without the necessity of any notice and grace period. If Lessor elects to treat such unapproved assignment or subletting as a noncurable Breach, Lessor may either: (i) terminate this Lease, or (ii) upon 30 days written notice, increase the monthly Base Rent and Percentage Rent Rate to 110% of the Base Rent and Percentage Rent Rate then in effect. Further, in the event of such Breach and rental adjustment, (i) the purchase price of any option to purchase the Premises held by Lessee shall be subject to similar adjustment to 110% of the price previously in effect, and (ii) all fixed and non-fixed rental adjustments scheduled during the remainder of the Lease term shall be increased to 110% of the scheduled adjusted rent.
 - (e) Lessee's remedy for any breach of Paragraph 12.1 by Lessor shall be limited to compensatory damages and/or injunctive relief.
 - (f) Lessor may reasonably withhold consent to a proposed assignment or subletting if Lessee is in Default at the time consent is requested.
- (g) Notwithstanding the foregoing, allowing a de minimis portion of the Premises, ie. 20 square feet or less, to be used by a third party vendor in connection with the installation of a vending machine or payphone shall not constitute a subletting.

12.2 Terms and Conditions Applicable to Assignment and Subletting.

- (a) Regardless of Lessor's consent, no assignment or subletting shall: (i) be effective without the express written assumption by such assignee or sublessee of the obligations of Lessee under this Lease, (ii) release Lessee of any obligations hereunder, or (iii) alter the primary liability of Lessee for the payment of Rent or for the performance of any other obligations to be performed by Lessee.
- (b) Lessor may accept Rent or performance of Lessee's obligations from any person other than Lessee pending approval or disapproval of an assignment. Neither a delay in the approval or disapproval of such assignment nor the acceptance of Rent or performance shall constitute a waiver or estoppel of Lessor's right to exercise its remedies for Lessee's Default or Breach.
 - (c) Lessor's consent to any assignment or subletting shall not constitute a consent to any subsequent assignment or subletting.
- (d) In the event of any Default or Breach by Lessee, Lessor may proceed directly against Lessee, any Guarantors or anyone else responsible for the performance of Lessee's obligations under this Lease, including any assignee or sublessee, without first exhausting Lessor's remedies against any other person or entity responsible therefor to Lessor, or any security held by Lessor.
- (e) Each request for consent to an assignment or subletting shall be in writing, accompanied by information relevant to Lessor's determination as to the financial and operational responsibility and appropriateness of the proposed assignee or sublessee, including but not limited to the intended use and/or required

INITIALS

modification of the Premises, if any, together with a fee of \$500 as consideration for Lessor's considering and processing said request. Lessee agrees to provide Lessor with such other or additional information and/or documentation as may be reasonably requested. (See also Paragraph 36)

- (f) Any assignee of, or sublessee under, this Lease shall, by reason of accepting such assignment, entering into such sublease, or entering into possession of the Premises or any portion thereof, be deemed to have assumed and agreed to conform and comply with each and every term, covenant, condition and obligation herein to be observed or performed by Lessee during the term of said assignment or sublease, other than such obligations as are contrary to or inconsistent with provisions of an assignment or sublease to which Lessor has specifically consented to in writing.
- (g) Lessor's consent to any assignment or subletting shall not transfer to the assignee or sublessee any Option granted to the original Lessee by this Lease unless such transfer is specifically consented to by Lessor in writing. (See Paragraph 39.2)
- 12.3 Additional Terms and Conditions Applicable to Subletting. The following terms and conditions shall apply to any subletting by Lessee of all or any part of the Premises and shall be deemed included in all subleases under this Lease whether or not expressly incorporated therein:
- (a) Lessee hereby assigns and transfers to Lessor all of Lessee's interest in all Rent payable on any sublease, and Lessor may collect such Rent and apply same toward Lessee's obligations under this Lease; provided, however, that until a Breach shall occur in the performance of Lessee's obligations, Lessee may collect said Rent. In the event that the amount collected by Lessor exceeds Lessee's then outstanding obligations any such excess shall be refunded to Lessee. Lessor shall not, by reason of the foregoing or any assignment of such sublease, nor by reason of the collection of Rent, be deemed liable to the sublessee for any failure of Lessee to perform and comply with any of Lessee's obligations to such sublessee. Lessee hereby irrevocably authorizes and directs any such sublessee, upon receipt of a written notice from Lessor stating that a Breach exists in the performance of Lessee's obligations under this Lease, to pay to Lessor all Rent due and to become due under the sublease. Sublessee shall rely upon any such notice from Lessor and shall pay all Rents to Lessor without any obligation or right to inquire as to whether such Breach exists, notwithstanding any claim from Lessee to the contrary.
- (b) In the event of a Breach by Lessee, Lessor may, at its option, require sublessee to attorn to Lessor, in which event Lessor shall undertake the obligations of the sublessor under such sublease from the time of the exercise of said option to the expiration of such sublease; provided, however, Lessor shall not be liable for any prepaid rents or security deposit paid by such sublessee to such sublessor or for any prior Defaults or Breaches of such sublessor.
 - (c) Any matter requiring the consent of the sublessor under a sublease shall also require the consent of Lessor.
 - (d) No sublessee shall further assign or sublet all or any part of the Premises without Lessor's prior written consent.
- (e) Lessor shall deliver a copy of any notice of Default or Breach by Lessee to the sublessee, who shall have the right to cure the Default of Lessee within the grace period, if any, specified in such notice. The sublessee shall have a right of reimbursement and offset from and against Lessee for any such Defaults cured by the sublessee

13. Default; Breach; Remedies.

- 13.1 **Default; Breach**. A "**Default**" is defined as a failure by the Lessee to comply with or perform any of the terms, covenants, conditions or Rules and Regulations under this Lease. A "**Breach**" is defined as the occurrence of one or more of the following Defaults, and the failure of Lessee to cure such Default within any applicable grace period:
- (a) The vacating or abandonment of the Premises. Lessee shall be deemed to have vacated the Premises if Lessee ceases to continuously operate its business in the Premises for a period of 5 consecutive days.
- (b) The failure of Lessee to make any payment of Rent or any Security Deposit required to be made by Lessee hereunder, whether to Lessor or to a third party, when due, to provide reasonable evidence of insurance or surety bond, or to fulfill any obligation under this Lease which endangers or threatens life or property, where such failure continues for a period of 3 business days following written notice to Lessee. THE ACCEPTANCE BY LESSOR OF A PARTIAL PAYMENT OF RENT OR SECURITY DEPOSIT SHALL NOT CONSTITUTE A WAIVER OF ANY OF LESSOR'S RIGHTS, INCLUDING LESSOR'S RIGHT TO RECOVER POSSESSION OF THE PREMISES.
- (c) The failure of Lessee to allow Lessor and/or its agents access to the Premises or the commission of waste, act or acts constituting public or private nuisance, and/or an illegal activity on the Premises by Lessee, where such actions continue for a period of 3 business days following written notice to Lessee. In the event that Lessee commits waste, a nuisance or an illegal activity a second time then, the Lessor may elect to treat such conduct as a non-curable Breach rather than a Default.
- (d) The failure by Lessee to provide (i) reasonable written evidence of compliance with Applicable Requirements, (ii) the service contracts, (iii) the rescission of an unauthorized assignment or subletting, (iv) an Estoppel Certificate or financial statements, (v) a requested subordination, (vi) evidence concerning any guaranty and/or Guarantor, (vii) any document requested under Paragraph 41, (viii) material safety data sheets (MSDS), or (ix) any other documentation or information which Lessor may reasonably require of Lessee under the terms of this Lease, where any such failure continues for a period of 10 days following written notice to Lessee.
- (e) A Default by Lessee as to the terms, covenants, conditions or provisions of this Lease, or of the rules adopted under Paragraph 2.9 hereof, other than those described in subparagraphs 13.1(a), (b), (c) or (d), above, where such Default continues for a period of 30 days after written notice; provided, however, that if the nature of Lessee's Default is such that more than 30 days are reasonably required for its cure, then it shall not be deemed to be a Breach if Lessee commences such cure within said 30 day period and thereafter diligently prosecutes such cure to completion.

 (f) The occurrence of any of the following events: (i) the making of any general arrangement or assignment for the benefit of creditors; (ii) becoming a
- (f) The occurrence of any of the following events: (i) the making of any general arrangement or assignment for the benefit of creditors; (ii) becoming a "debtor" as defined in 11 U.S.C. § 101 or any successor statute thereto (unless, in the case of a petition filed against Lessee, the same is dismissed within 60 days); (iii) the appointment of a trustee or receiver to take possession of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where possession is not restored to Lessee within 30 days; or (iv) the attachment, execution or other judicial seizure of substantially all of Lessee's assets located at the Premises or of Lessee's interest in this Lease, where such seizure is not discharged within 30 days; provided, however, in the event that any provision of this subparagraph is contrary to any applicable law, such provision shall be of no force or effect, and not affect the validity of the remaining provisions.
 - (g) The discovery that any financial statement of Lessee or of any Guarantor given to Lessor was materially false.
- (h) If the performance of Lessee's obligations under this Lease is guaranteed: (i) the death of a Guarantor, (ii) the termination of a Guarantor's liability with respect to this Lease other than in accordance with the terms of such guaranty, (iii) a Guarantor's becoming insolvent or the subject of a bankruptcy filing, (iv) a Guarantor's refusal to honor the guaranty, or (v) a Guarantor's breach of its guaranty obligation on an anticipatory basis, and Lessee's failure, within 60 days following written notice of any such event, to provide written alternative assurance or security, which, when coupled with the then existing resources of Lessee, equals or exceeds the combined financial resources of Lessee and the Guarantors that existed at the time of execution of this Lease.
- 13.2 **Remedies**. If Lessee fails to perform any of its affirmative duties or obligations, within 10 days after written notice (or in case of an emergency, without notice), Lessor may, at its option, perform such duty or obligation on Lessee's behalf, including but not limited to the obtaining of reasonably required bonds, insurance policies, or governmental licenses, permits or approvals. Lessee shall pay to Lessor an amount equal to 115% of the costs and expenses incurred by Lessor in such performance upon receipt of an invoice therefor. In the event of a Breach, Lessor may, with or without further notice or demand, and without limiting Lessor in the exercise of any right or remedy which Lessor may have by reason of such Breach:
- (a) Terminate Lessee's right to possession of the Premises by any lawful means, in which case this Lease shall terminate and Lessee shall immediately surrender possession to Lessor. In such event Lessor shall be entitled to recover from Lessee: (i) the unpaid Rent which had been earned at the time of termination; (ii) the worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that the Lessee proves could have been reasonably avoided; (iii) the worth at the time of award of the amount by which the unpaid rent for the balance of the term after the time of award exceeds the amount of such rental loss that the Lessee proves could be reasonably avoided; and (iv) any other amount necessary to compensate Lessor for all the detriment proximately caused by the Lessee's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom, including but not limited to the cost of recovering possession of the Premises, expenses of reletting, including necessary renovation and alteration of the Premises, reasonable attorneys' fees, and that portion of any leasing commission paid by Lessor in connection with this lease applicable to the unexpired term of this Lease. The worth at the time of award of the amount referred to in provision (iii) of the immediately preceding sentence shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of the District within which the Premises are located at the time of award plus one percent. Efforts by Lessor to mitigate damages caused by Lessee's Breach of this Lease shall not waive Lessor's right to recover any damages to which Lessor is otherwise entitled. If termination of this Lease is obtained through the provisional remedy of unlawful detainer, Lessor shall have the right to recover in such proceeding any unpaid Rent and damages as are recoverable therein, or Lessor may reserve the right to recover all or any part thereof in a separate suit. If a notice and grace period required under Paragraph 13.1 was not previously given, a notice to pay rent or quit, or to perform or quit given to Lessee under the unlawful detainer statute shall also constitute the notice required by Paragraph 13.1. In such case, the applicable grace period required by Paragraph 13.1 and the unlawful detainer statute shall run concurrently, and the failure of Lessee to cure the Default within the greater of the two such grace periods shall constitute both an unlawful detainer and a Breach of this Lease entitling Lessor to the remedies provided for in this Lease and/or by said statute.
- (b) Continue the Lease and Lessee's right to possession and recover the Rent as it becomes due, in which event Lessee may sublet or assign, subject only to reasonable limitations. Acts of maintenance, efforts to relet, and/or the appointment of a receiver to protect the Lessor's interests, shall not constitute a termination of the Lessee's right to possession.
- (c) Pursue any other remedy now or hereafter available under the laws or judicial decisions of the state wherein the Premises are located. The expiration or termination of this Lease and/or the termination of Lessee's right to possession shall not relieve Lessee from liability under any indemnity provisions of this Lease

INITIALS

as to matters occurring or accruing during the term hereof or by reason of Lessee's occupancy of the Premises.

- 13.3 Inducement Recapture. Any agreement for free or abated rent or other charges, the cost of tenant improvements for Lessee paid for or performed by Lessor, or for the giving or paying by Lessor to or for Lessee of any cash or other bonus, inducement or consideration for Lessee's entering into this Lease, all of which concessions are hereinafter referred to as "Inducement Provisions," shall be deemed conditioned upon Lessee's full and faithful performance of all of the terms, covenants and conditions of this Lease. Upon Breach of this Lease by Lessee, any such Inducement Provision shall automatically be deemed deleted from this Lease and of no further force or effect, and any rent, other charge, bonus, inducement or consideration theretofore abated, given or paid by Lessor under such an Inducement Provision shall be immediately due and payable by Lessee to Lessor, notwithstanding any subsequent cure of said Breach by Lessee. The acceptance by Lessor of rent or the cure of the Breach which initiated the operation of this paragraph shall not be deemed a waiver by Lessor of the provisions of this paragraph unless specifically so stated in writing by Lessor at the time of such acceptance.
- 13.4 Late Charges. Lessee hereby acknowledges that late payment by Lessee of Rent will cause Lessor to incur costs not contemplated by this Lease, the exact amount of which will be extremely difficult to ascertain. Such costs include, but are not limited to, processing and accounting charges, and late charges which may be imposed upon Lessor by any Lender. Accordingly, if any Rent shall not be received by Lessor within 5 days after such amount shall be due, then, without any requirement for notice to Lessee, Lessee shall immediately pay to Lessor a one-time late charge equal to 10% of each such overdue amount or \$100, whichever is greater. The parties hereby agree that such late charge represents a fair and reasonable estimate of the costs Lessor will incur by reason of such late payment. Acceptance of such late charge by Lessor shall in no event constitute a waiver of Lessee's Default or Breach with respect to such overdue amount, nor prevent the exercise of any of the other rights and remedies granted hereunder. In the event that a late charge is payable hereunder, whether or not collected, for 3 consecutive installments of Base Rent, then notwithstanding any provision of this Lease to the contrary, Base Rent shall, at Lessor's option, become due and payable quarterly in advance
- 13.5 Interest. Any monetary payment due Lessor hereunder, other than late charges, not received by Lessor, when due shall bear interest from the 31st day after it was due. The interest ("Interest") charged shall be computed at the rate of 10% per annum but shall not exceed the maximum rate allowed by law. Interest is payable in addition to the potential late charge provided for in Paragraph 13.4.

13.6 Breach by Lessor.

- (a) Notice of Breach. Lessor shall not be deemed in breach of this Lease unless Lessor fails within a reasonable time to perform an obligation required to be performed by Lessor. For purposes of this Paragraph, a reasonable time shall in no event be less than 30 days after receipt by Lessor, and any Lender whose name and address shall have been furnished to Lessee in writing for such purpose, of written notice specifying wherein such obligation of Lessor has not been performed; provided, however, that if the nature of Lessor's obligation is such that more than 30 days are reasonably required for its performance, then Lessor shall not be in breach if performance is commenced within such 30 day period and thereafter diligently pursued to completion.
- (b) **Performance by Lessee on Behalf of Lessor.** In the event that neither Lessor nor Lender cures said breach within 30 days after receipt of said notice, or if having commenced said cure they do not diligently pursue it to completion, then Lessee may elect to cure said breach at Lessee's expense and offset from Rent the actual and reasonable cost to perform such cure, provided, however, that such offset shall not exceed an amount equal to the greater of one month's Base Rent or the Security Deposit, reserving Lessee's right to reimbursement from Lessor for any such expense in excess of such offset. Lessee shall document the cost of said cure and supply said documentation to Lessor.
- 14. Condemnation. If the Premises or any portion thereof are taken under the power of eminent domain or sold under the threat of the exercise of said power (collectively "Condemnation"), this Lease shall terminate as to the part taken as of the date the condemning authority takes title or possession, whichever first occurs. If more than 10% of the floor area of the Premises, or more than 25% of the parking spaces situated within the parking area, is taken by Condemnation, Lessee may, at Lessee's option, to be exercised in writing within 10 days after Lessor shall have given Lessee written notice of such taking (or in the absence of such notice, within 10 days after the condemning authority shall have taken possession) terminate this Lease as of the date the condemning authority takes such possession. If Lessee does not terminate this Lease in accordance with the foregoing, this Lease shall remain in full force and effect as to the portion of the Premises remaining, except that the Base Rent shall be reduced in proportion to the reduction in utility of the Premises caused by such Condemnation. Condemnation awards and/or payments shall be the property of Lessor, whether such award shall be made as compensation for diminution in value of the leasehold, the value of the part taken, or for severance damages; provided, however, that Lessee shall be entitled to any compensation paid by the condemnor for Lessee's relocation expenses, loss of business goodwill and/or Trade Fixtures, without regard to whether or not this Lease is terminated pursuant to the provisions of this Paragraph. All Alterations and Utility Installations made to the Premises by Lessee, for purposes of Condemnation only, shall be considered the property of the Lessee and Lessee shall be entitled to any and all compensation which is payable therefor. In the event that this Lease is not terminated by reason of the Condemnation, Lessor shall repair any damage to the Premises caused by such Condemnation.

15. Brokerage Fees.

- 15.1 Additional Commission. In addition to the payments owed pursuant to Paragraph 1.13 above, Lessor agrees that: (a) if Lessee exercises any Option, (b) if Lessee or anyone affiliated with Lessee acquires from Lessor any rights to the Premises or other premises owned by Lessor and located within the Shopping Center, (c) if Lessee remains in possession of the Premises, with the consent of Lessor, after the expiration of this Lease, or (d) if Base Rent is increased, whether by agreement or operation of an escalation clause herein, then, Lessor shall pay Brokers a fee in accordance with the fee schedule of the Brokers in effect at the time the Lease was executed. The provisions of this paragraph are intended to supersede the provisions of any earlier agreement to the contrary.
- 15.2 **Assumption of Obligations**. Any buyer or transferee of Lessor's interest in this Lease shall be deemed to have assumed Lessor's obligation hereunder. Brokers shall be third party beneficiaries of the provisions of Paragraphs 1.13, 15, 22 and 31. If Lessor fails to pay to Brokers any amounts due as and for brokerage fees pertaining to this Lease when due, then such amounts shall accrue Interest. In addition, if Lessor fails to pay any amounts to Lessee's Broker when due, Lessee's Broker may send written notice to Lessor and Lessoe of such failure and if Lessor fails to pay such amounts within 10 days after said notice, Lessee shall pay said monies to its Broker and offset such amounts against Rent. In addition, Lessee's Broker shall be deemed to be a third party beneficiary of any commission agreement entered into by and/or between Lessor and Lessor's Broker for the limited purpose of collecting any brokerage fee owed.
- 15.3 Representations and Indemnities of Broker Relationships. Lessee and Lessor each represent and warrant to the other that it has had no dealings with any person, firm, broker, agent or finder (other than the Brokers and Agents, if any) in connection with this Lease, and that no one other than said named Brokers and Agents is entitled to any commission or finder's fee in connection herewith. Lessee and Lessor do each hereby agree to indemnify, protect, defend and hold the other harmless from and against liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any dealings or actions of the indemnifying Party, including any costs, expenses, attorneys' fees reasonably incurred with respect thereto.

16. Estoppel Certificates.

- (a) Each Party (as "Responding Party") shall within 10 days after written notice from the other Party (the "Requesting Party") execute, acknowledge and deliver to the Requesting Party a statement in writing in form similar to the then most current "Estoppel Certificate" form published by AIR CRE, plus such additional information, confirmation and/or statements as may be reasonably requested by the Requesting Party.
- (b) If the Responding Party shall fail to execute or deliver the Estoppel Certificate within such 10 day period, the Requesting Party may execute an Estoppel Certificate stating that: (i) the Lease is in full force and effect without modification except as may be represented by the Requesting Party, (ii) there are no uncured defaults in the Requesting Party's performance, and (iii) if Lessor is the Requesting Party, not more than one month's rent has been paid in advance. Prospective purchasers and encumbrancers may rely upon the Requesting Party's Estoppel Certificate, and the Responding Party shall be estopped from denying the truth of the facts contained in said Certificate. In addition, Lessee acknowledges that any failure on its part to provide such an Estoppel Certificate will expose Lessor to risks and potentially cause Lessor to incur costs not contemplated by this Lease, the extent of which will be extremely difficult to ascertain. Accordingly, should the Lessee fail to execute and/or deliver a requested Estoppel Certificate in a timely fashion the monthly Base Rent shall be automatically increased, without any requirement for notice to Lessee, by an amount equal to 10% of the then existing Base Rent or \$100, whichever is greater for remainder of the Lease. The Parties agree that such increase in Base Rent represents fair and reasonable compensation for the additional risk/costs that Lessor will incur by reason of Lessee's failure to provide the Estoppel Certificate. Such increase in Base Rent shall in no event constitute a waiver of Lessee's Default or Breach with respect to the failure to provide the Estoppel Certificate nor prevent the exercise of any of the other rights and remedies granted hereunder.
- (c) If Lessor desires to finance, refinance, or sell the Premises, or any part thereof, Lessee and all Guarantors shall within 10 days after written notice from Lessor deliver to any potential lender or purchaser designated by Lessor such financial statements as may be reasonably required by such lender or purchaser, including but not limited to Lessee's financial statements for the past 3 years. All such financial statements shall be received by Lessor and such lender or purchaser in confidence and shall be used only for the purposes herein set forth.
- 17. Definition of Lessor. The term "Lessor" as used herein shall mean the owner or owners at the time in question of the fee title to the Premises, or, if this is a sublease, of the Lessee's interest in the prior lease. In the event of a transfer of Lessor's title or interest in the Premises or this Lease, Lessor shall deliver to the transferee or assignee (in cash or by credit) any unused Security Deposit held by Lessor. Upon such transfer or assignment and delivery of the Security Deposit, as aforesaid, the prior Lessor shall be relieved of all liability with respect to the obligations and/or covenants under this Lease thereafter to be performed by the Lessor. Subject to the foregoing, the obligations and/or covenants in this Lease to be performed by the Lessor shall be binding only upon the Lessor as hereinabove defined.

INITIALS	

- 18. Severability. The invalidity of any provision of this Lease, as determined by a court of competent jurisdiction, shall in no way affect the validity of any other provision hereof.
- 19. Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Lease shall mean and refer to calendar days.
- 20. Limitation on Liability. The obligations of Lessor under this Lease shall not constitute personal obligations of Lessor, or its partners, members, directors, officers or shareholders, and Lessee shall look to the Project, and to no other assets of Lessor, for the satisfaction of any liability of Lessor with respect to this Lease, and shall not seek recourse against Lessor's partners, members, directors, officers or shareholders, or any of their personal assets for such satisfaction.
- 21. Time of Essence. Time is of the essence with respect to the performance of all obligations to be performed or observed by the Parties under this Lease.
- 22. No Prior or Other Agreements; Broker Disclaimer. This Lease contains all agreements between the Parties with respect to any matter mentioned herein, and no other prior or contemporaneous agreement or understanding shall be effective. Lessor and Lessee each represents and warrants to the Brokers that it has made, and is relying solely upon, its own investigation as to the nature, quality, character and financial responsibility of the other Party to this Lease and as to the use, nature, quality and character of the Premises. Brokers have no responsibility with respect thereto or with respect to any default or breach hereof by either Party.

23. Notices

- 23.1 **Notice Requirements**. All notices required or permitted by this Lease or applicable law shall be in writing and may be delivered in person (by hand or by courier) or may be sent by regular, certified or registered mail or U.S. Postal Service Express Mail, with postage prepaid, or by facsimile transmission, or by email, and shall be deemed sufficiently given if served in a manner specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specify a different address for notice, except that upon Lessee's taking possession of the Premises, the Premises shall constitute Lessee's address for notice. A copy of all notices to Lessor shall be concurrently transmitted to such party or parties at such addresses as Lessor may from time to time hereafter designate in writing.
- 23.2 **Date of Notice**. Any notice sent by registered or certified mail, return receipt requested, shall be deemed given on the date of delivery shown on the receipt card, or if no delivery date is shown, the postmark thereon. If sent by regular mail the notice shall be deemed given 72 hours after the same is addressed as required herein and mailed with postage prepaid. Notices delivered by United States Express Mail or overnight courier that guarantees next day delivery shall be deemed given 24 hours after delivery of the same to the Postal Service or courier. Notices delivered by hand, or transmitted by facsimile transmission or by email shall be deemed delivered upon actual receipt. If notice is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.
- 23.3 **Options.** Notwithstanding the foregoing, in order to exercise any Options (see paragraph 39), the Notice must be sent by Certified Mail (return receipt requested), Express Mail (signature required), courier (signature required) or some other methodology that provides a receipt establishing the date the notice was received by the Lessor.

24. Waivers.

- (a) No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any other term, covenant or condition hereof. Lessor's consent to, or approval of, any act shall not be deemed to render unnecessary the obtaining of Lessor's consent to, or approval of, any subsequent or similar act by Lessee, or be construed as the basis of an estoppel to enforce the provision or provisions of this Lease requiring such consent.
- (b) The acceptance of Rent by Lessor shall not be a waiver of any Default or Breach by Lessee. Any payment by Lessee may be accepted by Lessor on account of monies or damages due Lessor, notwithstanding any qualifying statements or conditions made by Lessee in connection therewith, which such statements and/or conditions shall be of no force or effect whatsoever unless specifically agreed to in writing by Lessor at or before the time of deposit of such payment.
- (c) THE PARTIES AGREE THAT THE TERMS OF THIS LEASE SHALL GOVERN WITH REGARD TO ALL MATTERS RELATED THERETO AND HEREBY WAIVE THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE TO THE EXTENT THAT SUCH STATUTE IS INCONSISTENT WITH THIS LEASE.

25. Disclosures Regarding The Nature of a Real Estate Agency Relationship.

- (a) When entering into a discussion with a real estate agent regarding a real estate transaction, a Lessor or Lessee should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Lessor and Lessee acknowledge being advised by the Brokers in this transaction, as follows:
- (i) <u>Lessor's Agent</u>. A Lessor's agent under a listing agreement with the Lessor acts as the agent for the Lessor only. A Lessor's agent or subagent has the following affirmative obligations: <u>To the Lessor</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessor. <u>To the Lessee and the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (ii) <u>Lessee's Agent</u>. An agent can agree to act as agent for the Lessee only. In these situations, the agent is not the Lessor's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Lessor. An agent acting only for a Lessee has the following affirmative obligations. <u>To the Lessee</u>: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Lessee. <u>To the Lessee and the Lessor</u>: (a) Diligent exercise of reasonable skills and care in performance of the agent's duties. (b) A duty of honest and fair dealing and good faith. (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth
- (iii) Agent Representing Both Lessor and Lessee. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Lessor and the Lessee in a transaction, but only with the knowledge and consent of both the Lessor and the Lessee. In a dual agency situation, the agent has the following affirmative obligations to both the Lessor and the Lessee: (a) A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Lessor or the Lessee. (b) Other duties to the Lessor and the Lessee as stated above in subparagraphs (i) or (ii). In representing both Lessor and Lessee, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating to either Lessee's or Lessor's financial position, motivations, bargaining position, or other personal information that may impact rent, including Lessor's willingness to accept a rent less than the listing rent or Lessee's willingness to pay rent greater than the rent offered. The above duties of the agent in a real estate transaction do not relieve a Lessor or Lessee from the responsibility to protect their own interests. Lessor and Lessee should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Both Lessor and Lessee should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.
- (b) Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this Lease may be brought against Broker more than one year after the Start Date and that the liability (including court costs and attorneys' fees), of any Broker with respect to any such lawsuit and/or legal proceeding shall not exceed the fee received by such Broker pursuant to this Lease; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.
- (c) Lessor and Lessee agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- 26. No Right To Holdover. Lessee has no right to retain possession of the Premises or any part thereof beyond the expiration or termination of this Lease. At or prior to the expiration or termination of this Lease Lessee shall deliver exclusive possession of the Premises to Lessor. For purposes of this provision and Paragraph 13.1(a), exclusive possession shall mean that Lessee shall have vacated the Premises, removed all of its personal property therefrom and that the Premises have been returned in the condition specified in this Lease. In the event that Lessee does not deliver exclusive possession to Lessor as specified above, then Lessor's damages during any holdover period shall be computed at the amount of the Rent (as defined in Paragraph 4.1) due during the last full month before the expiration or termination of this Lease (disregarding any temporary abatement of Rent that may have been in effect), but with Base Rent and Percentage Rent Rate being 150% of the Base Rent and Percentage Rent Rate payable during such last full month. Nothing contained herein shall be construed as consent by Lessor to any holding over by Lessee.
- 27. Cumulative Remedies. No remedy or election hereunder shall be deemed exclusive but shall, wherever possible, be cumulative with all other remedies at law or in equity.
- 28. Covenants and Conditions; Construction of Agreement. All provisions of this Lease to be observed or performed by Lessee are both covenants and conditions. In construing this Lease, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Lease. Whenever required by the

INITIALS

INITIALS

context, the singular shall include the plural and vice versa. This Lease shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.

29. Binding Effect; Choice of Law. This Lease shall be binding upon the Parties, their personal representatives, successors and assigns and be governed by the laws of the State in which the Premises are located. Any litigation between the Parties hereto concerning this Lease shall be initiated in the county in which the Premises are located. Signatures to this Lease accomplished by means of electronic signature or similar technology shall be legal and binding.

30. Subordination; Attornment; Non-Disturbance.

- 30.1 **Subordination**. This Lease and any Option granted hereby shall be subject and subordinate to any ground lease, mortgage, deed of trust, or other hypothecation or security device (collectively, "Security Device"), now or hereafter placed upon the Premises, to any and all advances made on the security thereof, and to all renewals, modifications, and extensions thereof. Lessee agrees that the holders of any such Security Devices (in this Lease together referred to as "Lender") shall have no liability or obligation to perform any of the obligations of Lessor under this Lease. Any Lender may elect to have this Lease and/or any Option granted hereby superior to the lien of its Security Device by giving written notice thereof to Lessee, whereupon this Lease and such Options shall be deemed prior to such Security Device, notwithstanding the relative dates of the documentation or recordation thereof.
- 30.2 **Attornment**. In the event that Lessor transfers title to the Premises, or the Premises are acquired by another upon the foreclosure or termination of a Security Device to which this Lease is subordinated (i) Lessee shall, subject to the non-disturbance provisions of Paragraph 30.3, attorn to such new owner, and upon request, enter into a new lease, containing all of the terms and provisions of this Lease, with such new owner for the remainder of the term hereof, or, at the election of the new owner, this Lease will automatically become a new lease between Lessee and such new owner, and (ii) Lessor shall thereafter be relieved of any further obligations hereunder and such new owner shall assume all of Lessor's obligations, except that such new owner shall not: (a) be liable for any act or omission of any prior lessor or with respect to events occurring prior to acquisition of ownership; (b) be subject to any offsets or defenses which Lessee might have against any prior lessor, (c) be bound by prepayment of more than one month's rent, or (d) be liable for the return of any security deposit paid to any prior lessor which was not paid or credited to such new owner.
- 30.3 **Non-Disturbance**. With respect to Security Devices entered into by Lessor after the execution of this Lease, Lessee's subordination of this Lease shall be subject to receiving a commercially reasonable non-disturbance agreement (a "**Non-Disturbance Agreement**") from the Lender which Non-Disturbance Agreement provides that Lessee's possession of the Premises, and this Lease, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises. Further, within 60 days after the execution of this Lease, Lessor shall, if requested by Lessee, use its commercially reasonable efforts to obtain a Non-Disturbance Agreement from the holder of any pre-existing Security Device which is secured by the Premises. In the event that Lessor is unable to provide the Non-Disturbance Agreement within said 60 days, then Lessee may, at Lessee's option, directly contact Lender and attempt to negotiate for the execution and delivery of a Non-Disturbance Agreement.
- 30.4 **Self-Executing**. The agreements contained in this Paragraph 30 shall be effective without the execution of any further documents; provided, however, that, upon written request from Lessor or a Lender in connection with a sale, financing or refinancing of the Premises, Lessee and Lessor shall execute such further writings as may be reasonably required to separately document any subordination, attornment and/or Non-Disturbance Agreement provided for herein.
- **31. Attorneys' Fees.** If any Party or Broker brings an action or proceeding involving the Premises whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "**Prevailing Party**" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred. In addition, Lessor shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is subsequently commenced in connection with such Default or resulting Breach (\$200 is a reasonable minimum per occurrence for such services and consultation).
- **32.** Lessor's Access; Showing Premises; Repairs. Lessor and Lessor's agents shall have the right to enter the Premises at any time, in the case of an emergency, and otherwise at reasonable times after reasonable prior notice for the purpose of showing the same to prospective purchasers, lenders, or tenants, and making such alterations, repairs, improvements or additions to the Premises as Lessor may deem necessary or desirable and the erecting, using and maintaining of utilities, services, pipes and conduits through the Premises and/or other premises as long as there is no material adverse effect on Lessee's use of the Premises. All such activities shall be without abatement of rent or liability to Lessee.
- **33. Auctions.** Lessee shall not conduct, nor permit to be conducted, any auction upon the Premises without Lessor's prior written consent, which consent shall be granted or denied at Lessor's sole discretion.
- 34. Signs. Lessor may place on the Premises ordinary "For Sale" signs at any time and ordinary "For Lease" signs during the last 6 months of the term hereof. All signs must comply with all Applicable Requirements. Lessee shall not place, construct, or maintain on the glass panes or supports of the show windows of the Premises, the doors, exterior walls or the roof of the Building, or anywhere else within the Shopping Center outside of the Premises, or on any interior portions of the Premises that are visible from the exterior of the Premises, any signs, advertisements, names, insignia, trademarks, descriptive material or any other items without Lessor's prior written consent, which consent shall be granted or denied at Lessor's sole discretion. Lessor shall designate the size, shape, color, design, and location of all exterior sign(s) to be installed by Lessee, and Lessee shall, at Lessee's sole cost and expense, fabricate, construct and install all such sign(s) in full compliance with Lessor's designation and in accordance with the Sign Criteria for the Shopping Center attached hereto, if any. Lessee agrees to submit plans and specifications for Lessee's sign(s) for Lessor's written approval within 30 days after the full execution hereof and to install such sign(s) prior to opening for business at the Premises. Lessor, at Lessee's cost, may remove any item placed, constructed or maintained in, upon or about the Premises or Shopping Center which does not comply with this paragraph. In the event there is a pole, pylon or monument sign for the Shopping Center, Lessor shall have the right, but not the obligation, to install lettering designating Lessee's business on such sign, at Lessee's expense, with Lessor's approval of location, size, style and color. All signs that are permanently attached to the Premises or Building shall become the property of Lessor at the expiration or earlier termination hereof; provided, however, that Lessee shall promptly remove all such signs if Lessor so elects, and Lessee shall prom
- **35. Termination; Merger.** Unless specifically stated otherwise in writing by Lessor, the voluntary or other surrender of this Lease by Lessee, the mutual termination or cancellation hereof, or a termination hereof by Lessor for Breach by Lessee, shall automatically terminate any sublease or lesser estate in the Premises; provided, however, that Lessor may elect to continue any one or all existing subtenancies. Lessor's failure within 10 days following any such event to elect to the contrary by written notice to the holder of any such lesser interest, shall constitute Lessor's election to have such event constitute the termination of such interest.
- **36. Consents.** All requests for consent shall be in writing. Except as otherwise provided herein, wherever in this Lease the consent of a Party is required to an act by or for the other Party, such consent shall not be unreasonably withheld or delayed. In those express instances where consent is within the sole discretion of a party, the party shall have no obligation to adhere to a standard of reasonableness. Lessor's actual reasonable costs and expenses (including but not limited to architects', attorneys', engineers' and other consultants' fees) incurred in the consideration of, or response to, a request by Lessee for any Lessor consent, including but not limited to consents to an assignment, a subletting or the presence or use of a Hazardous Substance, shall be paid by Lessee upon receipt of an invoice and supporting documentation therefor. Lessor's consent to any act, assignment or subletting shall not constitute an acknowledgment that no Default or Breach by Lessee of this Lease exists, nor shall such consent be deemed a waiver of any then existing Default or Breach, except as may be otherwise specifically stated in writing by Lessor at the time of such consent. The failure to specify herein any particular condition to Lessor's consent shall not preclude the imposition by Lessor at the time of consent of such further or other conditions as are then reasonable with reference to the particular matter for which consent is being given. In the event that either Party disagrees with any determination made by the other hereunder and reasonably requests the reasons for such determination, the determining party shall furnish its reasons in writing and in reasonable detail within 10 business days following such request.

37. Guarantor.

- 37.1 **Execution**. The Guarantors, if any, shall each execute a guaranty in the form most recently published by AIR CRE.
- 37.2 **Default**. It shall constitute a Default of the Lessee if any Guarantor fails or refuses, upon request to provide: (a) evidence of the execution of the guaranty, including the authority of the party signing on Guarantor's behalf to obligate Guarantor, and in the case of a corporate Guarantor, a certified copy of a resolution of its board of directors authorizing the making of such guaranty, (b) current financial statements, (c) an Estoppel Certificate, or (d) written confirmation that the guaranty is still in effect.
- **38. Quiet Possession.** Subject to payment by Lessee of the Rent and performance of all of the covenants, conditions and provisions on Lessee's part to be observed and performed under this Lease, Lessee shall have quiet possession and quiet enjoyment of the Premises during the term hereof.
- **39. Options.** If Lessee is granted any option, as defined below, then the following provisions shall apply.

INITIALS

- 39.1 Definition. "Option" shall mean: (a) the right to extend or reduce the term of or renew this Lease or to extend or reduce the term of or renew any lease that Lessee has on other property of Lessor; (b) the right of first refusal or first offer to lease either the Premises or other property of Lessor; (c) the right to purchase, the right of first offer to purchase or the right of first refusal to purchase the Premises or other property of Lessor.
- 39.2 Options Personal To Original Lessee. Any Option granted to Lessee in this Lease is personal to the original Lessee, and cannot be assigned or exercised by anyone other than said original Lessee and only while the original Lessee is in full possession of the Premises and, if requested by Lessor, with Lessee certifying that Lessee has no intention of thereafter assigning or subletting.
- 39.3 Multiple Options. In the event that Lessee has any multiple Options to extend or renew this Lease, a later Option cannot be exercised unless the prior Options have been validly exercised.

39.4 Effect of Default on Options.

- (a) Lessee shall have no right to exercise an Option: (i) during the period commencing with the giving of any notice of Default and continuing until said Default is cured, (ii) during the period of time any Rent is unpaid (without regard to whether notice thereof is given Lessee), (iii) during the time Lessee is in Breach of this Lease, or (iv) in the event that Lessee has been given 3 or more notices of separate Default, whether or not the Defaults are cured, during the 12 month period immediately preceding the exercise of the Option.
- (b) The period of time within which an Option may be exercised shall not be extended or enlarged by reason of Lessee's inability to exercise an Option because of the provisions of Paragraph 39.4(a).
- (c) An Option shall terminate and be of no further force or effect, notwithstanding Lessee's due and timely exercise of the Option, if, after such exercise and prior to the commencement of the extended term or completion of the purchase, (i) Lessee fails to pay Rent for a period of 30 days after such Rent becomes due (without any necessity of Lessor to give notice thereof), or (ii) if Lessee commits a Breach of this Lease.
- 40. Security Measures. Lessee hereby acknowledges that the Rent payable to Lessor hereunder does not include the cost of guard service or other security measures, and that Lessor shall have no obligation whatsoever to provide same. Lessee assumes all responsibility for the protection of the Premises, Lessee, its agents and invitees and their property from the acts of third parties. While Lessor does not assume any responsibility to provide any security measures or any liability for failure to provide security measures or for any inadequacy thereof, Lessor shall have the authority to institute or continue such security measures as Lessor in its sole discretion deems necessary or appropriate from time to time, the cost and expenses of which shall be considered Common Area Operating Expenses. To the degree directed by Lessor, Lessee shall coordinate its security measures at the Premises with the security measures instituted by Lessor, if any.
- 41. Reservations. Lessor reserves the right: (i) to grant, without the consent or joinder of Lessee, such easements, rights and dedications that Lessor deems necessary, (ii) to cause the recordation of parcel maps and restrictions, and (iii) to create and/or install new utility raceways, so long as such easements, rights, dedications, maps, restrictions, and utility raceways do not unreasonably interfere with the use of the Premises by Lessee. Lessee agrees to sign any documents reasonably requested by Lessor to effectuate such rights.
- Building Planning. Lessor shall have the right at any time or times, upon giving Lessee not less than 60 days prior written notice, to provide and furnish Lessee with space of comparable visibility located elsewhere within any of the buildings within the Shopping Center and to move Lessee into such new space, provided that the usable area of such new space is not less than the usable area of the Premises and provided that all of Lessee's reasonable out-of-pocket moving expenses (including but not limited to the cost of moving Lessee's personal property, the cost of reprinting Lessee's stationery or other business materials with the new address, and the cost to relocate and reinstall tenant improvements and Lessee's telecommunications and computer equipment) shall be paid by Lessor, and provided further that Lessor shall construct at Lessor's expense such improvements to such new space as shall be necessary to place it in a condition that is substantially comparable to the Premises. Except as provided in the immediately preceding sentence, Lessor shall have no obligation to improve such space or pay any other expenses incurred by Lessee as a result of such relocation. On such relocation, the terms and conditions of this Lease shall remain in full force and effect, including but not limited to the Base Rent payable hereunder and Lessee's Share (even if the usable area of such relocated Premises is in excess of the usable area of the Premises), except that the Premises shall be in such new location. Upon Lessor's request, the Parties shall execute an amendment to this Lease in form required by Lessor confirming the relocation of the Premises to such new location. If the new space does not meet with Lessee's approval, which approval Lessee shall give or withhold in accordance with Paragraph 36, Lessee shall have the right to cancel this Lease by giving Lessor written notice thereof within 15 days of receipt of Lessor's notification of its intent to relocate Lessee. Lessee's failure to give such notice within such 15 day period shall be deemed Lessee's approval of the new space. If timely notice is given by Lessee, then this Lease shall terminate unless Lessor rescinds Lessor's prior notice of its intent to relocate Lessee within 10 days after Lessor's receipt of Lessee's notice of cancellation.
- 43. Performance Under Protest. If at any time a dispute shall arise as to any amount or sum of money to be paid by one Party to the other under the provisions hereof, the Party against whom the obligation to pay the money is asserted shall have the right to make payment "under protest" and such payment shall not be regarded as a voluntary payment and there shall survive the right on the part of said Party to institute suit for recovery of such sum. If it shall be adjudged that there was no legal obligation on the part of said Party to pay such sum or any part thereof, said Party shall be entitled to recover such sum or so much thereof as it was not legally required to pay. A Party who does not initiate suit for the recovery of sums paid "under protest" within 6 months shall be deemed to have waived its right to protest such payment.

Authority; Multiple Parties; Execution.

- (a) If either Party hereto is a corporation, trust, limited liability company, partnership, or similar entity, each individual executing this Lease on behalf of such entity represents and warrants that he or she is duly authorized to execute and deliver this Lease on its behalf. Each Party shall, within 30 days after request, deliver to the other Party satisfactory evidence of such authority.
- (b) If this Lease is executed by more than one person or entity as "Lessee", each such person or entity shall be jointly and severally liable hereunder. It is agreed that any one of the named Lessees shall be empowered to execute any amendment to this Lease, or other document ancillary thereto and bind all of the named Lessees, and Lessor may rely on the same as if all of the named Lessees had executed such document.
- This Lease may be executed by the Parties in counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument.
- 45. Conflict. Any conflict between the printed provisions of this Lease and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions.
- Offer. Preparation of this Lease by either party or their agent and submission of same to the other Party shall not be deemed an offer to lease to the other Party. This Lease is not intended to be binding until executed and delivered by all Parties hereto.
- **47. Amendments.** This Lease may be modified only in writing, signed by the Parties in interest at the time of the modification. As long as they do not materially change Lessee's obligations hereunder, Lessee agrees to make such reasonable non-monetary modifications to this Lease as may be reasonably required by a Lender in connection with the obtaining of normal financing or refinancing of the Premises.
- Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY

OR ARISING OUT OF THIS AGREEMENT.	
49. Arbitration of Disputes. An Addendum requiring the Arbitration of all disputattached to this Lease.	ites between the Parties and/or Brokers arising out of this Lease is is not
50. Accessibility; Americans with Disabilities Act. (a) The Premises:	
determine whether the subject premises comply with all of the applicable construerequire a CASp inspection of the subject premises, the commercial property owner.	
	vas determined that the Premises met all applicable construction-related accessibility that it received a copy of the inspection report at least 48 hours prior to executing
have undergone an inspection by a Certified Access Specialist (CASp) and it w	vas determined that the Premises did not meet all applicable construction-related
	
INITIALS	INITIALS

accessibility standards pursuant to California Civil Code §55.51 et seq. Lessee acknowledges that it received a copy of the inspection report at least 48 hours prior to executing this Lease and agrees to keep such report confidential except as necessary to complete repairs and corrections of violations of construction related accessibility standards.

In the event that the Premises have been issued an inspection report by a CASp the Lessor shall provide a copy of the disability access inspection certificate to Lessee within 7 days of the execution of this Lease.

(b) Since compliance with the Americans with Disabilities Act (ADA) and other state and local accessibility statutes are dependent upon Lessee's specific use of the Premises, Lessor makes no warranty or representation as to whether or not the Premises comply with ADA or any similar legislation. In the event that Lessee's use of the Premises requires modifications or additions to the Premises in order to be in compliance with ADA or other accessibility statutes, Lessee agrees to make any such necessary modifications and/or additions at Lessee's expense.

LESSOR AND LESSEE HAVE CAREFULLY READ AND REVIEWED THIS LEASE AND EACH TERM AND PROVISION CONTAINED HEREIN, AND BY THE EXECUTION OF THIS LEASE SHOW THEIR INFORMED AND VOLUNTARY CONSENT THERETO. THE PARTIES HEREBY AGREE THAT, AT THE TIME THIS LEASE IS EXECUTED, THE TERMS OF THIS LEASE ARE COMMERCIALLY REASONABLE AND EFFECTUATE THE INTENT AND PURPOSE OF LESSOR AND LESSEE WITH RESPECT TO THE PREMISES.

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS LEASE OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

- 1. SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS LEASE.
- 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PREMISES. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING OF THE PREMISES, THE STRUCTURAL INTEGRITY, THE CONDITION OF THE ROOF AND OPERATING SYSTEMS, COMPLIANCE WITH THE AMERICANS WITH DISABILITIES ACT AND THE SUITABILITY OF THE PREMISES FOR LESSEE'S INTENDED USE.

WARNING: IF THE PREMISES ARE LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THE LEASE MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PREMISES ARE LOCATED.

The parties hereto have executed this Lease at the place and on the dates specified	above their respective signatures.
Executed at:	Executed at:
On:	On:
	Division .
By LESSOR:	By LESSEE:
Dur	Dur
By: Name Printed:	By: Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
By:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Address:	Address:
Federal ID No.:	Federal ID No.:
BROKER	BROKER
	
Attn:	Attn:
Title:	Title:
Address:	Address:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Federal ID No.: Broker DRE License #:	Federal ID No.:
\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ 	Broker DRE License #:
Agent DRE License #:	Agent DRE License #:

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2019 AIR CRE. All Rights Reserved.
SCLN-20.30, Revised 10-22-2020

Limited liability company: Manager(s)

Husband and wife: Community Property or Separate property

Joint tenants: All joint tenants

Tenants in common: All tenants in common

Corporation:

Board chairman, president or any vice president

and

Secretary, asst. secretary, CFO or asst. treasurer

General partnership: General partner with authority

Limited Partnership: General partner



PROPERTY INFORMATION SHEET

(For the sale or leasing of non-residential properties)

PREFACE:

Purpose: This Statement is NOT a warranty as to the actual condition of the Property/Premises. The purpose is, instead, to provide the brokers and the potential buyer/lessee with important information about the Property/Premises which is currently in the actual knowledge of the Owner and which the Owner is required by law to disclose.

Actual Knowledge: For purposes of this Statement the phrase 'actual knowledge' means: the awareness of a fact, or the awareness of sufficient information and

circumstances so as to cause one to believe that a certain situation or condition probably exists. TO WHOM IT MAY CONCERN: ("Owner"), owns the Property/Premises commonly known as (street address, city, state, zip) located in the County of generally described as (describe briefly the nature of the Premises or Property) (hereinafter "Property"), and certifies that: Material Physical Defects. Owner has no actual knowledge of any material physical defects in the Property or any improvements and structures thereon, including, but not limited to the roof, except (if there are no exceptions write "NONE"): Equipment. Owner has no actual knowledge that the heating, ventilating, air conditioning, plumbing, loading doors, electrical and lighting systems, life safety systems, security systems and mechanical equipment existing on the Property as of the date hereof, if any, are not in good operating order and condition, except (if there are no exceptions write "NONE"): Owner has no actual knowledge of any leases, financing agreements, liens or other agreements affecting any equipment which is being included with the Property, except (if there are no exceptions write "NONE"): Soil Conditions. Owner has no actual knowledge that the Property has any slipping, sliding, settling, flooding, ponding or any other grading, drainage or soil problems, except (if there are no exceptions write "NONE"): **Utilities.** Owner represents and warrants that the Property is served by the following utilities (check the appropriate boxes) public sewer system and the cost of installation thereof has been fully paid, \square private septic system, \square electricity, \square natural gas, \square domestic water, telephone, and U other: Insurance. Owner has no actual knowledge of any insurance claims filed regarding the Property during the preceding 3 years, except (if there are no exceptions write "NONE"): Compliance With Laws. Owner has no actual knowledge of any aspect or condition of the Property which violates applicable laws, rules, regulations, codes, or covenants, conditions or restrictions, or of improvements or alterations made to the Property without a permit where one was required, or of any unfulfilled order or directive of any applicable government agency or of any casualty insurance company that any work of investigation, remediation, repair, maintenance or improvement is to be performed on the Property, except (if there are no exceptions write "NONE"): Hazardous Substances and Mold. A. Owner has no actual knowledge of the Property ever having been used as a waste dump, of the past or present existence of any above or below ground storage tanks on the Property, or of the current existence on the Property of asbestos, transformers containing PCB's or any hazardous, toxic or infectious substance whose nature and/or quantity of existence, use, manufacture or effect, render it subject to Federal, state or local regulation, investigation, remediation or removal as potentially injurious to public health or welfare, except (if there are no exceptions write "NONE"): Owner represents and warrants that it is not currently, and never has been engaged in the business of hauling waste, and never stored hazardous substances on the Property, except (if there are no exceptions write "NONE").

C. Owner has no actual knowledge of the existence on the Property of hazardous levels of any mold or fungi defined as toxic under applicable state or Federal law, except (if there are no exceptions write "NONE"); Fire Damage. Owner has no actual knowledge of any structure on the Property having suffered material fire damage, except (if there are no exceptions write "NONE"): Actions, Suits or Proceedings. Owner has no actual knowledge that any actions, suits or proceedings are pending or threatened before any court, arbitration tribunal, governmental department, commission, board, bureau, agency or instrumentality that would affect the Property or the right or ability of an owner or tenant to convey, occupy or utilize the Property, except (if there are no exceptions write "NONE"): Owner has not served any Notices of Default on any of the tenants of the Property which have not been resolved except (if there are no exceptions write "NONE"): 10. Governmental Proceedings. Owner has no actual knowledge of any existing or contemplated condemnation, environmental, zoning, redevelopment agency plan or other land use regulation proceedings which could detrimentally affect the value, use and operation of the Property, except (if there are no exceptions write "NONE"): 11. Unrecorded Title Matters. Owner has no actual knowledge of any encumbrances, covenants, conditions, restrictions, easements, licenses, liens, charges or other matters which affect the title of the Property that are not recorded in the official records of the county recorder where the Property is located, except (if there are no exceptions write "NONE"): Leases. Owner has no actual knowledge of any leases, subleases or other tenancy agreements affecting the Property, except (if there are no exceptions write "NONF"): 13. Options. Owner has no actual knowledge of any options to purchase, rights of first refusal, rights of first offer or other similar agreements affecting the Property, except (if there are no exceptions write "NONE"): 14. Short Sale/Foreclosure. The ability of the Owner to complete a sale of the Property is contingent is not contingent upon obtaining the consent of one or more lenders to conduct a 'short sale', ie. a sale for less that the amount owing on the Property. (This paragraph only needs to be completed if this Property Information Sheet is being completed in connection with the proposed sale of the Property) One or more of any loans secured by the Property is is not in 15. **Energy Efficiency**. The Property has has not been granted an energy efficiency rating or certification such as one from the U.S. Green Building Council's Leadership in Energy and Environmental Design (LEED) or Seller/Lessor does not know if the Property has been granted such a rating or certificate. If such a rating or certification has been obtained please describe the rating or certification and provide the name of the organization that granted it: 16. Other. (It will be presumed that there are no additional items which warrant disclosure unless they are set forth herein):

INITIALS

INITIALS
© 2017 AIR CRE. All Rights Reserved.
PI-8.03, Revised 10-22-2020

Page 1 of 2

The statements herein will be relied upon by brokers, buyers, lessees, lenders and others. Therefore, Owner and/or the Owner's Property Manager has reviewed and modified this printed statement as necessary to accurately and completely state all the known material facts concerning the Property. To the extent such modifications are not made, this statement may be relied upon as printed. This statement, however, shall not relieve a buyer or lessee of responsibility for independent investigation of the Property. Owner agrees to promptly notify, in writing, all appropriate parties of any material changes which may occur in the statements contained herein from the date this statement is signed until title to the Property is transferred, or the lease is executed. Signatures to this Statement accomplished by means of electronic signature or similar technology shall be legal and binding.

Date:	OWNER
(fill in date of execution)	<u> </u>
	Ву:
	Name Printed:
	Title:
Buyer/lessee hereby acknowledges receipt of a copy of this Property Information	Sheet on (Fill in date received)
	BUYER/LESSEE
	
	By: Name Printed:
	Name Printed: Title:
AIR CRE * https://www.aircre.com NOTICE: No part of these works may be repr	* 213-687-8777 * contracts@aircre.com oduced in any form without permission in writing.

INITIALS



DISCLOSURE REGARDING REAL ESTATE AGENCY RELATIONSHIP

(As required by the Civil Code)

When you enter into a discussion with a real estate agent regarding a real estate transaction, you should from the outset understand what type of agency relationship or representation you wish to have with the agent in the transaction.

SELLER'S AGENT

A Seller's agent under a listing agreement with the Seller acts as the agent for the Seller only. A Seller's agent or a subagent of that agent has the following affirmative obligations:

To the Seller: A fiduciary duty of utmost care, integrity, honesty and loyalty in dealings with the Seller. To the Buyer and the Seller:

- (a) Diligent exercise of reasonable skill and care in performance of the agent's duties.
- (b) A duty of honest and fair dealing and good faith.
- (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the parties.

An agent is not obligated to reveal to either party any confidential information obtained from the other party that does not involve the affirmative duties set forth above.

BUYER'S AGENT

A Buyer's agent can, with a Buyer's consent, agree to act as agent for the Buyer only. In these situations, the agent is not the Seller's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Seller. An agent acting only for a Buyer has the following affirmative obligations:

To the Buyer: A fiduciary duty of utmost care, integrity, honesty and loyalty in dealings with the Buyer. To the Buyer and the Seller:

- (a) Diligent exercise of reasonable skill and care in performance of the agent's duties.
- (b) A duty of honest and fair dealing and good faith.
- (c) A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the parties.

An agent is not obligated to reveal to either party any confidential information obtained from the other party that does not involve the affirmative duties set forth

AGENT REPRESENTING BOTH SELLER AND BUYER

A real estate agent, either acting directly or through one or more salesperson and broker associates, can legally be the agent of both the Seller and the Buyer in a transaction, but only with the knowledge and consent of both the Seller and the Buyer.

- In a dual agency situation, the agent has the following affirmative obligations to both the Seller and the Buyer:

 (a) A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either the Seller or the Buyer.
 - (b) Other duties to the Seller and the Buyer as stated above in their respective sections.

In representing both Seller and Buyer, a dual agent may not, without the express permission of the respective party, disclose to the other party confidential information, including, but not limited to, facts relating to either the Buyer's or Seller's financial position, motivations, bargaining position, or other personal information that may impact price, including the Seller's willingness to accept a price less than the listing price or the Buyer's willingness to pay a price greater than the price offered.

SELLER AND BUYER RESPONSIBILITIES

Either the purchase agreement or a separate document will contain a confirmation of which agent is representing you and whether that agent is representing you exclusively in the transaction or acting as a dual agent. Please pay attention to that confirmation to make sure it accurately reflects your understanding of your agent's role. The above duties of the agent in a real estate transaction do not relieve a Seller or Buyer from the responsibility to protect his or her own interests. You should carefully read all agreements to assure that they adequately express your understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. If you are a Buyer, you have the duty to exercise reasonable care to protect yourself, including as to those facts about the property which are known to you or within your diligent attention and observation. Both Sellers and Buyers should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.

Throughout your real property transaction you may receive more than one disclosure form, depending upon the number of agents assisting in the transaction. The law requires each agent with whom you have more than a casual relationship to present you with this disclosure form. You should read its contents each time it is presented to you, considering the relationship between you and the real estate agent in your specific transaction. This disclosure form includes the provisions of Sections 2079.13 to 2079.24, inclusive, of the Civil Code set forth on page 2. Read it carefully. I/WE ACKNOWLEDGE RECEIPT OF A COPY OF THIS DISCLOSURE AND THE PORTIONS OF THE CIVIL CODE PRINTED ON THE BACK (OR A SEPARATE PAGE).

Buyer	Seller	Lessor	Lessee	\checkmark		Date:
Buyer	Seller	Lessor	Lessee			Date:
Agent: Real	DRI Estate Broker	E Lic. #: (Firm)				
Ву:				DRE Lic. #:	Date:	_
	erson or Broke				THE LEGAL VALIDITY OF AD	FOLIACY OF THIS FORM FOR ANY SDE

THIS FORM HAS BEEN PREPARED BY AIR CRE. NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ADEQUACY OF THIS FORM FOR ANY SPECIFIC TRANSACTION. PLEASE SEEK LEGAL COUNSEL AS TO THE APPROPRIATENESS OF THIS FORM.

NITIALS		INITIALS
2019 AIR CRE	All Rights Reserved	

DISCLOSURE REGARDING REAL ESTATE AGENCY RELATIONSHIP CIVIL CODE SECTIONS 2079.13 THROUGH 2079.24 (2079.16 APPEARS ON THE FRONT)

2079.13. As used in Sections 2079.7 and 2079.14 to 2079.24, inclusive, the following terms have the following meanings:

(a) "Agent" means a person acting under provisions of Title 9 (commencing with Section 2295) in a real property transaction, and includes a person who is licensed as a real estate broker under Chapter 3 (commencing with Section 10130) of Part 1 of Division 4 of the Business and Professions Code, and under whose license a listing is executed or an offer to purchase is obtained. The agent in the real property transaction bears responsibility for that agent's salespersons or broker associates who perform as agents of the agent. When a salesperson or broker associate owes a duty to any principal, or to any buyer or seller who is not a principal, in a real property transaction, that duty is equivalent to the duty owed to that party by the broker for whom the salesperson or broker associate functions. (b) "Buyer" means a transferee in a real property transaction, and includes a person who executes an offer to purchase real property from a seller through an agent, or who seeks the services of an agent in more than a casual, transitory, or preliminary manner, with the object of entering into a real property transaction. "Buyer" includes vendee or lessee of real property. (c) "Commercial real property" means all real property in the state, except (1) single-family residential real property, (2) dwelling units made subject to Chapter 2 (commencing with Section 1940) of Title 5, (3) a mobile home, as defined in Section 798.3, (4) vacant land, or (5) a recreational vehicle, as defined in Section 799.29. (d) "Dual agent" means an agent acting, either directly or through a salesperson or broker associate, as agent for both the seller and the buyer in a real property transaction. (e) "Listing agreement" means a written contract between a seller of real property and an agent, by which the agent has been authorized to sell the real property or to find or obtain a buyer, including rendering other services for which a real estate license is required to the seller pursuant to the terms of the agreement. (f) "Seller's agent" means a person who has obtained a listing of real property to act as an agent for compensation. (g) "Listing price" is the amount expressed in dollars specified in the listing for which the seller is willing to sell the real property through the seller's agent. (h) "Offering price" is the amount expressed in dollars specified in an offer to purchase for which the buyer is willing to buy the real property. (i) "Offer to purchase" means a written contract executed by a buyer acting through a buyer's agent that becomes the contract for the sale of the real property upon acceptance by the seller. (j) "Real property" means any estate specified by subdivision (1) or (2) of Section 761 in property, and includes (1) single-family residential property, (2) multi-unit residential property with more than four dwelling units, (3) commercial real property, (4) vacant land, (5) a ground lease coupled with improvements, or (6) a manufactured home as defined in Section 18007 of the Health and Safety Code, or a mobile home as defined in Section 18008 of the Health and Safety Code, when offered for sale or sold through an agent pursuant to the authority contained in Section 10131.6 of the Business and Professions Code. (k) "Real property transaction" means a transaction for the sale of real property in which an agent is retained by a buyer, seller, or both a buyer and seller to act in that transaction, and includes a listing or an offer to purchase. (I) "Sell," "sale," or "sold" refers to a transaction for the transfer of real property from the seller to the buyer and includes exchanges of real property between the seller and buyer, transactions for the creation of a real property sales contract within the meaning of Section 2985, and transactions for the creation of a leasehold exceeding one year's duration. (m) "Seller" means the transferor in a real property transaction and includes an owner who lists real property with an agent. whether or not a transfer results, or who receives an offer to purchase real property of which he or she is the owner from an agent on behalf of another. "Seller' includes both a vendor and a lessor of real property. (n) "Buyer's agent" means an agent who represents a buyer in a real property transaction.

2079.14. A seller's agent and buyer's agent shall provide the seller and buyer in a real property transaction with a copy of the disclosure form specified in Section 2079.16, and shall obtain a signed acknowledgment of receipt from that seller and buyer, except as provided in Section 2079.15, as follows: (a) The seller's agent, if any, shall provide the disclosure form to the seller prior to entering into the listing agreement. (b) The buyer's agent shall provide the disclosure form to the buyer as soon as practicable prior to execution of the buyer's offer to purchase. If the offer to purchase is not prepared by the buyer's agent, the buyer's agent shall present the disclosure form to the buyer not later than the next business day after receiving the offer to purchase from the buyer.

2079.15. In any circumstance in which the seller or buyer refuses to sign an acknowledgment of receipt pursuant to Section 2079.14, the agent shall set forth, sign, and date a written declaration of the facts of the refusal.

2079.16 Reproduced on Page 1 of this AD form.

2079.17(a) As soon as practicable, the buyer's agent shall disclose to the buyer and seller whether the agent is acting in the real property transaction as the buyer's agent, or as a dual agent representing both the buyer and the seller. This relationship shall be confirmed in the contract to purchase and sell real property or in a separate writing executed or acknowledged by the seller, the buyer, and the buyer's agent prior to or coincident with execution of that contract by the buyer and the seller, respectively. (b) As soon as practicable, the seller's agent shall disclose to the seller whether the seller's agent is acting in the real property transaction as the seller's agent, or as a dual agent representing both the buyer and seller. This relationship shall be confirmed in the contract to purchase and sell real property or in a separate writing executed or acknowledged by the seller and the seller's agent prior to or coincident with the execution of that contract by the seller.

(C) CONFIRMATION: The following agency relationships are confirmed for this transaction.

Seller's Brokerage Firm DO NOT COMPLETE, SAMPLE ONLY License Number
Is the broker of (check one): ☐ the seller; or ☐ both the buyer and seller. (dual agent)
Seller's Agent DO NOT COMPLETE, SAMPLE ONLY License Number
Is (check one): ☐ the Seller's Agent. (salesperson or broker associate); or ☐ both the Buyer's Agent and the Seller's Agent. (dual agent)
Buyer's Brokerage Firm DO NOT COMPLETE, SAMPLE ONLY License Number
Is the broker of (check one): ☐ the buyer; or ☐ both the buyer and seller. (dual agent)
Buyer's Agent DO NOT COMPLETE, SAMPLE ONLY License Number
Is (check one): ☐ the Buyer's Agent. (salesperson or broker associate); or ☐ both the Buyer's Agent and the Seller's Agent. (dual agent)

(d) The disclosures and confirmation required by this section shall be in addition to the disclosure required by Section 2079.14. An agent's duty to provide disclosure and confirmation of representation in this section may be performed by a real estate salesperson or broker associate affiliated with that broker.

2079.18 (Repealed pursuant to AB-1289, 2017-18 California Legislative session)

2079.19 The payment of compensation or the obligation to pay compensation to an agent by the seller or buyer is not necessarily determinative of a particular agency relationship between an agent and the seller or buyer. A listing agent and a selling agent may agree to share any compensation or commission paid, or any right to any compensation or commission for which an obligation arises as the result of a real estate transaction, and the terms of any such agreement shall not necessarily be determinative of a particular relationship.

2079.20 Nothing in this article prevents an agent from selecting, as a condition of the agent's employment, a specific form of agency relationship not specifically prohibited by this article if the requirements of Section 2079.14 and Section 2079.17 are complied with.

2079.21 (a) A dual agent may not, without the express permission of the seller, disclose to the buyer any confidential information obtained from the seller. (b) A dual agent may not, without the express permission of the buyer, disclose to the seller any confidential information obtained from the buyer. (c) "Confidential information" means facts relating to the client's financial position, motivations, bargaining position, or other personal information that may impact price, such as the seller is willing to accept a price less than the listing price or the buyer is willing to pay a price greater than the price offered. (d) This section does not alter in any way the duty or responsibility of a dual agent to any principal with respect to confidential information other than price.

2079.22 Nothing in this article precludes a seller's agent from also being a buyer's agent. If a seller or buyer in a transaction chooses to not be represented by an agent, that does not, of itself, make that agent a dual agent.

2079.23 (a) A contract between the principal and agent may be modified or altered to change the agency relationship at any time before the performance of the act which is the object of the agency with the written consent of the parties to the agency relationship. (b) A lender or an auction company retained by a lender to control aspects of a transaction of real property subject to this part, including validating the sales price, shall not require, as a condition of receiving the lender's approval of the transaction, the homeowner or listing agent to defend or indemnify the lender or auction company from any liability alleged to result from the actions of the lender or auction company. Any clause, provision, covenant, or agreement purporting to impose an obligation to defend or indemnify a lender or an auction company in violation of this subdivision is against public policy, void, and unenforceable.

2079.24 Nothing in this article shall be construed to either diminish the duty of disclosure owed buyers and sellers by agents and their associate licensees, subagents, and employees or to relieve agents and their associate licensees, subagents, and employees from liability for their conduct in connection with acts governed by this article or for any breach of a fiduciary duty or a duty of disclosure.

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com
NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS		INITIALS
© 2019 AIR CRE.	All Rights Reserved.	



RENT ADJUSTMENT(S) (ORIGINAL TERM)

(ORIGINAL TERM) STANDARD LEASE ADDENDUM

	Dated:			
	By and Between			
	Lessor:			
	Lessee:			
	Property Address:			
	rioperty riddicasi	(street address, city, state, zip)		
_				
Paragrap				
The <i>appropri</i> a		al Term of the Lease shall be increased	d by using the method(s) selecte	ed below (check method(s) to be used and fill in
☐ I.	Consumer Price Index.			
	a. The monthly Base Rent shall be	e increased on and eve	y months thereaf	fter during the Original Term ("CPI Increase
Term sha defined).	II be multiplied by a fraction the deno	minator of which is the Base CPI (as hitute the new Base Rent until the nex	erein defined), and the numerat	Rent scheduled for the first month of the Original tor of which is the Comparison CPI (as herein vent shall any such new Base Rent be less than the
	b. The term "CPI" shall mean the	Consumer Price Index of the Bureau	of Labor Statistics of the U.S. De	epartment of Labor for (select one): CPI W
All Items	(1982-1984 = 100). The term "Compa	rison CPI" shall mean the CPI of the	calendar month which is 2 full m	or the area in which the Premises is located nonths prior to the applicable Original Term CPI mmencement Date of the Original Term.
matter sh	most nearly the same as the CPI shall	l be used to calculate the Base Rent in merican Arbitration Association in acc	creases hereunder. If the Partie cordance with the then rules of s	t, bureau or agency or is discontinued, then insteades cannot agree on such alternative index, then the said association and the decision of the arbitrators
□ II.	Fixed Percentage The monthly Pas	a Pont shall be increased on	and ayany ma	onths thereafter during the Original Term
				e paid for the month immediately preceding the
	e Percentage Increase Date.	percent (withly base here selectained to be	paid for the month infinediately preceding the
	-			
L III.	Fixed Rental Adjustment(s) ("FRA")			
The mon	thly Base Rent shall be increased to th	e following amounts on the dates set	forth below:	
	On (fill in FRA Adjustm	ent Date(s)):	The r	new Base Rent shall be:
		- < _ \		
		- ~		
		- / \ \		
		- \		
		- \ /		
				
	/	_		
		_		
	S FEE: For each adjustment in Base Ree, paragraph 9 of the Sublease.	ent specified above, the Brokers shall	be paid a Brokerage Fee in accor	rdance with paragraph 15 of the Lease or if
	X			
<		CRE * https://www.aircre.com * 2		
) `			

INITIALS



OPTION(S) TO EXTEND TERM STANDARD LEASE ADDENDUM

Dated:	
By and Between Lessor:	
Lessee:	
Property Address:	
(street address, city, state, zip)	
to extend the term of the Lease ("Extension Option(s)"), with each Extension O	conditions and provisions of Paragraph 39, Lessor grants Lessee option(s) ption being for a term of months, commencing when the prior term ust give written notice of such election to Lessor and Lessor must receive such notice at
and proper notification of the exercise of an Extension Option is not given by Le Except as specifically modified, the terms, conditions and provisions of the Leas	at the applicable Option Term would commence, time being of the essence. If timely essee and/or received by Lessor, such Extension Option shall automatically expire. See shall apply during Option Terms but the amount of Rent during Option Terms shall be to the following option Terms of the amount of Rent during Option Terms shall be to the following option Terms shall be to the following option Terms of the following option Terms shall be to the following option Terms of the following option Terms option Terms of the following option Terms of the following option Terms of the following option Terms option Terms of the following option Terms of the following option Terms of the following
established by using the method(s) selected below (check method(s) to be used	ana jiii in appropriately):
follows: the monthly Base Rent scheduled for the month immediately precedin denominator of which is the Option Term Base CPI (as herein defined), and the	nmensurate with the increase in the Option Term CPI (as herein defined) determined a ig the first occurring Option Term CPI Increase Date shall be multiplied by a fraction the numerator of which is the Option Term Comparison CPI (as herein defined). The erm CPI Increase Date during the applicable Option Term, but in no event shall any
(b) The term "Option Term CPI" shall mean the Consumer Price Index of	f the Bureau of Labor Statistics of the U.S. Department of Labor for <i>(select one)</i> :
	shall mean the CPI of the calendar month which is 2 full months prior to the applicable
Option Term CPI Increase Date. The term "Option Term Base CPI" shall mean the	ne CPI of the calendar month which is 2 full months prior to (select one):
index most nearly the same as the CPI shall be used to calculate the Base Rent i	governmental department, bureau or agency or is discontinued, then instead the increases hereunder. If the Parties cannot agree on such alternative index, then the in accordance with the then rules of said association and the decision of the arbitrators
	the monthly Base Rent shall be increased on and every
months thereafter during such Option Term(s) ("Option Term Pero Base Rent scheduled to be paid for the month immediately preceding the applied	
base nent scheduled to be paid for the month immediately preceding the applications of the paid for the month immediately preceding the applications of the paid for the month immediately preceding the applications of the paid for the month immediately preceding the applications of the paid for the month immediately preceding the applications of the paid for the month immediately preceding the applications of the paid for the month immediately preceding the applications of the paid for the month immediately preceding the applications of the paid for the month immediately preceding the applications of the paid for the month immediately preceding the applications of the paid for the month immediately preceding the paid for the	able option ferriff ercentage increase bate.
Premises during such Option Term established pursuant to the procedures, term however, regardless of such Fair Market Value, Base Rent during an Option Term Starting as of Lessee's exercise of the applicable Extension Option (but not earli thirty (30) days ("Negotiation Period") attempt to agree upon the Fair Market Value, then the Fair Market Value shall be established pursuant to the procedur (b) Each Party shall, within fifteen (15) days after the end of the Negotia Market Value ("Submitted Value(s)"). If a Party fails to timely provide a Submitt both Parties timely provide Submitted Values, then each Party shall, within fifte other Party of such Party's selected arbitrator who shall meet the qualifications Arbitrator who is favorable to such Party's position and may, prior to or after ap a Party fails to timely and properly provide notice of such Party's chosen Advoca (c) If both Parties timely and properly designate Advocate Arbitrators, to thoose a third (3rd) neutral arbitrator who shall meet the qualifications set fort Lessor and Lessee. If Advocate Arbitrators fail to agree upon and timely appoint Arbitrator within fifteen (15) days after request by either Party. If the President an appropriate legal action for a judge with competent jurisdiction over the Par (d) The Advocate Arbitrators and the Neutral Arbitrator ("Arbitrators)" in which the Premises is located, shall have been active over the five (5) year pewithin the general real estate market of the Premises. The Neutral Arbitrators and shall not have previously represented in a real estate transaction a Party or Arbitrators shall be decided by a majority vote of the Arbitrators, with each Arb hearings and require briefs, including market data and additional information. (e) Within thirty (30) days after selection of the Neutral Arbitrator; Market Value. The Aelotrators' Market Value ("Selected Market Value"), then provide the Parties a Market Value. The Selected Market Value shall be the Fair Market Value. The Arbitrators' ight to complete estab	ation Period, in writing submit to the other Party such Party's determination of the Fair ted Value, then the other Party's Submitted Value shall be the Fair Market Value. If the (15) days after both Parties have exchanged Submitted Values, in writing notify the set forth herein ("Advocate Arbitrator(s)"). Lessor and Lessee may select an Advocate pointment of an Advocate Arbitrator, consult with such Party's Advocate Arbitrator. If ate Arbitrator, then the other Party's Submitted Value shall be the Fair Market Value. Then such Advocate Arbitrators shall, within fifteen (15) days after their selection, herein ("Neutral Arbitrator"). The Neutral Arbitrator shall be engaged jointly by the a Neutral Arbitrator, then the President of AIR CRE shall appoint such Neutral of AIR CRE does not timely appoint the Neutral Arbitrator, then either Party may file ties to appoint the Neutral Arbitrator. Is shall be duly licensed real estate brokers or salespersons in good standing in the state wind before their appointment in the leasing of properties similar to the Premises and additionally not be related to or affiliated with either Party or Advocate Arbitrator, anyone related to or affiliated with a Party. All matters to be determined by the itrator having one (1) vote. The Arbitrators may, as the Arbitrators determine, hold Arbitrators shall first determine the Fair Market Value established by taking into accountalue"), then decide which Party's Submitted Value is closer in monetary amount to the copy of the Arbitrators' Market Value and finally notify the Parties of the Selected Arbitrators shall have no right to decide a Selected Market Value which is a compromise all be binding upon the Parties. The Party whose Submitted Value is not the Selected Market Value, pay the fees and costs of all three (3) Arbitrators. The applicable Option Term, then Lessee shall continue to pay to Lessor rent in the Term and Lessor's acceptance of such rent shall not waive, adversely affect or prejudices aright to collect the full amount of
	into account rent rates, rent abatements, periodic rent increases, real property taxes, ther applicable allowances, building services, length of lease term and other factors
INITIALS	INITIALS

© 2017 AIR CRE. All Rights Reserved.
OE-7.01, Revised 10-13-2022

Last Edited: 10/12/2022 4:58 PM

able parties for space of comparable location, size, age, condition, qual and customary manner for a reasonable length of time on the open ma Lessee and guarantors (if any) of this Lease (as of Lessee's exercise of the used for the Agreed Use (or other reasonably comparable uses). The A transaction any of the following: a sublease, lease assignment, lease relandlord; or a lease of space that was subject to a right of first refusal, reduce the Fair Market Value on accoursolely by Lessee (in excess of any applicable improvement allowance, all Lessee's improvement of the Premises), shall not reduce the Fair Market Value.	wity, parking, visibility, view, signage and accessibility if the Premises were marketed in a normal bracket to be leased to a tenant with financial strength and credit worthiness comparable to the Extension Option) for a term comparable to the length of the applicable Option Term and rbitrators, in deciding the Arbitrators' Market Value, shall not consider as a comparable enewal or extension; lease with a tenant that has equity, is related to or affiliated with the right of first offer, expansion option or other encumbrances. The Arbitrators, in deciding the at of Alterations and improvements made by Lessee to the extent the cost thereof was paid bated rent in lieu of improvement allowance or other consideration provided by Lessor for the Value on account of any real estate brokerage commission savings by Lessor, and shall not epair of the Premises for which Lessee was responsible under the Lease but did not perform.
IV. Fixed Rental Adjustment(s) ("FRA"). The monthly Base Rent shall be increased to the following amounts on On (fill in FRA Adjustment Date(s)):	the dates set forth below: The new Base Rent shall be:
<u> </u>	
	< / `
V. Continuation of Original Term Adjustments. The monthly Base Rent during the Option Term(s) which start(s) on used to calculate increases in the Base Rent during the Original Term of	
BROKER'S FEE : For each adjustment in Base Rent specified above, the applicable, paragraph 9 of the Sublease.	Brokers shall be paid a Brokerage Fee in accordance with paragraph 15 of the Lease or if
AIR CRE * https://www.air	rcre.com * 213-687-8777 * contracts@aircre.com by be reproduced in any form without permission in writing.
) Y
N Y	
V	
	
INITIALS	INITIALS

professional real estate brokers and/or appraisers customarily consider in determining fair market rent of property in an arm's length transaction by ready, willing and



NOTICE OF ELECTION TO EXTEND TERM OF LEASE

	Dated:	
	By and Between	
	Lessor:	
	Lessee:	
	Property Address:	
		(street address, city, state, zip)
To the Lessor	:	
Notice is here	eby given that Lessee irrevocably o	elects to exercise its option to extend the term of this Lease by months, ie. from to
Please ackno	wledge receint of this Notice in th	e space provided below and return a copy to Lessee via mail, telefax (Lessee's fax number is).
	wiedge receipt of this woulde in th	is space provided below and retained copy to reside and in many televisive straining in section and retained copy to reside and in the section and retained copy to reside and in the section and retained copy to reside and in the section and retained copy to reside and in the section and retained copy to reside and in the section and retained copy to reside and in the section and retained copy to reside and in the section and retained copy to reside and in the section and retained copy to reside and retained copy to retain a section and retained copy an
Lessee		
	-	
By:		
	d:	
Title:		
Phone:		
By:		
Name Printed	i:	
Title:		
Phone:		
A al al		< \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \
Address:		
Federal ID No	J	
Lessor hereby	acknowledges that it received th	is Notice on
Lessor		
Ву:		· · · · · · · · · · · · · · · · · · ·
Name Printed		
Title:		
Phone:		\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \
By:	1	
Name Printed		
Title: Phone:		
1 110HE.		() '
Federal ID No	o.:	

WARNING: IF THE OPTION PROVIDES FOR A MARKET RENTAL VALUE ADJUSTMENT, LESSEE MAY WISH TO INVESTIGATE THE ANTICIPATED FAIR RENTAL VALUE OF THE PREMISES PRIOR TO SENDING THIS NOTICE.

AIR CRE recommends that Lessee hand deliver this Notice to Lessor or, if that is not practical, that the Notice be sent by Certified Mail (return receipt requested), Express Mail (signature required); courier (signature required) or some other methodology that provides the Lessee with a receipt establishing the date that the Notice was received by Lessor.

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com
NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS

INITIALS
© 2017 AIR CRE. All Rights Reserved.

NTETL-1.02, Revised 10-22-2020



OPTION TO PURCHASE

	STANDARD LEASE ADDENDUM
Dated: By and Between Lessor: Lessee:	
Property Address:	(street address, city, state, zip)
Paragraph:	
(a) Lessor hereby grants to Lessee an option t	to purchase the Premises upon the terms and conditions herein set forth.
(the " Option Period "), time being exercised, Lessee must deliver to Lessor a cashio of the Standard Offer, Agreement and Escrow In	nstructions for the Purchase of Real Estate.
	those relating to Lessee's Default set forth in paragraph 39.4 of this Lease are conditions of this Option. burchase as provided above, the transfer of title to Lessee shall occur on the close of escrow and until that time the
(d) If Lessee elects to exercise this option to p terms of this Lease shall remain in full force and	
(e) If Lessee elects to exercise this option to p	ourchase, the purchase price to be paid by Lessee shall be
	se is exercised, Lessor and Lessee shall give instructions to consummate the sale to, located at in the normal and usual escrow forms then used by such escrow holder, as follows:
(ii) Lessor shall deposit the check referre escrow no later than 2:00 P.M. on the last busin (iii) The parties agree to execute any add	ditional instructions as are normal and usual; ions of sale shall be as set forth in the AIR CRE "STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS FOR THE
	t of the other, execute, acknowledge and deliver to the other a short form memorandum of this Lease for recording all be responsible for payment of any fees applicable thereto.
· ·	s not exercised by Lessee in a timely fashion, the Lessee shall, upon request of Lessor, execute, acknowledge and see's interest in such option. Lessor shall be responsible for the preparation of such deed and the payment of any fee
WARNING:	
LESSEE SHOULD NOT EXERCISE THIS OPTION UP FINANCING, AND IS OTHERWISE IN A POSITION	NTIL LESSEE HAS COMPLETED SUCH INVESTIGATION AS MAY BE APPROPRIATE, OBTAINED ANY NECESSARY N TO COMPLETE SUCH PURCHASE.
	CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com part of these works may be reproduced in any form without permission in writing.

INITIALS © 2017 AIR CRE. All Rights Reserved. OP-6.03, Revised 10-22-2020



NOTICE OF ELECTION TO PURCHASE

	Notice of Election to Conclude
Dated:	
By and Between	
Lessor:	
Lessee:	
Property Address:	
	(street address, city, state, zip)
To the Lessor:	
	exercise its option to purchase the Premises. In accordance with the terms of such option Lessee has enclosed herewith able to as and for the Deposit referred to in paragraph 4.1 of the Standard Offer, Agreement and Escrow cinafter the "Purchase Agreement").
Lessee has also executed and enclosed herevyour files.	with 3 copies of the Purchase Agreement. Please execute 2 of said copies and return them to Lessee. The third copy is for
	of the Purchase Agreement from you Lessee will forward one of the copies to the Escrow Holder, i.e., with the Escrow Holder will advise you when an escrow has been opened so that you can deposit Lessee's check with the ions of paragraph (f)(ii) of the option.
Please acknowledge receipt of this Notice in	the space provided below and return a copy to Lessee via umail, telefax (Lessee's fax number is).
Lessee	
Ву:	
Name Printed:	
Title:	
Ву:	
Name Printed:	
Title:	
Federal ID No.:	
Lessor hereby acknowledges that it received	this Notice, Lessee's check and the copies of the Purchase Agreement on
Lessor	
Ву:	
Name Printed:	
Title:	
Ву:	
Name Printed:	
Title:	
Federal ID No.:	
	V
WARNING: LESSEE SHOULD NOT SEND THIS	NOTICE UNTIL LESSEE HAS COMPLETED SUCH INVESTIGATION OF THE PREMISES AS MAY BE APPROPRIATE, OBTAINE WISE IN A POSITION TO COMPLETE SUCH PURCHASE.

AIR CRE recommends that Lessee hand deliver this Notice to Lessor or, if that is not practical, that the Notice be sent by Certified Mail (return receipt requested), Express Mail (signature required); courier (signature required) or some other methodology that provides the Lessee with a receipt establishing the date that the Notice was received by Lessor.

> AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS © 2017 AIR CRE. All Rights Reserved. NTETP-1.02, Revised 10-22-2020

INITIALS



Paragraph:

RIGHT OF FIRST REFUSAL TO PURCHASE STANDARD LEASE ADDENDUM

Date:		
By and Betwe	een	
Lessor:		_
Lessee:		_
Property Add	lress:	
		(street address, city, state, zip)

	(a)	Lessor shall not, at any time prior to the expiration of the term of this Lease, or any extension thereof, sell the Premises, or any interest thereir	n, withou
first g	giving	s written notice thereof to Lessee, which notice is hereinafter referred to as "Notice of Sale".	

- (b) The Notice of Sale shall include the exact and complete terms of the proposed sale and shall have attached thereto a copy of the bona fide offer and counteroffer, if any, duly executed by both Lessor and the prospective purchaser.
- (c) For a period of 12 calendar days after receipt by Lessee of the Notice of Sale, Lessee shall have the right to give written notice to Lessor of Lessee's exercise of Lessee's right to purchase the Premises, the interest therein proposed to be sold, or the property of which the Premises are a part, on the same terms, price and conditions as set forth in the Notice of Sale. In the event that Lessor does not receive written notice of Lessee's exercise of the right herein granted within said 12 day period, there shall be a conclusive presumption that Lessee has elected NOT to exercise Lessee's right hereunder, and Lessor may complete the sale to the prospective purchaser, on the same terms set forth in the Notice of Sale.
- (d) In the event that Lessee declines to exercise its right of first refusal after receipt of the Notice of Sale, and, thereafter, Lessor and the prospective purchaser modify by more than 5%, (i) the sales price, or (ii) the amount of down payment, or if there is a material change in any seller financing offered, or in the event that the sale is not consummated within 180 days of the date of the Notice of Sale, then Lessee's right of first refusal shall reapply to said transaction.
- (e) In the event that Lessee declines to exercise its right of first refusal after receipt of the Notice of Sale, and, thereafter, the proposed transfer or sale is not consummated, the Lessee's right of first refusal shall apply to any subsequent transaction. If, however, said transfer or sale is, in fact, completed, then said right shall be extinguished and shall not apply to any subsequent transactions.
- (f) Notwithstanding the above, this right of first refusal is intended to apply only to voluntary transfers involving third party transferees. This right of first refusal shall not, therefore, apply: where the Premises are taken by eminent domain or sold under threat of condemnation, to intra-family or intra-ownership transfers, to transfers by Lessor to a trust created by Lessor, or, if Lessor is a trust, to transfers to a trust beneficiary.
- (g) NOTE: This right of first refusal cannot be exercised: (i) during the period commencing with the giving of any notice of Default and continuing until said Default is cured, (ii) during the period of time any Rent is unpaid (without regard to whether notice thereof is given Lessee), (iii) during the time Lessee is in Breach of this Lease, or (iv) in the event that Lessee has been given 3 or more notices of Default, whether or not the Defaults are cured, during the 12 month period immediately preceding the exercise of the right of first refusal.

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS



RIGHT OF FIRST OFFER TO PURCHASE STANDARD LEASE ADDENDUM

Dated:	
By and Between Lessor: Lessee:	_
Property Address:	(street address, city, state, zip)
aragraph:	

a. Subject to the provisions of this paragraph, during the lease term and any extensions thereof (the "First Offer Period"), Lessee shall have a right of First Offer to purchase the Premises. Lessee's right of First Offer shall be deemed to be an "Option" as defined in paragraph 39 and subject to all of the provisions and limitations set forth in such paragraph.

b. If, at any time during the First Offer Period, Lessor decides, in its sole and absolute discretion, that it is interested in selling the Premises Lessor shall notify Lessee in writing of such interest (the "First Offer Notice"). Lessor is not, however, under any obligation to sell the Premises.

c. If Lessor should send a First Offer Notice to Lessee and Lessee wishes to exercise Lessee's right of First Offer with respect to the Premises, then within fifteen (15) days of delivery of the First Offer Notice to Lessee, Lessee shall deliver notice to Lessor of Lessee's exercise of its right of First Offer. Such notice shall be in the form of a binding offer to purchase the Premises and Lessee in making such offer shall use the then most current "Standard Offer, Agreement and Escrow Instructions for Purchase of Real Property" form published by AIR CRE (the "Offer").

d. If Lessee does not deliver to Lessor its Offer with respect to the Premises within the specified delivery period, time being of the essence, then Lessee's right of First Offer shall terminate.

e. If Lessee delivers the required Offer in a timely fashion then Lessor shall review said Offer and decide whether or not it wishes to sell the Premises on the terms contained therein. If Lessor in its sole and absolute discretion determines that the Offer is unacceptable then Lessor shall be precluded for a period of 6 or months from selling the Premises to a third party on terms which are materially less favorable then those offered by Lessee. If Lessor determines that the terms of Lessee's Offer are acceptable then the parties shall proceed with the sale in accordance with the terms of Lessee's Offer.

f. In addition to the provisions of paragraph 39 if this Lease or Lessee's right to possession of all or any portion of the Premises shall terminate in any manner whatsoever, then immediately upon such termination the Right of First Offer herein granted shall simultaneously terminate and become null and void and of no force or effect whatsoever. Time is of the essence with regard to Lessee's Right of First Offer.

g. Lessee's right of First Offer is intended to apply only to voluntary transfers involving third party transferees and shall not apply therefore: where the Premises or any portion of either is taken by eminent domain or sold under threat of condemnation, to transfers to an entity related to the Lessor, to intra-family or intra-ownership transfers, or to transfers by Lessor to a trust created by Lessor or if Lessor is a trust to transfers to a trust beneficiary.

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS

INITIALS
© 2017 AIR CRE. All Rights Reserved.

OTP-1.03, Revised 10-22-2020



RIGHT OF FIRST REFUSAL TO LEASE ADDITIONAL SPACE STANDARD LEASE ADDENDUM
Date: By and Between Lessor: Lessee: Property Address: (street address, city, state, zip)
aragraph:
a) Subject to the provisions of Paragraph 39 Lessee shall have a one-time right to lease (the "Right to Lease") the space described in subparagraph (b) if such space ecomes available during the term of this Lease.
b) This Right to Lease shall only apply to the following space in the Project: (the "Additional Space").
Prior to offering the Additional Space to anyone else Lessor shall give Lessee written notice of the availability of the Additional Space and the date the existing enant or occupant, if any, is expected to vacate such space. For a period of three business days following delivery of such notice, time being of the essence, Lessee hall have the right to request from Lessor a written statement setting forth the basic economic terms, including, but not limited to, Lessor's determination of the nonthly Base Rent, an improvement allowance, if any, and all other economic terms and conditions (collectively, the "Economic Terms"), upon which Lessor is willing to lease the Additional Space. Lessee must make such request in writing. If Lessor has not received such a request from Lessee within such time period Lessee shall be conclusively presumed to have elected not to lease the Additional Space and Lessee's Right to Lease the Additional Space will thereafter be null and void and of not curther force or effect. In establishing the Economic Terms the Lessor will make a good faith determination of the fair market rental rate for the Additional Space. The term "fair market ental rate" as used in this subparagraph shall mean the rental rate that a willing, comparable, renewal tenant (excluding sublease and assignment transactions) woul ay, and a willing owner of a comparable quality building located in the same vicinity would accept, for comparable space taking into account (i) the age, quality and ayout of the existing improvements in the Additional Space, and (ii) items that real estate brokers customarily consider, including, but not limited to, rental rates,
pace availability, length of lease term, Lessee size, Lessee improvement allowances, operating expenses and allowance, and any other economic matters then being harged by Lessor or owners of similar buildings. In no event, however, shall the monthly Base Rent payable for the Additional Space be less, on a square foot basis, hen the Base Rent payable with respect to the original Premises.
For a period of three business days after receipt of the Economic Terms from Lessor, time being of the essence, Lessee shall have the right to give written notice to Lessor of Lessee's exercise of its Right to Lease the Additional Space upon the Economic Terms and the same non-Economic Terms as set forth in this Lease with espect to the Premises. If Lessee timely exercises its Right to Lease as provided herein, the parties will promptly thereafter execute an amendment to this Lease to include the Additional Space in the Premises and to document the lease terms thereof. If Lessor has not received such notice from Lessee within such time period essee shall be conclusively presumed to have elected not to lease the Additional Space and Lessee's Right to Lease the Additional Space will thereafter be null and oid and of no further force or effect.
AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.
NOTICE. No part of these works may be reproduced in any form without permission in writing.

INITIALS



GUARANTY OF LEASE

WHEREAS	(" Lessee ") desires to lease from	(" Lessor ") the premises commonly known a	s (street address, city, state, zip)
("Premises") pursuant to a	a lease dated for reference purposes only	as of (" Lease ").	
•		of Lessor executing the Lease, Lessor requires	("Guarantor") to execute this Guaranty
of Lease (" Guaranty "); Gu	arantor desires Lessee to consummate the	e Lease. Accordingly, contemporaneous with execut	tion of the Lease, Guarantor is executing this
Guaranty to induce Lessor	to execute the Lease with Lessee.		

NOW THEREFORE, Guarantor hereby unconditionally and irrevocably guarantees and promises to perform and be liable for any and all obligations and liabilities of Lessee under the Lease, including, but not limited to, payment of all rents and all other sums payable by Lessee under the Lease and performance by Lessee of each and every one of the terms, conditions and covenants to be kept and performed by Lessee under the Lease.

Guarantor hereby agrees that, without the consent of, or notice to, Guarantor and without affecting or in any way releasing Guarantor's obligations under this Guaranty: (i) Lessor and Lessee may, by agreement or course of conduct, amend, extend, renew or otherwise alter any term, covenant or condition of the Lease, or the Lease may be assigned by Lessor or Lessee (and their successors and assigns) and this Guaranty shall guarantee all obligations of Lessee under the Lease as so amended, extended, renewed, altered or assigned; (ii) Lessor may release, change or add a party to the Lease and/or a guarantor of the Lease; (iii) Lessor may exercise, not exercise, delay exercising, impair, modify, limit, terminate or suspend any of Lessor's rights or remedies under the Lease or this Guaranty; and (iv) Lessor may permit all or any part of the Premises to be subleased, or the Lease to be assigned, assumed, transferred, mortgaged or encumbered.

Guarantor hereby agrees that no notice of nonperformance or default by Lessee or by another guarantor need be given to Guarantor. This Guaranty is a continuing and irrevocable guarantee. Guarantor waives the benefit of any law allowing Guarantor to revoke this Guaranty. No provision of this Guaranty or rights of Lessor hereunder may be waived, nor may any Guarantor be released from any obligation under this Guaranty except by a writing duly executed by Lessor.

Guarantor hereby waives and agrees not to assert or take advantage of any right or defense based on: (i) notice of acceptance of this Guaranty; (ii) demands (including demands for payment or performance), presentation and protest; (iii) any statute of limitations; (iv) requiring Lessor to proceed against Lessee, any Guarantor or other guarantor or any other person (as herein defined) liable to Lessor, or to proceed against these persons in any order; (v) requiring Lessor to apply any security deposit or other security; (vi) requiring Lessor to proceed under any other remedy Lessor may have before proceeding against Guarantor; (vii) any right of subrogation that Guarantor may have against Lessee or other guarantors; (viii) any right or defense that may arise by reason of the incapacity, lack of authority, death or disability of Lessee, any guarantor or any other person; (ix) any right or defense arising by reason of the absence, impairment, modification, limitation, destruction or cessation (in bankruptcy, by an election of remedies, or otherwise) of the liability of Lessee, of the subrogation rights of Guarantor, or of the rights of Guarantor to proceed against Lessee for reimbursement; (x) limiting Guarantor's obligations to not exceed Lessee's obligations; and (xi) notices (including notices of adverse change in the financial status of Lessee or other facts that increase the risk to Guarantor).

The obligations of Guarantor under this Guaranty shall not be altered, limited or affected by any proceeding, voluntary or involuntary, involving the bankruptcy, insolvency, receivership, reorganization or liquidation of Lessee, Guarantor, any other guarantor, or by any defense that Lessee, Guarantor or any other guarantor may have by reason of any order, decree, or decision of any court or administrative body resulting from any such proceeding.

Guarantor subordinates any and all existing or future indebtedness of Lessee to Guarantor to Lessee's obligations owed to Lessor under the Lease. Any recovery by Lessor from any other guarantor or insurer shall first be credited to the portion of Lessee's indebtedness to Lessor which exceeds the maximum liability of Guarantor under this Guaranty. If a person comprising Guarantor is married, such Guarantor expressly agrees that recourse may be had against such Guarantor's separate or community property.

Guarantor shall, upon ten (10) days' prior written notice from Lessor, provide Lessor with Guarantor's financial statements for the current year and tax returns for the three (3) preceding years and, if such is the normal practice of Guarantor, such statements shall be prepared in accordance with generally accepted accounting principles and audited by an independent certified public accountant. Lessor shall maintain Guarantor's financial statements and tax returns in confidence but shall be permitted to disclose Guarantor's financial statements and tax returns to prospective lenders, purchasers or others for a bona fide business purpose.

Guarantor shall, upon ten (10) days' prior written notice from Lessor, in writing reaffirm that this Guaranty remains and continues in full force and effect, that no event has occurred that would invalidate the Guaranty or excuse Guarantor's obligations or performance under the Guaranty, and that the Guaranty applies to the Lease, as amended, extended, renewed, altered or assigned. Guarantor shall, upon ten (10) days' prior written notice from Lessor, execute such other and further documents and do such further acts as may be reasonably necessary or required by Lessor to effectuate the intent of the parties and carry out the terms of this Guaranty.

Guarantor agrees that any suit, action or proceeding arising directly or indirectly from the Guaranty or the Lease shall be litigated only in courts located within the county and state in which the Premises is located. Guarantor irrevocably consents to the jurisdiction of any local, state or federal court located within the county and state in which the Premises is located, and waives and agrees not to assert by way of motion, defense or otherwise in any suit, action, or proceeding any claim that Guarantor is not personally subject to the jurisdiction of the above-named courts, that such suit, action, or proceeding is brought in an inconvenient forum, or that the venue of such action, suit, or proceeding is improper.

Each person which comprises Guarantor consents to service of process by a nationally recognized overnight courier service (such as Federal Express) at either the Premises or at the Guarantor's address for notices set forth below, and shall be deemed served and received three (3) business days after deposit with the overnight mail service. Guarantor irrevocably waives and agrees not to plead or claim in any action or proceeding that such service of process was in any way invalid or ineffective. Lessor reserves the right to serve process in any other manner permitted by law.

This Guaranty shall be governed by the laws of the state in which the Premises is located and, for purposes of conflicts of law, Guarantor shall be treated as if resident of or domiciled in such state. Guarantor waives the right to a jury trial of any cause of action, claim, counterclaim or cross-complaint in any action, proceeding or other hearing brought against Guarantor. If and to the extent the Lease requires Lessor and Lessee to arbitrate any disputes, then Guarantor also agrees to arbitrate such disputes in the same arbitration to the extent related to the Guaranty and upon the same terms and conditions provided in the Lease.

If Lessor brings an action or proceeding against Guarantor to enforce this Guaranty, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "Prevailing Party" shall include, without limitation, the party who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other party of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred.

This Guaranty constitutes the entire agreement between Guarantor and Lessor with respect to the subject matter of this Guaranty and supersedes all prior agreements, understandings, negotiations, representations and discussions, whether verbal or written, of the parties, pertaining to such subject matter. Guarantor is not relying on any representations, warranties or inducements from Lessor that are not expressly stated in this Guaranty.

If any provision of this Guaranty is determined to be illegal or unenforceable, all other provisions hereof shall nevertheless be effective. The waiver or failure to enforce any provision of this Guaranty shall not operate as a waiver of any other breach of such provision or any other provisions of this Guaranty, nor shall any single or partial exercise of any right, power or privilege preclude any other or further such exercise or the exercise of any other right, power or privilege. Time is strictly of the essence under this Guaranty and any amendment, modification or revision of this Guaranty.

Each person executing this Guaranty represents and warrants to Lessor that (i) each individual executing this Guaranty has the legal power, right, and actual authority to bind Guarantor; (ii) Guarantor has the legal right, power and authority to enter into and perform this Guaranty; (iii) all requisite action by or on behalf of a corporation, limited liability company, partnership (general or limited), trust or other legal entity ("entity") has been taken for this Guaranty to be duly authorized by such entity or Guarantor; (vi) no consent of any partner, shareholder, member, creditor, investor, government, judicial or administrative body or other party is required in connection with Guarantor executing this Guaranty; (v) this Guaranty is a valid and legally binding obligation of Guarantor; (vi) the execution and delivery of this

INITIALS	INITIALS
© 2022 AIR CRE. All Rights Reserved.	

Last Edited: 10/12/2022 3:04 PM

Guaranty does not conflict with or result in the material breach of any terms, conditions or provisions of, or constitute a default under, any bond, note, or other evidence of indebtedness or any contract, indenture, mortgage, deed of trust, loan, partnership agreement or other agreements or instruments to which Guarantor is a party or by which Guarantor is bound; and (vi) Guarantor will derive a substantial and material economic benefit from Lessor's execution of the Lease. If Guarantor is an entity, Guarantor shall, concurrently with the execution of this Guaranty (or upon ten (10) days' prior written notice from Lessor), deliver to Lessor a certified copy of a resolution or other documentation evidencing that such entity is authorized or ratifies the entity's execution of this Guaranty.

The term "person" as used in this Guaranty shall include an individual and/or an entity. If more than one (1) person comprises Guarantor, the obligations of such persons shall be joint and several. The unenforceability of this Guaranty or Lessor's election not to enforce this Guaranty against one (1) or more persons shall not affect the obligations of the remaining persons which comprise Guarantor or the enforceability of this Guaranty against such remaining persons.

Any notice, request, demand, instruction or other communication to be given to Guarantor under this Guaranty shall be in writing and shall be delivered at either the Premises or the address set forth beneath the Guarantor's signature below, or to such other place as Guarantor may from time to time in writing designate by at least fifteen (15) days' notice to Lessor.

When the context and construction so requires, all words used in the singular in this Guaranty shall be deemed to have been used in the plural. The term "Lessee" as used in this Guaranty shall mean the Lessee named in the Lease, any assignee of Lessee's interest under the Lease and their respective successors and assigns. The term "Lessor" as used in this Guaranty shall mean the Lessor named in the Lease, any assignee of Lessor, whether by outright assignment, assignment for security or otherwise. Lessor may without Guarantor's consent assign this Guaranty, voluntarily or by operation of law. This Guaranty applies to, inures to the benefit of, and binds all persons that comprise Guarantor, their heirs, devisees, legatees, executors, administrators, representatives, successors and assigns.

Signatures to this Guaranty accomplished by means of electronic signature or similar technology shall be legal and binding. This Guaranty may be executed in counterparts.

Guarantor is advised to seek advice of legal counsel before signing this Guaranty. Guarantor acknowledges that AIR CRE, the real estate brokers or their agents or employees have not made any representation or statement as to the legal sufficiency or effect or tax consequences of this Guaranty or the Lease.

GUARANTOR	Executed At: On:	
ву:	Ву:	
Name Printed:	Name Printed:	•
Title:	Title:	
Address:	Address:	
Email Address:	Email Address:	

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com
NOTICE: No part of these works may be reproduced in any form without permission in writing

Last Edited: 10/12/2022 3:04 PM



LESSOR'S RIGHT TO TERMINATE ON ASSIGNMENT OR SUBLEASE BY LESSEE

STANDARD LEASE ADDENDUM			
Date: By and Between Lessor: Lessee: Property Address:			
(street address, city, state, zip)			
Paragraph:			
(a) In the event of an assignment of this Lease or a sublease of more than forty percent (40%) of the area of the Premises for at least one year (including extension options) (each a "Recapture Transfer"), then Lessor shall have the right to terminate this Lease as provided herein ("Recapture Right") with respect to the portion of the Premises that is the subject of a Recapture Transfer ("Recapture Area"). Lessor may exercise the Recapture Right within ten (10) days after Lessor has received the terms, conditions and proposed effective date of a Recapture Transfer, and such other information, documents and sums requested by Lessor pursuant to Paragraph 12.2(e) of the Lease.			
(b) If Lessor exercises the Recapture Right, then as of the proposed effective date of a Recapture Transfer ("Recapture Date"), (i) this Lease shall terminate with respect to the Recapture Area as if such Recapture Date was the Expiration Date of this Lease with respect thereto; (ii) Lessee shall vacate and deliver possession of the Recapture Area to Lessor in the condition required by the Lease (including Paragraph 7.4(c) of the Lease); (iii) the Base Rent, other amounts payable by Lessee (including Lessee's Share of Common Area Operating Expenses [if applicable]) shall be reduced proportionately based on the reduction of the size of the Premises; (iv) if the Recapture Area is less than the entire Premises, Lessor shall, at Lessor's expense, separately demise the Recapture Area, Lessee shall provide Lessor access to Premises as reasonably necessary therefor, and Lessee shall, at Lessee's expense, modify the portion of the Premises that Lessee was obligated to at Lessee's expenses maintain or repair a part of the Building (such as roof) or a Building system (such as electrical, plumbing, mechanical, fire sprinklers, life safety systems, heating, ventilating and air conditioning) that will after the Recapture Date be shared by or serve the Recapture Area and other space, then after the Recapture Date Lessor shall be responsible for the maintenance and repair of such shared Building parts or systems and Lessee shall, as additional rent, pay to Lessor within ten (10) days after billing, Lessee's pro rata share of such maintenance and repair costs; (vi) if prior to the Recapture Date Lessee was obligated to pay for any expenses of the Premises (such as Real Property Taxes, insurance premiums or utilities) which after the Recapture Date utilities be shared between the Recapture Area and other space, then Lessor within ten (10) days after billing, Lessee's pro rata share of such shared expenses; (vii) Lessee shall abide by and conform to reasonable rules and regulations which Lessor makes from time to time for the ma			
(c) If Lessor exercises the Recapture Right, Lessor shall have the right, in Lessor's sole discretion, to lease the Recapture Area to any person or entity (including anyone with whom Lessee negotiated or who was otherwise interested in the Recapture Area) or to not lease the Recapture Area. If Lessor chooses to enter into a lease or other occupancy agreement (including with Lessee's proposed assignee or sublessee), then Lessee shall have no right to any of the rents or other consideration payable under such other lease or occupancy agreement, even if such rents and other consideration exceed the Rent payable under this Lease by Lessee.			
(d) Lessor's election not to exercise the Recapture Right shall not waive Lessor's other rights, including approval rights under Paragraph 12.1 of the Lease, rights to receive any Transfer Profit and/or rights to exercise the Recapture Right for any other Recapture Transfers. The Recapture Right shall continue throughout the entire term of the Lease, as extended.			
(e) Lessor shall not have the Recapture Right in the event of (a) an assignment of this Lease or a sublease of the entire Premises to a person or entity acquiring Lessee's business (whether through acquiring the assets of Lessee and/or a controlling interest in Lessee), (b) an assignment of this Lease due to Lessee's death, (c) a change in the control of Lessee which is due to transfer of stock, membership interest or partnership interest (i) among those who were the holders thereof as of the Commencement Date, (ii) among those who were the holders thereof as of the Commencement Date and their respective spouses, parents, siblings, children, grandchildren or any of their other family members, (iii) by will or devise, or (iv) to any person who was as of the Commencement Date an employee, officer or director of Lessee.			
AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.			

INITIALS



PROFIT SHARING ON ASSIGNMENT OR SUBLEASE STANDARD LEASE ADDENDUM

Date:	_
By and Between	
Lessor: Lessee:	
Property Address:	
Property Address.	(street address, city, state, zip)
Paragraph:	
(a) If Lessee assigns this Lease of the percent of the control of	or subleases any part of the Premises, then as a reasonable condition ro Lessor's consent, Lessee shall pay to Lessor (<i>fill in</i> he amount by which the Transfer Income exceeds the Transfer Expenses (" Transfer Profit "). Lessee shall pay the Transfer Profit after receipt by Lessee. Lessee shall pay the Transfer Profit in addition to Rent otherwise due from and payable by Lessee under
nn assignee or sublessee in connection	shall mean the aggregate of the following payable to Lessee, Lessee's related or affiliated persons or entities by or on behalf owith an assignment or sublease: Rent or additional rent; consideration of any kind, whether denominated as rent, key money, ent in excess of fair market value for past, present or future services; and any payment in excess of fair market value for sale,
dditional rent (both prorated on a per out-of-pocket expenses paid to third pa noving, real estate brokerage commiss onsideration shall be in a form reason be amortized (without interest) on a st	(s)") shall mean the aggregate of the following in connection with an assignment or sublease: Rent (excluding Transfer Profit) of square foot basis in case of a sublease of less than the entire Premises); and Lessee's actual, reasonable and substantiated arties not related to or affiliated with Lessee, assignee or sublessee, such as advertising, tenant improvements, space planning, sions and attorney's fees. If part of the Transfer Profit is payable other than in cash, the Transfer Profit of the noncash ably satisfactory to Lessor. In determining the Transfer Profit, the Transfer Expenses (other than Rent or additional rent) shall raight-line basis as follows: in case of an assignment, over the then remaining term of this Lease (not including any extension er the original term of the sublease (not including any extension options).
thief financial officer, describing in reasond complete copies of all agreements, expenses, and Lessor shall have the riging underpaid the Transfer Profit, then Lessesor's reasonable costs of such reviews.	days after Lessor's request, furnish a complete statement, certified by an independent certified public accountant or Lessee's conable detail the computation of any Transfer Profit that Lessee has derived or will derive. Lessee shall maintain true, correct, documents and instruments related to an assignment or sublease, books and records of the Transfer Income and Transfer ht to review, audit and copy the same to verify the Transfer Profit. If Lessor's review or audit of Transfer Profit indicates Lessee see shall, within thirty (30) days after demand, as additional rent pay such underpayment to Lessor together with Interest and w or audit; provided, however, if Lessee underpaid the Transfer Profit by more than five percent (5%), then, at Lessor's option, are Lessee in material and incurable Breach of this Lease.
NOT	AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com TICE: No part of these works may be reproduced in any form without permission in writing.



WORK LETTER

Dated:	
By and Between	
Lessor:	
Lessee:	
Property Address:	
(street address, city, state, zip)	
1. The Improvements. The Premises shall be modified with the following im of the quality which is standard for the Building, which quality shall be determine	nprovements or work: (the "Improvements"). The Improvements shall ned by Lessor ("Standard Improvement" - See paragraph 12 below).
specifications for the completion of the Improvements ("Preliminary Plans"). The stimate of the cost any work required of Lessor which is not a Standard Improvements of the cost any work required of Lessor which is not a Standard Improvement of the preliminary Plans and the preliminary cost estimate shall constitute Lessee's and the Preliminary Plans cannot in good faith be modified within 10 days after terminate upon notice thereof by one Party to the other, Lessor shall refund to Plans, legal fees and other costs incurred by Lessor in connection with this Leas	ually executed and delivered this Lease, Lessor shall prepare preliminary plans and he Preliminary Plans shall itemize the work to be done by each Party, including, an vement. Lessee shall approve the Preliminary Plans and the preliminary cost estimate ys following delivery thereof by Lessor. Lessee's failure to timely approve or disapprove as approval thereof. If Lessee shall disapprove all or any part of the Preliminary Plans, such disapproval to be acceptable to both Lessor and Lessee, then this Lease shall Lessee any Security Deposit or prepaid rent by Lessee, less the cost of the Preliminary e, and neither Party shall thereafter have any obligation, liability or responsibility to thinary Plans, when approved by Lessee, shall supersede any prior agreement of the
completion of the Improvements ("Final Plans"), including an estimate of the comprove the Final Plans and the final cost estimate or specify with particularity Lessor. Lessee's failure to timely approve or disapprove the Final Plans and the disapprove all or any part of the Final Plans, and the Final Plans cannot in good and Lessee, then this Lease shall terminate upon notice thereof by one Party to less the cost of the Preliminary Plans, Final Plans, legal fees and other costs income.	final cost estimate shall constitute Lessee's approval thereof. If Lessee shall faith be modified within 10 days after such disapproval to be acceptable to both Lesse the other, Lessor shall refund to Lessee any Security Deposit or prepaid rent by Lesse urred by Lessor in connection with this Lease, and neither Party shall thereafter have seever having to do with this Lease. The Final Plans, when approved by Lessee, shall
obligated to start construction of the Improvements or perform any acts in furt	exceeds Lessor's estimated cost of the Standard Improvements, before Lessor is herance thereof, Lessee shall pay to Lessor in cash a sum equal to such excess. If the e pays Lessor for such excess, then Lessor shall construct the Improvements. Lessee ts.
jurisdiction of the Premises shall have made a final inspection of the Improvem connection therewith and the Premises are in a broom-clean condition. If the Commencement Date, this Lease shall not be invalid, Lessor shall not be subjective.	t to any liability therefore in any respect whatsoever, Lessor shall use commercially faith, efforts and all due diligence to cooperate with the Lessor to complete all phases
delay in transportation, unavoidable casualties, rain or weather conditions, gov majeure"), or by Lessee's Breach or Default, by any acts or omissions of Lessee, ordered by Lessee ("Lessee Delay"), then the Commencement Date established Delay; provided, however, Lessor may elect to designate the Commencement D	rogress of the planning or construction of the Improvements by strikes, lockouts, fire, ernmental procedures or delay, or by any other cause beyond Lessor's control ("force Lessee's agents, employees, contractors, including extra work, changes in constructio in the Lease shall be extended by the period of such force majeure delay and Lessee Date as the date Completion of the Improvements would have occurred if such Lessee is, meet with Lessor or Lessor's representatives and provide sufficient information for it timely and property construction of the Improvements.
governmental body, authority or instrumentality does or may require any work safety or any other systems, and the cost of such extra work is estimated to exwritten notice of such election to Lessee. In the event that the Lessor elects to notice of termination to give written notice to Lessor of Lessee's commitment to perform such work within 5 days thereafter. , If Lessee does not make the requirement to the same of the sa	s Work Letter, if at any time before the actual Completion of the Improvements, any to be performed outside of the Premises, such as the Building Common Areas, fire life ceed then Lessor shall have the right to terminate this Lease by giving terminate this Lease, Lessee shall have the right within 5 days of the receipt of such so pay for such extra work. Lessee shall provide Lessor with the necessary funds to lired commitment or does not forward the necessary funds within the time limits Security Deposit or Rent prepaid by Lessee, and neither Party shall thereafter have any er having to do with this Lease.
	Lessor, the Parties shall execute an amendment to the Lease confirming the date of the Premises to Lessee, the Commencement Date and Expiration Date of this Lease.
9. Work Done by Lessee. Any work done by Lessee shall be performed in a lift required by Lessor, all work by Lessee shall be done only with union labor and mechanical, electrical wiring and ceiling work are to be done only by contractor	
10. Early Entry. If Lessor notifies Lessee of the estimated Completion, Lessee construction of any improvements Lessee is to construct and to equip and fixtuentry by Lessee into the Premises under this paragraph shall be under all of the	rize the Premises, as long as such entry does not interfere with Lessor's work. Any
any items of the Improvements that Lessee deems incomplete or incorrect in o	that Lessor delivered possession of the Premises to Lessee, notify Lessor in writing of order for the Completion of the Improvements in the Premises to occur and the I the Premises and approved construction of the Improvements if and to the extent
12. Standard Improvements. Lessor's Standard Improvements are as follows	:
·	
a. Partitions:	
b. Wall Surfaces:	
c. Window Coverings:	
INITIALS	INITIALS
	HALLIALS

INITIALS
© 2017 AIR CRE. All Rights Reserved.
WL-2.02, Revised 10-22-2020

d.	FLoor Coverings:
e.	Doors:
f.	Electrical and Telephone Outlets:
g.	Ceilings:
h.	Lighting:
i.	HVAC Ducting:
j.	Sound Proofing:
k.	Plumbing:

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com
NOTICE: No part of these works may be reproduced in any form without permission in writing.



WORK LETTER TO STANDARD OFFICE LEASE

Date	ed:	
Ву а	and between:	
The	Premises shall be constructed in accordance with Lessor's Standard Improvements, as follows:	
1.	Partitions	
2.	Wall Surfaces	
3.	Draperies	
4.	Carpeting	
5.	Doors	
6.	Electrical and Telephone Outlets	
7.	Ceiling	
8.	Lighting	
9.	Heating and Air Conditioning Ducts	
10.	Sound Proofing	
11.	Plumbing	
12.	Entrance Doors	
13.	Completion of Improvements Lessor shall construct and complete improvements to the Premises in accordance with the plans and specifications prepared by, dated, consisting of sheets (the "Improvements")	
of ar estir of tir mod othe	Within days after the date of this Lease Lessor shall prepare at its cost and deliver to Lessee for its approval copies of preliminary and specifications for the completion of the Premises, which plans and specifications shall itemize the work to be done by each party, including a cost estimate my work required of Lessor in excess of Lessor's Standard Improvements. Lessee shall approve said preliminary plans and specifications and preliminary cost mate or specify with particularity its objection thereto within days following receipt thereof. Failure to so approve or disapprove within said period ime shall constitute approval thereof. If Lessee shall reject said preliminary plans and specifications either partially or totally, and they cannot in good faith be diffied within 10 days after such rejection to be acceptable to Lessor and Lessee, this Lease shall terminate and neither party shall thereafter be obligated to the er party for any reason whatsoever having to do with this Lease, except that Lessee shall be refunded any security deposit or prepaid rent. The plans and cifications, when approved by Lessee, shall supersede any prior agreement concerning the Improvement.	I
15.	Construction If Lessor's cost of constructing the Improvements in the Premises exceeds the cost of Lessor's Standard Improvements, Lessee shall pay to Lessor in cash before commencement of such construction a sum equal to such excess. If the final plans and specifications are approved by Lessor and Lessee and Lessee pays Lessor for such excess, then Lessor shall, at its sole cost and expense, struct the Improvements in accordance with said approved final plans and specifications and all applicable rules, regulations, laws or ordinances.	
16.	Completion 16.1 Lessor shall obtain a building permit to construct the Improvements as soon as possible. 16.2 Lessor shall complete the construction of the Improvements as soon as reasonably possible after the obtaining of necessary building permits. 16.3 The term "Completion", as used in this Work Letter, is hereby defined to mean the date the building department of the municipality having jurisdiction of Premises shall have made a final inspection of the Improvements and authorized a final release of restrictions on the use of public utilities in connection therewit the same are in a broom-clean condition.	
	16.4 Lessor shall use its best efforts to achieve Completion of the Improvements on or before the Commencement Date set forth in the Lease or within 180 dayer Lessor obtains the building permit from the applicable building department, whichever is later.	
orde any 17.	16.5 In the event that the Improvements or any portion thereof have not reached Completion by the Commencement Date, this Lease shall not be invalid, but her Lessor shall complete the same as soon thereafter as is possible and Lessor shall not be liable to Lessee for damages in any respect whatsoever. 16.6 If Lessor shall be delayed at any time in the progress of the construction of the Improvements or any portion thereof by extra work, changes in construction ered by Lessee, or by strikes, lockouts, fire, delay in transportation, unavoidable casualties, rain or weather conditions, governmental procedures or delay, or by other cause beyond Lessor's control, then the Commencement Date established in the Lease shall be extended by the period of such delay. Term Upon Completion of the Improvements as defined in paragraph 16.3 above, Lessor and Lessee shall execute an amendment to the Lease setting forth the date t Lessor delivered possession of the Premises to Lessee as the Commencement Date of this Lease.	
regu only	Work Done by Lessee Any work done by Lessee shall be done only with Lessor's prior written consent and in conformity with a valid building permit and all applicable rules, ulations, laws and ordinances, and be done in a good and workmanlike manner of good and sufficient materials. All work shall be done only with union labor and y by contractors approved by Lessor, it being understood that all plumbing, mechanical, electrical wiring and ceiling work are to be done only by contractors ignated by Lessor.	
	Taking of Possession of Premises Lessor shall notify Lessee of the estimated Completion date at least 10 days before said date. Lessee shall thereafter have the right to enter the Premises to amence construction of any Improvements Lessee is to construct and to equip and fixturize the Premises, as long as such entry does not interfere with Lessor's	
INIT	TIALS INITIALS	

© 2017 AIR CRE. All Rights Reserved. OFGWL-2.01, Revised 10-22-2020 work. Any entry by Lessee of the Premises under this paragraph shall be under all of the terms and provisions of the Lease to which this Work Letter is attached.

20. Acceptance of Premises

Lessee shall notify Lessor in writing of any items that Lessee deems incomplete or incorrect in order for the Premises to be acceptable to Lessee within 10 days following the date that Lessor delivered possession of the Premises to Lessee. Lessee shall be deemed to have accepted the Premises and approved construction if Lessee does not deliver such a list to Lessor within said number of days.

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2017 AIR CRE. All Rights Reserved.
OFGWL-2.01, Revised 10-22-2020

INITIALS